Introduction to Sahidic Coptic

BY THOMAS O. LAMBDIN





Introduction to Sahidic Coptic

by Thomas O. Lambdin

Copyright 1983 by Mercer University Press Macon GA 31207

All rights reserved

Printed in the United States of America

C115 La

940116

All books published by Mercer University Press are produced on acid-free paper that exceeds the minimum standards set by the National Historical Publications and Records Commission.

Library of Congress Cataloging in Publication Data

Lambdin, Thomas Oden. Introduction to Sahidic Coptic.

Bibliography: p. 364 Includes indexes.

1. Coptic language—Grammar. I. title.

PJ2035.L3 1982

493'.282421

82-14282

94:0

The present work is an expansion of a series of elementary lessons developed gradually during twenty years of teaching Sahidic Coptic at the college level. The Lessons are designed to provide a carefully graded introduction to the basic grammar and vocabulary of the language. The content of the Lessons and the mode of presentation were dictated by purely practical pedagogical considerations; the book is in no way intended to be a scientific reference grammar. The Reading Selections are furnished with glosses designed to facilitate the transition to unsimplified material. A thorough mastery of these and the Lessons will bring the student to the level at which any Sahidic text of average difficulty can be read with no trouble. The emphasis on basic matters has necessitated the omission of much technical linguistic data not immediately relevant to the needs of the average beginning student. Those who are interested in a detailed study of the phonology, in the relationship of Sahidic to the other Coptic dialects, or in the historical development of Coptic from ancient Egyptian may consult the standard works on these subjects as cited in the Bibliography.

A special effort has been made to provide a Glossary that will be useful to the student beyond his first year's study. In addition to covering the words used in the present text, the Glossary is intended to contain the full vocabulary of the Sahidic New Testament, including most associated phrases and idioms, as well as a generous selection of lexical items from other Biblical and literary texts. Deliberately excluded from the Glossary are words of a specialized nature, such as the names of plants, vessels, implements, drugs, and animals occurring only in technical texts that usually provide little clue to their precise meanings; nor has any effort been made to include the unusual lexical usage of Shenute. For these items the reader must consult the indispensible A Coptic Dictionary of W. E. Crum, which, together with M. Wilmet, Concordance du nouveau

 $\it testament\ sahidique$, is the main authority for the Glossary included here.

I would like to express my sincere thanks to my colleague, George W. MacRae, the Charles Chauncey Stillman Professor of Roman Catholic Theological Studies, Harvard Divinity School, for encouraging me to undertake this work and for his helpful comments on a large portion of the manuscript; to Mr. Gary A. Bisbee, for the exceptional skill and care with which he prepared the final copy for publication; to Mr. Watson E. Mills, Director of the Mercer University Press, for his part in initiating and publishing this work.

Thomas O. Lambdin

Cambridge, Mass.
June 1982

Table of Contents

	Pre	face			•	•	•	•	•	•	•	•	•	•	•	•	•	•	•	•	•	•	iii
	Abb	revia	tic	ns	ar	nd	Co	n	er	ıti	ior	ıs		•					•			•	νi
	Int	roduc	tic	n			•				•	•		•		•	•			•		•	vii
-	Les	sons	1-3	30		•		•				•	•	•								•	1
	Rea	ding	Sel	lect	ic	ns	;																
		Intro	duc	tor	у	Re	ma	ırk	cs				•	•						•		•	146
		Luke	I-V			•	•		•	•		•		•	•	•	•	•			•	•	149
		Apoph	the	gma	ıta	ı F	at	ru	ım			•		•	•	•	•	•					171
	1	Wisdo	m c	of S	o1	.on	or	ì	•		•	•		•	•	•		•			•	•	185
	•	The L	ife	e of	J	os	ep	h	th	ıe	Ca	rp	en	tε	er	•	•	•	•			•	196
÷	Glo:	ssary	•						•	•	•					•	•	•	•		•	•	209
	G1o	ssary	of	Gr	ee	k	Wo	rd	ls	•	•		•		•	•	•	•		•	•	•	359
	Bib	liogr	aph	ıy						•			•					•	•				364
	Gra	mmati	ca1	. In	de	x		•	•	•	•										•		366
	Tab	le of	Pr	inc	ip	a1	v	er	ba	1	Со	nj	ug	at	io	ns	;				•		371
	Sub	ject	Ind	lex																			373

Abbreviations and Conventions

. 11	adjective, adjectival	obj.	object
adj.		•	•
adv.	adverb, adverbial	oft.	often
aft.	after	p.c.	participium conjunc-
art.	article		tivum
bef.	before	part.	particle
Boh.	Bohairic	Perf. I	the First Perfect
c.p1.	common plural	pers.	person
caus.	causative	phr.	phrase
cf.	compare	p1.	plural
Circum.	the Circumstantial	pred.	predication, predicate
conj.	conjunction	prep.	preposition
Conj.	the Conjunctive	Pres. I	the First Present
coord.	coordinated, coordinating	prob.	probably
cpd.	compound, compounded	procl.	proclitic
dat.	dative	pron.	pronoun, pronominal
def.	definite	Q.	qualitative
e.g.	for example	q.v.	which see
eth.	ethical	recipr.	reciprocal
exclam.	exclamatory	reflex.	reflexive
f., fem.	feminine	Rel.	Relative Form
fig.	figuratively	s.	singular
foll.	following	s.v.	sub voce
Fut. I	the First Future	Sah.	Sahidic
Fut. II	the Second Future	sim.	similar(ly)
Fut. III	the Third Future	sing.	singular
Gk.	Greek	sthg.	something
Gr. In.	Grammatical Index (Coptic)	sub	under
Hab.	the Habitual	subj.	subject
i.e.	that is	suff.	suffix(ed)
idem	having the same meanings	tr.	transitive
	as the immediately pre-	usu.	usually
	ceding word	vb.	verb, verbal
imperf.	imperfect	Vocab.	Vocabulary
Imperf.	the Imperfect	W.	with
imptv.	imperative		with
indef.	indefinite		
indep.	independent		
Indep.	Infinitive	±	with or without
	Inflected Infinitive	+	rith plus and
intens.	intensive	т	with, plus, and
_		=	is fully equivalent in
interrog.	interrogative intransitive		f
Intro.	Introduction		function and meaning to
lit.	literally	r	
m., masc.	masculine	ine names	of specific conjuga-
n.	noun, nominal		inflected verbal forms
neg.	negative		alized throughout the
no.	number	book.	

Introduction

The political unification of Egypt took place around the beginning of the third millennium B.C. with the establishment of the First Dynasty at Memphis. Soon afterward written records began to appear in the hieroglyphic script, which together with its cursive derivatives. hieratic and demotic, remained the sole medium for writing the Egyp-At that time. tian language until the end of the second century A. D. the missionaries of the Church, then centered in Alexandria, undertook the translation of the Bible from Greek into Egyptian in order to facilitate their task of Christianizing the country. They abandoned the three-thousand-year-old hieroglyphic writing system, probably as much because of its complexity and imperfections as for its "heathen" associations, and chose instead to employ a modified form of the Greek alphabet. Egyptian in this new guise is known as Coptic, a modern term derived from Arabic $qubt\hat{i}$, itself a corruption of the Greek word (ai) qúpti (os). Egyptian.

The conquest of Egypt by Alexander the Great in 332 B.C. and the subsequent Greek-speaking administration of the country under the Ptolemies led to the thorough Hellenization of Lower (i.e. Northern) Egypt. Egyptian-Greek bilingualism was apparently commonplace in the Delta, and it is probable that much Greek technical, legal, and commercial terminology was introduced into spoken Egyptian at this time. Rough and unsystematic attempts to transcribe Egyptian in the Greek alphabet were made as early as the third century B.C. It was only natural, then, that the Coptic translators of the Bible not only adopted the Greek alphabet but also generously supplemented the native lexicon with many more borrowings from Greek. The Greek vocabulary of any Coptic text is significantly large.

Evidence of dialectal differences is found as early as the third millennium B.C., but the general conservativism of the hieroglyphic script and the practice of standardizing a particular form of the language for long periods of time (e.g. Middle Egyptian, New Egyptian) tend to obscure the great dialectal diversity that must have existed

in the spoken language as one traveled the 750 miles down the Nile from Aswan to the Mediterranean. The individual dialects first become recognizable when we reach the Coptic period and see the language spelled out in the Greek alphabet. The exact geographical location of the dialects is still a matter of scholarly debate, but the reader should become familiar with their names and the approximate chronological range of their use for literary purposes.

Sahidic, the dialect treated in this book, was the dialect chosen for the official translation of the Bible mentioned above. There is conflicting evidence on its geographical location; the name Sahidic. from Arabic $as-sa^{c}id$, Upper (i.e. Southern) Egypt, places it in the south (hence its alternate name: Theban, Thebaic); linguistic considerations, however, favor a northern locale, in the neighborhood of Memphis and the eastern Delta. One cannot rule out the possibility that both locations are correct; the fact that Thebes and Memphis alternated as the capital of Egypt through much of its history and were the chief centers of religious (priestly), building, and commercial activity could have led to the development of an "urban" dialect in these two areas, quite distinct from the dialects of the "rural" areas that lay between. By the fourth century A.D. Sahidic was firmly established as the standard literary dialect and retained this status until its demise around the tenth century. Surviving texts in Sahidic include, in addition to the New Testament and a large portion of the Old, a considerable corpus of Church literature and some remnants of secular literature, nearly all of which is translated from Greek. Of native works we have only the writings of Pachomius (c. 300). the founder of Egyptian monasticism; Shenute (c. 400), the administrator of the White Monastery in Upper Egypt; and Besa, a disciple of Shenute. The Coptic writings of Shenute, who attempted to mould the language into a literary vehicle comparable to Greek, are often referred to as the "classics" of Sahidic literature. Their syntactic complexity and unusual vocabulary usage, however, place them beyond the scope of the present work, which is based on the language of the more widely studied translation literature.

Bohairic replaced Sahidic as the standard literary dialect.

Bohairic texts are attested as early as the ninth century, but the dialect does not seem to have achieved wide usage until it was adopted as the official language of the Coptic Church in the eleventh century. Most Bohairic texts come from after this time, and many of them were translated from Sahidic originals. The term Bohairic comes from Arabic al-buhairah, Lower (i.e. Northern) Egypt; it is generally assumed that Bohairic was the dialect of the Western Delta, including Alexandria and Nitria. The designation Memphitic has also been used for this dialect.

Fayyumic, as its name implies, was the dialect of northern Middle Egypt in the vicinity of the Fayyum Basin. It is well attested in texts ranging from the fourth to the eleventh century, but it apparently never attained the status of Sahidic.

Achmimic, generally located in the area of Akhmim (Panopolis) in southern Middle Egypt, enjoyed only a brief literary period from the third to the fifth century.

Subachmimic, tentatively localized between Akhmim and Thebes, was used extensively in the fourth and fifth centuries for the translation of Manichaean and Gnostic literature. Its association with this heretical material probably had much to do with its early demise as a literary dialect. The Nag Hammadi texts are in Subachmimic or a variety of Sahidic influenced by Subachmimic in varying degrees.

For further details on the dialects the reader should consult the works of Worrell, Vergote, Kahle, and Till cited in the Bibliography.

The Arab conquest of Egypt in 641 A.D. and the subsequent suppression of the native Christian population resulted in the gradual dying out of the Egyptian language in favor of Arabic. We cannot be sure how long this process took, but it is safe to assume that by the fifteenth century Coptic had ceased to be a native spoken language, thus bringing to an end a continuous written record of over four thousand years.

The Coptic Alphabet

Sahidic Coptic is written in the Greek alphabet augmented by six letters borrowed from Demotic script, the last stage of Egyptian hieroglyphic writing. The letters of the full alphabet, together with their conventional transcription, are as follows:

λ	a	н	ē	N	n	T	t	ற	š
В	Ъ	θ	th	3	ks	Y	u	4	f
Г	g	1	i	o	О	ф	ph	2	h
A	d	ĸ	k	n	p	×	kh	x	j, ğ
e	е	λ	1	P	r	ψ	ps	6	č, c
z	Z	M	m	С	s	w	ō	†	ti

The following paragraphs deal with the Coptic, not the Greek, pronunciation of this alphabet.

Spelling and Pronunciation

a. The Consonants

B was apparently pronounced like English v in voice, but it is generally read simply as b in back.

r occurs only as a positional variant of κ in a very small set of forms. Pronounced like the g of good.

 Δ and z do not normally occur in standard Sahidic spelling. z may occur for c in a few words, e.g. ANZHBE for ANCHBE school. Pronounced d as in dog and z as in zoo respectively.

 ϕ , Θ , and x occur in Sahidic Coptic words only as combinations of two consonants: n+2, $\tau+2$, and $\kappa+2$ respectively. Θ is fairly frequent, e.g. $\pi\Theta\Theta\Theta\Theta\Theta$ evil, for $\pi\Theta$ 2009. Φ and Φ are rare and need not be used at all. The Copts seem to have used this same pronunciation for these letters in Greek words, contrary to the ordinary Greek pronunciation of Φ as Φ as Φ as Φ (German

ich, ach).

 κ , κ , and τ were like English k, p, t, but without aspiration. Thus, they were more like the k, p, t of skin, spin, stop than the aspirated sounds of kin, pin, top.

 λ , M, and N were probably the same as English l, m, and n.

- % is simply a combination of κ + c, rarely used. E.g. soyp ring.
- ${\tt P}$ is conventionally pronounced like English r in road. Its actual pronunciation is unknown.
 - c was like English s in see.
 - Ψ is simply n + c, rarely used. E.g. Ψιτε nine (psite).
 - m was the sh of shall.
 - 4 was the f of foot.
 - 2 was probably like English h in hope.
- ${\bf x}$ is conventionally pronounced like the j of judge. Its actual pronunciation was probably closer to that of the $[{\bf t}^y]$ of tune.
- 6, conventionally like the ch of church, was probably closer to the $[k^y]$ of cue, cute.
- + is merely a graphic symbol for $\tau + \iota$, but it was the normal way to spell this sequence of sounds. E.g. +Me village (time).

b. The simple vowels

- a like the a of father. E.g. a4 [af] meat.
- ϵ like the e of let. E.g. $2\epsilon N$ [hen] some.
- н probably like the a of hate. E.g. мнт [met] ten.
- I like the i of machine. This vowel is always spelled ϵ_1 in initial positions: $\epsilon_1 N \epsilon$ [in ϵ] to bring, $\epsilon_1 c$ [is] behold. Internally and finally the spelling alternates between ϵ_1 and ϵ_1 , but ϵ_1 is preferred.

¹ Brackets are used to indicate phonetic pronunciation in standard phonetic symbols. Do not confuse these with the conventional transcriptions.

xii

o like the o of log, fog, dog, off, on. E.g. ton [top] edge.

y does not appear as a simple vowel in Coptic words.
oy is the normal writing of the vowel [u], the oo of food. E.g. Noys noub [nub] gold.

w like the o of hope. E.g. 2wn [hop] to hide.

c. Semivowels and diphthongs

The consonants y and w of English yet and wet are often referred to as semivowels because they are the same sounds as the vowels [i] and [u] of beet and boot very briefly articulated. The Coptic vowels ϵ_1 (i) and ϵ_2 and ϵ_3 function as consonants in the same way. E.g. $\epsilon_1 \omega_1$ [yot] father, ϵ_1 or ϵ_2 to become pure.

The semivowels ϵ_1 (1) and oy combine with a preceding simple vowel to form various diphthongs. Many of the diphthongs have more than one spelling; the reader should follow the spelling used in the Lessons. The diphthongs should be pronounced carefully, with the value of the single vowel as given above plus a final y or w as the case may be.

λι, λει as in cλειν [sayn] physician, λιπωτ [aypót] I ran.

 $\lambda\gamma$ (rarely $\lambda\gamma$) as in $\lambda\gamma$ [naw] to see, $\lambda\gamma\eta\sigma\tau$ [awpot they ran.

e: (less commonly ee;) as in nerpone [peyrome] this man. Although a knowledge of the grammar is necessary fo making the correct distinction between e; = [i] and e; = [ey], the problem is not a serious one: in normal Sahidic spelling e; has the value e + ; (1) in the demonstrative adjectives ne; - re; - ne; - (Lesson 5), (2) in the first person verbal prefixes of the forms e; -, ne; -, me; - (Lesson 21 and following), and in a few isolated words like e; e [eye] (Lesson 29).

las of tra

men

1 1

The Gree

but

smal:

spel] for ,

respe binat

tivel 2007. Copts lette nunci:

 ϵ_Y (rarely ϵ_{OY}), as in ϵ_{YD} as $[\epsilon_W \S \hat{\alpha} t^y \epsilon]$ while they were talking.

HI as in HHI [pey] the house.

HY (less commonly Hoy) as in THY [tew] wind.

161, 6161 is very rare and is [yi] not [iy], e.g.
21618 [hyib] lamb.

ioy is rare, e.g. cloy [siw] star.

oer, or as in oyoern [woyn] light.

ooy as in mooy [mow] water, mooyr [mowt] dead.

 ω_1 as in $\varepsilon x \omega_1$ [$\varepsilon t^y \acute{o} y$] on me; rare except in final position.

ωογ as in τωογν [town] to stand up, exωογ [ϵt^y ów] on them.

ογι (rare) as in Νογι [nuy] mine; also possibly as [wi] in some words, e.g. κογι [kwi] small.

ογογ (rare) as in Μογογτ [muwt] to kill, Νογογ [nuw] theirs.

Double Vowels

The double writing of any of the simple vowels is generally understood to be an indication of the presence of a glottal stop, i.e. the complete but very brief stoppage of airflow in the glottis, conventionally indicated by 'in transcription. Thus make $m\hat{a}$ ' δb thirty, ceene $s\hat{e}$ ' δpe remainder, goon $\delta \hat{o}$ ' δp to be. The stress is on the first vowel; the vowel after the glottal stop was probably of very brief duration.

Vowel doubling occurs in diphthongs as well, e.g. many mā'āw mother, meeye mē'ĕwe to think. There is no sure way of knowing whether ooy indicates [ow] or [5°5w].

Syllabification and the Supralinear Stroke

One of the most distinctive features of Sahidic spelling is the short stroke placed over certain consonants or

groups of consonants. This supralinear stroke, as it is called, indicates a syllable, but there is some disagreement among Coptic scholars on how this syllabification actually sounded in the spoken language. When the stroke is used over a voiced consonant such as w, it probably meant that the consonant is functioning as the vowel, i.e. the most sonorous part, of the syllable in question, exactly like the final n of English button and sudden, phonetically [-tn] and [-dn]. Thus, g_{NT} (to seek me) was pronounced [\S_n t] and NT (to bring me) as [nt]. The voiced consonants capable of having this syllabic pronunciation are \mathfrak{s} , λ , \mathfrak{m} , \mathfrak{m} , and \mathfrak{p} , known mnemonically as the blemner consonants. Note that they are all voiced continuants, i.e. consonants whose voiced duration may be prolonged at will (remember that \mathfrak{s} is \mathfrak{v} , not \mathfrak{b}). E.g.

тисωти (we hear) [tnsotm] тит (fish) [tvt] ωτορτη (to disturb) [štortr] крири (to mutter) [krmin]

The stroke over the remaining consonants may be pronounced as a brief e or as a (the first vowel of English above) before the consonant over which the stroke is placed, e.g. $c \pi c \omega n \overline{q}$ [səpsopəf] to entreat him. This pronunciation may also be used with the blemner consonants for the sake of convenience.

In non-standard texts, of which there are many, the vowel e is often written instead of using the stroke (and vice versa), but most frequently in proclitic elements and initial clusters, e.g. cencon $\overline{q} = c\overline{n}con\overline{q}$, 26M nH = 2 \overline{M} nH 1. In standard spelling 6 is used regularly instead of the stroke only when the consonant preceding the consonant that would have had the stroke is a blemner; thus mokmek and nognes are words of the same pattern as conc \overline{n} and coxc \overline{x} . This convention may have been adopted to prevent incorrect syllabification: mokmek could be read as [$m \circ kmk$] or [$m \circ kmk$]. The chief exceptions are indeed words where a different

syllabication is required: wonnt [somnt] three, tonnt [tomnt] to befall. The n of these words is an intrusive (secondary) glide from the labial m to the dental t; the earlier forms were wont and tont. The convention likewise does not apply when the final consonant is also a blemner:

NA2MN [nahmn] to rescue us. Much of the variation between e and a stroke that occurs in the writing of certain verbal prefixes (e.g. NTEPT-, NTEPET-; MAPT-, MAPET-) probably results from inconsistent application of this rule.

The forms $\tau\omega\sigma\gamma N$ (to arise) and $coo\gamma N$ (to know) have been standardized in the Lessons. In the Reading Selections the orthography of the source has been followed.

Stress

Coptic is a highly compounding language, mostly by prefixation. All prefixal elements are proclitic, i.e. unstressed and bound, to the word which stands last in the sequence, regardless of its length, e.g.

 $2\overline{N}$ TEMMNTATTAKO = $2\overline{N}$ -TE-M-MT-AT-TAKÓ in his imperishability

Any element designated as prefixal in the course of the Lessons should be considered as proclitic. All simple prepositions are proclitic, like ${}_{2}\overline{\scriptscriptstyle N}$ in the above example, but for the sake of clarity they are written as separate words in this text.

The main stress, then, is on the word standing at the end of the compound. The successive application of the following rules will enable the reader to apply the correct stress in all but the rarest cases:

- (1) Stress is always on one of the last two syllables of a word.
 - (2) The vowels \mathbf{H}_{\bullet} o, and \mathbf{w} are always stressed.
- (3) Final simple $-\lambda$ and simple $-\epsilon_1$, $-\epsilon_1$ are always stressed.

- (4) Final -oγ is stressed except (1) when it is the suffixed pronoun of the 3rd person plural (a knowledge of the grammar will make this clear), and (2) in the words πλ2ογ (back), cποτογ (lips), cλ2ογ (curse), and ρλcογ (dream).
- (5) Final -6 is unstressed except in the adjectives introduced in Lesson 15 (thus, casé, wise, sāλē, blind, etc.) and in a few miscellaneous words like seké (wages), mntpé (witness), knté (figs), and namé (truly).
- (6) A final syllable marked by a stroked consonant is never stressed unless it is the only syllable of the word.

Assimilation

Assimilation, for our present purposes, may be defined briefly as the alteration of a sound due to its proximity to another sound, usually resulting in greater phonetic compatibility. The final \overline{N} of prefixal elements (e.g. prepositions, particles, articles) is regularly assimilated to \overline{N} before \overline{N} and \overline{N} , e.g.

*2 \overline{N} nH1 \rightarrow 2 \overline{N} nH1 in the house * \overline{N} MAGIN \rightarrow \overline{M} MAGIN the signs.

The assimilation of consonant -N also occurs but is not standard, e.g. Temnictic for Tennictic (our faith). In some texts the particle \overline{N} , which has several grammatical functions, assimilates completely to B, λ , and P, e.g. $\overline{N}B\overline{P}PE \rightarrow \overline{B}B\overline{P}PE$ (the young), $\overline{N}POME \rightarrow \overline{P}POME$ (the men). This is not considered standard, but it is not uncommon; numerous examples will be met in our reading selection from the Wisdom of Solomon.

Whatever the pronunciation of the supralinear stroke was, an alternate spelling with $-\lambda$ - often occurs before final -2: $\omega N \lambda_2 = \omega N \overline{z}$ to live. This represents an assimilation to the guttural quality of z.

An alternation between $-\omega$ - and $-o\gamma$ - in certain word

patterns is a result of an assimilation in the pre-Coptic stage. ω was altered to oy after M and N; thus, words like Moy2, Noyxe, Moyoyt, and Noyk originally had the same vowel as $\kappa\omega\tau$, $\kappa\omega\tau$ e, $\tau\omega$ oyN, and $\tau\omega$ k respectively.

The Spelling of Greek Words

Greek words in Coptic are usually spelled correctly. Certain types of errors do occasionally occur, however, resulting in part from the discrepancy between the classical spelling and the contemporary pronunciation, and they must be taken into account when consulting a standard Greek dictionary. The most frequent of these are confusions between (1) H and Y; (2) G and A1; (3) 1 and H; (4) o and ω ; (5) r and κ ; (6) 1 and G1; (7) H and G; (8) T and A; (9) initial 2 and zero. All of these are illustrated by the following words chosen from our Reading Selections.

εγκα = εκκα (βῆμα)

απγληση = απκαίοη (σπήλαιον)

περιχορος = περιχωρος (περίχωρος)

ορκανοη = οργανοη (δργανον)

περειερκάζε = περιεργάζε (περιεργάζε -)

† απαχε = αιστάζε (διστάζε -)

εθρίοη = αιθρίοη (αίθριον)

κγρισσα = κηράσσε (κηράσσε -)

εγάωνη = εμάνη (ήδονή)

πιθε = πείθε (πείθε -)

εσγάζε = εκτίχαχε (ἡσυχάζε -)

εελπίζε = ελπίχε (έλπίζε -)



1.1 Gender. There are two grammatical genders in Coptic: masculine and feminine. Nouns denoting male beings are usually masculine; those denoting females, feminine. The gender of other nouns cannot, in general, be deduced either from their form or meaning and must be learned for each noun. Examples:

1	masculine	feminine			
ElWT	father	МХХҮ	mother		
K A 2	earth, ground	ne	sky, heaven		
\$00Y	day	оуфн	night		

There are some pairs of nouns where a formal relationship exists between the masculine and the feminine form:

m	asculine	feminine				
CON	brother	cone sister				
о н р є	boy, son	феере girl, daughter				
2 \$\lambda \lambda 0	old man	2χλω old woman				
ОУ2ОР	dog (male)	ογεωρε dog (female)				

These will be noted in the lesson vocabularies. The derivational process involved is no longer a productive one in Coptic: such pairs cannot be formed at will.

1.2 Number: singular and plural. Only a relatively small number of nouns have preserved a distinct plural form. For example:

:	singular	plural				
610T	father	610T6	fathers			
CON	brother	CNHY	brothers			
X 01	ship	6ХНУ	ships			

The plural is otherwise made explicit by the form of the article (see below), the noun itself remaining unchanged.

Those plurals that are in common use will be given in the lesson vocabularies along with the singular. They should be learned as they occur, since there is no consistent pattern for their formation.

1.3 The definite article. The definite article has the forms

masc. sing. и, ие common plural N, не fem. sing. т. те

These are attached directly to the noun, as in

pome man пpome the man Mpome the men eix hand теіх the hand Neix the hands

The plural article appears as M before n and м (cf. Intro., p. xvi):

пе sky тпе the sky мпнуе the heavens млеги sign пмлеги the sign ммлеги the signs

Before nouns beginning with a vowel the plural article appears as either N or N:

EXHY Ships NEXHY OF NEXHY the ships

Before initial stroked consonants there are several possibilities:

MTON, 6MTON repose пентон, пМТОН, пЕМТОН the repose NKA, 6NKA thing N6NKA, NNKA, NNKA the things

The fuller forms n6-, т6-, N6- are used regularly before nouns beginning with two consonants:

κλομ crown nekλom the crown Nekλom the crowns c2 ime woman Tec2 ime the woman Ne2 ione the women

Note that oγ and (ε) i have a consonantal value (ω and y respectively) in certain initial situations:

oyzop dog neyzop the dog (pewhor) neyzoop the dogs zin road $au e_2$ in the road ($tehy\bar{e}$) neziooye the roads

The fuller forms are also used with certain nouns denoting periods of time:

пвоγовіφ the time твромпв the year пвігооγ the day твуфн the night (оуфн) твумоу the hour (оумоу)

Note that oynoy and oygm fall under the two-consonant rule above.

1.4 Prepositions. Coptic prepositions are proclitic (i.e. unstressed and bound) to the word they govern. In many texts some or all of the prepositions are printed as a unit with the following word: 21 mxo1 on the ship, 6 mm; to the house. In this text, however, all prepositions will be printed as separate words: 21 mxo1, 6 mm; An exception will be made only in the case of the preposition 6 (to, for) if it is ligatured orthographically to a following oy- as 6y-.

The preposition mN (with) is used as the conjunction "and" in joining two nouns: npome mN $Tec_2 lme$ the man and the woman.

A definite noun followed by a prepositional phrase or local adverb (e.g. WMAY there) constitutes a full predication (sentence) in Coptic:

The man is on the ship.

THE CLINE OF THE COMMAN IS IN THE WOMAN IS IN THE HOUSE.

NEXHY HMAY.

The ships are there.

In sentences of this type there is no overt equivalent of English "is/are." We shall refer to sentences of this type as sentences with adverbial predicates.

Vocabulary 1

In the lesson vocabularies all nouns will be given with the definite article, separated from the noun by a period. This device makes both the gender of the noun and the correct form of the article clear at a glance. To save space, the article is not included in the definition. Prepositions and particles which regularly have assimilation of final \overline{N} to \overline{M} before n and M will be noted, as e.g. $2\overline{N}$ ($2\overline{M}$).

n.pome man, person; mankind. те.c2:ме (pl. ме.2:оме) woman, wife.

π.2 λλο old man, monk.
Θλλω old woman (= τ.2 λλω).
π. χωωμε book, book-roll, document.

n.wne stone.

T6.21H (pl. N6.2100Y6) road,

way, path.

m. Tooy mountain;
monastery.

m. HI house.

m. NoyB gold.

2N (2N) in.

2L under.

2I on, upon.

2! xN (2! xN) on, upon.

MN with, together with,
in the company of;
and.

Exercises

- A.1. 21 T621H
 - 2. 21 ntooy
 - 3. 2N T621H
 - 4. 2M THI
 - 5. 2A RHI
 - 6. 21 none
 - 7. 21 RXOOME
 - 8. ны проме
 - 9. MN TEC21ME
 - 10. 21XN NE2100YE
- B.1. none 21xN T621H.
 - 2. N2XXO 21 T621H.
 - 3. N2XXW 2M THI.
 - 4. THOYB 22 HONE.
 -
 - 5. RXWWME 21 NWN6.

- 11. 2A NHI
- 12. 21XN NTOOY
- 13. 2N NHI
- 14. MN NE210ME
- 15. 21 MNOYB
- 16. πΝΟΥΒ ΜΝ ΝΧΟΟΜΕ
- 17. π2 ΧλΟ ΜΝ ΘΧλΦ
- 18. HPWHE MN TECZIME
- 19. NPWME MN NEZIOME
- 20. N2 XAO MN N2 XAO
- 6. пні 21хм птооу.
 - 7. NPWME 21XM NTOOY.
 - 8. TEC21ME MN NPWME.
 - 9. NXWWME 2M THI.

Lesson 2

2.1 The indefinite article. The indefinite article for nouns of either gender is oy in the singular, 26N in the plural, prefixed directly to the noun:

OYEO1 a ship

oypome a man

cenpume men, some men

oylih a road

cenziooye roads, some roads.

The plural indefinite article may be translated as "some, certain" or be omitted entirely in translation, as the context requires. The plural indefinite article is frequently written as ${}_2N$ and is easily confused with the preposition ${}_2N$. In the exercises to the lessons we shall always distinguish between the two, but in part of the Reading Selections the orthography of the source is maintained.

Because the use of the Coptic articles, both definite and indefinite, corresponds closely to the use of the articles in English, only exceptions to this general correspondence will be noted in the following lessons when appropriate. References to the omission of the article require special attention. For the present lesson note that indefinite nouns designating unspecific quantities of a substance require an indefinite article in Coptic where there is none in English:

ογμοογ water 26Νο6ΙΚ bread 26Νλ4 meat

The choice between the singular and plural article here is lexical, i.e. it depends on the particular noun. All such nouns, if definite and specific, may of course appear with the definite article: nμοογ, ποεικ, πλ4. Abstract nouns, such as με truth, often appear with either article (ογμε, τμε) where English employs no article.

2.2 Indefinite nouns cannot be used as subjects of

sentences with adverbial predicates unless introduced by the word $o_Y \overline{N}$ or its negative:

OYN-OYZXXO 21 TE21H. A monk is on the road.

 $o\gamma \overline{N}$ is actually a predicator of existence ("there is, there are"), and the sentence given may also be translated as "There is a monk on the road."

The negative of oyN- is mN- (also spelled $\overline{m}m\overline{n}$ -). In general, an *indefinite* article is deleted (omitted) in negation in Coptic:

 $M\overline{N}-2\overline{\lambda}\lambda O$ 21 T621H. There is no monk on the road. $M\overline{N}-POME$ $2\overline{M}$ DH1. There is no man in the house.

 $oy\overline{N}$ - and $m\overline{N}$ - are not used before definite nouns.

The sentence npome 2M nmi is negated by adding Am:

проме 2M пн: An. The man is not in the house.

2.3 The genitive (or possessive) relationship between two nouns is expressed by the preposition \overline{N} (of):

то сере N тес 2 IMe the woman's daughter

If the first noun is indefinite, however, the preposition \overline{N} is used instead of \overline{N} :

OYXWWME NTE $n_2 \overline{\lambda} \lambda 0$ a book of the monk OY2M2 $\lambda \lambda$ NTE n_{PO} a servant of the king

Vocabulary 2

n. 2 H 2 λλ, τ. 2 M 2 λλ (ΘΜ 2 λλ)

slave, servant.

n. TET fish.

n. Ppo (pl. N. Ppwoy) king;

τ. Ppw queen.

n. ωγοειν light.

n. χοι (pl. Ν. εχηγ) ship, boat.

τ. ne (pl. Μ. πηγε) sky,

τε. μρω (pl. Ν. προογε) harbor.

n. μοογ water.

N (Μ) of.

NTE Of. oyn- there is, there are. мм-, ммм- there is not, there are not.

өтве (prep.) about, concerning; for the sake of, because of. an not.

Exercises

A.1. 21XH UXOL

2. 2x THE

3. 2N Mnhy6

4. етве прро

5. MN OM2 AA

6. 2N TEMPW

7. 21 nelepo

8. STES THOYS

9. 2N N6MPOOY6

10. 2x nxo1

В.1. ПТЕТ М петеро

2. NEXHY N NPWME

3. NHI M NPPO

4. πρώη Μ π₂ χλο

5. noyogin N The

6. THOOY N TEMPO

C.1. OYN-OYOYOGIN 2N MUHYG. 6. MN-HI 21XM UTOOY.

2. MN-TET 2N TEMPω.

3. MN-MOOY 2M netepo.

4. OYN-OY2 XXO 21 T621H.

11. GTBG 2 GNPPWOY

12. 26NPWM6 MN 26N210M6

13. SENTODY MN SENSIONYE

14. 2A OYWNE

15. 21XN OYTOOY

16. 2N OYTET

17. 21 26NH1

18. MN ΟΥ2 XXW

19. етве оухффме

20. ETEE HOYOEIN

7. пран й тёрш

8. Νχωωμε Μ η₂ λλο

9. **пно**оу й петеро

10. 26NWN6 NT6 NTOOY

11. OY2 MZ AA NTG HPPO

12. TEC2 IME H 112 H2 AA

7. NEXHY 21 HELEPO AN.

8. MN-x01 21 ne1epo.

9. NXOOME 21XM NXOL AN.

5. ΟΥN-26NON6 2N TEMPO. 10. MN-NOYB 2M ΠΗΙ Μ Π2M2λλ.

Lesson 3

3.1 Relative clauses. As we shall see in subsequent lessons, relative clauses in Coptic exhibit a variety of forms, depending on the type of predication involved. In the present lesson we shall consider only those relative clauses associated with sentences with adverbial predicates. Note the transformation

проме 2 м пні → (проме) ет 2 м пні
The man is in the house. (the man) who is in the house
The relative pronoun er functions here as the subject of the relative clause; it is not inflected for number or gender:

TEC21MG 6T 21 T621H the woman who is on the road $\overline{N}_2\overline{\lambda}\lambda O$ 6T $2\overline{N}$ 96N66T6 the monks who are in the monastery Negation is with λN : $\overline{N}_2\overline{\lambda}\lambda O$ 6T $2\overline{N}$ 96N66T6 λN .

Relative clauses cannot be used to modify an indefinite noun. This is an important general rule of Coptic.

Any relative clause may be substantivized, i.e. converted to the status of a noun, by prefixing the appropriate form of the definite article:

neт 2 m пн: the one who (he who, that which) is in the house

тет мN понре the one (f.) who is with the boy

нет 2: пхо: those who (those things which) are on

the ship

Such constructions may refer to persons or things, depending on the context.

The relative clause of $\overline{M}M\lambda\gamma$, who (which) is there, is used to express the further demonstrative "that":

npone of Mmay that man nexhy of Mmay those ships

3.2 Greek nouns. The typical Coptic text contains a large number of Greek loanwords. Greek masculine and feminine nouns retain their gender; Greek neuter nouns are treated as masculine:

δ άγγελος παρτέλος the angel $\dot{\eta}$ έπιστολή τεπιστολή the letter $\dot{\eta}$ ψυχή τεψχή the soul το πνεύμα πεπνέγκα the spirit το δῶρον παφρον the gift

Greek nouns appear in the nominative singular form of Greek and are usually not inflected in any way. Occasionally, however, a Coptic plural ending is added to a Greek noun:

νεηιστολοογε the letters κεψγχοογε the souls

The Greek noun ἡ δάλασσα (the sea) was borrowed as τ. 2λλλccλ, i.e. e was taken as the definite article plus 2. Thus, "a sea" is ογ2λλλccλ.

Initial χ , ϕ , ϑ , ψ , ξ of Greek nouns are considered two consonants in attaching the definite article (cf. Intro., p. x).

T6.χωρλ the country T6.Ψχχη the soul n6.φιλοςοφος the philosopher T6.θγςιλ the offering.

Vocabulary 3

n.+нь (pl. нь.тнь) town, village.

n.po (pl. \overline{N} .pwoy) door, gate.

n.xocic (pl. N.xicooye) master, owner, lord;

w. art., the Lord.

п. ноуте god; w. art., God.

ne.кро (ме.крооу) shore, bank, margin-land.

n. Kake darkness.

п. gнрв son, child, boy.

T.96676 daughter, girl.

MMAY (adv.) there, in that place.

נוף אֹ (נוף אֹ) (prep.) at the mouth or entrance of.

אאבף אֹ, אואבף אֹ (אאבף אֹ) in the presence of, before.

Greek nouns:

ealacca (ἡ θάλασσα) sea, ocean. π.ταφος (ὁ τάφος) tomb.
τ.πολις (ἡ πόλις) city. π.καθητής)
τ.επιστολη (ἡ ἐπιστολή) letter. pupil, disciple.
π.αγγελος) τ.εκκλησια (ἡ ἐκκλησία)
angel, messenger. church.

Proper names:

πλγλος (Παῦλος) Paul.

8. HMOOY ET 2M HELEPO

ιнсογc ('Ιησοῦς) Jesus; almost always abbreviated in Coptic texts: ΤC, ΤΗC.

Exercises

A.1. 2 IPN TEKKAHCIA 11. ОУЕПІСТОЛН ЙТЕ ПЛУЛОС 12. MMAGHTHC N TC 2. NNA2PM nFPO 3. гм птафос 13. 2H HPAN H TECOGIC 4. MN MMAGHTHC 14. NAZPM THOYTE 5. **RPO M RHI 15.** пекро м петеро 6. Excelc M Excl 16. 21XM NEKPO N OANACCA 17. 2N OYKAKE 7. NXICOOYE N NEXHY 8. TOJEEPE M 112M2AA 18. 21рн про н птафос 9. 21 рм про м пн 1 19. \overline{N} XWWME \overline{M} \overline{M} XWHTHC 10. OYEKKAHCIA NTE NIME 20. Npwme \overline{N} netme B.1. HWNE ET 2N TEMPO 9. NAFFEROC GT 2N MINYS 2. MMAGHTHC 6T MN TC 10. Νέχλο ετ ε<u>Μ</u> πτοογ 3. HEARE ET 21XN THOLIC 11. NPOME N THOMIC OF MMAY 4. NEKKAHCIA ET 2N THOAIC 12. NETME ET MMAY 5. **noyoein** et 2 n manye 13. Милентис й проме ет Милу 6. NTET ST 2N SALACCA 14. NOHPE M 12 M2 AX 6T MMAY 7. $\pi_2\overline{\pi}_2$ ax et $\overline{\pi}$ nazp $\overline{\pi}$ nxoeic 15. Nexhy et $21x\overline{\pi}$ nekpo $\overline{\pi}$

петеро

- C.1. HNOYTE 2N THE.
 - 2. м**n**-вкканста 2**m** п+мв бт ммау.
 - 3. оүй-оүхггөхос гірй про й птафос.
 - 4. Νεπιστολή ΜΝ Νχώωμε.
 - 5. MN-XOOM6 MHAY.
 - 6. ОУМ-ОУМАӨНТНО ЙТЕ ПЛУЛОС 21РЙ ПРО.

- 7. MN-OYOGIN 2H HKAKG.
- 8. nxoeic H nhi 2H nhi An.
- 9. nxol 21xm nelepo xn.
- 10. Порнре № пфме гі тегін.
- ОУЙ-26НТАФОС 2Й НЕКРШОУ
 БТ ЙМАУ.
- 12. OYN-OYKAKE 21XN THOAIC.

Lesson 4

4.1 Pronominal possession is indicated by inserting a bound form of the appropriate pronoun between the definite article and the noun so modified. It is best to learn the forms, i.e. the article plus the pronoun, as a unit:

fem. sing. noun masc. sing. noun sg. 1 com. my father my mother TALLOT TAMAAY 2 masc. your father TERMANY your mother DEKEIOT 2 fem. your father TOYMANY your mother HOYEIWT 3 masc. his father TERMANY his mother печетот 3 fem. RECELOT her father TECHANY her mother pl. 1 com. TENMANY our mother our father HENELOT 2 com. петПетот vour father τετπηλλγ your mother 3 com. their father TEYMANY their mother печетот plural noun (my brothers, etc.) sg. 1 com. pl. 1 com. NACNHY NENCHHY 2 masc. 2 com. NEKCNHY NETNCHHY 2 fem. NOYCHHY 3 masc. NEGCNHY 3 com. NEYCHHY 3 fem. NECCHHY

Note that there is a gender distinction in the second and third persons of the singular but not of the plural. This is characteristic of all pronominal paradigms in Coptic. The term "common" (com.) refers to forms or categories where no gender distinction is made.

4.2 The nearer demonstrative "this" is expressed by

masc. sing. nei- fem. sing. Tei- com. pl. Neiprefixed directly to the noun:

neipome this man

Telczime this woman

Neichny these brothers

After a noun with a demonstrative adjective the genitive is usually expressed by NTG, as in

neixwone Ate nacon this book of my brother ('s)

4.3 The pronominal element -ke- inserted between the article and the noun expresses "other":

пкероме the other man Nкероме the other men
The indefinite article is omitted in the singular but not
in the plural:

керюме another man генкерюме (some) other men
-ке- may also be used after demonstrative or possessive
prefixes:

петкероме this other man пакежог my other ship -ке- is not inflected for number or gender in this usage.

Vocabulary 4

n.con (pl. ме.смну) brother; often of a brother monk. т.соме sister.

n. 61 ω (pl. $\overline{\mathbf{n}}$. 610 τ 6) father; (pl.) parents, ancestors.

T.MALY mother. n.NOB6 Sin.

n. HPH wine.

n.oeik bread; piece or loaf of bread.

n.ma place; m neima here, in this place.

Greek nouns:

trunk (h elphyn) peace. η. επισκοπος (δ έπίσμοπος) bishop.

212TN, 21TN (212TH) (prep.) near, with, beside.

NCA (prep.) behind, in back of.

N (₩) (prep.) in; mostly synon-

ymous with 2N.

η.κοςμος (δ κόσμος) world. η.μοναχός (δ μοναχός) monk. τ. εντολή (ἡ έντολή) command. commandment.

T. Aropa (h dyopd) agora. forum, marketplace.

11. THOYS THE REYMOGIC

12. HPAN N TETNMANY

14. 2X2TN NEITME

16. мп нет п+ме

15. 23 NOYNOB6

13. 2A2TN TEKEKKAHCIA

Exercises

- А.1. Иса нечмаентис б. Инагри пенжоетс 11. МИ петири
 - 12. MM N69CNHY 2. нагря пеужовіс 7. гі теуекканста
 - 2 1 РМ печтафос
 8. етве петй+ме 13. GTBG TENCONG
 - 4. MN TERMEPE 9. 2N TOYHOLIC 14. 2M RK6H1
 - 5. NCA necompe 10. 2M nelkochoc 15. 2N KEMA
- B.1. Nentoah N nenetote
 - 2. **ПРАН Н ПАВІЮТ**
 - 3. про ₩ пекна
 - 4. про и пкент
 - 5. STES NENNORS
 - 6. 222ТЙ ПВИН1
 - 7. 2N OYELPHNH
 - 8. NNA2PH HENERICKOROC
 - 9. THANY N TO
 - 10. $nhp\pi$ π neihonaxoc
- C.1. HENXOBIC 21 HXOL AN.
 - 2. MN-HPH M NOIMA.
- 19. 2N TATOPA N THORIC

17. TELENICTOAH NTE NAYAOC

18. neixol Ate nenxoeic

- 4. мй-вірнин гй петкосмос.
 - 5. HAGIOT MN TAMARY 2M HHI.
- 3. OYN-OYEXAO EIPN TOKKAHCIA.

- 6. OYN-ZENOEIK MMAY.
- 7. HENCON 21 HEKPO N BANACCA.
- 8. OYN-OYXOL 2A2TH HEKPO.
- 9. OYN-OY2 $\overline{\lambda}\lambda\omega$ 21 $\overline{p}\overline{m}$ npo \overline{m} neah1. 14. neaxwwe 21 nwne et
- 10. HENCHHY 21XM TTOOY.

- 11. πογcon 2 m πτλφος λη.
- 12. netneiot 21 naxo1.
- 13. nenxol 2N tempo.
- 14. **ПЕЧХФФМЕ 21 ПФМЕ 6Т ММА**У.
 - 15. MN-21H M MMA 6T MMAY.

Lesson 5

5.1 Sentences with nominal predicates. A second type of non-verbal sentence is illustrated by

nacior no. He is my father. It is my father.

TAMAAY Te. She (It) is my mother.

nachty ne. They are (It is) my brothers.

оуроме пе. He (It) is a man.

oyczime Te. She (It) is a woman.

2 GN EXHY NG. They are (It is) ships.

The pronominal subject is expressed by ne (m.s.), Te (f.s.), and Ne (pl.), the choice of which depends usually on the gender and number of the predicate noun. Simple two-member sentences like the above are relatively rare except in response to such questions as "Who is that?" "What are these?" where an answer giving the predicate alone is sufficient, the subject being understood from the context. Modifiers of the predicate, such as a genitive phrase, may optionally stand after the pronominal subject:

помире пе \overline{M} поунив. He is the son of the priest.

A nominal subject may be added to the basic predication, producing a three-member sentence in which ne, Te, we are reduced virtually to the status of a copula. If

the predicate is indefinite, the order is almost always predicate + no, the subject being placed before or after the whole unit:

OYCL2 NG NAGIOT.

NAGIOT OYCL2 NG.

My father is a teacher.

If the subject and predicate are both definite, the normal position of ne, re, we is between them:

nelpome ne nenca2. This man is our teacher.

Identification of subject and predicate in this case can be made only on a contextual basis. The rarer order, nerpone nench nerch nench nerch neighbor nerch neighbor nench nerch ne

In the event that there is a disagreement in the number or gender of subject and predicate, the copula ns, rs, Ns usually assumes the number and gender of the noun immediately preceding it.

All of the preceding sentences are negated by placing \overline{N} (\overline{N}) before the predicate and \overline{N} before the \overline{N} , \overline{N} 6:

Ħ πλειωτ λη πε. It is not my father.

 $\pi \lambda \varepsilon \iota \omega \tau \ \overline{N}$ oych2 in $\pi \varepsilon$. My father is not a teacher.

M nenca: an ne neipome. This man is not our teacher.

Note that in the case where both subject and predicate are definite, the nominal element negated is, by definition, the predicate.

Sentences with nominal predicates are converted to the status of relative clauses with ere. For the moment we shall restrict ourselves to those clauses where ere functions as the subject of the relative clause:

npone ete oych ne the man who is a teacher npone ete \overline{N} oych an ne the man who is not a teacher.

The phrase ere man me is frequently used to introduce explanatory material, much like English "namely, i.e.,

that is to say":

пенсотир ете пы пе ТС nexc our savior, i.e. Jesus Christ

5.2 The nearer demonstrative pronouns (this, these) are mai (m.s.), tai (f.s.), and wai (pl.). They are frequently employed as subjects in sentences with nominal predicates:

> These are his words. NAI NE NEGDAXE. This is my ship. HAL HE HAXOL. TAI OY2 HZAA TE. This is a maidservant. OYZMZAA TE TAI.

Vocabulary 5

n.ca2 teacher, master; scribe. n.2a1 husband.

п.оүнив priest (Christian or т. 60м power, strength. otherwise). T.MHT6 middle, midst;

п. гамфе (pl. N. гамфиче) carpenter.

п. ерпе, п. рпе (pl. N. рпнуе) temple.

п.минфе crowd, throng.

Greek nouns:

ne.xpιcτoc (ὁ χριστός) the Christ, regularly abbr. πc. π. εγλητελιον (τὸ εὐαγγέλιον) gospel.

τ.πλροενος (ή παρθένος) virgin; young woman.

T. OPINH (ή ὁρεινή) mountain district, hill-country.

π. λ c πλ c moc (δ άσπασμός) greeting.

π. correp (ο σωτήρ) savior, redeemer; sometimes abbreviated as COP.

Proper names:

GAICABET Elizabeth.

ZAXAPIAC Zacharias.

N/2N TMHTE N in the

middle/midst of.

мефак (adv.) perhaps.

came1.

 $\pi.61MOYA$ (f. $\tau.61M1YA6$)

MAPIA Mary. ιωςκφ

1 W 2 A N N H C John.

Joseph.

Exercises

- A.1. OY2 M2 AA TE NTE TAMAAY.
 - 2. OYTET NO.
 - 3. ОУЖФОМЕ ПЕ ПТЕ ПЕКСОМ.
 - 4. OYNAPSENOC TE.
 - 5. 26NOYHHB NG.
 - 6. nappe n tacone ne.
 - 7. TOEEPE M 12 AMOE TE.
 - 8. THANY M RENCOTH! TE.
- B.1. TALL TE TEOM \overline{M} RNOYTE.
 - 2. HAL M DECOAL AN DE.
 - 3. OYEAMAYAE TE TAI.
 - 4. NAI NG WOAXS W HEYAFFEA LON.
 - 5. neythe 2N topinh.

 - 7. NACIOT OYZAMOC NE.
 - 8. печфире оуоунив пе.
 - 9. 26 NOYHHB NG NG4CNHY.
 - 10. мефак песзат пе.
 - 11. noy221 21PM npo.
 - 12. NECPAN NE EXICABET.
- 13. EXICABET THANY TO \overline{N} 102 ANNHC.
- C.1. NETME ET NCA HTOOY
 - 2. приме ете оуепіскопос пе
 - 3. NEIGAME STE RECACRACHOC NE
 - 4. пиннов вт 21xm пекро
 - 5. $T2\overline{\lambda}\lambda\omega$ ete Oynapeenoc te
 - 6. NE2100YE ST 2N TOPINH
 - 7. NEXHY ET $2\overline{N}$ THHTE \overline{N} GARACCA
 - 8. next, ete nai ne nencothp
 - 9. перпе, ете пат пе пит й пхоетс
 - 10. neixowe, etc mli ne neylication

- 9. n61moya n6 H n6461mt.
- 10. N OYXOL AN NG.
- 11. oynose ne.
- 12. N OYNOYTE AN HE.
- 13. M HENHI AN HE.
- 14. 26NCA2 NE.
- 15. NENTONH NE M NENTOEIC.
- 14. IO2ANNHC RE ROHPE \overline{N} ZAXAPIAC.
- 16. петині зазти перпе.
- 17. мерак петроме пе пехс.
- 18. пат не паспаснос й марта.
- 6. RESHI \overline{N} THATE \overline{N} THOMIC. 19. MAPIA OYRAPOGNOC TE.
 - 20. OYN-OYMHHOE N TMHTE N тагора.
 - 21. мп-егрини й петма.
 - 22. OYN-zentme 2N TOPINH.
 - 23. MEGAK OYN-OYCA2 2H ITME.
 - 24. NAI NE NENNOBE.
 - 25. 26NOGIK NE NAI.

Lesson 6

6.1 The independent personal pronouns.

ANOK	I	YNON	we
йток	you (m.s.)	ЙтωтЙ	you (c.p1.)
ÑТО	you (f.s.)		
П ТОЧ	he, it (m.)	NTOOY	they (c.pl.)
NTOC	she, it (f.)		

These pronouns occur frequently in sentences with me, Te, Ne. When used as predicates in a two-member sentence, they are invariably followed by me:

ANOK NG. It is I. ANON NG. It is we. Wroc ng. It is she.

In three-member sentences they may appear in ordinary subject or predicate positions:

In sentences with an indefinite nominal predicate a special construction without no is used with the pronouns of the 1st and 2nd person; negation is with an alone:

ANOK OY2AMGE (AN) I am (not) a carpenter.

ANON 26NOYHHB. We are priests.

In this construction a reduced proclitic form of the pronoun is very often used:

 NTK you (m.s.)
 NT6TN you (c.pl.)

 NT6 you (f.s.)

as in

 $\overline{N}T\overline{K}$ -OY2TH5AOC (AN) I am (not) an angel. $\overline{N}T\overline{K}$ -OY2AM96. You are a carpenter. $\overline{N}T6T\overline{N}$ -26MMA0HTHC. You are pupils.

A 3rd person masc. form $\overline{w}\tau\overline{q}$ - also occurs, but is very rare. The reduced forms of the 1st and 2nd person pronouns may also be used with a definite predicate, but this construction is rather infrequent:

ANT-OHIZAA M nxoeic. I am the handmaiden of the Lord.

6.2 The interrogative pronouns.

NIM who? λφ what? oγ what?

These pronouns are used in sentences with ne, Te, Ne:

אוא ne? Who is it? עס ne? What is it?

или пе петроме? Who is this man?

NIM NG NGKPAN? What is your name? (note idiom)

OY NG NAI? What is this?
OY NG NAI? What are these?

The interrogative pronoun normally stands first. The choice of number and gender for the copula depends on the understood or expressed subject. The pronoun oy is also found with the indefinite article:

oyoy ne? What is it? (lit.: It is a what?)
26Noy Ne? What are they (lit.: They are whats?)

When the subject is a personal pronoun of the 1st or 2nd person, it may be placed before NIM or oy in normal or proclitic form:

Who are you?

Who are you?

What are you?

The personal pronoun may be repeated for emphasis:

ANT-NIM ANOK? Who am I?

Note that NIM may also be used in ordinary genitive constructions:

пфире N nim? whose son?

Vocabulary 6

n. wwc (pl. N. wooc) shepherd. n. maein sign, token;

τ.cogs field, open country. marvel, miracle.

N. GCOOY sheep (pl.). T.CHUE SWOTA.

n. 600y glory, honor.

Greek nouns:

n.λλος (δ λαός) people.

n.cyrrenhc (δ συγγενής) kinsman (usually plural).

n. nonoc (ô vóuoc) law.

π. 2 Η Γ ΕΜΟΝ (ὁ ἡγ Εμών) governor, one in authority.

Proper names:

T.CYPIA Syria (note article).

T. FARIARIA Galilee (note article).

לוגגאן Judea (ווגגאן; note article).

п. ісрана Israel, usually abbreviated as nтна (use article when it denotes the people).

Exercises

- A.1. $\Delta N \overline{\Gamma}$ -OY2 \overline{N} 2 $\Delta \lambda$ \overline{N} T6 \overline{n} 2 H \overline{r} 6 H \overline{n} 0 \overline{N} 7. \overline{N} 70 T6 T Δ C2 IM6.
 - 2. OYN-OYOUC MMAY 21 TCOOE. 18. N OYONE AN RE.
 - 3. **Т**ток пе пенсы 2.

 - 5. N ANON AN RG.

 - 7. NT6-NIM NTO?
 - 8. ANT-OYALLEYOR ME MNONTE. 23. ON THE TEANIALL NE.
 - 9. OYN-OYCHAE HMAY.
 - 10. OY NO NEIMAGIN?
 - 11. NTO9 N6 NOY2 X 1.
 - 12. πλι πε πνομος ή πνούτε. 26. Ντώτη πε.
 - 13. OYOY NE NEYATTEALON?
 - 14. nTHX n6 n64xxoc.
- 15. NTOOY No.
- 16. NIM NE? NECCYTTENHO NE.

neipome.

- 19. хнок пе тоснф.
- 4. ANON NE NEGECOOY. 20. NEOOY M TXOEIC 21XM
- 6. \overline{N} TOU RE RECOY \overline{M} REULACC. 21. LO RE RAI? OYMAEIN RE.
 - 22. MEMAK NTOU NE NEXT.

 - 24. nepne \overline{N} NIM ne nai?
 - 25. REIXOWME OYEYAFTEXION
 - ne.

 - 27. до те тегін?
 - 28. OYN-OY2HIGMON 2N TCYPIA.
 - 29. Am TE TELSOM?
 - 30. NTETW-2 SNOYHHB.

- 31. 0XXW NNA2PM П2НГЕМОИ.
- 32. MN-+ME 2N TOPINH ST HMAY.
- 33. πaι πε πνομός π πppo.
- 34. OYN-26NGOOC 21 HTOOY.
- 35. REIOYOGIN OYMAGIN RG.
- 36. Птос оупаробнос тб.
- 37. NEAMOYA N NIM NE?
- 38. IO2ANNHC H nexc AN ne.
- 39. пкероме пасуггение пе.
- 40. Пточ пе пфире М памфе.

- 41. ag ne nkake et 21xN tnox1c?
- 42. NTOOY NACHHY NE.
- 43. NAI NE ÑŒAXE Ħ
- 44. NK66COOY 2N TCOOG.
- 45. NIM NE NPAN M N2HFGMON?
- 46. TAI TE TACHEG.
- 47. OYMAGIN NTG TERGON NG.
- 48. NIM TO THANY N IWENNHC?

7.1 The First Perfect. Verbal inflection in Coptic is commonly, but not solely, of the form: verbal prefix + subject (noun/pronoun) + verb. The infinitive is the main lexical form of the verb and may occur in all of the verbal conjugations. Its uses and further modifications will be dealt with in subsequent lessons. The conjugation known as the First Perfect is the narrative past tense par excellence and corresponds to the English preterite (simple past: I wrote, I wept, I sat down) or, if the context demands, the English perfect (I have written):

λIBOK	I went	λNBΦK	we went	
AKBOK	you (m.s.) went	ATETNBOK	you (c.pl.) went	
APB ØK	you (f.s.) went			
AGBOK	he went	λΥΒΦK	they went	
ACBOK	she went			

The pronominal elements are for the most part familiar from the possessive prefixes of Lesson 4. In the 1st

person singular 1 is normal for most of the verbal system (contrast the -x- of nxelor). The pronominal element of the 2nd person feminine exhibits much variation and should be noted carefully for each conjugation introduced: xperok and xrok are also attested in the First Perfect.

If the subject is nominal, the verbal prefix is λ -:

a-npwms swk the man went

There are two other ways in which nominal subjects may be used in a verbal phrase: (1) they may stand before the verbal unit, which in the First Perfect still requires a pronoun as well:

TEC 2 IME ACROK the man went the woman went

or (2) they may stand after the verbal unit, again with a pronominal subject, introduced by the element $\overline{N61}$:

AGROK NG: TEC21MG the man went

All three constructions are common and differ only in the emphasis accorded the subject. When the verbal prefix is followed by the indefinite article, the resulting λ -o γ ... may be spelled $\lambda\gamma$..., as in

α-ογε⊼λο βωκ οτ αγε⊼λο βωκ a monk went

- 7.2 The prepositions ϵ , ω , and $\varepsilon \times \overline{N}$ are frequent after verbs of motion.
 - e indicates motion to or toward a place or person,
 less commonly motion onto or into:

жупот в теккансіх. They ran to the church.

жумає в пхої. They got on (or into) the ship.

Otherwise the preposition в is very frequent in a

Otherwise the preposition 6 is very frequent in a general referential sense: "to, for, in regard to,"

with many other nuances that will be noted in passing.

2) gh indicates motion to, up to; it is used more frequently with persons than places:

жипот од печетот. He ran to his father.

дивок од пеніскопос. We went to the bishop.

3) exm indicates motion onto, on:

He got on the donkey.

acze exm nkaz. She fell on the ground.

 $ex\overline{N}$ properly denotes motion onto, while $2ix\overline{N}$ denotes static location; the two are sometimes interchanged. The same contrast exists with the less frequent pair $2ip\overline{N}$ (at the entrance of) and $ep\overline{N}$ (to the entrance of).

Several of the prepositions we have already introduced also occur freely with verbs of motion. For example 21, 21xN (on or along a surface), NCA (behind, after), MN (along with), 2N (within a circumscribed area), NNA2PN (into the presence of), 2A2TN (up to, near). The preposition NCA often has the sense of English "after" in "to go after," i.e. to go to fetch, or "to run after," i.e. to try to overtake. The reader should give particular attention to the use of prepositions with verbs, since these combinations are sometimes quite idiomatic and unpredictable.

Vocabulary 7

Bwk to go.

Mooge to walk, go on foot (usually).

- en to come; en Now to come after, come to get.
- axe to go up, climb (onto, up to: ϵ); to mount (an animal: $\epsilon x \overline{N}$).

not to run, to flee; not Now to pursue.

φληλ to pray (for something: e, eτεe, exN, ελ; for someone: e).

PIME to weep (for someone: e, exN).

2MOOC to sit down (at: 6).

n. 61ω (pl. N. 600γ) ass, donkey.

ne.270 (f. T6.270pe; pl. N6.2700p) horse.

Wer subject marker (see the lesson).

ETBE OY why?

The prepositions ϵ , $\epsilon \times \overline{N}$, $\epsilon \setminus \overline{N}$, $\epsilon \setminus \overline{N}$ as given in the lesson.

Greek nouns:

πεσρονος (ὁ θρόνος) throne.

τε.τρλπεζλ (ή τράπεζα) table.

- 1. A-NECHHY BOK 6 THOLIC.
- 2. понрежиет в пенит.
- 3. ACBOK NG! TGGCONG GPM NPO N TEKKAHCIA.
- 4. AY61 NCA NEYOHP6.
- 5. а-течмалу моофе в птафос.
- 6. A-NEGMACHTHE ANG 6 EXOL.
- давшк фа кесон.
- 8. диет бры печит.
- 9. димоофе $\overline{\text{N}}$ 61 $\overline{\text{TC}}$ еж $\overline{\text{M}}$ пекро $\overline{\text{N}}$ едадсед.
- 10. MHONAXOC AYAA6 6 NTOOY.
- 11. а-мечмаентис пот е кема.
- 12. ачфана бтве мечфире.
- 13. GTBG OY ATGTNOOT GEN TG21H?
- 14. ANMOOGE MN NENZIOME 6 ПТМВ.
- 15. λγε Χλο Βωκ φλ πεπισκοπος.
- 16. ачфана етве меннове.
- 17. ACGI NGI OM2AA NNA2PH NGCXOGIC.
- 18. AMAR Well newsource the.

- 19. 6TB6 OY AKNOT NCA NAGIOT?
- 20. ETBE OY APPIME ETBE NPOME ET MMAY?
- 21. дагноос ежн пекро н петеро.
- 22. A-OXAW PIME 6 RECOMPE.
- 23. AYEMOOC MN NEYCHHY.
- 24. AIMOOGE 21 TE21H MN TAGEGPE.
- 25. AYBOK 21 TE21H ET HMAY E THOXIC.
- 26. A-neszto not oa teztope.
- 27. A4AA6 6XM N692TO N61 N6NICKOROC.
- 28. TERNALY ACEL OF HEPPO N THORIC.
- 29. ANOTHER 23 HENCHHY 6T $2\overline{H}$ HTOOY.
- 30. AIPIME EXT NANOBE.
- 31. д-прро змоос ежм печеронос.
- 32. A-MMAGHTHC 2MOOC 21PM NPO M NH1.
- 33. AYROT NG! NOHPG 21 TG21H 6 TEMPO.
- 34. a-neio mooge NCA negaceic.
- 35. ANALE EXT NENGOOY.
- 36. AYEMOOC 6 TETPAREZA NEI TETCEINE MN NETGEEPE.
- 37. A-ZENPOME EL ZAZTN TEKKAHCIA.
- 38. GTBG OY AKMOOGG NCA HELD 21XN TG21H?
- 39. A-TC 2MOOC 2A2TN N64MAGHTHC.
- 40. AYMOOGE NEI NGOOC NCA NEYECOOY.
- 41. л-пелмоух пот е тсоще.
- 42. ANOAHA 6 OYMA6IN.
- 43. λ -nalog \overline{N} thoaig 61 ga nehremon \overline{N} toyalla.
- 44. A-TOGEPE 2MOOC MN NECCYFFENHC.
- 45. давык Пбі пангеным в тсурід.
- 46. $n\lambda$ 1 ne nnoyte \overline{H} $n\overline{1}\overline{H}\overline{\lambda}$.

8.1 Directional adverbs. Coptic possesses a set of directional adverbs which correspond very closely to English adverbs of the type "up, down, in, out, over, along, etc." As with their English counterparts, the directional meanings found with verbs of motion are for all practical purposes the basic meanings (e.g. to go up, to sink down, to run in), but extended uses are equally common (e.g. to shut up, to quiet down, to think over). The Coptic directional adverbs consist formally of the preposition e plus a noun, with or without the definite article. Most of the nouns in question are seldom met outside of these particular expressions and will be considered in more detail in a later lesson. Because these adverbs are so frequent, we shall follow the practice of other editors and write them as single units. The following eight are the most important:

eson out, away een forward, ahead ezoyn in enazoy back, rearward ezpai up, down erne upward enecht down engoi upward.

The adverbs may be used alone, as in

A461 620YN. He came in.

A400 600YN. He ran back.

But they very frequently combine with a simple preposition to form a compound prepositional phrase. Among the most frequent of these are

GBOX 6 out to, away to.
GBOX 2N out of, out from in, away from;
(rarely) out into.
GBOX MMO same as preceding.

680A 21 away from on, out from on, away from at.

(1) away from (a person); (2) through, out through (a place); (3) through the agency of (a person or thing).

ezoyn e to, into, toward.

62PA1 6 up to, down to.

62 Pλι ex W up onto, down upon.

enecht e down to, down into, down onto, down on.

епеснт ехи down onto, down on.

enazoy e back to.

бөн б ahead to, forward to.

The meaning of most such compounds when used with verbs of motion is generally self-evident, but caution is in order when dealing with their use with other verbs. The dictionary should always be consulted to check on idiomatic and unpredictable meanings.

8.2 Clauses containing a First Perfect may be coordinated with the conjunction $\lambda\gamma\omega$ (and) or follow one another with no conjunction (termed "asyndeton"):

A42HOOC AY@ A4PIH6. }

He sat down and wept.

8.3 Many infinitives are used as masculine singular nouns. This usage will be noted without further comment in the lesson vocabularies from now on ("as n.m.:"). For the infinitives in Vocabulary 7 note the nouns ne.gaha prayer, n.nor flight, and n.pime weeping.

Vocabulary 8

- φακε to speak, talk (to, with: ε, ΜΝ; about: ε, ἐλ, ετεε; against: Νελ, ογεε); as n.m.: word speech; matter, affair.
- Twoyn to arise, get up (from: 680λ 21, 680λ 2N); to rise up (against: 6, 6xN, $62P\lambda$ 1 6xN).

T.PI cell (of a monk).

n. : ws (pl. Ne. : BHYE) work, task; thing, matter, affair. PAKOTE Alexandria.

21TN (prep.) (1) through, by means of, by the agency of; (2) from with. from by (a person).

oyse (prep.) against, opposite.

Ayo (coni.) and.

Greek noun: η, κημα (τὸ βῆμα) platform, dais, viewing or judgement seat.

Note: Only the less predictable combinations of verb and prepositional phrase will be given in the lesson vocabularies. Other combinations in the exercises should be self-evident from the meanings of the individual words involved.

$A.1.$ гіт \overline{N} мекарлі	H A	
-----------------------------------	-----	--

- 2. OYBG NENCHHY
- 3. 6BOA 21TN T6121H
- 4. 6BOX 6 H616PO
- 5. GROA 2N THOALC
- 6. GBOX 21 TETPANGZA
- 7. 6TB6 N642WB
- 8. OYBE NETHOLXE
- 9. гітп нечолже
- 10. егоун 6 ракоте
- 11. 62 PA 1 6 RBHMA
- 12. 62 РА 1 6 ЖМ ПТООУ
- 13. 620YN 6 TAPI
- 14. 23 NE12BHYE
- 15. фа нечилентис
- - 2. ANGI GROYN, ANTHOOC.

- 16. 21PN TEAP !
- 17. 6TB6 NOYPIM6
- 18. эй печиют
- 19. епесит е ехахсса
- 20. **Enecht e nmooy**
- 21. 6 mazoy 6 neythe
- 22. 60H 6 NTOOY
- 23. 6вол 21 пвима
 - 24. 62 PA 1 6XM 162 TO
 - 25. 680% 21TM NPO M NHI
 - 26. 6BOX 2M HELMA
 - 27. OYEG HPAN M HPPO
 - 28. GBOA 21TH REOGIC
 - 29. erse ner\nor
- B.1. AYTWOYN, AYRWT 650A. 4. A42MOOC AYW A4QAX6 M $\overline{\text{N}}$ NPOM6.
 - 3. давык витол в неяфие. 5. титоли ввом за тетрапека.

- 6. АНМООФЕ ВЗОУН В РАКОТВ.
- 7. NKEPOME ANALE EZPAL EXM HELD.
- 8. STES OY APROT SEON 2M ROYHI?
- 9. λ-π2 ΧλΟ ΒΦΚ 62 ΟΥΝ 6 Τ64 ΡΙ.
- 10. etbe oy atethoaxe NCA nenxoeic? > of
- 11. AI2MOOC 2A2TM NAGIOT.
- 12. LYWLX6 21 REYHP $\overline{\mathbf{n}}$.
- 13. OYN-26NGOOY MN 26N2TOOP MMAY.
- 14. A461 6R6CHT 6BOX 21 RBHMA.
- 15. ХУТФОУН П61 ПРФМВ 62 РА 1 6ЖМ ПРРО.
- 16. AIDAX6 ETBE NANOBE.
- 17. A-Neglome Mooge enecht e nekpo.
- 18. ANBOK 62PA1 6 NENTMG.
- 19. ACROT GBOX 21TH HPOMG GT HMAY.
- 20. ANBOK 620YN NNA2PH HEHICKOROC.
- 21. ETBE OY AKBOK ERAZOY E TAPOPA?
- 22. A4BOK 6BOA 2M REIKOCHOC.
- 23. АСФАЖЕ МП ТЕСМАЛУ ЕТВЕ НЕСФИРЕ.
- 24. AYAA6 NGI NPWM6 EZPAI 6XN NGYZTOOP.
- 25. AYOLK & HOELK MN NIET.
- 26. АУФАЖЕ ОУВЕ ИЕЧЕНТОЛН.
- 27. ATTWOYN 680A 21 1169PONOC.
- 28. Anmooge 60H $2\overline{M}$ TKAKE.
- 29. дала барлі в пвинл.
- 30. A 4 QAX 6 GTB 6 NE 2 BHY 6 M IMONAXOC.

9.1 Prepositions with pronominal suffixes. A pronominal object of a preposition is expressed by means of a suffixed form of the pronoun. The preposition itself assumes a prepronominal form which must be learned with each preposition. For example, the preposition e becomes epo' before pronominal suffixes. The mark is a convention used in Coptic grammars to indicate any form to which a pronominal suffix is to be added. The prepronominal forms of the prepositions introduced thus far are as follows:

Ħ	™мо ⁴	6	epo*	6 b <u>M</u>	e p w *
йs	N2HT €	NXI S	s 1xo"	ИдерП	NA 2 PA"
z 🏃	22602	6XN	6X0*	7T\$45	*HTS4S
2 l	2100(00) *	2 1 T N	2 1 TOOT 5	OYB 6	оувн"
м П	и <u>м</u> мх *	МСУ	NC o	6TB6	6ТВННТ°
ga	gapo*	2 1 P N	2 1 7 00 *		

Because the variety exhibited by the prepronominal forms is at first puzzling, the following comments may be of some help:

- 1) In epo' and 2xpo' the final syllable -po' is an original part of the preposition; it was lost in the normal prenominal form. Qxpo' is on analogy with these.
- 2) \overline{N}_2HT' and $21\omega'$ were originally compounds: \overline{N}_2HT' in the belly (of), $21\omega(\omega)'$ on the back (of). The prenominal form $2\overline{N}$ was originally a noun meaning "interior" (cf. the $20\gamma N$ of $620\gamma N$); 21 was a simple preposition.
- 3) 21xw and 6xw are 21 and 6 with the noun xw (head, top); the noun is reduced to -x- in $21x\overline{N}$ and $6x\overline{N}$, which also contain the genitival \overline{N} .
- 4) 2190° and 690° are 21 and 6 with the noun po° mouth, door; the latter is reduced to p in 219N and 69N, with genitival N.

- 5) 217007° is 21 plus the noun τ_{007} ° (hand); the latter is reduced to $-\tau$ in 217N, with genitival N.
- 6) NA2PA contains the noun 2PA (face). The initial element is obscure. NA2PW also contains the genitival W.
- 7) $2\lambda 2TH'$ consists of 2λ plus the noun 2TH' (heart, breast), reduced to -2T— with the genitival \overline{N} in $2\lambda 2T\overline{N}$.

The nouns contained in these expressions will be dealt with in a later lesson.

Typical inflections of these prepositions are as follows:

epoi to me epon to us

ерок to you (m.s.) брот \overline{N} to you (pl.)

epo to you (f.s.)

spoq to him spooy to them

spoc to her

иймы with me иймын \overline{N}_2 нт in me \overline{N}_2 нт \overline{N}

 $\mathsf{N}\mathsf{\overline{M}}\mathsf{M}\mathsf{A}\mathsf{K}$ etc. $\mathsf{\overline{N}}_2\mathsf{H}\mathsf{T}\mathsf{\overline{T}}\mathsf{H}\mathsf{Y}\mathsf{T}\mathsf{N}$

ийме \overline{N}_2 нте

үотнѕй Бтнѕи үкмым ымы

NMMAC N2HTC

оувиг against me оувим

оувик etc. оуве-тнути

[ОҮВНТӨ]

ОУВНЧ ОУВНУ

OYBHC

втвинт because of me втвинтя

6ТВННТ \overline{K} etc. 6ТВ6ТНYТ \overline{N}

6TBHHT6

бтвинт бтвинто у

6твннтС

The prepronominal form of 21 appears as both 2100 and 21000; typical forms are

2 1 0 0 T ME 2 1 0 0 N N 2 1 0 N 2 1 0 N 2 1 0 O N 2 1 0 O N 2 1 0 O Y 2 1 0 O C 2 1 0

The following details on the forms of suffixed pronouns in general should be noted for future reference:

- 1) The 1st pers. sing. suffix appears as ι after a single vowel, as zero (nothing) after $-\tau$, and as τ elsewhere.
- 2) The suffix of the 2nd pers. fem. sing. appears as zero after a single vowel other than $-x^2$, as -6 in place of a single $-x^2$, as -6 after -7^2 , and as -76 elsewhere.
- 3) The suffix of the 2nd pers. pl. appears as $-\tau \bar{N}$ after single -o', $-\lambda'$, $-\omega'$, with the change of -o' to $-\omega'$ and $-\lambda'$ to $-\mu'$. After $-\tau'$ one uses $-\tau \mu \gamma \tau \bar{N}$. Elsewhere $-\tau \mu \gamma \tau \bar{N}$ is usually added to the prenominal form of the preposition or other form in question, as in $2\lambda 2\tau \bar{N} \tau \mu \gamma \tau \bar{N}$.

Vocabulary 9

- gone to come into being; to take place, happen; gone MMo' to befall, happen to (someone). Acgone it happened that (followed directly by the principal verb, as in Acgone Agel gapon it happened that he came to us).

 MOY to die (of, from: 6TBG, 2A); as n.m.: death, manner of death.
- 26 to fall; 26 6 to fall to, upon, into; to find, chance upon; 26 680λ to perish, be lost, fall away.

 NKOTK to lie down, sleep; often a euphemism for dying.
 20N to draw near, to approach (someone or something: 6, 620γN 6).

21C6 to grow weary, exhausted; to be troubled, afflicted.
As n.m. labor, toil; weariness, affliction.

гко to grow hungry; as n.m.: hunger, famine.

n. KA2 earth, ground.

n.ga festival, feast day.

ne.200γ day; H nooγ (adv.) today (note loss of 2);

nooγ up until today, until now.

ne.6xo6 bed.

минсь (минсю (prep.) after (of time); минсюс (adv.) afterward.

Greek words

Ae (δέ) postpositive conjunction: but, however.

Frequently marks the introduction of a new subject or topic and has no translation value.

τ. cynarwrh (ή συναγωγή) synagogue.

Exercises

Translate the following prepositional phrases. Replace the nominal object with the appropriate pronominal suffix. E.g. 6TB 6 TELLOWS + 6TB HHT¶.

- A.1. exH nkaz
 - 2. OYBE REYXOEIC
 - 3. 21 **896306**
 - 4. гірн птафос
 - 5. MNNCA RESHOY
 - 6. NAZPH HNOYTE
 - 7. 2N N616PO
 - 8. 21TN NGI21CG
 - 9. 6 NTET
 - 10. 6тве пеузко
 - 11. e necpan
- 12. ga N692 M22A
- 13. 21XH HKA2
- 14. ПСА печфире
- 15. 21TM n62KO

- 16. 23 na6x06
- 17. минсь пениют
- 18. 2N T64P1
- 19. 21XM N6KXO1
- 20. 23 NETNZICE
- 21. 6TB6 NOA
- 22. 21PN TEKKAHCIA
- 23. 9λ Τ2 λλω
- 24. OYBE NAMAKE
- 25. NASPH TERMANY
- 26. 21TH NOYOGIN
- 27. ZAZTN TEMPO
- 28. ежй пвимъ
- 29. ₩ nmooy
- 30. MNNCA HOA ST HNAY

- B. 1. 680X 2100N
- 7. 62PA 1 6XON
- 11. WNAZPAL

- 2. NHMA1
- 3. 680 λ \overline{N}_2 HT-THYTN 8. OY86-THYTN
- 12. **ETBHHTK**

6. 212THN

13. NA2P6

4. OYBHI

- 9. NMMHTN
- 14. 210K

- 5. **6TBHHTN**
- 10. STECHT SXOL 15. NMMAN
- C.1. A-MMOY gone 2H MKOCMOC 6BOX 2 ITN NENNOBE.
 - 2. MNNCA NAI A6 A9MOY N61 NPPO 6T MMAY.
 - 3. A- MOHPE 26 6 MCCHT 6XM MKA2.
 - 4. 6TB6 OY A-N6121C6 wone MMO1?
 - 5. ATRKOTK EXM REGADE. ATPINE.
 - 6. A-NPWHE 2WN 620YN 6 THE ST 21XH TOOY ST HMAY.
 - 7. MNNcoc A6 AY26 6 N6YOHP6 2N TCYNAPOPH.
 - 8. AYMOY NG! NPOME N TEIROXIC 2A OYEKO.
 - 9. λ -OYOA GONE \overline{M} NEIMA \overline{M} NE2OOY OT \overline{M} MAY.
 - 10. ANZICE ZITH HEZKO AYW ANTWOYN, ANBOK E KETME.
 - 11. A 4 DAX 6 NAMAN GTEG NXWWHG GT 2N THOALC.
 - 12. λ 121C6, λ 1 \overline{N} KOT \overline{K} , λ 1 ϕ λ H λ 6 HNOYT6 6TB6 N61 ϕ λ x6.
 - 13. A42WN 620YN NGI N6200Y M NWA.
 - 14. A-112 XAO MOY 21PH 11PO N TEAP1.
 - 15. AY2KO AYW AY61 6RA2OY 6 RTM6.
 - 16. MNNCOC AG A-TENHOLIC 26 6BOX 21TOOTH.
- 17. датфоун понре евох 21 пкд2, данфт од печенфт.
- 18. MNNCA NEIGAME A-HENMOEIC NKOTK.
- 19. ACOUNG AG AC26 6 HNOYB 2A N66A06.
- 20. A-NEXHY 2WN 620YN 6 TEMPW.

10.1 The Direct Object. The direct object of a transitive verb is usually introduced with the preposition \overline{N} (\overline{N} , \overline{N} MHo'):

λακωτ Ν ογηι. He built a house.

AGKOT MMOG. He built it.

Many verbs, especially those denoting perception, employ 6:

ANCOTH 6 negroop. We heard his voice.

ANCOTM 6PO4. We heard it.

Occasionally other prepositions assume this function, as for example $\overline{\mbox{Nc}_{\lambda}}$ in

Augine NCA Teuczine. He looked for his wife.

Augine Mcoc. He looked for her.

The appropriate preposition for each transitive verb will be given in the lesson vocabularies when a verb is introduced.

10.2 The Indirect Object (Dative). An indirect object, if present, is introduced with the preposition \overline{N} (NA*, inflected like $N\overline{N}NA$ * in §9.1):

Alt \vec{H} nxwwme \vec{H} npwme. I gave the book to the man.

Ait NA4 Μ πχωωνό. I gave him the book.

וג ਜੋ ਜ਼ੋਅਰਥ אבא. I gave it to him.

It is unfortunate that the prenominal forms of the most frequent direct and indirect object markers are the same $(\overline{N}, \overline{N})$ before n and M). In general the direct object precedes the indirect object unless the direct object is nominal and the indirect object is pronominal. In the latter case either order is correct, but there is a preference for placing the indirect object first. The preposition $e(epo^*)$ also often marks what Greek and English regard as an indirect object (dative).

10.3 The Negative of the First Perfect. The negative forms of the First Perfect are not formally related to the positive forms:

Mnibok I did not go MnerNbok we did not go MnerNbok you (m.s.) did not go MnerNbok you (pl.) did Mnebok you (f.s.) did not go not go Mnerbok he did not go Mnorbok they did not go Mnecbok she did not go

With a nominal subject: \overline{M} ne-npone box the man did not go. Variant spellings such as \overline{M} n \overline{K} -, \overline{M} n \overline{M} -, \overline{M} n \overline{M} - are not uncommon.

10.4 As noted in Lesson 9, many prepositions consist of a simple preposition compounded with a noun. The nouns occurring in these expressions belong to a special group which take pronominal suffixes to indicate possession. We shall deal with the more important of these individually in later lessons, but for the moment note $\tau \cot \tau$, the presuffixal form of $\tau \cot \tau$ (hand). The absolute form $\tau \cot \tau$ survives only in the special meanings "handle, tool, spade" and in some compound verbal expressions (see Glossary); in the sense of "hand" it has been replaced by ϵix . Prepositions compounded with $\tau \cot \tau$, such as $\epsilon i \tau \cot \tau$, eroor, $v \cot \tau$, often employ a construction with an anticipatory pronominal object before the real nominal object, the latter being introduced by the particle $v \cot \tau$

This same construction is also occasionally found with the other prepositions introduced thus far.

Vocabulary 10

κωτ to build, erect (Μmo'). NAY to see, look at (6).

COTH to hear, listen to (6); to obey (NA', NCA).

gine to seek, look for, inquire after (Nch); to visit (6); to greet (6).

61N6 to find (MMO').

+ to give (Μmo'); to entrust (μmo'; to: ετη); + ογες, ተ mn to fight with; ተ mmo' פּצּסֹג to sell (to: פּ, אב"). xı to take, receive (Mmo').

41 to lift up, take, bear, carry (Mmo*); 41 mm to agree with; 41 2x to bear, tolerate.

TG. OTHN garment, tunic.

T.2061T6, N.2061T6 garment, cloak.

ne.2700y sound, voice.

NTN (NTOOT') from; used in a wide variety of expressions, but very frequently with verbs of receiving, accepting, hearing. The separative notion lies in the verbal idiom; thus, with other verbs it has the sense of (being) near, by, with, in the hand of.

GTN (GTOOT') to; used frequently with verbs of entrusting, giving, handing over, transmitting,

- A.1. AT NAI TOYOTHN.
 - 2. 11 NA9 H NAXO1 680A.
 - 3. λ 1+ \overline{M} NNOYB STOOT $\overline{4}$.
 - 4. ANT MN NPWMG GT HMAY.
 - 5. AYT OYBHN.
 - 6. даф й пноув й печфире.
 - 7. AY41 H HONG 680A 21PO4.
 - 8. дчя й печфире варді.
 - 9. Minengi MMMAY.
- 10. дан й печелов, давок евол. 22. дит мнос етооте.
- 11. LYKOT \overline{N} OYPRE $\overline{M}MAY$.
- 12. Mnoycoth NCA negroeic.
- 13. дісютя є перрооу я поуннв. 24. печенот де япечвок
- 14. Ипісфти етве песмоу.

- 15. ANNAY GYXOI MMAY.
- 16. AYOING NEW! 2N TAPOPA.
- 17. ANSING MMOG EXM HEKPO MN DENCAS.
- 18. AIDINE 6PO4 2N TEAPI.
- 19. AMNAY BYAFFEROC.
- 20. AIXI MMOOY NTOOTE M DAGIOT.
- 21. AYXI MMO4 NTOOT.
- 23. A461N6 MMO4 2A ne46x06.
- 620YN.

- 25. GTBG OY MRGTNNKOTK?
- 26. ANGI 24 Noice of MMAY.
- 27. Hneczwn ezoyn e ntadoc.
- 28. мпі+ м проєгте євох.
- 29. Mnit N6 N TEIGTHN.
- 30. етве оу йпеет фарот?
- 31. Мпенфаже иймау.
- 32. 6TB 6 OY AYT NHMHTH?
- 33. MRE426 GPON MMAY.
- 34. Mnoyale 6 пеужот.
- 35. $\overline{\mathbf{M}}$ net $\overline{\mathbf{N}}$ x i $\overline{\mathbf{N}}$ net $\overline{\mathbf{N}}$ net $\overline{\mathbf{N}}$ to o to $\overline{\mathbf{N}}$ net $\overline{\mathbf{N}}$ e i o te.
- 36. MINGATWOYN 680% 21 NKA2.
- 37. Ачсфтя в неугрооу, ачпот ввох.
- 38. дучі Тбі пиннає мп пепіскопос.
- 39. МПЕНСШТМ 6 НЕЧЩАЖЕ.
- 40. AUXI M NAZOGITE.
- 41. Мпе-печмаентис сфтм мач.
- 42. Мпе-пафире ет епагоу.
- 43. ALCOTH GTBG NAI NTOOTH M HACON.
- 44. λ 4+ \overline{N} TEMMANY ETOOTOY \overline{N} NEMMARKHE.
- 45. AY61NG $\overline{M}MO4$ \overline{N} TMHTG \overline{M} $\overline{M}MHMGG$.
- 46. A-HZAMOG KOT NAN N OYHI MMAY.
- 47. MGGAK A-ROYZAI MOY.
- 48. AINAY 6 ΠΕΟΟΎ Η ΠΧΟΕΙΟ ΜΗ ΤΕЧ6ΟΜ.
- 49. AYROT NHMAN 6 TOPINH.
- 50. AGDATE NHMAL GTBE REYALTERION.

11.1 The Forms of the Infinitive. The nominal or pronominal object of many transitive verbs may be suffixed directly to the infinitive without the use of a prepositional object marker, as in

Aleine M πAeiωτ. → Alen-πAeiωτ. I found my father.

Aleine Mmoq. → Alenτa. I found him.

Infinitives that allow this construction have three distinct forms: (1) the normal (absolute) dictionary form (61N6), (2) the prenominal form (6N-; note the conventional use of the single hyphen), and (3) the prepronominal form (6NT). For infinitives of some patterns these three forms are more or less predictable; for others there is much irregularity. We shall deal with the most important patterns individually in subsequent lessons. The forms of the pronouns to be suffixed to a given infinitive are very much the same as those suffixed to the prepronominal forms of the prepositions as given in § 9.1. Other examples will be given as required.

11.2 Infinitives of the pattern x: (i.e. consonant + 1). Infinitives of this pattern show some irregularities, but three of these verbs (41, +, x1) are especially frequent and their forms should be learned:

†	+-	TAA*	to give
ΧI	XI-	XIT"	to take
9 1	4 1 -	917	to raise, carry
g١	ற ≀ −	9 ≀ τ *	to measure.

The verbs c: (to become sated) and e: (to come) are intransitive and do not take direct objects. 41, 91, and c: may also be spelled as 461, 961, and c61; the spelling T61 for † is rare.

Object pronouns are attached to these verbs as

follows:

XIT	ΧlTN	TAAT	TAAN
ХITК	XI-THYTÑ	ΤλλΚ	† -тнүт ग
хітє		TAATE	
¥ιτΨ	жітоү	PAAT	ΤλλΥ
ХITC		TAAC	

Note that THYTN (2nd pers. pl.) is attached to the prenominal form of the infinitive; the prepronominal form is also found: x_1T^-THYTN .

Vocabulary 11

- gu gu- gur' to measure ($\overline{M}MO'$); to measure out; as n.m.: measure, weight, extent; moderation.
- cı to become sated, filled (with: MMO*).
- κω (1) to put, place, set (mmo*); (2) to leave, abandon (mmo*); κω mmo* mca to leave, abandon, renounce; κω mmo* na* esoa to forgive (someone: na*; something: mmo*).
- eine to bring (אַפּס , to a person: אזי, אַגא); פּוּאפּ פּפּסג to publish.
- $T\bar{N}NOO\gamma$ to send ($\bar{M}MO''$; to a person: $N\lambda''$, G); $T\bar{N}NOO\gamma$ $\bar{N}C\lambda$ to send for.
- Me to love, come to love $(\overline{M}MO^*)$.
- MICE to bear (a child: $\overline{M}MO^*$); as n.m.: giving birth; offspring, progeny, one born.
- gone to fall ill, become sick; as n.m.: sickness, disease.
- page to rejoice (at, over: ϵ , $\epsilon \times \overline{N}$, ϵ_2 par $\epsilon \times \overline{N}$); as n.m.: joy, gladness.
- те.уфн night (оуфн).
- n. 2AT silver, money, coins.
- пе. двир, те. двеере (pl. ме. двеер) friend, companion.
- ємьтє (adv.) very, very much, exceedingly; also ммьтє.

- А.1. АУТААТ БТООТБ.
 - 2. ANDITE NAY.
 - 3. AIXITT BTOOTOY.
 - 4. ATSTNTAA9 NAN.
 - 5. AYXITOY 6BOX NEHTC.
- 6. AK41T 62PA1.
- 7. AIGITOY NHTN.
- 8. AITANY NAC.
- 9. APPI-THYTH 680X.
- 10. AYTAAT6 6TOOT.
- B. Translate. Replace the nominal objects with the appropriate pronominal object.
 - 1. 11-1061K NAC. (AITAA9 NAC)
 - 2. Mn64+-n2AT NA1.
 - 3. $\lambda N + \overline{N} \times \omega \omega M G G T O O T \overline{q}$.
 - 4. 24-1164HI NAY 680X.
 - 5. асф-тефтии нан.
 - 6. ANXI-THOYS NTOOTOY.
 - 7. ATGTNXI-TG161PHNH 680A

- 8. ауфі-тегін.
 - 9. AIGHTOGIK NAY.
 - 10. AKG1-H2AT NAI.
 - 11. A441-HONG GBOA.
- 12. дчяг-пноув, дчишт бвод.
- 13. дсфі наі й пирп.
- 14. држі-пробіте бвох 2М
- С.1. понре ет Псюч
 - 2. nmooy 6T N2HT4
 - 3. newbhp et $n\overline{m}m \Delta q$
 - 4. NET 2N TCYNAFWIH
 - 5. TACONE MN TECOBEEPE
 - 6. ΠΝΟΥΒ ΜΠ H2AT
 - 7. πε200γ MN τεγφΗ
 - 8. 27 петомне
 - 9. e neigi
 - 10. пол И пеломе
- 11. пошс мп нечовеер
- 12. MNNCA RECHICE
- 13. тертии ет 21хм песлос
- 14. ngwhe et $2\overline{N}$ hecopy

- 15. птоотч п печенот
- 16. по 1 № пгът
- 17. п₂оєттє м пафвир
- 18. OYMICE \overline{N} TE RESHI
- 19. ϵ TOOTOY \overline{N} NAGBEEP
- 20. ne 2 POOY N TAMAAY
- $21. \ _{2}$ м икаке й теуфн
- 22. OYB6 NEYGAX6
- 23. 21TOOT9 H UXO61C
- 24. MNNCA HOA 6T MMAY
- 25. **пран М петфоне**
- 26. NPWME ET 222TH4
- 27. 2N 26NMA6IN
- 28. 2N OYCH46
- D.1. ACOONE $\overline{\text{N}}$ 61 TE4C21M6 EMATE.
 - 2. AIKW M THOYE MN TEAT EXN TETPAREZA.

- 3. ANKO N NENGBEEP NOON 2H 11-ME, ANBOK EOH E 11TOOY.
- 4. MANCOC AS AGONS NOT 12 XXO, AGMOY.
- 5. дарафе енате ийма ежи пмоу й прро.
- 6. A 4ME N TEC 2 IME ET HMAY EMATE.
- 7. λ -nnoyte thnooy \overline{M} negnonoc epocy.
- 8. ATTINOOY NEI HERICKOROC NCA HMONAXOC.
- 9. LIGI NAY \overline{M} HOGIK $M\overline{N}$ HP $\overline{\Pi}$.
- 10. RPWM6 AYOI \overline{N} TCWG6.
- 11. \overline{NN} -Q1 $2\overline{N}$ NEGQAX6.
- 12. AYCI, AYTWOYN GBOX 21 TGTPARGZA, AYBOK GBOX.
- 13. LATTINOOY \overline{N} NE210ME MN NEYOHPE EBOX $2\overline{M}$ THE.
- 14. 6TB 6 OY MIG-HNOYTE KW NHTH H NETHNOBE EBOX?
- 15. ANCI \overline{N} NEGGAXE AYW ANTHNOOY \overline{M} MOG GBOA.
- 17. LAGING $\overline{\mathbf{H}}$ neadher cen tei $\overline{\mathbf{H}}$ nmonaxoc.
- 18. Ageine \overline{H} heat gapon $2\overline{N}$ teyon.
- 19. АСМІСЕ Й ПЕСФИРЕ Й ПНЕ ЕТ ЙМАУ.
- 20. Mnechice \overline{M} mod \overline{M} neima.
- 21. A-n2 H2 AA M6 N TOGGEE N NG4X061C.
- 22. ETBG OY MIGGING NA! M NXWWHG?
- 23. AYEINE \overline{N} NEGENTOAH EBOX.
- 24. AYEINE HMOI NAZPAH.
- 25. STES OY AKTHNOOY NOW!?

12.1 The relative form of the First Perfect. When the First Perfect is used in relative clauses, it combines with the relative pronoun into a single unit:

entaicotm which I heard entancotm entakcotm etc. entatetmcotm

еитъресфтМ

бМТЪЧСФТ₩ **б**МТЪҮСФТ₩

ENTACCOTM

витх−проме сотМ

These forms are very frequently spelled with \overline{N} for initial en-, as $\overline{N}T\lambda I$ -, $\overline{N}T\lambda K$ -, etc.

The relative pronoun ent- of the preceding paradigm and et, which was introduced in § 3.1, cannot be preceded directly by prepositions or direct object markers. The real syntactic function of the relative pronoun within the relative clause must be expressed by a resumptive pronoun. The general construction is most clearly understood by "Copticizing" a few English examples:

the man who went + the man who he went npome GNTA 180K

the man whom I saw + the man who I saw him npome GNTA 1NAY 6PO9

the man to whom I gave the money + the man who I gave the money to him

npome GNTA 1 + 12 AT NA9

the boat into which we climbed - the boat which we climbed into it

nxo: GNTANAAG 6P09

the sound which they heard + the sound which they heard it

ne2pooy 6NTAYCOTM 6PO9

This use of resumptive pronouns is required in Coptic in all but a few instances which will be mentioned later on. Similar constructions with er will be treated in Lesson 19.

When a relative clause contains more than one verb, the relative pronoun need not be repeated:

πρωμε εΝΤΑ ΤΟΥΝ ΑΥΘΑ 4 BOK 6BOA the man who arose and left.

12.2 The relative pronoun has the form ere before the negative First Perfect:

проме ете мпечпот евох the man who did not flee мжооме ете мпечетие ммооу the books which he did not find

12.3 As mentioned in § 3.1, all relative clauses in Coptic may be substantivized by prefixing the appropriate form of the definite article. Resumptive pronouns are required. Study the following examples carefully:

the one who (or: he who) went

the one (m.) whom they sent

the one (m.) to whom I gave the money

that (m.) which I took from you

those who took it (m.)

the one (f.) whom they entrusted to us

nentage netage n

12.4 Infinitives (cont.). With the exception of the verbs treated in § 9.2, monosyllabic infinitives of the pattern consonant + vowel are relatively uncommon and do not constitute any sort of unified class. Some of these verbs are very important, however, and their forms should be learned:

CW	c e-	coo*	to drink
κω	KX-	KAA*	to put, place
6 I W	6 l A-	6177,	to wash
xω	xe-	x00*	to say
M6	M676-	мер і т	to love

Some verbs of this type are intransitive and do not take direct objects: e.g. Moy (to die), NA (to pity), gA (to rise: of the sun etc.), 2e (to fall), 6w (to remain).

12.5 When introducing a direct quotation, the verb x_{00} requires a "dummy" object (it: $\overline{M}MOC$, -c) followed by the conjunction x_{0} , as in

AGENOR TO MINAY GROY. He said, "I have not seen him." With this particular verb the alternate object form xo MMOC is not permitted in the First Perfect. xw may, of course, have a real direct object otherwise:

> I did not say these things. MILIXETNAL.

Vocabulary 12

cw ce- coo' to drink (Μmo'); often with esox 2N in partitive sense (drink some of); as n.m.: drinking, a drink. 610 611- 6111 to wash (MMO"); + 6801 idem.

xw xe- xoo' to say (Mmo').

Na to have pity, mercy (on: אבי, צא); to pity; as n.m.: mercy, pity, charity.

w or ww to become pregnant, to conceive (MMO').

φλ to rise (of sun etc.); as n.m. rising.

60 to stop, cease, come to a stop; to delay, tarry.

6ω ε to wait for; 6ω MN to wait with, stay with.

n.20 face.

T.OYEPHTE foot.

n.ph the sun.

п. дпот (pl. м. дпнт) сup.

n. w hair.

n.maaxe ear.

n. epore, r. epore milk.

xe (conj.) introducing direct quotation.

- А.1. пра витачропе мизу
 - 2. RESPONDE SNTA42MOOK SXW4
 - 3. HEIW ENTLIASE EXWY
 - 4. HEAT HTALTANG NO
 - $^{5}\cdot$ NE2TOOP ENTANXITOY $\overline{\mathtt{N}}$ TOOT $\overline{\mathtt{K}}$ 12. TEC2IME ENTACO AYO
 - 6. Понре витаумооре Псоч

 - 7. товере ентасоюне

- 8. **ПНР** П 6NTA- 112 XAO COO9
- 9. TEMTHN ENTACEIAAC
- $10. \overline{N}_{0}$ X6 6NTA9X00Y
- 11. NPOME ENTANNA 2APOOY
- - **ACHIC6**
- 13. HMA GNTA-MMONAXOC 60 No HTG

- 14. **ngwbhp** entay6w 6P04
- 15. $\pi_2 \overline{\lambda} \lambda O$ ENTATET $\overline{N} 6 \omega$ EPO4
- 16. ng2 to entaktaag nai 680a
- 17. пхооме ентаркази ехп тетрыпехы
- 18. NA NOT NTAICW 680A N2HT9
- 19. TEPWTE NTAHCW EBOA N2HTC
- 20. NEGREEP NTANKAAY NCWN
- 21. праже ентачраре ежрч
- 22. TEC21M6 ENTACE1A-NEGOYEPHTE
- 23. HOW ENTAICE MMOS

- 24. HEAT GNTA-BENKOGIC MTHM PTIO
- 25. проме Птакко нач П NEGNORG GROA
- 26. Проме витачет фарок гй T 6 YOH
- 27. NECHHY ENTAGMEPITOY
- 28. HPOME ENTAKNAY 6 H6420
- 29. Приме вита-пепіскопос ΦλΗλ ΕΧΦΟΥ
- 30. MAGHTHC ENTAYMOT NCW4
- B.1. NENTAYED $2\overline{n}$ \overline{n} the et \overline{n} hay 9. Nentay $2\overline{n}$ 6 20 yn 6 7 \overline{n} 7 1

 - 3. HENTAMNA HALLAYO ANT HALL 11. TENTAMMEPITC N OYOGIK
 - 12. NENTAYETA-NEY2OETTE
 - 13. нете тпоусотт в полже
 - 4. **ПЕНТАЧЖЕ- НЕ 10 ДЖЕ** 5. тентасю ймог 14. $\pi_{\text{NNTAIXOO4}}$ NHTN

 - 6. NENTANGO GPOOY
 - 7. пентачсе-пероте
- 15. Νέτε ΜπεατΝΝΟΟΥ ΜΜΟΟΥ
- 16. NETE MNOYEINE MMOOY EBOX
- 8. NENTAYBOK EBOX & THOXIC
- C.1. ACEIO H NEUZO MN NEUMAAXE.
 - 2. A4X00C X6 MINAY 6P04.
 - 3. MNNCA NAL AG ACO NGI TECZIME.
 - 4. Ачріме етве нечовеєр ентаумоу 21 тегін.
 - 5. π₂ λλο λ6 Μπε 4 C ω ε BOλ 2 Μ π Η Ρ Π .
 - 6. MNNCA MOA M MPH ANBOK 6BOA.
 - 7. λ-τωεερε ειω π πεсчω ¿π πμοογ π πειερο.
 - 8. A- HPH 61 62 PA 1 6XM HTOOY.
 - 9. 6TB6 OY MHETHELA-NETH20?
- 10. AYXOOC NAN X6 MR6NXITT.
- 11. ANGO NAMAY 2N TEYOH ET MMAY.
- 12. Ачфаже имман етве неченну ете мпочет иммач.
- 13. λ π PH $\phi\lambda$ λ YW λ MTWOYN, λ MBWK ϵ BO λ .
- 14. AICOTH 6 NGIGAXE 2N NAMAAXE.

13.1 The Temporal. A special conjugation is used to express a subordinate temporal clause (English "when" with a simple past or pluperfect verb), as in

MT6P1NAY 6P09, AIROT 9AP09. When I saw him, I ran to him.

NT6P69XWK Й П692WB 6BOA, A9BWK 6BOA. When he had completed his work, he left.

The full inflection of this form, called the Temporal Conjugation, is as follows:

NTEPICOTHwhen I heardNTEPICOTHNTEPEKCOTHetc.NTEPETNCOTH

Птересфтя

ПтеречсютМ

МтерессютМ

 $\overline{\mathsf{N}}\mathsf{T}\mathsf{e}\mathsf{p}\mathsf{e}\mathsf{-}\mathsf{n}\mathsf{p}\mathsf{o}\mathsf{m}\mathsf{e}$ cot $\overline{\mathsf{m}}$ when the man heard

Alternate spellings, such as \overline{N} Tepel-, \overline{N} Tep \overline{K} -, \overline{N} Tep \overline{K} -, are common.

The Temporal is negated by prefixing - $\tau \overline{\mbox{\scriptsize M}}$ - to the infinitive:

With a nominal subject, $-\tau \overline{M}$ - usually remains with the verbal prefix:

A Temporal clause usually stands before the main clause, but occurrences after the main clause are not rare:

NTEPICOTM 6 NEGGAX6, AIPING. I wept when I heard his words.

When a Temporal clause is continued with a second verb, the Temporal prefix is not repeated and the First Perfect is used: NTEPICOTM 6 RETEPOOY AYOU AINAY 6 RETEO... when I heard his voice and saw his face...

13.2 Relative clauses with ne, те, ме. Contrast the two sentences:

א-חאפוסד אאאד אואץ. My father left me there.
חאפוסד הפ פאדאקאאד אואץ. It was my father who left me
there.

The second sentence, known in English as a type of cleft sentence ("it was . . . that/who . . .), singles out the subject ("my father and no one else") as the actor, while the first sentence merely describes a past action with no special emphasis. The Coptic correspondent of the English cleft sentence employs no, to, we followed by the relative form of the verb. no, to, we usually combine with the relative form to produce nontai-, nontak-, etc.:

павіют пвитачкаат ММАУ.

This form should not be confused with the nominalized relative nentalcot \overline{n} etc., which consists of the definite article plus the relative form (see § 12.3). Contrast

תגפוסד הפאדגיקאד אוויץ. It was my father who left me there.
הואפוסד הפ הפאדגיקאד אוויץ. My father is the one who left
me there.

The second sentence is a normal ne sentence: n_{AE100T} is the subject, $n_{ENTAYKAAT}$ \overline{M}_{MAY} is the predicate.

The cleft sentence with \mathfrak{n}_{6} is a favorite one with the interrogative pronouns:

мім пвитъчтпиооу мнок? Who sent you? Who was it that sent you?

ΟΥ ΠΕΝΤΆΚΤΑΙ ΝΑΥ? What did you give to them? What was it that you gave to them?

Note that in this case the English cleft sentence pattern ("who was it that") is slightly different from the one

given above.

13.3 The preposition 6 is used before an infinitive to express purpose:

ыет в фаже иймак. I have come to speak with you.

The subject of the infinitive in this construction is usually the same as that of the main verb, but some laxness occurs, as in

13.4 Coptic has no real passive conjugations. The passive is expressed by using the 3rd pers. pl. of the active form in an indefinite sense:

аут Nooy Mmoi gapok. I have been sent to you.

That such expressions are to be taken in a passive sense is most evident (1) when there is no clear reference for the pronoun "they," or (2) when an agent is added, usually with 680λ 21 $7\overline{N}$, as in

λΥΤΝΝΟΟΥ ΜΗΟΙ 6ΒΟλ 2ΙΤΗ ΠΡρο. I have been sent by the king.

13.5 Infinitives (continued). Transitive infinitives of the pattern $\kappa\omega\tau$ have the following forms:

kerker-kor*to buildxekxek-xok*to completezenzen-zen*to hide, concealen-on*to count.

Transitive verbs of this pattern are quite common.

Vocabulary 13

- xok (forms above) \pm 680% to finish, complete ($\overline{\text{MMO}}$); as n.m.: end, completion.
- 2001 (forms above) to hide, conceal (HMO').
- on (forms above) to count (Мно"); to esteem, have regard for (мно"); to ascribe (someone or something: мно";

to: 6), to reckon as. BWA BGA- BOA' (1) to loosen, unfasten, undo (MMO'); (2) to interpret, explain $(\overline{M}MO^*)$; BOX GBOX = (1) and also: to nullify, annul (MMO'). TWM TEM- TOM' to close, shut (MMo'). ογωμ ογωμ- ογομ' to eat (Μηο'; partitive: 680λ 2N); OYOM NCA to gnaw at. gon gen- gon' (1) to receive, accept (MMo'; from: NTN); (2) to buy (Mmo'; for a price: 2). oyon to open (Mmo', 6). ת. אים meat, flesh (human or animal); piece of meat. ne.yzop (f. Te.yzwpe; pl. Ne.yzoop) dog (oyzop). n. BAA eye. n.way time, hour. n. Noya rope. Greek nouns: τε.χηρα (ή χήρα) widow. π.ορφωνος (δ δρφανός) orphan. τ.capa (ή σάρξ) flesh. τ. ηγλη (ἡ πύλη) gate. Proper names:

David (sometimes abbreviated ALA). өнөроүсланы Jerusalem (with def. art.), regularly abbreviated eTXHM.

- PTOOTH PROPIATION TONAL 7. THYAH GHTAYBOK GBOA 2. HEHMA GNTASTWOYN GEOX 21009 2 I TOOTC 3. news gntagek Mmog 680a 8. пашв витакфаже гароч 4. пні битаукот Т ТМАУ 9. TEXHPA ENTAIT NAC H 5. HNOYB ENTANZONT 2A HONE n2 AT 6. HONE ENTAGE ENECHT EXT 10. πεχλο εκτλγεφη εεογη ε NAOYEPHTE T69P1
- B.1. Albok 6 PAKOTE 6 NAY 6 DAGIOT.

- 2. AYEL 6 GAXE NHMAL.
- 3. ARBOK 6 NGIGPO 6 61A-NG42061T6 6BOA.
- 4. ALZMOOC 6 CW N TEPWT6 AYW 6 OYWM M HA9.
- 5. AYTHNOOY HMOI 6 GING NAK H REIZOOME.
- 6. AGEING W OYMOOY 6 GIA-NEGOYEPHTE MMOG.
- 7. A 4 T WOYN 6 B WK 6 MA 2 OY 6 9 T A HM.
- 8. NIM MENTAYXOK 680A N M61208?
- 9. NIM NENTAY20NT 2H NH1?
- 10. ΝειέΝτολοογέ ΝέΝτλγβολογ έβολ.
- 11. TEXHPA TE $\overline{\text{NTAY}}$ NAC $\overline{\text{M}}$ NOGIK.
- 12. NIM NENTAMBOA NHTH $\overline{\mathbf{M}}$ RECOUME?
- 13. Пточ пентачтом П тпуан.
- 14. NAI NG ENTAIGONT TTOOTOY.
- 15. NEYZOOP NE ENTAYOYOMT.
- 16. nexpictor nentagoyon \overline{N} nabax.
- 17. NIM NENTAYONOY 6 π +M6 ET $\overline{\mathbf{H}}$ MAY?
- 18. nca2 nentagramy nai.
- 19. RELACIACIOS RENTA-RAFFEROS XOOS NAS.
- 20. OY $\pi \in \mathbb{R}^{3}$ $\pi \in \mathbb{R}^{3}$ $\pi \in \mathbb{R}^{3}$
- 21. HEIFHE HE NTAYKOTT NEI NENEIOTE.
- 22. петроме пентасфаже иймач.
- С.1. Птере-пмонахос жок й печгов евох, ачтооун, ачвок евох.
 - 2. Answ NHMA4 ga nga $\overline{\mathbf{H}}$ nph.
 - 3. $\overline{\mathsf{N}}\mathsf{T}\mathsf{G}\mathsf{P}\mathsf{G}\mathsf{-}\mathsf{\Pi}\mathsf{P}\mathsf{H}$ $\mathsf{Q}\mathsf{A}$, $\mathsf{A}\mathsf{N}\mathsf{A}\mathsf{A}\mathsf{G}$ $\mathsf{G}\mathsf{X}\overline{\mathsf{N}}$ $\mathsf{N}\mathsf{G}\mathsf{N}\mathsf{2}\mathsf{T}\mathsf{D}\mathsf{D}\mathsf{P}$ $\mathsf{A}\mathsf{Y}\mathsf{D}$ $\mathsf{A}\mathsf{N}\mathsf{\Pi}\mathsf{D}\mathsf{T}$ $\overline{\mathsf{N}}\mathsf{C}\mathsf{D}\mathsf{G}\mathsf{-}\mathsf{N}$
 - 4. AYONT 6 HHI N AAY61A.
 - 5. Птерессоты в наг, аскі-песфире, асгоня.
 - 6. AYTOM NEI NPOME N MNYAH N THOAIC.
 - 7. NTEPIEN-MEAT, AITAAN NAY.
 - 8. Mnoywn N Nwaxe H nenxoeic.
 - 9. ACOUNG AS NTSPOYOYON N THYAH N THOAIC, A-HAAOC HOT SEOA 21 TOOT \overline{c} .
- 10. Ντεριεωκ, **λιχι Η ΠΝΟΥ2 ΝΗΜ**ΑΙ.
- 11. Птерпсоти е полхе ентачхооу, анрафе енате.
- 12. AGONN & NAME N STARM.
- 13. NTEPOYTHEINE N NEWOME ENTANZONOY 2H THI, AYBOK BEOA.

- 14. A-N6Y2OOP OYOM H NA9.
- 15. \overline{N} T G P G 4 2 G G P O O Y, A 4 B W A G B O A \overline{N} N G Y N O Y 2 .
- 16. NTEPINAY EPO4, AINA 2APO4.
- 17. \overline{N} TEP64XWK EBOX \overline{N} NEIGAXE, A4TWM \overline{N} NE4BAX, A4MOY.
- 18. NTEPENNAY EPOOY, ANXOOC NAY XE ETBE OY ATETNEL 6
- 19. A-REYZOP OYOM NCA TCAPA M REID.
- 20. ACGWHG AG NTGPG-HNAY M HECMICE 2WN 620YN, ACBWK 6 HHI N TECMAAY.
- 21. $\overline{\text{N}}$ Тероунау є поуобін $\overline{\text{M}}$ печго аую аусют $\overline{\text{M}}$ 6 неч ϕ ахє, аугє ϕ пкаг, ауге ϕ -н ϕ уго.
- 22. GTBG OY ATGTNEWA GBOA N NENTOAH N NETNGLOTE?
- 23. Птерестязь в тесфевре имау, асріме.
- 24. Птеречтноуми н про П течрі, дівок свол.

14.1 The Second Perfect. As we shall see in subsequent lessons, each "first" tense in Coptic has a counterpart called a second tense, the use of which places a special emphasis) on some element of the sentence other than the verb, usually an adverbial phrase. Contrast the following:

First Perfect: a-nal gone etehhtk.

This happened because of you.

Second Perfect: אדם-חשו ששחה הדוודה.

It was because of you that this happened.

As our translation indicates, the English cleft sentence is a handy way to render Coptic sentences with second tense verbal forms. Except for the special uses taken up below, the use of a second tense is not obligatory but depends on what the writer chooses to emphasize.

The Second Perfect has the same inflectional forms as the Relative of the First Perfect, but usually without the initial 6: NTAICOTH, NTAKCOTH, etc.

When phrases containing interrogative pronouns or adverbs are placed after the verb, a second tense is regularly used, but exceptions are not rare:

אדאגדאאססץ אאסט פדאה סץ? Why did you send him? אדאים אוא דפופחוכדסאו אוא? To whom did he bring this letter?

But if the interrogative phrase is place first, as is usually the case with GTBG OY, the first tense is used: GTBG OY AKTNOOY MMO4?

14.2 Further remarks on interrogative pronouns and adverbs. The interrogative pronouns NIM (who?) and oy (what?) may be used as subjects or objects of verbs and as objects of prepositions. When they are used as the subject of a verb, the verb is normally in the second tense form:

NTA-OY gone? What happened?
NTA-NIM BOK 620YN? Who went in?

Examples of object usage, again regularly with the second tense:

אדאאאא פ אוא?Whom did you see?אדאאאא א אוא?To whom did you give it?אדאאאארם אוא?What did he put there?

The construction introduced in §13.2 is used much more frequently than the preceding: NIM NENTAMBOK 620YN? NIM NENTAKNAY 6P04?

The interrogative adverbs <u>των (where?)</u>, <u>6 των</u> (whither?), <u>680λ των (whence?)</u>, and <u>τπνλγ</u> (or τνλγ, when?) occur regularly in <u>post-verbal position</u> with a second <u>tense</u>:

NTA-NEKELOT BOK 6 TON? Where did your father go?
NTAYMOY THNAY? When did they die?

14.3 Infinitives (continued). In infinitives of the type $\kappa\omega\tau$ the ω is modified to $o\gamma$ when the initial consonant is m or m:

moyp mep- mop' to bind moyn --- to remain.

The o of the presuffixal form i regularly replaced by a before stem final 2 and (usually) g:

OYW2 OYG2- OYA2' to put, place OYW9 OYG9- OYA9' to want, desire MOY2 MG2- MA2' to fill.

Vocabulary 14

- моүр мөр- мор' to bind, tie (someone: \overline{M} or suff.; with: \overline{M} мо', $2\overline{N}$; to: 6, $6 \times \overline{N}$, 620 у 6).
- NOYXE (Or NOYX) NEX- NOX' to cast, throw (MMO'; at, into:
 e); NOYXE EBOX to discard, throw away, abandon; NOYXE
 enecht to cast down.
- oywz oyez- oyxz* (1) to put, place, set (Μμο*); oywz exN to add to, augment; (2) intrans.: to settle, dwell, reside (in: 2N; with: μΝ).
- oyog oyeg- oyag to want, wish, desire (ΜΜο΄); as n.m.:
 wish, desire; Μ πεσογωσ of his own volition, as he
 wished. oyeg- may be compounded with another infinitive: oyeg-e: to wish to come, oyeg-coth to wish to hear.
- моуг мег- мъг' ± евох (1) to fill (something: ммо' or suff.; with: ммо', гм, евох гм); (2) intrans.: to become filled, full (of, with: ммо'). An indefinite noun after ммо' (that with which something is filled) normally has no article.
- † ммо° 21 to put (a garment: ммо°) on, to dress. ne.cnay2 bond, fetter.

пе. отеко (pl. не. отекооу) prison.

т.рметн (м.рметооуе) tear(s).

T.61x hand.

Twn (adv.) where? 6 Twn whither? 6800 Twn whence? $T\bar{N}N\lambda\gamma$, $TN\lambda\gamma$ (adv.) when?

пе.схних (Gk. τὸ σχήμα) fashion of dress; monk's habit;
моγр ммо м песхних to garb someone in a monk's
habit, to accept into monkhood.

Exercises

- 1. Μπογογωφ ε εων εεογν ερον.
- 2. NTA-REKEIOT MOY THNAY?
- 3. ACOUNTS AS NTSPIXOK SBOA M HAZOB, ALTOOYN, ALBOK SBOA.
- 4. NIM ПЕНТАУНОЖЯ В ПЕФТЕКО? ПЕНСА2 ПЕ.
- 5. **λ-**₩πηγε ΜΟΥ2 ₩ ΟΥΟ6ΙΝ.
- 6. Итачет еголи е иеткосмос и иелолюф.
- 7. ALBOX GBOX \overline{N} NECHAY2 ENTAYMEP-TPOME \overline{N} 2HTOY.
- 8. NTAK+ H NECKHMA 2100K THNAY?
- 9. NAI NE NOAME ENTAYOUNE M NEZOOY ET MMAY.
- 10. NTAKOYWE EN TERNOLIC THAY?
- 11. Асфоне де Птересге ероч, асчітч, асвок евол ийнач.
- 12. NTAP2OROY TON?
- 13. LYMOPT \overline{N} NLOYEPHTE \overline{N} NLEIX LYW LYKLLT \overline{M} NLY 21 HEKPO.
- 14. NIM ПЕНТАТЕТ $\overline{\mathbf{n}}$ \mathbf{g} \mathbf{e} \mathbf{n} \mathbf
- 15. THE THOY ENTRYOYAGT.
- 16. Мпечоуюг ехм пгат ентачталч нау.
- 17. MNNCA NAI A6 A9MOY2 N PAG6.
- 18. λ -текканста моуг \overline{N} негрооу \overline{M} пиннуе.
- 19. йты-иім кым гй пертеко?
- 20. NTATETHNEX-NEI206ITE EBOX ETBE OY?
- 21. $\lambda = \overline{N}$ pome moyp \overline{M} mmonaxoc ϵ nesto.
- 22. A-N69BAA MOY2 N PH61H.
- 23. Птачоующ в нау в нім?
- 24. NTA-NETNEIOTE 26 EBOA ETBE NEYNOBE.
- 25. MNNcoc A6 ANGI 620YN 6 TCYNAFOTH.

- 26. λ-πεγχοι MOY2 H MOOY.
- 27. OY $\pi \in NT \times KOYOM \overline{M} \rightarrow \pi \cap OOY$?
- 28. NTEPITOM H HPO, AIRMOOC AYO AION H HRAT GHTAYTAA4 HAI.
- 29. x-Ney200p oywm Ncx Negoyephte.
- 30. HAI HE HONE ENTAYNOUT EBOA.
- 31. $\overline{N}T\lambda = n\omega\lambda$ $\omega\omega n\varepsilon$ $\tau\overline{N}N\lambda\gamma$?
- 32. NIM HENTARBOA EBOA \overline{N} NEICHAY2?
- 33. λ -nenickonoc mop \overline{N} \overline{N} neckha.
- 34. \overline{N} TANKA-HEN+ME \overline{N} CON 6TB6 H62KO.
- 35. NIM NENTAHOYA 2 K 2 H NEIMA?
- 36. ACOY62-TECMEEPE EXM n66A06.
- 37. x-N649xx6 MX2OY N PAG6.
- 38. NTA-NEIZICE ZE EXWN ETBE OY?
- 39. \overline{N} TAYOYON \overline{N} THYAH \overline{N} THOALC THNAY?
- 40. OY $\pi \in \mathbf{NTA} = \mathbf{N} \oplus \mathbf{NTA} = \mathbf{NTA} \oplus \mathbf{NTA} = \mathbf{NTA} \oplus \mathbf{NTA} \oplus \mathbf{NTA} = \mathbf{NTA} \oplus \mathbf$
- 41. λ 1 NOYX6 \overline{M} NNOY2 enecht e π K λ 2.
- 42. Мпочочеф-сфтм в мафаже.
- 43. Мпінау є технра фа пооу.
- 44. A10YOO 6 NKOTK.

Lesson 15

15.1 Adjectives. Although there is some debate over the existence of adjectives as a grammatical category in Coptic, it is nevertheless convenient to retain the designation for the words treated in this lesson. Most attributive adjectives may either precede or follow the noun they modify, joined to the noun with a linking particle \mathbb{N} (\mathbb{N}). The noun and adjective form a close unit; any article, possessive adjective, or demonstrative stands before the whole unit:

оуноб м поліс, оуполіс N ноб a large city
памеріт N дире, падире м меріт my beloved son
псаве N рюме, проме N саве the wise man

These examples represent the normal attributive adjective construction. There are, however, some restrictions on certain special groups of adjectives:

- 1) Several adjectives show a distinct preference for the position before the noun in the given construction. These include NO6 great, KOYI Small, MHM Small, MOPR first, 226 last, and MEPIT beloved.
- 2) A few adjectives may be used after a noun without the linking N. These include ac old, No6 great, KOYI small, GHM small, OYOT single, OYOBO white. With the exception of a few fixed expressions, this construction is rare in standard Sahidic and should not be imitated.

Some adjectives have distinct feminine and plural forms; e.g.

каме	fem.	Камн	p1.		black
5 y 6		HKS		5766Å(e)	last
C786		CABH		C7866A(6)	wise
6776				67786Å(8)	lame
неріт				нерате	beloved
9 0 ρ π		90 P 11 6			first

20Y61T20Y61T620YAT6firstΦΜηΟΦΜηΟΦΜηΟalien, foreign

The plurals in -6676 also occur as -667. The fem. forms are used with sing. and plural fem. nouns. The plural forms occur mainly in substantivized usage: \overline{N} CABGEY6 the wise, \overline{N} 207AT6 the elders, prominent persons (e.g. of a city).

Greek adjectives may appear (1) in the Gk. masc. sing, form with nouns of either gender or number:

npwme \overline{N} araboc the good man the good woman

or (2) in the Gk. fem. sing. form if the modified noun refers to a female person:

TEC21M6 N AFAGH the good woman

or (3) in the Gk. neuter form with nouns of either gender if they denote non-humans:

τεψγχη \overline{N} τελειοη the perfect spirit.

Greek substantivized neuter adjectives are treated as masculine in Coptic:

παταθον good, that which is good (τὸ ἀγαθόν).

A noun may be modified by more than one adjective, with various orders:

проме \overline{N} 2 нке \overline{N} діклюс the righteous poor man плое \overline{N} \overline{p} po \overline{N} діклюс the great (and) righteous king.

All Coptic adjectives may be substantivized ("one who is . . ., that which is . . .") by prefixing the appropriate form of the article:

mesihn the poor man Nesihn the poor (people)
mesihke this poor man Telihke this poor woman
oycase a wise man 26NCAseGye wise men

When the first noun in a genitive construction is followed

by an adjectival phrase, \overline{N} TG may optionally be used instead of \overline{N} for the genitive:

понры N балы NTE промы the man's crippled child.

15.2 Adjectives as predicates are treated exactly like noun predicates. Note the obligatory use of the indefinite article:

OYATAGOC NG.

N 26NATAGOC AN NG.

They are not good.

The man is good.

NPWMG 26NAIKAIOC NG.

The men are just.

OYNOG TG TGINOAIC.

This city is large.

15.3 The cardinal numbers from one to five are

one	masc.	OYX	fem.	0761
two		CNXY		сйте
three		фомπτ		донт 6
four		чтооү		<u>4706</u> , 470
five		toy		ተε, ተ

The numbers from three upward stand before the noun with the adjectival $\overline{\mathbf{w}}$. The noun is in the singular form, as is the definite article when present:

фоныт и жої three ships
пфоныт и рро the three kings
тегромте и ромпе these three years

Note the absence of the indefinite article in the indefinite expressions.

The number one is construed in the same way, but the linking $\overline{\mathbf{w}}$ may be omitted:

(n) OYA POME, (n) OYA \overline{N} POME (the) one man.

The number two follows its noun, which is likewise in the singular; no \overline{N} is used:

con cnay, ncon cnay two brothers, the two brothers cone cnte, the two sisters, the two sisters.

Vocabulary 15

NO6 large, great, important.

κογι small, little; also of quantity: a little (e.g.

оукоу: \overline{N} об:к a little bread); with pl.: few (e.g.

генкоу и жооме a few books).

NEPIT (pl. MEPATE) beloved.

CABG (f. CABH; pl. CABGGYG) wise.

6226 (pl. 62266y6) lame, crippled.

2HK6 DOOr.

выни poor, wretched, miserable.

те. ромпе (pl. N. pmnooye) year; (N) тромпе this year.

 \overline{N} оуроние for a year. \overline{N} gomme \overline{N} pointe for three years. n.esor (pl. N.esare, N.esere) month.

п. 9 нре 9 нм small child (a frequent fixed expression). $n\omega_2$ to reach, attain (ϵ , 9 λ).

Greek adjectives:

Araeoc (áyaðóc) good.

Αικαιος (δίκαιος) just, righteous.

πιστος (πιστός) faithful, true, believing.

λπιστος (άπιστος) unbelieving.

πονηρός) bad, wicked.

And the numbers given in the lesson.

Exercises

A.1. OYXHPA N 2HKE

2. TEINOS M TYAH

3. OY2 M2 AA M HICTOC

4. OYPPO N AIKAIOC

5. $\pi_{61\lambda\lambda OC} \overline{N} \lambda \pi_{1CTOC}$

6. OHZAA M HONHPA

7. OYNOS \overline{N} 2HFEMON

8. π menmepit $\overline{\mathbf{N}}$ elot

9. TROY! N TET

10. OYEARE N 2HKE

11. OYKOYI N A4

12. OYZ XXW N CABH

13. нечмлентис м пістос

14. ne2 to N 6226

15. 26N210M6 N 6B1HN

16. TAIKAIOC \overline{N} POME

17. TAMEPIT H MAAY

18. OYKOYI \overline{N} OTEKO

19. негвнуе й поннрон

20. NOAKE N NCABERYE

- 21. **neinoe** N nomoc
- 22. HCABE N AIKAIOC
- 23. MINAPOENOC N CABH
- 24. неимерате й фире
- 25. пминфе й эпістос
- В.1. фомыт ы жол
 - 2. DOMTE \overline{N} GOTHN
 - 3. **петфомыт и 500**Å
 - 4. 4TOOY N 2061TE
 - 5. **neutooy** N tooy
 - 6. TOG N XHPA
 - 7. TELYTOE \overline{N} ENTOLH
 - 8. OYA H MONAXOC
 - 9. 0Y61 M TYAH

- 26. печфире й бале
- 27. NPM6100Y6 N N2HK6
- 28. Мпоннос мп паганос
- 29. очевіни п орфанос
- 30. TEC21ME M RICTOC
- 10. CHAY2 CHAY
- 11. ПЕЧВАХ СНАУ
- 12. POMRE CNTE
- 13. GBOT CHAY
- 14. neitoy N ebot
- 15. †oy N KOYI N XOI
- 16. TOY N PWME N ATABOC
- 17. C2 IME CNTE N AFABOC
- 18. nppo cnay
- C.1. NTAKHW2 6 TELHOLIC THNAY?
 - 2. AYKWT N OYKOYI N HOLIC MMAY.
 - 3. ANOYWE HMAY N 4TOS N POMNS.
 - 4. NTAP+ NMMA9 6TB6 OY?
 - 5. OY HE HPAN \overline{M} H † ME ENTATETHIND? EPOS \overline{N} TEYOH ET \overline{M} MAY?
 - 6. NIM MENTAMOINE NOWI?
 - 7. па пе пегропу $\overline{\mathbf{n}}$ пенмеріт $\overline{\mathbf{n}}$ фире.
 - 8. AI+ 所 N2AT N NGBIHN.
 - 9. $\overline{\text{M}}$ neare $\overline{\text{M}}$ neare et $\overline{\text{M}}$ may.
 - 10. OY HE HOL ENTARHOUS SPOR $\overline{ ext{N}}$ SI HELZ $\overline{ ext{N}}$ AO $\overline{ ext{N}}$ ALKALOC?
 - 11. NTAKNOXOY 6BOA TON?
- 12. ALGO 2M NGYTHG N OYPOHNG.

Lesson 16

- 16.1 The interrogative pronouns x_0 , oy, and NIM may be used adjectivally. This usage is most frequent in certain fixed expressions, the most important of which are
 - 1) AD M MINE (of) what sort? This phrase is used attributively, as in

ογλg \overline{M} MING \overline{N} xoι? what sort of ship? or predicatively (note obligatory use of indefinite article):

ογλφ M mine ne neipome? Of what sort is this man?

- 2) λg N 26 (of) what sort? N λg N 26 in what way? how?

 Ογλg N 26 ne neimλειη? Of what sort is this sign?

 Ν λg N 26 λκεινε Μμος? How did you find him?
- 3) ${}_{2}\overline{N}$ ${}_{3}\overline{D}$ \overline{N} oyoeig? at what time? Similar use of oy and Nih is rarer, e.g. Nih \overline{N} pome? what man? oy \overline{M} Mine? what sort? In special contexts these same or similar expressions may have an indefinite value: Nih \overline{N} pome such and such a person, ${}_{3}\overline{D}$ \overline{N} the some village or other, oy ${}_{3}\overline{N}$ oy this and that.
- 16.2 "Each, every" is expressed by NIM (not the same word as NIM who?) placed after a singular noun with no article: Pome NIM every man, everyone; 2008 NIM everything; the NIM every village. Pronominal resumption is usually in the plural:

ема и и и ситансот ерооү everything which we heard гов и и й поинрои ситачегре йнооү every evil thing that he did

But resumption in the singular is not rare.

anyone, anything. These are most frequent in negative contexts as "no one, nothing":

He gave me nothing.

אגגץ also appears with the indefinite article: סאָגגץ.
אגגץ is often used adjectively:

мпе-אבג א роне нау ерог. No man saw me. мпгреп-заху м жооме мтоотч. I received no book from him.

When (oy) and or phrases beginning with (oy) and are direct objects of transitive verbs (i.e. object with Mmo'), the use of the prenominal form of the infinitive is obligatory in the First Perfect and its negative. Thus Mnigwn N ally... is not permitted in the sentence above.

As a nominal predicate xxxy means "nothing," even when no negative is formally involved. The indefinite article is obligatory:

I am nothing. 26NAAAY NG NGYNOYTG. Their gods are nothing.

 \overline{N} Name or Name alone may be used adverbially in the sense "(not) at all":

 \overline{M} האבע \overline{M} אינגע. I didn't speak with him at all. Note also the expression of on Nim everyone, everybody.

16.4 "All, the whole (of)" is expressed by THP used in apposition to a preceding noun or pronoun. A resumptive suffix is required:

Npowe thpoy all the men (lit. the men, all of them) nkochoc thp \overline{q} the whole world, all the world Ayeı ezoyn thpoy. They all came in.

The pronominal suffixes are the same as those used on prepositions and infinitives; the 2nd pers. pl. form is THPTN. The 3rd pers. pl. THPOY may also be used for 2nd pers. pl. reference.

16.5 The numbers from six to ten:

 six
 masc. cooy
 fem. co, coe

 seven
 cagq
 cagqe

 eight
 gmoyn
 gmoyne

 nine
 vit, vic
 vite, vice

 ten
 mat
 mat

They are used like the numbers three to five in §15.3.

Partitive expressions with numbers employ the preposition \overline{N} (\overline{M} Mo°):

The number "one," oya (f. oye) is also used as an indefinite pronoun: a certain one, a certain man (or woman), as in

λ-ογλ вωκ φλ πλρχιεπισκοπος. A certain man went to the archbishop.

Vocabulary 16

- ee (τ.26) manner, way. N ee N prep. like, in the manner of; with pron. suff.: N τλ26 like me, as I do. N τε126 in this way, thus.
- T.MING kind, sort, type, species. $\underline{\lambda}$ M HING of what sort? \overline{N} TGIMING of this sort, such.
- ne.oyoeig time, occasion. \overline{N} oyoeig Nim every time, always. \overline{N} oyoyoeig once, on one occasion (in the past). \overline{N} neoyoeig at this/that time.
- пе.мто євох presence. \overline{M} пємто євох \overline{N} in the presence of; with pron. suff.: \overline{M} па \overline{M} то євох in my presence.

And the words and expressions treated in the lesson. Greek words and names:

τε.χωρα (ή χώρα) land, country.

т. єрнмос (ή έρημος) desert, wilderness.

п.кърпос (δ καρπός) fruit. п.ъркієреўс (δ άρχιερεύς) мωусис (Μωυσής) Moses. high-priest.

π.λγχιεπισκοπος (δ άρχιεπίσκοπος) archbishop.

Exercises

- A.1. 2061NG N TGIMING
 - 2. COOY N ECOOY
 - 3. n2xt thp4
 - 4. OYA N NGALEGYE
 - 5. OYAG H MING N 600Y?
 - 6. N CAOT N 200Y
 - 7. TEYOH THE
 - 8. 208 NIM GNTAIGONOY
 - 9. OYCHAE \overline{N} TEIMINE
 - 10. NETME THPOY NTE TCYPIA
 - 11. H REMTO GROA H NAPX 16P6YC
 - 12. \overline{N} 96 \overline{N} 0YN06 \overline{N} CABE
- 13. пазат тирч
- 14. POME NIM ENTARNAY EPOOY
- 15. COMTE M MINE
- 16. OYON NIM 6T $2\overline{N}$ TCYNAFOFH 37. \overline{N} OMOYN \overline{N} 680T
- 17. THOMOG M MWYCHC THP4
- 18. co€ N coo 6
- 19. TEIMHTE \overline{N} ENTOLH
- 20. Nexupa throy NTE nelkochoc
- B.1. MINIMALANAY EXN TETPANEZA.
 - 2. OYN-206ING N TEIMING 2M HOALC NIM.
 - 3. NTAKSING M NEKZAN N AQ N 26?
 - 4. OYAQ $\overline{\mathbf{M}}$ MINE HE HELXWWME?
 - 5. 2N AM N OYOGIM ACMICE H песфире?
 - 6. ANT-NIM ANOK? ANT-OYALAY.
 - 7. MITE 4 TOYALAY NAI.
 - 8. энкэ-очон или ысон.
 - $9. \ \text{2ensly ne negges} \ \overline{\text{M}}$ **TONHPON.**

- 21. П схоче п ромпе
- 22. N OF N OY2M2AA M HICTOC
- 23. ANON THPN
- 24. N2HK6 N THOLIC THPOY
- 25. 2N AAAY # MA
- 26. WMOYN N NOG N XOL
- 27. OYA MMOOY
- 28. max6 NIM 6NTA4XOOY
- 29. OYAO H MING N GONG?
- 30. OOC CNAY
- 31. 21XN XXXY N 21H
- 32. м печмто евох
- 33. AAAY N KAPHOC
- 34. 9TOOY M MAGIN
- 35. 6 Au N ut?
- 36. 2N TE4XOPA AYO 2N NE4+ME
- 38. 2N OYNO6 N PAGE
- 39. OYAPXIEPEYC M HONHPOC
- 40. NENGBEEP THPOY
 - 10. A-OYA EL MAPOS 2N теуфн.
 - 11. AC60 MN OYA N N6C-CYTTENHC.
 - 12. etse oy NTAPELPE N T6126?
 - 13. N OYOYOGIQ A-HAPXIenickonoc el e nentooy.
 - 14. AYEINE H HEYIT N POME 620YN 6P04.
 - 15. \overline{N} TAYT \overline{N} NOOY \overline{M} MOI 6 nelaboc thp4.

- 16. Япімере-хаху й пих ет йиху.
- 17. λ -2061N6 DONE EMATE M RECYCLIQ.
- 18. MINGHT-XXXY N OGIK NAN.
- 19. 2N λφ N ΟΥΟ61φ ΔΤ6ΤΝπω2 6 Π61MA?
- 20. MN-XXXY NHMA HMAY.

Lesson 17

17.1 The Imperative of most verbs is the same as the Infinitive, with no indication of number or gender:

MOOD6 NCW1. Walk behind me.

Mer-nxol e nome. Tie the boat to the rock.

Mere-nxosic. Love the Lord.

сютй є нафаже. Listen to my words.

Negation is with the prefix MnF-:

MnFgaxe NMMay. Don't speak with them.

Mηρεωκ 6 τηολις. Do not go to the city.

мприкотк и пени. Do not lie down here.

A few verbs have special Imperative forms with prefixed x-:

nay: anay look, see xo: axi-, axi' say, speak

oyon: xyon open eine: xni-, xni bring

eipe: apipe, api-, api do, make

The verb MA, MAT, MAT' (or MHGI') is used as the imperative of +, but + may also be used. The imperative of GI (to come) is expressed by AMOY, which has distinct feminine and plural forms: f. AMH, pl. AMHGITN.

- 17.2 The vocative is expressed by using a noun with the definite article or a possessive prefix: $n_{\overline{p}}$ 0 king! $n_{\overline{p}}$ 0 my son! The Greek vocative particle ω (Gk. Δ) may also be used, but not before a designation of God.
 - 17.3 Infinitives of the type MICE, with stressed

vowel -1- and final unstressed -6, have the following prenominal and presuffixal forms:

MICE MEC(
$$\overline{\tau}$$
) - NACT to bear (a child)

6196 69 $\overline{\tau}$ - A9T to hang up, suspend.

The prenominal forms of many of these verbs occur with or without the final -r. Several important verbs of this type have irregularities:

The final \overline{N} of \overline{N} -, $g_{\overline{N}}$ -, and $g_{\overline{N}}$ - may be assimilated to \overline{M} before a following \overline{N} or \overline{M} . Note that in \overline{N} \overline{T} , $g_{\overline{N}}$ \overline{T} and $g_{\overline{N}}$ \overline{T} the syllabic \overline{N} is the stressed vowel of the word. \overline{F} - is often written as $g_{\overline{F}}$ -. Suffixes are added to these forms regularly: $g_{\overline{N}}$ -, $g_{\overline{N}}$

17.4 There is a certain ambiguity surrounding the terms transitive and intransitive in classifying Coptic verbs. The strictest definition of a transitive verb requires (1) that its direct object be marked with the "preposition" N (MMo') and (2) that the general equivalence kut MMo4 = kot4 be attested for the verb, i.e. that the verb possess prenominal and presuffixal forms. A less strict definition would require a transitive verb to satisfy either, but not necessarily both, of the above criteria. This is approximately the position adopted by W. E. Crum in his Coptic Dictionary, the standard lexical work in the field. Verbs not satisfying either of these criteria are labeled intransitive or are left unlabeled.

In the present work the designation transitive is extended to include verbs having prenominal and presuffixal forms that correspond exactly in meaning to the infinitive with e or $\overline{N}c\lambda$ (e.g. $cwt\overline{N}$ e, $gine \overline{N}c\lambda$). Thus $cotheq = cwt\overline{N}$

epoq and onty = wine Nooq are taken as fully equivalent to the criterion kwt MMO4 = KOTT above. A verb like AMA2TE (to seize) is considered transitive because its direct object is marked by MMO', even though it does not have prenominal or presuffixal forms. It seems reasonable, therefore, to extend the designation transitive even further and to include verbs like may and sime (to understand), both of which normally have an object with 6, but neither of which has prenominal or presuffixal forms. In other words, as long as there is no lexical contrast requiring the preposition 6 to have the semantic force of a true preposition (for, in regard to), we have generally labeled verbs with 6-objects as transitive in the glossary of this work. Some subjectiveness remains, however, and one can sympathize with W. E. Crum in his desire to drop the terms transitive and intransitive altogether (op. cit., p. vii).

Vocabulary 17

2 A P 6 2 Vb. tr. to guard, watch (6; from: 6, 680A $2\overline{N}$); to keep, observe, preserve (6).

gime vb. tr. to understand (e); to know, realize (that: xe).
gmge vb. tr. to serve, worship (Nx*); as n.m. service,
worship.

MOYTE vb. tr. to call (e), summon, name. Note the constructions:

жүмоүте ероч же тоганинс. They named him John.

жүмоүте е печран же тоганинс. They called his name John.

жүмоүте ероч й пран й печ
тhey named him after his father.

אאגידה vb. tr. to grasp, seize, take possession of, take captive (אַאַהי); to learn by heart.

n.xxxe (pl. N.xixeeye) enemy.

п. мато: soldier.

τε.csω (pl. Nε.csooye) teaching, instruction, doctrine.

- ne.nka thing (in general); property, belongings; NKA NIM everything.
- xe (1) conj. that, introducing noun clauses after verbs of speaking, knowing, perceiving; (2) introduces proper name or epithet in certain contructions.

Greek words:

π.Αιλβολος (ὁ διάβολος) the devil.

τε.Ψγχη (ή ψυχή) soul.

ne.nneγma (τὸ πνεῦμα) spirit, nearly always abbreviated (ne.) nna.

т.параволн (ή παραβολή) parable.

ακαθαρτος (άκάθαρτος) unclean.

Exercises

- A.1. NAMOT ENTALENTS HMAY 6. REAPROC ENTACHTS \overline{M} necest
 - πνομος εντά-πασεις τάλα 7. παάχε εντά-μηλτοι εντά
 μ μφύς 8. πεάτ ετε μπε-μηλτοι εντά
 - 3. 208 NIM GNTAYAAY $\overline{N}61$ 9. HENTACMACT $\overline{4}$ 6 HeC2A1

 $\overline{\mathsf{M}}$ маентнс 10. неитау $\overline{\mathsf{N}}$ тоу фарон

- 4. прире ентасмаст \overline{q} 11. пентатет \overline{n} аач
- В.1. сфтм е тасвф.
- 14. Миромов м прро м поннрос вт ммму.
- се-терюте, пафире.
 <u>Nа</u> наі, пажоєїс.
 - 15. мер-мечоуернте гл метсилуг.
- 4. МПРХООС N XXXY N рюме. 16. ХІТЧ да пархібреус.
- 5. гарег 6 негентоан тироу. 17. амагте Ммоч.
- 6. gmg e m nxoeic пекноуте. 18. диду е при 2N тпе.
- 7. мпрво врог. 19. моуте в пексон, понрв.
- 8. 611-116K20. 20. 21762 670N 6 MMATO1.
- 9. МПРВШК 6 ТЕРИМОС. 21. МПРВШ М ПАМТО ВВОА.
- 10. 22 f tayyxh, haxofic. 22. \dagger na4 \vec{n} oykoyi \vec{m} mooy.
- 11. MA- \overline{N} KA NIM \overline{N} NEBIHN. 23. API-NAI \overline{N} TA26.
- 12. 60 NMMai $2\overline{N}$ Teygh. 24. ani-mht \overline{N} pome 6 neima.
- 13. ANI-COOY H MATOL NHMAK.

- 25. \overline{N} Oyosiq Nim Apips \overline{N} Teq26. 28. Ayon \overline{M} npo.
- 26. AMHELTN EZOYN E NETPNE. 29. MNPTEM-NPO.
- 27. AMH GAPOI, TAGEEPE. 30. 60 NHMAI \overline{N} GMOYN \overline{N} EBOT.
- C.1. MNOYEIME E MNAPABOAH ENTAYXOOY NAY.
 - 2. AYAMAZTE MMOU NEI MMATOI, AYMOPU, AYNOXU 6 ПЕФТЕКО.
 - 3. Mnoyeine Nei mmhhae xe Ntou ne nexpictoc.
 - 4. ПФАЖЕ НЕ НА І Н ПАІАВОЛОС. МПРСОТНОУ.
 - 5. дчетме й пеотоето же д-печетот мот.
 - 6. NIM HENTAGLAPEZ EPOTH GBOA 2N NXIXEEYE?
 - 7. \overline{N} TEPECW, $\lambda Y \overline{N} T \overline{C}$ E THI \overline{N} TECCONE.
 - 8. λαχωκ εβολ π Νεισογ π πεαφπώρε.
 - 9. AYMOYTE EPOI M HPAN N TANAAY.
 - 10. AYMOYTE 6 HPAN \overline{M} HOHPE XE \overline{IC} .
 - 11. $\lambda \log \overline{M} g \in N\lambda q \ \overline{N} \ C \lambda g q \in \overline{N} \ POM N e.$
 - 12. OYALAY NE 2008 NIM ENTAKALY. 16. NTAKMOYTE E NIM?
 - 13. λ -пајаволос йтч е тернмос. 17. йтатетибит и аф и 2е?
 - 14. ETBE OY MUETNEIME E NACBO? 18. NTAKEM-HELXWOME TON?
 - 15. LIMMAZTE \overline{H} EXCOME THE \overline{q} . 19. OYAM \overline{H} MINE TE TEICBO?

Lesson 18

18.1 The First Present (Pres. I):

THE I am weeping THE We are weeping KPIME you (m.s.) are weeping TETHPIME you (pl.) are TEPIME you (f.s.) are weeping Weeping APIME he is weeping CPIME they are weeping CPIME she is weeping

With nominal subject: npwme pime the man is weeping $oy\overline{n}-oypwme pime$ a man is weeping.

The prefix of the 2nd pers. fem. sing. also appears as

Tep- or Tp-. orn must be used to introduce an indefinite nominal subject.

The First Present usually describes action, activity, or process in progress at the time of speaking. It is therefore equivalent to the English progressive present (am weeping, am writing, etc.) except in those English verbs that do not normally use this form (e.g. think, know, see, hear, understand, wish, hope, believe), where its equivalent is the simple present: †GING I understand, †NAY I see, etc.

The infinitives $\epsilon\omega\kappa$ and $\epsilon\iota$ are not used in the First Present.

With the sole exception of oyom (to wish, love), the prenominal and presuffixal forms of the infinitive cannot be used in the First Present. Certain compound verbs are an exception to this rule and will be considered in a later lesson.

The pronominal prefixes of the First Present and its negative are also used before adverbial predicates:

† האו I am in the house.

אור אור אור אור אור ווא They are not in the house.

18.2 The First Future (Fut. I) is formed by prefixing NA- to the Infinitive. Inflection is exactly like that of the First Present, including its negative:

 †имриме, кимриме ...
 Neg. \overline{n} †имриме мириме мириме мириме мириме мириме мириме милериме милериме

The First Future corresponds to the English simple future (I shall write, I shall go) or to the intended (planned) future (I am going to write, going to go). The 2nd pers. pl. commonly appears as TETNA- for expected TETNNA-.

- 18.3 The term intransitive as applied to Coptic verbs requires a further comment (cf. § 17.4). Coptic has many intransitive verbs, such as verbs of motion (ει, κωκ, μοοφε) and verbs denoting activities involving no direct object (РІМЕ, МКОТК, etc.), whose classification is not problematic. But the intransitive use of verbs that are also transitive requires some attention. In certain situations any transitive verb may be used intransitively: the object may be omitted because it is understood from the context, or the speaker may wish to predicate the action of the verb without reference to any particular object (e.g. we plowed all day as opposed to we plowed the field). This usage is as commonplace in Coptic as it is in English and will not be noted in the vocabularies or final glossary. There is another type of intransitive usage, however, that is quite different. Compare the following:
- 1) \overline{N} TEPGYXOK \overline{N} NG4200Y GBOX when he had completed his days
- 2) NTEPE-NE4200Y XOK 680A when his days were completed.
- (1) is the normal active transitive use of xok GBOA; (2) involves a change in voice from active to passive (or mediopassive, as a more general term). For speakers of English this medio-passive usage offers no problem since many English verbs have the same ambiguity: he closed the door

vs. the door closed; he burned the paper vs. the paper burned. In the vocabularies and final glossary the designation intr. before the meaning of a verb whose transitive meaning is given first will always refer to this mediopassive usage. Of the transitive verbs introduced up to this point, the following have important medio-passive uses:

xωκ 680% intr. to be completed, finished, fulfilled; to die. 2ωπ intr. to hide (oneself).

BWA GBOX intr. to be melted, scattered, dispersed; to come undone, be loosened; to go to pieces.

TWM intr. to shut, close (subject: door, eyes, mouth, etc.). OYWN intr. to open.

ογως intr. to settle, dwell; to alight (on: $21x\overline{N}$, επεκητ $21x\overline{N}$). Moγ2 intr. to become filled, full (of, with: $\overline{M}MO^{2}$).

18.4 Infinitives of the type $\kappa\omega\tau c$ (to turn), with stressed $-\omega$ - and final unstressed $-\varepsilon$, have the same prenominal and presuffixal forms as the type $\kappa\omega\tau$:

KWTE KET- KOT' to turn.

Noγxe (to throw), with -oγ- for -ω- because of initial N (cf. p. xvi) also belongs to this type; the infinitive Noγx mentioned in Voc. 14 is a less frequent variant. Infinitives with -ωω- and final -e have similar forms:

yww6e yee6e- yoo6' to strike, wound.

18.5 Greek verbs occur frequently in Coptic texts. These have a single fixed infinitive form resembling the Greek imperative form and are inflected like any other Coptic verb. Examples:

πιστεύω	to believe (ε)
επιτιμάω	to rebuke (κλ*)
πειράζω	to tempt (κ̄κο*)
πατεύω	to fast
Δρχι	ἄρχω

Vocabulary 18

- κωτε κετ- κοτ' vb. tr. to turn (ΜΜο'; away: εΒολ; back: $eπλ_2ογ$); intr. to rotate, circulate; to surround, go around (ε); to consort (with: $μ\overline{μ}$).
- C2Al vb. tr. to write (MMO'; on, in: e, exN, 21, 21xN, 2N; to: NA', e, ga); to register; to draw, paint; as n.m. writing. letter.
- 6ωωπ vb. intr. to look, glance (at: ε, εxñ, ñcλ, ε2ογη ε);
 6ωωπ (εξολ) 2ητ to look forward to, expect, await.
 Often with εξολ, ε2ογη, ε2ρλι, επέςητ.
- cooy \overline{N} vb. tr. to know (\overline{N} mo'; about: eres; how to: \overline{N} + Inf.; that: xe); to recognize, be acquainted with; as n.m. knowledge.
- Meeye vb. intr. to think, suppose (that: xe; about: e); to ponder, consider (often + eBOX); as n.m. thought, mind.
- кωτε n.m. neighborhood, surroundings; м/гм пкωτε n in the neighborhood of, near, around; pron. obj. are expressed w. poss. prefixes: м печкωτε around him.
- 2HT° prep. forward to, before; used idiomatically with certain verbs, like 6ωg above and nωτ εκολ to flee (2HT°: from); anticipatory suffix is required.

esox xe, erse xe conj. because.

n.xxcic desert, wilderness.

Te. 6 poomne, ne. 6 poomne dove.

BXX6 (pl. BXX66Y, BXX6YC) adj. blind.

And the Greek verbs in §18.5 above.

Exercises

- (1) λ -neq2wb xwk gbox. (2) cenanoy2 \overline{N} page \overline{N} 61 nen4yxh.
- (3) $\overline{\Pi}$ +nicteye epok an. (4) +na2wn 2 \overline{M} nxaeie. (5) analaboaoc neipaze \overline{M} mod \overline{N} cay \overline{q} \overline{N} 200y. (6) nen \overline{N} \overline{N} akabaptoh oywy an e el ebox. (7) \overline{M} neoyoely neikochoc thp \overline{q} nabwa ebox. (8) a-nb \overline{N} xe xooc xe ha hal, naxoelc. (9) a-nhay \overline{M} neqy \overline{M} ye xwk ebox. (10) \overline{N} t \overline{N} haoyw2 an $2\overline{N}$ telxwpa.

(11) \dagger meeye as \overline{N} tok oyalkaloc ne. (12) \overline{N} baa \overline{N} \overline{N} baseye NAOYON. (13) KMGGYG x_{G} ANT-NIM? (14) GTRC OY TGTNKOTG MN 26NPWM6 \overline{N} T6IMIN6? (15) 2-N64CN2Y2 BWA 6BOA \overline{N} N64OY6PHT6. (16) $\Delta Y_2 \otimes \Pi \overline{N}61$ NECNHY & MMATOL M HPPO. (17) STBG OY Kenitima nai? (18) $\overline{\text{N}}$ tetnaeine an e nequapaboah. (19) мсенапістече є нафахе ан. (20) мхіхеече накоте є пенфие. (21) λ -THE OYWN, λ 461 680 λ NG1 OYNOG N OYOGIN. (22) λ -TEC 2 IME APXEL \overline{N} PIME. (23) MEWAK CENAKET-THYTH ENAZOY. (24) NWOOC 2AP62 6 NGCOOY 2N TCWWG. (25) A-NG4BAA TOM 2M nmoy. (26) $cenaoyw2 2\overline{m}$ $nkwten \overline{n} etahm$. (27) fnanex-nageNEY2OOP. (28) $T\overline{N}NAMOYTE EPO4 \overline{M} HPAN \overline{M} HE4ELOT. (29) ANGEN-$ TERICTORH ENTARCZAI MMOC NAN. (30) ETBE OY TETNINHCTEYE \overline{N} ογοειώ νιμ? (31) λ-νες λ μογλ πογλ π λ λ πογει TE21H AN. (33) NTEPERCUTH 6 HE2POOY, A46 ω QT 6BOA. (34) trime ebox xe x-nxcon moy. (35) x-teepoomne oyw exmпвима. (36) семесує же \overline{N} точ пе пе \overline{xc} . (37) \overline{N} тесооу \overline{N} ан \overline{N} c2al. (38) enitima nay etbe neynobe. (39) $\tau \overline{N} 6 \omega \omega \overline{\tau}$ ebox 2HT \overline{q} \overline{m} \overline{n} \overline{e} 2 \overline{O} γ \overline{e} \overline{m} \overline{m} γ . (40) \overline{m} \overline{n} \overline{e} q \overline{q} γ \overline{m} \overline{m} \overline{e} γ $\overline{$ \overline{n} energy examples and \overline{n} of \overline{n} oversome. (42) Nai ne \overline{n} waxe ENTAGE I PMOOY IN TEMPORAL (43) NTNCOOYN AN N NGAME N AZA. (44) $+\cos\gamma\bar{n}$ xe \bar{n} tok he higher \bar{m} hnoute. (45) $\lambda\gamma_2\omega n$ 620yn 6 THYAH AYW AYTOME. (46) NE2M HHI AN. (47) THEOOYN x_6 OYNO6 те течсвы. (48) тимпыт евох гитоу е пхаете. (49) ачгко EMATE EBOX XE AUNHOTEYE \overline{N} MMOYN \overline{N} 200Y. (50) ETBE OY TEIT хира $\overline{\mathbf{N}}$ гике моофе $\overline{\mathbf{N}}$ сwi? (51) ачархі $\overline{\mathbf{N}}$ фахе м $\overline{\mathbf{N}}$ пминфе. (52) ANKWIG \overline{N} \overline{N}_2 HKG GBOA. (53) +COOY \overline{N} \overline{M} MOK. \overline{N} TOK \overline{N} $\pi_{A1ABOAOC}$. (54) etbe oy kneipaze \overline{m} moi \overline{n} tel2e? (55) \overline{n} ce \overline{m} печкоте ан. (56) тетнасооу \overline{n} же \overline{n} та \overline{p} -на \overline{e} етве-тнут \overline{n} . (57) \uparrow NAWN \overline{N} NEXHY ET $2\overline{N}$ TEMPW. (58) \overline{N} TE \overline{N} -OYWW \overline{G} BWK GBOX, AYKOTN ENAZOY 6 NGNHI.

Lesson 19

19.1 The relative forms of the First Present and First Future employ the relative pronoun et, etc. When the relative pronoun is the subject of the relative clause, no further pronominal subject element is required:

проме ет ріме the man who is weeping those who hear my words

тиматої ет маммагте тимоч the soldiers who will seize him

проме ет маєїме тимої the men who will bring the silver.

When the relative pronoun is not the subject of the relative clause, a subject noun or pronoun and resumptive pronouns are required; the relative pronoun combines with the various subject elements as follows:

 ε+
 who/which I ...
 ετΝ

 ετκ
 who/which you ...
 ετετΝ

 ετε (ρ)
 etc.

 ετη
 ετογ (note this form)

 ετζ

With nominal subject: erepe-npowe who/which the man...
Study the following examples carefully:

 \overline{N} \underline{M} \underline{M}

the words which I am writing
the man whom you are seeking
the city in which they are settling
the child whom he will leave behind
the commandments which he will give
to us

אנאד פּדפּףפּ-תפּגפּושד the money which your father will אורא פּדססד \overline{k} entrust to you

When the verb of the relative clause is negative Pres. I or Fut. I, the relative pronoun is ere and subject as well as resumptive pronouns must be expressed in all constructions:

the men who do not heed me
the man who will not heed me
the words which we do not
understand
the villages which they will

метме ете <u>М</u>сенаамагте
Ммооу ам

not seize

19.2 The direct object of a transitive verb may be used in a reflexive sense:

I threw myself to the ground.

Δαειλλα 2Μ πμοογ Μ πειερο. He washed himself in the water

of the river.

Some verbs have special meanings in the reflexive, e.g.

ογλ2' Ncλ to place oneself in the following of, go in accordance with; also simply "to follow."

κοτ' (1) to return, go back (to: επλ2ογ ε, εκολ ε, εκολ ga, ε2ογη ε, ε2γλι ε); (2) to repeat an action, usually coordinated, as in

жикот чиртие he wept again

or with 6 + Inf., as in

MHENKOTH 6 NAY GPOC We did not see her again.

The verb $\tau\omega\sigma\gamma N$ occurs optionally with reflexive suffixes: $\lambda 4\tau\omega\sigma\gamma N\overline{4} = \lambda 4\tau\omega\sigma\gamma N$ (he arose). After stem-final -N the 2nd pers. masc. sing. suffix -K often appears as -r: $\lambda K\tau\omega\sigma\gamma N\overline{r}$ you arose.

The reflexive verb x26pxr', to stand, is actually a compound of x26 (a form of the verb w26, to stand) and the preposition 6pxr' to or at the foot/feet of. 6pxr' itself consists of the prep. 6 and the noun pxr' foot, which belongs to that small group of nouns that may take pronominal suffixes in a possessive sense: pxr my foot, pxrK, your foot, etc.

19.3 Infinitives of the type $c\omega r \bar{\eta}$, to choose,

constitute the largest class of verbs in Coptic and have the following prenominal and presuffixal forms:

COTH CETH COTH

When the final consonant of the infinitive is a blmnr consonant, the presuffixal form is usually written with -6-before the suffixes - τ , - κ , -q, -c: COTMEq, COTME τ , COTME κ , etc. When the final consonant is -2, spelling alternates between -2 and -32 in the unbound form: OYWN2 or OYWN2.

When the second consonant of the Infinitive is $_2$ (more rarely $_9$), the presuffixal form may have $_{-\lambda}$ - instead of -o-:

 $Ογω2\overline{M}$ $Ογ62\overline{M}$ Ογλ2M' to repeat $τω2\overline{M}$ $τε2\overline{M}$ τλ2M' to invite

When the infinitive begins with \varkappa or \varkappa , $-\omega-$ is replaced with $-o_Y-$:

Vocabulary 19

- wise ws (т)- ws т vb. tr. to change, alter (ммо'); intr. and reflex. to change, be altered (to: ε; into: ₂N; in form: N смот).
- λεερλτ' vb. reflex. to stand (before: ε; against: ε, $ex\overline{n}$, ογεε; with: $m\overline{n}$).
- oyων σογων σογον σογον
- пе.смот form, likeness, appearance; character, behavior.

TG.CMH VOICE, sound.

Π.Κω2 T fire.

Β P P G adj. new, young; Ν Β P P G recently, anew.

AC adj. old (not used of persons).

Greek words

ΤΕ.ΓΡΑΦΗ (ἡ Υραφή) writing, scripture.

Τ.ΕΧΟΥCΙΑ (ἡ Εξουσία) power, authority.

Τ.ΠΙCΤΙC (ἡ πίστις) faith, trust.

Π.ΜΥCΤΗΡΙΟΝ (Τὸ μυστήριον) mystery.

Exercises

- A. (1) none etoymour \overline{n} axol epoq (2) axoome etoyag \overline{q} (3) π_2 061T6 6T4+ $\overline{\text{MMO4}}$ 21004 (4) π_2 076K0 6T0YNANOXK 6P04 (5) MMOOY ET MOY2 \overline{M} HENXO1 (6) \overline{N} PWME ET NAHWT E HXAELE (7) TEXOPA $ETNN_2HTC$ (8) NET NAOYA2OY NCOM (9) TERLICTOAH ϵ +NAC2A1 MMOC GAPOK (10) Te21H ϵ TOYMOOGE 2100C (11) NCA2 ETE \overline{N} TET \overline{N} COOY \overline{N} \overline{M} MO4 AN (12) NET NA226PATOY \overline{M} HE4 \overline{M} TO EBOA (13) Incon ethnamop $\overline{4}$ m neckham (14) if po ethnam ebox 2 ht $\overline{4}$ (15) HET OYON \overline{N} $\overline{N}BAA$ \overline{N} $\overline{N}B\overline{A}AEEY$ (16) $\overline{N}POME$ $ET\overline{4}KOTE$ $N\overline{M}MAY$ (17) \overline{N} \overline{Q} \overline{M} \overline{Q} $\overline{$ TAAC NA4 (19) TRICTIC 6/61N6 MMOC 2N TEIC2IM6 (20) TEброом пе ет \overline{K} налу ерос (21) тефтни етет \overline{N} поуже \overline{M} мос евох (22) NETTNAERITIMA NAY (23) RMYCTHPION ETTNAGOART NAN ϵ -BOA (24) THE GTOYNAPOK $\overline{4}$ $\overline{1}$ $\overline{1}$ $\overline{N}CW^q$ (26) $\overline{N}x$ 1x 6 6 Y 6 T K W T 6 E T 6 N TO A 1C (27) TB HMA 6 T \overline{q} N A - $\lambda_2 \in P \lambda T = 21 \times \omega = (28)$ NET HELP $\lambda_2 \in \overline{M} M \omega T = (29)$ HKW2 $\overline{T} \in T \in P \in T$ шноуте нанож $\overline{4}$ еж \overline{m} пкаг (30) тесми етерсф \overline{m} ерос (31) \overline{m} -MATOL ET NAKOTK ENAZOY (32) NZOELTE \overline{N} BFPE ETCHAGONT (33) NET $n\omega_2 T \overline{M}MOOY \overline{N}N\lambda_2 P\lambda K$ (34) TETPAHH ETRICTEYE EPOC (35) прат етере-понре насе ероч (36) проме ет боот броун брои (37) $IMAEIN ETOYNAOYON 2\overline{4} EBOA (38) TEC2 IME ETAME MMOC$ (39) $\pi \in \text{CMOT} \in \overline{\text{TTOYWH}} \times \overline{\text{MMO4}} = 680 \lambda \times \overline{\text{N}} \times \overline{\text{HTT}} = (40) \times 200 \text{ GeV etcha-}$ NHCTEYE N2HTOY
- B. (1) Мпров-хаду \overline{n} модже еткнаейтоу $2\overline{m}$ петхооме. (2) дуное \overline{m} мустнріон оуши \overline{n} инт \overline{n} (3) \overline{m} пегооу ет \overline{m} нау сена-

 $6\omega\lambda\bar{n}$ $6BO\lambda$ $\bar{n}61$ $\bar{n}\omega\lambda\lambda\bar{c}$ \bar{m} nenxoeic. (4) $\lambda 49\bar{e}\bar{t}\bar{t}$ $\bar{n}61$ nalaboloc $2\bar{m}$ necmot \bar{n} oyaffeloc \bar{m} noyoein. (5) $\lambda 42\omega n$ 62oyn \bar{e} $n\bar{p}po$, $\lambda 4n\lambda_2\bar{t}\bar{t}$, $\lambda y\omega$ \bar{m} $necke-\lambda\lambda\lambda y$ \bar{n} $\omega\lambda\bar{c}$. (6) $\lambda \omega$ $\bar{t}\bar{e}$ teicmh \bar{e} teotmh epoc? (7) $\lambda 2ep\lambda\bar{t}\bar{k}$ $n\bar{m}$ nal oybh nal. (8) $\lambda 4\uparrow$ $21\omega\omega$ \bar{n} teothn \bar{n} λc $entainox\bar{c}$ $ebo\lambda$. (9) \bar{n} takeine \bar{m} $neix\omega\omega me$ \bar{n} λc twn? (10) $\lambda y\lambda_2ep\lambda$ toy \bar{m} nemto $ebo\lambda$ \bar{m} nnoe \bar{n} po. (11) $\lambda 4x$ ooc \bar{n} $entaino}$ $entaino}$, $entaino}$, e

Lesson 20

20.1 The Inflected (Causative) Infinitive.

тре-проме сотм that the man hear

Negation is with $\tau \bar{n}$ - placed either before the whole expression or before the infinitive: $\tau \bar{m} \tau_{P} \lambda c \omega \tau \bar{m}$ or $\tau_{P} \lambda \tau \bar{m} c \omega \tau \bar{m}$ that I not hear.

The Inflected Infinitive is used in the following ways:

(1) As a complementary infinitive, with e, after appropriate verbs of wishing or commanding when the subject of the infinitive is different from that of the main verb. Contrast

toyww ε εω π neima. I want to remain here.

†ογωφ ετρεκεω π пенма. I want you to remain here.

тпоγωφ εтπτρεκεωκ εεολ. We want you not to go away.

(2) Like the ordinary infinitive with e, the Inflected Infinitive is used in a wide range of result or purpose expressions, often corresponding to English "for... to..."

х-пеотоетф жок евох етренвок евох.

The time arrived (lit. was fulfilled) for us to leave.

A4+ NA4 N TEXOYCIA ETPERNOYME EBOX N 26NNNX N AKABAPTON.

He gave him the power (for him) to cast out unclean spirits.

Because of the frequent use of the Inflected Infinitive with ε , we shall spell this as a single unit, as in the preceding examples.

(3) With the preposition ${}_2\overline{N}$ + the definite article n- the Inflected Inf. has the force of a temporal clause with "while, as":

2 m птречмооф while/as he was walking 2 m птре-поунна фана as the priest was praying.

The tense of such "clauses" depends on the context. They occur frequently after introductory Acquine:

Acgune Ae 2M πτρειωνώς 2M περπε...

It happened, however, as he was serving in the temple, that...

(4) After the preposition MNNCL and without an article the Inflected Inf. is equivalent to a temporal clause with "after":

MNNCA TPANAY GPOOY after I saw them, ...
MNNCA TPG-NGGGIOT BOK GBOA after his father left, ...

(5) The Inflected Inf. is used frequently with the impersonal expressions treated in the following paragraph.

Other uses will be taken up in a later lesson.

- 20.2 Impersonal Expressions. The impersonal use of acomore was introduced in Vocabulary 9. There are several other impersonal expressions, some verbal, some anomalous, which occur frequently:
- (1) $2 \lambda n \overline{c}$ it is necessary (neg. \overline{N} $2 \lambda n \overline{c}$ λN), followed by the Inflected Inf. The subject of the infinitive may be anticipated with the preposition ϵ ; an untranslatable ne often co-occurs with $2 \lambda n \overline{c}$.
 - 2 λ Π C (π G) G T P G N Π W T G B O λ . It is necessary that we flee.

 2 λ Π C (π G) G P O I G T P A Q A X G

 N H M A K . With you.
- (2) oy \overline{N} -(ω)6om it is possible; neg.: \overline{N} -(ω)6om it is not possible. The subject of a following infinitive may be introduced with \overline{M} mo*, with the Inflected Inf., or both:

мм-доом є єїмє є мечдахє. It is not possible to understand his words.

MN-φ60M MMON ε ειμε. It is not possible for us to understand.

(3) gage (or cage) it is appropriate, proper, fitting; neg.: \overline{N} gage an or meage. The subject of the infinitive may be anticipated with prep. c.

gge epo4 6 кwk e2oyn. It is proper for him to enter.

м gge epwтм дм етрететнем It is not proper for you to
м пе м пе м . remain here.

The relative forms nete gge, Nete gge, what is proper (neg.: nete/Nete Megge) are often used as substantives.

(4) \overline{p} -xnx* to please, used impersonally with subject c- and an object suffix, or with a personal subject and a reflexive suffix. The suffix on xnx* is required; a nominal object is anticipated by a suffix and introduced with \overline{N} .

Study the following examples:

ACP-ANAS STRESS 620YN 6 nelkocmoc.

It pleased him to come (i.e. he came willingly) into this world.

АСТ-АНАЧ Й ПМННФЕ СТРЕУНАУ 6 1171It pleased the crowd (for them) to see this.

AIF-ANAI ETPACOTH 6 NGKWAXE.

It pleased me to hear your words.

Note also the partially synonymous verb F-2NA' to be willing, desire, which is used only with a personal subject and reflexive suffix:

NEIGAXE.

 $\lambda I \overline{P} = 2 N \lambda I$ etpace λI Nak \overline{N} I wanted to write to you (about) these things.

F-2NA" is not used in the First Present; F-ANA" has no such restriction.

20.3 The verb nexe-, nexx', followed by its subject, is equivalent to xw in the First Perfect, but is used only to report speech, with xe:

> The old man said. "... **πεχε-π2λλο χε...** He said to me, "... MEXAG NAL X6 ...

20.4 Infinitives of the types coack, to console, and gтортр, to disturb, have the following prenominal and presuffixal forms:

> COACT CXCX-CZCWX* **утортр** утртр− утртшр*

With the exceptions of the infinitives treated below in Lesson 26, the remaining types of transitive infinitives do not constitute regular classes of any significant size. The following verbs of minor types have occurred in the lessons up to this point:

ã <u>w</u> ã €	ბ പ്ൻ e−	ѹМ҃ѹнт″	to	serve
CSYI	C 6 5 -	C 2 A I *	to	write
сооүй	COYN-	COYWN	to	know

THNOOY THNEY- THNOOY' to send.

When the presuffixal form of the infinitive ends in a diphthong, as in $c_2\lambda_1$ and $\tau \bar{N}Noo\gamma$, the object suffix of the 3rd pers. pl. regularly appears as $-co\gamma$: $c_2\lambda_1co\gamma$ to write them, $\tau \bar{N}Noo\gamma co\gamma$ to send them. The -c- of this form sometimes appears also before other suffixes, e.g. $c_2\lambda_1c\bar{q}$ to write it.

Vocabulary 20

- coack ckck- ckcwa* vb. tr. to console, comfort (MMO*); intr. to be comforted; as n.m. consolation.
- утортр утртр- утртюр vb. tr. to disturb, trouble (мно);
 intr. to be disturbed, troubled; as n.m. trouble,
 disturbance.
- concπ cπcπ- cπcωn vb. tr. to beseech, entreat (MMo), often followed by erre-. The unbound and prenominal forms also occur as conc and cenc-. As n.m. prayer, entreaty.
- ογωφε ογωμε ογοφε vb. tr. to respond to (Μπο, Νλ); to answer.
- xnoγ xne- xnoγ' vb. tr. to ask, question (mmo'; for: ε; about: ετες).
- мокмек мекмоук vb. intr. or reflex. to think, ponder; as n.m. thought(s).

mocte mecte- mectω* vb. tr. to hate.

κω $\overline{\mathsf{M}}\mathsf{M}\mathsf{O}^{\mathscr{O}}$ ε + Inf.: to allow (someone) to do (something). Greek words:

п.сωма (τὸ σῶμα) body.

π. πειρωσμός) temptation.

And the impersonal expressions ελπζ, ggg, ογπ-(g)6ομ, μπ-(g)6ομ.

Exercises

А. (1) минса тре-пфире и врре вык евох (2) $2\overline{M}$ птреубыл евох и неімустиріон (3) $2\overline{M}$ птре-поунив агерат $21\overline{P}\overline{M}$ перпе (4) минса тресгансоу гіх пхынье (5) $2\overline{M}$ птреуноухе м печсыма ехм пкыг (6) минса тренсискомом (7) гм птреплаволос петрале ммоч ги омное м петраснос (8) минса тречтиноом м печраре м мертт фарон (9) минса тречомог евол и нечмаентне (10) минса транагт м печмто евол (11) гм птречеф евол гм папот и ас (12) минса тре-милтот рекупиення

B. (1) $2\lambda n\overline{c}$ ne epon etpen \overline{p} -nete gae \overline{n} oyoeig nim. (2) NAI NE NOAXE ETKNACZAICOY NAH. (3) NEXE-NZ $\overline{\lambda}$ O XE OYNOE TE TEKNICTIC, NAWHPE. (4) WE ETPENEMBETT N OYOELW NIM. (5) $2\lambda\Pi\overline{C}$ etpensw \overline{M} neima \overline{N} tromne. (6) \overline{N} Taytnnooyt warwt етрафахе $n\overline{m}$ н $T\overline{n}$. (7) пехе- $T\overline{C}$ нач хе +оуа $q\overline{q}$. (8) асфипе AE $2\overline{M}$ HTPEGNAY E NAI, AGRAZTY, AGGTOPTP. (9) MN-60M MMOI етраффинтк. (10) ганд брок пе етрексхех-тмаач м пфире. (11) NIM HENTAY NAY \overline{N} TEXOYCIA ETPEGEIPE \overline{N} NEIZBHYE? (12) \overline{N} теречсфт \overline{M} е нафаже, ачфторт \overline{P} , ачмекмоук \overline{q} . (13) +-оуфф етрекош иймы й фомйт й евот. (14) ффе етреусштй йсы $\overline{N}T\lambda T \in T\overline{N}_2$ G $T \in TP \rightarrow P$ \overline{N} ΔC $T \in N$ \overline{N} ΔN \overline{C} \overline{C} \overline{C} \overline{C} \overline{C} \overline{C} \overline{C} \overline{C} \overline{C} n+me \overline{m} $nooy. (18) оу петоунала <math>2\overline{m}$ $ntpeycwt\overline{m}$ e teqcmh?(19) λ -течсво \overline{N} помирои отртр-пархієпіскопос ємате. λ -toy \overline{N} koy! \overline{N} xo! 6! 620YN 6 T6MP ω . (21) coe 6PO! 6TP λ -60 ммме. (22) тмусистом стречтиност ером. (23) мм-ф6ом е солс⊼ й йлпістос. (24) дефопе гм птречетие е неумокмек, A4APXI \overline{N} enitima nay. (25) nexal nay xe $\overline{N}T\overline{K}$ -Oyaikaloc. (26) сенахноуч етве техоуста ет \overline{q} натахс нау. (27) мещие етреует егоун е перпе й тетге. (28) тмосте ймок мй некψλ×ε Μ πονηρον. (29) λγχνε-ογ2πλο χε λφ τε τπιςτις? (30) ACOWNE AS MNNCA TREGEWK SEON, AYARXI N WAXS STES MMASIN ентачалу \overline{N} теумите. (31) \overline{M} прместе-лалу \overline{N} роме. (32) а 1спсшпч етречоушов наг. (33) йпечки ймооу е же-хаху й фаже. (34) AYOYWOR NAU \overline{N} 61 HCON CNAY X6 \overline{N} 7 XNAAY 6POU 21 T621H. (35) $T\overline{N}NAXNOY4$ ETBE RESTOOY \overline{N} XWWME \overline{N} TE REYALLEYION.

Lesson 21

21.1 The Imperfect.

ngikwt I was building nenkwt nekkwt you were building neтeтñkwt nepekwt etc. neakwt неукwт

NEPE-HPWME KWT the man was building

The Imperfect is optionally, but often, followed by an untranslatable ne: Neikwt ne, Nekkwt ne, etc. Negation is with an: Neikwt an (ne), Nekkwt an (ne), etc.

The Imperfect is used to describe an action, activity, or process as in progress in past time and is normally the equivalent of the English past progressive unless idiom requires the simple past, e.g. Negcooy $\overline{\mathbf{n}}$ they knew (not: they were knowing). It also often conveys the meaning of habitual or recurring activity in the past: they used to build, they would build.

Relative clauses containing an Imperfect are introduced with the relative pronoun ere or, more frequently, with e- prefixed directly to the verbal form:

проме ете мечмоофе 21 те21н } the man who was walking on the road

пни ете меукот ммоч } the house which they were building

Pronominal resumption of the subject is required. In general, the prenominal and suffixal (prepronominal) forms of the infinitive may not be used in the Imperfect.

21.2 The Qualitative. Many verbs possess a second lexical form known as the qualitative. The qualitative describes a state or quality resulting from the action, activity, or process expressed by the Infinitive; it is

most conveniently taken as equivalent to English "to be" plus an adjective. The qualitative of transitive verbs is passive from the English point of view. E.g.

Inf. κωτ to build Q. κητ to be built (i.e. in a fully constructed state)
Inf. εωπ to hide Q. εμπ to be hidden, secret.

The form of the qualitative is more or less predictable for verbs belonging to the main classes:

- (a) type кот: Q. кнт; моүр: Q. мнр
- HIN to be reckoned, MH2, MG2 to be full ascribed to (6) OYH2 to live, dwell, be XHK (GBOX) to be finished, MHP to be bound done, perfect THM to be shut BHX to be loosened, un-OYHN to be open done, untied, dissolved GHN to be received, acceptable
- (b) type кютє: Q. кнт; ноужє: Q. ннж кнт to be turned, turning, circulating ннж to be lying, reclining (esp. at table); to be
- (c) type MICG: Q. MOCG
 MOCG to be born gose to be different, various
 - (d) type $c\omega\tau\overline{n}$: Q. $co\tau\overline{n}$; $n\omega_2\overline{\tau}$: Q. $n\lambda_2\overline{\tau}$

ογονε to be manifest, clear, plain 60λπ to be known, revealed, clear γοκε to be burned, destroyed by fire πλετ to be prostrated, bowing

(e) type coλcx: Q. cxcωλ; ωτορτφ: Q. ωτφτωρ
 cxcωλ to be consoled ωτφτωρ to be disturbed, upset.
 Otherwise, there is some irregularity:

кю: Q. кн to be situated, lying; to be

 c_{2} 1: Q. c_{H2} to be in writing, written c_{1} : Q. c_{HY} to be sated, full.

Note that KH, NHX, and OYH2 may all correspond to English "to be" when location or position is involved.

The qualitative is a verb and may stand in place of the Infinitive in the First Present and the Imperfect, together with their negative and relative forms. It is especially important to keep in mind that the qualitative does not express a passive action (cf. §13.4); it describes the state that the subject is (or was) in:

The qualitative may not be used in any of the other conjugations introduced up to this point, including the various constructions with the Infinitive and Inflected Infinitive.

21.3 Prepositional phrases with $2\overline{N}$ + a noun with the indefinite article occur very frequently as adverbs:

For gone, Me, and GenH see the Vocabulary below.

Vocabulary 21

MOYN vb. intr. (± GBOλ) to remain, last, endure; as n.m. perseverance, continuing. 2N ογμογη GBOλ continuously. cmoγ, Q cmamaat vb. tr. to bless (6); Q to be blessed. cmc ccc- coc Q chc vb. tr. to paralyze; Q to be paralyzed. τε.γνογ (ογνογ) hour. N τεγνογ adv. immediately, forthwith. τενογ adv. now. Φλ τενογ until now. ΧΙΝ τενογ from now on.

- GNG2 eternity; freq. as adv. forever (with neg.: never).

 gh eng2, gh Nieng2 idem (for Ni- see §30.8).
- \mathbf{x}_{1N} prep. from, starting from, since. \mathbf{x}_{1N} $\overline{\mathbf{M}}$ nooy each from today onward.
- gone occurs only in 2 n oygone adv. suddenly.
- 66HH vb. intr. to hurry, hasten (to: ϵ , ϵ_{PAT} ; to do: ϵ + Inf.). $2\overline{N}$ oy66HH adv. quickly, hurriedly.
- т.ме truth, justice; as adj. true. $2\overline{N}$ оүме adv. truly. Name idem.

Exercises

- (8) HENTOAH ET CH2 21 HELXWWHE (9) HMA ETOYNHX \overline{N}_2 HT $\overline{4}$
- (10) Ngaxe et 60 λ N ebox nan (11) neneiote et chamaat
- (12) HANCE ET CACON (13) WARE NIM ET CH2 2M HNOMOC
- (14) проме етинх $2\overline{M}$ печні є оуюм (15) поме ете метамоос 21204 (16) присе ете мемароч
- B. (1) NEGCOACT NET INCA? IN NEGROUPH (2) NEPE-NEXHY кн $2\overline{N}$ темрю. (3) неимосте \overline{M} мооу смате. (4) нере-плаос TELZE? (6) NEPE-MINYE MEZ \overline{N} OYOGIN. (7) TELEKKAHCIA NAMOYN 680A DA GNG2. (8) NEPE-REYHI KHT $2\overline{N}$ TOPINH. (9) гапо стрекиют свол гм очесин. (10) нере-печфире сис. (11) Acquire Ae $2\overline{N}$ Oyache Aycutm eynog \overline{N} 2pooy. (12) Nepeпоуннв отртор емате. (13) на не нечолке $2\overline{n}$ оуме. (14) ненску ан. (15) Птауфторт втве ммаети витачару. (16) NEKNOBE THPOY KH NAK EBOA. (17) NANAMOYN EBOA AN NEI петкосмос. (18) нере-печаюв жик евох наме. (19) нере-1022NNHC OYH2 21XN TEPHMOC. (20) NEYNA2 $\overline{\tau}$ M NEMTO 680A M mo6 \overline{N} \overline{p} $\overline{n}e$ τm . (23) $\phi \phi e$ $e p \phi \tau \overline{N}$ $e \tau p e \tau e \tau \overline{N} c \phi \tau \overline{M}$ $\overline{N} c \lambda$ $n e 1 e n \tau o \lambda h$. (24) Птерпсшты в печаспасмос, анбепн врати. (25) неганоос гіхм пекро п валасса. (26) мп-шеом ммої етраоушив ерок.

(27) меммоофе \overline{N} оуоуобіф 21 тегін є тполіс. (28) дофиле де $2\overline{M}$ птреухноуч, дчоуфф нду $2\overline{N}$ оугісе. (29) тпидсмоу є печрли фд и і енег. (30) д-пеіффиле соб \overline{N} мечоуєрнте. (31) теноу †сооу \overline{N} же \overline{N} ток пе пе \overline{N} С. (32) тді те идме теумоу \overline{M} печмоу. (33) дупфт \overline{N} теумоу ех \overline{M} пекро. (34) фд темоу \overline{M} мінемкот \overline{N} 6 иду єроч. (35) †идеф иймач \overline{N} саф \overline{N} гооу. (36) \overline{M} піфдже єме2 и \overline{N} гоеіме \overline{N} теіміме.

Lesson 22

- 22.1 Possession is predicated by the use of oy \overline{N} and $M\overline{N}$ compounded with the preposition \overline{N} TG, \overline{N} TA. There are two sets of forms:
- I have (B) oy\+-(A) OYNTAL ΟΥΠΤΑΝ OYNTN-OYNTAK you have OYNTHTN ογῆτϜоүйтетй~ OYÑT€ etc. OYNTE-OYNTOY-PATRYO OYNTAY OYNTT-OYNTAC OYNTC-

ογπτε-πρωμε the man has

And similarly for the negative: (A) MNTAL I do not have; (B) MNT-. Set (B) is actually a reduced proclitic form of (A). Both sets may be accompanied by an untranslatable MMAY (there).

If the possessor is pronominal (i.e. suffixal), an immediately following object is unmarked:

(A) OYNTAY OYC2IMG.(B) OYNTY-OYC2IMG.He has a wife.

But if some word intervenes (and this is possible only in set A), the object is marked with \overline{N} ($\overline{N}NO^{2}$).

(A) ογπται πμαγ π ογαιμε He has a wife.

If the possessor is a noun, the object is usually not marked:

оуыте-проме оусгие. The man has a wife.

Pronominal objects are used only with set (A) and are attached directly to the subject suffixes. These are generally limited to the third person forms:

m.s. -4, $-c\overline{4}$ f.s. -c c.pl. -coy

as in OYNTAIT, OYNTAICT I have it (m.), $OYNTAK\overline{C}$ you have it (f.), OYNTATCOY he has them.

We have seen that the genitive is expressed with \overline{N} TE after indefinite nouns (oy2 \overline{N} 2 $\lambda\lambda$ \overline{N} TE \overline{N} Ppo), nouns with demonstrative prefixes (neixwome \overline{N} TE \overline{N} CON), and nouns with a following modifier (\overline{N} GHPE \overline{N} GALE \overline{N} TE \overline{N} Pome). NTA* is used similarly when the possessor is pronominal:

oyeMe גגג אדאו a servant of mine
ne וצשטשה אדא this book of yours
gom אד א ghpe אדא three sons of his

אדה, אדג may be used predicatively:

ογ \overline{N} -ογ \overline{N} нι \overline{N} τ \overline{N} 4. He has a large house. πнι \overline{N} 7 the house that belongs to him.

 \mathfrak{goon} $\mathfrak{n}\mathfrak{s}'$ is also sometimes used to predicate possession:

MN-2 בד goon אבו. I have no money.

The occasional use of \overline{m} mo' to indicate possession should also be noted. We have already seen an instance of this in the idiom $07\overline{n}$ -/ $m\overline{n}$ -60m \overline{m} mo' lit., there is/is-not power in.

22.2 Possessive pronouns, corresponding to English mine, yours, his, hers, etc., are formed by adding the appropriate pronominal suffix to m.s. $n\omega^*$, f.s. $\tau\omega^*$, c.pl. $no\gamma^*$; thus, $n\omega_1$, $n\omega_K$, $n\omega_1$, $n\omega_K$, $n\omega_1$, $n\omega_1$, $n\omega_1$, $n\omega_1$, $n\omega_2$, $n\omega_1$, $n\omega_2$, $n\omega_1$, $n\omega_2$, $n\omega_3$, $n\omega_4$, $n\omega_1$, $n\omega_1$, $n\omega_2$, $n\omega_3$, $n\omega_4$, $n\omega_1$, $n\omega_2$, $n\omega_3$, $n\omega_4$, $n\omega_1$, $n\omega_2$, $n\omega_3$, $n\omega_4$, $n\omega_5$, $n\omega_1$, $n\omega_2$, and similarly for $\tau\omega^*$ and $no\gamma^*$. When used as predicates of nesentences, they serve to predicate possession:

NXWWMG ETE NOY4 NE no 1 ne . neixoi noa ne. NOYK NE. TWK TE.

the books which are his It is mine. This ship is his. They are yours. It (f.) is yours.

The proclitic pronouns na-, ra-, and na- are used to express "that of, that which pertains or belongs to." Number and gender are determined by an understood or expressed antecedent. The exact meaning must be gained from the context:

им-пметот

the affairs of my father мечфире мм ма-печсом his children and those of his brother

NA-THOAIC NATEIMINE the inhabitants of the city people of this sort

22.3 The qualitative (continued). Many intransitive verbs of motion or position (e.g. mooge, A26PAT", 2Mooc) do not have a strong contrast in meaning between infinitive and qualitative, the process and state involved being about the same thing. A26 and 2Mooc are in fact qualitative forms that have usurped the role of the infinitives was and 2Mce for all practical purposes. But note the following:

Inf. ROK Q. **BHK** to be going, be on the way there to be coming, be on the way here, be 61 NHY about to come, be about to arrive to be fleeing, running, in pursuit пнт пωт to be near, nigh, at hand 2 W N 2 H N 666τ to remain, wait, stay, be 6ω to be enduring, lasting, continual MOYN MH N to be riding, mounted **AYE**

The infinitives er and Box may not be used in the First Present and Imperfect; only the qualitatives NHY and BHK appear in these conjugations. For the other verbs the

qualitative is preferred, but the infinitive is also found. The future nuance of NHY is especially noteworthy.

There are many intransitive verbs for which the infinitive and qualitative bear a "becoming"/"being" relationship to each other:

Inf. gwns to become, come into existence; Q. goon to be, to exist.

Inf. ww to become pregnant; Q. GET to be pregnant.

Included among these are many verbs with -o- or $-\lambda-$ in the final stem syllable:

Inf. Ngot to become hard Q. NAGT to be hard
oyxal to become well oyox to be well

2KO to become hungry
Alal to increase of to be great
oyon to become holy oyaas to be holy

Vocabulary 22

oyxxi to become sound, whole, safe; Q oyox to be sound, whole, safe; as n.m. health, safety, salvation.

муют, Q мают to become/be hard, harsh, difficult.

MTON, Q MOTN to become/be at ease, at rest, relieved; as n.m. rest, relief. The Q is also used impersonally: CMOTN it is easy (to do: 6, 6TP6).

oyon, Q oyaas to become/be pure, holy, hallowed.

אוגו. Q or to increase (in age, size, quantity); Q to be great, honored.

λοι, Q oo to become/be numerous, many.

n.x26 lifetime.

киме Egypt.

2 λ 2 adj. of quantity: many, usually before sing. noun with \overline{N} , as in 2λ 2 \overline{N} powe many men.

n. 2HT heart, mind, intellect.

- M necway (they) both, both (of them); used appositionally to another pronominal element, as in AYBOK M nechay they both went. Sim. for other numbers: Μποφομητ all three of them.
- goon Q to be, to exist; a predicate adj. is introduced with \overline{N} and has no article: Neugoon \overline{N} nonHpoc he was wicked.

Exercises

- A. (1) MÑTAN EIPHNH $2\overline{M}$ HEIMA. (2) OY \overline{N} TE-HAEIWT WMOYN \overline{N} **XOI.** (3) OYNTAI MMAY \overline{N} OYKOYI \overline{N} 2AT. (4) OYNTAU 2A2 \overline{N} **χωωμε.** (5) ΟΥΝΤΑΥ ΜΜΑΥ Ν COOY Ν ΕCOOY. (6) ΜΝΤΟΥ-ΟΕΙΚ. (7) OYNTTOOY2061TE N AC. (8) OYNTE MMAY N OY2A1? (9) OYNT TAC $\overline{M}MAY \overline{N}$ CAMY \overline{N} SUPPE. (10) OYNT-OYMTHN \overline{N} BPPE.
- B. (1) neignor not ne. \overline{M} nok an ne. (2) naxol $M\overline{N}$ naпасом (3) ма-пра (4) печтафос \overline{M} ма-мечетоте (5) паоетк $m\overline{n}$ па-нафвеер (6) тетсиче ток те. (7) нетапит ноуоу не. (8) TNOY_2 row re. (9) rentime hattexhya (10) rnoys m TO AN HE.
- C. (1) AYW \overline{N} TEYNOY AGOYXAL \overline{N} 61 HET GWNG. (2) HOYPAN NAOYOU XIN TENOY WA ENEZ. (3) AMAIA NEL RPWME 2N NEMZOOY.
- (4) ANKOTN 6 KHM6 $2 \overline{N}$ OY66NH. (5) $12 \omega B$ A4MKA2 6MAT6 6XWN.
- (б) мёмотй ам ером етремчі за меімкооз. (7) міпечсе-нрії емег гм печаге тирч. (8) а-пгит м прро м киме мфот очвич.
- (9) своуож пв и некфирс. (10) нечфаке ий нечавнуе нафт.
- (11) $CMOK\overline{2}$ GTPARICTOYO G NOKWAXO MN NA-NOKWBOOP. (12) A4-MOY2 \overline{M} \overline{N} $\overline{$ (14) c_{2} at nan etbe nekoyxai. (15) thacmove nek-PAN ET OYAAB. (16) NENXIXEEYE Og. (17) $\pi \lambda \lambda z$ e xhk ebol. (18) про \overline{M} печні очни. (19) печран от $2\overline{N}$ тетхора тир \overline{C} .
- (20) хүкх-печеронос ехт пвимх.
- D. (1) NENZTOWP ZKAEIT. (2) NEPETT ANHY EXT OYEIW. (3) TEAC2INE EET. (4) NEYEET MN NEYCYPTENHC. (5) NEYECOOY NEYGOOD 2N TCOGE DE. (6) TELX M DECELOR NECGOOD NAME OF DECELOR NECGOOD NECGOOD NAME OF DECELOR NAME OF DECELOR NAME OF DECELOR NECGOOD NAME OF DECELOR N

(7) Nepe-ne200y $\overline{\mathbf{m}}$ nechice 2hn 620yn. (8) n2hremon nhy 6 pakote (9) Neqoon as ne $2\overline{\mathbf{n}}$ $\overline{\mathbf{n}}$ xais ga ne200y $\overline{\mathbf{m}}$ neqoyon $\overline{\mathbf{n}}$ ebox $\overline{\mathbf{m}}$ $\overline{\mathbf{n}}$ thx. (10) Nepe-222 $\overline{\mathbf{n}}$ poms nht 21 te21h. (11) $\overline{\mathbf{n}}$ +2kasit an. (12) Neygoon as ne $\overline{\mathbf{m}}$ nechay $\overline{\mathbf{n}}$ alkaioc $\overline{\mathbf{m}}$ nehto ebox $\overline{\mathbf{m}}$ nnoyte. (13) $\overline{\mathbf{n}}$ gooc bhk $\overline{\mathbf{n}}$ $\overline{\mathbf{n}}$ the. (14) $\overline{\mathbf{n}}$ 0y $\overline{\mathbf{n}}$ -222 $\overline{\mathbf{n}}$ tet $\overline{\mathbf{n}}$ each 2 $\overline{\mathbf{n}}$ each 2 $\overline{\mathbf{n}}$ nequive (15) nmonaxoc neq2hooc $2\overline{\mathbf{n}}$ teqpi. (16) $\overline{\mathbf{m}}$ matoi alhy 21x $\overline{\mathbf{m}}$ nxoi. (17) Nepe-noyhhb gaha $\overline{\mathbf{n}}$ tepisi e20yn. (18) Nepe-oynos $\overline{\mathbf{n}}$ chas $2\overline{\mathbf{n}}$ neasix. (19) nnohoc $\overline{\mathbf{m}}$ nxoeic mhn ebox ga niene2. (20) $\overline{\mathbf{n}}$ tepeanay xe ceet, aa $\overline{\mathbf{n}}$ te neahi.

Lesson 23

23.1 The Circumstantial.

 6 I C W T M
 I, hearing
 ENC W T M

 6 K C W T M
 you, hearing
 ETET N C W T M

 6 P E C W T M
 EY C W T M

 6 C C W T M
 EY C W T M

ере-прыме сыты the man, hearing

The Circumstantial is used only in subordinate clauses modifying either a particular element of the main clause or the main clause as a whole. Such clauses describe an activity or state existing simultaneously with the time designated by the verb of the main clause and do not, in themselves, have a tense. They correspond to various English constructions: nominative absolutes, participial modifiers, or temporal clauses with "as, while, when" and a progressive verb form, Typical uses in Coptic include

(1) subject complement:

EIA26PAT 2A2TM пбрпб, AINAY бУNO6 M миндб.
Standing near the temple. I saw a great crowd.

(2) object complement:

Aγ26 6 πρωμε 642MOOC 2N ΤΑΓΟΡΑ.

They found the man sitting in the marketplace.

ANNAY epooy symoogs 21 Te21H.
We saw them walking on the road.

(3) complement to the entire main clause:

epe-nench2 xw \overline{N} neighxe, hynos \overline{N} gnhpe gwns. As our teacher was saying these things, a great wonder occurred.

If the context requires it, circumstantial clauses may also be translated as causal, concessive, or conditional clauses.

There are several important special uses of circumstantial clauses in Coptic:

- (1) They are regularly used as relative clauses to modify an indefinite antecedent. Contrast

 npome et eime e nagaxe the man who understands my words oppome eteime e nagaxe a man who understands my words

 Such indefinite antecedents include xaay, oya, oyon, and 2061NG. Further examples will be found in the exercises.
- (2) Certain verbs are regularly followed by the Circumstantial of a complementary verb:

хүмоүм євох єуфахє \overline{N} теуфн тнр \overline{c} . They continued talking the whole night.

ACAO CCPIME. She stopped crying.

(3) The Circumstantial of xω MMOC xe is regularly used to introduce direct quotation after appropriate verbs:

λθογωφΕ Νλγ, εθχω ΜΜΟC xe... He answered them, saying...

The Circumstantial is not negated. Instead, the

circumstantial prefix ε -, also called the circumstantial converter, is added to the negative of the First Present:

e-n+cwтм an I, not hearing
e-nrcwтм an you, not hearing

After ϵ - the syllabic pronunciation of N is given up; the stroke is not needed, but is sometimes retained.

23.2 Nouns as adjectives. In Coptic, as in English, a large number of nouns may do double duty as adjectives (cf. pencil sharpener, bookstore, brick wall, etc.). The order is reversed in Coptic, with the modifying noun second, preceded by the adjectival linking \overline{N} (\overline{N}):

OYMA N XAGIG a silver cup

OYMA N XAGIG a desert place

REGEMOT N COMA his corporeal form (lit. body-form)

OYGIGPO N KO2T a fiery river

Such items are very frequent, but not as freely formed as their English counterparts. In some cases two translations are possible: ογληοτ π μρ<math>π a wine cup οr a cup of wine. Note that, as with adjectives, the construction differs from the genitive by the absence of an article on the second noun.

Several words form a large number of compounds whose meanings are more or less completely predictable. Among these are

мы N (place of), as in мы N gwne dwelling place
мы N оүшм eating place, refectory
мы M мооше road, path
мы N кы-оетк pantry (place for putting bread)

CAN (seller of, vendor of, dealer in), as in

CAN TET fish-monger CAN HPN wine-seller

CAN AG meat-seller CAN 2AT dealer in silver.

A glance through the final Glossary will provide dozens of

further examples.

The nouns pome and crime often occur redundantly in this construction; the order of the nouns may be reversed:

TERCONG \overline{N} C21MG his sister (lit., woman-sister)

R2AMGG \overline{N} pome the carpenter (lit., man-carpenter)

RPOMG \overline{N} XAXE the enemy (lit., enemy-man)

Noun-noun modification does not always correspond exactly to English idiom, but little difficulty will be met in translating these constructions. Most of them will not be given separate listing in the vocabularies or Glossary.

Vocabulary 23

MOYN GBOX + Circum.: to continue (doing something).

 6ω + Circum.: to continue, persist in (doing something).

xo vb. intr. (1) to cease, stop, come to an end; + Circum.: to stop (doing something); (2) to leave, depart (from: ΜΜΟ΄, 2Ν, GBOX 2Ν). This verb has special Imperative forms: m.s. xxxx; f.s. xxx; c.pl. xxxxν.

ογω vb. intr. to cease, stop, come to an end; + Circum.: to stop (doing something), to finish (doing something), to have already (done something).

n.ae wood.

n. Benine iron.

өемеете monastery, convent.

MOOYT (Q of MOY) to be dead.

n. וסאאבארc the Jordan River.

Greek words:

ne.coyo grain, wheat.

т.тыпро mouth (also fig.).

axxa conj. but.

n. 2ωs N 61x handwork,

handicraft.

τ. περιχωρος (ή περίχωρος) surrounding countryside.

τ.Μετανοια (ή μετάνοια) repentance.

τ. λ ποθηκη (ή άποθήνη) storehouse, barn.

π.Δεμων, π.Δεμων, π.Δεμον (ὁ δαίμων) evil spirit, demon.
πε.στλγρος (ὁ σταυρός) the Cross; usually written πες Ρος.

Exercises

A. (1) OYOY20P EMMOOYT (2) OYCYNAFWFH ECKHT $2\lambda 2T\overline{N}$

тагора (3) оурыне ечоуна 21 пхаете (4) оуамааа е-нчсшты ан пса печхоете (5) оувире фин еч оуве печсон (6) оуахаю ессие (7) оуант ечнафт (8) оуетрини е-нсмии евох ан (9) оусатие есеет (10) аенанке еуакаетт (11) оуати е-нсмоти ан (12) оупих ечоухав (13) песоуо ет ки ап тапоенки (14) оуминфе ечоф (15) оуматот ечахну ех \overline{n} оуато

В. (1) генфахе \overline{n} ме (2) оүро \overline{n} веніпе (3) оүс \overline{poc} \overline{n} фе (4) оүн \overline{n} шие (5) генсиауг \overline{n} веніпе (6) \overline{n} фахе \overline{m} метанова (7) пама \overline{n} фшпе (8) оүсмот \overline{n} аггелос (9) тпістіс \overline{m} ме (10) оүснче \overline{n} кшг \overline{r} (11) оүнустнріон \overline{n} ноуте (12) пенма \overline{n} оүшм (13) генматов \overline{n} хахе (14) оуапот \overline{n} ерште (15) оуна \overline{n} гарег

C. (1) ϵ_{N2MOOC} $2\overline{N}$ TATOPA, ANNAY ϵ π_2 Hremwn ϵ_{42} wn ϵ_{2} oyn. (2) THASO M HEIMA ELGOOT EBON 2HTT M HEZOOY M HXOELC. (3) EPE-NECNHY MOODE E BENEETE, AYZE SYPOME EMMOOYT EMKH 21XM $\pi \kappa \lambda_2$. (4) Aymoyn ebox eyeine \overline{M} necoyo e tanoehkh. (5) алити! ифоүшф им е сфты е зенфихе и тетитие. ероч ечвик евох $2\overline{N}$ өемеете. (7) м \overline{N} там соуо $2\overline{N}$ темапоенки, ALLA OYNTAN MMAY \overline{N} 2A2 \overline{N} \overline{g} 6. (8) COTM \overline{e} \overline{N} \overline{g} AX \overline{e} \overline{N} TATANPO, EBOX XE 26NME NE. (9) AINAY \overline{N} OYOY2OP E441 \overline{N} OYKOY1 \overline{N} 6 POOM TE $2 \overline{N}$ TE $4 T \lambda$ TPO. (10) λ YE $1 \phi \lambda$ PO $4 \overline{N}$ 6 1λ NET OYH $2 \overline{N}$ THE PIX WPOC $\vec{\mathbf{M}}$ HIOPAINHC THROY. (11) NAI NE $\vec{\mathbf{N}}$ PAN $\vec{\mathbf{N}}$ NECHHY ET HIT E KHME. (12) NEICOOYN AN HE XE NEIXWWHE NOYK NE. (13) бигни блом в тполіс, жиму бумнифе блоф блинт бвох 21 \overline{n} THYXH. (14) EMMOODE 2 λ 2 \overline{m} HEPHE, λ 4N λ Y EYB $\overline{\lambda}$ X \overline{n} 2HKE 64+ 6802 \overline{M} ne4208 \overline{N} 61x. (15) $M\overline{N}$ -60M \overline{M} MON 6 NOYX6 6802 \overline{N} 2 6ΝΔΔΙΜΌΝ Ν ΔΚΑΘΑΡΤΟΝ. (16) ΑΠΟΥΌ ΕΠΟΖΑΙ ΑΥΌ ΑΠΘΟΏΤ Ε2ΟΥΝ ϵ про \overline{N} течрі. (17) гап \overline{C} ером етремаю євоа г \overline{M} пеіма. (18) λ C6w ecrime exm imoy m nechepit n 21. (19) λ 4 λ 0 eqeine NAN \overline{H} neqzob \overline{N} 61x. (20) \overline{TC} Ae, each ebol \overline{H} \overline{HNA} eqoyalb, AUKOTT GBOX 2M ΠΙΟΡΑΔΝΗς, ΕΥΜΟΟΦΕ 2M ΠΕΠΝΆ 21 ΤΕΡΗΜΟς Ν 2λ2 й гооу, бупбірахе йноч гітй паіаволос, аую йпбчоуби-лалу гй NEZOOY ET MMAY. NTEPOYXOK AE EBOA, A42KO. (21) A12MOOC ϵ 1C 2 λ 1 \overline{N} gom to \overline{N} oynoy. (22) Lymoyn obol gyping \overline{N} toygh

тнрс. (23) мпенею енфана птерпсютм е на і. (24) й теуноу асоую есфине аую асоуха і. (25) а іло є і+ й геноє ік нау. (26) а фахе нямау ечеп і+ма нау. (27) мп-оуха і фооп й нете псесютм ан пса нечентоль. (28) ауагератоу гагтм печс+ос еуріме. (29) ачеп і+ма нау ечхю мнос хе, "мпрфахе й халу + рюме етве пе і гюв."

Lesson 24

24.1 The Second Present has exactly the same inflection as the Circumstantial. This ambiguity poses a serious difficulty for the reader of Sahidic Coptic which can be resolved only by a careful study of the context. The uses of the Second Present parallel those of the Second Perfect:

(1) emphasis on an adverbial element:

epe-Nal goon MMOI ette NaNore.

It is because of my sins that these things happen to me.

(2) preceding various interrogative expressions:

Whom do you seek?

GUPING G OY?

Why is he weeping?

GUTON?

Where is he?

When $\tau\omega n$ is used with a nominal subject, the usual idiom is eq $\tau\omega n$ N? Where is N?, without the expected $\pi \epsilon_1$:

ечтым пекстыт? Where is your father?

The alternate construction ($\epsilon_P \epsilon^- n \epsilon_K \epsilon_L \omega_T \tau \omega_N$?) is less frequent.

Clauses containing second tense forms are negated with an:

610YH2 2M N61MA AN. It is not here that I dwell.

NTALLAG NAK AN. It is not for you that I did it.

As may be seen from the translation, the negation applies to the adverbial element and is not a negation of the verb proper.

24.2 The Bipartite Conjugation (Present-Imperfect System). The First Present, its relative forms, the Circumstantial, the Second Present, and the Imperfect comprise a system:

Pres. I 4coth приме сотн Rel. Pres. I $\left\{ \begin{array}{ll} \text{етчсоты} & \text{приме соты} \end{array} \right.$ Сircumstantial счсоты еге-приме соты Pres. II $\text{ечсоты} & \text{ере-приме соты} \end{array}$ Імрегяест

Following the penetrating analysis of H. J. Polotsky (see Bibliography), Coptic scholars now refer to this system as the Bipartite Conjugation. This term arises from the fact that the base form, the First Present, consists only of subject + predicate, with no conjugational prefix. The remaining forms of the system consist of this bipartite nucleus preceded by a set of elements called converters: the relative converter et/etepe, the circumstantial converter e/epe, the second tense converter e/epe, and the imperfect converter Ne/Nepe. The term tripartite is applied to all other Coptic verbal conjugations, which consist of a verbal prefix + subject + predicate, e.g. the First Perfect A'q-cwtm, A-npwme cwtm. The First Future is a special case and will be treated in the following lesson.

The conjugations belonging to the Bipartite Conjugation may have three kinds of predicates: infinitives, qualitatives, or adverbial predicates (i.e. adverbs or prepositional phrases). In the tripartite conjugations only the infinitive may be used. The conjugations of the

Bipartite Conjugation, as we have already seen, characterize an action as durative, continuing, or (less commonly) habitual. The following features of the Bipartite Conjugation are equally distinctive:

- (1) The First Present requires the use of $oy\overline{N}$ (neg. $m\overline{N}$ -) before an indefinite subject (e.g. $oy\overline{N}$ -oypome $cot\overline{M}$). The use of $oy\overline{N}$ -/ $m\overline{N}$ is optional after the converters, e.g. Nepe-oypome $cot\overline{M}$ or Ne-oy \overline{N} -oypome $cot\overline{M}$.
- (2) Apart from the use of $m\overline{n}-$ just mentioned, negation is universally with (\overline{n}) ... λN .
- (3) An infinitive cannot, in general, be used in the prenominal or prepronominal form, i.e. prepositional direct object markers (MMO*, G, etc.) must be used. This rule, known as Jernstedt's Rule (see Bibliography), has the following exceptions:
 - (a) the verb ογωφ ογεφ- ογλφ', which may occur in all forms; e.g. †ογωφ Μπου οτ †ογλφΨ.

 - (c) certain types of compound verbs; see 26.1.

The Imperfect may be expanded into a subsystem of its own by the prefixation of the other converters:

Imperfect Neqcotm Nepe-npome cotm
Imperfect Rel. eneqcotm enepe-npome cotm

ete neqcotm

Imperfect Circum. 6-N64CWTM 6-N6P6-npwM6 CWTM

These forms have all the characteristics of, and belong to, the Bipartite Conjugation. The relative forms have already been introduced. The circumstantial forms are used syntactically exactly like the Circumstantial (of Pres. I). The past tense of the action is explicitly marked, however, while in the Circumstantial it must be gained from the context. Second tense forms of the Imperfect may occur, but

they are too rare for consideration here. All verbal forms containing the imperfect converter may be followed by me.

24.3 Numbers (continued). The 'teens are formed by prefixing MNT- to special forms of the units. MNT- is a proclitic form of MHT ten:

11 m. митоус; f. митоусі 15 m. f. митн

12 m. митснооус; f. митснооус(е) 16 m. f. митасе

13 m. f. мптфонте 17 m. f. мптсафч (e)

14 m. f. MNTAGTE

18 m. f. митфиниє

Construction is the same as that of the units:

м $\overline{\mathbf{n}}$ т \mathbf{g} ом \mathbf{t} \mathbf{g} $\overline{\mathbf{n}}$ \mathbf{g} \mathbf{g} \mathbf{m} \mathbf{g} \mathbf{m}

Vocabulary 24

rωz rez r paz r' Q paz r vb. tr. to strike, kill (μmo'); to strike down, cast down.

COBΤς CETOT Q CETOT vb. tr. to prepare, make ready (ΜΜο΄; for: ε); intr. and reflex. to get ready.

xice xecT- xxcT' Q xoce (± e2px1) vb. tr. to raise up, exalt (MMO'; over: e, exM, 21xN); intr. to be exalted; as n.m. heights. net xoce the Almighty.

DYGING vb. intr. to pass (subj. usually period of time).

КІМ КӨМТ- КӨМТ vb. tr. to touch (e; with: e); to move, shift, stir (ммо°, e); vb. intr. to move. stir. be moved.

дине vb. intr. to be ashamed (about: єтвє); as n.m. shame.
gine 2нт to revere, be humbled before.

goyeir Q to be empty, vain.

200Y Q to be bad, wicked.

TONTN TNTN- TNTWN' Q TNTWN vb. tr. to liken, compare (μmo'; to: ε, μn, εxn).

COTH COTH COTH Q COTH vb. tr. to choose, select ($\overline{M}MO'$); Q also = to be excellent, exquisite.

MOYOYT MGYT- MOOYT' vb. tr. to kill $(\overline{M}MO')$. n. THHEG finger.

6 oy why? for what reason?

giнт Scetis, the Lower Egyptian center of monasticism, in the Western Delta.

πε.προφητης (ὁ προφήτης) prophet.
π.λποστολος (ὁ ἀπόστολος) apostle.

Exercises

А. (1) 2 en2 ione 6-neyeiw \overline{n} 2 en2 of ite (2) песмот еntaquetty \overline{n} \overline{n}

В. (1) пийтсиооус й апостолос (2) петдомйт й маентнс (3) мйтачте й генеете (4) сарч й алиши (5) дмоун й нт еудоуетт (6) мнте й ноб й ехоуста (7) мйтсиооус й сгине (8) мйтн й гооу (9) мйтоует й ромпе (10) мйтоуе й евот

С. (1) ечтонтй ммок е нім? (2) естин тафтни й врре? (3) сенажасте егра ежй негіоне тироу йте пеікосмос. (4) ситй нак й чтооу й риме. (5) йтсре-теромпе ет ймау очеіне, аукотоу е пеутме. (6) й пегооу ет ймау тетнафіпе етве неіггимоу е оу? (7) ачге ежй пкаг аую йпечкім. (8) йтачтйтимоу е оу? (9) ере-наі фип е пенховіс етве печна. (10) ечоущь е рагт етве пентаілач оуве на-печтме. (11) оуйтоугооу ину ечгооу. (12) еутин ненфвеєр? (13) ачкім є течтапро е печтниве. (14) тйнасмоу е пекран ет хосе (15) ечтйтим еуфире фим. (16) мйносма ав а-несину котоу е фінт. (17) етве оу коущь е моуоут й неіриме? (18) ететйсовте ммитй е оу? (19) аумеут-оуон нім ете неуоунг гіт пты нім міт тперіхирос. (20) ауге е тетрапеха есебтют.

Lesson 25

25.1 The relative, imperfect, circumstantial, and second tense converters may be used with the First Perfect, the First Future, existential and possessive predications, and copulative sentences with me, TG, NG. The relative forms for all of these have already been discussed. The second tense of the First Perfect, i.e. the Second Perfect, was introduced in Lesson 14. The second tense forms of existential, possessive, and copulative sentences are too rare for inclusion here.

 (a) First Perfect
 A4CWTM
 Neg. МП64СWTM

 Perf. I Rel.
 6HTA4CWTM
 6T6 МП64СWTM

 Perf. I Circum.
 6-A4CWTM
 6-MП64СWTM

 Pluperfect
 N6-A4CWTM
 N6-MП64СWTM

 Second Perfect
 NTA4CWTM
 NTA4CWTM

The imperfect of the First Perfect (Ne-λαςωτῶ) corresponds to the English pluperfect: he had heard, he had written. The circumstantial of the First Perfect is used to describe an action as completed prior to the tense of the verb in the main clause.

6-хчгмоос, хчсгхі...Having sat down, he wrote...хмг 6 броч 6-хчмоу.We found him dead (lit.,
having died).

 (b) First Future
 чилсотй
 проме илсотй

 Fut. I Rel.
 етчилсотй
 етере-проме илсотй

 Fut. I Circum.
 ечилсотй
 ере-проме илсотй

 Fut. I Imperfect
 иечилсотй
 иере-проме илсотй

 Second Future
 ечилсотй
 ере-проме илсотй

The circumstantial of the First Future describes an action as imminent, about to take place, with respect to the tense of the main clause:

As I was about to leave, he summoned me.

ANZE GPO4 GANAMOY. We found him on the point of death.

The imperfect of the First Future describes an action as imminent in past time:

אבגאום e nxoi (ne). I was about to get on the ship.

This form is commonly called the *imperfectum futuri*. The Second Future $(\epsilon \eta N \lambda C \omega T \overline{M})$ has all the normal uses of a second tense form. Special uses of both these conjugations will be mentioned later on.

The First Future and its related system are formally an off-shoot of the Present System, with NA- inserted before the infinitive. It has no other characteristics of the Bipartite Conjugation, however: (1) it is not durative (except with certain aspectually neutral verbs, e.g. page); (2) only the Infinitive may occur in predicate position;

- (3) the prenominal and prepronominal forms of the Infinitive occur freely.
 - (c) Existential and

Possessive OYN-OYNTAY MN-MNTAYRelative ETE OYN-OYNTAY ETE MN-MNTAYCircumstantial E-OYN-OYNTAY E-MN-MNTAYImperfect E-OYN-OYNTAY E-MN-MNTAY

The circumstantial forms describe a state simultaneous to the tense of the main clause:

6-мм-обік ммау, амвшк бвоа. There being no food there, we left.

אונה פּףסי פּ-א \overline{N} -60א אואס פּ שאב. We found him unable to speak.

The imperfect forms simply place the state in past time: $N\varepsilon - O\gamma \overline{N} - (Or N\varepsilon \gamma \overline{N} -) O\gamma \rho \omega M\varepsilon \overline{M} M\lambda \gamma (n\varepsilon).$ There was a man. $N\varepsilon \gamma \overline{N} T\lambda 4 \ 2\lambda 2 \ \overline{N} C 2 IMC (n\varepsilon).$ He had many wives.

(d) Copulative sentences with πε, τε, κε:

Relative $\varepsilon \tau \varepsilon$ oych2 $\pi \varepsilon$ $\varepsilon \tau \varepsilon$ \overline{N} oych2 λN $\pi \varepsilon$ Circumstantial ε -oych2 $\pi \varepsilon$ ε -N oych2 λN $\pi \varepsilon$ Imperfect $N \varepsilon$ -oych2 $\pi \varepsilon$

The circumstantial and imperfect are used as above.

The circumstantial forms of all the subsystems listed above have a frequent use as relative clauses after indefinite antecedents:

оуром с с-лчкет-оун a man who had built a house оумустнр он бүнлболп свол a mystery which is about to be revealed оухнрл с-мптлс фирс ймлү a widow who has no son оуфирс с-оухирл те течмллу a boy whose mother is a widow The circumstantial converter срс- is sometimes used

improperly for c- before copulative sentences.

25.2 The Conjunctive.

 (N) ТАСФТМ
 ПТПСФТМ
 ПТЕ-приме сФТМ

 NT СФТМ
 ПТЕТИСФТМ

 NT GC ФТМ
 ПС GC ФТМ

 \overline{Nr} , \overline{Nq} , and \overline{Nc} also appear frequently as \overline{Nr} , \overline{Nq} , \overline{Nq} . The conjunctive is used to continue the force of a preceding verbal prefix. In a sense, it is no more than an inflected form of the conjunction "and." It is especially frequent after a First Future or an Imperative:

It may be used to continue the force of virtually any preceding verbal prefix except that of the affirmative First Perfect, but even this restriction does not hold in

the relative forms. It is also used after an Inflected Infinitive, as in

гапс ерон етренвык птпраже иймач.

It is necessary that we go and speak with him.

In many instances, especially where there is a change of subject, the Conjunctive clause has the meaning of a purpose or result clause:

This usage depends very much on the presence of an injunctive (imperative) force, implicit or explicit, in the first clause. For the conjunctive with Greek conjunctions, see Lesson 30.

The Conjunctive resembles the Tripartite Conjugation: only the Infinitive may be used as its verbal component. Negation is with $-\tau \overline{\mathsf{M}}-$ before the Infinitive. If the Conjunctive continues a negative verb, however, the negation may carry over.

Vocabulary 25

- оує, Q оуну vb. intr. to become/be distant, far (from: є, тмо , єкох тмо); as n.m. distance. є поує away, to a distance. т по є at a distance.
- TA20 TA26- TA20' Q TA2HY vb. tr. (1) to cause to stand; to create, establish (Μmo'); (2) to reach, attain, catch up to (Μmo'); to seize, arrest (Μmo').
- $6ων \overline{\tau}$, Q $6ον \overline{\tau}$ vb. intr. to become/be angry, furious (at, against: ε, $εx\overline{ν}$); as n.m. wrath, fury.
- \overline{M} π νb. intr. to be worthy, deserving (of: \overline{M} M o'; to do: \overline{N} , \overline{G} + Inf.).
- тако таке- тако Q такну vb. tr. to destroy, put an end to $(\overline{N}$ Mo'); intr. to perish; as n.m. destruction, perdition.
- ωμ \overline{c} \overline{c}

n.whn tree. п.тър branch. n. excove grape. $n.2\lambda\lambda$ HT (pl. 2 $\lambda\lambda$ ATE) bird. $n.MA \overline{N} \in \lambda$ OOAE vineyard. T.XENERWP roof.

2ωλ. Ο 2Hλ vb. intr. to fly. τ. Bω N ελοολε grape-vine. T.Bw tree, vine. Bw is used when type of tree is mentioned: use why otherwise.

Exercises

(1) THE \overline{N} EXOCAE OYHY AN EBOX $2\overline{N}$ THE. (2) ϵ -AYTAKO \overline{N} THONIC, AYNO GBON. (3) 2AH \overline{c} STPSKCOBTS NAM \overline{N} OYMA \overline{N} NKOTK. (4) $\lambda YT\lambda 26 - \overline{N}QHPE EYMHP \overline{N}N\lambda 2P\overline{M} \Pi 2HFEMON. (5) <math>\lambda MHEIT\overline{N} \overline{N}TET\overline{N}$ сф $\overline{\mathbf{n}}$ е течсво. (6) негазерат $\overline{\mathbf{n}}$ поче в 16 ф $\overline{\mathbf{n}}$ е пмннфе. (7) +NABWK NTAGNTY. (8) NEPE-NECHTY EINE M HKAPHOC E THOALC NCET MMOY EBOX 2N TAPOPA. (9) NEYNAPW2T MMOY NEI MMATOL $\overline{\mathbf{n}}$ xaxe. (10) a-n2akht 2wa e the ayw a40yw2 ex $\overline{\mathbf{n}}$ oytap $\overline{\mathbf{n}}$ te TOWNS (11) NAI NE TOAXE ENTAGEAICOY 21 THAS 2M TESTHEBE. (12) ϵ -ачтооун $\overline{\mathsf{N}}$ 61 n 61 n 61 a 48 w 6 b 60 c 4 e 4 p 4 w 6. (13) n 6-0 v 7тач \overline{M} мау \overline{N} оукоу \overline{N} фире ечене. (14) сепатагоч \overline{N} сепох \overline{q} е пефтеко. (15) не-оуапістос пе пеу \overline{p} ро. (16) кнафіне \overline{n} сфі \overline{H} nezony et \overline{M} ny \overline{H} ne \overline{M} noi. (17) \overline{H} \overline{H} no an etpey-COTHT. (18) ANZE E HMA \overline{N} EXODAE ESTAKHY. (19) 6-A-GOM \overline{N} T \overline{N} GBOT OYGING, ACKOT \overline{C} G RECHI. (20) NE-M \overline{N} -460M \overline{M} MO4 G таге-мечфвеер. (21) е-ачеши $\bar{\tau}$ ех \bar{m} печсои, ачтшоүн ехшч, A4MOOYTH. (22) NEPE- \overline{N}_2 AAATE \overline{N} THE OYOM EBOA $_2\overline{N}$ NEAOOAE. (23) ANNAY ϵ negation equipment fine the normal parameter ϵ TAKON. (25) GINANKOTK, A-MAZMZAA GING NAI N TEKEMICTOAH. (26) AYARE $2\overline{N}$ OYGERH E TXENERWP. (27) NEYN-OYFPO NCABE ϵ -оүйтач фомит $\overline{\mathbf{n}}$ фире. (28) גואוג $\overline{\mathbf{n}}$ оүноб $\overline{\mathbf{n}}$ гахнт ϵ чоүнг 21XN OYBW N 620026. (29) TNNANAY NTNEIME NTNGINE EMATE. (30) AUKIM E NEUTHHBE E $\overline{N}BAA$ \overline{M} $\overline{N}B\overline{A}AE$. (31) \overline{N} TEYNOY AUAO EMBONT. (32) AYEL E BHONGEM EBON ME NE-AYCOTH ETBE THICE н пенсытнр. (33) א- חאאסו שאב זיה петеро. (34) не-оуп-оу-NO 6 \overline{N} gtopt \overline{p} 2 \overline{N} tholic. (35) $\lambda - \overline{N}$ 2 alate by each tenenup м пн п. (36) меумпу м дооп м пеппа ет оузыт гм пеугнт. (37) NGYXI N NTAP NCGNOYXG MMOOY 21 TG21H.

Reading

The following selection is from the Sayings of the Fathers. See p.146 for a brief description of this text.

Note: The term $\lambda n \lambda$ is a title of respect, ultimately from Aramaic 'abb \overline{a} , father. MAKAPIOC is a proper name.

Lesson 26

26.1 Compound verbs. Coptic vocabulary is particularly rich in compound verbs. Most compound verbs consist of a simple infinitive in the prenominal form plus a nominal element, usually without an article, e.g. †-eooy to praise, xi-banticma to be baptized. Meanings are for the most part predictable from those of the components.

The verbs most frequently occurring in compounds are +- to give, \times 1- to take, 41- to raise, carry, $6\overline{N}$ - to find, KA- to put, and \overline{F} - to do, make. Some examples:

+- kapnoc to produce fruit

+-метамога to repent; to humble or abase one's self

†-εοογ κλ° to praise

+-csw Nx* to teach someone (something: 6)

x1-c8 to receive instruction, be taught (something: ϵ) $\epsilon \overline{N-M}$ to find rest

6 N-2ωB NN to have dealings with

 $6\overline{N}-60M$ ($6\overline{N}-60M$) to have power, prevail (over); to be able (to do: 6+ Inf.)

41-pooyg to take heed, be concerned (for, about: 6, NA', 6TB6, 2A).

Compounds with \overline{p} - are the most frequent of all and fall into two groups. In the first group \overline{p} - has its basic meaning "to do, make, perform":

 \overline{P} -NOBE to sin (against: ϵ) \overline{P} -NAI to do this, thus \overline{P} -OY to do what?

 \overline{P} -X \overline{N} powne (X is a number) has two meanings: (1) to reach the age of X; (2) to pass X years.

In the second group of \overline{p} - compounds \overline{p} - has the meaning "to become," e.g. \overline{p} - \overline{p} -p to become king (over: $ex\overline{n}$). The second element may be virtually any noun or adjective in the language, so that a complete catalogue is impossible. Qualitatives are uniformly o \overline{n} , as in o \overline{n} \overline{p} -p0 to be king. Further examples:

 \overline{F} -2 \overline{X} AO to grow old; O \overline{N} 2 \overline{X} AO to be old \overline{F} -2Hremwn to become governor; O \overline{N} 2Hremwn to be governor. \overline{F} -xOGIC to become lord, master (over: 6, \overline{exn}); O \overline{N} xOGIC

to be lord, master.

The distinction between these two groups is often blurred, however, with qualitatives of the o \overline{N} type being extended to the first group as well, e.g. \overline{p} -gnhpe to marvel, become amazed (at: \overline{M} MO°, e, erse, $\overline{ex}\overline{N}$), to admire; Q o \overline{N} gnhpe to be amazed.

Less frequently the nominal element of a compound verb has the definite article:

F-πωεφ to forget (N)
F-πμεθγε to remember (N)
+-Θε Νλ΄ to provide the means to someone (so that: ε,
ετρε).

In the case of \overline{p} -nws \overline{g} , \overline{p} -nmesys, and many others of this type a pronominal object is expressed by a possessive prefix on the noun: \overline{p} -neqws \overline{g} to forget him, \overline{p} -neqmesys to remember him.

Because compound verbs employ the prenominal form of the infinitive, the question arises concerning their occurrence in the Bipartite Conjugation, where the prenominal form is usually prohibited. In general, compound verbs are an exception to Jernstedt's Rule and may be used freely as they stand in the Bipartite Conjugation. Two types of compounds, however, do tend to follow Jernstedt's Rule:

(1) the type F-nmeeye, with the definite article on the noun. In the Bipartite Conjugation the full form of the infinitive is used. Contrast

> жир-печмесус. I remembered him. фенре й печмесус. I remember him.

(2) many compounds whose nominal element is a part of the body. Contrast

λι†-τοοτ̄c.
I helped her.
†† ν̄ τοοτ̄c.
I am helping her.

26.2 The element \mathfrak{g} -, $\mathfrak{e}\mathfrak{g}$ -, originally a full verb "to know, know how to," may be prefixed to any infinitive to express "can, be able." E.g.

Mпечшвшк He was not able to go.

 \overline{N} + NA \overline{M} + TOOTK AN. I shall not be able to help you. It occurs redundantly and optionally in the compounds of 60M: $O(\overline{N})$ (9) 60M, \overline{N} (9) 60M, \overline{N} (9) 60M.

26.3 Infinitives of the type take. There is a fairly large group of verbs whose infinitives begin with τ - and end in -0, e.g. take take- take Q takey to destroy. At an older stage of Egyptian these verbs were compound causatives with a form of \uparrow (to give) plus a verbal form inflected by suffixation. Thus, the original construction

involved two verbs (e.g. I caused that he pay a fine) which coalesced into a single verb with two objects (I caused him to pay a fine). Traces of the older construction survive in Sahidic, e.g. Luke 3:14 \(\text{NnPTTG-XXXY}\) oce Do not make anyone pay a fine (i.e. suffer a loss). TTO TTG- is the causative of \(\text{†}\) itself. The lack of an object marker on the second object is characteristic of the construction, but the absence of an article in this particular example stems from its association with the compound verb \(\text{†-oce}\) to pay a fine, suffer a loss. In general, however, there is no need to take the older construction into account in Coptic, since most of these verbs are simply transitive. Some examples:

about: e, erse; that: xe); causative of eime.

TANO TANO Q TANHY (± 62 PA1) vb. tr. to cause to go up, cause to board, cause to mount; to raise up, offer up, send up (Μπο); caus. of λλ6.

Tanzo Tanzo Q Tanzhy vb. tr. to bring (back) to life, let live, keep alive (\overline{M} Mo'); caus. of \overline{w} No.

τ + φ results in initial x:

xno xno xno vb. tr. to give birth to (mmo); to acquire, obtain, get (mmo; often with reflex. dative nx for one's self); caus. of gone.

xnio xnie- xnio' Q xniht vb. tr. to put to shame, to blame, scold, reproach (ΜΜΟ'; for: eTBG, eXN, 2λ); caus. of wine.

Sometimes the initial r- is lost, as in

κτο κτε- κτο' Q κτηγ vb. tr. to turn; this verb has become completely synonymous with its base κωτε.

A few verbs have retained a final -c or -oy (a frozen subject suffix):

xooy xey- xooy' vb. tr. to send (\overline{M} Mo'; to: \overline{G} PAT', NA', \overline{G} X \overline{N} , \overline{G} X); + \overline{G} Roox away, out, off; + \overline{G} X \overline{N} ahead.

TNNOOY to send (already introduced). Originally xοογ meant "to cause to go" (caus. of go to go) and τηνοογ meant "to cause to bring" (caus. of εινε).

τογνος τογνος τογνος vb. tr. to awaken, arouse, raise up (Μπο΄); caus. of τωογν (probably).

The Imperative of these verbs may optionally have a prefixed ma-: matamo, matamo, etc. Cf. §17.1.

Vocabulary 26

(The compound verbs given in 26.1, the prefix g- in 26.2, and the verbs tamo, ta

wв g єв g- ов g Q ов g vb. tr. to forget, overlook, neglect (ммо"); intr. to sleep, fall asleep; as n. forgetting, sleep.

 $\omega N\overline{z}$, Q $ON\overline{z}$ vb. intr. to become/be alive, live; as n.m. life. o the Q of ergs.

π. ροογω care, concern, anxiety. \overline{p} -ροογω (Q o \overline{n}) to become/be a care or concern (for: $n \times r$).

те. финре wonder, amazement, miracle.

†-тоот", † \overline{N} тоот" to help, assist (object suffix is required; nominal object with \overline{N}).

τε. θγειλ (ή θυσία) offering, sacrifice.

π. βλητισμα) baptism. +- βλητισμα to baptize.

Exercises

(1) петыни ае и \overline{q} — карпос аи. (2) и \overline{q} ей — еом ай е тако \overline{n} ие— фухн \overline{n} йа іка іос. (3) петса 2 петйх і — свю йтоот \overline{q} . (4) а \overline{q} — метанота ечхю ймос же а і \overline{p} — нове, пахое іс. (5) тйиах ісе ймоч ей — еооу \overline{m} печрай ет оуаав. (6) етиа \overline{p} — оү? (7) й тере \overline{q} — ий тоооус \overline{n} ромпе, а — иечетоте йт \overline{q} е перпе. (8) с 2 ати и і и \overline{r} тамот етве и \overline{g} в нуе еткетре ймооу ймау. (9) аую \overline{n} теуноу а — тво \overline{n} ехооле \overline{q} — гар \overline{n} карпос. (10) ейиа \overline{n} — \overline{n} тои \overline{m} петкосмос? (11) \overline{q} і — рооуф етве и егіфре \overline{n} тет \overline{n} гарег ерооу евол \overline{g} пперооу. (12) и \overline{q} — с \overline{n} и и \overline{n} е и ей \overline{n} пхое іс.

(13) \vec{N} +OYOO AN 6 6 \vec{N} -208 MM NA-TEIMINE. (14) \vec{M} MIIF-NOBE ερωτ \overline{N} ενες. (15) \overline{N} ΤΟ4 πετ Νλ \uparrow - \overline{M} ΤΟΝ ΝλΝ. (16) πελλητ δε $\overline{\mathsf{M}}$ ΠΕΘΦ $\overline{\mathsf{M}}$ -60M 6 2WA 6BOA. (17) ε-ΔΘΕΙ 62ΟΥΝ ε ΠΕΡΠΕ, ΔΥΤΆλΟ \overline{N} OYOYCIA. (18) CGNACHOY EPOH \overline{N} CET-600Y NAH. (19) \overline{M} RE-OYOGIQ TETNAQ6M-60M G TANZG-NGT MOOYT. (20) HAI NG HMA ETCNAXIO \vec{M} necompe \vec{N}_2 HT \vec{q} . (21) where \vec{p} of the transfer of \vec{N}_2 HT \vec{q} is a constant. 22 NEXHPA MN NOPPANOC. (22) A4X110 NA4 N NKA NIM GNTA- π_{642} HT OYAGOY. (23) NTAKF-OY 2N THOALC? (24) 61NA+-CBW инт \overline{N} е оү? (25) λ -Nequaxe жпіооу дую дупют свод. (26) Япе-THE THE THE PLANT OF THE PLANT $\pi G = 2 \omega B$, $\lambda q K T O q G \pi G = 1 m G = 0$ NAM. (29) \overline{N} \overline{N} \overline{N} TATET \overline{N} P- \overline{N} 1 6 OY? (31) AIP-MATO1 6P6-2HPWAHC O \overline{N} 2HCGMWN. (32) \overline{N} TEPESTOYNOC \overline{M} MOC, \overline{A} COYXAI \overline{N} TEYNOY. (33) 640 \overline{N} 2 \overline{X} \overline{A} 0, мй-бом ймоч в вык буполіс бсоуну. (34) сеплеі йсетако й петрие. (35) Мирр-пово \overline{N} нентолн \overline{M} пномос. (36) дутагоч ечноофе мй мечмаентнс. (37) $\vec{m} \vec{n} \vec{p} \vec{p} - \vec{n} \vec{w} \vec{w} \vec{m} \vec{a} + c \vec{w}$. (38) toyωσ ε τάμοκ αε πεκφηρε ογοά. (39) Νίμ πεντά t-θε Νητ \overline{N} етрететикот и оуні и теіміне? (40) фиар-пекмеєує итатиовок. (41) KNAP-XOEIC E NEIMOKMEK \overline{M} HONHPON. (42) λ -NEYMAGHTHC TAXOU 6 DIXOI. (43) \overline{N} TEPOYCOTM 6 NAI, AYP-ORHP6. (44) OY HET NATOR NAN ETPENONE DA NIENEZ? (45) HAZTE ZAPAT NEFжобіс ежі негелоуста тироу. (46) птерпр-печменує, анархет $\overline{\mathbf{n}}$ pime. (47) $\overline{\mathbf{n}}$ tepe-ne200y $\overline{\mathbf{n}}$ nechice xwk ebox, acxno $\overline{\mathbf{n}}$ оуфире \overline{M} пес 2 λ 1. (48) не 2 виуе \overline{N} \overline{N} λ 1 к λ 10 с нажи о \overline{N} не 900 у. (49) TETNACOOYN NTETNP-wither. (50) NTOU AS N OYNOYTS AN ng \overline{N} te ngt mooyt, all ngt on \overline{z} . (51) oynog \overline{N} gihpe te tal.

Lesson 27

27.1 Negative adjective compounds. The prefix AT- is used to form negative adjectives from verbs and nouns:

λτοογπignorantλτουτπdisobedientλτηογimmortalλττλκοimperishableλθΗΤsenseless,λτ6οΜpowerless, impotentfoolishλτογωνεκολ invisible.

This prefix was originally a negative relative pronoun; a trace of this older usage is found in the resumptive pronoun required in some expressions, e.g.

עראבע פּרְס" unseeable, unseen אדשַאבּבּ פּרְס" ineffable; without פּרְס": speechless אדגנוּא פּרְס" immovable.

The resumptive pronoun agrees with the modified noun:

оүмүстнріон \overline{N} атфаже єроч an ineffable mystery оүбон \overline{N} аткім єрос an immovable power.

Nearly all $_{\lambda T^{-}}$ adjectives freely compound with \overline{p} - (Q o \overline{N}), as in \overline{p} - $_{\lambda T}$ cooy \overline{N} to become/be ignorant, \overline{p} - $_{\lambda T}$ oy $_{\omega N}$ \overline{z} 680 $_{\lambda}$ to become/be invisible.

- 27.2 Compound nouns. The distinction between a compound noun and a noun + \overline{N} + noun phrase is somewhat arbitrary. As a working definition we shall assume (1) that the first noun of a true compound noun must be in a reduced form different from the free (unbound) form, if indeed the latter exists; (2) that the linking \overline{N} be absent or at least optional. The most productive compounding prefixes are \overline{N} \overline{N} , \overline{N} ,
- (a) peq- forms agent or actor nouns; the second element is normally a simple or compound infinitive, but occasionally a qualitative:

PERMOOYT dead person reaxiove thief речтако destroyer; perishable

These may be used nominally or adjectivally, e.g.

OYCZIME N PESP-NOBE a sinful woman оуписумы й речтыко a destructive spirit TEICAPA N PETTAKO this perishable flesh.

and may be formed freely from virtually any appropriate verb in the language.

(b) pm-, pmn-, a reduced form of pwne พ, man of:

рмикние an Egyptian PMN2HT a wise, discerning person PMNNAZAPEO a person from Nazareth рмтюм a person from where? as in мтетм-гемримтюм? Where are you from?

(c) MNT- is used to form feminine abstract nouns from adjectives or other nouns. Compounds in MNT- are extremely numerous; the following is a typical sampling:

мМтоунны priesthood less frequent. MNTCABE Wisdom митврре youth; newness

митрии wisdom, prudence митеро kingdom, kingship; мителло old age (of a man) the spelling MNTFPO is MNT2XXw old age (of woman) MNTNO6 greatness: seniority MNTHONAXOC monkhood митаттако imperishability; incorruptibility.

MNT- is also used to designate languages:

мйтрийкние Egyptian митоуестении Greek MNT268PAIOC Hebrew MNT 2 POMA LOC Latin

(d) 61N- is used to form a feminine noun of action or gerund from any infinitive. The meaning ranges from concrete to abstract, e.g. GINNAY sight, vision; GINOYOM food (pl. 61NOYOOM). These are so predictable in meaning that they have been systematically excluded from the Glossary

unless they have acquired meanings not immediately obvious from that of the base verb.

Less frequent compounding prefixes are $\lambda N-$, eien-(eione), eiez- (eiwze), p λ - c+- (ctoi), goy- (g λ Y), geF-(gehp), gN- (ghpe), gCN- (c λ g), and 2λ M-. The reader may check these out in the Glossary.

Nominalized relative clauses are sometimes taken as compound nouns, occurring with an extra article, e.g.
(n) net govert vanity, (n) neeooy evil, oyner oyaas a saint.

A similar usage is found with $680 \times 2\overline{N}$, designating origin or affiliation (the def. art. appears as Ne^- , Ne^-):

OYGBOX 2N TCYPIX NG. He is a Syrian.

NGGBOX 2M NHI N AXYGIA NG. They are the ones from the house of David.

27.3 There is a form of the verb known as the participium conjunctivum (proclitic participle) used only for forming compounds with a following nominal element:

cω p. c. cλy-μpπ wine-drinking, a wine-drinker
ογωм ογλμ-ρωμε man-eating

xice χλc1-2μτ arrogant

moone μλν-εcooγ shepherd, tender of sheep.

It is uniformly vocalized with -x-. For most verbs the p. c. is rare or non-existent; a few verbs like the above account for most of the examples encountered. Note especially the compounds of Me: MAI- (one who loves):

MAI-GOOY desirous of fame or glory
MAI-NOYB, MAI-2AT desirous of wealth
MAI-NOYTE pious, God-loving
MAI-PWME kind, philanthropic
MAI-OYWM gluttonous.

27.4 The Third Future and its negative:

616C@TH	енесштМ	neg.	NNACWTM	И иенсфт <u>М</u>
екес фт й	өтетиесютМ		Миексот	П иетПСФТН
өр 6 С ФТ Й			Ñиесфт Й	
ечесютѬ	бүвс фт М		Мивчсфт М	Пиєусют м
есесютН			Пиессот М	

ере-проме сотт Пис-проме сотт

The negative forms are also spelled as ϵ nna-, ϵ nnek- etc. The 1st pers. sing. also occurs as \overline{N} nelc ω t \overline{M} .

The Third Future is an emphatic or vivid future with a wide variety of nuances; in an independent clause it describes a future event as necessary, inevitable, or obligatory. The English translation will depend on the context: 646cotm he shall hear, he is to hear, he is bound to hear, he must inevitably hear, he will surely hear, and similarly for the negative. The 2nd person is often used in commands and prohibitions:

NNEKHEIPAZE & RXOEIC HEKNOYTE.
You shall not tempt the Lord your God.

etetne22pe2 e nelentoxH.
You shall keep these commandments.

One of the most frequent uses of the Third Future is to express purpose or result after the conjunctions xe and xekx(x)c:

l have written to you so that you may know what has befallen me here.

 \overline{n} их \overline{n} иооү \overline{q} бр \overline{u} т \overline{n} х ε вч \overline{u} х ε и \overline{m} н \overline{n} т \overline{n}

We shall send him to you so that he may speak with you.

The same type of clause may be used as an object clause instead of the Inflected Infinitive after verbs of commanding, exhorting, and the like:

ANCHOUNT X6KAC WN64XOOC 6 AAAY.
We entreated him not to tell it to anyone.

It may occasionally replace the Inflected Infinitive in other situations:

The Third Future is tripartite; only the infinitive may be used in the verbal slot. The Second Future is sometimes used instead of the Third Future after xekac and xe.

Vocabulary 27

[The adjectival and nominal compounds given in 27.1, 2.]

- xloye vb. tr. to steal (\overline{M} mo'; from: $2\overline{N}$, eBOA $2\overline{N}$); as n.m. theft. \overline{N} xloye adv. stealthily, secretly.
- п.ммтре witness, testimony. т.ммтммтре testimony. ¬¬ммтре to testify, bear witness (to, about: ммо°, етве,
 ехм, е, 2ҳ, мм).
- T. 20 Te fear. AT20 Te fearless. F-20 Te (Q o N) to become/be afraid (of: e, exN, eтве, 2HT'). peqF-20 Te fearing, respectful. мПтречF-20 Te fear, respect.
- 2ων ετοοτ' to command, order someone (to do: ε, ετρε, χεκλο).
 τ. ρλοογ dream.
- TALGO TALGE- TALGO' Q TALGHY Vb. tr. to heal, cure ($\overline{\text{H}}\text{MO}'$; of, from: $2\overline{\text{N}}$, 6BOA $2\overline{\text{N}}$).

n.caein physician.

n.coma (τὸ σῶμα) body; the indef. art. is often deleted with this word in prep. phrases.

P-oyogin to shine, make light.

 \overline{P} -Kake (Q o \overline{N}) to become/be dark.

Exercises

м песнау е пні жекас бус+-тоотч м пеубіют й 2хло. (7) нерейфире и полине о и этсфін. (8) еничиолже евоу и петсфия и PEGTAKO TNNAY? (9) A-HEYXOGIC 2WN GTOOTOY N NGG2 \overline{M} 2AA XGKAC eyeeine $\overline{\mathbf{m}}$ $\overline{\mathbf{m}}$ $\overline{\mathbf{m}}$ $\overline{\mathbf{m}}$ $\overline{\mathbf{n}}$ $\overline{\mathbf{n}}$ $\overline{\mathbf{n}}$ $\overline{\mathbf{n}}$ $\overline{\mathbf{n}}$ $\overline{\mathbf{n}}$ HEKCHHY. (11) AYEL N ZIOYE N TEYGH AYW AYXL M HEYCOMA EBOA $2\overline{M}$ ntadoc. (12) to \overline{N} at 60M \overline{M} nemto ebox \overline{N} oypome \overline{N} tei-(13) OYMAI-OYWM RE REKCON. (14) NIM RET NAP-MNTPE ϵ thictic \overline{M} M6? (15) neimatol 2 enabote Ne. (16) a-haffe-AOC 61 NAI 2N OYPACOY N TEYOH AYW ATTAMOL ETBE NELWAXE. (17) \overline{N} TEYNOY A-THE \overline{P} -KAKE. (18) NIM HENTAGTAGOK GBOA $2\overline{M}$ пекфоне? (19) петроме оуслети \overline{N} саве пе. (20) \overline{M} прр-20те, πλώμρε. (21) λύω \overline{N} τεύνου λά \overline{P} -λτούων $\overline{2}$ εβολ \overline{N} 61 πλιλβολος. (22) OYNO6 TE TEHM \overline{N} TEPO. (23) $2\overline{N}$ TEHM \overline{N} T $2\overline{\lambda}$ O NEGELPE \overline{M} meeye ам \overline{N} мегооу \overline{N} течм \overline{N} тфире фим. (24) Мпенпістеуе е TEYMNTHNTTPG. (25) AYP-WIHPE AYW AYNOG \overline{N} 20TE WWIIG \overline{N} TEY-MHTE. (26) $\lambda 42 \omega N$ STOOTOY STREYMOYP M HOHPS NCENOXY S HEgтеко. (27) \overline{N} т \overline{K} -оүмаі-еооу ечgоуеїт. (28) оур \overline{M} \overline{N} тg \overline{N} ток? $\lambda N\overline{\Gamma}$ -оүр $\overline{M}\overline{N}$ кние. (29) \overline{M} пвс \overline{G} бом \overline{G} тоунос \overline{G} . (30) $26N2\overline{M}2\lambda\lambda$ \overline{N} peq \overline{P} -2076 Ne. (31) M \overline{N} - ϕ 60M \overline{M} MOI e $\phi\lambda$ xe N \overline{M} MHT \overline{N} $\vec{\mathbf{M}}$ MNTOYEEIENIN. (32) AXIC NAM XEKAC EMEXOOY $\vec{\mathbf{M}}$ noeik e $\vec{\mathbf{N}}$ -2HKG \overline{N} THOOLIC. (33) +NAGW NMMAK XGKAC \overline{N} NGYMOOYT. (34) OYATTAKO NE NNOMOC \overline{M} RXOGIC. (35) \overline{N} TA-MAPIA TAMAAY XNOI 2 N OYMYCTHPION N ATGAXE EPO4, 6-MN-XAAY N POME 2M HKOCHOC THPT NACIME GPO4. (36) AYMOY2 AC THPOY N GONT 2N TCYNACO-TH GYCWTM G NAI. (37) AXIC M REIWNE XE GGEF-OEIK. АУСІНЕ Й ОУНОЕ Й СЛЕІН СТРЕЧТАЛЕО Й ПОНГЕ, АЛЛА ЙПСЧЕЙ-60М 6 TAX604.

Lesson 28

28.1 The Habitual and its negative.

φλ≀сωтΉ	ФУИСФ1Щ	Neg.	МЕІСФТĦ	MENCOTH
φλκсωт₩	дътет Йсфт₩		Мексют й	метет пс от п
фур (е) сфты			мересфт	
фτωορά	σλ Υ сσ τΗ		МТФЭРЭМ	меусфтМ
φλССФТѬ			мессфт	
фуре-проме	сштН		мере-прюме	сютМ

The Habitual (or praesens consuetudinis) describes an action or activity as characteristic or habitual. It may usually be translated by the English general present (I write, I work, etc.):

The Habitual forms a regular system with the converters:

relative: { ega4cwtm Neg. ete мечсwtm ete ga4cwtm circumstantial: e-ga4cwtm 6-мечсwtm imperfect: ме-ga4cwtm ме-мечсwtm second tense: ega4cwtm

The Habitual is basically tenseless (hence the designation aorist in some grammars) and gains its translation value from the context. The imperfect converter makes a past tense explicit, e.g. NG-ga4C2A1 he used to write. Note that subject resumption is required in the relative form:

NPWMG GGA4P-NA1 the man who does thus. The Habitual belongs to the Tripartite Conjugation: only the Infinitive may be used in the verbal slot.

28.2 Emphasis. The typical non-emphatic word order

in a verbal clause is

(verbal prefix) + subject + verb + object + adverbial elements We have seen that the conversion of the verbal prefix to a second tense form places a strong emphasis on the adverbial element, requiring in most cases a cleft sentence in the English translation. The use of the Coptic cleft sentence pattern, with me, Te, Ne + a relative form is a further device for giving special prominence to a subject or object. A somewhat weaker emphasis is achieved by placing a specific element of the clause at the beginning. Such preposed elements are usually resumed pronominally within the clause unless they are simple adverbial phrases. This transformation, known also as fronting or topicalization, is very common in Coptic; examples abound on every page. The element preposed may be completely unmarked as such, but the Greek particle As is ubiquitous in this function. Fronted personal pronouns are always in the independent form. E.g.

анок де йпечение ймог.Me he didn't find.печенре де дурдетч.His son, however, they killed.йток де й†на† нак ан й пеат.I will not give the moneyto you.

The independent pronouns may be used appositionally to emphasize any suffixed pronoun, e.g. $2\overline{M}$ ntplowTM A6 lnok but when I heard; 6TBHHT \overline{K} \overline{N} TOK for your sake. We have already mentioned the repetition in \overline{N} T \overline{K} -NIM \overline{N} TOK? Who are you? They may even stand before a relative clause, as in \overline{N} NOK \overline{C} \overline{M} MOY the place which I am in.

The particles 61c and 61c 2HHT6 add a certain vividness or immediacy to a following statement. If an element is topicalized, 61c generally occurs before nouns and 61c 2HHT6 before pronouns.

Behold, I am the maidservant of the Lord.

eic гните екефопе екко N рок.

Behold, you shall remain (being) mute. (Cf. §30.11)

eic гните темью мтежно N оуфире.

Behold you shall conceive and bear a son.

The translation "behold" is purely conventional, but it is difficult to find a better English equivalent. The forms eic 2HHRE, eicte, eicne, and eic 2HHTE eic also occur. eic has several other functions: (1) with a following noun, as a complete predication:

GIC TEKCONG. Here is your sister.

- (2) as a "preposition" before temporal expressions, as in eld gomte \overline{N} pomme \overline{M} mennay epo4. We have not seen him for three years.
 - 28.3 Emphatic and intensive pronouns.
- (a) אגץגא, אבץגד, less frequently ογλλ(τ), is used in apposition to a preceding noun or pronoun: alone, sole, self, only. E.g.

I alone, I by myself, only I to him alone, to him only the king himself, the king alone.

(b) 2000 (1 c.s. 200 or 2000τ; 2 f.s. 2000τε, 2 c.pl. 200τ-τηγτη), similar to the preceding, but often with the added nuance of "also, too, moreover." E.g.

NTOK AG 200K, пафире, сенамочте ерок же пепрофитис м пет жосе. And you, moreover, my son, will be called the prophet of the Most High.

GIC GAICABET TOYCYFFENHC NTOC 2000C ON ACO N OYOHPE 2N TECMNT2 $\overline{\lambda}\lambda\omega$. Behold, Elisabeth your kinsman has also conceived a child in her old age.

The form 2004 also serves as an adverb/conjunction "however, on the other hand" without any pronominal force. $\overline{N}TO4$ is used likewise.

(c) MMIN MMo, an intensive pronoun, used in apposition to a preceding pronoun, usually possessive or reflexive:

my own house

2 m neu+me mmin mmou in his own village.

28.4 The reciprocal pronoun "each other, one another" is expressed by possessive prefixes on -ερμγ (fellow, companion), e.g.

אאאושַה אא אפּאפּףאץ. We fought with one another. אפּץשּאַגּה אא אפּץפּףאץ. They were talking with each other.

28.5 Further remarks on - κ e-. In addition to the use of - κ e- as an adjective "other, another" introduced in 4.3, - κ e- may have a purely emphasizing function, e.g.

TREPOME the man too, the man as well.

Both uses are frequent, and the correct translation will depend on a careful examination of the context.

There is a related set of pronouns: m.s. 66 or ket, f.s. kete, c.pl. kooye. These occur alone mostly in negative expressions, e.g. Mninay 6 66 I saw no one else. Otherwise the articles are added, as in there the other one (f.), Nkooye the others, 26Nkooye some others. For the indefinite singular keoya and f. keoyei, another (one), are used.

- 28.6 Nouns with pronominal suffixes. It was noted earlier that there is a small group of nouns which take pronominal suffixes in a possessive sense. Among the more important of these are
- (a) xw' head, mostly replaced by ane in normal usage, occurs frequently in compound expressions. The prepositions $ex\overline{n}$, exw' and exp(x) and exp(x) have already been introduced. Note also exp(x) before, in front of; exp(x) to raise one's head; exp(x) to submit (reflex.), to compel (not reflex.); exp(x) exp(x) e to submit to; exp(x) to bow the head. There are other similar verbal compounds.

- (b) 61A, 61AT' eye; mainly in compounds, e.g. KT6-61AT' to look around; M62-61AT' MHO' to stare at; TOYN-61AT' 6BOA to instruct, inform; cf. also NAIAT' in the following lesson.
- (c) pw mouth. The unbound form n.po appears often in the sense of "door, entrance," but in the sense of "mouth" it is usually replaced by τληρο except in compounds, e.g. the prepositions ερΝ, ερω and ειρΝ, ειρω; κλ-ρω, κω Ν ρω to become/remain silent (Q κλρλειτ); τΝ-ρω idem (as imptv.); χι-ρω ΜΝο to obstruct, block.
- (d) τοοτ' hand, already commented upon in \$10.4. The more important verbal compounds include †-τοοτ' (Vocab. 26), κλ-τοοτ' εδολ to cease (doing: Circum.), and 21-τοοτ' to begin (see Vocab. below).
- 28.7 The nouns underlying the directional adverbs of Lesson 8 are used in several other important adverbial and prepositional expressions. With N, 21, and ca they form adverbs of static location: e.g. N BOA outside, 21 20γN inside, ca-necht underneath, below. Each of these may be converted into a prepositional phrase by adding N, NHMO': 21 BOA N outside of, beyond; ca-20γN N within, inside of. Nearly all the possible combinations occur: (N, 21, ca) + (BOA, 20γN, 2γA1 up, 2γA1 down, necht, The, nA20γ, ngω1) ± NHMO' (sometimes also + 6). Their meanings are usually obvious from the context. The noun n.ca in these expressions means "side, direction." It is the same ca we have in NGA and MNNGA. Note also the phrase (N) ca ca NIM on every side, everywhich way.

Vocabulary 28

(eic, eic 2hhte, mayaa", 2000", \overline{m} hin \overline{m} ho", ka-po", t \overline{m} -po", 2enkooye, \overline{n} kooye, -ephy from the lesson)

cwoyz ceyz- cooyz Q cooyz vb. tr. (\pm e2oyn) to gather, collect (\overline{M} Mo"; at: e, $6x\overline{N}$, $2\overline{N}$); intr. idem.

canno canno canno Q canage vb. tr. to nourish, rear, tend to (mmo'); Q to be well-fed.

ח. אף ואפ fault, blame. $6 \overline{N}$ אף ואפ $6 \overline{N}$ to find fault with, blame. פּעפרוחק. $6 \overline{N}$ פעפרוחק. פעפרוחק. פעפרוחק. פעפרוחק. פעפרוחק. פעפרוחק.

2τοογε dawn, morning. ε/N/21 2τοογε at dawn.

pacts tomorrow. npacts, \overline{N} pacts, \overline{S} pacts, \overline{M} neapacts adv. tomorrow.

21-τοοτ' to begin, undertake (to do: 6 + Inf.); for 21see Glossary sub 210γ6.

on adv. again, further, moreover.

Exercises

(1) NIM HE HEIPMNOYTE EQAYMOYTE EPO4 XE IW2ANNHC? (2) $\overline{\text{N}}$ TO4 де гооч илет фарои и печрасте. (3) имге ероч ечмоофе мауа-At e sensets. (4) λ -2061NS micteys spot, 26NKOOYS AS MMOYпістеуе. (5) мере-паікатос сфтм в праже п пречр-нове. (6) NG-WAPG-MMONAXOC +-NGY2WB N 61X GBOA 2N NG2OOY GT MMAY. (7) нере-очное м минфе сооча зарм печил. (8) мпрем-драке epol, naelwt. $\overline{\mathbf{m}}$ ni $\overline{\mathbf{p}}$ -xaay. (9) $\overline{\mathbf{n}}$ poyze a-ncon kto4 on 6 te4-(10) мм-хаху м профитис фип 2м печтие мили ммоч. (11) ANOK A6 20 THA21-TOOT 6 C2A1 N NOAX6 ENTAYOUNG. \mathfrak{g} даська $\overline{\mathfrak{q}}$ и неч \mathfrak{q} нре $\overline{\mathfrak{q}}$ өе $\overline{\mathfrak{q}}$ оуег \mathfrak{q} $\overline{\mathfrak{q}}$ харос. (13) дахоос NAI x_6 tm-pok nfbok $_{6BOA}$. (14) akeipe \overline{n} nai \overline{n} tok mayaak? (15) NIM NET NACANOYON $e^{-\lambda}$ -NENGIOTE MOY? (16) $\lambda 4 k \lambda$ -poq, MUCAOAEQE-YYY. (17) 102 ANNHC A6 2004 AMOYN 680A 640YH2 гі пхлете. (18) йтере-роуге де фоле, нечилентис лусшоуг 2H HMA GT HMAY. (19) \overline{N} To Yww an etpekew \overline{M} neima. Kot \overline{K} 6 пекні мін мін мін к (20) $\lambda \gamma z$ і тоотоу є кот \overline{N} оуноє \overline{N} \overline{p} пє єре-TEYANG NAMOS 6 THE MAYAAC. (21) λ -gom $\overline{N}\tau$ \overline{M} MOOY 60 \overline{M} MAI, NKOOYE AE AYKTOOY E THOLIC. (22) NKOOYE AE CH2 2N KEXOOME. (23) Прече-нове де меуслуф-неуфире гй нентоли й пхоетс. (24) AYKOTOY ON 6 CTCONT. (25) 6TB6 OY TETTMID6 MT NETT-6PHY N TEL26? (26) 21 2TOOYS AS A-NPWMS N THOLIC COOYS ϵ taropa. (27) ac \overline{p} -20te ϵ 800 xe a-nec2al $\epsilon \overline{N}$ -apike ϵ 90c.

(28) NIM REGARTANGE-RCAGIN MAYALA? (29) ANSE E NENCHHY EYCANAGT THROY 6-MN-OYA GASKAGIT NSHTOY. (30) THNAGO ENWAHA WA POYSE.

Reading

(from the Sayings of the Fathers)

 λ -оух й ненетоте тйнооу й печилентис є мез-мооу. Меретдоте де пе оуну й трі ймате. λ чр-пово де є хі-пноуз иймач. Йтеречеї де ехй тооте, ачетие хе йпечетне иймач й пноуз. λ четре й оуфана, ачмоуте ечхо ймос хе, "порні, паетот пет хо ймос хе, 'моуз й паггтон й мооу.'" λ ую й теуноу λ -пмооу ет епоот, λ -псон моуз й печоофоу, λ ую λ -пмооу змоос он е печил.

New words: τ.φωτε, π.φηι well, cistern.

M62-M00γ to fetch water.

π.λΓΓΙΟΝ (τὸ άγγεῖον), π.φοφογ names of vessels.

Lesson 29

29.1 The Conditional and conditional clauses.

бідансютйif I hearвидансютйвкрансютйif you hearвтетйдансютйбредансютйetc.

бЧ**ФД**NСФТМ

6CQANCOTM

брфан-прфм6 сфтм

Negation is with -TM-: 649ANTHCOTH, 6P9ANTH-NPONG COTH.

9AN may be omitted in the negative: 64THCOTH, 6P6TH-NPONG
COTH. The Conditional occurs only in the protasis of conditional sentences. Only the Infinitive may occur in the verbal slot.

Conditional sentences in Coptic fall formally into two clearly defined groups: (1) real, and (2) contrary-tofact. The protasis of real conditional sentences in present time has a variety of forms:

(a) a clause with the Conditional:

вкумилистере в мых if you believe this

(b) egone (if) or egone (if) followed by the First Present, the Circumstantial, the Conditional, or any type of nonverbal predication:

egwne/egxe	KHICTEYE 6 NAI	
11	екпістеуе в илі	if you believe this
**	вкфунцістеле в илі	
**	Пток пе печетот	if you are his father
er	OYNTAK N2AT	if you have the money
**	и с с еп м +и	if I am not worthy

(c) the Circumstantial alone often serves as protasis:

enm neima,... since we are here....

The apodosis of such conditions may be any variety of verbal clause appropriate for the required sense (e.g. Fut. I, III, III; Habitual; Imperative). The apodosis may optionally be introduced with 616 (6616). For examples, see the exercises.

The protasis of contrary-to-fact conditions is in fact an Imperfect circumstantial clause, or, in the case of nonverbal clauses, a circumstantial of the clause with the imperfect converter:

6-и640 <u>й</u> Бро	if he were king
о <mark>ч</mark> ён эп воти-эи-э	if he were the king
е-ие-оүйтаи оүрро	if we had a king
6-иетет им пеіма	if you were here

In past time 6-N6- is followed by the affirmative Second Perfect or negative First Perfect:

6-N6-NTAK+-n2AT NAI if you had given me the money
6-N6-MN6KXI-n2AT if you had not taken the money

If the clause is nonverbal, 6-N6- alone is used. Thus, 6-N6KM N6KM means both "if you were here" and "if you had been here."

The conditional prefix e-ne- is not to be confused with the particle ene which serves to introduce a question, e.g. ene aknay epoq? Did you see him?

The apodosis of both tenses is in the imperfect of the Future:

е-мекпістече, мере-паі мафоле ам.

If you believed, this would not happen.

6-ме-мтакпістече, мере-паі маффпе ам.

If you had believed, this would not have happened.

The Greek conjunctions ϵ_{IMHTI} (ϵ_{L} $\mu \eta$ τ_{L}) and κ_{AN} ($\kappa \tilde{\alpha} \nu$) are also used to introduce protases of both real and contrary-to-fact conditions.

NCABHA X6 (except that, unless, if not) is often used to introduce the protasis of a contrary-to-fact condition; the clause usually contains a Pres. I, Perf. I, or non-verbal predication:

ПСАВНА ЖЕ КПІСТЕУЕ if you did not believe

" AKПІСТЕУЕ if you had not believed

" NTOK ПЕ ПАБІФТ if you were not my father

29.2 Inflected predicate adjectives. There is a small set of predicate adjectives inflected by means of pronominal suffixes or by proclisis to a nominal subject, e.g.

NECE-TEGC21ME. His wife is beautiful.

NECEC. She is beautiful.

The more important of these are NAA- NAA" great, NANOY-NANOY" good, Nece- Necw" beautiful, Neceww" wise, NAGE-NAGW" numerous, Neew" ugly. When used in relative clauses, they are treated like the First Present: npwne et NANOY4

the good man, npome etc nece-tegezime the man whose wife is beautiful. They may also be preceded by the imperfect and circumstantial converters: ne-nanoya (ne) he was good; oy-pome e-nanoya (ne) a good man. naiat (blessed is/are) belongs to this group, but a following nominal subject must be anticipated with a suffix: naiatoy N Npeap-eiphnh blessed are the peacemakers.

29.3 The comparison of both attributive and predicate adjectives is expressed by placing the preposition 6 before the item on which the comparison is based: NOG 6 NAI greater than this, CAB6 6 NG4CNHY wiser than his brothers. In addition to simple adjectives, both Coptic and Greek, the predicate adjectives of the preceding paragraph as well as appropriate qualitatives and other verbal constructions may be used in this construction. E.g.

Nego \overline{N} Noe 6 Negonhy. He was more important than his brothers.

чжосе в печжовіс. мечо \overline{N} оуовім в при. месюс в тессоме. He is more exalted than his master.

It was brighter than the sun.

She is more beautiful than her sister.

A comparison may be strengthened by using 2070 (more) in various combinations: \overline{N} 2070 e, 6 2070 e, 6 2076, all meaning "more than." \overline{N} 2070 alone may express an absolute comparative: \overline{N} 2070 the greater.

The Greek preposition napa (or \overline{M} napa) may be used instead of e. Suffixes may be attached: napoi, napok, napo etc.

- 29.4 Nouns with possessive suffixes (continued).
- (a) par' (foot) was mentioned in §19.2 in connection with epar' and arepar'. Other compounds include 2a par' prep. under, at the foot of; κ_{A} -par' to set foot (+ eboa: to start out); Mooge \overline{N} par' to go on foot.
- (b) 2PA* is the presuffixal form of two words: (1)
 20 2PA* face; (2) 2POOY 2PA* voice. Both of these words

are common in their unbound forms. Compounds worth noting are e_2pN $e_2p\lambda'$ prep. toward (the face of); (N) $N\lambda_2pN$ (N) $N\lambda_2p\lambda'$ prep. in the presence of; $x_1-2p\lambda'$ $(Q x_1-2p\lambda_0)$ to amuse oneself, be diverted, distracted (suff. is reflex.); $q_1-2p\lambda'$ to raise one's voice, utter (± 680 λ , 62 $p\lambda$ 1).

- (c) 2TH' is the presuffixal form of (1) 2HT heart, mind, and (2) 2HT tip, edge. Compounds using the form include †-2TH' to observe, pay attention to (ε, εχΝ); φΝ-2TH' to have pity (on: εχΝ, ε2γλι εχΝ); and the prep. 2λ2TN 2λ2-TH'.
- (d) 2HT' is the presuffixal form of (1) T.2H belly, womb, and (2) T.2H front. 2HT' (belly, womb) may be used in its plain sense, as in $2\overline{N}$ 2HTC in her womb; otherwise it appears only as part of the prep. $2\overline{N}$ \overline{N} 2HT'. 2HT' (front) is used as a preposition with certain verbs, e.g. gine 2HT', \overline{P} -20TE 2HT'.
- (e) $\tau \circ \gamma \omega'$ (bosom) is found in the prepositions $\epsilon \tau \circ \gamma \overline{N}$ - $\epsilon \tau \circ \gamma \omega'$ and $2 \cdot 1 \tau \circ \gamma \overline{N}$ $2 \cdot 1 \tau \circ \gamma \omega'$ near, beside. The latter is
 frequent in the relative construction $n \in \tau$ $2 \cdot 1 \tau \circ \gamma \omega'$ neighbor,
 e.g. $n \in \tau$ $2 \cdot 1 \tau \circ \gamma \omega'$ his neighbor.

Other nouns used with pronominal suffixes are aphx end, $\kappaoyN(\tau)$ bosom, $pin(\tau)$ name, $coyN\tau$ price, and max = max =

Vocabulary 29

(egone, egxe, \overline{N} CABHA xe, NANOY-, Nece-, NAGe-, NAIAT', \overline{N} 20Y0 e, 41-2PA', $g\overline{N}$ -2TH' ex \overline{N} , net 21TOYW' from the lesson)

- wck, Q ock vb. intr. to delay, tarry; to be prolonged, continue; + Circum.: to continue (doing).
- COO26 CA26- $CA2\omega(\omega)$ Q CA2HY vb. reflex. + 680A to withdraw, leave (from: $\overline{M}MO^2$).
- TAMIO TAMIG- TAMIO' Q TAMINY vb. tr. to create, make; to prepare, make ready (MMO'); as n.m. creation, creature.
- овыю овые- овыю Q овыну vb. tr. to humble, humiliate; intr. and reflex. to become humble; as n.m. humility

(often + N 2HT).

- ns.2MoT grace, gift, favor; gratitude. $g\overline{n}-2MOT \overline{N}T\overline{N}$ to give thanks to (for: $6x\overline{N}$, 21, 2A); $6\overline{N}-2MOT$ to find favor.
- gop \overline{n} (f. gopne) adj. first, before or after n. with \overline{n} . \overline{n} gop \overline{n} adv. formerly, at first.
- \overline{P} -20YO 6 (Q o \overline{N}) to exceed, be more than; to be in excess, more than enough for.
- 2N OYOPx adv. firmly, surely, certainly, diligently.

Exercises

- А. (1) бтв в пекоб в 10 \overline{N} 2 нт чиду \overline{N} -2 тнч бхок. (2) и и пентачтаніо \overline{M} прор \overline{n} \overline{N} роме? (3) и до е-и в 1 роме \overline{N} 20 у о в рои. (4) и в с е-те і поліс \overline{N} 20 у о. (5) и лілтоу \overline{N} \overline{N} 2 нкв. (6) \overline{N} пиоуте тамі в-тпе м \overline{N} пкаг. (7) и ім пе пет гітоую к? (8) таї те трорпе \overline{N} в и толь. (9) и диоу- $\frac{1}{2}$ -2 г \overline{M} метаної д. (10) сеидсмоу в рок \overline{N} 20 у о в роме и ім. (11) и в у \overline{N} тач оу с 2 іме в-и в сос в в мате. (12) д стаміо и дч \overline{N} оу коу і \overline{N} в і и оу мом. (13) и ділт \overline{N} \overline{M} пентач- \overline{N} в плагов іс. (14) оу пет и дроб в і ок? (15) оу пет \overline{N} дала х є в в в оу х ді? (16) с д 2 в т тут \overline{N} в в ох \overline{M} ної.
- B. (1) ϵ 4 ϕ λ 10 ϵ 7, 4 λ 20 γ 7 ϵ 7 ϵ 8. (2) ϵ 8 ϕ λ 10 ϵ 7, ϵ 16 NACNHY NAGONT MMATE. (3) GROANKAAT G BOK, TNAKTOI G GIHT. (4) 64 MAN 2 WN 6 TOOTK 6 TP 6 KAAC, 6 K 6 AAC $2 \overline{N}$ O Y WP \overline{X} . (5) 6 M W I G TOOTK 6 TP 6 KAAC, 6 K6 AAC $2 \overline{N}$ O Y WP \overline{X} . ММАТОІ 61 620YN 6 ТПОЛІС, СБИДРА2ТЙ ТНРЙ. (6) БООПЕ ЧСОТЙ 6 TEKCHH, 4NACA204. (7) EDONG OYNTHTN 2ENOEIK MMAY GYP-2040 EPOTN, TETHETALY \overline{N} NET 2KAEIT. (8) EPOAN-NECHHY KTOOY 6 ΠΤΗ Ε 21 ΡΟΥ26, ΤΗ ΑΒΟΚ ΝΉΜΑΥ. (9) ΕΦΙΕ ΠΕΚΕΙΟΤ ΕΠΙΤΗ ΑΝΑΚ, инеквоит. (10) воже пексои \overline{p} -певооу илк, еке \overline{p} -ппет илиоуч . PAH (11) ϵ PDAN-TEKCONE EL DAPOL \overline{N} PACTE, \dagger NATAMOC ϵ TB ϵ 11619 λ X6. (12) 6-N6-OYAIKAIOC N6 NTOK, N6KNAEIP6 N T6126 λ N. (13) $AP6\overline{N}-2MOT$ $\overline{N}NA2P\overline{N}$ $\overline{N}NOYTE$. (14) $E-NGK\overline{M}$ $\overline{N}EIMA, NEPE-NACON$ NAMOY AN RE. (15) $T\overline{N}Q\overline{\Pi}-2MOT$ $\overline{N}TOOT\overline{K}$ 2A REKNOE \overline{N} NA. 6-N6-NTAIGIME XE NTOK NE NPPO, NEINANA2T NA2PAK NE EIGINE 2HTK. (17) GOXE KOCK EKO N PETP-NOBE, NCENAON-2THY EXOKAN м пегооу ет ммау. (18) едопе сеегре м пет наноуч, сенаей-2MOT NNA2PM TEOGIC. (19) 6-N6-NTA-NGIZICE WCK, NENNAMOY

пе. (20) едопе супіствує $2\overline{N}$ оуюрх, сенастомоу. (21) дде ерон етренд \overline{n} -2мот йтоот \overline{q} \overline{N} оуосід нім. (22) 2ап \overline{c} ерон етрен \uparrow -тоотоу \overline{N} нет 2 ітоуюн. (23) смок \overline{z} 6 с2а і \overline{M} м \overline{N} три \overline{N} -киме. (24) єїс пеїноє \overline{M} маєїн набол \overline{n} євол ни \overline{n} . (25) \overline{N} Савна же \overline{N} ток пе паєїют, неїнамооут \overline{K} .

Reading

(from the Sayings of the Fathers)

- 1. A 4 X OOC NG 1 OY 2 TAO X6 " $2\overline{M}$ RIPACMOC NIM MRPGN-APIKE E-PWME, AAAA 6 N-APIKE EPOK MAYAAK EKXW MMOC XE 'EPE-NAI WOOR MMOI ETBE NANOBE.'"
- 2. λ -оуд \overline{N} $\overline{N}2\overline{N}$ ЛО вюк g_{λ} $K62\overline{N}$ ЛО дую пехдч \overline{M} печндентно хе, "таміо нан \overline{N} оукоуі \overline{N} дрфін." дую дчтаміоч. пехдч хе, "гер $\overline{\Pi}$ -геноєїк нан." дую дчгорпоу. \overline{N} Тооу де думоун євод буфдхе є не $\overline{\Pi}\overline{N}$ Ткон \overline{M} пегооу тнр \overline{M} теуфн тнр \overline{C} .
- 3. AYXOOC \overline{M} 61 \overline{M} 2 $\overline{\Lambda}$ 30 X6, "KAN NAME GPDAN-OYAFFGAOC OYWNA2 NAK GBOA, \overline{M} \overline{M}

New words: π.πιρωσμος (ὁ πειρασμός) temptation.
π.ωρφικ lentils.

2ωρπ 2ωρπ 2ωρπ 2ωρπ vb. tr. to moisten. $\overline{n}\overline{n}$ \overline{n} \overline{n}

Lesson 30

30.1 The Injunctive (also called the Optative):

марісюты let me hear марысюты let us hear маречсюты let him hear мареусюты let them hear марессюты let her hear

маре-проме соты let the man hear

The Injunctive occurs only in the 1st and 3rd persons in standard Sahidic. The 1st person corresponds to the cohortative, the 3rd person to the jussive; theoretically, the Imperative may be said to occupy the 2nd person position. The negative of the Injunctive is expressed by using the negative Imperative prefix Mnp- with the corresponding form of the Inflected Infinitive: Mnprequox don't let him go, Mnprequox don't let them kill him. The Injunctive is tripartite and is used only with the Infinitive. The free form of the 1st person, Mapon, is used alone in the sense "Let's go."

30.2 The Future Conjunctive of Result (also called the Finalis).

ТАРЕСФТЯ

тароусштй тороусштй

TAP6CCWTM

 \overline{N} may occur optionally before all of these forms. For the 1st person sing, the simple Conjunctive τ_{λ} - may be used.

The Future Conjunctive is basically a result clause; it is especially frequent after an Imperative, e.g.

сотн брог тарек - case. Listen to me and you will become wise (or: so as to become wise).

Although the Conjunctive itself may occasionally have the

value of a result/purpose clause after an Imperative, the Future Conjunctive always has this meaning. The nuance of the form can best be understood if it is viewed as the transformation of an underlying conditional sentence:

амоу тарекнау - вкранет вете кнанау.

It may also occur after a question, e.g.

ити пентачилу броч таречфахе броч?

Who has seen him so as to be able to describe him? If the question is rhetorical, as in this example, negation is generally implied: "No one has seen him so as.... If the question is real, the implication is "Tell me the answer so that...," as in

ечтом пексом тарборже иммач?

Where is your brother that we may speak with him?

30.3 The Clause Conjugations. A distinction is made between sentence conjugations (Bipartite and Tripartite) and clause conjugations. The latter are so named because they correspond to a conjunction plus a clause in normal translation. To this category belong the Temporal, the Conjunctive, the Conditional, the Future Conjunctive of Result, and most uses of the Inflected Infinitive (GTPGG-COTM, 2M NTPGGCOTM, MNNCA TPGGCOTM). Characteristic of this category is (1) negation with -TM-, and (2) the use of the Infinitive only.

A further clause conjugation is $gantqcwt\overline{m}$ (until he hears):

 gantcoth
 until I hear
 gantncoth

 gantcoth
 until you hear
 gantetncoth

 gantecoth
 etc.

 gantoycoth
 gantoycoth

φλητε-πρωμε cωτπ until the man hears

Translation is regularly with "until," e.g.

TNNAGO M neima gantagi. We shall remain here until he comes.

Similar in appearance to a clause conjugation is the form $xin(\overline{N}) \tau \lambda 4 cot\overline{M}$ (from the time that he heard). This consists, however, of the conjunction xin followed by the Second Perfect. Even more frequent are the compound expressions with $\kappa\lambda\tau\lambda$ see and \overline{N} see (as, according as, just as), both of which are followed by relative constructions, e.g.

They found it just as he had told them.

KATA OE ENTALAC NHT \overline{N} , ETETNEAAC 2 ϖ T-THYT \overline{N} ...

According as I have done to you, you too are to do ...

КАТА ӨЕ ЕТ CH2 ЕТВИНТЯ

as it is written concerning him

ката өе етоүнафсфти имос

according as they would be able to hear (i.e. understand)

The feminine resumptive -c in these constructions refers back to ee and should not be translated as a pronominal object. If a real pronominal object is required, the resumptive -c is omitted, e.g.

KATA OF NTA-NACIOT TNNOOYT, ANOK 20 \uparrow XOOY NMOUTN. Just as my Father sent me, so I too am sending you.

Other constructions with oe are treated similarly, e.g.

TAL TE OF NTA-MXOFIC ARC NAL.

Thus has the Lord acted for me.

30.4 When the Inflected Infinitive is used instead of a simple Infinitive after a verbal prefix, it has the value of a causative (hence its alternate name, the Causative Infinitive):

T caused them to enter.

†NATPEKPIME.

I shall cause you to weep.

30.5 The form MnxTqcotM describes an action as expected but not yet done. It is conveniently translated as "he has not yet heard." The form is fully inflected:

МПА+СФТЙ МПАТПСФТЙ МПАТ6-ПРФМ6 СФТЙ

МПАТКСФТЙ МПАТЕТПСФТМ

พีกมาecwrพี

МПАТЧСФТЙ МПАТОУСФТЙ

MNATECOTH

It may occur in circumstantial clauses with the circumstantial converter e-; the resultant form appears ambiguously as e-Nnate- or simply Mnate-. In this usage it is best translated as an affirmative clause with "before":

TNNATA204 6-MHATTHO2 6 THOAIC.

We shall overtake him before he reaches the city.

With the imperfect converter NG-MNATTCOTH corresponds to the pluperfect: he had not yet heard.

30.6 An untranslatable dative with NA* or epo* occurs optionally with many verbs, especially in the Imperative. This reflexive dative is called the ethical dative, following standard terminology. E.g.

BOK NAK & NEKHI. Go home!

CO NHTN. Drink!

Verbs with which this occurs with some frequency are noted in the Glossary.

30.7 Higher numbers, ordinals, and fractions.

- 20 χογωτ (f. χογωτε) χογτ- 70 gue, coque, gbe
- 30 MARE (f. MARE) MAR- 80 SMENE, SMINE-
- 40 2MG 90 η<u>ς</u>τλιογ
- 50 τλίογ 100 φε 200 φητ 60 ce 1000 φο 10,000 τελ

The tens combine with the forms of the units used in the 'teens (§24.3). The -r- of -rH (5) is not repeated after

another -T-:

 жоүтоүе
 21
 мавчіте
 39

 хоүтн
 25
 фчетн
 75

An intrusive -T- appears before -ATTE (4) and -ACE (6):

навтачте 34 сетасе 66

The numbers ge 100, go 1000, and TBA 10,000 are masculine:

 go CNAΥ
 2000
 MNTCNOOYC N TBA
 120,000

 gomNT N go
 3000

Proclitic forms of the units are frequent here, e.g.

φήτ-φο 3000 cey-φο 6000

Combinations of these higher numbers with tens and units vary in form, e.g.

96 малв = 96 м \overline{N} малв 130 сеу-90 хую 9моун \overline{N} 96 6800

Ordinal numbers are formed from the cardinals with the prefix M62-. The ordinals are treated as adjectives before the noun with linking \overline{N} . Gender distinctions are maintained:

THE2CNAY N 200Y the second day
the2CNT6 N POMB6 the second year

For "first" the adjectives gop (f. gopne) and 20Ye1T (f. 20Ye1Te) are used.

Fractional numbers worth noting are τ . Nage (half) and 60c, 61c- (half). Other fractions are expressed by perefixed to the denominator, as in pe-MHT one-tenth, or with oyon (oyN-), as in oyN-N-4TOOY a fourth.

30.8 The remote (or further) demonstrative pronouns (that) are m.s. nH, f.s. TH, and pl. NH. These occur much less frequently than nAI, TAI, NAI because of the preference for using phrases with 6T MMAY, such as net MMAY.

The prefixal forms ni-, +-, and Ni- are usually described as the reduced forms of nH, TH, and NH, parallel

in usage to nei-, tei-, and Nei-. While such a formal relationship may exist, the use of ni-, \uparrow -, and Ni- in standard Sahidic is quite restricted. The form ni- occurs mainly in a few temporal and local adverbial expressions, such as \overline{M} nioyoeig (at that time) and nich (that side, as opposed to this side). The form Ni- occurs most frequently in expressions involving comparison with \overline{N} of \overline{N} (like) or \overline{P} -of \overline{N} (to become like); it sometimes corresponds more closely to an English generic noun, e.g. \overline{N} of \overline{N} Niepoomne like doves, like a dove. It is also found in the expression \overline{N} Niene2. Elsewhere ni-, \uparrow -, and Ni- are frequent as scribal variants of nei-, tei-, Nei- or have the force of an emphatic article.

30.9 When it is necessary to express a durative or continuous process or state in the future, a periphrastic construction is employed using the Circumstantial. Contrast

אאסיסים you will become holy איז you will be holy

бкекх-рокyou shall become silentекефопе бкко N рокyou shall remain silent

The difference is sometimes slight, but not infrequently spelled out. The same construction occasionally appears with other tripartite conjugational forms. A full discussion of the aspectual problem involved here lies beyond the scope of this book.

30.10 Greek conjunctions, adverbs, and prepositions that occur frequently in Coptic (for reference only). The term postpositive means that the word in question must follow immediately after the first element of the sentence, as in nrome As Aybox.

λλλ άλλά but, rather.

λρλ ἄρα (introduces question).

rap γάρ for, because, since (postpositive).

141 A6 δέ but, however (postpositive). eimhti el un ti (1) if not, unless, except that (+ Conj.); (2) elliptically, e.g. Mnoyxey-2HAIAC WA AAAY MMOOY GINHTI 6 CAPENTA Elias was not sent to any of them except Sarepta. Note the independent pronoun in this usage: אויסאג א pome אבואר פרסץ פואאדו אואסא No one will understand it but me. eite ... eite eite ... eite either ... or. ener έπεί because, since. energh éneron because, since, when. eneralner έπειδήπερ inasmuch as, since. єті Ёті yet, still, while yet (+ Circum.). н ñ or. KAI FAP Hal Yap for truly. κλιτοι καίτοι although, albeit. หมท หลับ even if. ката ната (prep.) in accordance with, according to; also in distributive sense, e.g. kata cassaton every sabbath. Note the absence of the article here. men... Δε μέν... δέ balances two statements: on the one hand ... but on the other. Both postpositive. พห นท์ introduces a rhetorical question presuming a simple yes or no answer. ми поте μήποτε so that not, lest (+ Conj.). мнише μήπως so that not, lest (+ Conj.). мнт: цить like мн. but with strong element of surprise. moric μόγις hardly, scarcely. OYN OUN therefore (postpositive).

OYAG ΟὖΘΕ and not, nor; the negation is often repeated in Coptic as well.

OYTG ... OYTG ΟὖΤΕ ... ΟὖΤΕ neither ... nor.

nnoc made (nman) was 1 1 1.

προς πρός (prep.) used like κατα.

noc πως how? why?

τοτε then, thereupon, next.

20ΑλΝ ὅταν when, whenever, if (+ Cond.).

20CON, N20CON ὄσον as long as (+ Circum.).
20C ὡς (1) as if; (2) although; (3) when, while as (all + Circum.).
20CTE ὥστε so that (+ Conj. or Infl. Inf.).
xωρις χωρίς (prep.) without; a following noun has no indefinite article.

- 30.11 Final remarks on Coptic conjunctions and particles.
- (a) The main coordinating conjunctions are Ayw and MN. MN is used primarily to join nouns or nominalized expressions; Ayw is used elsewhere. Ayw is sometimes used for MN, but this poses no particular translation problem. Ayw often appears redundantly before the Conjunctive or before the apodosis of a conditional sentence. When nouns have no article (for whatever reason), they may be joined with the preposition 21 instead of MN, as in MN-MOOY 21 OGIK MMAY There is neither water nor food. 21 is also used to form compound nominal expressions of a special type, e.g. CAPX 21 CNO4 flesh and blood. These expressions function as a unit: any article occurs only with the first word, as in 26NCAPX 21 CNO4 NG They are flesh and blood.
- (b) The main uses of the conjunction xe have already been introduced: (1) in naming-constructions (see Vocab. 17); (2) to introduce noun clauses (object clauses) after appropriate verbs of speaking, perception, and the like; (3) to introduce purpose/result clauses with the Second or Third Future. xe is also frequent in the sense "for, since, because," which is less ambiguously expressed by GBON XE and GTBE XE. In many instances xe is the equivalent of English "namely, i.e." in introducing explanatory appositions, e.g. oyeycll...xe oycoelg N GPMngan an offering... namely a pair of turtle-doves. XE is also used in some compound conjunctions, such as NCABHA XE (if not, unless) and N GE XE (as if, as though).
 - (c) eguxe and ele, in addition to their role in

conditional sentences, may be placed before any statement to mark it as a question.

- (d) 66 is a postpositive particle with very much the same function as Greek A6. It is especially frequent in the phrase TENOY 66 and now, so now therefore.
 - (e) NTOOYN: then, thereupon, next, forthwith.
- (g) Certain temporal expressions may occur with a following relative clause without resumptive pronouns.

 These function virtually as compound conjunctions. E.g.

the day when this will happen on the day when he looked.

(h) The Conditional is frequently used in a temporal sense: when, whenever.

Vocabulary 30

- τελμλ vb. intr. to rejoice (over: $εx\overline{N}$); as n.m. joy.
- τλφο τλφο τλφο vb. tr. to increase (Μπο); often prefixed to another Inf.: to do something more, much. τλφε-οειφ to preach, proclaim (Μπο).
- TEBO TEBO TEBO Q TEBHY vb. tr. to purify, cleanse, heal (MMO'; of, from: 6, 680A 2N, 2A); as n.m. purity, purification.
- TAYO TAYO" (± 680%) vb. tr. to send forth, cast forth, proclaim, tell (MMO"). TAYO-KAPHOC to produce fruit.
- TAGIO ΤΑGΙΘ- ΤΑGΙΘ' Q ΤΑGΙΗΎ vb. tr. to honor, respect. value, esteem (ΜΗΘ'); Q to be honored etc., valuable.
- TAMPO TAMPO TAMPO Q TAMPHY vb. tr. to strengthen, confirm $(\overline{M}MO')$; intr. to become strengthened, firm, resolute.
- n.com time, occasion. \overline{N} oycom once. 21 oycom all at once, altogether. \overline{N} kecom again. com NIM always, on every occasion. \overline{N} 222 \overline{N} com many times, often. KATA com \overline{N}

(+ Inf.) on every occasion of.

m.200YT male (of animals or humans); freq. as adj.: male, wild, savage. c21M6 is used as the corresponding female.

ne.kaom crown, wreath. \uparrow -kaom ex \overline{n} to crown. x_1 -kaom to receive a crown, become a martyr.

eleibc shade, shadow. P-2leibc to shade, protect (e, exN).

n.ceene remainder, rest (often in plural sense). A redundant -ke appears frequently: nkeceene the rest.

F-xpix to need (Μmo'); to have to (do: ε + Inf.); xpix is Gk. ή χρεία

6M-ngine N, 6M-n(')gine to search out, visit.

2 PAI is often used to reinforce a following preposition, esp. $2 \overline{N}$, with no real difference in sense.

Exercises

(1) nexay xe mapon, teynoy 2hn e20yn. (2) agtpe-neg2M2AA Tamio nau $\overline{\mathbf{n}}$ oykoyi $\overline{\mathbf{n}}$ oeik. (3) autage-oeig $\overline{\mathbf{n}}$ neyaffealon $2\overline{N}$ TEXUPL THE GANTACASON EBOX N2HTC. (4) MINTER N2AXATE OYOM 6BOA $2\overline{N}$ Nelexooxe. (5) λ -Nal throy wore kata 96 ft CH2 2M TXWOME. (6) 2PAI 2N TME2MNTCHOOYC N POMME N TEUм \overline{N} теро дчмоу \overline{N} 61 пен \overline{P} ро. (7) дмнег \overline{N} фдрог тдрет \overline{N} 61 не \overline{M} (8) гапо брои бтренфахе иммач в-мпате-течтапро том $2\overline{M}$ TMOY. (9) MAP6-TXO61C $0\overline{N}$ -2TH4 EXOK NTTAX6OK. (10) 0AP6оуфии ϵ -илиоуч тауе-карпос ϵ -илиоуч. (11) \pm иафс $\overline{\kappa}$ \overline{m} п ϵ има ϕ алтчкточ. (12) етве па і тетнах і \overline{m} пеклом \overline{m} пеооу $_{2}\overline{n}$ \overline{m} пнуе. (13) \overline{N} gas an etertie-net 2 itoywk swnt. (14) nkecesne as λγλρx61 \overline{N} pime 21 ογcon. (15) λ -π λ $\overline{\Pi}$ Ν \overline{N} τελήλ εχ \overline{M} πηογτε пасютну. (16) а-пет фоне хоос нач же пхоетс, oyn-60м ммок E TBBOI. (17) \overline{N} NGT \overline{N} TAYE-NENTATET \overline{N} NAY E700Y E XAAY. \overline{N}_2 AAATG \overline{N} THE $\underline{\omega}$ AYOY $\underline{\omega}_2$ 2A $\underline{\Theta}$ A18 \overline{C} \overline{M} Highn GT \overline{M} MAY. (19) Hete оу \overline{N} тъч фтни с \overline{N} те мъреч+оуе \overline{N} пете м \overline{N} тъч (20) т \overline{N} сооу \overline{N} x_6 nx_{06} ic nating m nengthe m kecon m netroy et m_{MAY} . (21) Mneneipe kata of entagemn etooth ϵ arc. (22) autherпкесеепе гмоос жекас бубсшты в течсвю. (23) чил-клом $ex\overline{n}$ **net** $na\overline{p}$ - $m\overline{n}$ тре етве печран ет оуав. (24) теом \overline{m} пет

хосе тет $na\bar{p}$ -2ae18 \bar{c} еро. (25) сф \bar{m} е nagaxe таректахро $2\bar{n}$ тпістіс $2\bar{n}$ оуфр \bar{x} . (26) а-пхоеіс тафе-печла м \bar{m} мас. (27) оу200ут $m\bar{n}$ оус21ме ачтаміооу \bar{n} 61 пхоеіс. (28) \bar{n} тачеі е т \bar{e} вооу евоа $2\bar{n}$ меунове. (29) мар \bar{n} рафе \bar{n} т \bar{n} теана \bar{n} т \bar{n} +-еооу na4. (30) \bar{n} т \bar{n} -хріа ам \bar{n} мекфахе ет таеіну. (31) ме-оу \bar{n} -тач 26n2 \bar{m} 2a3 \bar{n} 200ут $m\bar{n}$ 26n2 \bar{m} 2a3 \bar{n} с21ме. (32) оу \bar{n} -60м ае \bar{n} пноуте е тафе-2нот \bar{n} 1a4. (33) паі пе прам емта-паггеаос таач \bar{n} 4 се \bar{n} 6 —мпате-течмаху ф \bar{m} 6 \bar{n} 6. (34) пеі 2061те таеіну \bar{n} 6 2070 е \bar{n} 6. (35) ас \bar{p} -хнра фант \bar{c} 7 2меметачте \bar{n} 6 ромпе. (36) фаухоос же емехі-клом $2\bar{n}$ 6 \bar{n} 7 \bar{n} 8 и меі мо \bar{n} 9 21се.

The Lord's Prayer

пеновиют ет $2\overline{N}$ мпнуе, маре-пекран оуоп. текм \overline{N} Τρρο маресет. Пекоуюю маречюю \overline{N} өе ет $\overline{q}_2\overline{N}$ тпе \overline{N} 490 \overline{m} 6 ον 212 \overline{M} πκλ2. Пеноеік ет мну 2 и \overline{r} 4 ммоч най \overline{m} пооу, и \overline{r} κω най евол \overline{N} иет ерон \overline{N} \overline{n} θε 2ωων ον ετ \overline{n} κω εвоλ \overline{N} иете оу \overline{n} ταν εροοу, и \overline{r} τ \overline{m} 7 χιτ \overline{N} 620 \overline{N} 9 \overline{n} 6 πείρας αλλά ν \overline{n} 0 \overline{n} 1 εвоλ 21 τοοτ \overline{q} \overline{m} 1 πιονή \overline{n} 1 που \overline{n} 2 τωκ τε τεом \overline{n} 1 πεοού \overline{n} 2 νι ενες. \overline{n} 3 που \overline{n} 4 ενος \overline{n} 5 τως \overline{n} 6 τον \overline{n} 6 που \overline{n} 6 σου \overline{n} 6 που \overline{n} 7 που \overline{n} 7 που \overline{n} 7 που \overline{n} 8 τως \overline{n} 8 τως \overline{n} 9 που \overline{n} 9 που \overline{n} 9 νι ενες \overline{n} 9 συν \overline{n} 9 που \overline{n} 9 που \overline{n} 9 που \overline{n} 9 συν \overline{n} 9 που \overline{n} 9 συν \overline{n} 9

- 1. The repetition of the verb is apparently an attempt to clarify what was felt as an awkward construction in the Greek.
- 2. et nhy renders Gk. έπιούσιον "for the coming (day)."
 Note that the 2nd pers. Conjunctives continue, with
 the force of Imperatives, the 3rd pers. Injunctive
 forms at the beginning.
- 3. The prep. e has the special sense of "due from (as indebtedness)." Thus, NET EPON "those things which are due from us," NETE OYNTAN EPOOY "those from whom we have (something) due."
- 4. NOY2N vb. tr. to rescue, save.

Reading Selections Introductory Remarks

A. Luke I - V

The text given here is based on that of G. Horner, The Coptic Version of the New Testament in the Southern Dialect, otherwise called Sahidic or Thebaic (Oxford, 1911-24), Vol. II, pp. 3-95. The only orthographic changes made are in the division of the words in order to bring the text into conformity with the style of the present work. The Coptic version should be studied in conjunction with the original Greek; only in this way can the reader gain a clear understanding of the translation techniques employed and of the influence the original has had on the grammar, vocabulary, and style of the Coptic translation. The opening verses are rather difficult, but the remainder of the text is fairly simple and straightforward.

B. Apophthegmata Patrum

The Sahidic version of the Apophtheamata Patrum, or Sayings of the Fathers, survives in a single manuscript, parts of which are preserved in five different European libraries. The largest fragment, some forty-four leaves, now in the Biblioteca Nazionale of Naples. was published by G. Zoega in his Catalogus codicum copticorum manu scriptorum qui in Museo Borgiano Velitris adservantur (Rome, 1810). Sayings from this particular set of pages are often denoted by the siglum Z. These and the smaller fragments of Paris, Vienna, Venice, and London have all been assembled and edited by M. Chaîne, Le manuscrit de la version copte en dialecte sahidique des "Apophthegmata Patrum" (Cairo, 1960). The enumeration and text of this edition, which is unfortunately not without printing errors, have been followed for the selections given here. Chaîne supplies a French translation of the text and a valuable concordance of each "saying" with extant Greek and Latin versions, which the interested reader may wish to consult.

The contents of the Sayings are quite varied, including anecdotes about individual desert Fathers, the miracles they unwittingly performed because of their excessive virtue, their pithy statements on the perfections and imperfections of fellow-monks and the monastic way of life, and even quite serious digressions on important theological issues of the day. The collection is probably no more "historically authentic" than any similar collection of traditional material, but it does, as a whole, shed light on the early days of Christian monasticism and on the personalities of the dedicated men and women of the Egyptian desert communities.

Apart from a revision of word division, very few changes have been made in the text: (1) $2\overline{N}$ and $2\varepsilon N$ have been adjusted throughout; (2) gone for the unusual goone, passim; (3) NGK- for NGK- on the first econthylon of No. 5; (4) $\Delta Y_2 \varepsilon$ for $\Delta N_2 \varepsilon$ in No. 17; (5) ε entoah for NTOAH in No. 24; (6) \overline{N} NCYNKAHTIKOC for \overline{N} enc- in No. 26; (7) ε oyoyon for oyon in No. 31; (8) ε and ε for ε for ε and ε in No. 38; (9) two lines transposed in No. 38 (a printing error in Chaîne); (10) restore [\overline{M} MO4] for Chaîne's [ε soal in No. 70; (11) ε for ε in No. 175; (13) ε oyoxyna for oyoxyna in No. 175; (14) ε ε \overline{N} $\overline{N$

C. Wisdom of Solomon

The text given here is based on P. de Lagarde, Aegyptiaca (Göttingen, 1883), pp. 65-82. Sapienta Solomonis, or The Wisdom of Solomon, well preserved in Greek, Latin, Syriac, Coptic, and Armenian versions, is an intertestamental work in the tradition of Hebrew wisdom literature (Proverbs, Ecclesiastes, Ben Sirach), but by a writer well acquainted with the major schools of Greek philosophy. The date and provenance of the work are both disputed, and the interested reader may consult the discussion in R. H. Charles, The Apocrypha and Pseudepigrapha of the Old Testament (Oxford, 1913), Vol. I, pp. 518-68, where an annotated translation and an extensive bibliography may also be found. The short essay of Moses Hadas in The Interpreter's

Dictionary of the Bible (Abingdon Press, Nashville, 1962), sub Wisdom of Solomon, may also be read with profit. No changes have been made in the text other than in the division of the words. The minor restorations of Lagarde have been accepted without comment.

D. The Life of Joseph the Carpenter

As an apocryphal work dealing with the life, but mainly the death, of Joseph, the father "according to the flesh" of Jesus. The Life of Joseph the Carpenter is one of that large number of spurious gospels, acts, epistles, etc. that sprang from the imaginative pens of Christian writers attempting to fill in biographical details missing from the canonical New Testament. Although useless in a quest for "the historical Jesus," each of these works has its own intrinsic interest, reflecting as it does the peculiar doctrinal, nationalistic, sectarian, or other preoccupations of its writer and his circle. The Life of Joseph is fully preserved in a Bohairic Coptic version and a brief Arabic paraphrase, both of which were published by P. de Lagarde, Acauptiaca (Göttingen, 1883), together with the Sahidic version of Chapters 14-21.1. Two further fragments (Chapters 5-8.1; 13) of the Sahidic version were published by F. Robinson, Coptic Apocruphal Gospels (Texts and Studies IV, 2; Cambridge, 1896), where a full translation of the Sahidic version may be found. The second fragment (Chap. 13) has been omitted from the text given here because of its poorly preserved state. S. Morenz has devoted a short monograph to the study of certain motifs in this text, especially the Egyptian background of the death scene in Chapters 21-23; that work, Die Geschichte von Joseph dem Zimmermann (Texte und Untersuchungen 56; Berlin, 1951) also contains a German translation of Chapters 14-24.1 of the Sahidic version. The text is presented as it appears in the published sources except for the division of the words. There are many unusual spellings, but the reader should be able to cope with them by this stage. The text is narrated by Jesus, who delivers a brief aside to his apostles in 22:3.

GYAFFGAION KATA AOYKAC

Chapter I

- (1) ENELAHNEP $\lambda-2\lambda^2$ 21-TOOTOY E C2 λ 1 N Ngaxe etbe ne2bhye entaytwt N 2ht 2pal N2htN, (2) kata be entaytaac etootN N61 nentaynay 2N neybaa xin N gopN, e-aygwne N 2ynepethc M ngaxe, (3) λ 1 \overline{p} -2nal 2w, e-aloya2 $\overline{\tau}$ NGA 2wb nim xin N gopN 2N oywpX, etpac2alcoy nak oya oya, kpaticte beoblae,
- (4) XEKAC GREEINE 6 ROPX \overline{N} NOAXE ENTAYRABHEEL MMOR \overline{N} 2HTOY.
- (5) Agome $2\overline{N}$ Ne200y \overline{N} 2HPWAHC \overline{n} PPO \overline{N} †0YALLA \overline{N} 61 OYHHB G- \overline{n} 64PAN \overline{n} 6 ZAXAPIAC, 64H \overline{n} 6 Ne200y \overline{N} ABIA, G- $\overline{0}$ 9 \overline{N} 7 $\overline{4}$ 0YC2IM6 6BOA $2\overline{N}$ \overline{N} 9 $\overline{0}$ 66PG \overline{N} APWN e- \overline{n} 6CPAN \overline{n} 6 GAICABET.
- (6) неущооп ас пе $\overline{\mathbf{m}}$ песнау $\overline{\mathbf{n}}$ акалос $\overline{\mathbf{m}}$ пенто евох $\overline{\mathbf{m}}$ пноуте, еумоофе $2\overline{\mathbf{n}}$ лентолн тнроу $\mathbf{m}\overline{\mathbf{n}}$ лакалома $\overline{\mathbf{m}}$ пхоетс еуоуаав. (7) ауш не-милтоу фире $\overline{\mathbf{m}}$ мау пе, евох хе не-оуабрин те ехісавет, ауш лтооу $\overline{\mathbf{m}}$ песнау не-ауалал пе $2\overline{\mathbf{n}}$ неу200у.
- (8) асфине де $2\overline{M}$ птречой $2\overline{M}$ с \overline{M} нечголу \overline{M} пемто евол \overline{M} пмоуте, (9) ката псим \overline{M} тийтоуння асратич е тале-фоугние вграі, е-ачвик егоум є перпе \overline{M} пховіс.

^{1.} energhner (Émelőhner) conj. inasmuch as. Tot tettot Q that to become agreeable; to agree (on, upon, to: e, $ex\overline{n}$; with: $m\overline{n}$); tot \overline{n} 2ht 2pl 2 \overline{n} to become agreeable, acceptable to or among.

^{2.} π. εγπερετικ (ὑπηρέτης) assistant; custodian.

^{3.} κρατιστο: voc. of κρατιστος (κράτιστος): 0 most excellent Theophilos.

^{4.} κλθηγει Μπο΄ επ (καθηγέομαι) to instruct in.

^{6.} π. Αικλιωμα (τὸ δικαίωμα) act of justice; ordinance.

^{7.} AGPHN (adj. or n.f.) barren (woman).

^{8.} τ.τλεις (ἡ τάξις) order, arrangement; rank, post.

^{9.} n.cwnT custom; GIPG M ncwnT to follow the custom.
ACPATW4: it became his turn; an impers. expression, the
exact analysis of which is uncertain. n.goy2HNG incense.

- (10) душ нере-пинные тиря й пллос флил й псл вол й пилу й проугние. (11) д-плггелос де й пхоетс оуших или евол ечлгелтя й сл оунам й перустастирном й проугние.
- (12) AUGITOPT AS $\overline{N}61$ ZAXAPIAC \overline{N} TEPSUNAY, AYW AY2OTS 26 62PA1 SXW4. (13) NEXS-NAFISOC AS NAM X6

мпрр-20те хахаріас, же аусютм є пексопс. Тую тексгіме елісавет снажпо нак й оудире, нёмоуте є печран же ішганне. (14) оуй-оураде надюпе нак мй оутеана, аую оуй-гаг нараде ежй печхпо. (15) чнар-оуное гар й пемто евох й пховіс, аую йнечсе-ирй гі сікера, аую чнамоуг евох гй пенйа ет оуаль хін ечгй гитс й течмаху. (16) аую чнакте-оуминде й йдире й пінх є пховіс пеуноуте. (17) аую йточ чнамооде га течги гй пепйа мй теом й гнаїас, є кто й йгит й йеіоте є неудире аую йатсютм гй тмптрййгит й йаікаіос, є совте й оуалос й пховіс ечсётют.

- (18) Lyw nexe-zlxlpilc \overline{m} neffeloc xe $_2\overline{n}$ by the the e hal? Lnok flp lip-2xlo lyw tecline lclil $_2\overline{n}$ nec2ody.
- (19) х-пхггехос де оуфуб, пеххч ихч хе хиок пе гавріна, пет агератч й пенто євох й пиоуте. хутйнооут є фахе ийнак хуф є тафе-оеіф илк й ихі. (20) еіс гинте екеффпе еккф й рфк, ймй-феом ймок є фахе фа пегооу етере-ихі ихффпе, етве хе йпкпістеує є ихфахе, ихі ет иххфк євох гй пеуоуоеіф.
- (21) плаос де нечефу $\overline{\tau}$ 2нт $\overline{\tau}$ и хахартас пе, ауф неу \overline{p} -фпнре итереческ 2 \overline{m} перпе. (22) итеречет де свол, ипечфе \overline{m} -бом е

^{11.} πε. θγειλετηρίον (τό θυσιαστήριον) altar.

^{13.} π.coπc entreaty, prayer; coπc ceπc- or concπ cπcπ-cπcωπ* Q cπcωπ to entreat, implore (ΜΜΟ*).

^{15.} π. εικερα (τὸ σίμερα) strong drink.

^{22.} жюрм, Q хорм to make a sign, beckon (to: 6, оув 6; with: ммо, гм). мпо, емпо adj. dumb, mute; р-мпо (Q о м мпо) to become mute.

фахе мймау, ауш ауетме же йтачнау бубшай бвох $2\overline{M}$ перпе. йточ аб нечжшрй оувну пе, ауш ачеш бчо й быпо. (23) асфшпе аб йтере-негооу й печфффе жык бвох, ачвык бграт б печнт. (24) мййса нетгооу аб асш йет батсавет течсгіме, ауш асгопс й фоу й бвот, бсхы ймос (25) же

TALL TO BE $\overline{N}TA$ - $\overline{N}XOEIC$ AAC NAL $2\overline{M}$ $\overline{M}E2OOY$ $\overline{N}TA46\omega\phi\overline{T}$

- (26) $2\overline{M}$ nme2cooy as \overline{N} grot ayxooy \overline{N} rabpiha harreadc groalthin noyte synolic \overline{N} Te traditala s-nechan ne nazapes, (27) wa oynapesnoc s-ayw \overline{M} -toot \overline{C} \overline{N} oyaal s-nespan ne lwch ϕ
- (27) 94 ОУПАРОСНОС $G-\lambda Y9\overline{n}-TOOT\overline{C}$ \overline{N} ОУ2 λI G-HG4PAN HG I WCH φ $GBO\lambda$ $2\overline{M}$ HH \overline{N} $A\overline{\lambda}A$, λYW HPAN \overline{N} THAPOGNOC HG MAPIA. (28) λYW $\overline{N}TGPG4BWK$ NAC G2OYN, HEXAY NAC XG

XAIPE, TENTACEN-2HOT. TXOGIC NAME.

(29) $\overline{\text{N}}$ тос де асфторт \overline{p} сх $\overline{\text{M}}$ пфахе, ауш несмокмек $\overline{\text{M}}$ мос хе оуаф $\overline{\text{M}}$ міне пе петаспасмос. (30) пехе-паггелос нас хе $\overline{\text{M}}$ п $\overline{p}\overline{p}$ -готе, марта. аретне гар $\overline{\text{N}}$ оугмот $\overline{\text{N}}$ нагр $\overline{\text{M}}$ пноуте. (31) ауш етс гните тенаш, $\overline{\text{N}}$ техпо $\overline{\text{N}}$ оуфнре, $\overline{\text{N}}$ темоуте е печран хе $\overline{\text{TC}}$. (32) пат чнафшпе $\overline{\text{N}}$ оуное, ауш сенамоуте

ероч же пфире \overline{M} пет жосе. пжосіс пиоуте их \uparrow их \overline{M} пефронос \overline{N} длусід печеїшт. (33) душ чид \overline{p} - \overline{p} ро ех \overline{M} пні \overline{N} ілкив фл иіенег, душ \overline{M} М \overline{m} -ган идфиле \overline{N} течи \overline{M} теро.

- (34) $\pi e x e^{-m x} p$ a a \overline{m} $\pi a r r e x o c$ xe \overline{n} ag \overline{n} 2 e πa 1 nagwing $\overline{m} m o$ 1? $\overline{m} \pi e$ 1 coy \overline{n} 2 o o y t.
- (35) λ -паггелос оуфб, нехач нас же оуппа ечоуаль пет нну егра ежф, ауф теом \overline{M} пет хосе тет на \overline{P} -галь \overline{C} сро. етве пал пстенахноч чоуаль.

^{25.} NOGNE NEGNEG- NEGNOYE' to mock, reproach (ΜΜΟ'); as n.m. reproach, scorn.

^{27.} gπ-τοοτ' NA' lit., to grasp the hand of (someone) for, i.e. to betroth (a woman) to (a man); the Q is expressed as τοοτσ ghn NA4, she is betrothed to him (II, 5).

^{28.} χαιρε (χαῖρε) Greetings. 33. Text has αναρρο.

^{34.} coyn-200yr to know a man (sexually); cooyn + 200yr.

сепамоуте ероч же прире \overline{N} плоуте. (36) ауш етс елісавет тоусуггение \overline{N} тос гшшс он асш \overline{N} оутире г \overline{N} тесн \overline{N} тесн \overline{N} тех \overline{N} х ω , ауш песмегсооу \overline{N} евот пе пат, тат едаумоуте ерос же табрии, (37) же \overline{N} ие-хазу \overline{N} даже \overline{P} -атбом \overline{N} иагр \overline{M} плоуте.

(38) пехас де N61 марта хо етс гинте амб-өмгах й пхоетс. маресфюпе нат ката пекфахе.

ауш а-паггелос вык евох 21тоотс. (39) астшоумс де $\overline{\text{N}}$ 61 марта $2\overline{\text{N}}$ метгооу, асвык е тортин $2\overline{\text{N}}$ оубени е тполіс $\overline{\text{N}}$ тоудата. (40) асвык егоум е пит $\overline{\text{N}}$ хахартас, асаспаде $\overline{\text{N}}$ елісавет. (41) асфыпе де $\overline{\text{N}}$ тере-елісавет сытй е паспасмос $\overline{\text{M}}$ марта, а-пырре фим кім граї $\overline{\text{N}}$ 2 нтс, ауш а-елісавет моуг евох $2\overline{\text{M}}$ пепла ет оуалв. (42) асчі-грас евох $2\overline{\text{N}}$ оумоб $\overline{\text{N}}$ син. пехас же

тесмамаат $\overline{\text{N}}$ то $2\overline{\text{N}}$ мегіоме, аую чемамаат $\overline{\text{N}}$ 61 пкарпос $\overline{\text{N}}$ 2 нте, (43) же ам $\overline{\text{T}}$ -мім амок же ере-тмаау $\overline{\text{M}}$ пахобіс бі ерат? (44) біс гнте гар $\overline{\text{N}}$ тере-тесми $\overline{\text{M}}$ поуаспасмос таге-мамааже, а-пфире фим кім $2\overline{\text{N}}$ оутеана $\overline{\text{N}}$ 2 нт. (45) аую маіат $\overline{\text{C}}$ $\overline{\text{N}}$ темтаспістеує же оу $\overline{\text{N}}$ -оужюк євох мафюпе $\overline{\text{N}}$ ментаужооу мас гіт $\overline{\text{M}}$ пхобіс.

(46) AYW NGXG-MAPIA X6

а-тафухн хісе $\overline{\mathbf{N}}$ пхоєїс. (47) а-папла теана ех $\overline{\mathbf{M}}$ пноуте пасштну; (48) же ачещу ех $\overline{\mathbf{N}}$ пен $\overline{\mathbf{E}}$ віо $\overline{\mathbf{N}}$ теч $_2\overline{\mathbf{M}}_2$ ал, еїс гинте гар хін теноу сенатнаїої $\overline{\mathbf{N}}$ 61 генеа нім, (49) же ачеїре наї $\overline{\mathbf{N}}$ 2 енм $\overline{\mathbf{N}}$ тное $\overline{\mathbf{N}}$ 61 петеу $\overline{\mathbf{N}}$ 7-60м $\overline{\mathbf{M}}$ 60ч, ау $\overline{\mathbf{M}}$ 64 оуаже. (50) печна хін оужем да оужем єх $\overline{\mathbf{N}}$ нет

^{40.} Δεπλζε (ἀσπάζομαι) to greet.

^{48.} ΤΜΑΙΟ ΤΜΑΙΕ- ΤΜΑΙΟ Q ΤΜΑΙΗΥ to justify (ΜΜΟ), to consider just or justified; intr. to become justified.
Τ.ΓΕΝΕΑ (ἡ ΥΕΝΕά) generation.

^{50.} π.χωм generation.

 \overline{p} -20TG 2HT \overline{q} . (51) AGGIPG \overline{N} OYGOM 2 \overline{M} REGGEOI; AGXOUPG GBOX \overline{N} NXACI-2HT 2 \overline{M} REGGYG \overline{N} NEY2HT. (52) AGGOPG \overline{p} \overline{N} RAYNACTHC 2 \overline{N} REGPONOC; AGXICG \overline{N} NET OF BIHY.

- (53) ачтете-иет гкаетт й аганой; ачжеу-примаю суфочетт. (54) ач+-тоот \overline{q} м п \overline{H} а печг \overline{m} гаа е \overline{p} -пиненуе \overline{m} пиа (55) ката не ентачувже м \overline{n} иенеготе аврагам м \overline{n} печеперма фа енег.
- (56) λ -маріа де єю гагтно й фомйт й бвот, аую аското еграї є посні. (57) λ -посусеї ф де жюк бвол й блісавет стресмісе, аую асжпо й суфире. (58) аусютй де йбі несрійраун мій нессуггенно же λ -пхобіс тафе-печил міймас, аурафо міймас. (59) асфюпе де гій пметфисум й госу ауєї бунаське й пфире фим. аумоуте броч й прам й печеїют же гахаріас. (60) λ -течмалу де суффі, пежас же

MMON. AXXA GYNAMOYTE EPOT XE 102ANNHC.

(61) пехху де нас же

MN-XXXY 2N TOYPAITE GYMOYTE GPOU M HEIPAN.

(62) NEYXWPM AE OYBE HEHELWT XE KOYEW-MOYTE EPOH XE NIM?

^{51.} nc.680: arm (of man), leg (of animal). xwwpe xcepe-xoop* Q xoope (± 680λ) to scatter, disperse (Μπο*); also more generally: to bring to naught.

^{52.} gopgp gpgp- gpgwp Q gpgwp to overturn, upset (νησ); as n.m. overthrow, destruction. π.Δγνωστης (δ δυνάστης) ruler.

^{53.} λγλθον (τὸ άγαθόν) n. good, what is good.

^{55.} ne.cneρμα (τὸ σπέρμα) seed; offspring, issue.

^{58.} $p\overline{m}p\lambda\gamma H$ cpd. of $p\overline{M}$ (27.2) and τ . $p\lambda\gamma H$ neighborhood, town-quarter; hence: neighbor.

^{59.} CEBE CEBE- CEBHT' Q CEBHY to circumcise ($\overline{\text{M}}\text{MO}'$); as n.m. circumcision. 60. $\overline{\text{M}}\text{MO}$ No. equine $\overline{\text{M}}\text{MO}$ otherwise.

^{61.} T. PAITE kin, kindred; PMPAITE kinsman.

(63) ачаттет де \overline{N} оуптиакте, ачегат ечхю \overline{M} мос же тюганине пе печран. Аую ду \overline{p} -фпире тироу. (64) д-рюч де оуюн \overline{N} теумоу \overline{M} печаде, ачфаже, ечемоу е пиоуте. (65) дуготе де фюпе ех \overline{N} оуон итм ет оуиг $2\overline{M}$ пеукюте, дую $2\overline{N}$ тортин тир \overline{C} \overline{N} †оудата исуфаже пе $2\overline{N}$ истраже тироу. (66) д-иентаусют \overline{M} де тироу каду $2\overline{M}$ пеугит, бужю \overline{M} мос же

EPETHELOHPE OHM NAPTOY?

кат гар тетх \overline{m} пхоетс несфооп нямач пе. (67) λ -хахартас де печетот ноуг евох $2\overline{m}$ пепя ет оудав, ачпрофитеуе, ечхо мнос (68) хе

чемамаат $\overline{\text{N}}$ 61 пноуте $\overline{\text{M}}$ $\overline{\text{ПНА}}$, же ачей-печфіне ауш ачеїре $\overline{\text{N}}$ оусште $\overline{\text{M}}$ печалос. (69) ачтоунос $\overline{\text{N}}$ оутап $\overline{\text{N}}$ оужаї нан $2\overline{\text{M}}$ пні $\overline{\text{N}}$ лаубіл печ $2\overline{\text{M}}$ 2ал. (70) ката $96\overline{\text{N}}$ тачфаже $21\overline{\text{N}}$ ттапро $\overline{\text{N}}$ нечірофитне ет оуаав жін ене2, (71) $\overline{\text{N}}$ оуоужаї евох $21\overline{\text{I}}$ ненхаже ауш евох $2\overline{\text{N}}$ теіх $\overline{\text{N}}$ оуон нім ет носте $\overline{\text{M}}$ мон, (72) є біре $\overline{\text{N}}$ оуна $\overline{\text{M}}$ ненбіоте, є $\overline{\text{F}}$ -пмебує $\overline{\text{N}}$ течалавнки ет оуаав, (73) панаф $\overline{\text{N}}$ тачф $\overline{\text{K}}$ $\overline{\text{M}}$ моч $\overline{\text{N}}$ аврагам пенеїшт, є $\overline{\text{T}}$ -96 нан (74) ах $\overline{\text{N}}$ готе, $\overline{\text{E}}$ -анноу $\overline{\text{E}}$ $\overline{\text{E}}$ евох $21\overline{\text{N}}$ ненхаже, є $\overline{\text{M}}$ ще нач (75) $2\overline{\text{N}}$ оуоуоп $\overline{\text{M}}$ оуаїкаїосуни $\overline{\text{M}}$ печ $\overline{\text{E}}$ но $\overline{\text{M}}$ 0 евох $\overline{\text{N}}$ ненгооу тироу. (76) $\overline{\text{N}}$ ток де $2\overline{\text{E}}$ шжк,

^{63.} AITEI (αἰτέω) to ask, ask for. π.πινλκια (ὁ πίναξ) writing-tablet.

^{65.} waxe 2N to talk of, about.

^{66.} και ταρ (καὶ γάρ) conj. for, for truly.

^{67.} προφητεγε (προφητεύω) to prophesy.

^{68.} cwtc cct- cot' to redeem, rescue (ΜΝΟ'); as n.m. redemption; cipe N ογαντο Νλ' to make a redemption for.

^{69.} n. Tan horn; trumpet.

^{72.} ειρε \overline{N} ογηλ $M\overline{N}$ to do a kindness to, for. τ.ΔιλθΗΚΗ ($\hat{\eta}$ διαθήμη) will, testament, covenant.

^{75.} τ.Δικλιος γνη (ή δικαιοσύνη) justice, righteousness.

парире, сенамоуте ерок же пепрофитис й ппет жосе.

кнамооре гар 21 он й пжоетс е совте й нечгооуе;

(77) е † й оусооуй й оужат й печалос 2й пкш евол й неунове (78) етве тмйтрй-2тич мй пна й пенноуте 2й нетечилей-пенртие йгнтоу йет поуоети евол 2й пжтсе,

(79) е р-оуоети е нет 2моос 2й пкаке мй нет 2моос 2й олго й пкаке мй нет 2моос 2й олго й пкаке й пет 2моос 2й олго й пкаке ий нет 2моос 2й олго й пкаке ий нет 2моос 2й олго й пкаке ий нетечине.

(80) прире де дин ачаузане ауш ачей-бом 2й пепйа. нечрооп де пе 2й йжле фа пегоуший евол й птих.

Chapter II

(1) асфипе ав $2\overline{N}$ мв200у ет ммау ауаогма ет евох $21\overline{T}\overline{M}$ прро аугоустос етре-тоткоуменн тнр \overline{C} с211 мса местме. (2) тат те тфорпе \overline{N} апографн ентасфипе ере-куртнос о \overline{N} 2нгемим е тсурта. (3) ауш меувык тнроу пе поуа поуа е с2114 мса течполіс. (4) ачвык е2рат 2шшч мет толіс \overline{N} алуета, евох $2\overline{N}$ махарее тполіс є фоуаліа є тполіс \overline{N} алуета, тефаумоуте ерос же внелеем, же оуевох $2\overline{M}$ пні пе м \overline{N} тпатріа \overline{N} алуета, (5) етречтали е20ум м \overline{N} марта, тетере-тоот \overline{C} фип мач, есеет. (6) асфипе ае $2\overline{N}$ птреуфипе $2\overline{M}$ пма ет \overline{M} маухык евох \overline{N} 61 мегооу етресмісе. (7) асхпо \overline{M} песфире,

^{79.} COOYTN COYTN- COYTWN Q COYTWN to straighten, stretch out (ΜΜΟ΄); intr. to become straight, upright; COOYTN ΜΜΟ΄ 6 to direct toward, make fit for.

^{80.} ΑγχΑΝε (αύξάνω) to grow up.

^{1.} π.λογμα (τὸ δόγμα) decree. τ.οικογμανη (ἡ οίμουμένη) the world. $c_2 \lambda_1$ να to register by, according to; note the medio-passive intransitive use of $c_2 \lambda_1$.

^{2.} τ. απογραφή (ἡ ἀπογραφή) enrollment, registry.

^{4.} τ.πατρια (ἡ πατριά) family, clan; people, nation.

^{5.} τλλ4 ε20γN reflex.: to register himself (from +).

^{7. 6}ωωλα 666λα- 600λ° Q 600λα to swathe, clothe (ΜΝΟ΄).

Τ.τοεια rag, piece of cloth; swaddling-clothes. ***πτο- ***πτο- Q **

песфрп-м-місе, асеоолеч \overline{N} гентовіс, асхточ $2\overline{N}$ оуоуом \overline{V} , хе не-мм \overline{N} -ма фооп нау пе $2\overline{M}$ пма \overline{N} боіле. (8) неу \overline{N} -генфоос де пе $2\overline{M}$ пма ет мнау, еуфооп $2\overline{N}$ тсюфе бугарег $2\overline{N}$ поурфе \overline{N} теуфи е пеуоге \overline{N} есооу. (9) а-паггелос \overline{N} пховіс оуфи \overline{V} нау евол, ауф а-пеооу \overline{M} пховіс \overline{V} -оуобім ерооу; ау \overline{V} -готе $2\overline{N}$ оунов \overline{N} готе. (10) пехе-паггелос де нау хе

мпрр-20те. его 2ннте глу \uparrow таде-осід интй й оуноб й раде, па і ет надшпе й палос тирй, (11) же аужпо интй й пооу й псштир, ете па і пе пехс пхоєїс, 2й тполіс й алуета. (12) душ оунаєти интй пе па і: тетна 2 е еудире фим ечеоле й 2 ентоєїс ечки 2й оуоуом $\overline{4}$.

(13) Lygwine $2\overline{N}$ Gygcne $\overline{M}N$ harreloc \overline{N} 61 Gymhhge \overline{N} tectpltil \overline{N} the eyemgy e hnoyte eyem \overline{M} moc (14) Xe

пеооу \overline{M} пиоуте $2\overline{M}$ иет хосе, хую \uparrow рнин 21х \overline{M} пк λ 2 $2\overline{M}$ \overline{M} рюме \overline{M} печоуюф.

(15) acquire as \overline{N} teps- \overline{N} affence but espai sitootoy s the, neps- \overline{N} good game MN neysphy ms

MAPNBOK DA BHORGEM, NTNNAY 6 REIGAXE ENTAGORE ENTAGORE OYON $2\overline{4}$ EPON.

(16) жубепн де, жуеї, жуге є маріх мії ішснф мії піднре финенк гії поуомії. (17) ї тероунку де, жубіме є підже є підже є піджооч нау єтве піднре фин. (18) жуш оуон нім битаусштії журтфінре єжії нента-піфоос жооу нау. (19) маріх де несгарег є неі фаже тироу пе, єски мінооу граї гії пісстит. (20) жукотоу де пісії піфоос, єуфтеооу жуш бусмоу є піноуте єжії нентаусотноу тироу жуш хунау ката ве єнтаужоос нау. (21) ї терефіноун де пі гооу жик євох єтреуєвніті, жумоуте є пістра же тіс, пісітатії євос тали єрои ємпатій міноч гії вн. (22) жуш пітероужшк євох пісі нісгою тість пість пість

^{8.} τ.ογγφε watch. π.ο2ε flock, herd; pasture; fold.

^{13.} τε. στρατια (ή στρατιά) army, host.

^{14.} \overline{M} πεσογωφ: this renders Gk. εύδομίας (men of his favor) rather than the alternate reading εύδομία.

(23) КАТА ӨЕ ЕТ СН2 $2\overline{N}$ ПИОМОС \overline{N} ПХОЕІС ХЕ 200УТ NІМ ЕТ НАОУФИ \overline{N} ТООТЕ ЕУНАМОУТЕ ЕРОЧ ХЕ ПЕТ ОУАЛВ \overline{M} ПХОЕІС, (24) ДУФ Е $\frac{1}{N}$ ПООРОСІЛ КАТА ПЕНТДУХООЧ $2\overline{M}$ ПИОМОС \overline{N} ПХОЕІС ХЕ ОУСОЕІФ \overline{N} ЕРЙПФАН Н МАС СНАУ \overline{N} ЕРООМПЕ. (25) ЕІС 2ННТЕ ДЕ НЕЎ \overline{N} —ОУРФИНЕ ПЕ $2\overline{N}$ ӨІЕРОУСЛАНИ Е-ПЕЧРАН ПЕ СУМЕФИ. ДУФ ПЕІРФИЕ ИЕУДІКАЛОС ПЕ \overline{N} РЕЧФЙФЕ \overline{M} ПИОУТЕ, ЕЧЕФФ \overline{T} ЕВОЛ 2НТ \overline{T} \overline{M} ПСОЛС \overline{N} М П \overline{T} \overline{N} , Е-ОУ \overline{N} —ОУП \overline{N} Л ЕЧОУДЛВ ФООП ИЙМАЧ, (26) Е-ДУТАНОЧ ЕВОЛ 21 \overline{T} ПЕП \overline{N} Л ЕТ ОУДЛВ ЖЕ ИЧІЛЬМОУ ДІ С-МП \overline{T} ИЛУ Е ПЕ \overline{X} С \overline{M} ПХОЕІС. (27) ДУФ ДЧЕІ $2\overline{M}$ ПЕП \overline{N} Л Е ПЕРСЕ \overline{M} ПСФИТ \overline{M} Л ПОМОС $2\overline{N}$ РОЧ, (28) \overline{N} ТОЧ ДЕ ДЧХІТ \overline{T} Е ПЕЧ $2\overline{M}$ НР, ДЧСМОУ Е ПИОУТЕ, ЕЧХФ \overline{M} МОС (29) ХЕ

теноу кнако свол \overline{M} пек $_2\overline{M}_2$ ал, пхоеіс, ката пекфахе $_2\overline{N}$ оусірнин, (30) же а-навал нау є пекоухаї, (31) паї ситакс \overline{B} тот \overline{q} \overline{M} пемто свол \overline{N} ладос тнроу, (32) поуосін субол \overline{n} евол \overline{N} \overline{N}_2 сенос ауо є пеооу \overline{M} пеклаос п \overline{I} \overline{I}

61C ПАІ КН 6Y26 МЙ ОУТШОУН Й 2A2 2 \overline{M} П $\overline{H}\overline{X}$, AYW ОУМАЄІН 6 ОУШ \overline{M} 21 \overline{U} 21 \overline{U} 400 А6 ОУЙ \overline{M} 600 СІТЙ ТОУЧУХН, ЖЕКАС 6Y66 \overline{U} 6 ВОЛ Й61 ЙМОКМЕК Й 2A2 Й 2HT. (36) НС \overline{U} 0YПРОФНТНС А6 Ж6 АННА ТФЄЄРЕ \overline{M} ФАНОУНА ТЕ

^{23.} T.OOT @ Womb.

^{24.} τε.θγειλ (ἡ θυσία) sacrifice. π.coειφ pair.
τε.θγπηφλη turtle-dove. μ (ἡ) conj. or. π.μλς the young of any animal.

^{28.} n. 22NHp embrace, arms.

^{32.} π. ε ε θ Noc (τὸ ἔθνος) nation, people.

^{34.} OYW2 \overline{M} 21 to contradict, object to; note OYW2 \overline{M} OYBG in the same meaning.

^{36.} The exact function of τε is not clear; it is not required in the sentence as it stands. τε.φγλη (ἡ φυλή) tribe, people, nation. τ.Μπτροογης virginity; π.ροογης

GBOA 2N TGOYAH N ACHP. TAL AG ACALAL 2N 26N2OOY 6-NAGWOY, E-ACP-CAMPE N POMNE MN NECZAL XIN TECHNTPOOYNE (37) AYW $\lambda C \overline{P} - \chi H P \lambda$ WANT $\overline{C} P - 2 M E N E T \lambda 4 T E \overline{N}$ POMME. TALLAG MECC $\overline{N} - \overline{M} = \overline{M} + \overline{M$ евох. есфйфе й теуфн ий пегооу гй генинстета ий генсопс. (38) 2N TEYNOY AG ET MMAY ACAZEPATO, ACEXZOMOROFGI M TIMOGIC, AYW NECHAME MN OYON NIM ET 600T GBOA 2HTY M HCWTE N OTAHM. (39) NTEPOYXWK AE 6802 N61 2WB NIM KATA MNOMOC H EXOCIC, AYKTOOY CZPAI C TEANIALIA C TCYHOAIC NAZAPCO. (40) notice we will attain by Ne46 \overline{M} -60M, eamer \overline{M} codia, epe-texapic $\overline{\mathbf{M}}$ innoyte 21x04. (41) nepe-negetote as but he трромпе е отаны ы пара ы ппаска. (42) ытеречр-мытскооче де \overline{N} pomne, eynabwk ezpai kata ncwn \overline{T} \overline{M} nwa, (43) ayw $\overline{\text{NT}}$ GPOYXWK 680 λ $\overline{\text{N}}$ N6200Y, 6YNAKTOOY, A46W $\overline{\text{N}}$ 61 $\overline{\text{H}}$ 0HP6 $\overline{\text{H}}$ HM $\overline{\text{IC}}$ $2\overline{N}$ $\Theta \overline{1}\overline{\lambda}\overline{H}\overline{M}$. \overline{M} \overline{N} Θ \overline{N} Θ \overline{N} Θ \overline{N} Θ \overline{N} Θ \overline{N} Θ \overline{N} \overline{N} Θ \overline{N} \overline{N} TE21H NMMAY. \overline{N} TEPOYP-OY200Y AS \overline{N} MOOGS, AYGINS \overline{N} COY $2\overline{N}$ NEYCYTTENHC MN NET COOYN MMOOY. (45) AYW NTEPOYTM2E GPO4, AYKTOOY e_{2} PAL e θ $\overline{1}$ AHM eY ϕ ING \overline{N} C ϕ 4. (46) AC ϕ ϕ Re Ae M \overline{N} RCA CONNT N 200Y AY26 6PO4 2M HEPHE, 642MOOC N TMHTG N NCA2, ETCOTH EPOOY, ETENOY MMOOY. (47) AYF-QUILPE AS THEOY \overline{N} 61

virgin, virginity.

^{37.} cine cn- caat to pass through, across; cine mmo eBoa to leave, pass out of. τ. NHCTEIA (ἡ νηστεία) fasting.

^{38.} εχομολογει (έξομολογέω) to confess, acknowledge.

^{40.} τ. coφιλ (ή σοφία) wisdom. τε.χλρις (ή χάρις) grace.

^{41.} τ̄ρροκπε, τ̄κροκπε adv. yearly, annually. π.πωςχω (τὸ πάσχα) Passover.

^{42.} GYNABUK is difficult. If Circumstantial of Fut. I, there is no main verb; if Fut. II, the tense is incorrect. It appears to be due to a slavish rendering of the Gk., but fails to carry the construction into the next verse, as the Gk. requires.

^{44.} \overline{p} -0 γ 200 γ \overline{N} Nooge lit., to spend a walking-day, i.e. to walk for a day.

ист сфтм броч бхй тбчийтсав в ий ибчбіноуффв. (48) аунау де броч, аур-фпире. Пбхб-тбчидау нач хв пафире, йтакр-оу най 21 на 1? віс 2ннте анок ий пбкбіфт бимок \overline{x} й 2нт бифіне йсфк.

- (49) пехач ас илу же етве оу тетйшие йсш!? йтетйсооүй ли же глис етраеш ги ил-плеішт?
- (50) $\overline{\text{NTOOY}}$ де $\overline{\text{МПОYEIME}}$ в праже $\overline{\text{NTAYXOOY}}$ нау. (51) ачет де спеснт ийнау еграт в нахарев, ауш нечештй йешоу. Течналу де несгарег в неграже тироу гй песгит. (52) $\overline{\text{1C}}$ де нечирокоптет гй теофта мй өнаткта мй техарге нагри пноуте мй йршив.

Chapter III

(1) 2N тс \overline{n} м \overline{n} де \overline{n} өнгемоній \overline{n} тівіріос клісар, ечо \overline{n} гигемом ех \overline{n} фоудалій йеї помтіос пілатос, ере-гиродис о \overline{n} тетрархис ех \overline{n} тгалілата, ере-філіппос печсом птетрархис сх \overline{n} ідоурата м \overline{n} ттрахомітіс \overline{n} хора м \overline{n} лусамідс птетрархис ех \overline{n} тавілини, (2) ере-динас пархієреус пе м \overline{n} кліфас, \overline{n} -пудхе \overline{n} пиоуте уюпе уа імганинс пунре \overline{n} хахаріас гате теримос. (3) ачеї еграї є тперіхорос тир \overline{c} \overline{m} піорадинс

^{48. 21} NAI adv. in this way, thus.

^{52.} προκοπτει (προκόπτω) to progress, advance. ΘΗλΙΚΙΣ (ἡ ἡλικία) age, time of life.

^{1.} cπ- or cen-, proclitic form of a f. noun meaning year in date formulas: τcπ-мπτ the fifteenth year.

ΘΗΓΕΝΟΝΙΑ (ἡ ἡγεμονία) rule, administration. π.τετρλαρχης (ὁ τετράρχης) tetrarch, petty prince. The circumstantial clauses ερε-φιλιππος ... and ερε-ληγις ... are not grammatically correct as they stand.

^{2. 2}ATE, 2ATN 2ATOOT prep. near, by, with; a synonym of 2A2TN, with which it is virtually interchangeable.

^{3.} κγρισσλι (μηρύσσω) to announce, proclaim.

ечкуріссьі $\overline{\mathbf{M}}$ пвыптісмы $\overline{\mathbf{M}}$ метаноїх $\overline{\mathbf{N}}$ кх-нове евох, (4) $\overline{\mathbf{N}}$ ее ет снz 21 пхююме $\overline{\mathbf{N}}$ муже $\overline{\mathbf{N}}$ нсы іхс пепрофитнс же тесми $\overline{\mathbf{M}}$ пет юф евох z $\overline{\mathbf{N}}$ теримос же с $\overline{\mathbf{B}}$ те-теz ін $\overline{\mathbf{M}}$ пхобіс; сооут $\overline{\mathbf{N}}$ $\overline{\mathbf{N}}$ нечмы $\overline{\mathbf{M}}$ моофе. (5) еїх нім намоуz, $\overline{\mathbf{N}}$ те-тооу нім z1 сів $\overline{\mathbf{T}}$ нім еввіо; хую нет бооме нафипе еусоутюм $\overline{\mathbf{M}}$ нет наф $\overline{\mathbf{T}}$ е zenz100уе еусхебаює. (6) хую пеооу $\overline{\mathbf{M}}$ пхобіс наоуюм \overline{z} евох, $\overline{\mathbf{N}}$ те-сар \overline{x} нім нау є поухаї $\overline{\mathbf{M}}$ пноуте.

(7) NEYXW 66 MMOC HE N MMHHWE ET NHY EBOX E BANTIZE EBOX 2 LTOOTY XE

межпо \overline{N} мезчю, мім пентачтамот \overline{N} є пот євох знт \overline{C} \overline{N} торгн ет мну? (8) арі—земкарпос ає єуйпора \overline{N} тметамоїа, йтетйтмархеї \overline{N} хоос же оуйтам пенеї от авразам. +хоймос мн \overline{N} же оуй—бом \overline{N} пмоуте є тоумес—земонре \overline{N} авразам євох $2\overline{N}$ меї оме. (9) жім темоу пкедевім кн за тмоуме \overline{N} йонм. Они мім єте \overline{N} чма+-карпос ам \overline{C} -мамоуч семакоореч \overline{N} Семох \overline{T} є пко \overline{C}

- (10) λ - \overline{M} MHH ϕ e λ e xNOY4, ϵ Yx ω \overline{M} NOC x ϵ OY 6 ϵ π e \overline{M} N λ λ 4 x ϵ eneoyx λ 1?
- (11) ачоушу \overline{e} , e^{4} х ω ммос илу хe пете оу \overline{n} т \overline{q} - ω тни с \overline{n} те илреч+-оуeі й пете ийтач, лу ω пете оу \overline{n} т \overline{q} -оeік илречeірe ои 21 илі.
- (12) $\lambda = 2 \in NKETE \lambda WNHC$ AG E1 E XI-BARTICNA EBOX 21TOOT $\overline{\P}$.

^{4.} ωφ εφ- οφ' εκολ to cry out; to read, recite.

^{5.} n.GIA valley, ravine. T.CIBT hill. GOOMG Q of GOOMG to twist, pervert (MNO'); intr. to become crooked, twisted. CAGGAGG Q of CAOGAG to make smooth; intr. to become smooth.

^{7.} βλητιχε (βαπτίζω) to baptise; note active form with passive meaning. 204 (f. 24ω; pl. 280γι) n.m. snake, serpent. τ.οργι (ἡ ὁργή) wrath.

^{12.} π. τελωνης (ὁ τελώνης) tax-collector.

DX PAN YAXON

TCA2, ENNAPTOY?

- (13) \overline{N} TO4 A6 \overline{N} REALY NAY X6 \overline{N} ROYO \overline{N} RENTAYTO9 \overline{A} NHT \overline{N} .
- (14) AYXNOY4 A6 $\overline{\text{N}}$ 61 NET O $\overline{\text{M}}$ MATO1 X6 ENNA $\overline{\text{P}}$ -OY 2000 ON?

пехач нау же

 \overline{M} \overline{M}

(15) EFE-RAACC GOOT GBOA, GYMGGYG THPOY $2\overline{N}$ NGY2HT GTBG $1\omega_2$ annhc ag mggak \overline{N} toy ng ng \overline{x} , (16) a- $1\omega_2$ annhc oyog \overline{g} , eyaw \overline{M} moc \overline{N} oyon nim ag

амок мен енвантихе ймютй $2\overline{N}$ оумооу. Чину де йен пет хоор ерон, пан е-и+йпфа ам \overline{N} вфа евоа й пмоус й печтооуе. Йточ пет навантихе ймютй $2\overline{N}$ оуп \overline{N} ечоуаль мй оукф $2\overline{T}$, (17) пан етере-печга $2\overline{N}$ теченх е твво й печхнооу, е сфоуг егоун й печсоуо е течапофики. Птф2 ас чиарок $2\overline{T}$ гоусате е-месфф.

(18) 2N 2GNKGYAXG AG G-NAGWOY $NGUNAFAKAAGI <math>\overline{M}MOOY$, $GUTAYG-OGIY <math>\overline{M}$ NAAOC. (19) 2HPWAHC AG \overline{M} \overline{M}

^{13.} τως τες- τος Q της to bound, limit, determine, fix (Μης).

^{14.} ττο ττε- ττο* to make (someone: first object) give (second object). π.οcε fine; loss, damage; ττε-λλλγ οcε to force payment out of someone. π.λλ slander; 21-λλ to slander (ε). 2ω ε to be satisfied with; used with ethical dative ερο* (530.6). π.οψωνιον (τὸ ὁψώνιον) wages.

^{16.} xoop Q of xwwpe to become strong, powerful. n.moyc strap, band. n.tooye shoe, sandal.

^{17.} π. 2x winnowing fan. πε. xnooy threshing-floor.
π. τω chaff. τ. caτε fire. ωρῶ ερῶ- ορω Q ορῶ to quench (πιο*); intr. to become quenched.

^{18.} παρακάλει (παρακάλέω) to exhort $(\overline{M}MO^2)$.

^{19.} T.ZIME Wife.

ммоч евох 2 ітоот етве 2 нрюдіхс, етме м печсом хую етве 2 шв мім м помиром емта—2 нрюдис длу, (20) дчоує 2 — петке ехм мечкооує тироу: дчет $\overline{\pi}$ —1 шезаминс егоум є потеко. (21) дсоюпе де $2\overline{m}$ птре—палос тир \overline{q} хі—вантісма дую мтере— $\overline{1C}$ хі, дчодих, д—тпе оуюм. (22) д—пеп \overline{n} д ет оудав єї епесит ехоч $2\overline{n}$ оусмот \overline{n} сома \overline{n} ее \overline{n} оуброомпе, дую дусми оропе евох $2\overline{n}$ тпе хе

 \overline{N} ток по пафире, памеріт. \overline{N} та іоую \overline{N} гит \overline{K} . The remainder of Chap. III is genealogy and has been omitted.

Chapter IV

- (1) ТС ДЕ ЕЧХНК ЕВОХ \overline{M} П \overline{N} Х ЕЧОУДДВ, ХЧКОТ \overline{q} ЕВОХ $2\overline{M}$ ПІОРДДИНС, ЕЧМООФЕ $2\overline{M}$ ПЕП \overline{N} Х 21 ТЕРНМОС (2) \overline{N} 2ME \overline{N} 200Y, ЕУПЕІРДХЕ \overline{M} МОЧ 21Т \overline{M} ПАІДВОЛОС, ХУФ \overline{M} П \overline{q} ОУЕМ-ХДДУ $2\overline{N}$ NE200Y ЕТ \overline{M} МДУ. \overline{N} ТЕРОУХФК ДЕ ЕВОХ, ХЧ2КО. (3) ПЕХЕ-ПАІДВОЛОС ИЛЧ ХЕ ЕФХЕ \overline{N} ТОК ПЕ ПФНРЕ \overline{M} ПИОУТЕ, ХХІС \overline{M} ПЕІФИЕ ХЕ \overline{q} 4 \overline{p} 7-ОЕІК.
- (4) A 4 O Y W W B NA 4 NG 1 TC X G 4 CH 2 X G 6 P G T P W M G NA W N $\overline{2}$ AN G NO G I K MMATG.
- (5) дчхіт \overline{q} де егрлі, дчтоуоч є мм \overline{n} терфоу тнроу \overline{n} то ікоуменн г \overline{n} оустігмн \overline{n} оуоєі \overline{g} . (6) пехе-пдідволос де \overline{n}

++ нак \overline{n} тегезоуста тнр \overline{c} м \overline{n} пеуеооу, же \overline{n} таутаас наг, ауш фагтаас \overline{m} пе+оуаф \overline{q} . (7) \overline{n} ток ее екфаноушф \overline{r} \overline{m} пайто евол, снафшпе нак тнр \overline{c} .

^{20.} ωτπ ϵ τπ- σ τπ' Q σ τπ $(\pm \epsilon_2 \sigma \gamma N)$ to imprison, enclose, shut in $(\overline{M}M\sigma')$.

^{22.} Ντλιογωφ is Perf. II since this is an independent clause.

^{5.} τογο τογο to show, teach (someone: Μπο'; something: ε). τε. στιγμή (ή στιγμή) moment.

^{7.} ογωφτ to worship, greet, kiss (mmo*, na*).

- (8) λ - $\overline{1C}$ оушу $\overline{8}$, пехач нач же чсн2 же екнаоушу $\overline{7}$ \overline{M} пхоетс пекноуте, ауш екнаумуе нач оуаач.

емже йток пе пюнре й пиоуте, чобк епеснт 21xй пенма, (10) чсн 2 гар же чиа 2 wn етоотоу й нечаггелос етвинтк етреугарег ерок. (11) ауш сенач ітк ехй неубіх, мипоте игжюрй сушне й текоуерите.

- (12) $\lambda = \overline{1C}$ As symmetry nexts and the largest meknoyte.
- (13) $\overline{\text{N}}$ теречхек-петрасмос де нім євох, д-патаволос сагшч євох ймоч фа оуоуобіф. (14) дую дчкточ йбі $\overline{\text{тс}}$ гй тбом й пенйа є тгалілата. Д-псобіт єї євох гй тперіхюрос тир $\overline{\text{с}}$ етвинт $\overline{\text{ч}}$. (15) йточ де неч \uparrow -свю пе гй неусупагюги, сре-рюме нім \uparrow -вооу нач. (16) дчеї єграї є нахара, пма єнтаусаноуф $\overline{\text{ч}}$ йгит $\overline{\text{ч}}$, дую дчвюк єгоун ката печсюн $\overline{\text{г}}$ й негооу $\overline{\text{м}}$ псавватон є тсупагюги. Дчтюоун де є юф. (17) ду \uparrow нач $\overline{\text{н}}$ пхююме $\overline{\text{п}}$ нсатас пепрофитис. Дчоуми $\overline{\text{н}}$ пхююме, дчге є пма єт сиг (18) же

пепил и пхобіс єгрлі єхші. Єтвє плі лятагст, лятиносут є булггеліге й йгнке, є тафе-обіф й сукш євол й йліхмалштос ни сунау євол й йвхле, є хосу й нет сусф \overline{q} ги сукш євол, (19) є тафе-обіф й теромпе и пхобіс єт фип.

^{8.} $\pi.\tau\bar{N}_2$ wing; wing of a building. 4066 466- 406° Q 486 to leap, move quickly; reflex. idem.

^{11.} xwpπ to stumble; tr. to strike (Μmo*) against (ε).

^{14.} n.cocit fame, report.

^{16.} π. carbaton (τὸ σάββατον) the sabbath.

^{18.} τως τες τες τας Q τας to anoint (Μησ'; with: ς Ν, Μησ'). π.λιχηλωστος (ὁ αίχμάλωτος) prisoner, captive.
Ογωμ σογεφ σογοφ σογοφ σογοφ το wear down, destroy; also intr. to be worn down, destroyed.

- (20) дакб-пхююме де, датада м пгупиретис, дагмоос. иере-мвад \overline{N} буон иім ет г \overline{N} тсундгюги бюрт ероч.
- (21) AMAPXET AS \overline{N} XOOC NAY XS \overline{M} NOOY A-TETPPAGH XWK GBOX $2\overline{N}$ NET \overline{N} MAAXE.
- (22) Lyw nepe-dyon nim \overline{p} -mntpe nnmlq, ey \overline{p} -whipe n nwlxe n texlpic et nhy ebol $2\overline{n}$ pwq, eyxw mhoc xe

мн м пфире и тосиф и пе па 13.

- (23) пбхач аб нау же пантюс тетнахи на і \overline{n} тенпараволн, же псабін, арі-пагре брок. Нентансют \overline{m} брооу же ауфипб $2\overline{n}$ кафарнаоум арісоу гюсу г \overline{m} пбіна $2\overline{m}$ пбк \dagger не.
- (28) AYMOY2 AG THPOY \overline{N} GONT $2\overline{N}$ TCYNAFOFH GYCOTM G NAI.

^{20.} $\kappa\omega B$ $\kappa\varepsilon B$ -, $\kappa\overline{B}$ - κOB Q κ HB to make double; to fold $(\overline{M}MO^{\circ})$.

^{22.} τε.χωριο (ή χάρις) grace, favor.

^{23.} πλιτως (πάντως) adv. wholly, altogether. F-πλιρε to heal (e); π.πλιρε drug, medicament. Note reflex. εροκ.

^{24.} ελμην (άμήν) adv. indeed, verily.

^{25.} gtam vb. tr. intr. to shut, close ($\overline{\text{M}}$ Mo*); to close, become sealed. $\pi.26-8\omega\omega$ M famine, bad harvest; cpd. of 26 season, $8\omega\omega$ M adj. bad.

^{27.} $n.cos_{\overline{2}}$ leper; $cos_{\overline{2}}$, $Q cos_{\overline{2}}$ to become leprous; $n.cos_{\overline{2}}$ leprosy. Note 21 at the time of; $\overline{N}c\lambda$ except for.

(29) AYTWOYN, AYNOX \overline{q} GBOX RBOX \overline{N} THOXIC, AY \overline{N} T \overline{q} \overline{y} A RKOO2 \overline{M} RTOOY etepe-teyhoxic kht 21xw4 2wcte etpeyhox \overline{q} GBOX \overline{N} XO4 \overline{N} N. (30) \overline{N} TO4 Ae A461 GBOX 2 \overline{N} Teyhhte, A4BWK.

(31) дчет епеснт е кдфдридоум тполіс йте тгалілдід, дую иечтовю пе 2й йсдвватой. (32) дуў-финре де тнроу 62 рдт ехй течсвю, же нере-печфаже фооп пе 2й оублоуста. (33) дую неуй-оурыме пе 2й тсунагыгн ере-оупйд й длімойтой й акафартой й2нт $\overline{4}$. Дую дчх $\overline{1}$ -фкак євох 2й оуноб й смн (34) же

 λ_2 рок ийман, $\overline{1C}$ прйнадарее? аке е такон. \dagger сооуй же итк-ин итк, пет оуаав и пноуте.

(35) λ -TC λ 6 emitima na4, 64xw \(\text{MMOC} \) x6 τ M-PWK NF61 680\(\text{N}_2\) HT\(\text{T}_4\).

ачноуже ймоч йет паатмонтон е тмнте, ачет евох й $_2$ нт $_4$ е-мп $_4$ валитет ймоч хаху. (36) ауфторт $_7$ ае фюпе ежй оуон итм, ауфаже мй неуерну, еужю ймос же

(37) λ -ncoeit as mooge etbhht $\overline{4}$ $2\overline{M}$ ma nim \overline{N} thepixopoc.

(38) ачтшоун де евох $2\overline{N}$ тсунагшгн, ачвшк егоун е пні \overline{N} сімшн. тфшне де \overline{N} сімшн неу \overline{N} -оуное \overline{N} гишс пе. аусепсшту де етвинт \overline{C} . (39) ачагерат \overline{Y} гіхшс, ачепітіма \overline{M} пегмон, ачкаас. \overline{N} теуноу астшоун, асалаконеї нау.

^{29.} π.κοο₂ angle, corner. πχουτπ adv. headlong.

^{33.} x1-gkak 680a to cry out; ne.gkak cry, shout.

^{34.} Note use of reduced form NTK for NTOK.

^{35.} enitima ma* (êπιτιμάω) to rebuke, reprove. вланте: \overline{H} MO* (βλάπτω) to harm, injure.

^{36.} OY62-Ch2NG to order, command (NA*; that: 6, etp6).

^{38.} T.gome mother-in-law; n.gom father-in-law. 2mom, Q 2HM to become hot; ne.2mom heat, fever.

^{39.} ΔΙΔΚΟΝΕΙ ΝΑ΄ (διακονέω) to wait on, serve.

(40) ере-при де нагити, отон нім ете отптот-рюме будюме г \overline{n} гендюме будове атптот нач. Пточ де ачтале-тоот \overline{q} ехм пота пота ммоот, ачтал600г. (41) нере-палімоніон де нну евол г \overline{n} гаг пе, бухі-дкак євол, бухю ммос хе

 \overline{N} ток пе прире \overline{M} пноуте. ауш нечепітіма нау с-н \overline{q} кю \overline{M} нооу ан є фахе, хе неусооу \overline{N} хе \overline{N} точ пе пе \overline{X} С. (42) \overline{N} тере- $\overline{2}$ тооує де фшпе, ачеї евоа, ачвык бума \overline{N} хате. нере- \overline{M} минфе де фіне \overline{N} Сыч пе. аубі фароч, ауама $\overline{2}$ те \overline{M} ноч е т \overline{M} Вык є каау. (43) \overline{N} точ де пехач нау же

2את \overline{C} פּדף אפּץאררפּאוצפּ \overline{N} \overline{N} אנפּחסאונ \overline{N} דא \overline{N} דפּף \overline{M} האסץ דפּ, אנפ \overline{N} דא איז איז איז איז איז איז פֿר פּוּציע.

(44) NEUKHPYCCE AG NE $2\overline{N}$ \overline{N} CYNAFWFH \overline{N} †0YAA1A.

Chapter V

- (1) асфиле де $2\overline{M}$ птре-пиннуе дого ехич йсеситй є пуахе й пиотте, йточ де неча2 брату пе 2 атй танин й генинсарев.
- (2) ATHAY E XOI CHAY EYMOONE 2ATH TAIMHH, $E-A-\overline{N}OYW2E$ RE E1 E2PA1 21WOY, EYEIW \overline{N} NEYWHY. (3) ATAAE AE E OYA \overline{N} $\overline{N}XOI$

^{40.} $2\omega\tau\bar{n}$ $2\varepsilon\tau\bar{n}$ — $2\sigma\tau\bar{n}$ Q $2\sigma\tau\bar{n}$ vb. tr. to reconcile, adjust ($\bar{m}mo^*$; to: ε , $m\bar{n}$); intr. (1) to become reconciled; (2) to set (of the sun, etc.). Note pome in indef. pron. sense "anyone," with plural resumption in $\varepsilon\gamma g\omega n\varepsilon$.

^{1.} goyo goye- goyo' vb. tr. to pour, empty out $(\overline{M}MO'';$ out of: $680 \times 2\overline{N});$ intr. to flow, pour forth. T.XIMNH $(\mathring{\eta} \ \lambda \ell \mu \nu \eta)$ lake.

^{2.} MOONE MGNE-, MANE- Q MANOOYT Vb. tr. to bring (boat) to land, into port; to moor (MMO'; at, to: ε); intr. to come to land, into port, be moored. n.ογω2ε fisherman. ne.gne (pl. Ne.gnhy) net.

^{3. 2} ING to row (GBOA $\overline{\mathbf{N}}$: away from).

 $e-n_{\lambda}-c$ імши пе. λ чхоос илч етречгіне євол й пекро й оукоу і. λ чгмоос де гі пхої, λ ч \uparrow -св ω й ймни ω е.

- (4) $\overline{\text{N}}$ теречоую ає бчюдже, пехач $\overline{\text{N}}$ сімон же кет-тнут $\overline{\text{N}}$ є нет юнк, $\overline{\text{N}}$ тет $\overline{\text{N}}$ хала $\overline{\text{N}}$ нет $\overline{\text{N}}$ онну є бюпе.
- (5) λ -cimon as Gyoge, hexay nay be nca2, ang \overline{n} -21ce \overline{n} teygh the \overline{c} , \overline{n} \overline{n} $\overline{6}$ \overline{n} -alay. Stee hexala \overline{n} neghty.
- (6) \overline{N} тероур-па 1 де, дусшоуг егоун \overline{N} оуминфе \overline{N} тет е-нафшоу. Нере-неуфину де напшг пе. (7) духшр \overline{M} е неуфвеер ет 21 пкехої стреуєї \overline{N} Се \uparrow -тоотоу н \overline{M} Мду. дуеї де, думег-пхої снау гшсте етреушм \overline{C} . (8) \overline{N} Тере-сімши петрос нау е па 1, дапагт \overline{T} га \overline{N} Оуєрнте \overline{N} \overline{T} \overline{C} , ечхш \overline{M} Мос же

CAZWK EBOA MMOI, XE ANT-OYPWME N PETP-NOBE, MXOEIC.

(9) NG-LY2OTE FLY TL2OU RE MN OYON NIM ET NMMLU EXN TCOOY2C N NTET ENTLYSONC. (10) 20M01WC AE RKE ILKWBOC MN 1W2ANNHC, NGHPE N ZEBELLIOC, NEYO N KOINWNOC N CIMWN. REXE-TC N CIMWN XE

MULL TENON CHARGE CREEN-LONG.

(11) AYMANG-NEXHY AG G HEKPO, AYKA- \overline{N} KA NIM \overline{N} COOY, AYOYA2OY \overline{N} COU4. (12) ACGOHG AG, G42 \overline{N} OYGI \overline{N} MHOAIC, GIC OYPOMG G4MG2 \overline{N} COB2 A4NAY G \overline{I} C, A4HA2T \overline{I} EX \overline{M} HG42O, A4C \overline{I} COUT \overline{I} , G4X \overline{M} MMOC XG

TXOGIC, GKQANOYWQ, OYN-60M MMOK 6 TEBOL.

^{4.} ушк у в к у о к $^{\prime}$ Q унк to dig deep; Q to be deep; и в у и к the deep places. $^{\prime}$ в о $^{\prime}$ Сатс $^{\prime}$ Сатс $^{\prime}$ (хаха (хаха) to let down, lower.

^{5.} gn-21ce to labor, work with difficulty.

^{6.} $\pi\omega_2$ $\pi\varepsilon_2$ - $\pi\lambda_2$ Q $\pi\mu_2$ vb. tr. and intr. to burst, tear, break ($\overline{N}MO^2$).

^{9.} T.cooy20 gathering, collection; catch (of fish).

^{10. 20}Μοιως (ὁμοίως) adv. likewise. π.κοινωνος (ὁ κοινωνός) partner.

(13) λ 4COYTN-T6461X λ 6 680 λ , λ 4X ω 2 6 ρ 04, 64X ω MMOC X6 \uparrow 0Y ω 9. T880.

аую \overline{n} тбүноү а-псюв \overline{z} каач. (14) \overline{n} точ аб ачпараггенаб нач же

 \overline{M} П \overline{p} ХООС 6 ХЛХУ, ДХХД ВФК, N \overline{r} ТОУОК 6 ПОУННВ, N \overline{r} ТЛХО 62 \overline{p} 2 \overline{M} Пект \overline{g} 80 ката өб ентачоуб2-сл2 N6 \overline{M} МОС \overline{N} 61 МФУСНС 6 \overline{Y} М \overline{N} Т \overline{p} 6 NAY.

- (15) мере-пуахе де моофе \overline{n} гоуо етвинт \overline{q} , дую мере- \overline{m} минфе сфоуг егоум е сф \overline{m} броч дую е талбооу $z\overline{n}$ меуфиме.
- (16) \overline{N} TO4 A6 NE4C126 \overline{M} MO4 NE 6 26NMA \overline{N} XA16, 64 \overline{Q} AHA.
- (17) асфине де, еч+-свы й оугооу, ере-генфарісалос гмоос мй генномодіалскалос, нал ентауєї евох гй †ме нім йте тгалілала мй †оудала мй өтляй, нере-теом де й пхоєїс фооп пе етречталео. (18) етс генрыме де ауй-оурыме гіхй оуелое ечсне, ауш неуфіне пе йса хітч егоун е калч й печйто евох. (19) е-мпоуге де е тегін є хітч егоун етве пминфе, аувык еграї є тхенепыр, аухала ймоч епеснт гітй йкерамос мй пеєлоб є теумите й пемто євох й $\overline{\text{тс}}$. (20) ачнау де є теупістіс, пехач хе

проме, некнове ки нак евоа.

(21) a-nerpammateye as \overline{m} nepapicatoe apxet \overline{m} mokmek, \overline{e}_{YXW} \overline{m}_{MOC} as

NIM HE HALL ET XI-OYA? NIM HETE OYN-60M MHOU N

^{13.} $x\omega_2$, Q xH_2 vb. tr. to touch (e),

^{14.} παραγγέλλω) to order, command.

^{16.} C126 C62- CA2T' vb. reflex. to withdraw, go away; also intr. to be removed.

^{17.} Νε.φαρισαίοι (οἱ φαρισαῖοι) Pharisees. π.ΝοΜΟΔΙ-Αλοκαλος (ὁ νομοδιδάσκαλος) teacher of the law.

^{19.} π.κεραμός (ὁ μέραμός) tile.

^{21.} πειγαμματεύς (ὁ γραμματεύς) scribe, clerk. χι-ογα, χε-ογα to blaspheme (against: ε); πιογα blasphemy.

KA-NORE EBOX NCA HNOYTE MAYAA4?

(22) $\overline{\text{NT}}$ бе е е е меумокмек, пехач мау же агроти тетимене $2\overline{\text{N}}$ метигнт? (23) ар гар пет моти е хоос пе, же мекмове кн мак евоа, жи е хоос пе, же тооум итмооре? (24) жекас ае ететиение же оуйтепринре и проме схоуста 21хи пкаг е ка-мове евоа — пехач и пет сне же

 ϵ_1 хю ммос нак же тфоун нтч 1 м пекблоб; вфк ϵ пекн 1. (25) й теумоу ав ачтфоун м пеумто евох, ачч 1 м печблоб, ачвфк ϵ печн 1 еч \dagger -вооу м пноуте. (26) ауг-фпн 1 а 1 тнроу, ау \dagger -вооу м пноуте, аумоу 1 готе, 1 вужф ммос же, аннау 1 1 готе, 1 гот

- (27) MNNCA HAI AGEI GBOA, AGNAY GYTGAWHHC G-HGGPAN HG AGYGI GG2MOOC $2\overline{M}$ HGGTGAWHIOH. HGXAG HAG XG OYA $2\overline{K}$ HCWI.
- (28) $\lambda 4 K \lambda \overline{N} K \lambda \Delta \varepsilon$ NIM $\overline{N} C \omega 4$, $\lambda 4 T \omega O Y N$, $\lambda 4 O Y \lambda 2 \overline{4}$ $\overline{N} C \omega 4$.
- (29) AYW A-REYET F-OYNOS \overline{N} GONG GPO4 $2\overline{N}$ RESHIT. NEYN-OYMHHQE AS \overline{N} TERWINC $M\overline{N}$ 2 SHKOOYS $N\overline{M}$ MAY SYNHX.
- (30) а-нефарісатос мі неграмматеус крітрі єгоун є нечмантис, єухо імос же

ETBE OY TETÑOYWM AYW TETÑCW MÑ ÑTEAWNHC AYW $\Bar{Npe4P}$ -nobe?

(31) A-TC AG OYWOB, NGXA4 NAY XG

NGT THK \overline{p} -XPIA AN \overline{m} NCAGIN, AXXA NGT MOK \overline{z} NGT \overline{p} -XPIA

NA4. (32) NTAIGI AN \overline{G} TG2 \overline{m} -NAIKAIOC AXXA \overline{N} PG4 \overline{p} -NOBG

^{23.} xw conj. or.

^{27.} π. τελώνιον (τὸ τελώνιον) tax-house.

^{29.} T. your a reception, entertainment, banquet.

^{30.} κρῶρῶ vb. intr. to murmer, complain (against: ε, εἰογν ε, εἰκ, κα).

^{31.} TWK TEK- TOK' Q THK vb. tr. to strengthen, confirm; reflex. and intr. to become strong, firm, hale, hardy.

^{32.} $\tau\omega_2\overline{H}$ $\tau\varepsilon_2\overline{M}$ $\tau\lambda_2M'$ Q $\tau\lambda_2\overline{M}$ vb. tr. to summon ($\overline{M}MO'$, ε); vb. intr. to knock at the door. Metanogi ($\mu\varepsilon\tau\alpha\nu\circ\varepsilon\omega$) to repent.

6 METANOEL.

- (33) \overline{N} TOOY AS \overline{N} NA4 XS \overline{M} MASHTHC \overline{N} 102ANNHC NHCTSYS \overline{N} 2A2 \overline{N} CON AYO CSCONC, \overline{N} TOOY $\overline{M}\overline{N}$ NA-NSSAPICATOS. NOYK AS OYOM, CSCO.
- (34) пеже- $\overline{1c}$ илу же мн оуй-вом ймштй етре-йфире й пмл й фелеет инстече, ере-пл-тфелеет иймлу? (35) оуй-генгооү ле ину буилч й пл-тфелеет йтоотоу. Тоте сеилинстече $\overline{2}$ й иегооү ет ймлу.
- (36) дажи де нау \overline{N} кепараволн х з мере-хаду с \overline{X} п-сутое ic 21 суфтни \overline{N} дал натор п \overline{C} суфтни \overline{M} пасе. ефине \overline{M} мон, чилие2-ткефтни \overline{N} дал, душ \overline{N} тетм-тое ic \overline{N} дал \overline{p} -дау е тихее. (37) душ мере-хаду ноуже \overline{N} сунр \overline{n} \overline{N} в \overline{p} ре е 2 енаскос \overline{N} дс. ефине \overline{M} мон, даре-пнр \overline{n} \overline{N} в \overline{p} ре пе2- \overline{N} аскос, напшие евол, \overline{N} те- \overline{N} кеаскос тако. (38) дала ефаунеж-нр \overline{n} \overline{N} в \overline{p} ре е 2 енаскос \overline{N} в \overline{p} ре. (39) мере-хаду де суеф-нр \overline{n} \overline{N} в \overline{p} ре, ечсе-нр \overline{n} дс. фачхоос гар же неч \overline{p} -перп-дс.

^{34.} T. geneet bride; Ma \overline{N} geneet bridal chamber; (n.) \overline{N} Tuenet the groom.

^{36.} cwxπ cxn- coxn° Q coxπ vb. tr. to break off, cut off (ΜΜο΄); intr. to break, burst. gal adj. new. τωρπ τορπ΄ Q τορπ vb. tr. to sew (ΜΜο΄; to: ε). π.πχεε rag; gthn Μπχεε tattered garment. π.gay use, value, profit; γ-gay to be useful, of value, to prosper.

^{37.} π.λcκος (ὁ ἀσμός) wineskin. πων(ε) π $\overline{\text{N}}$ -, πεν- πον' Q πην (± εβολ) vb. tr. to pour ($\overline{\text{M}}$ Mo'); intr. to pour, flow.

Apophthegmata Patrum

- 4. ДЧХООС \overline{N} 61 ДПД ПОІМНИ Х6, "Д-ОУСОИ ХООС \overline{N} ДПД ПДНС6 Х6, "GINДР-ОУ \overline{M} ПД2НТ 64ИДФ \overline{T} ? \overline{N} † \overline{P} -20T6 ДИ \overline{N} 2НТ $\overline{4}$ \overline{M} ПИОУТ6. "ПЕХД4 ИД4 Х6, "ВФК ИРТОБ \overline{K}^1 6УСОИ 64 \overline{P} -20T6 \overline{N} 2НТ $\overline{4}$ \overline{M} ПИОУТ6, ДУФ 6ВОД 2 \overline{N} ТМ \overline{N} 7Р-20T6 \overline{M} ПИСУТ6."
- 5. λ -Oya xng-Oy2 $\overline{\lambda}$ 30 xg, "etbe oy, gi2mooc 2 \overline{m} nama \overline{n} gwng, na2ht kwtg ca ca nim?" λ 40ywg \overline{g} naq \overline{n} 61 n2 $\overline{\lambda}$ 30 xg, "gboa xg cgwng \overline{n} 61 nekgcohthpion et 21 boa: tginnay, tgincwt \overline{m} , tgingwa \overline{m} , 2 tgingaxg. Nai 6g gwng gkyanxno \overline{n} tgygngpii \overline{a} 2 \overline{n} 0ym \overline{n} tkabapoc, \overline{a} gape- \overline{n} kggcohthpion et 21 20yn gwng 2 \overline{n} 0yc \overline{g} 32 \overline{r} 5 m \overline{n} 0yoyxai.
- 6. λ -OYA ON XNG-OY2 $\overline{\lambda}$ NO XG, "GTBG OY \uparrow 2MOOC 2 \overline{H} NAMA \overline{N} GWNG, \uparrow 2 λ \overline{H} NWN?" λ 4OYWG \overline{E} NAM XG, "GBOX XG \overline{H} NATEKGIWP \overline{z}^2 \overline{M}

^{3. (1)} λπλητά ε (ἀπαντάω) to meet, confront.

^{4. (1)} Twee Tee- Toe' Q THE vb. tr. to join, attach ($\overline{\text{M}}$ No'; to: 6); used reflex. here.

^{5. (1)} π.εσεμτηρίου (τὸ αίσθητήριου) sense-organ. (2)

9ωλΗ vb. tr. to smell. (3) π.εμεργία (ή ένεργία) function,
action. (4) κλελρος (καθαρός) pure; μητκλελρος purity.

⁽⁵⁾ cepa27 vb. intr. to pause, rest, become still.

^{6. (1) 2} λ 0 π 1 π 7, Q 2 λ π 1 λ 0 π 0 vb. intr. to become despondent. (2) ϵ 1 ω 9 ϵ 2 ϵ 1 ϵ 9 ϵ 7 ϵ 1 ϵ 9 ϵ 7 vb. tr. to perceive, see (π 1 π 0°).

 $n\overline{n}$ тон $6t\overline{n}_2$ бапіге $\frac{3}{6}$ броч оуде тколісіс $\frac{4}{6}$ бт надшпе. 6- не-ак 6 ібр \overline{z} -наі $2\overline{n}$ оушр \overline{x} , душ \overline{n} те-пекма \overline{n} дшпе моуг \overline{n} в \overline{n} брок дантоупше вераі 6 некнаєщ бераі \overline{n} 2 неоу н \overline{t} 7 не н \overline{t} 9 не н \overline{t} 9 не н \overline{t} 1 гантоу пе н \overline{t} 9 не н \overline{t} 1 гантоу пе н \overline{t} 3 гантоу пе н \overline{t} 3 гантоу пе н \overline{t} 4 гантоу пе н \overline{t} 5 гантоу пе н \overline{t} 5 гантоу пе н \overline{t} 6 гантоу пе н \overline{t} 7 гантоу пе н \overline{t} 8 гантоу пе н \overline{t} 9 гантоу пе н \overline{t} 9 гантоу пе н \overline{t} 9 гантоу пе н \overline{t} 1 гантоу пе н \overline{t} 1 гантоу пе н \overline{t} 2 гантоу пе н \overline{t} 3 гантоу пе н \overline{t} 3 гантоу пе н \overline{t} 4 гантоу пе н \overline{t} 4 гантоу пе н \overline{t} 5 гантоу пе н \overline{t} 6 гантоу пе н \overline{t} 7 гантоу пе н \overline{t} 7 гантоу пе н \overline{t} 8 гантоу пе н \overline{t} 9 гантоу

- 9. ATXOOC ON X6, "TNHCTIA HE HEXALINOC 1 M HMONAXOC 64+ OYBE HNOBE. HET NOYXE N TAI CABOA MMON OY2TO N AAR-C2IME HE.
- 10. Lyzooc on x6, "ncwml et goywoy 1 nte nmonlxoc eqcwk 2 n teyyxh e2pli 2n ngik 3 nte necht, lyw nqtpe-n2ylwh 4
- 11. Layrooc on Xe, "imonlyoc \overline{N} 2lk gayt-klom exwa 2 \overline{M} inkl2, lyw on 2 \overline{N} \overline{M} inkye gayt-klom exwa \overline{M} impto ebol \overline{M} inhye."
- 12. LAXOOC ON X6, "THONLXOC ET LML2TE LN \overline{M} THEYLL MATTAL THE HING EP-XOEIC E LLY \overline{M} THEOC 2 ENG2."
- 13. LANGOO ON ME, "MULTAOVE-NAMY M GAME 64200Y 680A 2N Tektanpo. TBW N 620026 rap mectaoye-gonte 1 6802."

^{(3) &}lt;sub>2</sub> exnize G (έλπίζω) to hope for. (4) τ.κοχλοίο (ἡ κόλασοις) punishment, correction. (5) τ.αΝτ (τ.εΝτ) worm.

⁽⁶⁾ п.моте neck.

^{9. (1)} ne.xxxinoc (ὁ χαλινός) bridle. (2) xxb-c2ime adj. lusty, lecherous; lit. female-crazed, from xibe, Q xobe to rage, be mad, p.c. xxb-.

^{10. (1)} gooye, Q goywoy vb. intr. to become dry, dry up. (2) cwk cek- cok' Q chk vb. tr. to draw, drag, impel (Μπο'); also intr. to be drawn, move swiftly, flowingly. (3) n.gik depth(s). (4) θγωνη (ἡ ἡδονή) pleasure, delight.

^{11. (1) 2}λκ adj. sober, mild, prudent.

^{12. (1)} ΜΑΧΙCTA (μάλιστα) adv. especially. (2) π.πΑΘΟC (τὸ πάθος) suffering, misfortune, calamity.

^{13. (1)} T. GONTE the acacia nilotica, a thorn tree; hence: thorns.

- 14. λ 4x00C on x6, "NANOY-OY6M- λ 8 1 λ 7 ω 6 C6-HP $\overline{\Pi}$ NFT \overline{M} -OY ω M 2 λ 6 \overline{N} \overline{N} CAPX \overline{N} NEKCHHY 21T \overline{N} TKATAXAXIA."
- 15. ачхоос он же, "нта- π_2 оч 1 коскес 2 е еуга 3 дантоунож евол г $^{\rm H}$ ппарадісос. 4 ере-пет каталалі 5 $^{\rm H}$ печсон
 тітшн е па і. дачтако гар $^{\rm H}$ течухн $^{\rm H}$ пет сшт $^{\rm H}$, ауш течкеоуе і 6 $^{\rm H}$ ін $^{\rm H}$ моч мечтангос.
- 16. дуфа де фопе \overline{N} оуовіф $2\overline{N}$ фінт, дуф ду † \overline{N} оудпот \overline{N} нр \overline{n} оу $2\overline{N}$ до. пехдч же, "чі євод ймої й пімоу." \overline{N} тєре-пкесеепе де мду ет оуфи мймач, йпоухі.
- 17. AYXI AG ON \overline{N} OYCAIAION \overline{N} HP \overline{n} \overline{N} Anapxh \overline{N} XG GYGTAAG \overline{N} HGCNHY KATA OYANOT \overline{G} ROYA. A-OYA AG \overline{N} NGCNHY BWK G2PAI GX \overline{n} TKYNH, \overline{N} AUNOT GBOA 2IXWC, AYW \overline{N} TGYNOY AC2G \overline{N} GI TKYNH. AYBWK AG \overline{G} NAY GTBG NG2POOY \overline{N} TAUGWNG, AY2G \overline{G} NCON GUNHX 2I NGCHT. AY2I-TOOTOY \overline{G} CW \overline{G} \overline{M} MMOU, GYXW \overline{M} MOC XG, " \overline{N} T \overline{K} -OYNAI-GOOY GUGOYGIT. KAXWC \overline{N} A-NAI \overline{G} WMOK." A-NZ \overline{N} AO AG WAM \overline{G} GPO4, GYXW \overline{M} MOC XG, "AXWT \overline{N} 2A NAWHPG. OY2WB I'AP \overline{G} -NANOYUNG \overline{N} TAUAAAU. 4ON \overline{G} \overline{N} GI NXOGIC XG \overline{N} NGYKGT-TGIKYNH \overline{G} \overline{M} NA-OYOGIQ TAPG-TOIKOYMGNH THP \overline{G} GIME XG AYKHNH \overline{G} \overline{G}

^{15. (1)} π.204 (f. τε.24ω) snake, serpent. (2) κοςκες = κλςκς to whisper. (3) εγ2λ Eve. (4) π.πλρλλισος (ὁ παρά-δεισος) Paradise, Eden. (5) κλτλλλι (ματαλαλέω) to slander. (6) ογει is used pronominally: his own one (soul).

^{17. (1)} π. CAIAION (Τὸ σαΐτιον) keg. (2) Τ. ΑΠΑΡΧΗ (ἡ ἀπαρχή) first-fruits; μρπ π απαρχΗ new wine. (3) Τ. ΚΥΠΗ, Τ. ΚΗΠΗ arch, vault, vaulted place. (4) comp comp Q chy vb. tr. to scorn, treat with contempt (ਜ Μο΄). (5) κααως (μαλῶς) adv. well. (6) ωαπ ελμ΄ Q ολπ vb. tr. to embrace (ε). (7) An oath: "As the Lord lives,..."

OYANOT N HPH.

- 18. LYCON KIM 2M NG46WNT 620YN 6 0YL. L4L26PLTT 6 NG9LHL, L4L1TEI 6 XI N 0YNNT2LP92HT 6 EXM NG4CON LYW 6 NLPLTE 2 M nniplchoc LXM NWLL. 3 LYW N TGYNOY L4NLY GYKLNNOC 64NHY 6BOL 2N TG4TLNPO. NTGP6-NLI L6 9WNG, L4LO 64GONT.
- 19. ДЧВФК Й ОУОБІФ ЙБІ ПБПРБСВУТБРОС 1 Й ФІНТ ФД ПДРТ ЖНЕПІСКОПОС Й РДКОТЕ ДУФ ЙТБРБЧКТОЧ Є ФІНТ, ДУЖНОУЧ ЙБІ НБСИНУ ЖЕ, "БРБ-ТПОЛІС \overline{P} -ОУ?" ЙТОЧ ДБ ПБЖДЧ НДУ ЖЕ, "ФУСІ, 2 НДСИНУ, ДНОК ЙПІНДУ Є П2О Й ДДДУ Й РФМЕ ЙСД ПДРТ ЖНЕПІСКОПОС МДУДДЧ." ЙТООУ ДБ ЙТБРОУСФТЙ, ДУТДЖРО 3 БТВБ ПФДЖЕ ЖЕ БУБ2ДРБ2 БРООУ 2ДВОД 2Д ПЖІ-2РДЧ 4 Й ЙВДД.
- 21. λ -0ya \overline{N} $\overline{N}2\overline{X}$ 00 bwk \underline{y} 2 k62 \overline{X} 30, ayw пехач \overline{M} печмаентис же, "таміо мам \overline{N} оукоуі \overline{N} аруін," ауж ачтаміоч. пехач же, "2 ep \overline{n} -2 eno ei k aymoyn \overline{M} nezooy thp \overline{q} \overline{M} teygh thp \overline{c} .
- 23. L4XOOC $\overline{\text{MG}}$ 1 ANA 1CAK X6, "M6M61OT6 M6N ANA NAMBO N6Y ϕ 0P61 2 $\overline{\text{N}}$ 26M9THN $\overline{\text{M}}$ 16A66 6Y2 $\overline{\text{N}}$ $\overline{\text{MTO}}$ 1061C $\overline{\text{MN}}$ 26M9THN $\overline{\text{N}}$ 9 $\overline{\text{M}}$ 7 B $\overline{\text{MN}}$ 6. 3 $\overline{\text{NTOTN}}$ AG TENOY TET $\overline{\text{M}}$ ϕ 0P61 26M9THN 6YTA61HY. BOK

^{18. (1) 2}μρω-2μτ adj. patient, long-suffering; μντ2μρω-2μτ patience. (2) πλριτε (παράγω) to pass, pass by, away.

⁽³⁾ $n\omega\lambda\overline{z}$ $no\lambdaz^*$ Q $no\lambda\overline{z}$ vb. tr. to wound, damage, offend.

⁽⁴⁾ π.κληνος (ὁ μαπνός) smoke.

^{19. (1)} πρεσβγτερος (ὁ πρεσβύτερος) elder. (2) ψγςι an expletive of some sort, but cf. gloss 175(5) below. (3) ταχρο ταχρο ταχρο Q ταχρης vb. tr. to affirm, confirm, strengthen (Μηο΄); intr. to be confirmed, resolute. (4) χι-2ρα΄ to amuse or divert self; as n.m. diversion, distraction.

^{21. (1)} אוּסַקּגוּ lentil(s). (2) ניסף פֿר פֿרָה פֿרָה פֿרָה פֿרָה lentil(s). (2) פֿרָה פֿרָה פֿרָה פֿרָה ער פֿרָה ער ניין איז ער ניין איז ניין פֿרָה ער ניין איז ניין איז ניין פֿרָה ער ניין איז ניין פֿרָה ער ניין ער ניין פֿרָה ער ניין פּרָה ער ניין ער ניין ער ניין ער ניין ער ניין ער ניין ער ני

⁽³⁾ NG. ΠΝ (GYMAT) ΙΚΟΝ (Τὰ πνευματικά) spiritual matters.

^{23. (1)} Perhaps insert μπ before ληλ ηλμέω. (2) φορεί (φορέω) to wear. (3) π.g δ δ Νης palm-fiber.

NTWTN M NEIMA! ATETNTAKOT."

- 24. GYNABOK AG G $m_2\overline{c}$, n_{exa4} nay xe, n_{habok} an g kott g +-Gntoah nht \overline{n} ; \overline{n} tet \overline{n} 2apg2 гар an."
- 25. $\overline{\text{NTLM}}$ on LMXOOC XE, "L-LML MLMEW XOOC XE, 'TLI TE BE GTE 996 6 MMONLXOC 6 POPEL $\overline{\text{N}}$ NEG201TE: 20CTE 6 NEXTEM9THN $\overline{\text{M}}$ MBOL $\overline{\text{N}}$ TEMPL $\overline{\text{N}}$ 90M $\overline{\text{N}}$ T $\overline{\text{N}}$ 200Y, $\overline{\text{N}}$ TET $\overline{\text{M}}$ -LLZY TLIOC $\overline{\text{N}}$ 6 41T $\overline{\text{C}}$, tote embor! $\overline{\text{M}}$ MOC."
- 26. LYXOOC NGI LIL KLCILNOC XE, "OYL N NCYNKAHTIKOC, 1 E-LYLNOTACCE N NGYPHMA THPOY, LYTLLY N N2HKG. LYKL-26HKOYI NLY 6TBE TEYXPIL MLYLLY. MIGYOYOG E ONZ 2N OY-HNTLNOTAKTIKOC ET XHK 6BOL NTE NEOBBIO N 2HT. NLI LE LYXO N OYGLXE NLZPLY NGI BLCIMOC, NGT GOON 2N NGT OYLLB, GYXO MHOC XE, 'THNTCYNKAHTIKOC LKCOPMEC, 5 LYO THNTMONLXOC MIGK2E 6POC. 1 "
- 27. λ -Oya \overline{N} necnhy xne-ana nactamon xe, "Oy netinalay, xe cebaibe \overline{N} mhoi ei+ \overline{M} nazub \overline{N} eix eboa?" λ 40ywg \overline{B} \overline{N} 61 nz $\overline{\lambda}$ 0, nexay xe, "nke-ana xixwi m \overline{N} nkeceene gay+-neyzub \overline{N} 61x eboa. Nai \overline{N} oyoce an ne. ekgannoy 2 ae e +, axi-+mh 3

^{24. (1)} $\omega_2 \overline{c}$ $\varepsilon_2 \overline{c}$ - $o_2 c'$ vb. tr. to reap, harvest; as n.m. harvesting, reaping. 2 and c are often interchanged in this word. Note - τ for zero (1st pers. obj.) on KOTT.

^{25. (1)} The sense is that if no one thought it worth taking, it was suitable to be worn by a monk.

^{26. (1)} CYNKAHTIKOC (συγμλητικός) adj. of noble rank;

Τ.ΜΝΤΟΥΝΚΑΗΤΙΚΟΟ nobility. (2) ΔΠΟΤΔΟΟΘ (άποτάσσω) to renounce, give up. (3) πε.ΧΡΗΜΔ (τὸ χρῆμα) goods, money.

(4) π.ΔΠΟΤΔΚΤΙΚΟΟ (άποτακτικός) anchorite, hermit monk;

Τ.ΜΝΤΔΠΟΤΔΚΤΙΚΟΟ status of anchorite. (5) Cωρκ Cepk- Copk() COPK vb. tr. to lose (ΜΝΟ); intr. to go astray, be lost.

^{27. (1)} exise (θλίβω) to afflict, distress; passive construction here. (2) NOY vb. intr. (aux.) to be about to, be going to (do: e + Inf.). (3) τ. †ΜΗ (ή τιμή) price, value.

N OYCON N OYWT NTE HIAOC. 4 EKBANOYWB AE E KA-OYKOYI EBOA 2N COYNTH, 5 NTOK ET TWB. TAI TE BE ETEKNAEN-MTON." HEXERON NAH XE, "EBWHE OYNTAI TAXPIA MMAY, KOYWB ETMTPAHEI-POOYB 2A 2WB N GIX?" AHOYWBE NGI H2XAO XE, "KAN OYNTAK 2WB NIM, MHPKA-H2WB N GIX EBOA. HETE OYN-60M MMOK E AAH, APIH, MONON 8 2N OYWTOPTP AN."

- 28. λ -оусон хие-апа сарапіон хе, "ахі-оуфахе ероі." пехе-п $2\overline{\lambda}$ ло нач хе, "еінахе-оу нак? хе акчі-пенка й йгнке мй нехнра мй йорфанос, аккалу $2\overline{\mu}$ пфоуф $\overline{\tau}$." ачнау гар е пфоуф $\overline{\tau}$ ечме2 й хффме.
- 31. NG-OYN-OYN AG NTG NGT OYNNB GONYMOYTE GOOU XG ϕ 1
 ANTPIOC GUOYH2 2N Θ 1XHM, GUF-2WB 2N OY2ICG GANTGUXHO NAU M

 HEUGEIK MHIN MMOU. N2WCON AG GUA2GPATT 2N TATWPA G \uparrow M

 HEUZWB N GIX GBOA, GIC 2HHTE 2N OYWCNG AUGING NOYBANANTION
 GYN-MHT N GG N 20AOKOTTINOC 2 21WWC. AUA2GPATT M HEUMA, GUXW MMOC XG, "2AHC HE GTPG-HENTAUCOPMEC GI." AYW GIC HET

 MMAY AUGI GUPIMG. AUGONT AG NGI H2XAO, AUXITU N CA OYCA,

 AUTANC NAU. HGT MMAY AG AUAMA2TG MMOU, GUOYWG G \uparrow N OY
 OYWN NAU. H2XAO AG MHGUOYWG G XI. TOTG AU2I-TOOTT G XI
 GKAK GBOA, GUXW MMOC XG, "AMHITN NTGTNNAY GYPWMG NTG HNOYTG

 XG NTAUF-OY." H2XAO AG AUHWT N XIOYG, AUGI GBOA 2N THOAIC

 XG NHGYCOYWHT.

38. AHBOK NGI ANA MAKAPIOC HNOG DA ANA ANTONIOC, AYO

 ⁽⁴⁾ π. ιΔος (τὸ είδος) kind, sort. (5) coynτ' price, value
 (w. suff. only); κλ-ογκογι 68ολ 2π to deduct a little from.

^{(6) 41-}ροογω to be concerned, anxious (about: e, etbe, 2λ), to care about. (7) κλη (μάν) even if. (8) ΜοΝΟΝ (μόνον) only, alone; but (w. neg.).

^{28. (1)} n.woyoT window; niche, alcove.

^{31. (1)} Βλλλατιον (τὸ βαλλάντιον) purse; note resumption as fem. in 21ωως, copmec, τλλς. (2) π.2ολοκοττινος (δ δλοκόττινος) a gold coin. (3) π.ογων part, share.

 $\overline{\text{NTEPERKON}}^{1}$ 6 HPO, AREL 6BOX GAPOR, HEXAR NAR X(6), " $\overline{\text{NTK}}$ -HIM?" NTOU AS AUCYOOF SUXO MHOC XS, "ANOK HE MAKAPIOC." AYW AGWTAM M RPO, AGBWK GZOYN, AGKAAG. NTGPGGNAY 6 TGG-2YTOMONH, 3 A4OYWN NA4, AYW A4OYPOT 4 NMMA4, 64XW MMOC X6, "GIC OYNOG N OYOGIQ GIOYWQ G NAY GPOK. ALCOTH TAP GTBHHTK." WALLEY TOWN OUT - PAG 18 OYMTTMY OF 18 POR POWER WYL EBOX 2N ZENNOG N ZICE. NTEPE-POYZE AE GONE, A-ANA ANTONIoc zwp \overline{n} nag \overline{n} zenkoyi \overline{n} bht. \overline{n} exe-ana makapioc nag xe, "KGAGYG NAI TA2WPH NAI MAYAAT." NTOG AG NGXAG XG, "2WPH." AYW A4TAMIO N OYNOG N WOA N BHT, A420PHT. AY2MOOC, AYфахе е тм \overline{n} роуге. аүнов-TOY, AYW THERE 10 ACEWE ENECHT 6 RECHYAHON 11 680A 21TM ATHAY 6 HAWAI 13 N THERTS N AHA MAKAPIOC, ATP-WHIPS, AYO $\lambda 4 + \pi 1^{14} \in \mathbb{N}_{6} \times \mathbb{N}$ and makaping, equo fimog xe, " $\lambda - 2\lambda 2 = \mathbb{N}_{6}$ for CI CBOX 2N NEIGIX."

48. NG-OYN-OYCON AXN CEPA2T 2N OY26NGGTG. 2A2 AG N CON GAGKIM GYOPTH. NGXA4 GG 2PA1 N2HTH XG, " \uparrow NABWK TAGW MAYAAT GIANAXWPGI. AYW 2M NTPATMGN-2WB MN AAAY \uparrow NACEPA2T AYW NNAGOC NAAO N2HT." A4G1 AG GBOA, A4OYW2 MAYAA4 2N

^{38. (1)} κωνΣ κλ2- κολ2* Q κολΣ vb. intr. to strike, knock (at: e). (2) gram vb. tr. to shut (ΜΜΟ*). (3) τ.2γπο- ΜΟΝΗ (ἡ ὑπομονή) patience, endurance; he apparently made him wait a long time. (4) ογροτ, Q ροογτ vb. intr. to be happy, glad. (5) π. βΗΤ palm leaves (moistened and used for weaving). (6) κελεγε (κελεύω) to order, bid, command. (7) π. goλ bundle. (8) †-2Ηγ to benefit, profit; ρεσ†-2Ηγ beneficial; μπτρεσ†-2Ηγ beneficial; μπτρεσ†-2Ηγ beneficial. (9) ΝογβΤ ΝΟΒΤ* vb. tr. to weave (ΜΜΟ*). (10) τ. ΝΗΒΤΕ weaving, basketry. (11) πε. cπγληοι (τὸ σπήλαιον) cave. (12) Νλκλριος (μακάριος) blessed; used here as epithet of Apa Antonios; do not confuse with Apa Makarios. (13) π.λ.gλι multitude, large amount. (14) †-πι to kiss (e).

^{48. (1)} αναχωρέι (άναχωρέω) to retire, withdraw; to go

оуспуллюн. $2\overline{N}$ оусоп де динег-печкелюх 2 \overline{M} мооу, диоудги е пкдг, дую \overline{N} теумоу дискорк \overline{F} . 3 \overline{N} теречеми \overline{T} де, дичт \overline{T} , диоуоеп \overline{T} . 4 д-печгит де ег ероч, диегме же племом пет $^+$ иммач, дую пежди же, "егс гинте он $^+$ диджюрег маудат дую $^+$ беом \overline{T} . егидемк \overline{N} тооум е еемеете. \overline{C} -хргд гдр е игре ероч \overline{M} мл иги дую \overline{N} гоуо гупомине 5 е теонега \overline{M} \overline{M} пиоуте." дикточ де, дием е печма.

- 70. 1-оусом жі й песхних, 141 махюреї й теумоу, ечжю ймос же, "1 мр-оулмахюрітнс." 1 дусютй де йеї йгало, 148 мк, 141-тоотоу 2 ймоч, 140 хутречкюте 3 е йрі й месмну ечметімої, ечжю ймос же, "кю мії евол. 1 мр-оулмахюритно ім, 1221 лиг-оурюме й речр-мове хую й врре."
- 71. пежду де \overline{N} 61 \overline{N} 2 \overline{X} 00 же, "екфаннау буфире фим бченк бара етпе $2\overline{M}$ печоуф \overline{M} Мин \overline{M} Моч, беп-течоубрите, сок \overline{M} епесит \overline{M} Мау; с \overline{F} -новре гар нач ан."
- 102. $\[676-\] \]$ макартос мооф $\[67700\] \]$ оуовту $\[77700\] \]$ пките $\[77700\] \]$ его $\[77700\] \]$ $\[77700\]$ $\$

- 70. (1) π. ΑΝΑΧωριτης (ὁ ἀναχωρητής) anchorite; the status of a true anchorite was viewed as a very advanced stage of spiritual development. (2) †-τοοτ΄ Μπο΄ to lay hold of (suff. on τοοτ΄ is reflex.). (3) In causative sense: "they made him go around to the cells..."
- 102. (1) π .zexoc (tò ëxog) marsh. (2) twoyn as tr. vb. to carry (\overline{M} Mmo*). (3) twn \overline{M} T, Q tom \overline{M} T to meet, befall (e). (4) π .oz \overline{C} scythe. (5) x1 \overline{M} Mmo* \overline{N} son \overline{C} to ill-treat, harm,

into the desert and live as a hermit monk. (2) π.κελωλ jar, pitcher. (3) cκορκρ cκρκρ- cκρκωρ Q cκρκωρ to roll away (tr. or intr.). (4) ογωσπ ογεσπ- ογοσπ Q ογοσπ νb. tr. to break, smash (Μπο΄). (5) εγπομικε (ὑπομένω) to be patient (with, under: 6), submit to; to endure, last. (6) τ.βομοιλ (ή βοήθεια) help, aid, support.

ммок, же мй-бом ймог врок. 6 віс гните гар гов мім втеквіре ймооу фвіре ймооу го. Йток факинствув й генгооу; анок де метоуом в птира. 7 факр-оуон й ровіс 8 й генсоп; анок де метйкотй вивг. Оугов й оуот петекхравіт врог йгита. "
пеже-апа макартос же, "оу пе?" йточ де пежач же, "пекобъто пе. анок де метей-бом в объто вивг. бтве пат

124. 124000 Пбі апа гюрсінсі хв, "оутюювв¹ П оме² буфаннох бусйтв³ гатй пібро, ненагупоміне ан й оугооу й оуют. Ттерпосе⁴ ав фасмоун євох й өв й пюне. Таі те өб й проме 6-оуйтач ймау й печмебує й міткосмікон. ⁵ ніпосе⁶ ан гй өөте й пноуте. 64 фанеї 62 раі бумйтноб, ⁷ фачвох євох. 212 гар не йпірасмос й на-теіміне махіста 64 фооп гй тмите й йроме. Наноус ав бтре-проме соуен-печфі ймін ймоч, 67 бтречнот ав бвох й пегроф⁸ й тмйтноб. Нет тахрну ав гітй тпістіс 26 маткім 67 ооу не.

do violence to; to constrain; $xi \ \overline{N} \ 60N\overline{C} \ (xin60N\overline{C}) \ n.m.$ violence, physical constraint. The genitive (my) is objective here: "the constraint I feel from you." (6) $M\overline{N}$ -60M \overline{M} MOI 6POK I have no power over you. (7) 6 \overline{N} THP \overline{T} (not) at all. (8) POGIC vb. intr. to remain awake, keep watch (over: 6).

^{124. (1)} π.τωωκε, τ.τωωκε brick. (2) π.ομε, τ.ομε clay, mud. (3) τ.ς Ντε foundation. (4) τ.τερποςε(Ν) baked brick. (5) κοςμικός (κοσμικός) worldly, secular; μΝτκοςμικός worldliness. (6) πιςε πες (Τ) - πλςτ Q ποςε vb. tr. to bake, cook (Μμο). (7) In sense: "if he achieves a position of importance." (8) πε. 2 ροφ burden, responsibility.

^{141. (1)} ne. npwaction (to mpodotetov) suburbs, environs.

ΘΕ Ν ΟΥΆ ΕΒΟΆ 2Ν ΤΑΑΣΙς. 2 ΝΤΕΡΟΥΒΌΚ ΑΕ Ε2ΟΥΝ, ΆΥΦΑΝΑ, ΑΥ2ΜΟΟς. ΑΡΑΡΧΕΙ ΝΕΙ ΠΡΡΟ Ν 20Τ2 3 ΜΜΟΥ, ΕΥΆ ΜΗΟς ΧΕ, "ΝΕΝΕΙΟΤΕ ΕΤ 2Ν ΚΗΜΕ 7 -ΟΥ?" ΝΤΟΡ ΑΕ ΠΕΧΑΡ ΧΕ, "СЕФΑΝΑ ΤΗ-ΡΟΥ ΕΧΉ ΠΕΚΟΥΧΑΙ." ΑΥΌ ΑΡΧΟΟς ΝΑΡ ΕΤΡΕΡΟΥΘΉ Ν ΟΥΚΟΥΙ Ν ΟΕΙΚ. ΑΡΗ-ΟΥΦΗΝ Ν ΝΕ2 4 2 Ι 2 ΜΟΥ ΝΑΡ, ΑΡΟΥΘΉ. ΑΥΌ ΑΡΗ-ΟΥΦΗΜ Η ΜΟΟΥ ΝΑΡ, ΑΡΟΘΌ. ΠΕΧΑΡ ΑΕ ΝΑΡ ΝΕΙ ΠΡΡΟ ΧΕ, "ΚΟΟΟΥΝ ΧΕ ΑΝΓ-ΝΙΗ?" ΝΤΟΡ ΑΕ ΠΕΧΑΡ ΧΕ, "ΠΝΟΥΤΕ COOΥΝ ΜΜΟΚ." ΤΟΤΕ ΠΕΧΑΡ ΧΕ, "ΑΝΓ ΠΕ ΘΕΘΑΟCIOC ΠΡΡΟ," ΑΥΌ Ν ΤΕΎΝΟΥ ΑΡΠΑΣΤΤΗΝ ΧΕ ΤΕΤΝΟ Ν ΑΤΡΟΟΥΦ 6 2 Η ΠΕΙΚΟCHOC. 2 Ν ΟΥΜΕ ΝΧΙΝΤΑΥΧΠΟΙ 2 Ν ΤΗΝΤΡΡΟ ΗΠΙΜΕ2- 2 ΗΤ 7 Ν ΟΕΙΚ ΕΝΕ2 ΟΥΑΕ ΜΟΟΥ Ν ΘΕ Η ΠΟΟΥ, ΟΥΑΕ ΜΠΙΕΙΜΕ ΧΕ CE2ΟΛΕ 8 Ν ΤΕΙ2Ε ΧΙΝ ΠΕ2ΟΟΥ ΕΤ ΜΜΑΥ." ΑΡΑΡΧΕΙ Ν Η-ΕΟΟΥ ΝΑΡ ΝΕΙ ΠΡΡΟ. Π 2 ΧΛΟ ΑΕ ΑΡΤΦΟΥΝ, ΑΡΠΟΤ, ΑΡΚΤΟΡ ΟΝ ΕΚΗΜΕ.

⁽²⁾ sense here: the ranks of ordinary soldiers. (3) 2072 T 2672 T 2672 WT Q 2672 WT vb. tr. to examine, inquire into (MMO*). (4) n.N62 oil. (5) n6.2 MOY salt. (6) ATPOOY adj. carefree, free from anxieties. (7) M62-2HT NMO* to be sated, satisfied with. (8) 2006, Q 2006 vb. tr. to be sweet. pleasant.

^{175. (1)} peqp-2ws worker, doer; here in monkish sense: ascetic, practitioner. (2) Δφελλης (ἀφελής) simple. (3) gwq τ (gwb τ), Q goq τ (gob τ) vb. intr. to stumble, err. τ. μπτειλιωτης being uninformed; ίδιώτης non-professional, layman, uninformed person. (4) n.μλ here = the altar. (5) φγςι in fact, for real (φύσει by nature, naturally); τε. φγςις (ἡ φύσις) nature. (6) n.bloc (ὁ βίος) life. (7) bλλ-2ητ guileless, innocent; μπτελλ-2ητ guilelessness.

OVMNTATHOL. 8 AYW AYEL WAPON, AYXOOC NAN XE, "ANA, ANGWTM ETRE OYMAXE N ANICTON, XE A-OYA XOO4 XE NOEIK ETNXI MHO4 $\frac{9}{200}$ x6 NTO4 NAME AN RE ROWNA M REXC AXXA RESCMOT RE." $\pi_2 \overline{\chi}_{AO}$ as $\pi_6 \chi_A \chi_6$, "anok alke-mai." $\overline{\pi}_{TOOY}$ as aykwpu GPO4, GYXW MMOC XG, "MNWP. MNFTAXPOK 2M HAI, AHA, AAAA KATA OG GTGPG-TKAOONIKH 11 GKKNHCIA XW MMOC HICTGYG XG HOGIK GTNXI MMOU NTOU HE HEWMA M HEXE 2N OYME, AYW 2N OYCMOT AN, AYW RELEATED 12 RESCNOS RE $2\overline{N}$ OYME AYW $2\overline{N}$ OYCXYMA 3 AN. ANNA \overline{N} 96 \overline{N} TAPXH 6-A4X1 \overline{N} OYKA2 680N 2 \overline{M} 15 Aunacce \overline{M} mpwme kata teq21kwn \overline{N} ayw M \overline{N} -60M \overline{N} אבגג 18 אבגגע א אססכ אפ א פואסא א דפ און א אודס וויסן איז אין איז איז א דייטן א דער א דער א איז איז א דער א KATANYMITOC HE N ATTARON, TAL ON TE GE N HOELK NTARKOOC XE HAL HE HACOMA. THRICTEYE XE 2N OYME HAL HE HOWMA M $ne\overline{xc}$." nexa4 $\overline{N}61$ $n_2\overline{\lambda}\lambda0$ xe, "etet $\overline{N}T\overline{M}$ nige \overline{M} MO1 $ebo\lambda$ $2\overline{M}$ $\pi_2\omega_8$, \overline{n} +natwt an \overline{n}_2 ht." \overline{n} tooy as π_2 exay xe, "mapentwba2" H HNOYTE 2N TELZEBAWMAC ETBE HELMYCTHPION, AYW THRICTEYE TE HOOTE NAGOANT NAN 680A." π_2 TAO AE A4 ϕ T- π_0 AX6 6 ρ O4 2 π OYPAGE, AYW AGCORE M THOYTE EGAM MMOC XE, TRADEIC, NTOK ET

COOYN ME N GIO AN N ANICTOC KATA OYKAKIA 21 AAAA ME NNGIпадна 22 $_{2}$ $_{3}$ $_{1}$ оүм $_{1}$ тапістос м $_{1}$ оүм $_{1}$ татсооун, 6 $_{2}$ наі 6 $_{3}$ вод, TECHLO TO HEEC." NOTAD AG ON AYBOK & NEYPI, AYTOBAS H THOUTE, GYZW MMOC X6, "TO THEXO, GREEWATH GROW H THEIZ TAO H neimycthpion xe evenicteye ayw N可TH十一Oce 23 M Hevzice." **»-п**иочте де сфтя ерооч гі очсоп. Птере-обдюмые де жфк 680X, AYEL 6 TEKKAHCIA \overline{N} TKYPIAKH, 24 AY2MOOC \overline{M} TGOM \overline{N} T MAYAAY 21 $\langle OY \rangle OYPOM$ \overline{N} OYOT. NEPS-112 \overline{N} AO AS $2\overline{N}$ TSYMHTS. AYOYUN NGI NEYBAA ET 21 20YN, AYU NTEPOYKU E2PAI M HOEIK $\mathbf{E}\mathbf{X}\mathbf{N}$ TETPANYZA ET OYAAB, AGOYWNA: EBOA \mathbf{M} N $\mathbf{G}\mathbf{O}\mathbf{N}\mathbf{N}\mathbf{T}$ MAYAAY \mathbf{N} OE N OYOHPE KOYL, AYO NTEPE-HEHPECBYTEPOC COOYTN EBOA N TE4-61× 6 ×1 $\overline{\text{M}}$ noeik e nog $\overline{\text{q}}$, 26 eic oyaffexoc agei eboa 2 $\overline{\text{N}}$ $\overline{\mathbf{m}}$ пнує, є-оу $\overline{\mathbf{n}}$ -оубортє $\overline{\mathbf{n}}$ $\overline{\mathbf{n}}$ $\overline{\mathbf{n}}$ $\overline{\mathbf{n}}$ $\overline{\mathbf{n}}$ $\overline{\mathbf{n}}$ $\overline{\mathbf{n}}$ whee, λαπω2 29 M neachor 6 nnothpion. Ntepe-nenpecbytepoc AG GP-ROGIK \overline{N} FAACMA KAACMA, 30 NGPG-RAFFGAOC 2004 ROO \overline{M} пфире коут фим фим. Тую итероун и печочот 31 е жт евох 2 и NET OYALB, AGEL \overline{N} 61 $\overline{\Pi}$ 2 $\overline{\lambda}$ 80 \overline{N} Oykaacma nag egiho \overline{N} Choq, ayo NTEPERNAY, ARP-20TE, ARXI-GRAK EBOX XE, "+ TICTEYE, TXOEIC, X6 HOELK HE HEKCOMA AYOU HHOTHPION HE HEKCHOY." AYOU N TEYNOY A-RAY GT 2N TEYGIX P-OGIK KATA RECOY M RMYCTHPION. $\lambda 4 NOX \overline{4}$ 620YN 6 PW4, λYW $\lambda 4XI$ 646YXAPICTI 32 \overline{M} NXO6IC. $\mathsf{nexa}_{\mathsf{q}}$ naq Ne_{l} $\mathsf{N}_{\mathsf{2}}\mathsf{T}\mathsf{Xo}$ xe, "nnoyte cooyn n tempcic n Npome xe

T.26ΒΑΦΜΑς, ΘΈΑΦΜΑς (ἡ ἐβδομάς) week. (21) Τ.ΚΑΚΙΑ (ἡ κα-κία) evil, badness. (22) πλλΝΑ (πλανάω) to deceive, lead astray; middle: to err. (23) †-oce to suffer a loss (of: N). (24) Τ.ΚΥΡΙΑΚΗ (ἡ κυριακή) Sunday. (25) ογρωΝ var. of ΜρωΝ) pillow, seat. (26) πωω πεω- ποω Q πιω νb. tr. to divide (ΜΜο). (27) Τ.60ρΤε knife, sword. (28) ωωωτ ω εετ- ωλλτ Q ωλλτ νb. tr. to cut, slay (ΜΜο). (29) πω2Τ, πε2Τ-πλ2Τ Q πλ2Τ νb. tr. to pour (ΜΜο). (30) πε.ΚλλαΜΑ (τὸ κλάσμα) piece; repeated to express distributive: into pieces; cf. the following ωμμ ωμμ into small pieces. (31) †- Η π()ογοι to advance, proceed (suff. is reflex.). (32) εγχλριστι (εύχαριστέω) to give thanks.

мп-бом ммооу в оубм-дв бчоуют. 33 стве плі флятре-печсюмл фине м повік дую печсмоч й нрй й мет хі ммоч гй оупістіс. " дую дуф \overline{n} -гмот 34 йтй пмоуте гіхм пентляфюне, хе мпечкл- $n_2\overline{x}$ ло й рюме в \uparrow -осе м печгісе, дую дувюк й пфомйт є меурі \overline{z} й оурдфе.

240. A-ARA CAPARION NAY SYROPNH. 1 REXAS X6, "THEY DAPO M HNAY N POY26. CETOTE EBOX." AYO NTEPER CEI> NAC 620YN, N6XL4 NAC X6, "6w 6POI N OYKOYI, X6 OYNTAI-OYNOMOC MMAY, MANTAOKA 6803." NTOC A6 NEXAC X6, "KAXWC, NA61WT." \overline{n} to 4 agapx61 \overline{m} yaxx61 \overline{n} yaxx61 \overline{n} yaxx61 \overline{n} yaxx61 \overline{n} yaxx61 \overline{n} **ЄВОХ Й ПОСТАТОУ Й ЧАХМОС, АУЮ КАТА СОП Й КА-РОЧ ЄВОХ ФАЧ-**ELPE N MONNT N KXX- HAT. NTOC 2000C AC60 6CMAHA 21 HAZOY MMO4 2N OY2OTE MN OYCTWT. AMMOYN AE EBOX 649XHX 2APOC TAPECOYXAI, AYW A-NNOYTE CWTM EPO4. TEC21ME AE ACHA2TC 2A-PATOY \overline{N} NEGOVERHTE ECPINE ECXO \overline{M} MOC XE, "API-TALARH, 5 RA-GIOT. THA GTGKCOOYN XG THAOYXAL \overline{N}_2 HT \overline{q} XIT GMAY. \overline{N}_1 TA-THOUTE FAP THOONK WAPOI E HAI." AYW AMXITE BYZENEGTE H TAPOENOC. \overline{N} TELCONE, AYW MIPTANG-NA287 EXWC H GNTOAH, ANN N OG GTGCOYAQC HAPECAAC. KAAC 2M NXOEIC." AYW MNNCA 26NKOYI N 200Y NEXAC X6, "ANOK OYPE4F-NOBE. GIOYOO G OYOM N OYCON M MHNE." MNNCA KEOYOGIQ ON NEXAC X6, "GIOYOQ G OYUM N OYCON KATA CABBATON."8 MNNCOC ON NEXAC XE, "ENIAH AIP-2A2 N NOBE,

⁽³³⁾ oyor vb. intr. to be raw, green, fresh. $\mathfrak{g}\overline{\mathfrak{n}}$ -2MOT $\overline{\mathfrak{n}}\overline{\mathfrak{n}}$ to thank.

^{240. (1)} τ.πορημ (ἡ πόρνη) prostitute. (2) ΨΑλλει (ψάλλω) here: to recite psalter; πε.ΨΑλΜΟΣ (ὁ Ψαλμός) psalm. (3) κΧΧ-πΑΤ bow, genuflection; κωλ νb. tr. to bend, bow; τ.πΑΤ knee, leg. (4) πε.στωτ trembling. (5) ΑΡΙ-ΤΑΓΑΠΗ be charitable, do a kindness; τ.ΑΓΑΠΗ (ἡ άγαπή) love. (6) Ογεσιστε π παροσιος a convent. (7) π.ΝΑΣΕ yoke; here in monastic sense: imposed penance. Η (ἡ) or. (8) once a week. (9) σπιλΗ (ἐπειδή) because. since.

 $0π\overline{\tau}^{10}$ 620γη 6γρι λγω π6 \uparrow ηλογομ $\overline{\tau}$ πλι ηλι $2\overline{h}$ ογφογ $\overline{g}\overline{\tau}$ μ \overline{h} πλ τ 2ωβ \overline{h} 61χ." λγω λγειρε 21 ηλι, λγω λ¢ $\overline{\tau}$ 7 πλημ \overline{h} πηογτε, λ¢ \overline{h} πκοτ \overline{h} λε $2\overline{h}$ πλη ε \overline{t} πλοεις.

⁽¹⁰⁾ on t is for otn't, from wtπ.

TCODIA N COLOMON

Chapter 1

- (1) мере-танка посумн, мет криме \overline{M} пкаг. ари-пменуе \overline{M} пхонис $2\overline{N}$ оум \overline{N} таганос, \overline{N} пет \overline{N} оим \overline{N} сом $2\overline{N}$ оум \overline{N} тгалаоус \overline{N} те пет \overline{N} гант.
- (2) xe gayse epon $\overline{N}e$ 1 nete $\overline{N}e$ 6 eneipaze \overline{M} mon an. ganoyon \overline{z} as ebox \overline{N} nete $\overline{N}e$ 6 o \overline{N} atnaste epon an.
- (3) gape-imeeye far eeooy norxoy e inoyte, are tereom et oron \overline{z} ebox gacxiele-naeht.
- (4) WE MEPE-TCOOL TAP BOK 620YN GYYYXH 6C200Y, OYAG MGCOYO2 $2\overline{N}$ COMA \overline{P} PG4 \overline{P} -NOBG.
- (5) $\pi e \pi N \lambda$ гар ет оудав \overline{N} тсофід фачпот євох \overline{N} кроч, аую фачоує \overline{N} \overline{M} мокмек \overline{N} \overline{N} наθит, аую фачх $\pi 1 e^- \pi x 1 n 6 o n \overline{c}$ ечодиє 1.
- (6) ОУМАБІ-РЮМЕ ГАР ПЕ ПЕП \overline{N} \overline{N} ТСОФІА, АУМ М \overline{M} МАТМАІЄ-ПЖІ-ОУА АМ $2\overline{N}$ МЕЧСПОТОУ; ЖЕ ПМОЎТЕ ПЕ ПМ \overline{N} ТРЕ \overline{N} МЕЧЕЛОТЕ, АУМ ПЕТ МОЎФ \overline{M} МАМЕ \overline{M} ПЕЧ2НТ, АУМ ПЕТ СФТ \overline{M} Е ПЕЧААС.
- (7) же пепых м пхобіс хамег-тоікоуменн, хую пет фюп м птира чсооун м пеугрооу.

I. (1) μρίνω to judge. ἀπλοῦς adj. simple, frank, sincere. (2) Νλ2ΤΕ, Q Ν̄2ΟΥΤ νb. tr. to believe, trust (ε); λΤ-Νλ2ΤΕ adj. unbelieving. (3) πωρΣ περΣ- πορχ' Q πορΣ νb. tr. to divide, separate (ν̄Μο'; from: ε). (5) πε.κρο4 deceit, guile. ογε, Q ογμγ νb. intr. to be distant (from: ε, ν̄Μο'), remain aloof from. (6) πε. cποτογ lip(s), shore, edge. Gλωτ (pl. 6λοΤΕ, 6λΟΟΤΕ) π.m.f. kidney; here in OT sense as seat of emotions. Μογω̄Τ ΜΕΦΤ- ΜΟΦΤ' Q ΜΟΦΤ νb. tr. to to examine, search out (ν̄Μο'). (7) πτηρ̄Ψ the universe, everything.

- (8) GTBG ПАІ М \overline{N} -ХАХҮ НА2 $\overline{\omega}$ П GY $\underline{\omega}$ XG 2 \overline{N} OYXINGON \overline{C} , OYAG N \overline{q} NA \overline{p} -BOX AN G TGKPICIC GT \overline{N} NHY.
- (9) CENAGM-HOINE FAP M HOOMNE M HACEBHC, AYOU HACOUTH E NEUDAME 6 HOYON \overline{z} GBOA N HEUAHOMIA.
- (10) WE HMANE \overline{M} HEYERS GAYCOTH E SOB NIM, AYO HESPOOY \overline{N} NEKPMPM NASOH AN.
- (11) 2 APG 2 GG GPWTN G HEKPNPN GT GOYGIT, 2 AYW + CO G HETNALC GBOA 2 N TKATANANIA; x G MN-OYWAXG GHYOYGIT NA2WN. OYTANPO GCXI-GOA WACTAKG-TGYYH.
- (12) $\overline{\text{M}}$ $\overline{\text{M}}$
- (14) ПТА 4 СОНТОУ ГАР ТНРОУ 6 ТР 6 У 6 М ВОЛ АУМ 6 ТР 6 У ОУЖА 1 ПО 1 ПСШИТ И ПКОСМОС. НИЙ-ПА 2 Р 6 И МОУ 2 РА 1 П2 НТОУ, ОУА 6 МПТ 6 Р 0 П АМПТ 6 2 1 ЖИ ПКА 2.
- $\{(15)$ TAIKAIOCYNH FAP OYATMOY T6.
- (16) $\overline{\text{N}}$ acebhc ae $2\overline{\text{N}}$ neyeix m $\overline{\text{N}}$ neyeaxe aycotn $\overline{\text{q}}$ nay;

⁽⁸⁾ F-sol e to avoid, escape. Wnhy for nhy. (9) going vb. intr. to take counsel (concerning: e); as n.m. counsel. άσεβής adj. ungodly, impious. ἡ ἀνομία lawlessness. (10) π.κω2 envy, jealousy; vb. intr. to be envious, jealous, zealous (for: e). (11) †-co e to restrain; to refrain from. x1-sol to tell a lie. (12) ἡ πλάνη error, erring. (14) cont cut- cont Q cont vb. tr. to create, found (mmo); as n.m. creation, creature. gl sol adv. forever, for good. πλ2ρε π moy poison. λημίτε Hades, Hell. (15) Verse 15 is intrusive and incomplete. Omit.

АУТАЛЯ МАУ \overline{N} ФВНР, АУВФА 6ВОА, АУСМІМЄ \overline{N} ОУДІЛӨНКН МЕМЛЯ, ХВ СЕЙПФА \overline{N} ТМЕРІС \overline{M} ПЕТ \overline{M} МАУ.

Chapter II

The Reasoning of the Wicked

- (1) $\lambda\gamma \times 000$ гар, $\theta \lambda\gamma \times 66\gamma \theta$ граі \overline{N} гитоу $2\overline{N}$ оусооут \overline{N} ам, $x\theta$ оукоуі п θ п θ п λ 2 θ , θ ч θ проме, θ ч \overline{N} п \overline{N} оуа θ \overline{N} тоу \overline{N} θ \overline{N} θ \overline{N} θ \overline{N} θ \overline{N} θ \overline{N} \overline
- (2) x6 Ντληφωπ6 6 ππ6τ φογ61τ.
 ΜΝΝΟΦΟ 6ΝΝΑΡ-Θ6 Ν Ν6Τ6 Μπογφωπ6,
 x6 ΟΥΚΑΠΝΟΟ π6 πΝ146 6τ 2Ν φλητη,
 λγω ογήκ π6 πφλχε 6τ κιμ 2Μ π6ν2μτ.
- (4) \overline{N} сер-пюв \overline{g} \overline{M} пенран $2\overline{M}$ пеноуовіg, \overline{N} тет \overline{M} -ладу вр-пмебує \overline{N} ненавнує, дую пенаге наоувіне \overline{N} об \overline{N} оуклооле, дую чилхююре євол \overline{N} об \overline{N} оуніче \overline{G} -ачвюл євол \overline{Z} пактін \overline{M} при,

. РОЖА ФОЧ 3 ЭММ 2 РОТ - «Э ФУА

(5) OYZAGIBEC 6-ACOYGING NG NGNOYOGIW,

⁽¹⁶⁾ CHING CHN- CHNT' Q CHONT vb. tr. to establish, set up ($\overline{\text{M}}\text{Mo}$). $\hat{\eta}$ µEP/G portion, share; party, faction.

II. (1) 2Ν ογοοογτη λη incorrectly, not rightly.

Κ λγημ = Ν λγημ; ἡ λύπη grief, pain. (2) ϝ-φε Ν to become like. π.νισε breath. φλητ nose. π. +κ spark. (3) τ.

Κῶβες (glowing) coal. ὁ, ἡ ἀήρ air, atmosphere. (4) τε.

Κλοολε cloud. π.λκτιη (ἡ ἀκτίς, -ῖνος) ray, beam. τ. 2Μμε heat. 2ροφ, Q 2ορφ vb. intr. to become heavy, difficult. (5) τ. 2λιβες shadow, shade.

- (22) $\lambda \gamma \omega$ мпоусоуй-мустиріон и пноутв, оуде ипоука-2тну є пвеке и талкалосуни; ипоупістеує є птало и нечухи и нет оудав.
- (23) x_6 mnoyte againt \overline{M} mpone symutattako, ayo agamiog $2\overline{N}$ bikon \overline{M} negeine.
- (24) 2H REPRONOC AS H HALLBOADC A-THOY SI SZOYN S TKOCHOC.
- (25) сепетрате де ммоч мет тмеріс м пет ммау.

Chapter V

The Remorse of the Wicked at the Judgement

- (1) тоте панканос нал зерат \overline{q} очно \overline{m} пар знсна \overline{e} наф \overline{w} пемто евох \overline{n} нентачение \overline{m} ментачанет \overline{n} неч знее.
- (2) CENANAY, \overline{N} CEGOTOPT \overline{P} \overline{N} OYZOTE ECNAGT, \overline{N} CERWOO \overline{C} \overline{E} X \overline{N} TMOELZE \overline{M} REPOYXAL,
- (3) \overline{N} CEXOOC 2PAI \overline{N} 2HTOY, GYMGTANOI

 AYW GYAG-A2OM GTBG ПАФХ \overline{Z} \overline{M} ПGYП \overline{N} \overline{X} ,

 XG "ПАІ ПЕНЕНСФВЕ \overline{N} СФЧ \overline{M} ПІОУОСІФ,

 GЧGООП НАН \overline{M} ПАРАВОАН \overline{N} НОБИGG \overline{N} НІАӨНТ,
- (4) ENOT \overline{M} REGAZE GYALBE, AYO REGMOY GYCOO.

⁽²²⁾ κλ-2τη ε to set one's mind on/to. π. в εκε reward, pay. (24) ὁ φθόνος ill-will, jealousy. (25) τ. μερις is taken as collective: "those who belong to that one." πειράζω in the sense "to experience."

V. (1) ἡ παρρησία freedom, openness; 2N ογπλγ2μςιλ openly, publicly. άθετέω to disregard. (2) πωσζ πεωζ΄ ποως Q ποως vb. tr. to amaze (Μμο΄); intr. to be amazed (at: exN). τ.μοει2ε wonder, marvel. (3) λω-λ2ομ vb. intr. to sigh; as n.m. sigh. π.λωχ anguish, oppression. cωδε vb. tr. to mock, ridicule (Μμο΄, Ναλ). πλρλδολμ in sense: model, exemplar. (4) λ1δε as n.m. madness.

- (5) \overline{N} \overline{N} 26 \overline{N} \overline{N}
- (6) $_{6616}$ $_{\overline{N}}$ $_{\overline{N}}$
- (7) $\lambda NMOY2$ \overline{N} $\lambda NOM1\lambda$ 21 $T\lambda KO$ \overline{N} N6N2100Y6. $\lambda NBWK$ 21 $T\overline{N}$ $\overline{N}X\lambda$ 16 6M6YMOOG6 \overline{N} 2HTOY; T621H $\lambda 6$ \overline{M} $\overline{N}XO61C$ $\overline{M}\overline{N}\overline{N}\overline{N}\overline{O}YWN\overline{C}$.
- (8) $\overline{\text{NTAC}}$ -OY $\overline{\text{MMON N}}$ OY $\overline{\text{NGI}}$ $\overline{\text{NMNTXACI-2HT}}$ H THNTPMMAO HN THNTBABG-PWHG $\overline{\text{NTAC}}$ -OY NAN?
- (9) λ -NH THPOY OYGING \overline{N} GE \overline{N} OY2 λ IBGC, λ YW \overline{N} GE \overline{N} <OY>OYW G- λ 4 $\Pi\lambda$ 7 λ FG,
- (10) н й өб й оухол басбар гй оуговим й мооу б-мй-өб й бй-мбатьбсб н тбгін й пбатоп гй йгобім.
- (11) H N GE N OY2AAHT E-A42WA GBOA,

 {E-MEYEN-MAGIN M NE42WA GBOA}

 E4210YE N NE4TN2 E NAHP ET ACWOY,

 E4 NW2 MMO4 N GONC 2M NOYOE1,

 E4KIM N NE4TN2, E42HA,

 MNNCWC E-MEYEN-MAGIN M NE42WA GBOA.
- (12) H \overline{N} 96 \overline{N} OYCOTE 6-AYNOX \overline{Y} 6 HCOOYT \overline{N} , 6-A4H62-HAHP, \overline{N} TEYNOY ON A4T066, 6-MEYCOY \overline{N} -TE421H.

⁽⁶⁾ πσιρς, Q πορς vb. intr. to come forth; to shine (of un). (8) †-ογ ΜΜΟΝ Ν ογ is not clear; read perhaps †-ογ πλη as at end of verse. βλεσ-ρωμό boaster; μντβλεσ-ρωμό boastfulness. (9) π.ογω news, report. (10) σση vb. intr. to sail. π.2οσιμ wave. τ.τλοσο foot-print, track, trace. π.τοπ keel. (11) λσωογ (Q of λσλι) vb. intr. to be light, wift. π.ογοσι rush, swift movement. πω2 in sense: to split, cleave. (12) π.σοτε arrow. ε πσοογτί straight ahead), on target. τωσε: i.e. the air joins (or closes

- λγω Μην-κτο φοοη η πενμού; xε λγτωσεε ερφου, λγω νε-λλλυ κοτ $\overline{4}$.
- (6) динетти бе ититстои и идганом ет фооп, итихрю и тектістс 2и оубенн и не и оумитврре.
- (7) МАРЙТСІОН Й НРЙ 6-НАНОУЧ 21 СТІ-НОУЧ6, АУФ МПРТРЕУСЛАТИ Й61 ЙКАРПОС Й ПАНР.
- (8) MAPNT EXWN N 2NKAOM N OYPT EMMATOY206B,
- (9) МПРТРЕ-ХАХУ МНОМ ФОПЕ М ПВОХ \overline{N} ИЕММІТФИХ. МАРЙКА-СУМВОУХН \overline{N} ОУМОЧ $2\overline{M}$ МА ИІМ, ЖЕ ТАІ ТЕ ТЕММЕРІС ХУФ ПЕМКХНРОС.
- (10) OY2HKE \overline{N} AIKAIOC MAP \overline{N} XIT \overline{N} GONG.

 MRPTPEN+CO 6 TEXHPA,

 OYAG \overline{M} RPTPENGING 2HTOY \overline{N} NECKIM \overline{N} OY2 \overline{N} AO \overline{N} NO6 \overline{N} A26.
- (11) Mape-tensom game nan \overline{N} nomoc \overline{N} aikaiocynh; th \overline{N} teub rap egayxnioc 2 ω c atgay.
- (12) ΜΑΡΠΘωΡΘ Θ ΠΑΙΚΑΙΟΟ,

 ΧΕ «ΜΟΚΣ Θ Ρ-[ΧΡΗΟΤΟΟ] ΝΑΝ,

 ΑΥΨ «† ΟΥΒΕ ΝΕΝ2ΒΗΥΕ.

 «ΝΟΘΝΕΕ ΜΜΟΝ Ν ΝΕΝΝΟΒΕ 21ΤΝ ΠΝΟΜΟΟ,

 ΑΥΨ «ΟΥΨΝΣ ΘΒΟΑ Ν ΝΕΝΝΟΒΕ 21ΤΝ ΤΕΟΒΦ.

⁽⁵⁾ τωωδε τοοδ Q τοοδε νb. tr. to set a seal (on: ΜΜΟ΄, ερΝ). (6) χράομαι to use. ἡ κτίσις the world, creation. (7) cti-νογαε perfume, incense (cf. ctoi). π.ληρ is probably Gk. error for έαρ springtime. (8) ογρτ rose. 2ω6Ε 266Ε-206Ε Q 206Ε νb. tr. and intr. to wither. (9) μΝτωνλ profligacy. cγμδογλη prob. for cγμδολον τὸ σύμβολον mark, token. ογνοα νb. intr. to rejoice; n.m. joy. ὁ κλήρος portion, share, inheritance. (10) πε. ckim gray hair. (11) μΝτεωδ weakness; εωδ adj. weak. (12) εωρδ, Q εορδ νb. to hunt, waylay, ambush (ε). ρ-χρηστός useful. beneficial.

- (13) чх ω миос х ε †сооүн й пиоүт ε , хү ω ч ε Ір ε миоч й днр ε й пхо ε Іс.
- (14) gargore nan eyxfio \overline{N} nenmeeye, $420P\overline{g}$ nan e nay epo4,
- (15) $xe \ \overline{m}$ nearioc eine an \overline{m} natoyon nim, ayw neariooye cewore.
- (16) СИНП ПТООТ \overline{q} С 2 СИХООУТ, $\Delta \gamma \omega$ ЧС Δz Н γ СВОХ \overline{N} ИСИz100 γ С \overline{N} ӨС \overline{N} И1 Δ К Δ Ө Δ РС1 Δ . ЧМАК Δ Р1ZС \overline{N} Ө Δ Н \overline{N} \overline{N} Δ 1K Δ 10C, $\Delta \gamma \omega$ Ч ω 0 γ 9 ω 0 γ \overline{M} МОЧ ZС "П Δ 1 ω Т ПС ПИО γ ТС."
- (17) MAPNNAY XE 2 \overline{M} ME NE NE \overline{M} 9 \overline{M} X \overline{M} 0 \overline{M} T \overline{M} 1 \overline{M} 1 \overline{M} 1 \overline{M} 2 \overline{M} 5 \overline{M} 5 \overline{M} 5 \overline{M} 5 \overline{M} 5 \overline{M} 6 \overline{M} 7 \overline{M} 6 \overline{M} 7 \overline{M} 6 \overline{M} 9 \overline{M} 7 \overline{M} 9 $\overline{M$
- (18) едже пліклюс глу пе прире й пиоуте, чилфопа ероч, нападиеч йтоотоу й нет † оувич.
- (19) MAPNZETAZE MMOH ZN ZENGWO MN ZENBACANOC, ZEKAC ENEGEIME E TEHMNTZAK, AYW NTNAOKIMAZE N TEHMNTZAPOZZHT.
- (20) мар \overline{n} тбаеточ $2\overline{n}$ оумоу еченф; сенаб \overline{m} -печфине гар ката нечфаже.
- (21) NAI AYMGEYE GPOOY AYW AYCWP \overline{M} ; A-TEYKAKIA FAP TWM \overline{M} NGY2HT.

⁽¹⁴⁾ πιορφ: "he is hard for us to look at (i.e. countenance)."
(15) εινε νb. tr. to resemble, be like (ΜΜΟ΄); as n.m.
likeness, aspect. (16) χοογτ adj. base, rejected. ἡ άκαθαρσία uncleanness; νι- §30.8. μακαρίζω to bless, deem
blessed. 9ογφογ νb. intr. to brag, boast. (18) νογιπ
νειπ- νλιπ' Q νλιπ νb. tr. to save, rescue (ΜΜΟ΄). (19)
εστλικ έτάζω to examine, test. 9ω9 νb. tr. to twist; here
apparently as n. torture. ἡ βάσανος torture, anguish.
δοκιμάζω to prove, test. (20) τελειο τελειε- τελειο' Q
τελεικγ νb. tr. to condemn, disgrace (ΜΜΟ΄).

- (13) TAI TE BE 2000 ON 6-AYXTON ANOXN; MINZE GYMAEIN N APETH 6 OYON2 $\overline{4}$ GBOA. 2PAI AE $2\overline{N}$ TENKAKIA [...
- (14) же велпіс \overline{M} пасевно [о \overline{N}] ве \overline{N} оуфгіб ере-птну ч[і \overline{M} Ммоч],

λγω \overline{N} θ6 \overline{N} ΟΥ2λλΟΥC 64900M6, 6-λΥΘλΟ4 68Ολ 21 \overline{N} ΟΥ2λΤΗΥ,

Н \overline{N} ӨӨ \overline{N} ОҮКАПИОС \overline{G} -АҮЗАТНҮ ВОА \overline{q} \overline{G} ВОА, \overline{N} ӨӨ \overline{M} П \overline{p} -ПМӨӨҮӨ \overline{N} ОҮР \overline{M} МӨОӨ ІАӨ \overline{N} ОҮЗООҮ ОҮФТ \overline{G} -АЧПАРАГЕ.

Chapter VII

The Attributes of Wisdom

- (22) ОУМ-ОУНИХ ГАР Й2НТС 640УАХВ, Й Р64МОІ, И ОУСМОТ Й ОУШТ, Й АТ6-СМОТ, 64АСШОУ, \overline{P} Р64 \overline{P} -2 \overline{B} 8, 646 \overline{M} -60М, 640 \overline{N} АТТШАЙ, \overline{N} САВ6, \overline{N} АТМОВ6, \overline{M} МАІ-АГАӨОИ, 64ТО \overline{P} 7, 6-М6 \overline{Y} 9 \overline{M} 4 \overline{M} 2 \overline{T} 6 \overline{M} 604, \overline{P} 7 Р64 \overline{P} 7—П6Т ИАМОУ4,
- (23) \vec{M} маі-рюме, ечтахрну, ечор \vec{X} , ечо \vec{N} атрооуф, ечем-вом е гов и ім, ечем \vec{D} ех \vec{M} птнр \vec{V} , ечхоте гіти непих тнроу ет оуалв, \vec{V} речної, ет фооме.

VII. (22) ATG-CMOT adj. of various sorts. τωλ νb. tr. to defile, pollute; ATTωλν unpolluted. τωρ , Q τορ νb. intr. to become sober, alert. (23) χωτε χετ- χοτ νb. tr. to penetrate, pierce, permeate.

up) after the passage of the arrow. (13) ωxπ exπ- oxn' vb. tr. to destroy; intr. to perish, cease to be. ἡ ἀρετή goodness, virtue. The end of the verse is missing: "In wickedness [we were utterly consumed.]" (14) ἡ ἐλπίς hope. ne.y216 dust. n.2λλογc spiderweb. goome Q to be light, fine. Θλο vb. tr. to cause to fly, chase away. τ.2λτηγ whirlwind. ρΜπ6οειλε lodger; εοειλε vb. intr. to dwell, visit, sojourn.

- (24) $TCO\phi$ і дар кім беоуб мет кім тнроу; схоте дую смну бвод 2 ітм птнр \overline{q} стве пест \overline{g} во.
- (25) ECNHY PAP EBOX 2 ITN TEOM W HNOYTE, $\lambda \gamma \omega$ EBOX 2W HEOOY ET OYAAB NTE HHANTOKPATWP. ETBE HAI MEPE-XAAY E4XA2W TWWNT EPOC.
- (26) OYEINE TAP TE NTE NOYOEIN \overline{N} \underline{y} \underline{x} \underline{y} \underline{y}
- (27) с-оубі де те, всбм-бом є гов мім; луш всбеєт гарігарос, всеїре м птиря м врре; луш ката гемба свик бгоум є мещухи м мет оуалв, ссіре ммооу \overline{n} овир є пмоуте луш мпрофитис.
- (28) \overline{N} THOOTE FAP MG \overline{X} NALY AN GIMHTI THET OYH 2 $2\overline{N}$ TCOOLA.
- (29) TAL FAP NECWC 620Y6 RPH, AYW 620Y6 RECMINE \overline{N} NCIOY THPOY. GYWANTNTWN \overline{C} 6 ROYO6IN, CNA \overline{P} -WOP $\overline{\Pi}$ 6PO4:
- (30) пат мен гар фаре-теуфн ет е печна; тсофіа де мере-ткакіа бмеом ерос.

Chapter IX

(Solomon's) Prayer for Wisdom

(1) THOOTE \overline{N} NACIOTE, TROCIC \overline{M} THA, TENTLATAMETER THE \overline{Q} \overline{M} THE \overline{Q}

⁽²⁵⁾ δ παντοκράτωρ the Almighty. xω2M x62M- xλ2M Q xλ2M yb. tr. to defile, pollute (MMO*); intr. to become defiled. (26) τ.61λλ mirror. (27) 2λρ12λρο* intensive pron. (she) alone, by (her)self. M sppe adv. anew. κλτλ renea from generation to generation. (29) ne.cmine here prob. in ense: constellations, order. (30) 61 6 n(*)mλ to succeed, take place of.

- (2) $\lambda KC\overline{N}T \Pi P \omega M \in 2\overline{N}$ TEKCO $\phi I \lambda$, $\chi_{GK} = \chi_{GK} = \chi_{GK}$
- (3) NTP-2MM6 M ПКОСНОС 2N ОУТВВО MN ОУДІКЛІОСУНН, NTKPIN6 N ОУ2ЛП 2M ПСООУТН N ТЕЧЧУХН,
- (4) MA NAI \overline{N} TCO ϕ IA, TAI GT A2GPAT \overline{C} G NGKOPONOC, NFT \overline{M} T \overline{C} TOGI GBOA $2\overline{N}$ NGK $2\overline{M}$ 2AA,
- (6) KAN OYTEREIOC HE OYA $2\overline{N}$ NOHPE N Frome, $E-M\overline{N}TAH$ MMAY N TEKCOPIA, GYNAONY GYXAAY.
- (7) \overline{N} ток аксотп \overline{T} бурро \overline{M} пекадос, аую оурбч \uparrow -гап \overline{N} некфире \overline{M} некфебре.
- (8) акхоос ϵ кшт нак \overline{N} оурпе $2\overline{N}$ пектооу ϵ т оуахв, ауш оувусіастнріон $2\overline{N}$ тполіс \overline{M} пекма \overline{N} дшпе, пеіне \overline{N} текскнин ϵ т оуахв ентакс $\overline{\epsilon}$ тшт $\overline{\Upsilon}$ хін \overline{N} дор \overline{n} .
- (9) $\lambda \gamma \omega$ ере-тсофіл мемлк, тет сооум й мекавнуе, $\lambda \gamma \omega$ меслаератс пе йтеректаміе-пкосмос, ессооум же оу пет \overline{p} -лилк й пекйто евол, $\lambda \gamma \omega$ оу пет соутом $2\overline{N}$ мекемтолн.
- (10) МАТЙНООУС БВОА $2\overline{N}$ НЕКПНУЕ ЕТ ОУАЛВ АУШ БВОА $2\overline{M}$ ПЕӨРОНОС \overline{M} ПЕКЕООУ, ЖЕКАС ЕСЕЩ \overline{N} —2ICE N \overline{M} MAI, EC2ATHI,

IX. (3) F-2MM6 to steer, guide (MMO'). n.2λn judgement.

(4) τστο τστο- τστο' Q τστην νb. tr. to bring back (MMO');

+ 680λ: to reject. (5) άσθενής weak, without strength.

π.λ2ε lifetime. gλλτ Q to be lacking (in: MMO', 2N); "I

am intellectually lacking in (knowledge of) judgement and

law." (6) τέλειος perfect, complete; perhaps read εγτ. for

ογτ. (8) xooc ε + Inf. to order, command (that something

be done). ή σμηνή tent, "tabernacle." (10) φπ-21cε MN to

ПТА 6 IM 6 X6 ОУ ПЕТ ФИП ПНА 2 РАК.

- (11) \overline{C} COOYN Γ AP \overline{N} TOC \overline{N} 208 NIM, AYO CNO! \overline{M} MOOY, AYO CNAXI-MOGIT 2HT $2\overline{N}$ NA28HYG $2\overline{N}$ OYM \overline{N} TP \overline{N} N \overline{C} 2APG2 GPO! $2\overline{M}$ \overline{C} COOY,
- (12) $\overline{\text{NTG-NA2BHYG}}$ филе буфил, $\overline{\text{NYW}}$ † $\overline{\text{NAKPING}}$ $\overline{\text{M}}$ пеклаос $2\overline{\text{N}}$ оудіка іосуни, $\overline{\text{NTAфWNG}}$ $\overline{\text{GIMTO}}$ $\overline{\text{N}}$ неөронос $\overline{\text{M}}$ пасішт.
- (13) NIM FAP \overline{p} pome net nacoy \overline{n} -ngoxne \overline{n} nnoyte? H NIM net naeime xe oy netepe-nxoeic oyag \overline{q} ?
- (14) MMOKMEK FAP N Prome 6008, AYO CEROOSE NEI NEYMEEYE.
- (15) Howen the heatako grapog ex \overline{n} teyyxh, hyw ha \overline{n} gwhe \overline{n} te hkle grap-kake e ϕ ht \overline{n} alipooyg.
- (16) MOFIC ENTONTH H NET 21XH HKL2; ENGINE <N> NET 2 NENGIX 2 H OY21CE. NET 2 H MHYE AE NIM HENTA42ET2WTOY?
- (17) н или пеитачелие е пекфожие \overline{N} Савна же \overline{N} Ток ак \dagger \overline{N} тсофла, акт \overline{N} Инооү \overline{M} текп \overline{N} А ет оуаав евоа $2\overline{M}$ пжлсе?
- (18) TALL THE OF WTAYCOOYTN WELL NG2 100YF N NET 21XM NKA2, A-NPWME CEO E NET \overline{F} -ANAK, AYW AYOYXAL $2\overline{N}$ TCOOPLA.

labor, toil with. 2λτΗ' = 2λ2ΤΗ'. (11) NOI ΜΉΝΟ' to understand. xI-MOGIT 2ΗΤ' to guide; π.ΜΟGIT road, path. (14) GOOB Q to be weak, feeble. λΟΟΘΕ Q to be in a state of collapse or decay. (15) πρεστακο is in apposition to π. COMA. P-κακε ε to darken. σαι-ροογφ adj. full of cares. (16) μόγις adv. with great difficulty, hardly, scarcely. ΤΟΝΤΝ ΤΝΤΝ- ΤΝΤΟΝ' Vb. tr. to speculate about (ΜΉΝΟ', ε). (18) CBO ε to learn.

The Life of Joseph the Carpenter

V

(1) $2P\lambda$ 1 AG $2\overline{N}$ TMG $2M\overline{N}$ TA4TG \overline{N} POMTG \overline{M} MAP1A TA-MALY A1G1 $2\overline{M}$ MAOYWW, A1OYWZ \overline{N} ZHTC KATA HETG $2N\lambda$ 1, G-ANOK HG HGT \overline{N} WHAZ. (2) AYW \overline{N} TGPGC \overline{P} -WOM \overline{N} T \overline{N} GBOT \overline{N} W, A-HATKPO4 1WCH φ , HAMGPIT \overline{N} 1WT, G1 G2OYN 2WW4 $2\overline{N}$ NGMA \overline{N} KWT, A4GGN-TAMALY \overline{N} TACOYWN \overline{Z} GBOA XG CGGT. A4GP-2OTG AYW A4WTOPT \overline{P} . A4OYWW \overline{G} NAXGC GBOA \overline{N} X1OYG. (3) AYW GBOA 2GN TAYHH, A4- \overline{N} KKOT \overline{K} , \overline{M} HG4OYGM-AALY \overline{G} HTHP \overline{A} \overline{G} TGPOY2G \overline{G} \overline{M} MOOY.

VI

(1) $2\overline{N}$ тпафе де \overline{N} теуфн еіс пархаггелос гавріна ачвик фароч $2\overline{N}$ оурасоу $21\overline{N}$ техоусіа \overline{M} паішт \overline{N} агафос. пехач нач же, "ішснф, пфнре \overline{N} дауеіа, \overline{M} пр \overline{p} -20те. Жі \overline{M} маріа текс2іме; петеснахпоч гар оуаав. (2) ауш екемоуте е печран же \overline{IC} . \overline{N} точ пет намооне \overline{M} печалос $2\overline{N}$ оубершв \overline{M} пеніпе." (3) ачтшоун ае \overline{N} 61 ішснф евох $2\overline{M}$ п2іннв, ачеїре ката фе \overline{N} Тач2шн етоот $\overline{4}$ \overline{N} 61 паггелос \overline{M} пхоеїс. ач2аре2 е тпарфенос ет оуаав є20ун е печні.

VII

(1) MNNCA NAI AYAOFMA 61 CBOX 21TM $\Pi \overline{p}$ po ayroyctoc $\epsilon \tau p \epsilon$ -

V. (1) KATA NETG 2NAI according to my desire; see Glos. sub 2NG-. (2) $2\overline{N}$ NGMA for $2\overline{N}$ MMA (\overline{N} KWT) from the workshops. \overline{N} TACOYWN $\overline{2}$ is presumably a relative form instead of an expected circumstantial: "he found that my mother had become evident as being pregnant." NAXEC = NOX \overline{C} .

⁽³⁾ Μποογ is an error for Μπλγ.

VI. т.пафе half; т.пафе й теуфн midnight. (2) мооне мене- маноу vb. tr. to pasture, shepherd (Ммо). п.берфв (рl. бероов) rod, staff. пеніпе = веніпе ігоп. (3) гіннв vb. intr. to sleep; as n.m. sleep.

ТОІКОУМЕНН ТНРС С221С ЙСА МЕСТМЕ. (2) АЧТШОУМ ДЕ ЙЕІ ІШС СНФ, ПА-ТМЙТ2 \overline{X} ДО ЕТ МАМОУС, АЧХІ Й ТПАРВЕМОС Й СЕМИН, АЧСІМЕ ЙМОС Е2РАІ Е ПЕЧНІ ЙМІМ ЙМОЧ Е ВНОЛЕЕМ, Е-ДС2 ω М Е2ОУМ Е МІСЕ. АЧАПОГРАФН Й ПЕЧРАМ 2 λ TЙ МЕГРАМАТЕУС Й ВНОЛЕЕМ, СПОСНФ ПФИРЕ Й ІДКОВ МЙ МАРІА ТЕЧС2 χ ІМЕ МЙ χ С ПЕУФИРЕ, С-МЕЄВОЛ МЕ χ Й ПНІ Й АДУЕІД, ПА-ТЕФУЛН Й ЕІОУДА. (3) χ С МАРІА ТАМАДУ МІСЕ ЙМОІ Й2ОУМ Е ПМА Й БОІЛЕ Й ВНОЛЕЕМ 21-ТОУФЧ Й ПТАФОС Й 2 χ РХИХ ТЕС2 χ ІМЕ Й ІДКОВ ППАТРІДРХИС, ПСІФТ Й ІФСНФ МЙ ВЕМІДМІМ.

VIII

(1) a-ncatanac cymboyagye $2\overline{M}$ n2HT \overline{N} 2HPWAHC NNO6, NIWT \overline{N} apxH[xaoc ... (end of fragment I).

XIV

(1) асфиле де, мтеречхе-илі, дчтюсуи, дчеї е печні нахарею, тполіс етечоунг \overline{N}_2 нт \overline{C} , дую \overline{N} тегге дчхто е проие стечилмоу \overline{N}_2 нт \overline{q} прос пет кн егрлі \overline{N} роме иім. (2) дую еїс гните мере-печфоме гор \overline{g} емате \overline{N} гоуо е соп иім мтачфоме хімтаухпоч е пкосмос. (3) таї те таластрофи \overline{M} памеріт \overline{N} стот тоснф. (4) дч \overline{p} -гме \overline{N} ромпе \overline{M} патоухі-сгіме идч, дую ксутте \overline{N} ромпе дчаду \overline{Z} пкосмос \overline{M} течсгіме, дую мтересмоу, дч \overline{p} -керомпе ечебет мауадч. (5) д-тамеріт \overline{M} маду \overline{p} -кес \overline{N} готпе \overline{Z} печні хімтауф \overline{m} -тоот \overline{C} идч \overline{N} сгіме, е-дугом етоот \overline{q} гіт \overline{M} йоуння хе, "гарег ерос фа пеоуобіф \overline{N} тфехеет."

VII. (2) σεμνή f. adj. holy, august. ἀπογράφω to register. (3) π.μ. Ν 601λ6 inn. ὁ τάφος tomb. εγλχηλ Rachel.

VIII. (1) συμβουλεύω to advise, give counsel. Αρχή[λλος Archilaus.

XIV. (1) xτο xτε- xτο' Q xτηγ vb. tr. to lay down (Μηο'); intr. to lie down, to succumb (to: e). προς πετ etc.: "according to what is ordained for every man." (3) ἡ ἀναστροφή lit. turning; here = lifetime, biography.

(6) $\Delta \gamma \omega$ $\Delta - M \Delta P$ $1 \Delta \gamma$ $\Delta \gamma \Delta \gamma$ $\Delta \gamma$

χV

(1) мегооу де тироу \overline{M} плетот тоснф, пл-ти \overline{M} тг \overline{X} ло ет смамалт, сеегре \overline{N} же митоует \overline{N} ромпе ката поуег-сагие \overline{M} плетот. (2) д-пегооу \overline{M} пече \overline{M} -пфіме ет мач, ете соу-хоутасе пе \overline{M} певот епиф, (3) <дую дчархет \overline{M} фіве \overline{M} 61 пмоув ет сот \overline{M} 7, ете тсаря те \overline{M} плетот тоснф, дую> д-пгат поюме, ете пмоус пе \overline{M} 0 тсофід. (4) дчиюмие е пкеліюм, дч \overline{P} -пов \overline{M} 0 тоуюм м \overline{M} 1 псю, е-д-тсофід \overline{M} 0 тмиттехмітнс кот \overline{M} 0 еусормес \overline{M} 0 оум \overline{M} 1 топом. (5) дсфюпе де, \overline{M} 1 тере-поуовім дрхет \overline{M} 1 сюр евох \overline{M} 1 пегооу ет \overline{M} 1 ду, д-памеріт \overline{M} 1 ют тоснф дрхет \overline{M} 1 фторт \overline{P} 1 емате

⁽⁶⁾ \overline{N} TA-MAPIA should perhaps be emended (with Lagarde) to \overline{N} TAC A-MAPIA, "in her 15th year Mary bore me." This verse makes much better sense if mycrhpion is read for cnealion; this is supported by the Bohairic version. ϵ_N for \overline{N} . AT2 ϵ T2 ϵ T2 ϵ T7 inscrutable, unfathomable.

XV. (2) "the day of his visitation," i.e. of his final illness and death. coy- day (in datings), prefixed to the number: coy-xoytace the 26th day. enhh, enhn Coptic month name. (3) The portion in <> is missing from the ms.; I have restored it on the basis of the Boh. version. nowne neene- noone' Q noone vb. tr. to turn, change, transfer (Mmo'); intr. to change, be altered. ὁ νοῦς mind. (4) ὁ αἰών period of time; age, generation; eternity, world. ὁ τεχνίτης craftsman; mnttexnithe technical skill. κοτ' e to turn into, become. τ.copnec error. ἄτοπος strange, odd; mntatonon confusion, unreasonableness. (5) cop cepcop' Q chp vb. tr. and intr. (± εδολ) to scatter, spread.

21 XM $\overline{\text{H}}$ $\overline{\text{H}$

XVI

(1) "ОУОІ ИЛІ Й ПООЎ. ОЎОІ Й ПЕЗООЎ ЙТА-ТАМАЛЎ ЖПОІ \vec{N}_2 НТФ. (2) ОЎОІ Й МЕКІВЕ ЙТАІТСЙКО \vec{N}_2 НТОЎ. (3) ОЎОІ Й Й-ПАТ ЙТАІЗМООС ЗІХФОЎ. (4) ОЎОІ Й МЕБАООТЕ ЙТАЎЗЛООЛЕ ЙМОІ ФАНТЕІЕІ Е ПТЕ ЙТАМЕТЕХЕ Е ПИОВЕ. (5) ОЎОІ Й ПАЛАС МЙ ИАСПОТОЎ, ЖЕ АЎБЛОМАЙ Й ОЎМННФЕ Й СОП ЗЙ ПХІМБОЙТ МЙ ТКАТАЛАЛІА МЙ ТМЙТЛАС СИЛЎ МЙ ФАЖЕ НІМ Й ВФЛ ЕВОЛ. (6) ОЎОІ Й МАВАЛ, ЖЕ АЎБФФТ ЗЙ ОЎСКАМАЛЛОЙ АЎФ АЎМЕРЕТМЙТРЕЧЕІРВООМЕ. (7) ОЎОІ Й ИЛМАЛХЕ, ЖЕ АЎМЕРЕТМЎЛЕЧЕЇ Й КАСКТ МЙ ЙФЛХЕ ТНРОЎ Й ПФМТ. (8) ОЎОІ Й ИЛБІХ, ЖЕ АЎЗФВТ Й МЕТЕ МОЎІ ЛИ МЕ. (9) ОЎОІ Й ЙМЛЗТ МЙ ӨН, ИЛІ ЕТ ЕПЕІЎМЕІ Е ЗЕПТРОФН Й МОЎІ ЛИ МЕ, ЛЎФ ЗОЛЛИ БЎДЛИБЙТЙКА МІМ, ФЛУТ

ne.282 difficulty, straits.

XVI. (1) oyou war woe unto.... (2) T.GKIBG breast. тс ко vb. tr. to nurse, suckle (ммо°); here intr. to nurse. (4) 6x00T6 in sense: internal organs in general. 2x00x6 vb. tr. to nurse (a child: MMo'); to carry (a child) during pregnancy. 61 6 nre to grow up; n. Te time, season. μετέχω to partake (of: 6). (5) δλομλή δλήλωμ $^{\circ}$ Ο δλήλωμ vb. intr. to become twisted, implicated, involved. MNTλως cnay deceit (lit. two-tonguedness). (6) τὸ σκάνδαλον impediment (a term applied to any behavior or situation that can be regarded as an impediment on the road to perfection). MNTP6461F-BOONG greed; see Glos. sub BOONG, 612. (7) Noxxe M nome lit. the words of sinking; this curious expression comes from Ps. 51:4 τὰ δήματα καταποντισμοῦ. taken to mean "destructive words." (8) 2087 = 2097 2697-2047' vb. tr. to steal (אוֹאס'). (9) ת.אבד bowels, intestines. έπιθυμέω to desire, be eager (for: ε). ἡ τροφή food, nourishment.

рокгоу \overline{N} гоуо бүтрір \overline{N} кшг \overline{T} . (12) біль \overline{p} -оу тбиоу? $\overline{\lambda}$ ішр \overline{b} бгоун \overline{N} са са нім. (13) ахношс оуоі оуоі \overline{N} ршмб нім бт на \overline{p} -новб. (14) \uparrow хш \overline{m} нос инт \overline{n} , \overline{u} надире \overline{m} надфебре, хе пиоб \overline{N} гва \overline{n} таінау броч гіт \overline{m} пабішт ізкшв бчину бвох г \overline{n} сшма \overline{n} точ пе паі \overline{n} тачтш \overline{m} \overline{n} броі г \overline{u} \overline{m} пооу, анок пеібвіни \overline{n} тахаіпшрос ауш \overline{n} гнкб. (15) ахха пхобіс паноутє пе пибссітне \overline{n} тачухн \overline{m} пасшма \overline{m} пап \overline{n} ."

XVII

(1) нат де ечже мнооу \overline{n} 61 памеріт \overline{n} егет геснф, агтеоун, агноофе ероч ечйкот \overline{k} ечфтртфр $2\overline{n}$ течфухн м \overline{n} печ \overline{n} \overline{n} . Пехаї нач хе, "хаїре, памеріт \overline{n} егет, па-тм \overline{n} \overline{n} 2 \overline{n} 0 ет наноус." (2) ачтафе-оуфф наі $2\overline{n}$ 0 оуноб \overline{n} фторт \overline{p} м \overline{n} 0 оугва м \overline{n} 0 оуготе \overline{n} те пмоу, ечхе \overline{m} мос хе, "хаїре \overline{n} 0 оуннере \overline{n} 0 соп, памеріт \overline{n} 0 фре. а-тафухн \overline{m} тон ерої \overline{n} 0 оукоуї \overline{n} теретексин ет нот \overline{m} 1 тагої. (3) \overline{n} 0 паховіс, \overline{n} 0 паречноуг \overline{n} 0 ме, \overline{n} 0 паречтоухо, \overline{n} 0 паречноуг \overline{n} 0 \overline{n} 0 пет скепахе \overline{n} 1 птер \overline{q} 1, \overline{n} 1 \overline{n} 2 пет амагте \overline{n} 1 птер \overline{q} 2 \overline{n} 1 поуф \overline{n} 1 течм \overline{n} 1 \overline{n} 2 \overline{n} 2 пет раз ечкіфоу емате, \overline{n} 3 \overline{n} 4 пооу, анок пека \overline{n} 4 егол \overline{n} 5 пімахе ет сфт \overline{n} 6 сфт \overline{n} 6 ерої 2 ф \overline{n} 1 пооу, анок пека \overline{n} 2 \overline{n} 3 \overline{n} 4 оуме \overline{n} 4 пооу, анок пека \overline{n} 5 \overline{n} 5 \overline{n} 6 \overline{n} 6 \overline{n} 6 \overline{n} 7 оуме \overline{n} 0 оухок ката ое \overline{n} 7 пекатгелос

mediator, intercessor.

ρωκε in sense: to consume. τε.τριρ furnace, oven. Vss. 10-11 (acc. to Boh. version) are omitted in the ms. (12) ωρε ερε- ορε Q ορε vb. tr. to enclose, shut in (Μπο΄); intr. to be shut in. (13) άληθῶς adv. truly. (14) ταλαίπωρος wretched, miserable. (15) ὁ μεσίτης

XVII. (2) τλως- + Inf. to do sthg. much; τλως-ογως το answer profusely. Νογτή, Q Νοτή νb. intr. to be sweet, pleasant. (3) τογχο τογχε- τογχο' Q τογχηγ νb. tr. to make sound, whole (Μηο'); to rescue, save; as n.m. safety, salvation. σκεπάζω to cover, shelter. κήνε νb. intr. to be sweet, fat. κιωογ Q to be fat, soft, productive, fertile.

TOYN-IAT EBOA N 2A2 N CON, N 20YO AE NE200Y NTA-NA2HT MKA2 ерот етве оуменуе й ийтриме етве тет сманаат марта тпарое-HOC, AIREPETEPKAZE XE ACOW, AYW NEIXW MMOC XE, GEPAT EN TELOYOH, THANOXE GBON N XIOYE. (5) HAL AS SINESYS SPOOY, A-HAFFEROC OYONZ NA! EBOX 2H OYPACOY, EYXO MMOC NA! XE, 'ιωτηφ πωμρε π Δλγειλ, Μπρρ-20τε <e> x1 Μ Μλριλ τεκτειμε, OYAG MIPTCTAZE EXN TECGINOW, XE NTACOW PAP EBOA 2N OYUNA CHOYALB. (6) CHANTO AS TO OYOHPS NEMOYTS S TESTS XS TO. (7) TENDY AS, MAXOSIC, NCOTHP \overline{N} TAYYXH M \overline{N} NANNA, MNPSN-APIKE EPOI, ANOK ANT-HEKZMZAA AYW HZWB N NEKEIX. NTAIHEPEI-EPRAZE AN, W NAXOGIC, (8) ANNA MNA+COYN-NGOOY M NGINOG M MYCTHPION, GTG MGKXNO GT OYAAB NG, OYAG ON MNGICWTM GNG2 xc wape-crime ww exm rooyt. (9) w haxoeic arw hanoyte, THE HTWO M REINOS M MYCTHPION AN RE, NTEINARICTEYE EPOK AN TO ME HE TEKENO ET OYALB, TATEODY N TENTACEMOK, MAPIA, TELcalibe \overline{M} Me. (10) terpe \overline{M} nmeeye \overline{M} nzooy \overline{N} Ta-Tkepacthc оуюм \overline{M} понре онм е течоуернте, ачмоу. (11) а-мечроме COOY FOR GYOYOO 6 60HK NCGTAAK N 2HPOTHC HHAPANOMOC. (12) AYW A126 EPOC, A-TEKMNTHOYTE TA204, A4WNZ, AYW 2M ITPEKTOYNOC \overline{q} e negelote ayno6 \overline{n} page gwie nay. (13) alxnoyk A6, ω hamepit \overline{N} whpe, we ecymaze \overline{M} mok $2\overline{N}$ 2ω b nim. Alamazte

⁽⁴⁾ τογη-ιατ' εΒολ to inform, instruct. περιεργάζομαι to be overly concerned, meddlesome. (5) διστάζω to hesitate, be in doubt. (7) There is an ellipsis or omission after aimepeiepkaze an: "I became overly concerned not (for any other reason than that) I did not yet know..." (8) εχη for αχη. (9) The meaning of this verse is obscure. If ene introduces a contrary-to-fact condition, the apodosis should be neinanicteye; cf. vs. 14 below for a similar problem. τειελίβε for τειλίβε; τ.λίβε lamb. (10) ή κεραστής horned-(viper). Ογωμ here: to bite. (11) nequence = his kin etc. παράνομος lawless, unjust. (12) Perhaps read τανλόμος for ταλόμο. (13) ήσυχάζω to be still, quiet; μμοκ is reflexive.

 \overline{M} пекмаахе \overline{N} оунам, а ісок \overline{M} . (14) акоуюр \overline{B} екхю \overline{M} мос на і хе, \overline{N} псавна хе \overline{N} ток пе паеїют ката саря, епеї †натамок хе аксексек-памааже \overline{N} оунам. (15) теноу де, \overline{W} памеріт \overline{N} унре, паховіс аую памоуте, ефюпе \overline{N} такчі-логос нямаї етве перооу ет \overline{M} мау, актре-неїмавін \overline{N} готе еї еграї ех \overline{W} 1, †пара-калеї \overline{M} мок, \overline{W} паховіс \overline{N} агафос, к \overline{W} наї евох н \overline{T} тячі- \overline{W} 1 нямаї. (16) анок гар ан \overline{T} -пек \overline{M} 2 ал, анок пунре \overline{N} тек \overline{W} 2 ал. (17) екфансфай \overline{N} намерре, †науфот нак \overline{N} оуфусіа \overline{N} смоу, ете таї те тромологеїа \overline{N} текм \overline{N} тноуте, хе \overline{N} т \overline{K} -оуноуте \overline{Z} \overline{N} 0 оуме, аую \overline{N} т \overline{K} -оуховіс \overline{Z} \overline{N} 0 оужок."

XVIII

(1) на і де ечхю минору пе і па-типте хало ескі шору павіют еіюснф, мпе і ефе пореф ріме еінар ероч е-ачорю ечхі ген \mathbb{R}_2 абе м пиор аро еісютм пфахе м митевіни етсчхю минору на і. (2) минса на і агр-пиевре м пегору м памор, м пиар етереченоралі наталої е пес \mathbb{P} ос га порхаї м пкосмос тир \mathbb{T} . (3) м тернору аівюк е пефріон ет гі вол, аро астюори \mathbb{R} 61 маріа

⁽¹⁴⁾ For enei †Natamok read prob. Neinamooyt Mmok: "If you were not my father according to the flesh, I would kill you..." or Neinaenei†ma nak: "I would rebuke you." Joseph obviously believes that his present illness is a result of this incident. cokck cekcek- cekcwk vb. tr. to pull, stretch. (15) 41-2000c mm to hold accountable (for: etec); 41-wn mm idem. (17) cwam can- coam Q coam vb. tr. to break off, cut off. theppe bond, fetter. your yeer-yaat Q yaat vb. tr. to cut, slaughter (as sacrifice). ↑ θυσία sacrifice, victim. ↑ ὁμολογία confession, agreement.

XVIII. (1) Noyeg (N) prep. without; 6ω Noyeg is best translated "to keep from (weeping)"; cf. vs. 4 below. xi is for Q xhy caught. n.2λ66 snare. e is required before Nyaxe. (2) Neigyali the Jews. ταλο ταλε- ταλο΄ Q ταληγ vb. tr. to raise up, offer up. (3) τὸ αΐθριον atrium, courtyard.

ТАМАДУ, АСЕІ ЕВОХ Є ПМА ЄТСІЙ 2 ІТ Й, ПЕХАС ИЗІ 2 Й ОУНОБ Й АУПН МЙ ОУШАС Й 2 НТ ЖЕ, "ОУОІ ИЗІ, ПАМЕРІТ Й ФИРЕ, АРНУ ЄЧИАМОУ ЙЕІ ПА-ТМЙТ 2 ХАО ЄТ НАНОУС І ШСНФ, ПЕКЕІШТ КАТА САРЗ."

(4) ПЕХАІ НАС ЖЕ, "Ш ТАМЕРІТ Й МАДУ, НІМ ЄНЕ2 2 Й ПГЕНОС Й ЙРШНЕ ЙТАУФОРЕІ Й ТСАРХ ПЕТ НАЄШ ЙОУЕФ МОУ? (5) ПМОУ ГАР ПЕ ПАРХШИ Й ПКОСМОС ТНРЯ МЕХРІ ЙТО, Ш МАРІА ТАМАДУ ЄТ СМА-МАДТ. (6) ТАНАГІН ТЕ ЄРО ЙТЕМОУ 2 ШШТЕ Й ОЄ Й РШМЕ НІМ.

(7) АХАД ЄІТЕ ПАМЕРІТ Й ЄІШТ ІШСНФ ЄІТЕ ЙТО, Ш ТАМЕРІТ Й НАДУ, ЇІ ОУМОУ АН ПЕ ПЕТЙМОУ, АХАД ОУШИТ ВДА ЄНЕ2 ПЕ. (8)

АХАД АНОК 2 ШШТ ТНАЖПІ-МОУ 2 Д ПТНРЯ ЄТВЕ ТСАРХ ЙТАІФОРЕІ ЙМОС. (9) ТЕНОУ 66, Ш ТАМЕРІТ Й МАДУ, ТШОЎЙ ЙТЕВШК Є2 ОЎН \mathbb{C} САЗТЙ \mathbb{C} ПЗ ХДО ЄТ СНАМАДТ ЙТЕНДУ Є ПТШО ПЕЄВОД \mathbb{C} Й ТПЕ."

XIX

(1) AYW AITWOYN, AIBWK 620YN 6 REOPION 6T $\overline{4}$ NKOTK $\overline{8}$ 2HT $\overline{4}$, Alec grou 6-a-magin $\overline{8}$ mmoy oywn $\overline{2}$ 6box $\overline{8}$ 2HT $\overline{4}$. (2) anok as alemone 2a2th teuars, a-tamerit $\overline{8}$ maxy 2mooc 2a2th neqoystate. (3) auth $\overline{8}$ neqbax 62pal 2a razo, Mrsuggemeom 6 waxe 1mmal 6box xe a-tm $\overline{8}$ tmo $\overline{8}$ mmoy $\overline{9}$ -xoeic 62pal 6xwu. (4) auth $\overline{8}$ teuar $\overline{8}$ oynam, auggeneinos $\overline{8}$ ag-azom 64nagt. (5) auggeneinos $\overline{8}$ teuarazote $\overline{8}$ tasix $\overline{8}$ oynam 64ciop $\overline{8}$ ncwi $\overline{8}$ oynos $\overline{8}$ nay 2wc curve $\overline{9}$ groi xe, "w raxoeic, $\overline{8}$ reator $\overline{8}$ cut." (6) aitel $\overline{8}$ tasix $\overline{8}$ oynam 64ciop $\overline{8}$ ncwi $\overline{8}$ oynos $\overline{8}$ itel $\overline{8}$ fasix 62oyn 2a reacthooc, aleg $\overline{8}$ teutyxh 6-acta26-teutyous $\overline{8}$ cynaent $\overline{8}$ 62pal, ayw 6pe-nebal-wing $\overline{8}$ rmoy 6wg $\overline{8}$

π.ογωλο π 2μτ discouragement. λρμγ perhaps; often simply indicates question, as here: "Is he to die?" Note Fut. II.

(4) το γένος race. (5) ο άρχων ruler, Archon. μέχρι prep. even up to, even including. (6) ἡ άνάγκη necessity;

Γληλεγιμ το ερο + Conj. is an impersonal construction: "It is necessary that (you) die also ..." (7) είτε ... είτε either ... or, whether ... or. (8) κπι- aux. vb. must; usu. refixed to Inf., as here. For ne.ebox see 27.2 (end).

XIX. (3) ΜΠΤΉ πο muteness. (6) τὸ στῆθος chest, breast.

1. ΨΟΥΨΑΕ throat. Βλι-ψιΝΕ = 4λι-ψιΝΕ messenger.

ERON 2HT \overline{q} etpeqel eron 2 \overline{N} comm, and \overline{M} me-bah \overline{N} oynoy xok eron, xekac eqganel \overline{N} 61 mmoy, \overline{M} m \overline{N} \overline{q} -anoxh \overline{M} may, xe epe- negtopt \overline{p} oyh2 \overline{N} coq ayw mpime m \overline{N} mtako \overline{N} net mooge 2 λ teq2 λ 1.

χх

(1) A-TANANY \overline{N} BAX-2HT NAY 6POI 6160M6 \overline{M} 6 RE4COMA, ACGOMGM 2000 NGON N NEGOYEPHTE, AYO ACZE EPOOY E-A-HNIBE \overline{M} nermom kary. (2) nexac nal $2\overline{N}$ oym \overline{N} tatcooy \overline{N} xe, "nekrmot φΗΠ ΤΈΝΟΥ, ω ΠλΜΈΡΙΤ \overline{N} φΗΡΈ, χ6 ΧΙΝ ΤΈΥΝΟΥ \overline{N} ΤλΚχ6-ΤΈΚ6ΙΧ 680λ 2M Heycoma, a-nkw2T ep-20Te, aganaxopel nag. (3) elc NEGGOT MN NCHBE N PATT AYWE TOWAYK NO N GE N OYKAYCTAAAOC $\overline{\mathsf{MN}}$ OYXIWN." (4) AIKIM 6 TAANG AYW AIMOYTG 6 NGGOHPG GIXW ΜΜΟΟ ΧΕ, "ΤΦΟΥΝ ΝΤΕΤΝΦΑΧΕ ΜΝ ΠΕΤΝΕΙΦΤ ΕΤ ΟΜΑΜΑΑΤ ΧΕ ΠΕΟΥ-Of IQ \vec{N} quite is in this term of the \vec{N} quite in in \vec{N} quite in EBIHN TWM." (5) TOTE AYTWOYN NGI NOHPE MN NOEEPE M NAMEPIT \overline{N} lut eluche, lyel yl neyelut, lyee epou eukinlyneye e \overline{N} 6-дчгши єгоуи є пшаб євол $\overline{\mathbf{M}}$ півіос. (6) дсоушу $\overline{\mathbf{M}}$ $\overline{\mathbf{M}}$ 61 дусід TERMOS \overline{N} weeps, etc tca \overline{N} xhos ts, nexac \overline{N} necchy xs, "ΟΥΟΙ ΝΑΙ, ΝΑΟΝΗΥ, ΠΑΙ ΠΕ ΠΦΦΝΕ ΠΤΑΦΦΠΕ Ν ΤΑΜΕΡΙΤ Μ ΜΑΑΥ, λύω ωλ τένου Μπενκότ $\bar{\mathbf{N}}$ ε νλυ έρος. (7) πλι ον τένου πέτ νλфеене-пенетот ерои е тмиху ероч фх енег." (8) тоте хучт-2 РАУ ЕВОА, АУРІМЕ 21 ОУСОП Й61 ПФИРЕ МП ПФЕЕРЕ М ПАЕІШТ ELWCHA, AYW ANOK 200T ON MN MAPIA TAMAAY M NAPHENOC NEN-PIME NMMAY NE, ENCOOYN XE ATTEYNOY M MMOY EL.

For xekic read xe. ἡ ἀνοχή a holding back. Νηετ = νετ.

XX. (1) εομεν ενευμένουν νε. tr. to touch (ε). ε required before νεοπ. τ.εοπ sole of foot. π.νιβε = π.νιβε. πενομο heat, fever, warmth. (2) πεκιμοτ φμπ thanks be to you; a Q equivalent of φπ-ιμοτ. χο χε- χο΄ εβολ νε. tr. to extend (νωο΄). (3) chβε ν γλτ΄ shin-bone. ωες νε. tr. to become cold. κβο, Q κμβ νε. tr. to make cool; intr. to become cool. ὁ κρύσταλλος ice. ἡ χιών snow. (5) κινδυνεύω to be in danger (of: ε). πωλε πλε- πολε΄ νε. tr. to free from; intr. to be freed from (ε, ν, ιν). (7) φωωμε φερμε- φορμέν νε. tr. to remove (νωο΄; from: ε); to deprive (ε) of (νωο΄).

(1) TOTE ALGOOT M HEA M HPHC M HPO, AINAY & HMOY, A461 CFG-AMNTE OYH2 NCWY, GTG HAI HG HET O N CYMBOYAOC AYW THANOYPROC, MAIABOAGE XIN TEROYETTE, EPE-OYMHHOE N WAB-N-TO N TEKANOC OTHE NOWS, GYXI-LOK N KOET THPOY, G-MN-HIG огооу, еретоуени ми оукапнос и кюгт нну евох ги теутапро. (2) μ-πλειώτ ειώς μφ 6ω $\overline{\mathbf{n}}$, λάνλη ε μεντλήει $\overline{\mathbf{n}}$ ζωά εγο $\overline{\mathbf{n}}$ NYMOC EMATE KATA HE EWAYMOYE EN OPPH 21 GONT 620YN G YYXH HIM II POME ET NHY EBOX 2N COMA, N 20YO AE N PETF-NOBE, enume eymansine \overline{N} dymasin e-mody he \overline{N}_2 ht \overline{q} . (3) \overline{N} tepe-ma-THIT 2 XXO GT NANOYC NAY & NENTAY & NOW4, A 4 W TOPT F AYW Aисчвах \uparrow - \overline{p} метн. (4) λ -те ψ үхн \overline{m} паетот тосн ϕ оуюф ет евох OYNOG N 28A, AYW GCWING NCA MA N 2011 THTT MIGC 26 MA. (5) $\overline{\mathsf{N}}\mathsf{T}\mathsf{E}\mathsf{P}\mathsf{E}\mathsf{I}\mathsf{N}\mathsf{N}\mathsf{Y}$ as e ino6 $\overline{\mathsf{N}}$ wtopt $\overline{\mathsf{P}}$ $\overline{\mathsf{N}}\mathsf{T}\mathsf{N}\mathsf{T}\mathsf{N}\mathsf{T}\mathsf{N}\mathsf{Z}\mathsf{E}\mathsf{T}\mathsf{E}\mathsf{\Psi}\mathsf{Y}\mathsf{Y}\mathsf{H}$ $\overline{\mathsf{M}}$ in $\overline{\mathsf{N}}\mathsf{E}\mathsf{I}\mathsf{W}\mathsf{T}\mathsf{H}$ ϵ 1 ω CH ϕ , λ Y ω λ 6 λ 4 θ 6 ω P ϵ 1 \overline{N} 26NMOP ϕ H ϵ Y ϕ 08 ϵ 6M λ T ϵ 6 ϵ 7OY ϵ 2OT ϵ THE HAY EPOOY, ALTWOYN N TEYNOY, ALEREITMA M HET O N OPKANON M HALABOAOC MN NTAXIC ET OYH2 NGW4. (6) AYHWT 2N OYNOG N WINE. (7) AYW MINE-XAAY N PWMG 2N NET COOYZ 6 HAGIWT CIWCHO CIME, OYAE MAPIA TAMAY. (8) NTEPERNAY AE NEI MOY XE ALGREITMA N NEXOYCIA M REAKE GT OYHZ NCWY, ALNOXOY GBOA, AYW XG MHTAY AAAY N GROYCIA GROYN G HAMEPIT N GIWT IWCHO, **ΑΥΡ-20ΤΕ ΠΕΙ ΠΜΟΥ, ΑΥΠΩΤ, ΑΥ20ΠΨ 21 ΠΑ2ΟΥ Μ ΠΡΟ. (9) ΑΙΤΩΟΥΝ**

XXI. (1) π.ρμc the south. δ σύμβουλος counsellor. δ σανούργος villain. τε.2ογειτε the first, the beginning.

ΦΑΚ-Ν-20 fearsome (lit. changing of face). οἱ δεκανοί a group of 36 divinities (or demons) who ruled over the Zodiac; originally an Egyptian astronomical division for time computation, but later debased into astrology. χι-2ωκ Ν to be girded with. τ. η η ε η μπρετ. πε. θη η sulfur, brimstone. (2) γ-θγμος (Q ο Ν θγμος) to be wrathful. εν for Ν.

ΟΥΜΑΕΙΝ σ-πωογ πε a token of their own (lit. which is theirs). (4) Read ε ει for ει. μα Ν 2οπς Ν2ητα a place in which to hide. (5) θεωρέω to look at, observe. ἡ μορφή form, shape. τὸ δργανον instrument.

 \overline{N} teynoy, limb \overline{N} oynpoceyxh ε haeiwt \overline{N} araboc, ε imb \overline{M} moc me,

XXII

(1) "naelwt, thoyne thec N TMNTACASOC, neiwt N The, півда й речнау, пімааже й речсютй, сютй є пекфире й меріт ETE ANOK NE, ELCONO MMOK ETBE NZWB N NEKSIX ETE NASIWT IW-CHO RE, NETNOOY NAI \overline{N} OYNOG \overline{N} XAIPOYBIN M \overline{N} REXOPOC \overline{N} NAFFEROC MN MIXAHA, HOLKONOMOC N NAFAGON, MN FABPIHA HBAIT $ω\overline{M}$ -ΝΟΥ4 \overline{M} Νλιων ε πογοείν, \overline{N} Сероєї с \overline{C} τεψΥΧΗ \overline{M} πλείωτ είω-CHO, NCEXI-MOEIT 22 TEC2H WANTGCOYWTE M HCAQ N AIWN N KAKE, AYW \overline{N} CENAPARE \overline{N} NE2100Y ET 2 \overline{T} M \overline{T} WM, NA1 EYNO6 \overline{N} 20TE THE MODING \overline{N}_2 HTOY AYW OYNOG \overline{N}_2 BA THE NAY \overline{N} AYMIOC GT 21XWOY. MAPE-RELEPO \overline{N} KW2 \overline{T} EP-OG \overline{N} OYMOOY AYW \overline{N} TE-OALACCA \overline{N} PEYфар ογω есенохреі. (2) маречфωπε 2π ογμπτ2ημερος ε2ογη ϵ τεψухн \overline{M} πλειωτ ιως Φ , $X\epsilon$ τλι τε τεγνού ϵ τ \overline{q} \overline{p} - X \overline{p} \overline{M} πνλ \overline{N}_2 HTC." (3) $+x\omega$ \overline{M} MOC NHT \overline{N} , ω NAMEPOC ET OYAAB, NAAHOCTO-ACC ET CMAMAAT, ME POME NIM ETCYNAMICH E RKOCMOC A4EIME G ппет маноуч мп ппенооу. ечфанр-печоуоејф тирч ечафе еграј NCA NEADOYE H NEGBAR, EGGANET EGNAMOY, GF-XPIA H HNA H 2100Y6 AYW T61NARONOFIZE $2\overline{M}$ RBYMA 6T 2A 20T6. (4) RNHN \uparrow -

⁽⁹⁾ ή προσευχή prayer.

XXII. (1) τ.ΝογΝε root. χλιρογειΝ Cherubim. ὁ χορός chorus, choir. ὁ οίκονόμος steward, manager. ελι- = 4λι-.

φΜ-Νογ4ε good news. ροεις vb. tr. to guard, keep watch (ε).

ογωτε ογετε- ογοτε Q ογοτε vb. tr. to pass through (ΜΜο*).

2τοΜΤΜ, Q 2ΤΜΤωΜ to become dark. ὁ δήμιος executioner.

ρε49λλρ demon; as adj. ένοχλέω to trouble, disturb. (2)

ήμερος mild, tame; κῶτ2κμερος calm. (3) τὸ μέρος part; used fig. here of the apostles as Christ's members. ειφε εφτ
λφτ Q λφε vb. tr. to hang, suspend (ΜΜο*); Q with Ναλ: to be captivated by. ελοογε prob. pl. of ελω, λλω snare, trap.

ει + Circum.: to be about to. τ.ειμηλριε passage, passing.

τ.εικληολογίζε defense. π.εγκλ = π.εκκλ. ετ 2λ 20τε fear
ful. (4) πλήν here as conj.: but, however.

NAKTOL GEPAL GEN TELNEWK GRON M HAGIOT LOCHO, HA HELEP-HUGGYG GT NANOY4.

IIIXX

(1) ACOUNG AG, NTEPEIXO M NZAMHN, EPE-MAPIA TAMEPIT M HALV OYWEM NOW! N TACHE N NA-MINYE, (2) AYW N TEYNOY EIC HINAHA MN TABPIHA MN NEXOPOC N NACTEROC AYEL EBOX 2N THE, ΔYCI , $\Delta Y < \Delta > 2$ EPATOY EXM ROWN M RAGIOT INCHÁ. (3) ΔYO N TOYNOY ATTEGRED AND NEXT THE TOOM EXWY EMATE, AYW ALEIME $x \in \lambda$ -ternor of the el. (4) by before eathering to be \overline{N} tot HAMICE, GPE-Π2AX + NCOU N OE N OYTHY EUNAOT MN OYKO2T EUор счочим мсь оугнын есоф. (5) пмоу де гиич Мпетооте казч n 61 620YN 6XM NCWA M NAMEPIT N IOT 10CH NTOPXT 680A, YO 69600T 620YN 69NAY 6POL 612MOOC 2A2TN T69AH6, 61AMA2T6 EXT HEREMAY. (6) AYW NTEPELEINE ME ASP-20TE NEL MMOY N EL COYN ETBHHT, ALTWOYN, ALBOK 6 NCA N BOA M NMA M NPO, $3.16\,\overline{\text{N}}$ TH EHGEET MAYAAH $2\,\overline{\text{N}}$ OYNOG $\overline{\text{N}}$ 20TE. (7) AYW $\overline{\text{N}}$ TEYNOY 116231 NAP X6, To Hentagel GBOA 2N NTOHOC M HCA M HPHC, BOK HAK GROYN TAXH NTXWK GBOA M HENTA-HAGIWT OYGR-CARNE MMOG иль. (8) дала роетс ероч й ое й почоети й неквал, же йточ THE TIME FOR KATA CAPA, AYO A4OTT-21C6 NHMAI 28N N6200Y N тамитонре фим, ечпит ийма ввох гй оума бума бтве тепе!-

never- for neip-.

MXIII. (1) π.2μμην the amen. ογω2ν νδ. intr. to repeat, inswer, respond (to: 6, 6χΝ, ΝΔ΄, Ναλ). τ.λαπ6 tongue, language. (2) 6χΝ often means "by, beside, at" with verbs i standing or stopping. (3) τ.26λ2ιλ6 death-rattle. π6-χελ2μα panting, exhaustion. χμη Q to be bitter, sharp.

4) †-ΝλλΓ6 to be in labor; τ.Νλλκ6 labor pains. π.2μχ reaning uncertain; prob. related to 2ωχ to be in straits, lying. † Ναλ to pursue. π.τηγ wind. † ὅλη woods, forest.

5) π6. ακλγ temple (of head). (7) τάχα adv. quickly. (8)

101. † άφελία advantage, profit.

воухн \overline{N} гурфинс, душ діхі-свю птоот \overline{N} не \overline{N} пунре тнроу, вфдре-неуе іоте ті-свю нду є теуффехід. (9) тоте двядтон дявык егоун, дяхі \overline{N} теухн \overline{N} плеішт єїшснф, дяєїне \overline{M} нос ввох $2\overline{N}$ сшма \overline{M} пилу \overline{M} при ечидфа $2\overline{N}$ течвасіс, \overline{N} соу-хоут-асе \overline{M} певот епнп $2\overline{N}$ оуе ірини. (10) негооу тироу \overline{M} пши \overline{N} памеріт \overline{N} еїшт ішснф севіре \overline{N} фе м \overline{N} тоуєї \overline{N} ромпе. (11) д-міхана дмагте \overline{M} птоп снау \overline{N} оумаппа \overline{N} голосіліком естаєїну, а гавріна дмагте \overline{M} пкетоп снау. Дуаспаге \overline{N} тефухи \overline{M} па-меріт \overline{N} єїшт єїшснф, дуталс епеснт є тиаппа. (12) мпе-хаду де $2\overline{N}$ нет гмоос гагти єїме же ачмоу, оуде ткемаріа тамалу \overline{M} песеїме. (13) душ дітре-міхана \overline{M} гавріна роєїс \overline{M} тефухи \overline{M} памеріт \overline{N} єїшт ішснф єтве \overline{N} речтшр \overline{M} ет $2\overline{N}$ нег гмоос. Ауш дітре-наггелос \overline{N} асшматос єш еугумиєує га

XXIV

(1) душ діктої єхм псшмд ечинх евол \overline{N} ее \overline{N} оукоуфон, дігмоос, діеїме \overline{N} мечвал епеснт, дієще \overline{T} епеснт єхшч \overline{N} оумоє \overline{N} мду, еїріме ероч. (2) пєхді... (End of Fragment III).

⁽⁹⁾ ΔΒΒΑΤΟΝ Death, Abbadon (Gk., ultimately from Hebrew). ἡ βάσις here: course. (11) π.τοπ edge, hem. τ.μαππλ cloth, handkerchief. ὁλοσηρικός silken. (13) ρεστωρπ plunderer; τωρπ τερπ-τορπ νb. tr. to seize, rob (μωσ). ἀσώματος incorporeal. ὑμνέω to sing hymns.

XXIV. (1) το κοῦφον (empty) vessel.

Glossary

Words are arranged alphabetically according to the order given on page x, with the following exceptions: (1) initial e_1 - and o_1 - occupy the place of 1 and o_2 respectively; in all other positions they are alphabetized simply as e_1 and o_2 + o_3 + o_4 + o_4 + o_4 + o_5 + o_6 + o_7 + o_8 +

Verbs are entered under the free (unbound) form of the Infinitive. In the rare instances when this form is not attested, a suppositious entry is used when there is no doubt about its pattern; otherwise the entry is under the first actually attested form. Other parts as speech are entered under their unbound forms when they are attested. The Grammatical Index should be consulted for most of the prefixed elements. I have followed Crum (A Coptic Dictionary) in listing most work and and nominal compounds under the final element. Also following from, derivatives are listed under leading verbal entries; the cross-inferences must be consulted in locating these. In order to provide space for less predictable compounds, nouns of action in 61N-, which we be formed freely from nearly any verb, have been systematically excluded, as have many agent nouns with peq-.

Where space has permitted, a selection of variant forms has been given. Under verbal entries these are placed in parentheses; otherwise they are listed serially after the main entry. These variants will into two types: (1) simple spelling variants, especially between and a supralinear stroke; (2) dialectal or "substandard" spellings occur in otherwise fairly standard texts. The latter have been haded to increase the utility of the Glossary; many of them are not to be encountered in non-standard manuscripts.

Cross-references are grouped at the end of each letter. The mpletely predictable forms of the two verb types kot ket- kot' Q ket and cot\(\overline{\pi}\) cet\(\overline{\pi}\)- cot\(\overline{\pi}\) does been systematically exlanded from the cross-references; all other bound forms and qualitatives have been listed. Many spelling variants involving $\epsilon i/i$ and troke/ ϵ have also been excluded.

A adv. of approximation, as in A toy about five, A OYHP about how much? Cpd. as NA, as in NA 96 M MA26 to the extent of about a hundred cubits.

alc, ac n. a blow, slap (usu. on face). +-lac, + \overline{N} oylac to slap, strike (Na*). $0\overline{C}-\overline{N}-$ lac n. = lac.

ABAGHEIN, ABIGHEIN, ABAKHINE, A466EENE n.m.f. glass.

ABO (pl. ABOOY6) n.f. net (for fishing or hunting).

ABOK, ABOK (f. ABOKE; pl. ABOOKE) n.m. crow, raven.

AGIK n.m. consecration. x1-AGIK to consecrate; as n.m. consecration.

Q to be great, honored; as n.m. increase, growth.

AGING, AHC, AGIC n.f. greatness, size, quantity.

AKHC, AKEC, AKIC n.m. girdle, clothing.

Aκω, Arω, rω n.f. filth; carrion; anything ruined.

אג adj. deaf. דְּ-אג (Q o אֹג) to become deaf.

אג n.m. pebble; hail-stone in אג א חום.

n. only in ω-ω-ω n.m. a cry (cf. ω-ω).

ANAY, ANGY, ANHY adj. white.

אגם אאס ' Q אארץ (imptv. אאסדא) vb. intr. to go up, ascend (to, up to, onto: e, e²ףאו e, exh, e²ףאו exh); to mount (an animal), to board (a ship); rarely tr. with אאס '. ףפּקאאָבּ rider.

AXIA n.m. field-mouse or sim.

aake, aakh n.m. last day of month; \overline{N} aake (\overline{N}) on the last day of (+ month name).

Aλοκ n.m. corner, angle; prob. not Sah. (cf. κλα6).

AXOM n.m. bosom.

λλογ n.m. child, servant; not properly Sah. (cf. φηρ6).

axo6 n.m. thigh.

ANTENAC n.m. bone-marrow.

λλω, ελω (pl. λλοογε, ελοογε) n. snare, trap.

λλω, λλογ (pl. λλοογε, λλλγε, λροογε) n.f. pupil of eye.

אטגע n.f. forced labor; term of service; a measure.

λλωογε n.pl. bunch (of grapes) or sim.

AMA Ama; fem. title of respect or reverence; cf. AMA.

AMA; fe vb. intr. to prevail, take control, rule (over:

exN, 21xN); to be valid, hold good; to persevere, continue; vb. tr. to grasp, seize, take possession of

(MMO'); to retain, detain, take or keep captive; to

learn by heart; to hold (MMO') liable (for: e). As

n.m. power, possession. ATAMA; te unrestrained, uncontrollable; MNTATAMA; te lack of restraint, incontinence.

peqama; te self-controlled person.

AHE (pl. AMHY, AMHYE) n.m. herder, herdsman.

אא n.m. the underworld, Hades.

ληογ 2nd pers. m.s. imptv. of 61; f.s. λημ; pl. λημιν, λημειτ \overline{N} .

анге, амрн (pl. амрну) n.m. baker; митанре baking.

ынгнге, ымреге n.m. bitumen, asphalt.

AN neg. part. not; for uses see Gr. In.

אר- one who is in charge of; only in cpds. with nos.: אר- мнт decadarch, אח-שס chiliarch, אוּתא-שס chiliarch.

יאר prefix for forming collective nouns from numbers, as in יום (group of) a thousand, אור (group of) .50.

שואו vb. intr. to become pleasing, better. As n.m. beauty; אואבו to be(come) pleasing, good. אואבי to please; see § 20.2 for usage. או in c+באר n.m. perfume (cf. ctol).

אראם (pl. אראבעם) n.m. oath; used with eige, chine, $\omega p \overline{\kappa}$, †, тאף האראם under oath.

ANOK, ANT- indep. pron. 1st pers. sing.: I.

ANON, AN-, ANN- indep. pron. 1st pers. pl.: we.

λΝογροε n.m. watchman, guard (λη + ογροε).

ANCHEG, ANCHE (GN), ANZHEG, ANZHE (GN) n.f. school.

אראש n.m. sneeze.

ANTHAM, ANTEAM n.m. skull.

ANE, ONE n.m. courtyard.

Ana Apa, masc. title of respect or reverence (saints,

martyrs, respected monks, etc.).

anac adj. old; syn. of ac q.v.

Ane (pl. Anhye) n.f. head (lit. and fig.); total sum (of money), capital; also n.m. chief, village head.

митапе headship. Atane headless. F-Ane to become head. leader (of: 6). чи-тапе to behead.

anoт (pl. anнт) n.m. cup; са п anoт cup-maker, cup-seller.

априте n.f. period of time; \overline{N} оукоу! \overline{N} априте for a little while; \overline{N} оунов \overline{N} априте for a long time; \overline{p} -оунов \overline{N} априте to spend a long time.

лю, ню, выс n.f. a number (of), several.

ърнв, брив, рив n.m. pledge, deposit, guarantee.

λρηγ, ελρηγ adv. perhaps.

אףאב" n. end, limit (suff. required, as in אףאבּע א nκλε the end of the earth; 3rd pers. pl. sometimes -Nογ); אדאףאב" boundless.

apike n.m. fault, blame; aтаріке blameless; 6N-apike to find fault (with: e), to blame (e); речей-аріке fault-finder; митречей-аріке criticism.

APIM n. name of an edible plant.

Apooye, Apwoy n.pl. burrs, thistles; cp-Apooye idem.

Apon to become cold; as n.m. cold, chill.

אבשקא n. name of a skin disease.

Apφικ n.m. lentils.

ac, aac adj. old (usu. not of people); нрпас, ерп-ас old wine; митас oldness; р-ас to become old.

acai, Q acωογ, aceιωογ vb. intr. to become light, slight, casual; to be swift; as n.m. lightness, hastiness, alleviation. 2N ογαςαι easily, casually.

ACHP n.m. one's belongings.

ACIK, CIK n.m. an illness, related to fever, chills.

acoγ n.f. price, value; +-acoγ to pay; F-acoγ 2a to set a
price on.

acne n.f. language, speech; acne \overline{N} aac idem.

AT- prefix for the formation of negative adj.; §27.1.

- ATO, ATA n. a lot, multitude; usu. with indef. art.; ATO W CHOT. ATE-CMOT adj. phrase: of various or many sorts.
- λΥ, λΥε, λΥεις, λΥει imptv. vb. (1) bring here, give! All forms occur prenominally; with pron. suff.: λΥεις.
 - (2) come! come, let's ...! (with Conjunctive).
- אבץג, איבן, איבן, איבן, איבן, איבן, איבן, ayan, ayan to tend toward (a certain color); אים איב איב (of or in) a variety of colors.
- λγεικ, λγλκ n.m. (ship's) cargo.
- AYHT n.m. company of people; monastic congregation.
- λγω conj. and; for uses see Gr. In.
- λφ n.m. furnace, oven.
- μφ interrog. pron. what? See §§ 14.2, 16.1 for usage.
- אנאָגו, Q og vb. intr. to be(come) many, numerous, to multiply; Q is very frequent. As n.m. multitude, amount.
- A4, AA4, AB n.m. a fly; A4 N GBIW bee; A4 N OY2OP dog-fly.
- ріесе of flesh, meat. са й ач meat-seller. gar-ач butcher. оуєм-ач to eat meat. gen-ач to buy meat.
- λ26 n.m. lifetime, extent of lifetime; P-λ26 to pass one's life; gap-λ26 short-lived; P-gap-λ26 (Q o N gap-λ26) to be short-lived; MNTgap-λ26 a short life.
- λεε vb. intr. to be in need (of: NA*).
- λεο (pl. λεωωρ) n.m. treasure, treasure house, storehouse.
- Azon n. only in $\lambda g \lambda_2 o m$ to sigh, groan (at: e, $\epsilon x \overline{n}$, $\epsilon_2 p \lambda_1 \epsilon x \overline{n}$); as n.m. groan, yawn, roar.
- λεβ, λεβε n.m. marsh herbage, sedge.
- A:Po* interrog. adv. requiring anticipatory suff. referring to subject of clause. (1) with foll. verb:
 why? as in λ2 poτΝ τετΝριμέ why do you weep? (2) with suff. alone or with foll. noun: what about ...? what's the matter with ...? (3) with μΝ: what has ... to do with ...?
- λιωμ, λιωμε, λιωωμε n.m. eagle (originally: falcon).

axe, aaxe (or oyaxe?) n. blow, cuff.

 $xx\overline{n}$, $ex\overline{n}$ ($xx\overline{n}\tau'$, $ex\overline{n}\tau'$) prep. without; a foll. indef. n. has no article.

AGREC, ATREC n.f. moisture.

AGOATE, AKOATE n.f. wagon, cart.

жерни n.f. a barren woman; also adj.; митьерни barrenness; г-херни to become barren.

11 61P6 YOMA : NIHMA ACWOY: ACAI AMHY(E): AME JSQ : PSJK AAC: AC AMOY: EMOY ATA, AT6: ATO PK : PKK ATBEC: A6BEC AAXG: AX6 TWLESE: TWLHSE NONY: -NY ATECMOT: ATO AB: A9 וגאג :אגλΥλλΝ: λΥλΝ ABGN: OBN ABIGHEIN: ABAGHEIN ANA": ANAI AYAN: AYEIN ABOK: ABWK ANAY: NAY AYE, AYEL: AY ABOKE: ABWK ΑΝΑΥω: ΑΝΑΦ AYEIN: AYAN ABOOKE: ABOK ANT-: ANOK AYEIC(1): AY YNEINE: EINE **λφ': 61φ6** ABOOY6: ABO ANZHB (GN): ANCHBE ABOYI: A4 **AUAA: AA** ANIT, ANINE: GING **৯** ୮መ: እKመ λφλ2ΟΜ: ωφ, λ2ΟΜ ANI": EINE ACHC, ACIHC: AIAI 706: 6106 AHC: AIAI NONA: - MNA አወዘ: አወኔ ! ANXIP: XIP AKEC: AKHC **እ**ወ₭፞፞፞፞፞፞፞፞፞፞፞፞፞፞፞፞፞፞፞ቜጜ፞፞፞፞፞፞ጜ AKIC: AKHC YNXOX: XOX AUT-/: 6106 AOYHP: OYHP A46666NE: ABAGHEIN **λκολτέ: λ6ολτέ λλλΥ6: λλω** AOYON: OYON YOOTP : 3TPA **λ**λθΥ, λλΗΥ: **λ**λλΥ TORK: THRK PK : IYOPK **ANHY, ANO[®]: ANG** AUHAE: YUE λ2λ: 626 **λ**λ0: λ0 AP62: 2AP62 λ26: 626, **ω**26 λλοκ: λο APOOYE: POOYE 226PAT": ₩26 APOOYE: AAW **λλοογε: λλω** A200M6: A20M ALOY: ALW APWOY: APOOYE A2OOP: A2O $\lambda\lambda\omega\tau\overline{N}$: λO , $\lambda\lambda \varepsilon$ AC: AAC 1 -/ -: xω AM-, AME: OME ACEIWOY: ACAI XXW: EXW AMH, AMHEITN: AMOY ACKE: WCK

В

BA, BAG, BAGI, BOI n.m. branch of date-palm.

BAABG, RABW" (ΒΑΒWW") Q BABOT (ΒΑΒW) vb. tr. to despise (ΜΜΟ"), regard as foolish; intr. to be insipid, foolish.

ΒΑ(Α) ΒΕ-ΡωΜΕ n.m. boaster; ΜΝΤΒΑΑΒΕ-ΡωΜΕ boastfulness.

вымпе, вымпе n. goat; вымпе \overline{N} 200үт he-goat; вымпе \overline{N} c21мe she-goat; фамр \overline{N} вымпе goatskin; ч ω \overline{N} вымпе

goat's hair; mane-mamme goatherd.

BAI, או n.m. night raven, screech-owl.

- האא n.m. eye. אדאא shameless; אאדאדאא shamelessness.
- BAROT n.f. skin garment; skin bag.
- варот, варот, валот, варате n.m. brass, bronze; 20 м \overline{N} т (\overline{N}) варот idem or sim.
- BACNG, BACENG, BACNHG, BACHG, BACIG, BACING N.M. tin.
- BAGOP, BAGAP, BAGAAP, BAGOOPE n.f. fox.
- вафотр n.f. saw; ра й вафотр adj. saw-toothed.
- sagoyo n.m. rue.
- exace n.f. heifer.
- intr. to well up, be poured forth.
- вске (pl. векнуе, векебуе, векебуе) n.m. wages; †-веке, †

 м пвеке to pay, reward; ты-веке employer; реч†-веке
 idem. xi-веке, xi м пвеке to receive wages; xi е веке
 to hire (ммо); хы-веке hireling. смм-веке to fix
 wages. рмвеке, рммвеке hireling.
- of fetters, sword. +-Benine to put in irons, to fetter (e). x1-Benine to be put in irons.
- κεγω n.f. whirlpool.
- RECENHET, BECNAT (pl. BECNATE) n.m. smith; $m\overline{N}$ TBECNHET the work or craft of a smith.
- кн n.f. grave.
- BHB, BHBG n.m. cave, hole, den, nest (of animals).
- BHKG n.m. woof (of loom).
- EHGE n.f. swallow (bird); xxx-BHNG swallow-sparrow.
- sur n.m. palm-leaf. sur-cnip n.f. rib.
- вн6, вс6, вб n.m. falcon.
- rip (pl. врноуе) n.m. basket (of palm-leaf). віре, выре n.f. idem.
- single piece (of fruit, etc.).
- blind person; MNTBXxe blindness; \overline{F} -BXxe to become blind

- (Q o N BXA6), to make blind.
- Bλ2MOY (pl. βλ2MOOYe) name of a people (Gk. Βλέμυες); usu. located on east bank of Nile in Nubia. Other spellings include βλ62MOY, β6λ62MOYe, βλλ62MOY.
- BXx6, BXxx6 n.m.f. pottery, earthenware; P-BXx6 to be made of clay.
- BNNE n.f. date-palm, date(s); BN-payne virgin palm; BNgooye dried dates; BA N BNNE palm-branch; BAA N BNNE
 date-stone; GBIW N BNNE date-honey; KA4 N BNNE Stem,
 trunk of palm; AOOY N BNNE cluster of dates; CA N BNNE
 date-seller; CF-BNNE date-palm thorn; TAG N BNNE date
 cake; gN-BNNE, gG-BNNE, gE-BNNE, COYN-BNNE palm-fiber.
 BOINE n. harp or sim. musical instrument.
- BONE Τ ΕΣΕΦΑ (p.c. BANE T-) vb. tr. to dig, dig up, dig out (ΜΜΟ'); to burrow, delve; vb. intr. to be undermined.

 BONE Τ GEON vb. tr. to undo, take apart; intr. to be dug up, out.
- BONTE, BANTE (BNT-) n.f. gourd, cucumber; gourd-garden(?); сарвонте gourd-seller. вNT-N-66A06 pumpkin.
- BOC∓ Q to be dry, parched.
- BOYBOY vb. intr. to shine, glitter; as n.m. shine, glitter. BOY26, Bω26 n.m. eyelid.
- врвр vb. intr. to boil; врвр егра: ммо° to boil up with, cast up; as n.m. boiling.
- врефну, верефну, врфну n.m. coriander seed.
- въре, выре adj. new, young; мытвъре youth, newness; ы въре adv. anew, recently; ътвъре to renew; to become new. въреосут, вересфоут n.f. chariot.
- вю n.f. tree (when fruit is specified; otherwise use дни).
- Bωκ, Q βμκ vb. intr. to go, depart; to die; to be about to (+ ε + Inf.). Used with most prep. and directional adv. in regular senses. Note βωκ εγλτ to visit; βωκ 21 to undergo (as well as "to go upon").
- вωке vb. tr. to tan (leather: ммо'); вык-φыν tanner. вωх вех- вох' Q внх (± евох except when indicated) vb. tr.

to loosen, untie, unfasten (MMO'); (not + eBOX) to explain, interpret; to weaken, enfeeble; to nullify; to dissolve. Vb. intr. to be(come) loosened, undone, loose, scattered, melted, dissolved, weakened, paralysed, faint; to become dissolute; to be terminated, to die, perish. As n.m. solution, interpretation (not + eBOX); weakening, slackening; laxness, unrestraint; dissolution, destruction. BOX MN to come to terms with.

BOX n.m. the outside. nBOX, \overline{M} nBOX prep. on or to the outside of, outside; independent of, beyond, free from; contrary to. 680% adv. out, outward, away; usu. with verbs, but occasionally after prep. phrases with sense: onward, and so on, henceforth; for eBOX + prep. see sub prep. eBOX xe conj. because. N BOX adj. phr. outer, external; adv. outside, extant, in existence. CABOA, CA \overline{N} BOA, \overline{N} CABOA, \overline{N} CA \overline{N} BOA, \overline{M} HCA \overline{N} BOA (1) prep. (+ N, 6) outside of, beyond, away from; (2) adv. outside, on the outside. q_{ABOA} (1) prep. (+ \overline{N}) to the outside of; (2) adv. to the end, forever, finally, utterly. 22802 N prep. from, away from. 21802 (1) adv. outside, on the outside, from the outside; (2) prep. (N) outside of, beyond, except for; er 2180% adj. phr. external. F-BOX, F-BOX vb. intr. to avoid, escape (from: ϵ , \overline{N} , $2\overline{N}$). KA-BOA vb. tr. to vomit ($\overline{M}MO^{\epsilon}$).

 \overline{N} CABHA G/\overline{N} (1) prep. except for, outside of; (2) conj. except that (+ Conj.). \overline{N} CABHA xe except that, unless, if not. $\overline{N}B\overline{\lambda}$ ($\overline{N}B\overline{\lambda}\lambda\lambda^{s}$) prep. ($\pm \overline{N}$) without, except for, beyond.

**www adj. bad, mainly in fixed expressions such as c†-εωων (cτοι), coy-εωων (cιογ), gp-εωων (ghpe), ελγον εωων.

**cone n.f. evil, misfortune; p-εοονε to act badly, evilly (toward: νλ*); μπτρεσβ-εοονε evil-doing.

**ciep-εοονε (1) the evil-eye; (2) adj. envious, greedy; μπτειερ-εοονε envy, greed; p-ειερ-εοονε to become

envious of (e. exN): pequep-soons enchanter, one who casts evil-eye; mntp64616p-Boone greed; x1-616p-Boone to receive the evil-eye.

Boope Beepe- Boop Q Boope vb. tr. to push, drive (Mmo): to repel (NCA); to prevail over, defeat (MMO'); intr. (+ 680A) to swell up, protrude. As n.m. protuberance.

вюте (чюте чюбе) вет- (вот- воот-) Q вит vb. tr. to pollute (MMO'), befoul; to abominate. BOTE n.f.(m.) abomination; P-BOTE (Q o N BOTE) to become hateful; x1-BOTE to loathe, abominate (6).

Bωω Beω- Bοω (Bλω) Q BHω vb. tr. to strip, divest, flay (MMO'); to lay bare, unsheathe; to loosen, unfasten, undo, release; to despoil; to forsake; intr. to be undone, loosened; Q to be naked. pequo robber, despoiler. $(B \oplus 2)$ $B \in 2 (n \in 2-)$ O $B \mapsto 2$ Vb, tr, to bow (the head). $(B\omega_2\overline{N})$ $BG_2\overline{N}$ Q $B\lambda_2\overline{N}$ vb. tr. to roof over $(\overline{M}MO^4)$, cover with

awning. BW2N, BA2N n.m. canopy, awning.

BABG-: BAABE BETT: BOTE BABOT: BAAB6 866: BH6 BABO, BABOO': BAABE BHNNE: BNNE BAG, BAGI: BA вире: врре BAIPE: BIP BHT: BOTE BAKT: BOKE BH6: 4066 $BA\lambda B\overline{\lambda}$: $BO\lambda B\overline{\lambda}$ BING: BHNG **8** እእወተ፡ **8** እ **P**ወተ BINIBE: BENINE BANING: BENING BIPG: BIP BANTE: BONTE $B\lambda\lambda X G: B\overline{\lambda} X G$ ВАРАТЕ: ВАРОТ B₩-: B₩N6 варот: варот BNT-: BONTE **Βλφ*: Βω**φ BNT: 4NT вафаар: вафор BOI: BA ВАФАР: ВАФОР BOX: BOX BA2N: BW2N BO: 40 Вврнее: еврнее BOO: 40 8686: 866B6 BOONE: BOON веере-: вооре BOOP (*): BOOPE BEKE (E) YE: BEKE BOOTT: BOTE векнуе: веке BOPBF: 2BOPBF верефнує: врефну BOTT: BOTE BEPEGWOYT: BFGOOYT BOT6: BOT6

вра: бвра BPBOPT: 2BOPBP BPBWP: 2BOPBP 876-: 6877 врнуе: евра 8P1-: 68PA BPHOYE: BIP врнбе: еврнбе в Рону: врефну BTOOY: 4TOOY BW: 40 воте: чоте BW26: BOY26 8066: 4066 **8**₩6℃: 4₩6℃ BG: BH6

воте: чоте

RO6C: 4066

rn": 61N€ rpomne: 6poomne **Γω: λκω**

- e (epo*) prep. (1) reference: to, for, as regards, in respect to; (2) purpose: for, as; + Inf. in order to; (3) direction: to, toward, into; (4) hostility: at, against; (5) debt: against, due from; (6) ethical dative with many verbal expressions (cf. §30.6); (7) comparison: than (cf. §29.3); (8) temporal: at, in; (9) other meanings in combination with individual verbs, e.g. direct object, instrument, separation.
- сви n. darkness, only in F-вын to grow dark.
- свіни adj. poor, wretched; м \overline{n} тєвіни misery, wretchedness; \overline{p} -євіни to become wretched.
- נהוש (נפופ-) ח.m. honey; נפוש אום pure honey; נפופ-200ץ π wild honey; או אום הוש honey grove.
- CROT (pl. 682T6, 686T6) n.m. month; may be followed directly by month name without \overline{N} . $2\overline{P}$ -680T every month.
- свра, вра (вре-, ври-; pl. еврнуе, врнуе, врнуе) n.m. seed; евра-сшфе n.f. seed-grain.
- смрисс, Вврисс, вчрисс, врисс n.f. lightning; +-вврисе to lightning.
- сківс, ківє, кієвє n.f. breast; перківє n.f.m. idem; †єківє to suckle; хі-єківє to be suckled.
- ελκω, σλκο n.m. fruit of sycamore.
- vine; ελελ-, λελ-, λεελ-, λιλ-) (1) n.m. grape, grapevine; ελελ-2Μx sour grapes; ελελ-900γε dried grapes,
 raisins; εω Ν ελοολε grape-vine; μλ Ν ελοολε vineyard;
 ειε2-ελοολε idem; ελελ-κημε n.m. bruise. (2) n.f. tonsil (?); pupil of eye (but cf. λλοογε).

exame n.m. heron.

ene n.f. hoe, plow.

CHAPE n.f. inundation (of the Nile).

CHHO n.m. anvil.

enice n.m. dill, anise.

CHNT, GIMNT, GMNTG n.m. the west.

enoy, amoy (pl. emooye) n.f. cat.

ENG, NG, GN interrog. part.; see §29.1.

- ene conditional part. if; see §29.1.
- enez, eneze, enhze (1) n.m. eternity, age, era; (2) adj.
 eternal; (3) adv. forever (with neg.: never). ga enez
 (1) adv. forever (neg.: never); (2) eternity; (3) adj.
 eternal (N ga enez, NN ga enez). ga nienez adv. for-
- ever. ga eng₂ \overline{N} oyogig idem. xin eng₂ from of old. en₂, \overline{N}_2 n. eyebrow. $\overline{M} \times \overline{N}_2$, $\overline{G} M \times \overline{N}_2$ n.m. idem.
- 600γ n.m. glory, honor; 2λ 600γ adj. phrase: honorable,
 glorious. †-600γ to glorify, give honor to (Nλ*); as
 n.m. glorifying; μπτρεψή-600γ glorification. χι-600γ
 to be glorified. μλι-600γ desirous of glory.
- єпни, єпети, єпнф name of 11th Coptic month.
- enpa n.pl. vanities; as adj. vain; as adv. in vain; $m\overline{n}\tau$ enpa vanity.
- enw n.f. part of door fastening.
- ернт (рнт, еррнт, ррнт) vb. tr. to vow, promise, devote (ммо"; to: ма", є); as n.m. (р1. ератє) vow, promise.
- брну n.m.f. fellow, companion; usu. with possessive prefixes as a recipr. pron.: each other, mutually. See 28.4.
- epore n.m.f. milk; p-epore to give milk; t-epore na to suckle; oyem-epore to feed on milk; 2a пероте still sucking, not weaned.
- echt n.m. ground, bottom, lower part; enecht adv. down, downward, to the ground; enecht e prep. down to, down into; н песнт adv. below, down below, at the bottom of (нмог); сл-песнт adv. on the lower side, below; as prep. (+ N). гл песнт adv. underneath, below ground; prep. (+ N) under. гл песнт adv. on the ground, from on the ground. хл песнт adv. from below. Р-песнт to go under (N).
- ecooγ n.m. (f. ecω) sheep; μλη-ecooγ shepherd.
- ет, ете rel. pron.; see Gr. In.
- etbe (etbht") prep. because of, on account of; concerning, about; for the sake of. etbe xe conj. because. etbe nai adv. therefore. etbe oy adv. why?

GTO2, GTA2 n.m. garment or length of cloth.

εγω, εογω, ογω n.f. pledge, surety; † Μπο' Ν εγω to give as a pledge; ογω ν Μπο' Ν εγω to deposit as a pledge; κω Ηπο' Ν εγω idem; χι Νπο' Ν εγω to take as a pledge.

 $\epsilon \omega$, $\epsilon \omega$,

 ϵ ϕ x ϵ , ϵ ϕ x π ϵ (1) conj. if (§29.1); (2) as if, as it were;

(3) exclam. how! (4) adv. surely, indeed; (5) before apodosis: then.

eze, 2e, λ2e, λ2h part. yes; indeed, verily; also used to introduce questions. εωχε/εωωπε ε2ε if indeed, if so.

 e_2e (p1. e_2ooy , e_2Hy , e_2ey) n.f.m. ox, cow.

exw, axw n.f. tongs, pincers.

n.m. a Nubian, Cushite, Ethiopian.

CRATE: 6BOT CRUTE: CROT CRIET: 6810 CBOA: BWA CELHAC: CELY ERT-: WAT свю: Мпо CROC: OBO CCBT: G1687 CCIBT: 616B〒 cete: 616 CELOYA: GLOYA ccio: eto LET: ww INATE: KOT TROTE: KOT LENT: KOT CACAT: GAOOAE CANOLE: SYOUN ' 10026: A120026 ' \0: AXO THATE: HATE CHAY: MMAY OWKH : 004111 THOOYE: EMOY thingo: Mpw THOO: OME · HXH2: 6H2 THE CHE

· #-: 61NG

ene-: whe 6NH26: 6N62 GNOYNT: ENT GNTAIP: NOYTE ентир: ноуте 600Y: 61W GOY: EIW eoyo: eyo enein: enun епесит: есит епітй: еітй επογφλη: ογφλη euc: yuc ep-: elpe EPAT": PAT" брате: брит epse: wpB **€РВТ:** РВТ 6P62: 2AP62 €РНВ: ДРНВ ephs: sypes epn: po 6P0': 6 epo: Ppo брй-: нрп бррит: брит EPTO4: PTOB epw": po epw, epwoy: Ppo

€C: 61C

6CO: 6COOY 6T: WW 6TA2: 6TO2 6T6: 6T **ӨТВННТ": ЕТВЕ** ETN: TWPE etoot': Twpe ετογÑ-: τογω" **6ΤΟΥΦ΄: ΤΟΥΦ΄** erne: ne etno: wtm ETPIM: TPIM emod: elm EQATE: QOT εφλγ: εφω **ნ**თი: ნთთ ефоте: фот €goy: €gw 6g) T . € 1 G) € egone: gone 600T: 00T ефтеко: фтеко **சேற்ற ∈: அற ∈** egxne: egxe 64-: W46 64РА: ЕВРА ечрнее: еврнее 6267: 626 62H: 2H

62HY: 626

€2 PN: 20 **6XH: X06** 62161B: 2161B 62 TO: 2 TO 6XHY: XOI 62KO: 2KO 65 N7 : 5 N7 , 62 TOOP: 2 TO 66270: 6600 628A1: 2TA1 66206: 6206 EZNET: ZNET ezony: eze EXN: XXN 6600a: 660a exN, exw": xw" 6600a6: 6600a 6200YT: 200YT EXNT": AXN 6600e: 6600 62 by : 50

н

ні n.m. house; household, family. мес гм ні one born in household. рміні, реміні n.m. (1) member of household, kinsman; (2) monastic superintendent; р-рміні to be akin. ні n.m. pair, couple.

ни п.т. аре.

нрп (брп-, рп-) n.m. wine. нрп ас, брп-ас old wine. саунрп, сб-нрп, wine-drinker. р-нрп to become wine. нрх n.m.f. small bird, chick.

Hee n.m. leek. CA N Hee leek-seller.

Ηλ: ωλ Ηπ: ωπ Ημπω: Μπο Ηπε: ωπ нис: Уис

61 (1)

ei, Q NHY (§22.3); imptv. AMOY (q.v.) vb. intr. to come, go; to be about to (+ Circum.); to . . . gradually (+ N + Inf.). Used with full range of prep. and directional adv. in normal senses. Note also the following expressions: ei epat to come to a superior; ei etN to come into the power of; ei exN to be applicable to; ei NCA to come to fetch; ei eBOA e to sue; ei eBOA 21TN to leave, quit (a place); ei e2PAI exN to befall.

EIA (ειερ-, ειατ", ειαατ") n. eye, mostly in cpds.: κτεειατ" to look around; με2-ειατ", μογ2 ν ειατ" to stare,
look intently (at: Μμο"); cμν-ειατ" εκν to fix eye on;
ταβε-ειατ", ταββε-ειατ", ταβο ν ειατ" εβολ to instruct, inform; τογν-ειατ", τογνε-ειατ" εβολ idem;
αι-ειατ" ε2ραι to raise eye; 2α ειατ" before one's
eyes; κω 2α ειατ" to intend to do. ναιατ" exclam.
pred. blessed is/are...! μντναιατ" blessedness.

- cialy, eiw, iw n.m. linen, linen garment. NGIALY, NILLY, NALY idem.
- CIABE, GIAABE, IAABE, IAIBE n.f. pus.
- CIAN, IAN, IHA n.f. mirror.
- ϵ_{18} , ϵ_{1618} , ϵ_{168} (pl. ϵ_{168H}) n.m. hoof; claw; stinger; nail, talon.
- cise, Q ose vb. intr. to thirst, become thirsty (for: $\overline{M}MO'$); as n.m. thirst.
- cie, ссie, сi (1) conj. introducing apodosis: then (§29.1); before neg.: unless, without; (2) conj. or; сiс...н either...or; (3) modal or interrog. part. introducing statement; translation depends on context: well then, so.
- CHERT, GEHRT, GIRT, GIRT, HAT, GERT n.m. usu. with def. art.: the east. c_{λ} -neight on the east side (of: \overline{N}).
- brightness. 16226 n. brightness, light.
- CIME, EIMME vb. tr. to know, understand, realize (e; that: xe). ATEIME ignorant; innocent, unaware; F-ATEIME (Q O N ATEIME) to become ignorant, unaware; to be unconscious; мNTATEIME ignorance. GINEIME knowledge. NAGT-GIME, NAGTHME, NAGTIME adj. presumptuous, impudent; obdurate, stubborn; мNTNAGTHME stubbornness, presumptuousness.
- Vb. tr. to bring (MMO'), bear. Used with many prep. and directional adv. in ordinary senses. Note the following expressions: GING MMO' EXN OF GZPAI EXN to liken something to, compare with; N-TOOT' EXN to seize; CING MMO' NCA to bring an accusation against; GING MMO' CEOA to complete; to extradite; to publish; to introduce; GING MMO' E THATE to recall, bring up (in one's mind); GING MMO' GZOYN introduce; as n.m. reception.

ness, aspect. MNTP6461N6 resemblance.

tine n.f. adze.

eine, ine n.f. thumb; big toe.

eine n.m. chain.

- eion, ion n.m. (1) sea (rare in Sah.); (2) winepress.
- croop, croope n.m. canal. xroop vb. tr. to ferry (ΜΜΟ')
 across (to: e); to cross, ford (a river: ΜΜΟ'); 61Nxroop fording, transit. xroop n.m. a ford, crossing;
 ferryboat; ferryman. ειερο, ιερο (pl. ειερωογ, ιερωογ)
 n.m. river; often spec. the Nile.
- 6100γN, 100γN (f. 100γN6) a title (m.f.); meaning unknown. 610n6 n.f. a liquid measure.
- with 2nd element to designate particular craft or its product, as in elem-ca sculpture, work in relief; elem-oyoele tillage, tilled land, produce of tillage; elem-ge woodwork; elem-noye goldwork; elem-n-2at silverwork; elem-got trade, trading, merchandise; p-elem-got to engage in trade. 2ap-elome adj. variegated. p-elome to spin; peqp-elome craftsman; mntpeqp-elome craft.
- eloya, eeloya, eleoya, loya n.m.f. hart, hind.
- GIPE P- (GP-) λλ' Q o vb. tr. to do, make, perform, produce, fashion (ΜΜΟ'); intr. to act, function, behave. For GIPG in cpd. vbs. see §26.1; these are listed under 2nd element. As n.m. doing, performance; peqGIPG doer, maker. τλφε-GIPG to do or make even more, increase in doing.
- EIC, EC part. behold, lo; here is/are... (properly only before nouns). EIC 2HHTE, EIC 2HHTE idem (before pron. or verb). EIC 2HHTE EIC idem (before n.).
 EIC 2HHNE idem (before n. or pron.). EICTE = EIC 2HHTE.
 EICNE = EIC 2HHTE.
- EIT \overline{N} , IT \overline{N} , ITNE, TNH n.m. ground, earth, dust; dirt, rubbish. E NEIT \overline{N} = ENECHT; \overline{M} NEIT \overline{N} adv. below, underneath, at the bottom.
- $ei\omega$ $ei\lambda^-$ (ix-) $ei\lambda\lambda^*$ Q eih (\pm $eBo\lambda$) vb. tr. to wash (\overline{M} Mo'); $ei\omega$ $eBo\lambda$ as n.m. washing; peqei ω $eBo\lambda$ washer (in bath).

GIA-TOOT' NCA to renounce, despair of (suff. is reflex.). CLW, GELW, GLOY, GOY (pl. GOOY, GOOY) n.m.f. ass, donkey. 612-N-tooy wild ass, onager. 612-200yt idem. стюрн (тюрн) Q сторн (± свох, сгрхг) vb. intr. to stare (at: e, $\overline{N}(\lambda)$, stare in wonder, be astonished, dumbfounded. ειωρέ (ειωρλε, ιωρέ) ειερέ- (ιλρέ-) ειορε (ιορε) vb. tr. to see, perceive (Mmo'); to look (toward: Ncx). As n.m. sight, vision; view, opinion; peacrop one who can see; MHTP6461@PZ perception: 61N61@PZ vision, power to see. ciepze n.f. ray (of light), sight (of eye). clut, lut (elt-; pl. elote) n.m. father (lit. and fig.); pl. parents, forefathers. Often used of abbots, elders and other revered persons. ATELWT fatherless. F-ELWT to become father. MNTGIOT fatherhood, family. WNειωτ relative on father's side. CIWT, GIOYT, IWT n.m. barley. clore n.f. dew. (1026, 1026 (6162-, 6102-; pl. 61220y, 61220y6) n.m. field. For 6162-620026, -ghn, -Bepsoper (280psp) see 2nd element. cT-61026, cT026 n.f. a field measure. g1-61026

to measure a field; as n.m. a field measure.

(1φε εφτ- (λφτ-) λφτ' (λφ') Q λφε vb. tr. to hang, suspend (Mmo"; on: e; by: NCA), all ± 62PA1. Q to be suspended; to be captivated (by: NCA); to depend (on: $z\overline{u}$); + eBOX: to overhang.

CI: CIC ELEOAY: ELOAY 610T6: 61WT 11X-: 610 eren-: erone 610Y: 610 1127; CIM 616P-: 61X 610YT: 61WT · IAARG: GIAB6 eleb-Roone: Roone elcue: elc 1 1 x p 2 -: 6 1 m p 2 616PO: 6100P elcte: elc I IAAT": GIA 616bmoA: 6100b 61T-: 610T I IAT": GIA elepse: elwpz 610: 61XXY ' 1750A: 61056 6162-: 61026 614T: 04T · IBH: GIB 6162-620026: 620026 ተነላቸ: ወ4ቸ 61H: 610 · IRT: GIEBT 61HBT: 616BT INIBG: GIABG · ICE: GIB 61MM6: 61M6 16226: 616262 CICIS: GIB EIMNT: EMNT 1HA: 61AA · 1686168: 616868 610PM: 610PM 1H4T: 616BT

KAIPE n.f. gullet.

KAKE n.m. darkness; F-KAKE to become dark.

KANA2H n.f. womb; belly.

kaneae, kaneeae, kanhae, keneeae n.f. wooden sounding board struck to assemble congregation.

KANKIN, KANKEN, KENKIN, 66361N, 6618661N n.m. wheel.

κλλωπογ, κλλοπογ, 6λλοπογ n.m.f. small dog.

KAM n.m. reed, rush.

кып n.m. (1) thread, string, strand; (2) letter (alph.).

кароус adj. curled (of hair); meaning not certain.

кас, кеес, кис, кис (pl. кеес, каас) n.m. bone; fruitstone. мар-кас, мер-кас n.m. bone-setter.

KAC n.m. carat (a coin).

KACE, KECE, KHCE n.m. shoemaker.

κλακα (κοακεα) vb. intr. to whisper (to: ε); as n.m. whispering. ρεγκλακα whisperer.

като n.f. boat, skiff.

каю n.m. reed, reed pen, reed staff or pole; n.f. = ма й каю place where reeds grow. † ммо є пкаю to fence with reeds.

KAWABEA n.m. earring.

кач n.m. trunk of tree.

κα2 n.m. earth, soil; the ground; land, country; γ-κα2 to turn to dust. ρΜΝκα2 a man of the earth.

κλ2κ \overline{z} κ62κ62- κ62κω2' \overline{Q} κ62κω2 vb. tr. to hew out, clear, smooth out (\overline{M} Mo'); to cause (a wound) to heal; intr. to heal. P64κ62κ62- hewer. κλ2 \overline{K} κ62 \overline{K} - vb. tr. = κλ2κ \overline{z} .

ква n.m. vengeance. ¬¬-ква, вірє м п(*)ква to do vengeance (for: ма*, мм, 2м). +-ква to avenge. хі-ква to take vengeance (on: ммо*, 2м); as n.m. retribution, compensation; вірє м пхі-ква, хі м пхі-ква to take revenge; речхі-ква avenger.

кво (ква) кве- Q кнв vb. tr. to make cool; intr. to

become cool; as n.m. coolness. \uparrow -kBo to make cool; x_1 -kBo to become refreshed, get coolness.

KG (1) adj. other, different; prefixed directly to noun, as in kepome, genkepome, nkepome, Nkepome. In some temporal expressions: next, as in TREPOMNE next year; again, in addition, as in N kecon once again; Kekoyi a little more, a little longer. (2) adv. also, even, moreover; positioned as in (1), but only with def. art. This usage has led to isolation of nke (f. TKG) as an independent element that may be prefixed to pronouns, THE ANON even we, or personal names, THE TAYLOG EVEN Paul, or used in vb. cpd. F-nke- before another Inf. or Q in sense "also, even to do or be." 66 pron. another (one), (the) other (one); pl. 26NKOOY6 some others, (\overline{n}) kekooye the others. ket (f. kete) pron. another; with def. art. the other. Keoya pron. another one. кеке n.m. child; n.m.f. (var. каке, кааке) pupil of eye. KGAGBIN, KENABIN, KANABIN N.M. AXE.

κελωλ, κογλωλ, κολολ n.m. jar, pitcher. κελοολε n. idem. κεξτε n.f. hip, loin.

кние, вние, венн n.f. vaulted place, cellar, canopy; palate (of mouth). Cf. Gk. μύπη, γύπη.

ким кемт- кемт vb. intr. to move, stir; vb. tr. to touch (є); to move, shift, stir (physically or emotionally: є, мно); as n.m. movement. аткім immovable. єїнкім movement. км-то n.m. earthquake.

KITE n.f. double drachma (half a stater), coin and weight.
GIC-KITE one drachma.

KIWOY Q to be fat, soft, weak; to be fertile, productive.
KNAA, KAGA, KAHA n.m. chain, esp. on neck.

клачт, касчт, кавт n.f. hood, cowl.

KAG, KGAH n.m. vessel for liquids.

καις, κα n.m. bolt; knee, joint. κελεηκες n.m. elbow; ελη-κελεηκες bolt-smith, smith. ελη-καλε idem.

KAO n.m. poison (for arrows).

клом n.m. crown, wreath, circle. †-клом to crown (ехп, 21хп). х1-клом to receive, bear crown; to become a martyr. речхро-клом victoriously crowned; чл1-клом crown-bearer.

KAOONE n.f. cloud.

KAOOME n.f. bruise.

 $κ\overline{x}$ Ψ n. a blow; \overline{p} - $κ\overline{x}$ Ψ νx°, $\frac{1}{7}$ - $κ\overline{x}$ Ψ ε $x\overline{n}$ to strike; $g\overline{c}$ - \overline{n} - $κ\overline{x}$ Ψ a blow.

кмкм, коукм, коумкм vb tr. to strike, beat (6: a musical instr.); to make a repeated sound. As n.m. drum.

кмом, Q кнм vb. intr. to become black. каме, камн (f. камн; pl. камауєї) adj. black; usually after n. with \(\overline{n}\), rarely without \(\overline{n}\). \(\overline{p}\)-каме (Q o \(\overline{n}\) каме) to become black. кме (?) = кмнме n. darkness. кнме n.m.(f.) Egypt; рмпкнме an Egyptian; мптрмпкнме Egyptian (lang.).

אואא n.m. sheaf.

κπικε vb. intr. to be fat, sweet; as n.m. fatness, sweetness. פּ-κπικε to become fat; γ-κπικε to make fat, to salve. anoint.

KNOC, KNOOC, KNOOC Q KON \overline{c} vb. intr. to become putrid, to stink: as n.m. stench.

 $\kappa\overline{n}$ те n.m. fig; в ω \overline{n} $\kappa\overline{n}$ те fig-tree.

 $\kappa \overline{n}_{2}\varepsilon$ n.f. architectural term, precise meaning not certain: porch, shrine, side (??).

коетс, каетс n.m. vessel for liquids.

коєїг, калг n.m. sheath, case, cover; brick-mold (?).

κοιλεκ, κιλεκ, χοιλ(2)κ, χοιλχ name of 4th Copt. month.

које, каје, каеје, кој n.f. field; pmntkoj farmer.

комме, коми, коми, киме, ким(м) e n.m. gum.

коонч, комч n.m. blight.

κοογ, κωογ, κλγ n.m. length of time; ογκογι \overline{N} κοογ a little while.

 κoo_2 , $\kappa\omega_2$ n.m.(f.) angle, corner; point, tip, prow; piece. $\kappaoc\kappa\overline{c}$ $\kappaec\kappa\omega c'$ eBox vb. tr. to lay out, extend ($\overline{m}no'$; also

reflex.); to entwine self (reflex.).

KOT. KAT n.m. basket.

κογι (κογ-) (1) adj. small, young; a little, few; used before noun (usually with N) or after (usually without N). May be cpd. as κογ-N. κογι N 2HT adj. impatient, easily discouraged. (2) adv., usually N ογκογι a little; (N) κεκογι yet a little, a little more; μηνικά ογκογι after a little while; 2λθΗ Ν ογκογι a little before; 9μπ ογκογι, πλρλ κεκογι almost, more or less; προς ογκογι for a little while; κλτλ κεκογι occasionally; N/κλτλ κογι κογι little by little; P-κογι (Q o N κογι) to become small, few, young; μηντκογι smallness, youth.

κογκλε n.f. hood, cowl.

κογμ(τ)", κογογμ(τ)", κογομ", κογωμ", κεμ" n. bosom, breast (suff. obligatory); also sometimes: genitals.

κογηγ n.m. a plant: lawsonia inermis.

KOYP n.m. pivot, hinge.

Koyr adj. deaf.

KOYXOY, KOYNXOY, KONXOY n.f. a type of vessel.

κρι, και n.m. a fragrant substance.

κρικο n.m.f. ash, soot, dust; ρ-κρικο to become ashes, dust. ρ-λτκρικο to leave no ash (on burning).

крыры vb. intr. to murmur, mutter in anger or vexation (against: exm, ысл, е, егоум е); as n.m. complaint, murmuring. речкрыкры murmurer.

крытс, крытс n.m. smoke, mist; darkness, obscurity.

hpo, κλλ (pl. κρωογ) n.m. shore (of sea, river); limit or margin (of land); hill, dale.

кромры vb. intr. to become dark (in shade or color); Q крырым to be dark. As n.m. darkness.

kroyr n.m. frog.

Kroyx, kpox n. a cake.

кгоч n.m. guile, deceit; ambush; as adj. false, guileful.

**TKPOЧ guileless. м\nткроЧ guile. \nable - кроЧ (Q o \nabla кроЧ)

to be guileful, lie in ambush (for: e); реЧ\nabla - кроЧ

deceiver, traitor. ca ν κροσ deceiver. x1-κροσ to use guile, lie in wait; x1 μμος ν κροσ to take by guile. κρωμ n.m. fire (rare in Sah.). κωρμ n.m. smoke (?). κοογρ, 20γρ, 600γρ n.m. finger-ring; key. ca ν εκοογρ key-maker.

KTHP n.m. calf.

кю ка- ках (кее", ке") Q кн vb. tr. to put, place, set (ммо"; with local prep. in plain sense); to appoint, make (ммо"; as: м); to obtain, get (ммо"; with reflex. dat. nx"); to preserve, keep; to allow, permit, grant (ммо"; to do: 6 + Inf. or Circum.; that: xe); to bequeathe (ммо"; to: nx"); to leave, abandon (ммо"); to go to (a place). Q to be situated, to lie; to be loose, unrestrained. ма м ка- a place for putting (something).

κω Μησ' εβολ (1) to release (to: Ναλ), loosen; (2) to expel, dismiss; (3) to forgive (w. νλ' of pers.); (4) to leave, abandon; (5) to omit, leave out; (6) intr. to become loose, dissolved; to become desolate, deserted. As n.m. forgiveness, remission; κλ Ν κω εβολ mercy-seat; ρεμκώ εβολ one who forgives.

κω ΜΜΟ επες Ητ to lower, let down. κω ΜΜΟ επλ 2 ΟΥ to leave behind. κω ΜΜΟ ε2 ΟΥΝ to put or bring in; to bring into port; εινκω ε2 ΟΥΝ entrance (to a house). κω ε2 Ρλι to put down, lower; to publish, expose, set forth; Q to exist, be, be extant; εινκω ε2 Ρλι nature, fashion, what is established. κω 218 Ολ to excommunicate. κω ΜΜΟ ΝΤΟΟΤ το keep, preserve, hold in esteem (suff. is reflex.); to entrust to (suff. is not reflex.).

κωδ κδ- (κ6β-) κοβ Q κΗβ vb. tr. to double, fold, close by folding (ΜΜΟ); intr. to double, become twice the amount; as n.m. double, double amount; repetition. κδβ ε n. fold, crease. κοοβ ε q, κωβ ε q n.m. doubling. κωβ 2, κοοβ 2 n. sinew, cord. β X-κωβ 2 to cut sinews;

B2, KOOB2 n. Sinew, Cord. Bx-κωβ2 to cut sinews; ρενωρετ-κωβ2 hamstringer.

- κωκ κεκ- κοκ (κλκ, κλλκ) Q κηκ (± εβολ) vb. tr. to peel, strip of, divest (ημος); intr. to peel, become bare; as n.m. barrenness, nakedness. κωκ ημος λεηγ to strip, make naked (obj. removed: Ν); Q κηκ λεηγ to be stripped, naked; as n.m. nakedness. κω ημος κλεηγ (κλλ κλεηγ) to strip, make naked. κογκε n.f. rind.
- кыл келп- колп Q колп vb. tr. to steal (ммо); as n.m. theft, stolen object; אדאשת inviolable. колп n.f. theft.
- κων καν καν καν νο. tr. to strike (ΜΜο΄), clap; to hammer in, fix; to knock (at door: e, e2ογν e); as n.m. blow, stroke. καν n.f. blow.
- κωλῶ κᾶχ- (6ᾶχ-) κολχ Q κολῶ (60λῶ) vb. tr. to bend, twist (Μησ'); reflex. to bow; intr. to bend, become bent; as n.m. perversion, depression. κᾶχ-πλτ, κᾶχλης to bow. κλλλχτωρῆ n.f. part of a house. κᾶχς n.f. corner.
- KONG (KOONG) KENG- KONG Q KONG vb. tr. to pierce, slay; as n.m. slaughter; peqkong slayer.
- κωπ, Q κμπ vb. tr. to hide ($\overline{M}MO^s$); intr. to be hidden; as n.m. concealment. Rare in Sah.; use 2ωπ.
- κωρ n. measure of money.
- кωνώ (εωνώ) κενώ- κονω vb. tr. to request, persuade, cajole (ε); as n.m. entreaty, persuasion; γεчκωνώ flatterer; міτρεчκωνώ flattery. κονω n.m. flatterer.
- र्षा (бшрव) Q корव vb. tr. to bring to naught, destroy, cancel (ммо); intr. to be idle, deficient.
- kwT кєт-кот Q кнт vb. tr. to build, form (ммо); to edify, encourage (ммо); intr. to become edified; as n.m. act of building; a building; rule, precept. ма м кwт workshop. речкwт builder. хі-кwт to receive edification. екwт (pl. екоте, екате) n.m. builder, mason;

potter. секют, сікют n.f. potter's workshop.
кюте кет- кот' Q кнт (1) vb. tr. (a) to turn, direct
(ЙМО'); + евох to turn sthg. away; + епхгоу to turn

(2) vb. reflex. to turn (self) around, to return; to repeat, do again (+ e + Inf. or + coord. vb.); + eBOA to turn away; + eNAZOY to turn back, return; + eZOYN e to return to; + eZPAI e to return to.

sthg. back: + 620YN e to convert to, bring around to.

(3) vb. intr. to turn, rotate, revolve; to circulate, go or move in a cyclical way (e.g. watch, visit); to visit (e); to go around, form circle; κωτε exN to circulate among; κωτε мN to consort with, stick with; κωτε NCA to seek, go about seeking; κωτε GBOA to go away; to turn, return; κωτε e20γN to turn or incline inward; κωτε e2γA1 to turn around. κωτε ε to surround.

κωτε n.m. (1) turning, circuit; (2) surroundings, environment; (3) seeking, inquiring; $\overline{M}/2\overline{M}/6$ πκωτε adv. round about; $\overline{M}/2\overline{M}$ πκωτε \overline{N} , $\overline{M}/2\overline{M}$ π(°) κωτε prep. around, in the neighborhood of; about, concerning. κοτ n.m. circular motion, turn, visit; \overline{p} -κοτ, \uparrow -κοτ to make a turn, make a visit. κοτ n.m. wheel. κοτ \overline{c} n.f. circuit, turning; a turn, bend; knot, twist; crookedness, guile; $\overline{e}_1 p e_1 \overline{N}$ ογκοτ \overline{c} , \overline{p} -κοτ \overline{c} to make a turn, to circumvent; \uparrow -κοτ \overline{c} to make a circuit, circulate; x_1 -κοτ \overline{c} to be crooked; $c_2 \overline{N}$ κοτ \overline{c} a guileful person; $\overline{M}\overline{N}$ Τ \overline{c} 2 \overline{N} κοτ \overline{c} 3 guile, dishonesty.

кто кте- кто" Q ктну (ктоеіт, ктаєїт) vb. tr. to cause to turn (ммо"; to: e); this verb has the same range of meanings as кюте above, including reflex. and intr. uses; as n.m. turning, return; аткто" irrevocable; ммтречкто good conduct.

кштч кетч- котч Q котч vb. tr. to gather (ммо").

кшшве (кшшче) кееве- (кааве-) коов vb. tr. to force, compel, seize by force (ммо"). ква n. compulsion, forced labor; у-ква to do forced labor.

- κωωρε κεερε- (κερε-) κοορε vb. tr. to cut down, chop down (ΜΜΟ'); intr. to be cut down.
- κωως (κωως κωνς) κοος (κοονς κοος) Q κης νδ. tr. to prepare (a corpse: Μησ) for burial; as n.m. burial, funeral; corpse. ρα η κωως α 20 γν raiser of the dead, necromancer. Γ-κωως to become a corpse, die. καις, καις, καις, καις, καις, shroud; (3) corpse. ς καις εffigy.
- κωωφε, Q κοοφε vb. tr. to break, split (Μπο*); intr. to become split, broken. Rare in Sah.
- κω2, Q κH2 vb. intr. to become jealous, envious (of: 6); to become zealous, eager; to emulate, try to equal (6); as n.m. envy, zeal. ρε4κω2 zealot; rival, imitator.

 --κω2 to cause (Νλ*) to envy etc. (6). κοι2ε n.f. rival woman.
- $(\kappa\omega_2)$ $\kappa\varepsilon_2$ $\kappa\lambda_2$ Q $\kappa\kappa_2$ vb. tr. to level, smooth out $(\vec{n}mo^*)$; to tame, accustom $(\vec{m}mo^*; to: \varepsilon)$.
- $κω_2\overline{\tau}$ n.m. fire. \overline{p} - $κω_2\overline{\tau}$ (Q o \overline{N} $κω_2\overline{\tau}$) to become fire. \uparrow - $κω_2\overline{\tau}$ to set fire (to: ϵ).

Κλ-: ΚΦ **Κλλ**^{*}: **Κω** KAABET: KOOBE **KAAK*: KOK** KAAKG: KGK6 KAAM: 6WM KAAC: KAC KAGIG: KOIG KAGIC: KOGIC KAIE: KOIE KAICE: KOOG KA12: KO612 KAK": KOK KAKE: KEKE KANABIN: KENEBIN ^እ እ እ ኔ ተ : 6 እ እ ል 2 ተ KANHAG: KANGAG KANWOY: 6061X6 KAHAYGI: KMOM KAHAYAG: GAMOYA KAME: KMOM канн: кмом KAHHAG: GAMOYA

каноочав: баноча KANKAW: GINGAW кап: 60п **Κλπ*: 6ωπε** KANIXE: 6ANEIXE KAPAGIT: PO **የአ**ያወ4: PO KAC: TKAC кат: кот KAY: KOOY KAYNAKEC: 6WNA6 KAYON: 6AOYON КАЧКАЧ: бАВбАВ KA2HY: KWK KAZK: KAZKZ KAXII: 6AXII KBA: KWWB6 KBA: KBO KBBE: KWB KBET: KBO K6*: KW

KEET: KW

KEEBET: KOOBE кеере-: кооре KEEC: KAC KEICE: KAICE **Κ6λ: 6ωλ** KEXEEXE: KAXEXE KENENKEZ: KXNE кехн: кхе KEAKIA: KAAKIA KENKWN": 60N6\(\overline{N}\) KEAAHC: XIAAEC келы: 67м KEAMA: 6EAMAI κελοολε: Κελωλ KENOIT: 60GING KEMT-/": KIM **κεη**": κογη(τ)" KEOYA: OYA кеп-, кеп: 6шпе κερε-: κωωρε KECE: KWWC, KACE K€T=: KWT, KWT€ кет: ке

KETE: KE K62K-: KA2K2 KEZKEZ-: KAZKZ K62KW2 (*): KA2KZ кн: κω KHB: KBO KHM: KMOM KHM6: KOMM6, KMOM **КНП: 600П6** KHC: KAC KHC: KOOC KHC6: KAC6 KHT: KOT, KOTE KIA2K: KOIA2K KIRG: GKIRG KIEBE: EKIBE KIN-: 61N-KINBHA: 6INOYHA KIC: KAC $K\overline{\lambda}$: $K\overline{\lambda}\lambda\epsilon$ **KXX:** 6XX каа: кро KABT: KAA9T **KAEA: KAAA** KA64T: KAA4T кана: кааа KAL: KPI **KAOF6: 6A0066 K**₹26: **K**₩λ₹ KME: KMOM кмиме: кмом киме: комме KMTO: KIM KN: XIN

KNNOYT: 6 INMOYT KNOOC: KNOC KNT': 6INE KNOOC: KNOC KO126: KO2 **ΚΟλ': 6 Φ λ** ΚΟλΟλ: ΚΕλΦλ **ΚΟλΠ(*): 6ωλΠ** KONTE: KONT KOM: 60M комч: коомч KONXOY: KOYXOY KOOB : KWOBE KOOBET: KOB KOORT: KORT KOONC': KWWC KOOPE': KWWP6 KOOC': KWWC KOOYE: KE ΚΟΟΦ6: ΚΦΦΦ6 κοπ': 6ωπε корте: борте короч: коро KOP4*: 60P4* KOC: 6OC KOT": KWT, KWTE кот: коте KOTC: KWT6 KOYK: 60YX KOYKE: KOK коукі кі кі кі ΚΟΥλωλ: ΚΕλωλ

KOYON": KOYN(T)" KOYOYN(T)": KOYN(T)" KOYWN*: KOYN(T)* KOYX: 60YX KOX*: 606 KPMNTC: KPMTC KPO2: 6PW2 крож: кроуж KPWOY: KPO Kpww6: 6p06 **KPW2: 6PW2** KTAGIT: KOTE KTET: KOTE ктнү: коте кто(*): коте ктоеіт: коте KYAMAN: 66AMAI KOBE4: KOB κωλ: 6ωλ KWA6: 60WA6 κωλπ: 6ωλπ KONT: KOOC KWOY: KOOY κωπ', κωπε: 6ωπε кшрі: кршм KOPZ: 6WPZ **KWTZ:** 6WTZ KWWNT: KWNT KOO46: KWWB6 KW2: KOO2 XOIAX: KOIA2K XOIA (2) K: KOIA 2 K

λ

коүмкй: кйкй

KOYNXOY: KOYXOY

אג n.m. envy, slander. אאדאג slander. נו-גג to slander (e); as n.m. slander; אר slanderer; אאדף פּקנוראג slander; אגר eager for slander.

nne; something; may take article as n., e.g. ογλλλγ ghm a little something. κελλλγ any other. λλλγ nim everyone, everything. (2) as adj. any (usu. bef. n. w. π); λλλγ Μπωτπ any of you. (3) Neg. context: none, no one, nothing. (4) As pred. ογλλλγ, 26Νλλλγ = nothing, no one, even when neg. is not present. (5) λτλλλγ π prep.

lacking, without; (\overline{N}) xxxy adv. (not) at all.

AABOI, AABAI n.f. lioness; she-bear.

AAIN, AAGIN, AGGIN n.m. steel.

pieces; F-AAKM AAKM to break or tear into pieces.

NAKOOTE, NAKOTE, NAAKOTE n.f. a liquid measure (wine). $\lambda \lambda \kappa \overline{\lambda}$ n.m. corner, edge, extremity, top.

to apply (paint, overlay: ΜΜΟς; to: ε); to paint, smear.

λακατπ, λακατπ, λακατ, λακατ n.m. tar, pitch.

xac n.m. tongue; language (also acne \overline{n} xac); any tongue-shaped object. xac cnay deceitful; m \overline{n} Txac cnay deceit. xac n.m. tow, flax.

ALTEG, ALTEGO n.f. a patch; 21-ALTEG 6 to put a patch on.
ALYO, ALEW n.m.f. sail; curtain, awning. 610-ALYO n.f.
half-sail.

λλφλης (pl. λλφηηγ, λλφητογ) n.m. village magistrate. λλ2μ n.f. a liquid measure.

ALGE vb. tr. to remove, cause to cease (6).

λελογ (pl. λελλγε, λλγε, λλλγ) n.m.f. young man or woman. Λεκκιμφε n.m. warrior, champion.

ментии, женони, жанони n.m. saw.

AGON n. earring, bracelet.

λεΨ, λιΨ n.m. person afflicted with eye-disease.

λειικέ, λεψε, λιπόε n.m. fragment.

λοηλισε n.f. crumb, fragment.

ver n.m. care, anxiety.

λειλως Q to be high, tall. λλιλει n.m. haughtiness.

NING AGBT' Q AOBG (AAB-) vb. intr. to be mad, rage (at: GROYN G, NCA; from: NTN, 2A, 2N, 21TN); rarely tr. to make mad. AAB-AA see AA. AAB-MA2T gluttonous. AAB-CRIMG lecherous. AAB-2H greedy; MNTAAB-2H greed; P-AAB-2H to become hungry, greedy.

NIKT' in P-λικτ' to veil, cover; π λικτ' prep. covering.

Alaooze, Aeawze, eaooze, Aoze n.f. gum resin (or tree).
Almnh n.m. portrait, image.

TARB, GAARB, TARG n. festing, buffoonery.

xo (imptv. xxok, f. xxo; pl. xxot \overline{N}) vb. intr. (1) to cease, stop, come to an end, be terminated; + Circum.: to stop doing, no longer do. (2) to leave, quit, depart (from: \overline{M}MO', 2x, 2xbox \overline{N}, 21, 2\overline{N}, 6Box 2\overline{N}, 21\overline{N}, 21\overline{N}); sometimes + untranslatable \overline{M}MAY. x-N642HT xo \overline{M}Mo4 he fainted.

AO126 n.m. (f.) mud, filth.

אסופּפּ n.f. cause, excuse, reason; אַדאסופּפּ without cause. ל-אסופּפּ אַגְיּ to provide excuse or occasion to. פּאּר-אסופּפּ to find excuse. פּאּר-אסופּפּ idem.

xok, xo6 n.m. cup, bowl; also as measure. ωπ-xok idem.
xokxκ x6κxωκ Q x6κxωκ vb. intr. to become soft; rarely tr.
to make soft, smooth (Μmo°); as n.m. softness. xxκxxκ
n. a kind of confection.

ADOME, ADYME, MODAE n.f.m. bait.

λοογ, λοογε, λωογ, λλγ n.m. curl; fringe, hem; cluster. λοο4ε, λοο8ε, λοβε Q to be decayed, about to collapse.

λογλλι n.m. shout. ωg/εg-/Νεχ-/Τωκ λογλλι εβολ to shout.

хочх $\overline{\Psi}$ (хочхеч, ховхеч) х६чх $\Psi\Psi$ Q х६чх $\Psi\Psi$, х६чх Ψ (± ६४०х) vb. intr. to rot, perish by decay or corruption; vb. tr.

to destroy, cause to rot (mmo'); as n.m. decay, rot.

xoxxx (xo6xex) xexxwx* (xe6xw6*) Q xexxwx (xe6xw6) vb. intr. to languish, be sickly; vb. tr. to make sick (ΜΜΟ*); as n.m. sickness.

λο6, λλ6 n. in \overline{p} -λο6 e to importune; $m\overline{n}$ τλο6 persistence. λο6λ66 n.m. girder, frame, joint.

хово, Q хово vb. intr. to glow red-hot; tr. to heat red-hot (ммо'); as n.m. glow.

λωκο n.m. crown, battlement; as vb. tr. to crown, adorn. λωκ, Q ληκ vb. intr. to become soft, be fresh. λωκο (λω6ο, λωχ, λογχ) λχ- λοκο vb. tr. to bite, stab,

- pierce ($\overline{N}MO'$); + $\overline{N}CA$: to bite or snap at; as n.m. bite. PERAWKE biter, biting. $\overline{QC}-\overline{N}-\lambda \overline{QK}$ piercing blow.
- λωκώ, Q λοκώ vb. intr. to be weak, ineffectual; as n.m. weakness.
- NUME (NAMEC) Q NOME vb. intr. to become foul, to stink; as n.m. foulness, putrescence.
- NOTE (NOOTE) vb. intr. to become hard, callous (of skin).
- λωωμε (λωωμ, λωμ) Q λοομε (λομε) vb. intr. to wither, fade; to become filthy, dirty, muddy; as n.m. filth; withered appearance. λτλωωμε unfading. Q also λλλμ.
- Nowc (Noc) Nec- Q NNC(6) vb. tr. to crush, bruise (ΜΜΟ΄); vb. intr. to become crushed, bruised.
- νω2 π λ62 m- λ02 m' Q λ02 m vb. tr. to boil (mmo*); vb. intr. to be boiled. λλ2 m6c n. boiled food (?).
- \wx λεx- λοx' Q λοx vb. tr. to crush, bruise (Μ̄μο'); to
 lick (Μ̄μο'); vb. intr. to be sticky, adhesive; to stick
 (to: ε, ε₂ογη κ̄).
- λωχκ (λωχε, λωχτ) λοκκ (λοκε, λοκκ, λοκτ) Q λοχτ vb.
 intr. to become sticky, adhesive; to stick (to: 6); vb.
 tr. to stick, join (Μηος; to: 6); also to lick.
- λωχ̄ λεχ̄- λοχ₂* Q λοχ̄ (1) vb. tr. to crush (Μ̄μο*); intr.
 to be crushed, effaced; as n.m. anguish, oppression;
 (2) vb. tr. to lick (Μ̄μο*).
- $\lambda \omega 6 G$ $\lambda 6 G$ λ

\AAKOTG: AAKOOTG AANOHN: AGNTHN /**ሃ**ጆን። ሃይሃዐላ AAOYE: AAAY YYYC(E): YOOC **XAY: X00Y** \AB=: A186 AAY6: A6AOY, AAAY 10841: XX801 አልወነዘ: ወነልነ \ARW: AAYO ΑλΦΝΗΥ: ΑλΦΑΝ6 $\lambda \lambda K \lambda \lambda K$: $\lambda O K \lambda \overline{K}$ **λλ2λ62: λ62λω2** /YYHA: YYYE AAZM6C: AWZM _ሃ ሃ ሃ ሃ ሃ ሃ ሃ ፅ TILXMER : TILEXE AYOOA: YYYE **AA6:** A06 \\\\(\o)': \\\6 AA6NT: AAKNT WHEC: YOUR AGBT": AIBG

 λ661N:
 λλ1N

 λ66λ-:
 6λ00λ6

 λ6λ-:
 6λ00λ6

 λ6λλγ6:
 λ6λ0γ

 λ6λω26:
 λ1λ0026

 λ6C-:
 λωωC

 λ6T:
 λλΤ

 λ64λ04T:
 λ04λT

 λ66-:
 λω66

 λ1λ:
 P1P

20076: 7776
20086: 70086
2008: 70086
2008: 70086
2008: 70086
2008: 7008
2008: 7008
2008: 7008
2008: 7008
2008: 7008

30086: 30046

λωΜ: λωωΜ6 λωC: λωωC λωΟΥ: λΟΟΥ λωωΤ6: λωΤ6 λωϫΤ: λωϫΓ λωϫΓ λωωΓ λωνΓ λωνΓ λωνΓ λωνΓ λωνΓ

λ06λ6Χ: λ0Χλ6Χ

M

MA n.m. place; often in spec. senses: dwelling-place, temple or shrine; neima this world; nkema the other world. n(*)ma ne it is (one's) lot or duty (to do: ε). For cpds. of ma N see 2nd element. ε nma N prep. to, toward; regarding, concerning; instead of, in the place of. εγma to one place, together. κατα nma in various, different places. Φα neima so far, up to now/here. 2α nma N as regards. MA NIM everywhere. κα-(n)ma Nα* to give an opportunity to. P-nma N to take the place of, succeed. +-ma Nα* to allow, permit, give opportunity to. xi-ma NTN to usurp the place of. εN-ma to find opportunity. See also § 23.2.

MAN MA- MAT' (MHEI") imptv. of +, q.v. See also §26.3.

MANE (MANE-, MAE-; f. MANE) number: thirty. See §30.7.

MANY, MAY n.f. mother; also fig. and as title. g\overline{N}-MANY,

ghn \overline{H} MANY child having same mother as another. AT
MANY motherless. \overline{P}-MANY to become mother.

MALKE n.m. ear; handle. κ_{A} -MALKE 6, PIKE \overline{M} NHALKE 6 to give ear to, incline ear to.

MAAXE, MAXE (MAX-) n.f. a dry measure.

magin n.m. sign, mark; wonder, miracle. ¬—маgin to become marked, remarkable; to indicate (g). речетре м пмасти wonder-worker.

†-масти to indicate, point at, signify (g); to give a sign (to: ма*); реч-масти augur.

XI-MAGIN to practice divination, augury; as n.m. divination; peaxi-magin augur, diviner; mutpeaxi-magin augury, divination.

MAKOT, MAKWT, MAKA(A)T, MARAA n.m. lance, javelin.

מאנד, אאχ, אסגע n.m. neck. ל-ח(*)אאגע נא to submit to.
אאמיד-אאגע adj. stiff-necked; אאזראאסיד-אאגע stiff-neckedness; ף-אאמיד-אאגע to be stiff-necked.

MANGARE, MANGARH, MANXARE n.m. pick, hoe; winnowing fan. MAPOYOGE, MEPOYOGE, MPOYOGE n.f. jawbone.

HAPXOXE (pl. MAPXOOXE) n. name of woman's garment.

нате in емате, mmare adv. very much, greatly; only.

мате (маате, мете) Q матфоу vb. tr. to reach, attain, obtain, enjoy (ммо'); intr. to hit the mark, be successful (in doing: e, м + Inf.); as n.m. success. —мате = мате tr.

mator, matoer n.m. soldier. \overline{p} -mator (Q o \overline{n}) to become a soldier. $\overline{m}\overline{n}$ tmator soldiering, warfare.

HATOY n.f. poison. BAK-MATOY poisonous, venomous.

אבץ ', אבץ ', אבץ ', mayar' intens. pron. self, self alone, ownself; used appositionally to preceding n. or pron.; see §28.3. אאשני n.f. balance, scales.

мащо in вмащо adv. very, greatly. Миащо idem.

MAGET, MAGEPT n.m.f. cable.

MA2, MAA2 n.m. nest, brood. MA2-ογΑλ, -ΒΑλ, M62-ογΗλ n.m. idem.

MAZE n.m. cubit. GIC-MAZE half cubit.

MARG n.m. flax. eqpa-mare linseed.

 $^{\text{MA}2}\overline{\text{T}}$ n.m. bowels, intestines. Me2T-0 great intestine.

Maxe n.m. axe, pick.

MAXKG, MIXKG, MEKXG, MIXGG n. a woman's garment.

мес, мес, мне n.f. truth, justice; freq. as adj. true, real, genuine; truthful, righteous. млтме truth, righteousness. маме adv. truly, in fact. 2N оүме idem. рммме an honest person. Р-(т)ме to become true, verified.

ме (меі) мере- меріт (р.с. маі-) vb. tr. to love, desire, wish (ммо); мере- may be used with another Inf. goy-меріт worthy of love. For cpds. with маі- see 2nd element. As n.m. love. меріт (pl. мерате) adj. beloved. мере n. midday, noon. м меере at noon.

NEGYG (MGYG, MGGY) vb. intr. to think (about: 6; that: x6), often w. ε as reflex. or ethical dative; to be about (to do: N + Inf.); as n.m. (± N 2HT) thought, mind.

MGGYG G2OYN ε to plot against. MGGYG GBOX to ponder, consider. MNTATMGGYG absence of thought. PG4MGGYG ONE who thinks. †-(n)MGGYG NA* to remind. P-n(*)MGGYG to think of, remember (N); as n.m. remembrance.

мелют (pl. мельте) n.f. ceiling, canopy.

мерег, мегр n.m. spear, javelin. gc-N-мерег thrust of spear. чал-мерег spear-bearer.

местигнт, месент n.f. breast, chest.

Mec 2 wa n.m. a file.

месорн, месорн, месоурн name of 12th Coptic month.

меще- мещь vb. not to know; usu. in меще-мім so-and-so, such-and-such; мещьк, мншык adv. perhaps.

MEGITIET n.m. hinge of door.

мегро n.m. manure; реч+-мегро one who manures. межпюме, межпююме, мефпюме n.m.f. ulcer, eruption.

местох n.m. tower.

мн, мі n.f. urine; мн м мооу idem. мн обік excrement. F-мн to urinate; to defecate. мы м F-мн anus; latrine. мннює n.m. crowd, multitude; as adj. many, great, much.

мниє, мнине in m мние adv. daily, every day. m мние (m) мние idem.

мнр n.m. shore, opposite shore (not properly Sah.).

мнт (f. мнтв) number: ten. млт- prefix for 'teens; see §24.3. соү-мнт tenth day. рв-мнт (pl. рв-мьтв) a tenth part, tithe.

MHTG, MHHTG n.f. middle. G TMHTG to, into the midst of (\overline{N}) , between; adv. forward, to a position in front.

- $2\overline{N}/\overline{N}$ TMHTG in the midst (of: \overline{N}); between; at the front. $6BO\lambda$ $\overline{N}/2\overline{N}$ TMHTG from the midst of (\overline{N}), from among. 21 TMHTG in through the midst (of: \overline{N}). MAP-MHTG n.f. belt. MH2G, MG2G n.m. feather.
- mike vb. intr. to rest; also reflex. (with \(\text{MMO}' \)); as n.m. rest. \(\dagger-\text{mike na'} \) to give rest to.
- MINE, MEINE n.f. kind, sort, species, quality, manner. MINE \overline{N} adj. sort of, kind of, manner of; kemine \overline{N} other sort of; mine nim \overline{N} every sort of; \overline{N} mine \overline{N} what sort, what kind of? \overline{N} Teimine of this sort, as follows, thus.
- мио° pred. with 2nd pers. suffixes: миок, мию, миют Наle! Be well! Greetings!
- MICE MEC($\overline{\tau}$)- (MAC-) MECT' (MACT') Q MOCE; p.C. MAC-, MECvb. tr. to bear (mmo'), give birth to; Q to be newly born; as n.m. offspring; giving birth. As 2nd member of cpd.: born, as in GANG N MICE born lame; birth-, as in MA H MICE birth-place, 200Y H MICE birthday, OFN-Hмисе first-born child; митфрп-(м)мисе status or right of first born. MICE 62PAI, + 6 MICE to bear, bring forth. peamice one who bears; MNTpeamice bearing, birth. ATMICE unborn. MHCE n.f. pregnant woman. MAC, MACE n.m. young animal; esp. bull, calf; MNTMACE likeness of a calf. MHCG, MHHCG n.f. usury, interest; + E MHCE to lend at interest: x1-MHCE to take interest; ATMHCe without interest. MGC-2N-H1 n.m.f. one born in household. Mecio Mecio vb. tr. to bring to birth, act as midwife for. Mecio, Mecio n.f. midwife; F-Mecio to act as midwife. For cpds. with MAC- see 2nd element.
- Mige, метge vb. intr. to fight, struggle, quarrel (with, against: мм, оуве, е; for, on behalf of: ехм, егры ехм) to attack (е); to strike (upon: ехм); as n.m. quarrel.

 мы м міge arena; речміде fighter; р-речміде to be hostile, quarrelsome; єтміде art of fighting.
- Πκλ2, Q Μοκ2 vb. intr. to become painful, difficult; to be
 in pain, grieved (in: 6); Q to be difficult (to do: 6,

- \overline{N} + Inf.; etpe); as n.m. (pl. \overline{M} KOO2) pain, difficulty, grief. \overline{P} - \overline{M} KA2 to become pained, grieved, difficult. $g_{\overline{N}}$ - \overline{M} KA2 to suffer pain. \overline{M} KA2 \overline{N} 2HT vb. intr. to be pained or troubled at heart; as n.m. pain, grief. $\frac{1}{7}$ - \overline{M} KA2 \overline{N} 2HT to grieve, vex (NA*). MOK2 \overline{C} , MOX \overline{C} n.f. grief.
- אב (pl. אבססי) n.m. battle, -array, troops; quarrel. בּקּ-אָר (Q אַבּאב chp) פּפּסא to set up battle-array. בּוּ-אַבּאב to fight; אַפּיאַבו-אַבּאב fighter.
- MMAY adv. there, in that place; from there, therefrom; thence. 680% MMAY thence, from there. 6MAY thither, to there. Sometimes without translation value (§22.1).
- MMA2 prep. before (a deity; in making offerings).
- MMIN MMO' intens. pronoun, appositional to a preceding pron., as in пані Ммін Ммої my own house. See §28.3.
- Mmon adv. or conj. for, for surely.
- м $\overline{\mathbf{n}}$ (archaic $\mathbf{n}\overline{\mathbf{m}}$) $\mathbf{n}\overline{\mathbf{m}}\mathbf{m}\mathbf{a}^*$ (1) prep. with, together with, in the company of; (2) conj. and, usu. joining nouns; sometimes $\mathbf{a}\mathbf{y}\mathbf{w}$ $\mathbf{m}\overline{\mathbf{n}}$.
- мм-, ммм- pred. of nonexistence: there is/are not (§2.2); used before indef. subj. in Present System (§18.1); for мм-сом, мм-сом see сом.
- Mmon, mon neg. part. no (in answer to question); (едопе)
 Mmon adv. if not, otherwise; xN Mmon, xIN Mmon or
 rather, rather than.
- \overline{M} NOYT (f. \overline{M} NOTE, \overline{M} NOTE) n.m.f. porter, doorkeeper. \overline{M} n.m. a grain-measure.
- MNT- prefix (f.) for forming abstract nouns; see §27.2.
 MNT- prefix for forming 'teens; see §24.3.
- ммть- ммть neg. of pred. of possession; see §22.1. Also used as nonliterary vb. prefix: lest, that not, unless.
- ммтре, метрн (pl. ммтресу) n.m. witness, testimony. ммт-ммтре n.f. testimony; р-ммтре to testify, bear witness; to testify (about: ммо°, етве, ехм; to a person: мм°; against: є; for, in behalf of: є, 2λ, мм).
- мо imptv. vb. (sing. мо, мо, ймо; pl. ймнетт) take!(e).

- modit n.m. road, path; rarely: place. Modit N di decoyn entrance; Modit N di dbox exit. xi-Modit entrance; to lead, guide; pay-Modit leader, guide; xxy-Modit idem; MNT-payxy-Modit leadership; p-xxy-Modit to be leader.

 Modit N Mooge track, path.
- MOGIZ, MOIA2 n.m. name of a measure.
- Mosize, Moize n.m.f. wonder; \overline{p} -Mosize to wonder, be astonished (at: \overline{N} , $6 \times \overline{N}$, $2 \overline{N}$).
- мокмек мекмоук vb. intr. to think, ponder, meditate; to intend (to do: етре); reflex. idem, to consider (that: xe). мокмек евох е to reflect on, ponder. As n.m. thought. хтмокмек unthinkable, inconceivable (еро). мооме n.f. nurse; as adj. foster-.
- moone mene- (mane-, manoy-) Q manooyτ (± ε2ογη) vb. tr. to bring into port, bring to land (mmo'; to: ε); vb. intr. to come to port, moor (to: ε). MA M MOONE harbor.
- moone мене- маноу" (маноуоу"), p.c. мане- vb. tr. to tend, feed, shepherd (ммо"); to feed on, devour (ммо"); vb. intr. to feed, graze (subj. cattle). ма м мооне pasture. речмооне shepherd; ммтречмооне shepherding. мане (ман-; pl. манну) n.m. herdsman, shepherd. For ман- in cpds. see 2nd element.
- Mooy (pl. Moyeih, Moyheie, Moyelooye, MoyNelooye) n.m. water; spec. the Nile inundation. In cpds. may mean juice, exudation, semen, urine. Atmooy waterless. Mec-mooy water-containing. Me2-mooy to draw water; MA M Me2-mooy place to draw water; peqMe2-mooy water-drawer. F-mooy to become water, liquify. Cek-mooy to draw water. CF-mooy to distribute water. +-mooy to give water; MA N +-mooy water source. TCE-mooy to slake. BAI-mooy water-bearer. XI-mooy to receive water. 21-mooy to rain.
- Mooge (Moge) vb. intr. to walk, go; used with many preparate and adv. in normal senses; as n.m. going, journey. Note Mooge \overline{N} to consort with; Mooge \overline{N} to be in the

following of. ATMOOGE pathless; ма й нообе road, path; моетт й нообе road, journey; 2тн й нообе road, path; р-2тн й нообе to go, walk. 2007 й нообе day's journey. морт n.f. beard. ATMOPT beardless. р-морт to grow beard. мосте месте- месте" (р.с. маст-) vb. tr. to hate (йно"); as n.m. hatred, object of hatred. маст- in cpd. hater of. фоу-мосте deserving of hatred. месте (f. местн) n.m. hated person.

моу, Q мооут vb. intr. to die (of: eтве, NTN, 2A, 2N, 2ITN; for: exN); as n.m. death; plague, pestilence. речмоу adj. mortal, dead; митречмоу mortality. речмооут dead person or thing. пад-моу adj. half-dead. атмоу immortal; митатмоу immortality.

Moye, moyer, moy, moye n.f. island (usu. in Nile). Moye, moyer n.m.f. lion(ess); mac $\overline{\text{M}}$ moye lion cub. Moyk vb. tr. to destroy; intr. to be destroyed.

MOΥΚΣ ΜΕΚΣ- ΜΟΚ2" vb. tr. to afflict, oppress (ΜΜΟ"); reflex. to be afflicted, oppressed; to humble oneself.

MOYA \overline{z} MGA \overline{z} - MOA \overline{z} Q MOA \overline{z} vb. tr. to make salty; to convert to salt (\overline{N} MO'); Q to be salty. M \overline{x}_2 , MEA \overline{z} , MHP \overline{z} n. salt. MGAz6 n.f. saltiness.

MOΥλ2, MOΥλλ2, MOΥΡ2 n.m. wax; candle; honey-comb.

moγx2 mox2 °Q mox2 vb. tr. to involve, enmesh (mmo °); vb. intr. to become hooked into, attached to (g, mmo °, 2 π).

MOYN, Q MHN (MHNG) vb. intr. ± 680λ to remain, last, endure; with Circum.: to continue doing. As n.m. (± 680λ) perseverance, continuing; 2Ν ογΜογη 680λ continuously.

moynk (moynt) menk- monk (mont) Q mont vb. tr. to form, fashion, make (mmo); as n.m. thing made; formation, fashioning; fashion, make; moynk n six handmade objects; λτμογηκ n six not handmade.

моуоут меут- мооут vb. tr. to kill (ммо); речмоуоут killer. моур мер- (мр-) мор Q мнр (р.с. мар-) vb. tr. to bind, tie (ммо ; to: ммо , с, схм, гм; with: ммо , гм); моур ммо м песхных to gird in monastic habit; to bind by

oath, adjure; Q to be bound, girt. As n.m. band, strap, girding. MOYP MN to be at enmity with. MA M MOYP prison. MAP, MAAP, MGP, MHP n.m. bundle. MAIPG, MHPG n.f. idem. MPPG n.f. chain, bond, joint. MOPG n.f. binding, restriction; purse. For cpds. with MAP- see 2nd element. MOYC n.m. strap, band; belt, girdle; thong. PGTTAMIG-MOYC strap-maker.

MOYCK MACK O MOCK vb. tr. to strike (MMO).

MOYT, MOTE n.m. sinew, nerve; joint; neck, shoulders.

моγтє vb. intr. to call, name (ε, rarely ммо°); see Vocab.

17 for usage. As n.m. call, incantation. речмоγтє enchanter; ммтречмоγтє enchantment. моγтє єхм/оγве to call upon, to; моγте євох to call out; to summon. моγтє є
20γN to call in (to: ε); to invite in.

моурт мерт- морт (Q морт (р.с. мьрт-) vb. tr. to examine, search out (ммо); to visit; reflex. (± скох) to reflect, ponder. As n.m. consideration, opinion. ътморт inscrutible.

MOY2 MG2- (MA2-) MA2" (MO2") Q MG2 (MH2) Vb. tr. to fill (MMO"; with: MMO", 2N, GBOA 2N); to fulfill, complete; to pay, repay (debt: MMO" ± GBOA; with: 2N; person: obj. suff. only); vb. intr. to become full, filled (of, with: MMO", 2N, 2A, 21TN); to get paid; + G2PA1 to flood (of the Nile); as n.m. fullness, contents; inundation. MG2-PW" to fill the mouth (with: MMO", 2N, G, GXN). MG2-TOOT" to fill hand, seize (MMO"). MG2-2HT to become sated. For MG2- as ordinal prefix see §30.7.

MOY2 vb. intr. to look (at: e).

MOY2 vb. intr. to burn, glow (with fuel: ΜΜΟ*).

MOΥΧਓ (MΟΥΧΚ, MΟΥΧΤ) MEXT- MOX6" (MOΧΚ", MOΧΤ") Q MOX6 (MOXΤ, MAXΤ) vb. tr. to mix (MMO"; with: MN); intr. to be mixed (with: 6, MN, 21, 2N). As n.m. mixture. γεσμογχε mixer, confuser, disturber.

MOSTE, MOOSTE n.pl. parts, neighborhood. Moosse idem. $MOX\overline{2}$, $MOX\overline{2}$, $MOXK\overline{2}$ (and -4 for -2) n.m. girdle (of monk or

soldier.

Mnai n.m. spindle.

พิทธ neg. part. It was not so (in answer to question in past tense). xพิ พิทธ or not (in double question, coord. with preceding positive statement); egone พิทธ if not.

Mno, емпо, евю (f. нмпю) adj. dumb, mute. ммтмпо muteness; F-мпо (O o N) to become mute.

Mπωρ exclam. imptv. part. Don't! By no means! No! Also used like Mπρ- as prefix for neg. imptv. See §17.1; 30.1.

мпфа, емпфа, мфа vb. intr. to be worthy, deserving (of: ммо"; of doing: м, е + Inf.); as n.m. worth, deserts, fate. атмпфа worthless, undeserving; ммтатмпфа unworthiness. управа то become worthy, deserving.

MPIC n.m. new wine, must.

мроф (мраф) Q морф vb. intr. to become red/yellow. мнрф,мерф, мрф adj. red, ruddy; р-мнрф (Q o n) to be ruddy.мрф, емрф, емрф (pl. мрооуе) n.f. harbor, landing stage.мрфм, оурфм, оухфм n.m. pillow.

MCA2 (pl. MCOO2) n.m. crocodile.

(Mcwse) encwse, Mcwne n.f. large needle.

 \vec{N} то евох n.m. presence, in prep. \vec{N} пемто евох \vec{N} , \vec{M} п(*) \vec{N} то евох in the presence of, before.

MTON (6MTON) Q MOTN vb. intr. to become at ease, at rest, content, relieved, well; Q also: to be easy (to do: 6 + Inf.); often impers. it is easy (6, 6Tp6). Vb. reflex. (with MMO") to rest self; to go to rest, die; as n.m. rest, ease, relief; MAI-MTON loving ease; MA N MTON a place to rest. P-n(")MTON to be or put at ease. MTON N 2HT to become content; as n.m. rest, satisfaction. +-NTON NA" to set at ease, give rest/respite to. XI-MTON to get rest, be relieved. MOTN n.m. health, ease. MOTNGC n.f. ease, contentment; +-MOTNGC NA" to give relief to; P-MOTNGC NA" idem; XI-MOTNGC to get relief; 2N OYMOTNGC With ease, easily. MOYTN MGTN- MOTN" Vb. tr. to set at rest (MMO"); also reflex.

мтю, ємтю, мто n.m.f. depth (of the sea); вык м мтю to founder, sink; в нвык м мто shipwreck.

мугр, вмугр, мефтр n. name of 6th Coptic month.

мотите, міштите n.f. comb.

H2AAY, H2AOY n.m. tomb, cavern.

 \overline{H}_2 IT, \overline{G} M2 IT n.m. north. \overline{G} HGH2 IT northward. \overline{M} HGH2 IT \overline{N} on the north of. \overline{G}_2 - \overline{H}_2 IT (on) the north side. THY \overline{N} \overline{M}_2 IT northwind.

MX12T, MX1TZ, MX12X n.f. mortar (vessel).

MXWA, 6MXWA n.m. onion.

M: eine MAA2: MA2 маар: мочр MAAT6: MAT6 MAB-: MAAB магаа: макот MX1-: M6 MAIPE: NOYP MAKA(A)T: MAKOT MAKOT: MAKOT MAN-: MOONE MANG(-): MOONE манну: мооне MAHOOYT: MOONE HAROY-: MOONE MANOYOY": MOONE MANXAA6: MAN6AA6 MAP (-): MOYP MAPHC: PHC MAC (-): MIC6 MACE: MICE MACT": MICE MACK': MOYCK MACT-: MOCTE MAT": MA MATH-: MTON матоет: матот MATWOY: MATE MAY: MAAY MAYAT": MAYAA" мафбрт: мафрт HAUT -: HOYUT MA2-/*: MOY2 MAXT: MOYXE MYX-: MYYXe муже: мууже

 $MAX\overline{4}: MOX\overline{2}$ $M\lambda X\overline{2}: MOX\overline{2}$ ME6: MG MEI: ME мекмоүк": мокмек MEKZT: MOYKZ MEKKE: MAKKE MEXATE: MEXOT $MG\lambda\overline{2}(-): MOY\lambda\overline{2}$ MEXSE: MOAYS Mene-: Moone MENK-: MOYNK MEP: MOYP MGP-: MOYP мере-: ме M6PIT(*): M6 мера: Мроф MEPE: MEPE2 Mec-: MICE MECIO(*): MICE MECIO: MICE MECTE(-): MOCTE MECT-/*: MICE MECTH: MOCTE местю": мосте MECOHT: MECTN2HT MECOYPH: MECOPH месфри: месори мете: мате METN-: MTON метри: мптре MEYE: NEGYE MEYT-: MOYOYT мефа": мефе мефак: мефе

мефір: Мфір

MEOT-: MOYOT медуе: дос M62 (-): MOY2, MA2 M626: MH26 MG2T-: MA2T MEXTT: MOYXE MH6: M6 MHEI": MA MHHNE: MHNE MHHCE: MICE MHHTE: MHTE MHN(E): MOYN MHPE: MOYP мирю: Мроф MHPZ: MOYAZ MHCE: MICE MHT: ME MHTE: MHT MI: MH MIW: MIO" MIWTN: MIO* MIXKE: MAXKE MIQTOTE: MOTOTE MX2: MOYAT **M**λ002: Mλλ2 MMATE: MATE **Минетти:** мо **Миние:** мние -MMM-: -MMM йно: мо MMO": N йм : иоий MNOOTE: MNOYT MNOTE: MNOYT **MNT-:** MHT

мефпфие: межпфие

MNTA": MN MNTACE: COOY мйте-: мй MNTH: TOY MNTOYE: OYA митресу: митре MO175: WO615 MOKE: MAKE MOK2": MOYKE MOKZ: MKA2 MOK2C: MKA2 MOX2 (*): MOYX2 MONT': MOYNK MOOYE: YOUNE MOOYT: NOY MOOYT ! MOYOYT MOOD (T) 6: MODT6 MOP": MOYP MOPE: MOYP морф: Мроф MOCE: MICE

MOCK: MOYCK MOTE: MOYT мотії: йтом MOTN €: MTON MOTNEC: MTON MOY: MOY6 MOYEL: MOYL, MOYE MOYETH: MOOY MOYELOOYE: MOOY MOYHELE: HOOY MOYI: MOYE MOYNT: MOYNK MOANETOOAE: WOOA MOYPE: MOYAE MOYPENA2: ENA2 нотй: йтон MOYXK: MOYXE MOYXT: MOYXE MODE: MOODE мофт(*): моуфТ MO2": MOY2

MOXK": MOYXE MOXK2: MOXZ можт (*): моужб MOXT: MOXZ $MOX6(*): MOYX\overline{6}$ HnF-: Hnwp MP-: MOYP Wody : Wrdy Прооче: Про MPPG: MOYP Mpwze: Mpwwe мую йроф MCE: WMC MCOO2: MCA2 Mcone: Mcose **Мто:** Мто MO: NO **Μ**ወአ: Μπολ майре: фре MXATZ: MXA2T MXN2: 6N2

N

 \overline{N} prep. marking the genitive; see 2.3.

 \overline{N} (Na*) prep. to, for (dative; see 10.2); also in Ma Na1* Give me (+ pron. suffix).

 \overline{N} linking noun and adjective (15.1), noun and noun (23.2).

 \overline{N} linking noun to proleptic suffix (10.4).

 \overline{N} ... AN negation; see Grammatical Index.

N (NMO*) prep. (1) place: in, into, from in; (2) time: in, on, during; (3) agent, instrument: with, by; (4) used to form adverbs (S.V.); (5) GBOX N out of, from within; (6) as marker of direct object (10.1); (7) partitive: of.

NA (NAA, NAE, NAI) vb. intr. to have pity (on: NA', 2A); as n.m. pity, mercy, charity. elpe й оуна, p-nna to treat charitably, kindly (мй, e2pal exi). Atha pitiless; p-atha (Q o й) to become pitiless. мйтна pity, charity; p-мйтна to do charity; cp-мйтна to distribute charity; ф-мйтна to give charity; x1-мйтна to receive charity; ga(A)т-мйтна to ask for charity. Na-нт adj. compassionate; p-na-нт to be compassionate; ма1-на-нт

charity-loving; ммтна-нт pity, charity.

NA vb. intr. to go (to: в, врат'); NA в тым to go whither?
NA вгоум to enter (в, уа); NA вгра to go up. NA ...
NHY to come and go.

אבא (אבר) אבר pred. adj. to be great (29.2).

nacio, naio, neio n.f. peg, stake.

NAAKE n.f. labor pains; pains in general. †-NAAKE to be in labor (with: ΜΝΟ*).

NANOY- (NANG-) NANOY' pred. adj. to be good, fair, just.

NANOYC impers. it is good, right (e, etpe). net nanoya
that which is good; mal-net nanoya loving what is good;

F-net nanoya to do good (to: na'; mm); peaF-net nanoya
benefactor; mmtpeaF-netnanoya benefaction.

mange, wenge n.f. grain, seed.

NAT, NET, NHT n.m. loom, web.

אמץ (imptv. אמץ) vb. tr. to look at, see, behold (6; that: x6); to seek out, get. אמץ פּפּסא to be able to see (i.e. not be blind). As n.m. sight, vision, view. אדאמץ פּפְס unseen, unseeable. פּפּיאמץ seer.

"μας Τε Νεετ- Q Νεογτ (Νεοτ) vb. intr./tr. to believe, trust (in: ε, εΝ, εΧΝ); Q to be trustworthy, faithful; as n. m. trust, faith. ατημετε unbelieving; γ-ατημετε to be mistrustful, unbelieving. γεθημετε believer. Νεοτ n. trust, faith; ο Νεοτ (Q) to be trustworthy.

NAX26, NAAX26, NA(A)x6, Nexe n.f. tooth.

месч, мсч, месв, мн(н)ч, мн(н)в n.m. sailor.

"εσεωω" pred. adj. to be wise. Cf. 29.2.

NHCe n.f. bench.

- NECE- NECO' (NECO') pred. adj. to be beautiful. HET NECOY, NET NECOY that which is beautiful. Cf. 29.2.
- NG2, N $\overline{2}$, NH2 n.m. oil. ATNG2 without oil. (n) ep-NG2 oil-press. +-NG2 to pour oil. ca \overline{N} NG2 oil-dealer.
- N62 N6 vb. intr. to mourn (for: 6, $6 \times \overline{N}$); as n.m. mourning. N62 C6 vb. tr. to awake, rouse (\overline{M} MO'); also reflex.: vb.
- intr. (\pm 62 pa1) to awake, arise (from: 2λ , $2\overline{N}$, 680 λ $2\overline{N}$). Ne6 ω ° pred. adj. to be ugly, unseemly, disgraceful. Cf. 29.2. NHHB6 (NHB6, NIB6, NI46) vb. intr. to swim, float.
- NHY ($\overline{N}NHY$) Q to be coming, about to come, to be on the way. Used as Q of ϵ_1 , q.v. for prep. and adv. complements.
- NIM (1) interrog. pron. who? what? NIM \overline{N} adj. what? (2) indef. pron. so and so; NIM $M\overline{N}$ NIM idem. See Gr. In.
- NIM adj. every, each, used with articleless noun, often with pl. resumption. See 16.2.
- миче (миве) мачт" (мечт", мичт") vb. tr. to blow (ммо"; аway; евод); vb. intr. (subj. wind, breath) to blow, with prep. in normal senses. As n.m. breath. †-миче to give breath; 2м-миче difficult breathing.
- NKA n.m. thing(s) in general; food; vessel; property, belongings. $\overline{N}KA$ NIM everything.
- йкотк (енкотк, икоте) vb. intr. to lie down (on: e, exu, zıxu); to die; as n.m. sleep, death. атикотк sleepless; ма и икотк couch. речикотк one who lies.
- NNO exclam. no, it shall not be so!
- nobe n.m. sin. atnobe sinless. мат-nobe sin-loving. Fnobe to sin (against: e); речF-nobe sinner; митречFnobe sinfulness.
- NOGIN vb. tr. to shake (ммо'); intr. to shake, tremble.

 атновін unshaken. As n.m. shaking.
- NOGIK n.m. adulterer. F-NOGIK to commit adultery (with: e, мм); ммтносік adultery.
- NOKNEK vb. intr. to have affection (for: e20YN e); as n.m. affection.

NOM, NAM n.m. pine, tamarisk.

NOMTE n.f. strength, power.

Noy vb. to be about to, be going to $(+ \epsilon + Inf.)$.

NOYB, NOY4 n.m. gold; money, coin. MAI-NOYB gold-loving. 2AM-NOYB, 2AY-NOYB, 2OY-NOYB n.m. goldsmith.

(NOYBT) NOBT vb. tr. to weave. NHBTE n.f. plait; basketwork.

NOYN n.m. the abyss of hell, the depths of the sea or earth.
NOYNG n.f. root. NEX-NOYNG to put forth roots. XI-NOYNG
(± 680%) to take root.

Noype n.f.m. vulture.

моут n. receptacle, pool.

- nort нат vb. tr. to grind, pound (ммо). на й nort mill. речногт grinder. Noeit n.m. meal, ground grain.
- NOYTE (pl. ΝΤΗΡ, εΝΤΑΙΡ) god. πΝΟΥΤΕ God. ΑΤΝΟΥΤΕ godless; ΜΝΤΑΤΝΟΥΤΕ godlessness; F-ΑΤΝΟΥΤΕ to be godless. ΜΝΤΝΟΥΤΕ divinity. ΜΑΙ-ΝΟΥΤΕ God-loving; ΜΝΤΜΑΙ-ΝΟΥΤΕ piety, godliness. ΜΑC-ΝΟΥΤΕ, ΧΠΕ-ΝΟΥΤΕ God-bearing. ΜΑCΤΕ-ΝΟΥΤΕ God-hating. ΡΜΝΝΟΥΤΕ godly person; ΜΝΤΡΗΝ-ΝΟΥΤΕ godliness. 2ΑΤΕ-ΝΟΥΤΕ God-slaying. ΡΕΦΘΕΜΘΕ-ΝΟΥΤΕ God-serving; ΜΝΤΡΕΦΘΕΜΘΕ-ΝΟΥΤΕ piety.
- моутм, Q мотм vb. intr. to be sweet, pleasant; as n.m. sweetness. xir-noytм sweet olive.
- NOYTY (NOYTY) NETY- (NETE-) Q NOTY (NOTE) vb. tr. to loosen, relax (אוסף); אפדע-ףω, אפדע-ה() 20 to smile; vb. intr. to become relaxed, loosened; (subj. face, mouth) to smile. As n.m. relaxation.
- ноурп мерп- морп' (марп') Q морп vb. tr. to frighten (ммо'), overawe; intr. to be frightened. моурп евох, N савох to frighten away (from: ммо').
- моүфс мофс vb. tr. to benumb; to strike, rebuke; as n.m. numbness. мофс n.m. one who strikes.
- NOY97 vb. intr. to become heavy, hard, difficult. Cf. Ngor.
- Nογηρ Q Nοηρ vb. intr. to be good. Nεηρ- pred. adj. to be good. Nοηρε, Nospe n.f. good, profit, advantage; ρ-

- NO 4P6 to be profitable (to, for: NA'; to do: 6, 6TP6).
 NO Y 4 adj. good; rare except in cpds. (cτοι, 26, φικ6).
 NO Y 4 (NO Y 8 T) vb. intr. to swell, be distended.
- NOY2, Nω2 n.m. rope, cord. cp-Noy2 εξολ to stretch measuring cord. ck-Noy2 as n.m. portion measured by cord.

 geg-Noy2 to make (lit. twist) rope.
- NOY2 NA 2 B- NA 2 B' Q NA 2 B vb. tr. to yoke up (a wagon:

 MMO'), to yoke (an animal: MMO'; to: 620YN 6). NA 2 B,

 NA 2 6 B n.m. yoke. NA 2 B 6 q n.m. idem. 4A 1-NA 2 B beast of

 burden. NA 2 E, NA 2 T, NA 2 B 6 n.f. shoulders, back, neck;

 g1 N NA 2 E shoulder's height. 2 E c-NA 2 E shoulder-covering.

 NOY2 E vb. intr. to copulate.
- NOY26 (Nω26, NOY2, NOY) NG2- NA2' Q NH2 (NG2) vb. tr. (1) to shake, cast off (ΜΜΟ'; ± 6ΒΟλ); (2) to separate, set apart (ΜΜΟ' ± 6ΒΟλ); vb. reflex. to separate self; to turn, return; vb. intr. (± 6ΒΟλ) to come apart, loose. NOY26 n.f. sycamore.
- NOY2M N62M NA2M Q NA2M vb. tr. to save, rescue, preserve (พีพo"; from: e, etw, พิปพ. 2N, eBOA 2N, eBOA 21TN); vb. intr. to be saved, rescued (preps. as above); Q to be safe and sound. As n.m. safety. peqNoY2M savior.
- Noyx adj. lying, false (usu. aft. n. w. \overline{N}); as n.m. liar. $M\overline{N}$ TNOYX falsehood. ca \overline{N} M \overline{N} TNOYX lie-monger.
- NOYXE (NOYX) NEX- NOX" Q NHX Vb. tr. to throw, cast (MHO"), used with full range of prep. and adv. in normal senses; Q to be situated, lying, reclining (at table). NOYXE MHO" 6 to cast into (prison), to launch a (ship) in (water). NHX 6 to rely on. NOYXE MHO" 6XN to impose (sthg.) upon (someone); to put (clothes) on (someone). NOYXE MHO" 620YN to put in, introduce. As n.m. throw.
- NOΥΧΚ NOΧΚ" (NOX6", NOX") vb. tr. to sprinkle, asperge
 (ΜΜΟ"; upon: 6ΧΝ); dir. obj. may be substance scattered
 or object receiving it. As n.m. sprinkling, scattering.
- NOY6 \overline{c} N66 \overline{c} Q NO6 \overline{c} vb. intr. to become angry, furious (at, against: c, $c \times \overline{n}$); as n.m. wrath. \overline{r} -NOY6 \overline{c} to make angry.

pequoyec wrathful person. +-noyec nx to make angry.

nother needed needoys vb. tr. to reproach, mock (\overline{m} no'); as n.m. reproach; \overline{r} -nother to become a reproach.

HOG adj. big, great, large; elder (son, brother, sister); bef. or aft. n. with N; aft. n. without N; as n.m. great person or thing, old person. MNTNOG greatness; seniority; F-MNTNOG to do great things. F-NOG (Q o N) to become great; to grow up, become of age; MAI-F-NOG ambitious. NOG & greater, older than; F-NOG & to become older than, superior to. F-OYNOG, gone N OYNOG to become great. NOG N pome full-grown; old; as n.m. elder, notable; MNTNOG N pome old age. NOG N C21MG sim. NTG NTA* prep. expressing genitive and possession; Gr. In. NTHG, enths n.m. plant, herb, weed; F-NTHG to become

weedy. x:-NTH6 to sow plants.

пто indep. pers. pron. you (f.s.).

йток indep. pers. pron. you (m.s.).

птооу indep. pers. pron. they; cf. пточ.

 $\overline{\mathfrak{n}}$ τοογ $\overline{\mathfrak{n}}$, $\overline{\mathfrak{n}}$ τωογ \mathfrak{n} adv. then, next, thereupon; therefore, so.

Nтос indep. pers. pron. she, it (f.); cf. Nточ.

 \overline{N} TO4 (1) indep. pers. pron. he, it (m.); (2) adv. but, rather, on the other hand; again, further; \overline{N} TOC and \overline{N} TOOY may be used sim. with f. or pl. subject reference.

NTO4 NTO4 me he (it) is one and the same.

πτωτπ indep. pers. pron. you (pl.).

Nyot (ενφοτ) Q ναφτ νb. intr. to become hard, strong, difficult; Q to be hard, harsh, difficult. ναφτ-2ρα impudent. ναφτ-(Ν)-2μτ hard-hearted; μντναφτ-2μτ hard-heartedness; γ-ναφτ-2μτ (Q ο Ν) to become hard-hearted. As n.m. harshness, boldness; 2ν ογνφοτ harshly, roughly; γ-νφοτ ν/ε κ(*)2μτ to encourage. ναφτε n.f. strength, protection; γ-ναφτε (Q ο ν) to become protector.

MG: particle introducing subject in post-verbal position.

N-: n-N-: 61N6 NA": N N. -: NA-, A KA: NA NAXY: 61AXY NAAX(2)6: NAX26 N76-: N77-NYE: NY AN ; IAN : IAN NAIAT': EIA NAM: NOM NAT": NOYT пфоки: поки тофи : (-) Тф4и NAOTE: NOOT NAOTIMME: 61ME NAWTMME: 61M6 BPIN : TPAN NYSE NOASE NA 2B": NOY2B NA2 (6) B: NOY2 B NA2M("): NOY2M NA 2 PA ": 20 NA2PN: 20 BSYON : PSAN NYCCE: NOACC $\overline{N}B\overline{\lambda}-$, $\overline{N}B\overline{\lambda}\lambda\lambda'$: $B\omega\lambda$ N6: 116 Ne: ene NEIYYA: EIYYA NEID: NYEIO иепре: мапре NET: NAT NETE-: NOYT¶

NETT: NOYTT иеф∏-: ноуфП N64: N664 NEAL-: NOVAL NEGT": NIGE N € 2 (-): NOY 2 € NEZMT: NOYZM Nex-: NOYXE NEXE: NAX26 NE6C-: NOY6C NH: NH NHBE: NHHBE NHBTE: NOYBT иннв: иееч иння: иеея NHT: NAT NH2: NE2 NH 2: NOY 2 6 NHY: 61 инж: иоуже NIAAY: GIAAY NIBE: NHHBE NIBE: NIGE NIGE: NHHBE NIAT : NIAE NAIKT": AIKT" NM: MN йм: *км NOBPE: NOYAL NOBT": NOYBT NOEIT: NOYT NOTE: NOYTE ноты: ноуты NOTT: NOYTT NOY": nw"

NOYBT: NOY9T Νογεών: ολφφ MOYOGI: OYOGI NOY4: NOYB NOY46: NOY4F νογττ Νογττ NOY2: NOY26 NOYX: NOYX6 ПФХОИ : (,) нФОИ NOOC': NOYOT MOYF: NOYYF NOTPE: NOYTE NOX4: NOAXE NOX": NOYXK NOXK": NOYXK NOX6": NOYXK NO6C: NOY6C NCABHA: BOX ΝΟΔΒΟλ: Βωλ NCA N BOA: BOA NT": EINE NTA": NT6-NT6: TWP6 йтй: тшре NTOOT': TOPE **ΝΤΨΟΥΝ: ΝΤΟΟΥΝ** NW2: NOY2 Nose: Nose NE: NEZ, SNE N2 GT-: NA2TE N2HT": 2N N2OT: NA2TE N2OYT: NA2TE Nxe: xe

NOY: NOY2 €

o

o, ω adj. great; archaic except as final element in cpds.: see ειερο, ρρο, 2λλο, 2λλω, ρτω, 2ρογο.

OBN, OBEN, ABEN n.m. alum.

OB26, OB2 n.m. tooth, tusk; (?) hoe.

OGIK n.m. (1) bread; loaf or piece of bread; (2) dung (cf.

MH). MA \overline{N} KA-/ \uparrow -/OY62-O61K storeroom, pantry. \overline{p} -O61K to become bread. p64TAM16-O61K baker.

оеік n.m. reed.

oeime, oime, oeim n.f. hook.

ogine, oine n.f. ephah (a grain measure).

oeig n. cry, only in cpds.: Ag-oeig loquacious; τλge-oeig to preach, proclaim (ΜΜΟ'); as n.m. preaching, proclamation; γεστλασε-oeig preacher, herald; γ-γεστλασε-oeig to become preacher, herald; ΜΝτγεστλασε-oeig proclaiming.

oxele, olxele n.m. ram.

оме, оме, жие n.m.f. clay, mud. F-оме to become mud. жи-пиры red clay; жи-гат white clay. оужи-оме n.f. name of a rodent; gangrene, ulcer; F-оужи-оме to spread like gangrene.

on adv. again, also, still, further, yet.

oor vb. (Q?) to groan or sim.

oote, ote n.f. womb.

ooyo n.m. gruel (of bread or lentils).

002, 02, w2 n.m. moon.

opse n. wafer, thin cake.

oce n.m. loss, damage; a fine. +-oce to suffer loss (of: MMO'); to be fined.

oee n.f. outlet (for water); way, course.

oze, ooze, wze n.m. courtyard; cattle pen, fold; pasture; herd, flock.

0: 61P6 ονς: ων OC24: 02C OB *: W46 OMK": WNK OT6: 00T6 086: 6186 ONT': WHK OT6": W6T OBGN: OBN оиф: фф 044: 046 оиос: Фид OBT(*): ወ4ቸ ዕወ፡ አወአነ ОВமுՇ: ω₿௵ ONZ: ANZ 02: 002 01: 2121 OOME: OME OXT': WOT Olyele: Oyele OONO: ONO 068C: W68 OYEKC: MYK 0026: 026 064: 06E ዕላለኛ: ወአኛ OP9": WPE

n

n-, t-, n- the def. article; see 1.3.

"λ-, τλ-, νλ- absolute relative pronoun, that of, that which belongs or pertains to; see 22.2.

пыт, тыт, мыт dem. pron. this, these; see 5.2.
пытоб, пытоб, петоб, петоб п.б. name of a disease.
пыко (пыко) Q пок(о) vb. intr. to become light, thin; w.
2HT: to become poor, mean (at heart). пок п.т. thin
sheet. plate.

nanot, manat n.m. bird, chicken.

naone, noone, naane name of 2nd Coptic month.

napmorte, napmort name of 8th Coptic month.

 $\pi \lambda P \overline{M}_2 O T \overline{\Pi}$, $\pi \lambda P G M_2 O T \overline{\Pi}$, $\pi 2 O T$, $\pi 2 \lambda T$ ($\overline{\Pi}$) name of 7th Coptic mo.

nat n.f. leg, shin, knee, foot. κλχ-πar to bend the knee.

πατάλλο n. unknown ethnic (?) term, abusive; prob. = pagan.
πλώνις, πλώνι, πλογνι name of 10th Coptic month.

naω n.m. trap, snare. naω n. idem.

nayonc, nagwnc, naxonc name of 9th Coptic month.

πλερε, πλερ n.m.(f.) drug, medicament; paint, color. ρπλερε to heal, cure (ε); ρεφρ-πλερε magician; μντρεφρπλερε magic. ψ-πλερε to heal, cure. χι-πλερε to take
medication, be healed; to take color, be dyed. μλ ν
χι-πλερε place of healing.

nazoγ n.m. back, hind part, buttocks; as adj. past. enazoγ adv. back, backward. enazoγ e prep. back to. ca-nazoγ = enazoγ. N ca-nazoγ adv. behind, back, from behind.

21 H120Y adv. in the past. 21 H120Y behind; prep. + $\overline{M}MO'$. NAGCE, NATCE n.f. spittle; NEX-NAGCE to spit.

пе, те, ме pron./copula. See Gr. In.

ne (pl. пнує) n.f. sky, heaven. Note adj. use in ауам м пе sky-blue, ах м пе hailstone(s), рммпе man of heaven. гроу-м-пе thunder, гооу м пе rain. тпе that which is above; етпе adv. upward; м тпе (1) adj. upper; (2) adv. above; (3) prep. above (+ м). гм тпе idem (2,3). (м) са-тпе idem (2,3). г тпе idem (2,3). р-тпе to surmount, rise above (ммог). м петпе м, гм петпе м prep. above, over. Р-петпе = Р-тпе.

net, ni n.f. kiss. \uparrow -nei to kiss (e, ep \overline{N} , ex \overline{N}). neipe (nipe) Q nope (npeiwoy, nepeiwoy) \pm ebox vb. intr.

to come forth (subj. light, blossom, hair); to bloom, blossom; to shine, be radiant. As n.m. coming forth, shining; tale; epithet. MA M neipe place of sunrise.

nenne, nenne n. bug.

nepinepoi n.m. royal palace.

nexe- nexx* vb. said (suff. is subj.), usu. + xe; used only to introduce direct speech.

пн, тн, мн dem. pron. that, those; see 30.8.

пире, пира n.m. quail. ги м пире brood of quails.
пиро n.m. red substance; rust, blight. ам-пиро red clay.
пи n.m. mouse.

nice (nic) $nec(\overline{\tau}) - nac\tau'$ (nict') Q noce (nuc) vb. tr. to cook, boil, bake ($\overline{m}mo'$); to melt (e.g. wax, metal, glass); vb. intr. to be cooked, to melt; as n.m. anything cooked. nace n. cooked food.

nite n.f. bow (for arrows); אַ אַ הווד loop-hole.

піба, пібн, пібі n. vanity; мПтпіба idem.

 $n\overline{\lambda}66$, $ne\lambda xe$, $n\overline{p}6e$ n.m. rag, torn cloth; as adj. old, worn. $n\overline{\lambda}66$ \overline{N} toesc idem; patch. $\overline{p}-n\overline{\lambda}66$ (Q o \overline{N}) to become torn, ragged.

пмин, пин n.f. doorpost, threshhold.

noeig n.m. rung, step.

no: n.m. bench.

 $nop\overline{\kappa}$ n.m. outer mantle of clerics, pallium.

nopk, nopek n.m. foal, calf. мес-nopk, мас-nopk mule.

потит vb. tr. to fell, cut down (ммо'); intr. to fall, fall away.

ηρω n.f. winter. F-τεπρω to pass the winter.

Ψις, Ψιτ (f. Ψιτε, Ψιςε) number: nine. με2Ψις ninth.
παταιογ ninety; ψαιτ- idem in cpd. nos.

nω", τω", Noy" poss. pron.; see 22.2.

 $\pi\omega\lambda\overline{6}$ $\pi\overline{\lambda}6-$ ($\pi\epsilon\lambda\kappa-$) $\pi0\lambda6''$ ($-\kappa''$, -x'', $\pi\lambda\lambda6''$) \pm $\epsilon\delta0\lambda$ vb. tr.

- to decide, settle (a matter: $\overline{M}MO'$); to relieve, free (from: $6BO\lambda$ \overline{N} , $6BO\lambda$ $2\overline{N}$, 2λ); vb. intr. to strive for or reach satisfaction or agreement (with: $M\overline{N}$); to reach conclusion; to be relieved of or freed from ($6BO\lambda$ 6, $6BO\lambda$ \overline{N} , $6BO\lambda$ $2\overline{N}$). $M\lambda OG$ n.m. part, portion. $MO\lambda G\overline{C}$ n.f. clod, lump.
- now (nown, nowne) new- (neen-, newe-) Q num vb. intr.
 to pour, be poured, flow (± 680x: out, forth); as n.m.
 pouring, outflow.
- חשאה (חשאה, חשאה) וההר (הפאה) nonk' vb. tr. (± פוסא) to draw, bail (water, breath: אוסי); to move, transfer, carry (אוסי; onto, upon: פּבּה; from: אוֹם; into: פּצַסיְא פּ).
- πωρ $\overline{\kappa}$ π $\overline{\rho}$ κ- (περ $\overline{\kappa}$ -) πορκ* (± εδολ) vb. tr. to pluck out, uproot ($\overline{\kappa}$ Μοσ*; from: $2\overline{\kappa}$); vb. intr. to be uprooted, destroyed. As n.m. plucking out.
- (πωρδ) Q πορδ vb. intr. to stretch, strain (uncertain).
 πορδ n.f. curtain (?), mat (?). προσραστ Q to be stiff
 (of hair).
- порф прф (перф-) порф Q порф vb. tr. to spread, stretch, extend (ммо"; ± евох out, forth); with е, ехм, 21, 21хм, мхгрм, оуве in normal senses; vb. intr. to spread, extend, be spread (prep. as preceding). As n.m. thing spread, mat, coverlet; + евох: spreading, extending.

 мх м порф соисh, bed. порф п.f. spread table. приф n.m. thing spread, mat, cloak, cover.
- nwpx npx- (nepx-) nopx* Q nopx ± esox vb. tr. to divide, separate (ΜΜο*; from: e, ΜΜο*; into: e; in half: e τ(*) NΗΤε; vb. intr. to divide, become divided; to part, depart. As n.m. (± esox) parting, separation; elpe N ογηφρχ to make a division; †-ηφρχ, † N ογηφρχ idem.

 λτηφρχ undivided, indivisible; ρ-λτηφρχ to become inseparable (with: e); ΜΝΤλτηφρχ indivisibility. Μλ Μ ηφρχ frontier. ρεψηφρχ divider.
- пют, Q пнт vb. intr. to run, flee; to run a course; used with full range of prep. and adv. in normal senses.

- nωτ Ν̄ca to pursue. nωτ Ν̄τΝ̄ to flee from; nωτ 2μτ' idem. As n.m. course, flight. μα Μ̄ nωτ place of refuge; racecourse; + 680λ: exit. peqnωτ runner.
- пютс, Q потс vb. tr. to split, divide, crack (ммо°, є).
 патсе n.f. plank, shelf.
- note note Q note vb. tr. to carve, engrave, depict (Μmo").

 noune (none) neene- (nene-) noone (naane") Q noone (1) vb.

 tr. to turn (Μmo"); to transfer, change, translate, copy (Μmo"; to, into, over to: ε, ε ε ογν ε, ε ε ρ ι ε); +

 ε δ ο λ: to remove, carry out, take out (Μmo"; from: Ν, ε Ν,

 ε ι χ Ν). (2) vb. intr. to turn, change, become altered

 (from: ε Ν; to: ε); + ε δ ο λ: to move away, depart. πουνε

 ε δ ο λ n.m. removal, change, death. λ τ πουνε immovable,

unchangeable; MNTATHOWNE immutability. peanwone, peanooned changeable person: MNTpeanwone changeableness.

пωωπε пъпе- пъпω' vb. tr. to make bricks (тωве). м\(\bar{n}\) тпъпетωве brick-making. мъ \(\bar{m}\) пъпе-тωве brickyard. пъпетт п. brick-maker.

nownec, nooned n.f. movement.

- nowpe nepe- noop' vb. tr. to dream (+ ρλcογ). ρεчηωωρε dreamer.
- nwm (nwme) neg- nom Q neg vb. tr. to divide (mmo; at, into: e; among: 2N, MN); to share (with: e2pN, exN, NA); vb. intr. to be divided, shared, apportioned (prep. as in preceding); as n.m. division. Atnow undivided, indivisible; peqnwm divider. name, neme (name, neme (name, neme) n.f. half, division; name N teymh midnight. F- name (Q o N) to be half, midway (+ N + Inf. or w. Circum.); also cpd. as F-name, as in F-name-moy be half dead.
- ทพฐพิ ทธฐพิ- ท๐ฐพ Q ท๐ฐพิ vb. tr. to ordain (พีмо"; as: พี);
 vb. intr. to serve as priest; as n.m. service, ordination. peqnωgพี servant. กมฐพธ n.f. service.
- תשמב (תשכים) הפעיד הסמני (תאמני) Q המעיד (המכיש) vb. tr. to amaze (אומים); to turn aside (אומים); vb. intr. (± פּבּפּג) to become amazed, beside oneself (at: פּבּאוֹ, אוֹב); to

turn aside, be turned (to: c). As n.m. amazement. πωως π 2 HT to be amazed, disturbed (at: εxπ, πcx): as n.m. amazement.

 $n\omega_2$ $n\varepsilon_2 (n\lambda_2-)$ $n\varepsilon_2$ $(n\lambda_2)$ Q $n\varepsilon_2$ $(n\varepsilon_2)$ $v\varepsilon_2$ tr. to burst, split, break, tear (MMO'); vb, intr. idem; as n.m. division, piece. o m nw2 nw2 Q to be in pieces. peqne2- splitter, divider. na26 n.f. fragment; N na26 naze in pieces. nazc n.f. prey; F-nazc, eige m nazc to make as prey. nx24 n.m. cleft.

 $n\omega_2$ $n\varepsilon_2$ - $(n\overline{z}$ -) Q nH_2 (1) vb. tr./intr. to reach, attain (e, e20YN e, φ A2PA1 e); to come upon $(\varepsilon \times \overline{N})$; to reach to (ωλ), refer to (ωλ); to mature, ripen. (2) aux, vb. + Inf.: to do for once, succeed in doing, just manage. to do.

πω₂ c ne₂ c- Q no₂ c vb. tr. to bite (mmo*); as n.m. bite. $n\omega_2 \overline{\tau}$ $n\varepsilon_2 \overline{\tau}$ $(n\lambda_2 \overline{\tau})$ $n\lambda_2 \tau'$ Q $n\lambda_2 \overline{\tau}$ (1) vb. tr. to bend, bow (MMO'); intr. and reflex. to bow, prostrate self. Used with ϵ , $\epsilon \times \overline{N}$, $\epsilon_2 P = 1$ $\epsilon \times \overline{N}$, N = 2, 2 = 2 in usual senses. (2) vb. tr. (± 680x) to pour, shed (MMO'; with 6, 6xN, 62PAI GEN in usual senses); vb. intr. to pour, flow (like preceding); nozī євох мī to abandon oneself with. As n.m. pouring, shedding. Δτπες T-cnoq not shedding blood. peanez T-cnog shedder of blood.

nωx6 (nωxκ, nωx7) Q nox7 vb. tr. to beat flat; as n.m. breadth, flat part.

πωσε ποσ' (ποκ', ποr') Q ποσε vb. tr. to break, burst (M-MO"); intr. idem. nose, nose, nose, nase n.f. fragment.

በአል₭6: በል₭6 эпоки зэпкки HYONG: HYONG በልአ6*: በወአቒ HANIN: BENINE проли: проие urue-: ummue пъпетт: поопе **⊔Уµതൂ:** ⊔തത⊔€ HAPEM2OTH: HAPM2OTH HAZE: HW2

патсе: паесе משח: דשאח π**λ**φε: πωφ пуфие: иффи magar: mag 3ωn :*/-s κπ U756: U05

uyce: uice

nact*: nice

ηλ₂τ(-/*): ηω₂τ πλ2Ψ: πω2 u76e: u06e neen-: nwn HEAKT: HWAE ueyxe: u<u>y</u>ee nen(e)-: nwn, nwwnc nenine: Benine uebe-: ummbe nepelwoy: nelpe

HEPKIBE: GKIBE

ποκ♥: πω66 1066: 1066 nec(T)-: NIC6 петве: тоове nok(e): make $n\overline{N}r-: (n\omega N\overline{K})$ пин: пйин петпе: пе **110K6: 11066** 16000X: 500A noka: uyke пиме-: пооме HELWOY: HELFE 1606: UY106 102K : 1026 nez-: (Bwz) noax: noa6 HPECPACT: HWPT nexx*, nexe-: xw HEHO: HOPO HONEC: HONE nuc: nice HOONE !: HOONE π**7**6€: π**λ**6€ HOONEC: HOONE YAIT-: YIC HHY6: NG noone: naone HCTAIOY: YIC пнфе: пфф HOOP": HOOPE HONT: HONK nı: neı HININE: BENINE 1007: 500A πωνε: πωωνε HONE: HONE HICT': HICE nope: neipe 1106: UY106 πωςω: πωως nopajo: nwpaj πωων(ε): πωωνε noce: nice пке: ке $n\omega x K: n\omega x \overline{6}$ በአ06: በወአቼ noco': nooc noxŦ: nwxe ποτ*: πω6€ noxī: noxē nore: nwee no6*: nw66

P

PA n.m. state, condition; cpd. with n. or vb. to give abstract or local sense, e.g. $PA-(\overline{N})-gA$ the east, $PA-\omega_2\overline{C}$ the harvest. gA nPA to the extent (of: \overline{N}), until (+ Rel.), even.

рыте n.f. kin, kindred. pырыте kinsman. בו-pыте to be akin.

γλΝ, γιΝ (ρͼΝ-, ρῆ-; γιΝ*, ρῆτ*, ρͼΝτ*, γλΝτ*) n.m. name, fame, reputation. +-ρῆτ* (ͼ) ϫͼ, +-ρῆ-πρλΝ ϫͼ to call, name. λτ+-ρλΝ Νλ* unnamed. For мογτε see Vocab.
 17. τλγε-ριΝ* to pronounce name, call by name. ρΜῆρλΝ dignitary, notable. 200γ Ñ ρλΝ holiday, name-day.

PAMMEI, PAMMI, PAMMI N. ring.

PAHONE, PAMOYNE n.m. part of a door.

PACTE n.m. morrow. расте, прасте, N расте, е расте, М печрасте on the morrow, tomorrow. NCA/MNNCA (печ)расте after tomorrow. ga (печ)расте until tomorrow.

Pacoy n.f. dream. F-pacoy to dream. peqoye2-pacoy interpreter of dreams.

PAT" n.m. foot; lowest part, bottom. $p\overline{m}(\overline{N})paT$ " footman. KA-paT" to set foot; + 680 λ to set out. Mooge \overline{N} paT"

to go on foot. N-pat' to track; atN-pat' untraceable.

cap-pat' to defecate. + epat' to put (shoe) on. oye2pat' to set foot. x1-pat' to impede. eN-pat' to trace,
search out; ateN-pat' unattainable. epat' prep. to, to
the foot/feet of. 2apat' prep. under. 21pat' toward.

paye, paye, phye n.f. town-quarter, neighborhood. pMpaye
neighbor.

pag only in pmpag mild, gentle person. митрира gentleness: P-pmpag (Q o N) to become gentle.

ף אשָּכּ vb. intr. to rejoice (over, at: פּאַת, פּצְףְאוֹ פּאַת; with: אַתּ); vb. tr. to mock, deride (אַאַסיֹּר); as n.m. joy.

PA2TE, PO2TE, AO2TE n.f. cauldron.

PARTOY n. some sort of monk's garment.

рыт, брыт, рчт adj. used with фтни garment.

PH n.m. sun; (alchemy) gold.

phc n.m. the south. 6 phc southward. м прис ммо° on the south of. 2x/21 прис ммо° idem. пхс-(N)-рис the south side. сх-рис on the south (of: N). рмрис southerner. мхрис п.т. Upper Egypt.

рнте n.m. manner, fashion. Rare in Sah.; use 26.

PI, PEI n.f. cell (of monk, of prison); room (of house).

ріке рек($\overline{\tau}$) – рак(τ) " (рек τ ") Q роке vb. tr. to bend, turn, incline ($\overline{\mathsf{M}}\mathsf{M}\mathsf{M}\mathsf{M}$ "; toward: \mathfrak{e} , $\mathfrak{e}\mathsf{x}\overline{\mathsf{N}}$, $\mathsf{N}\mathsf{x}$ ", $\overline{\mathsf{N}}\mathsf{c}\mathsf{a}$, $\mathfrak{g}\mathsf{a}$; away: $\mathfrak{e}\mathsf{B}\mathsf{o}\mathsf{a}$; away from: $\mathfrak{e}\mathsf{B}\mathsf{o}\mathsf{a}\mathsf{v}$ $\overline{\mathsf{N}}/2\overline{\mathsf{N}}$, $\mathfrak{g}\mathsf{1}$); vb. intr. and reflex. idem; as n.m. turning, inclination. $\mathfrak{p}\mathsf{a}\mathsf{k}\mathsf{T}\overline{\mathsf{c}}$ n.f. bent, direction.

pime vb. intr. to weep (about, for: e, exN, NA'); as n.m. weeping. xi-pime to weep. pMein, pMeie (pl. pMeiooγe) n.f. tear(s); †-pMein to weep.

PIP, PHA, AIA ($p\overline{p}$ -) n.m. swine, pig. PIP \overline{N} tooy wild swine. MANG-PIP swineherd. CA \overline{N} PIP pig-dealer.

 $p\bar{\kappa}pike$, $2p\bar{\kappa}pike$ n.f. nodding (in sleep); $\uparrow -p\bar{\kappa}pike$ na^* to give sleep to; $x_1-p\bar{\kappa}pike$ to doze off.

рымью n.m. rich man, important personage; мытрымью wealth; р-рынью to become rich.

рмонт, рмоот n.f. chills, ague.

pm26 (f. pm2+; pl. pm2eeye) n.m.f. free person. ммтрм2e freedom. кω (ввол) м pm2e to set free. p-pm2e (Q o N) to become free; to make free (from: 2N, ввол 2N). po n.m. goose.

po n.m. strand, ply (of cord).

PO (PΦ'; pl. PΦΟΥ) n.m. mouth; door, gate; edge (of sword);

ATPΦ' not speaking the language. κω Ν ΡΦ', κΑ-ΡΦ' (Q

κΑΡΑΘΙΤ) to remain silent; κΑ-ΡΦΨ n. silence; ΑΤΚΑ-ΡΦΨ

never silent; χι-ΡΦΨ to block off, obstruct (ΜΜΟ'); to

interrupt. πΑ-προ doorkeeper. 2ΑΝΡΟ n.m. doorway.

PA-, PG- forms fractions w. foll. no.: PA-ΘΦΟΜΝΤ a third.

GPN (GPΦ') prep. to the entrance of. 21PN (21PΦ') prep.

at the entrance of, on, at. 2ΑPN (2ΑΡΦ') prep. before,

usu. of setting food before. GBOA 21PN from before.

poeic, Q phc vb. intr. to remain awake, vigilant; to watch, keep watch (over: e); to guard (e; from: e, eвох 2N); as n.m. guard, watch. речровіс watchman. ма N ровіс watch, watch-tower. оущн N ровіс vigil; P-оущн N ровіс to keep vigil.

ромпе, рампе (р \overline{m} пе-; pl. р \overline{m} пооуе) n.f. year. етс генромпе many years ago. \overline{n} оуромпе for a year. ката ромпе per year. (\overline{n}) тромпе this year. \overline{r} ромпе, \overline{t} \overline{n} ромпе, \overline{t} еоромпе yearly, annually. \overline{p} -x \overline{n} powne to reach age of x; to pass x years.

POOYE, APOOYE n.m. stubble. ce-pooye n.f. stalk.
POOYNE n.m. virginity, virgin. μπτροογης idem; puberty.
POOYO n.m. care, concern, anxiety. κα-pooyo να to exercise care (suff. is reflex.). νεκ-ροογο ε to transfer cares to. F-pooyo (Q o N) to become a care/concern (for: να'); to become anxious (να' reflex.); to give heed (to: να'). μι-ροογο to take heed, take care (to, for: ε, ετες, να', ελ, or poss. prefix); as n.m. care, anxiety; μλι-ροογο guardian, one who cares (for: ελ); μητηλι-ροογο providence. λτροογο carefree; μητλτροογο freedom from care.

- POY26 n.m. evening: poy26, $6/\overline{N}/21$ poy26 in the evening.

 ga poy26 until evening. $nNAY \overline{N}$ poy26 the evening. $x_1-poy26$ to spend evening. 2Apoy26, 21poy26 = poy26.
- Pne, epne (pl. Pnhye) n.m. temple. goa ν ογρης, gl-pne to rob a temple. peqgl-pne, gaa-pne temple-robber.
- Fro, ero (f. Frw, erw; pl. Frwoy, erwoy) n.m.f. king, queen; as adj. royal. μπτερο, μπτερο (pl. -Frwoy, -erwoy) kingdom. F-Fro (Q o N) to become king; to rule (over: exN). ειρε Μμο΄ Ν Fro to make king.
- Pcω, ερcω n.f. fold (for sheep or cattle).

Pтов, ертов, (в) рточ n.m. grain measure.

ртш, ертш n.f. span (as measure).

- Pw, Pww enclitic part. of emphasis, usually of contrast: but, but then, on the other hand, on the contrary; in neg. context: not even, not at all. May follow other particles: λρηγ, μεφλκ, εφωπε, εφχε, ενε.
- Рωκε ρεκε- ροκε (γακε) Q ροκε vb. tr. to burn (ммо); vb. intr. to burn (aft., in pursuit of: мса, ε); as n.m. burning, fervor. ροκε n.f. fuel.
- роме (ром-, рм-, рем-) n.m. man, person, human being; indef. usage: anyone, no one; as adj. human; male (often redundant). атроне friendless; without a person; ммтатроне friendlessness. ман-роме kind. масте-роме misanthropic. ммтроме humanity; humanitas. у-роме to become man. For cpds. in рм-, реч- see 2nd elem.; cf. 27.2.
- рот рет- Q рыт vb. intr. to sprout, grow (subj. plants etc.); to become covered with vegetation, become overgrown (with: ммо°); as n.m. (pl. рате) vegetation; wool.
- Pwys regr- ray vb. tr. to satisfy, make content (ΜΜο΄);
 vb. intr. to suffice, be enough (for: ε, να΄); to assume responsibility (for: ε), deal with. As n.m. sufficiency, enough; ε πρωμε adv. enough, sufficiently.
 F-πρωμε to become enough, do enough, suffice.
- Pωze, Q paze vb. tr. to wash, clean (Μ̄μο'); paz T n.m.f. fuller, launderer.

Pω2Τ ρ62Τ- γλ2Τ' (ρο2Τ') Q γλ2Τ vb. tr. to strike, strike down, kill, cast down (ΜΜο'; upon, on: ε, ε2ογη ε, εxπ, ε2γλι εxπ; also + επεςκτ); vb. intr. to be struck, fall; Q to lie. As n.m. stroke, blow. γλ2Τς n.f. slaughter. Ψφωη, εγφωη n.m. cloak, covering.

P-: 61P6 P-ANA": ANAI PA-: PO PAK(〒)-/*: PIKC PAKTC: PIKE PAK2": PWKZ рампе: ромпе PANHI: PAMHGI PANT": PAN PATE: POT LYOLL: (Lmode) **የኢኒፎ: የመኒፎ ₽**▲2ቸ∶ ₽ω26 **₽**λ2Τ′: **₽**ω2Τ የኢኒፐሮ: የወኒቸ PB6: WPB **Рвн6:** еврн6е P € -: PO

ре-мит: мит PEK(T)-: PIKE PEKT": PIKC PEM-: PWM6 PEN-: PAN PENT": PAN peyī-: pwyc рнв: дрнв PHA: PIP PHC: POGIC рнт: брнт PHYE: PAYH PIN(*): PAN PM-: PWM6 рйетн: ріме PME 100YE: PIME

рйпе-: ромпе

рйпооче: ромпе

PMPAO: PAO PHOYA: 0YOGIG PMOYE: OYOELE PN-: PAN PN-, PNT': PAN POKE: PIKE POK26: PWKZ POOYT: OYPOT PO2T6: PA2TE ₹n-: ирП Purc: vc pp-: pip **РРНТ:** ЕРНТ pw', pwoy: po PWM: MPWM PUT: PET **Ргни:** ⊼гни PX-: WPX

С

- side, everywhich way. nick (N) ck ck nim on every side, everywhich way. nick (NN) nai, neick... (NN) nai, nck nck, nick... nikeck this way and that, this side and that. For the cpds. of ck (N) indicating direction or location, see 2nd element and § 28.7. keck elsewhere, apart. (N) ck oyck aside, apart, alone. (N) ck kaky N ck on any (no) side. Nck (Ncw*) prep. (1) behind; after (place or time); (2) after (= in search of, in pursuit of); (3) with some vbs.: against, at; (4) except, except for, other than. MNNCA (MNNCW*) prep. after (of time); MNNCWC adv. afterward.
- CA, Q CAGIOOY Vb. intr. to become beautiful; as n.m. beauty. CAGIG, CAGIH adj. beautiful (bef. or aft. n., usu. w. N); N/G CAGIG adv. thoroughly. F-CAGIG (Q O N) to become beautiful. MNTCAGIG beauty. +-CA to beautify (G, NA').

ca n.m. in cpds. maker of, dealer in, possessor of. See 2nd element and §23.2.

сдам (сдам сдам сдам од сдам од сдам од vb. tr. to nourish, rear, tend, maintain (ммо"); vb. intr. to be alive; Q to be nourished, well fed. As n.m. nourishment. ма й сдам од feeding place. речедам од nourisher, nurse; митречедам од геатing. Q сдам од теат

caace, cace n. tow, flax.

CABG (f. CABH; pl. CABGGY, CABGGYG) adj. wise; as n. wise person; bef. or aft. n. w. N. MNTCABG Wisdom. P-CABG (Q o N) to become wise. CBOY! n.m. disciple, apprentice. CBW (pl. CBOOYG, CBWOYG) n.f. instruction, doctrine; +-CBW to teach, instruct (person: NA*; subject: e); MA N +-CBW school; peq+-CBW teacher. x1-CBW to be taught (a subj.: e) peqx1-CBW pupil; MA N x1-CBW school. ATCBW ignorant. MA1-CBW loving learning. pM-NCBW knowledgeable person.

caein, caine n.m. physician; $m\overline{n}$ to make a show.

caλo, caλω, capo n.f. basket.

CAMIT n.m. fine flour.

CAMNT n.f. pool.

сырыкоте, сырыкоте n. wanderer, vagrant.

capaswoyg, capanswg, calaswwg n.m. hare, rabbit.

CAT, CHT n.m. tail. CHT, CEET n.m. penis.

CATBE Vb. intr. to chew, ruminate.

cate, caate, cote n.f. fire. gap \overline{N} cate flame of fire. \overline{p} -cate (Q o \overline{N}) to be fiery.

careepe n.f. stater (coin or weight).

CATO, CATO n.f. fan. cate vb. to fan.

слоч (f. слоче) number: seven. митслоч (f. -слоче) seventeen. мегслоч seventh. оче, ове, соче seventy.

cay n.m. yesterday. cay w 2007 idem. w cay idem.

ca2, ca24 n.m. aw1, borer.

CA2N- vb. tr. to bring near.

- calne n.m. supply, provisions. OYE2-calne to command (something: ммо"; someone: ма", етм; to do: e, етре); as n.m. command.
- CA2TE vb. tr. to kindle, to burn; as n.m. fire. ATCA2TE unheated. MA N CA2TE kitchen.
- calor (calore) czorp- (czorep-) czorwp Q czorop vb. tr. to curse (mmo'); as n.m. curse; e/2 ncalor under a curse. x1-calor to be cursed. peqcalor curser.
- cese cese- cesht* Q ceshy(т) vb. tr. to circumcise; as n. m. circumcision. Atcese uncircumcised; o N Atcese Q to be uncircumcised. мNTATCESE being uncircumcised.
- CBXTE, CXBTE, CX9TE vb. intr. to roll about.
- CBOK, Q COBK vb. intr. to become few, small; as n.m. fewness, smallness. $c\bar{\epsilon} K \epsilon$ n.m. fewness.
- стує, суве, утує n.f. shield.
- ce affirmative particle: yes, yes but; indeed, verily.
- ceene, cene, cine vb. intr. to remain over, be left over (of, from: ϵ , $2\overline{N}$). As n.m.f. (also chane, chae) remainder, rest; often with redundant $-\kappa\epsilon$ -.
- cei, cie n.f. name of a tree (oak?).
- ceaenin n.m. (1) spleen; (2) little finger or toe.
- cepcop Q to be displayed.
- сетн, стте, смте n. state in development of fig.
- ce2cw2' Q ce2cw2 vb. tr. to plane, rub down.
- CHBE, CH4E n.f. reed. CHBE \overline{N} par* shin-bone; greave. CHBE \overline{N} xw reed flute.
- CHNE, CENH, CENE n.f. granary, bin.
- снт, соте Q to be spun. снте n.f. spun fabric.
- CHY (coy-) n.m. time, season, age. Μπιση at this time.
 Ν ογσηγ once, at one time. Ν σηγ νιμ always. Μ/2Μ ησηγ at the time when. κατα σηγ from time to time. ατσηγ timeless. coy- is cpd. with no. to indicate day of
 month or other specified period. coγα = coy-ογα.

 CH46, CH86 n.f. sword, knife.

cms n.m.f. foal.

CI (CGI), Q CHY Vb. intr. to become sated, satisfied (with: MMO*, 2A, 2N); to enjoy; as n.m. fullness, surfeit. ATCI insatiate, greedy; MNTATCI greed. F-ATCI to be greedy.
CIB n.m. tick (insect).

CIB∓ n.f. hill.

CIKE CAKT Q COKE (COOKE) vb. tr. to grind, pound ($\overline{M}MO^e$); as n.m. grinding. KOT \overline{N} CIKE mill-wheel. WHE \overline{N} CIKE millstone.

сим (см-) n.m. grass, fodder, herbs; radish.

сімсім, смсім, смсм п.m. sesame.

CINE CN- (CEN-, CAAT-) CAAT" (CAT", COT", CNT") Vb. tr. to pass through/across; + 680A to pass out of, leave; vb. intr. idem (2N: through; 680A: out; 680A 2N out through); ATCINE not passing.

cine n.f. plowshare.

clooyn, claoyn n.f. bath.

cloy (coy-) n.m. star. coy- \overline{N} -2 τ 00 γ 6 morning star; coy- \overline{N} poy2 ϵ 6 evening star; coy- \overline{N} -2 ω p Orion; cloy cloy speckled.
cloyp n.m. eunuch.

cip n.m. hair; line, stripe.

cip, cep(e) n.m. leaven.

cip, caeip(e), chpe n.m. colostrum; butter.

сіт, сітє n.m. basilisk, serpent, dragon.

CITE CET- (CAT-) CAT" (CET", CIT") Q CHT VB. tr. to throw, Cast (MMO"; upon, on, in: exN, 21xN, 21; at, after: e, NCA), esp. to sow (grain); used with adv. in usu. senses. COTE, COOTE (pl. COOTE) n.m.f. arrow, dart; NEX-COTE to shoot arrow; MA N NEX-COTE archery range. PETK-COTE archer. x1-COTE to be struck by arrow. 2NAAY N KA-COTE quiver.

cige, Q cage vb. intr. to become bitter, like gall. As n.m. bitterness; 2π ογcige bitterly. †-cige to make bitter. cigq, cigs, giq n.m. flake, chip.

стче, снче, стве, снве n.m. tar; фе П стче cedar wood.

- clie ce2- chit' (ce2t') vb. tr. reflex. to remove self, withdraw; vb. intr. to be removed, displaced.
- ckal cek- cok' vb. tr. to plow (ммо'; with: ммо', 2N); as n.m. plowing. 2BB6 N ckal plow. речскы plowman.
- ckopkp ckpkp- ckpkwp Q ckepkwp vb. tr. to roll (Μησ); vb. intr. to roll, be rolled; as n.m. rolling. Used with various prep. and adv. in usu. senses. ckapakip, ckopakip, ckeaakip n. steep slope.
- CXAT, CXAAT, GOT, C2AT n.m. marriage gift (from groom).
- CRAATE vb. intr. to stumble, slip; as n.m. stumbling. +craate to cause to stumble (Na*).
- chonnen chennon' (± 680x) vb. tr. to tear asunder.
- схобат, Q схебаюб (схекаюк) vb. tr. to make smooth (ммо'); vb. intr. to become smooth; as n.m. smoothness.
- CMAY n.m. temples (of head); eyelids.
- CMA2 n.m. bunch (of fruit, flowers, etc.).
- cmн n.f. voice, sound. +-cmн to give voice, utter sound. х:-cmн to listen (to: 6). хтсмн voiceless, soundless. ххсТ-смн loquacious.
- CMINE CMN- (CMEN-) CMNT' Q CMONT Vb. tr. to establish, construct, found (MMO'); to set up, set right; to compose, write; to draw up (a document); vb. intr. to be established, put right, put in order; Q to exist, be standing, extant; to be correct, in good order. As n.m. establishing, confirmation, agreement. CMN-TOOT' MN to consort with. CMINE MMO' MN to settle (sthg.) with, to come to an agreement with..on... CMINE MMO' & to fabricate against. CMNTC & to resolve on (n. or Inf.).
- chme vb. intr. to make an appeal (to: NA°, 2A2TN; for, concerning: 2A, 62PA1 2A, 6TB6); to make an accusation (against: 6, 0YB6); as n.m. appeal, accusation. ANCHMG n. ordinance.
- смот n.m. form, likeness; appearance; pattern; character;

customary behavior. CMOT \overline{N} a kind of, sort of. ATCMOT formless. \overline{F} -(OY) CMOT to become as though (xe); \overline{F} -CMOT NIM to assume every aspect; \overline{F} -NEICMOT to behave thus; \overline{F} -necMOT \overline{N} to behave like. \uparrow -CMOT \overline{E} to give form to. XI-CMOT \overline{N} to become like.

CMOY, Q CHAMAAT (CMAAT, CMAMAANT) vb. tr. to bless (ϵ); as n.m. blessing, praise; \uparrow -cmoy to give blessing, give sacrament; x_1 -cmoy to receive sacrament; to greet, salute (someone: $\overline{N}T\overline{N}$).

CNAGIN vb. intr. to skip, stroll, wander (also reflex. with $\overline{\text{MMO}}$).

CNAT vb. intr. to be afraid (of: 2HT').

CNAY (f. cNTe) number: two (§15.3). MNTCNOOYC (f. -CNOOYCE) twelve, and sim. with higher nos. M necnay, N TCNTE adv. both together. MG2CNAY (f. -CNTE) second. 20 CNAY adj. two-edged. 2HT CNAY doubt; P-2HT CNAY to become doubtful; MNT2HT CNAY state of doubt. P-CNAY to become two; MNTPG4P-CNAY duality.

choya, choys n. last year.

cno4, cnos (pl. cnow4) n.m. blood. Atcnow bloodless. F-cno4 to become blood.

cNcN (cencen) vb. intr. to resound, echo; as n.m. echo.

co n. in +-co to spare, restrain (6); to avoid, refrain from (6 + n. or Inf.). As n.m. forbearance, restraint; AXN +-co unsparingly. MNTAT+-co lack of restraint.

 $\cos \overline{N}$ vb. intr. to fan, make cool breeze.

COBT n.m. wall, fence. KTG-COBT 6 to wall.

coste (coqte) cete- (ceste-) cetwr Q cetwr vb. tr. to prepare, make ready (ΜΜο'; for: e); vb. intr. to become ready, prepared; vb. reflex. to get ready. As n.m. preparation, what is prepared; furniture; F-coste to make preparations. Δτο Ετωτ unfurnished.

coeit n.m. fame, report. \overline{p} -coeit (Q o \overline{N}) to become famous. \uparrow -coeit to celebrate, give fame (to: $N\lambda^{\sigma}$, ε ; for, in: $2\overline{N}$). $p\overline{M}\overline{N}$ coeit famous person.

coeig, coeiz n.m. pair, couple.

col n.m. back (of man or animal).

coi n.m.f. beam; oyez-coi n.f. roof(-beam).

coκ, cook, caκ, cω(ω)κ n.m. sack, bag; sackcloth.

COKCEK CEKCEK- CEKCOK' vb. tr. to pull, stretch.

coa n.m. wick. MA \overline{N} †-coa wick-opening.

 $(coac\overline{x})$ $c\overline{x}c\overline{x}$ - ceacwa vb. tr. to adorn $(\overline{M}Mo^*; with: 2\overline{M})$.

CONCT CTCT- CTCWN' Q CTCWN vb. tr. to comfort, console (MMO'; for, concerning: ε, ετεε, εχΝ, ελ); vb. intr. to become comforted, consoled, encouraged; as n.m. consolation; amusement, diversion. gwnε (Q goon) Ν conc become a consolation. χι-conc to take comfort.

coaq, copq, coalb, coaq n.m. sieve.

 $com\overline{c}$ vb. intr. to look, see, behold (rare in Sah.).

con (cn-, cen-; pl. cnнү) n.m. brother (lit., fig.); моб м con elder brother; коү м con younger brother; con м е ют uncle; фм-сон пернем; con м мооне foster-brother. Freq. as monk's title. ммтсон brotherhood, brotherliness. ммтмы-сон brotherly love.

CONTE n.m. resin.

coone n.m. robber. MA W coone den of thieves.

- cooy (cey-; f. co, coe, cooye) number: six. MNTACE sixteen. ce sixty. Me2cooy sixth. Me2ce sixtieth. See 16.5; 24.3.
- cooyπ coyπ (coyωn-, coyen-) coyωn* vb. tr. to know (ϻμο*, ε; about: ετες; that: xε; how to: π + Inf.); to recognize, be acquainted with; to know sexually; as n.m. knowledge. ωρπ-cooyπ foreknowledge. Δτοοογπ ignorant; μπτωτοογπ ignorance; ρ-ωτοοογπ (Q o π) to be ignorant (of: ε, μμο*). ρμποοογπ an acquaintance. xι-cooγπ to get knowledge.
- cooytπ coytπ- (coyτωn-) coyτωn Q coyτωn vb. tr. to make straight, straighten (Μπο*); ± 680λ: to stretch (Μπο*; to: 6, 9λ, 620γη θ); vb. intr. to become straight, erect; to stretch; to be right (for: 6; with: μπ). As

n.m. uprightness. $2\overline{N}$ oycooyt \overline{N} uprightly; forthwith. \overline{N} cooyt \overline{N} just now. ε ncooyt \overline{N} straight, on target.

cooy26 n.f. egg; crown of head. Mooy N cooy26 egg-white.

- coole (cole) cale- calw (1) vb. tr. to set upright, set up (Μπο); vb. intr. to be set up, set upright. (2) vb. tr. to correct, reprove (Μπο); vb. intr. to be corrected. reproved.
- coo2e ch2e- ch2ω(ω) Q ch2Hy vb. tr. to remove; usu. reflex. w. 680h: to depart, withdraw; also w. 6πλ2ογ, e-20γN, 62Ph1. As n.m. departure.
- con (ch-, cen-; pl. coon, coon) n.m. time, occasion; turn, round (e.g. of reading or prayer). con...con now... again (oft. with men...a6). e ncon occasionally; all at once. M ncon at the time when (+ Rel.). M neicon on this occasion. N oycon on one occasion, once. N oycon eycon from time to time. 21 oycon altogether, all at one time. kata con from time to time. (N) kecon again. M neikecon yet once more. oymhhge/2a2 N con many times. Thincon idem. con nim always.
- conc cnc- (cenc-) vb. tr. and n.m. = conc \overline{n} q.v. \overline{p} -conc to make prayer. x_1 -con \overline{c} to receive comfort.
- concπ cπcπ- cπcωn' Q cencwn vb. tr. to entreat, implore (ΜΜΟ', ε); to pray (for: ετεε, εχΝ, 2λ, 21χΝ); to comfort; as n.m. prayer, entreaty, consolation.

соры, съры n.m. dregs.

copT, capT n.m.f. wool.

COT, COOT, CAT, CAATE n.m. dung, excrement.

cotbeq, catbeq, cot(2) \overline{q} n.m. tool, weapon; $+\overline{n}$ nige weapon. $coy\overline{n}\tau'$ n.m. price, value; $nage-coy\overline{n}\tau'$ of great value (vb.). coyo n.m. grain, wheat.

- (coyoloy $\overline{\lambda}$) coyeloywa" (coylwa", coylox") Q coylwa vb. tr. to wrap (\overline{M} Mo"; in: \overline{M} Mo", $2\overline{N}$; around: e).
- coype (coyp-, cep- cxp- cp-) n.f. thorn, spike, dart, aw1, needle. For cpds. see 2nd element.

coycoy n.m. point, moment.

- co6, ch66, c666 n.m. fool; adj. foolish. gx-co6 foolish talk. mnTco6 folly. F-co6 to become a fool; to make a fool.
- cosn (cken-) n.m. ointment. митсов anointing. nec-cosn to make ointments; пъс-/речпес-сов perfumer. скем-е-пісе cooking grease.
- $c\bar{n}$ -, cen- n.f. year, in date formulas, prefixed to number, e.g. $2\bar{n}$ (τ) $c\bar{n}$ - $q\tau ce$ in the 4th year.
- cnip (pl. cnipooye) n.m. rib. BHT-cnip rib. $\overline{N}/21$ ca-cnip \overline{N} MMo* prep. beside. 21 oycnip aside.
- спотоу n.m. lips; shore, edge (used as sg. or pl.).
- срач, серев n.m. wound, sore.
- cpit cpat' (cpit') vb. tr. to glean (mmo'); to ravage.
- сромр \overline{m} ср \overline{m} р ω м* Q ср \overline{m} р ω м vb. tr. to daze, stupefy (\overline{m} мо*); vb. intr. to become dazed, move dazedly; as n.m. stupefaction.
- срочреч (сроврев) сречрюч (среврюв) vb. tr. to dissipate (ммо); vb. intr. to wither, fall useless; as n.m. falling, withering. сречриче, срчриве, сречреч n. droppings, crumbs.
- стче (стве) Q сточт (стовт) vb. intr. to be at leisure, unoccupied; to have time for, be occupied with (e); as n. m. leisure, perseverance. митатстче lack of leisure.
- стим n.m. stibium, antimony, kohl.
- стої (c+-, сты-, сте-) n.m. smell, fragrance, incense; c+-моуче perfume, incense; c+-ым idem; c+-воом stench; ¬-стої to stink.
- стртр n.m. trembling.
- стю n.f. river bank.
- crowt vb. intr. to tremble (at: $2HT^{\sigma}$, 2λ); as n.m. trembling. co n. (mat of) soaked reeds.
- cw (coy) ce- (cey-, coy-) coo' vb. tr. to drink (ммо'); as n.m. drinking. мы м сw place for drinking. речсе-/сыу- a drinker of.

- cose vb. tr. to mock, deride, ridicule (ммо'); vb. intr. to laugh (at: exm, мсл), to sport (with: мм); as n.m. laughter, derision, sport. речсове mocker, jester. ух-сове jesting speech.
- cose, come n.f. edge, fringe.
- cook cek- (ck-, cak-) cok* Q chk (1) vb. tr. to pull, draw (mmo*); to beguile, attract; to protract, draw cut; to bring, take, lead; (2) vb. intr. to move with smooth, gliding motion, hence: to flow, be blown; to be drawn; to go, proceed; vb. reflex. = intr. Used with full range of prep. and adv. in usu. senses. cook 2x to submit to, move along with. cook as n.m. drawing; (± GBOX) death.
- cwx \bar{n} c $\bar{\chi}$ n- (cexn-) coxn* Q cox \bar{n} (tesox) vb. tr. to break off, cut off (\bar{m} no*); to decide; vb. intr. to be broken off, cut off; to burst, break; as n.m. separation. c $\bar{\chi}$ ne n. strip.
- cωνδ coν6' Q coν(ε)ε vb. tr. to smear (ΜΜο'; on: ε) to wipe out, obliterate; as n.m. obliteration.
- сшн сом" (сым") Q сим to pound, press, subdue.
- COMPT (COMNT) COMT' (COMNT') Q COM(N)T Vb. tr. to stretch, extend (MMO'; \pm 680%); to bind (to: e); vb. intr. to be stretched; to delay, tarry.
- cone n.f. sister. NO6/KOY1 \overline{N} cone elder/younger sister. cone \overline{M} MA2 \overline{T} real sister; cone 2A 610 \overline{T} step-sister.
- cwnk (cwnr, cwmk, cwmr) vb. tr. to suck (אַחסי).
- cwnT (cwwnT) cNT- (cenT-) cont" (coont") Q conT vb. tr. to found, create (MMO"); vb. intr. to be created; as n.m. creature, creation. ATCONT" uncreated. peqcwnT creator. ga-cwnT first in creation. cNTe n.f. foundation; κλ-cNTe to lay a foundation; cMN-/+-cNTe idem.
- CONT n.m. Custom. nconT \overline{N} ne_2 lone menstruation. elpe \overline{M} nconT to follow a custom.

- CONZ CONZ Q CONZ vb. tr. to fetter, bind (MMO'; to: 6, MTM; as to, e.g. feet: 2N, MMO'); vb. intr. to be bound: as n.m. bond. fetter. cnay2 n.m. bond. fetter.
- cwoy2 cey2- cooy2* Q cooy2 vb. tr. (± e2oyn) to gather, collect (ΜΜο*; to, at: e, exN, 2N; with: μΝ; against: e); vb. intr. idem, to be gathered. As n.m. gathering; assembly; F-ncwoy2 to attend service; μλ Ν cwoy2 meeting place. cooy2 c n.f. congregation, collection; F-cooy2 c to be collected.
- cwn cπ- (cen-) con vb. tr. to dip, soak (ммо ; in: 2N).
- cwp cp- (cep-) cop* Q chp (p.c. cap-) ± eBox vb. tr. to scatter, spread, extend, distribute (ΜΜο*); cwp Ντα to spread (report) against; vb. intr. to scatter, spread, esp. of sunlight; as n.m. (± cBox) spreading, laying out.
- сwpм сepм- copм' Q copм vb. tr. to lead astray, mislead.
 lose (ммо"; ± 6вол); сwpм ммо" мтоот" (reflex.) to lose;
 vb. intr. to go astray, get lost, err (from: 2м, м слвол м); as n.m. error. речсwpм one who leads astray.
 моу-м-сwpм torrent. срме n. wanderer, vagrant. сормес
 n.f. error.
- coc coc' Q chc vb. tr. to upset, overthrow ($\overline{M}MO'$); vb. intr. to be overthrown.
- cwr cor* (coor*, cxxr*) (1) vb. intr. or reflex. to repeat, do again (+ ε + Inf. or + Circum.); (2) vb. intr. to reach (to: ε; ± εεολ).
- сюте сет- сот" (соот", сът") vb. tr. to rescue, redeem (м-мо"; from: мтм, е, гм, евох гм); as n.m. ransom, price.

 †-сюте to redeem. хі-сюте to receive ransom. речсюте redeemer.
- соты сеты соты vb. tr. to hear, listen to (є); to obey, heed (ма, мса); to hear from, at hand of (єты, мты, гіты); as n.m. hearing, obedience. атсоты unhearing, disobedient; мытатсоты disobedience; р-атсоты (Q о ы) to be disobedient. мытречсоты obedience. стынт, сетынт, сынт adj. obedient; атстынт disobedient; мытстынт

obedience; \overline{p} - \overline{c} TMHT (Q o \overline{N}) to be obedient.

cot \overline{n} cet \overline{n} - cot \overline{n} Q cot \overline{n} vb. to choose, select (\overline{M} Mo'); Q to be chosen, elect; excellent, exquisite; often in comparative w. 6, \overline{n} 20 γ 0: to be better, choicer, more advantageous. As n.m. chosen or elect person; oft. adj. $\overline{M}\overline{n}$ to \overline{n} election, choice; superiority.

COTF, Q COTF vb. intr. to turn, twist.

- cωτ σ c c τ σ c c τ σ ' Q c c τ σ vb. tr. to purify, filter, strain, pour (Μπος; ± ε βολ); vb. intr. to be purified, pure, clear; to pour. As n.m. purity. ρεσιωτ σ purifier.
- cwwq ceeq- (ceese-, ceq-) cooq (coos, coq) Q cooq vb. tr. to defile, pollute (MMo); vb. intr. to become defiled, polluted; as n.m. pollution, abomination.
- cwg ceg- (gec-) cog' vb. tr. to strike. cag (cg-, gc-, cc-, gce-, ge-, gte-; pl. chge) n.m.f. blow, stroke; sore, wound; γ-cag (Q o N) to cover, be covered, with sores, wounds. +-cag to give a blow (to: Na'). x1-cag to be wounded. For cpds. w. reduced form + N see 2nd element.
- сшу сеу- (уес-) соу (уос) Q сну (унс) vb. tr. to despise, scorn (йно); vb. intr. to be despised, scorned, humbled; as n.m. shame, contempt, scorn. речсшу one who scorns; митречсшу scorn. †-сшу мъ to scorn, despise.
- compe vb. tr. to drag ($\overline{M}MO'$); vb. intr. to drag, creep.
- сюще n.f. field, open country. рынсюще country man.
- сшум (ушсм) Q соум (ушсм) vb. intr. (± N гнт) to be faint, (for: Ncx; from: гх), discouraged; to be annoyed (at, with: є, мN); as n.m. faintness.
- cwg τ ceg τ cog τ (cxg τ) Q cog τ (cxg τ) vb. tr. to stop, hinder (Μπο ; from: e + [neg.] Inf.); vb. intr. to stop, be hindered, impeded (from: e + Inf.).
- coop coop coop vb. tr. intr. = coop despise, q.v. coo n.m. deaf person. \overline{p} -coo to become deaf.
- cωz ε caz T- caz T' Q caz T (cag T) vb. tr. to weave (Μησ', ε;

- onto: ϵ_{2} OYN ϵ); as n.m. weaving. ϵ_{λ} 2 $\overline{\tau}$ in cpds.: weaver of. ϵ_{λ} 2 $\overline{\tau}$ n.m. weaver.
- COUZH CAZH Q CAZH vb. tr. to press down, crush, overwhelm (MNO'); vb. intr. to sink, be pressed down, crushed; to recede (from: 2A). CAZMEC n. pestle.
- $c\omega_2\overline{n}$ $c\varepsilon_2\overline{n}$ $c\lambda_2n$ vb. tr. to drink, suck in $(\overline{M}NO^*)$; vb. intr. to sink in, be swallowed. $c\iota_2n\varepsilon$ n. drop.
- $c\omega z \overline{p} ce z \overline{p} ca z p'' (co z p'') Q ca z \overline{p} vb. tr. to sweep (<math>\overline{m}mo''$); as n.m. sweeping.
- сює сєє- соє Q сне vb. tr. to stiffen, harden, paralyze (ммо'); vb. intr. to become rigid, paralyzed.
- C221 (C1261, C2HT) C62- C211" (C211C", C21T", C12", C12", C2T", C62T") Q CH2 Vb. tr. to write (MMO"; on, upon, in: 6, 6XN, 21, 21XN, 2N; to: 6, N1", 6P1T", 91; for, on behalf of: 6, 6TN, 21); to register; to draw, paint. C211 MMO" NC1 to ascribe to; C211 NC1 to take down in writing. C211 as n.m. writing, letter, epistle; letter of alph.

 X1-C211 to receive a letter. 1C211 illiterate. P64-C211 scribe. C12 n.m. scribe, writer; teacher, master, master craftsman; MNTC12 skill, craft; P-C12 (Q o N) to become master, skilled. C120, C1200 n.m.f. village scribe; also = C12.
- CZBHHTG, CBHHTG, CBGGTG, 2BHHTG n.m. foam. TAYG-CZBHHTG GBOA to foam.
- c21M6 (pl. 210M6) n.f. woman, wife; female. 21M6 n.f. wife.
 MNTC21M6 womanhood. ק-c21M6 to become wife (to: אגר).

 x1-c21M6 to take wife; as n.m. marriage.
- сенр (скир, фенр, фенр, фкер) vb. intr. to sail (with prep. in usu. senses); as n.m. sailing, voyage.
- сбрагт (сбрегт, убрагт, субрагт) vb. intr. to pause, become still, quiet, tranquil; also reflex. with ммо"; as n.m. quiet, rest; гм оусбрагт at rest, quietly.

 CART*: CONT
 CARHA: BOA
 CARIE: CA

 CAAT*/-: CINE
 CARO(*): TCARO
 CARIOOY: CA

 CAATE: CATE
 CAROA: BOA
 CARIF(6): CIP

CAK: COK	CEET: CAT	скир: сенр
CAK-: CWK	C 6 6 4 -: C W W 4	CKOPAKIP: CKOPKF
CAKT": CIKE	CEK-: CKAI	CXBTG: CBXTG
CAXO: C2A1	СЕКФТ: КФТ	CX6KXWK: CX06X 6
САХА6шшу: САРА6шОҮШ	сехети: Ф <u>х</u> еом	c <u>v</u> ue: cmv <u>u</u>
CAM": CWH	CENT: CON	схчте: свуте
CA-N-BOX: BWX	CEN-: CINE	cM-: CIM
СУИОХФ,: СУИФ	CENG, CENH: CHNE	CMAAT: CMOY
CYNĀ: CYYNĀ	CENCEN: CNCN	СМАМААТ: СМОУ
CAP-: CWP	cen-: con	CMENT: CMINE
CAPT: COYP	cen-: cn-	смит: сфт
САРН: СОРН	сепе: сеепе	CMN-: CMINE
CAPO: CAAO	cenc-: conc	CMNT": CMINE
CAPT: COPT	CEP-: COYP	CMONT: CMINE
CACE: CAACE	CEP (E): CIP	CMCIM: CIMCIM
CAT: COT	Cepes: SPA4	cn-: con
CAT-/*: CITE	CET-: CWTE	CNAY2: CONZ
CAT': CINE	CET-/*: CITE	CNHY: CON
CAT': COTE	сетинт: сфт	CNKO: TCENKO
CATBEM: COTBEM	Cey-: cw	CNOB: CNO9
CATE: CATO	CGY-: CW	CHOOLC (E): CHAY
CAY-: CW	cey-: cooy	CN-: CING
СХУ-нрП: нрП	C64-: C004	CNT-/": CINE
CAG: COG	C62-: C271	CNTE: CWNT
CAME: CIME	C62-: C126	CNTE: CNAY
CAOT: CO26	C62T*: C2A1	сите: сиду
Cλg(τ) ε CωgΤ	C62T": C126	CNOW4: CNO4
CA2 (*): C2A1	C666: C06	CO: COOY
CA26-: COO26	CHRE: CH46	COBK: CBOK
C7561: C571	CHEG: CH46	COE: COOY
CA2HY: COO26	синие: сеепе	COEI2: COEID
CA2M(*): CW2M	сние: сеепе	COK': CKAI
CA2M6C: CW2M	CHP6: CIP	COKE: CIKE
CA2n': Cw2n	CHT: CAT	CONE: CINC
Cysh : cms <u>b</u>	CHT: CITE	COME COMME
CA2T : C126	CHY: CI	COME . COMME
CX2T(-/*): Cw26	снув: Сфр	COO! CW
CA2T': C2A1	снче: снве	COOK: COK
C X 2 W (W) *: COO 2 6	CH46: C146	COOKE: CIKE
CA2 4: CA2	CH2: C2XI	COOKE: CTKC
CB66T6: CZBHHT6	CH66: CO6	COOT': CWTG
Свинте: Сувинте	CIAOYN: CIOOYN	COOT: COT
CBK6: CBOK	CIBE: CITE	COOT': CINE
CBO: TCABO	CIHY: TCIO	COOT6: CITE
CBOOY6: CBW	CIK: ACIK	
CBOY1: CAB6	CINE: CEENE	COOY2 C: CWOY2 COO4 (*): CWW4
CEPOOYE: POOYE	CIT': CITE	
CETE-: COBTE	CITE: CITE	COPMEC: CWPM
CBTWT(*): COBTE		COPT: CONT
CBW: CARE	CISUE: COSU	COT COTE
Ce: COOY	CKAPAKIP: CKOPKP	COT : CINE
CET: CW	CKENAKIP: CKOPKP	COT6: CAT6
	cken-: cosn	COTE: CHT

COT6: CITE COTT: COTEGY COTE 4: COTE 64 coy(-): cw COYT: QOY COY-: CIOY, CHY COYA: CHY COYBNNE: BNNE **COYEYOAMY: COΛΟΥΟΣ** COYENT: COOYN $COYAWA(f): COYOAOY\overline{A}$ COYN-: COOYN COY-OYHP: OYHP COYCOOYWE: WOYCOOYWE COYTN-: COOYTN COYTWN (-/"): COOYTÑ COYWN-/1: COOYN CO4*: COO4 COUTE: COBTE CO26: COO26 cn-: con cnc−: conc CP-: COYP

CFAPOOYE: APOOYE CPAT": CPIT CPBG: CP4G CPBHNE: BNN6 сречриче: срочра CPMG: COPH сроврев: срочря сровт: срчс CP04T: CF46 срчрів€: срочрч cc-: cww CTAATE: TAATE CTAIT: CTOI сте-: тсто CTET: CTOI CTEIW26: 61W26 CTHY: TCTO с+-: сто і стинт: сфтй сто(*): тсто(*) CTW26: 61W26 COO: TCTO

CONT: CONT COMT: CONK CWMK: CWNK COMNT: CONT CONF: CONK cone: cose COONT: CONT coo-: coo COB6: CBOE core: മാമാ€ сфие: ФСие COOT: CXAT **Γ**QΔΟ : 3 P QO . СФ6 PA 2 T: С 6 PA 2 T CALIC": CALL C2AIT": C2AI C2AT: CXAT C2HT: C2Al CZOYOPT: CAZOY CZOYP-: CAROY CZOYWP": CXSOY C61M: CKIM C60A: 960P

T

COK: COK

T-, T6- def. art. fem. sing.; see 1.3.

TA- absolute rel. fem. sing.; see 22.2.

TAATE (TOOTE) vb. intr. to shine (with: MMO'; ± 680%).

TAATE, CTAATE vb. tr. to clap (hands); to spread (MMO').

TABIP n.m. sanctuary (of the temple).

TAGIO (TAIO) TAGIG- (TAIG-) TAGIO (TAIO) Q TAGINY, TANY

Vb. tr. to honor, pay respect to (MMO); to esteem,

have high regard for, regard as precious; Q to be

honored, esteemed, excellent, valuable, precious; as

n.m. honor, honored state; complimentary gift. peq
TAGIO honored person. +-TAGIO (NA) to honor, give com
plimentary gift to. xi-TAGIO to receive honor or gift.

TAI adv. here, in this place.

тако тако тако Q такну (такнут) vb. tr. to destroy, put an end to (ммо); vb. intr. to perish, be lost, destroyed; as n.m. perdition, destruction. аттако

- indestructible, imperishable; митаттако incorruptibility. речтако (1) destroyer; (2) perishable.
- TAX, TOX n.m. heap, hillock.
- TANO (TANG, TAPO) TANG- TANO' Q TANHY Vb. tr. (± 62PAI) to lift, raise up, offer up, send up (MMO'; upon: exN, 2I-xN); to cause to mount (an animal); to take aboard; to weave (MMO'); vb. intr. to go up, ascend, mount, board; as n.m. raising up, offering.
- талбо талбо- талбо Q талбну vb. tr. to heal, cure (ммо; of, from: 2N, GBOA 2N); vb. intr. to become healed; as n.m. curing, healing. атталбо incurable. речталбо healer. речталбо idem. митречталбо healing power.
- tamio tamie- tamio" Q taminy vb. tr. to create, make $(\overline{n}no")$; to prepare, make ready; as n.m. thing made, creation.
- Tamo Tame- Tamo" vb. tr. to tell, inform (someone: $\overline{M}MO$ "; thing told: e, erse; that: xe).
- тано тена- (тна-) тано vb. tr. to make, create (ммо); to draw up (a deed).
- танго танге- танго" Q тангну vb. tr. to bring to life, keep alive, let live (ммо"); vb. intr. to become alive; as n.m. keeping alive, saving. речтанго savior, lifegiver; митречтанго life-saving.
- Tangoyr Tanger- Tangoyr Q Tanghyr vb. tr. to believe, trust ($\overline{M}MO^*$; that: xc); to entrust ($\overline{M}MO^*$; to: c, $cx\overline{N}$; also reflex.); to confide (in: $M\overline{N}$).
- Tan n.m. horn, trumpet. $na-n\tau an \overline{n}$ oyot unicorn. 204 \overline{n} tan horned snake. $ag-\tau an$ to sound trumpet; as n.m. trumpet blast.
- TANN, TENN, TENNE N. Cumin.
- rango n.f. mouth; also fig. of well, sword, tomb. (\overline{n}) rango n rango mouth to mouth, face to face.
- TAP n.m. sprig, branch.
- тарко (терко) тарке- тарко" (терко", т \overline{p} ко") vb. tr. to adjure, cause to swear (\overline{m} mo"; by: \overline{m} mo", ката, ε).
- tayo (taoyo) taye- (taoye-) tayo" vb. tr. (1) \pm eboa: to

send, send forth (ΜΜΟς; to: ε, ΝΑς, φΑ; after, for: Ν̄CA); to put forth, produce; (2) to cast (ΜΜΟς; forth: εΒΟΛ; down: εΠΕCΗΤ, ΕΖΡΑΙ); (3) to tell, proclaim, repeat, recite (ΜΜΟς; to: ε, εΤΝ, ΝΑς). As n.m. mission; + εΒΟΛ: product. ΑΤΤΑΥΟ indescribable, inexplicable.

тадо тадо- тадо vb. tr. to increase (ммо); тадо- + Inf. to do something much, more; increase in doing.

TAY n.m. spittle. Nex-/cer-TAY to spit.

TA20 TA26- TA20° Q TA2HY Vb. tr. to cause to stand, set up, create (MMO°); to reach, attain, meet, catch up to (MMO°); to arrest; to befall (someone: obj. suff.; that: e, erpe); to assign (MMO°; to: e); vb. intr. to be able, to manage (to do: e + Inf.). ATTA20° unattainable, incomprehensible; MMTATTA20° incomprehensibility. peqTA26- catcher. TA26 (c)par° to set up, establish (MMO°); as n.m. establishment, right order.

 $T\lambda 2\overline{T}$, $T\lambda 2T\overline{2}$, $T\lambda 0$ n.m. lead.

TART? (TART) TERTWR' Q TERTWR vb. tr. to mix, confuse $(\vec{M}MO^2)$; as n.m. mixture, confusion.

Taxo vb. tr. to judge, condemn; as n.m. judgement.

TAXPO TAXPO TAXPO Q TAXPHY (TAXPAGIT) vb. tr. to strengthen, affirm, confirm, make fast (ΜΜΟς; in, with: 2Ν; on, to: e, exN); to direct firmly (ΜΜΟς; toward: e); vb. intr. to become strengthened, resolute; to rely (on: exN); as n.m. firmness, strength, resoluteness; 2Ν ογταχρο firmly, certainly. †-ΤΑΧΡΟ to give strength.

ΧΙ-ΤΑΧΡΟ to receive confirmation.

тає, так n.m. lump, cake.

TAGGE, TATGE n.f. sole of foot; foot-print. $y\overline{c}-\overline{n}-\tau$ AGGE n.f. foot-print. $x_1-\tau$ AGGE NCA to follow. ATX1-TAGGE not to be tracked.

TBA n.m. ten thousand; see 30.7. GICTBA five thousand.

THE RO TERG- TERO' Q TERHY Vb. tr. to make pure, purify $(\overline{M}MO''$: of, from: G, GROA $2\overline{N}$, 2λ); vb. intr. to become pure, clean, clear; as n.m. purity, purification; $2\overline{N}$

OYTEBO in a ritually pure way; MA \overline{N} TEBO place of purification.

твыр n.m. a kick. мех-твыр евох to give a kick. ф-твыр to kick (at: егоүн є/гі); речф-твыр kicker.

TEKE- TEKO to send.

твин (pl. твиооує, тевиноу, тчину, твису) n.m. beast, domestic animal. па-итвин cattleman. миттвин bestial nature. речсанф-твин cattle-breeder.

тыт, тныт, тнчт n.m. fish. Gn-тыт to catch fish; речептыт fisherman. са n тыт fish-monger.

re fem. sing. pron. and copula; see 5.1.

те, тн n.m. time, season, age. $\overline{H}/2\overline{M}$ печте at the proper time. ε_1 ε пт ε to come of age. \overline{P} - τ ε idem.

TEAHA vb. intr. to rejoice (over: $\epsilon \times \overline{N}$); also used reflex. with $\overline{N}NO^*$; as n.m. joy.

τερποσεκ, τερποσε n.f. baked brick.

TERNE n.f. forehead.

τεστωσ Q to be pressed down.

тные, тыве, теве n.m. finger, toe; as measure: finger's breadth, any small quantity.

THUE n.m. dam, dike.

тыр adj. all, all of, the whole, every; normally follows noun in apposition; see 16.4. птыр the whole of creation, everything; the All (Gnostic); є птыр wholly, completely, (not) at all.

- THY, THOY, TGY (TOY-) n.m. wind, breath. TOY-PHC southwind. κω Η ΠΤΗΥ, κλ-ΠΤΗΥ to die; NGX-ΤΗΥ GROX idem. λΝΨ-/λΝΞ-ΤΗΥ, CGK-ΤΗΥ to breathe, draw breath. 2H-ΤΗΥ n.m. breeze. 2λ-ΤΗΥ n.f. whirlwind. ΧΙΝΤΗΥ n. windblight; F-ΧΙΝΤΗΥ to become blighted by the wind.
- † (τσι, †ι, †σι) †- τλλ (†) Q το (τω) (imptv. μλ μλμλτ, κησι) vb. tr. to give (μκο; to: κλ, σ); to
 pay out; to sell (for: ελ); to put, place; vb. reflex.
 to go, betake self (to: σ, σρπ, σεογκ σρπ, σεογκ σ), to
 begin (to do: σ, στρο); vb. intr. to move, go; Q impers.

CTO it suits, befits (someone: NA*; to do: 6 + Inf.).

As n.m. gift, bounty; pear giver, fighter; Mutpear generosity; F-pear to become a giver. TAI- (p.c.) one who gives. XI-+ to buy and sell; to hesitate; as n.m. exchange. (Transitive idioms) (obj. MMO*): GTN, NTN: to entrust to, to enjoin, command. GXN: to add to, apply to.

21: to dress in, put on, don (Q TO 21 being worn by).

GROA: to sell, give away. GRECHT: to put down (into: 6).

G20YN: to hand in, give in; to invest. G2PAI: to send, give up (to: 6). Intransitive idioms:) GXN: to fight for. MN: to fight against, struggle with. NCA: to pursue. OYBG: to fight against. GON: to move forward.

G20YN & to strike upon, against; to oppose. For cpds. with }- see 2nd element.

ተቴሮ, ተፍቴሮ, ተቼር n.f. heel; x_1 -ተቴሮ to trip (፴ሐ০°); мቫፐ x_1 -ተቴሮ tripping.

†не, тіме (pl. тме) n.m. village, town. рмтне villager.

taloy fifty. Me2tor fifth. Me2ταιογ fiftieth.

the n.f. loins.

+26, Q τλ26 vb. intr. to become intoxicated (with: 2λ, Μηο*, 2N); as n.m. drunkenness. p64+26 drunkard.

truer n.m. box; bee-hive.

tee n.f. gourd, vegetables. MA N 2APC2 N tee gourd-bed.

TKAC, KAC n.m. pain. \uparrow -TKAC to give pain (to: ϵ); as n.m. pain.

тан n. drop; нрп п тан filtered wine.

TAOM, TAOOM, TNOM n.m. furrow.

 $\tau \overline{\lambda} \tau \overline{\lambda}$ vb. tr. to let drip ($\overline{M}MO^{\sigma}$); intr. to drip. $\tau \overline{\lambda} \tau \iota \lambda \varepsilon$ n.f. drop.

TH- negative prefix for Temporal, Conjunctive, Conditional, and Infinitives. See Gr. In.

TMAGIO (TMAIO) TMAGIC- (TMAIG-) TMAGIO" (TMAIO") Q TMAGIHY (TMAIHY) vb. tr. to justify (MMO"), to regard or hold as justified; intr. to become justified; as n.m.

- justification.
- THH n.f. reed mat. carT-(T)HH mat-weaver.
- тймо (тмо) тймє— (тмє—) тймо" (тймє", тмо") Q тймну Vb. tr. to feed, nourish (ймо"; with: ймо", zй). тймє" осік to feed (someone) bread.
- тыты тыты- Q тытым vb. intr. to become heavy; + GBOA: to resound, reverberate.
- TM20 TM26- TM20" vb. tr. to kindle, set afire (MMO"); intr. to burn, blaze; as n.m. burning, heat.
- THNO (TNO, TNA) THNO (TNO, TANA) Q THNHY Vb. tr. to pound, tread down (אור to be beaten, trodden; Q to be contrite; as n.m. breaking, contrition.
- THROOY THREY- (THEY-) THROOY" (THROOY", THOOY") Vb. tr. to send (HMO"; to: ϵ , ϵ pat", μ a", μ a; for, after: π ca); also W. ϵ Boa, ϵ 20YN, ϵ 2PA1.
- $T\overline{N}_2$, TNA_2 , $TGNA_2$ n.m. wing, fin; also fig. of building, ship, etc. $\overline{p}-T\overline{N}_2$ to become winged. $pGT-T\overline{N}_2$ (Q pHT \overline{N}_2) idem. $x_1-T\overline{N}_2$ to take wing.
- το ετΕ τΕτωε vb. tr. to form, fashion, fabricate.
- τος, το, τοις, τλ, τλς n.f. part, portion, share. x_1 -τος to partake of $(2\overline{N})$. Μλι-τος \overline{N} 20γο covetous. Μπτηλι-τος \overline{N} 20γο covetousness.
- TOE, TO n.f. spot. \overline{p} -TO (Q o \overline{N} TO) to become spotted. TOELC, TOELC n.f. piece of cloth, patch, rag; purse. TOELT vb. intr. to mourn (for: ε , $\varepsilon x \overline{N}$); as n.m. lament. TOING vb. intr. to rise up.
- ток, тоб n.m. knife, razor.
- TOM n.m. reed mat.
- TONTN ΤΝΤΝ- ΤΝΤΟΝ (ΤΝΤΟΝ (ΤΝΤΟΝΤ) (1) vb. tr. to liken, compare (ΜΜΟ"; to: ε, ΜΝ, εχΝ); intr. to be like, comparable to. (2) vb. tr. to estimate (ΜΜΟ", ε), speculate about. As n.m. likeness, similitude; oracle.

 †-ΤΟΝΤΝ to guess, surmise. ΔΤΤΟΝΤΝ without comparison.

 P Ε Θ ΤΟΝΤΝ diviner.
- TON2 O TON2 vb. reflex. to become entangled (in: ϵ); to

converse (with: MN). TOOBER n.m. foliage. TOOTE (TAATE) vb. tr. to turn; intr. idem (cnagoy: back). TOOY n.m. mountain; monastery; desert cemetery; as adi. hill-, wild, desert-. PMNTOOY mountain man. 22NTOOY n.m. mountainous country. Toyer pl. of tooy. TOOY TEY- TOOY' vb. tr. to buy. τοογε n.m. shoe, sandal; pair of shoes. Moyc π τοογε shoelace. Gon W Tooy shoe-sole. TOOYTE TOYHT' O TOYHT vb. tr. to collect, gather. τοπ, τωπ n.m. edge, border, hem; keel; bosom, embrace. τορτρ τρτρ- τρτωρ Q τρτωρ vb. tr. to drive in (nail, sword: \overline{M} Mo"; into: 6, $2\overline{N}$); to pierce (\overline{M} Mo", 6). TOYA n.m.f. doorpost, lintel. τογειο (τογιο) τογιο vb. tr. to repay, give back (μωσ); as n.m. repayment. $xwwme \overline{N}$ toyero bill of divorce. (TOYNO) TOYN- (TOYNO-) TOYNO' (TOYNOY') vb. tr. to open. TOYNOC (TOYNOYC) TOYNOC TOYNOC Vb. tr. to awaken, raise up, set up (ммо"; from: ввох гм); to incite (ммо"; against: exīi); as n.m. raising; peatoynec- one who raises. τογο τογο' vb. tr. to show, teach (to someone: Μ̄μο'; something: ε; or vice versa); intr. to learn. Also = τλγο. τογω' n. bosom. στογπ-, στογεμ-; στογω' prep. at, near, beside. 21TOYN-, 21TOYEN-; 21TOYW' idem. τογωτ, τογοοτε n.m. pillar; idol. TOYMO TOYME- TOYMO' Q TOYMHY vb. tr. to make whole (MMO'); to save, rescue (from: e, eTN, MMO", 2N, EBOA 2N, 21TN); intr. to be saved, safe; as n.m. safety, salvation. TPA, 6pa n.f. extremity (of limbs); joint. трвнии, тирвнии, тервееии, терчееии n.m. papyrus plant. τρε, τρα n.m.f. kite (bird). TPIM, ETPIM n.m. clover. TPIP n.f. oven.

TPO TPG- vb. tr. to cause to do; rare except as prefix of inflected (causative) infinitive. See 20.1.

- TPPE, Q TPEIWOY vb. intr. to be afraid (of: 2HT' \overline{N}). TPWM n.m. hurricane.
- тсаво тсаве- (тсеве-) тсаво" (тсво") Q тсавну(т) vb. tr. to make wise, teach, show (ммо" of person; e of thing taught or vice versa); as n.m. teaching, instruction. ммтречтсаво teaching. саво (сво) саво" to learn (e).
- TCAGIO (TCAIO) TCAIG- TCAGIO Q TCAIHY Vb. tr. to make beautiful.
- TCANO (TCNO) TCANG- TCANO' Q TCANHY(T) vb. tr. to set in order (MMO'); to adorn; to furnish, provide; as n.m. propriety, order. MAI-TCANO' fond of adorning self.
- το $\overline{\textbf{κ}}$ κο το $\overline{\textbf{κ}}$ κο το $\overline{\textbf{κ}}$ κο $\overline{\textbf{V}}$ $\overline{\textbf{V}}$
- тсенко (тсйко, сйко) тсенко vb. tr. to nurse, suckle (Ммо"); аттсенко not giving milk (of breasts).
- $\tau \bar{c}$ io ($\tau \bar{c}$ eio) $\tau \bar{c}$ ie- $\tau \bar{c}$ io Q $\tau \bar{c}$ iny (ciny) vb. tr. to sate, make satisfied (\bar{m} mo'; with: \bar{m} mo', $2\bar{n}$).
- тсо тсе- тсо Q тснү vb. tr. to give a drink to, slake thirst of, water (мно ; with: мно); as n.m. watering. ма м тсо drinking-place; речтсо drink-giver.
- TCTO (CTO, COO) TCTG- (CTG-) TCTO' (CTO') Q TCTHY (CTHY)

 vb. tr. to bring back, return (MMO'); reflex. to go

 back. TCTO GBON vb. tr. to reject, throw out (MMO');

 as n.m. rejection. TCTG GBON, CTG GBON, CTGBON n. what

 is rejected. TCTO GRN2OY to turn (MMO') back; also re
 flex. COO in MAN COO n.m. lodging, retreat.
- тте- тто' vb. tr. to cause to give, require of (ммо').
- TWEE TEEC- (TEC-) TORC vb. tr. to goad, incite ($\overline{M}MO^*$, ε); as n.m. pricking of conscience, compunction.
- TWB \overline{z} (TWB λz) TGB \overline{z} (TWB z-) TOB z' vb. tr. to pray, make entreaty (to: $\overline{M}MO^*$; for: \overline{e} , \overline{e} $\overline{E$
- Tωκ τεκ- τοκ Q τηκ vb. tr. to strengthen, confirm (ΜΜΟ); to stiffen, thicken; vb. intr. to become strong, firm, thick; vb. reflex. to strengthen oneself. τωκ ε2ΟΥΝ to persist, be confident (in doing: Circum.); to endure (ε).

- †-τωκ to strengthen, fortify (NA*). x_1 -τωκ to take courage. τωκ \overline{N} 2HT intr. to become strong of heart, take courage, rely (upon: $ex\overline{N}$, $2\overline{N}$); as n.m. confidence. †-τωκ \overline{N} 2HT to give confidence (to: NA*); x_1 -τωκ \overline{N} 2HT to take courage.
- τωκ (τω66) τεκ- (τκ-) τεκ- Q τηκ νb. tr. to throw ($\overline{κ}ηο-$); τωκ εβολ to cast forth, exude ($\overline{κ}ηο-$). ρε4τκ-coτε archer.
- тык (тык, -е) ток (так, тое) vb. tr. (1) to kindle (fire), stoke (oven), obj. w. ммо от е. ма м тык stoke-hole (of bath-house); речтык stoker. (2) to bake (bread: ммо); as n.m. baking. ма м тык bakery. тік n.m. spark. тыкм текм токм Q токм (такм) vb. tr. to pluck (ммо); to
- тшкн текн- токн° Q токн (такн) vb. tr. to pluck (мно°);
 draw (sword). тшкн н гнт to become troubled.
- TWKC (ΤωςC, ΤωτC) ΤΕΚC- ΤΟΚC' (ΤΟςC') Q ΤΟΚC (ΤΟςC, ΤΟΤC)

 vb. tr. to pierce, goad, bite (ΜΜο'); to drive (nail:

 ΜΜο'; into: ε); to point (finger: ΜΝο'; at: ε); Q to be

 nailed, fastened (to: ε), to be inlaid (with: ΜΜο'); to

 be pierced (with: 2Ν). ΤωκC, ΤωςC n.m. piercing. ΤωςC,

 τοοτC n.m. a fixed seat. Τλλ n.m. molar tooth.
- τωλκ τολκ vb. tr. to pluck out (ΜΜΟ΄).
- ΤωλΜ τολΜ Q τολΜ vb. tr. to defile, besmirch, pollute (ΜΜο΄); vb. intr. to become defiled etc. (with, by: 2Ν, ΜΜο΄); as n.m. stain, pollution. †-ΤωλΜ ε to stain.

 λΤτωλΜ stainless, unpolluted.
- TWAC, Q TOAC Vb. intr. to become stuck, sink (in: ϵ , $2\overline{N}$).

 TWM TEM- (TM-) TOM' Q THM Vb. tr. to close, shut ($\overline{M}MO'$);

 Vb. intr. idem.
- TWM, Q THM vb. tr. to sharpen; vb. intr. to become sharp.
 TWMNT (TWMT), Q TOMNT vb. intr. to meet, befall (someone:
 6, 620YN 6); as n.m. meeting, event.
- TWMNT vb. intr. to become amazed, stupefied.
- TWMC TEMC- (TNC-) TOMC' (TOMEC') Q TOMC vb. tr. to bury (MMO'; in: ϵ , $2\overline{N}$); MA \overline{N} TWMC burial place.
- Two adv. where? how? ϵ two whither, where to? \overline{N} two = Two. GBOX two whence? $2\overline{N}$ two where? QA two whither?

- xin τωπ from where? pmnτωπ a person from where?
 τωπ in †-τωπ vb. intr. to quarrel, dispute (with: μπ, ογες,
 ε2ογπ ε2ρπ, ππλ2ρπ; about: ετες, εxπ); †-τωπ as n.m.
 dispute, strife. xt+-τωπ without strife. ρεч+-τωπ
 quarreler; μπτρεч+-τωπ faction; discrepancy.
- TWNOY, TWNE, TWNA, TONOY, TONW, TONNE, TONNOY adv. very, greatly; certainly; ce TWNOY yes indeed.
- TWOYN TOYN- TWOYN' vb. intr. and reflex. to arise, rise, rise up (± 680λ, ± 62 pλ1; against: 6, 6xN; from: 21, 21xN, 2N); TWOYN 2λ to lift up, bear; vb. tr. to raise, carry (ΜΜΟ'); as n.m. rising, resurrection (± 680λ).
- τωπ τοπ' vb. tr. to stop up, plug, caulk (\overline{M} Mo'). Me2-τωπ needle, peg. 2λΜ \overline{N} τωπ idem.
- τωπε τεπ- τοπ (τωπ) vb. tr. to taste (\overline{M} Mo). †πε π.f. taste; x_1 -†πε to taste (\overline{M} Mo); as n.m. tasting.
- τωρε n. willow. κω π τωρε willow tree.
- Twpe, tope n.f. (hand); handle; spade, pick, oar. F-twpe to clap. stamp. ωπ-τωρε (to grasp hand), to be surety, stand as surety (for: Mmo'; to: Nx'); as n.m. surety; x1-gπ-τωρε to take as surety; †-gπ-τωρε to give surety; $peqq\overline{n}$ -twpe guarantor. $qpwpe, qtope = q\overline{n}$ -twpe; $p\overline{m}qtwpe$ guarantor. Toot hand, in literal sense with many verbs (cf. kw, moy2, cwk, ama2Te, elw, elne, cmine, oyw2). eipe N a (na) toot', F-a (na) toot' to endeavor, make an effor (to do: ϵ , $\epsilon \tau \rho \epsilon$). $\kappa \omega \overline{N} \tau \cot' \epsilon \delta \delta \lambda$, $\kappa \lambda \tau \cot' \epsilon \delta \delta \lambda$ to cease (doing: Circum.); to despair; to stay one's hand; ATKA-TOOT' GBOX unceasing. + N TOOT', +-TOOT' to give a hand, to help; the suff. pron. on root usu, refers to the object, as in Ait-Toota I helped him. +-N TOOT' MMO'/6 to lay hold of, seize (suff. on TOOT' is reflex.). +- N TOOT' MN to assist, give aid to. +-TOOT' as n.m. help; peq+-roor' helper, assistant. ωπ-τοοτ' to grasp hand (in greeting, promising etc.); to betroth (obj. suff. of woman; Na to man); Q TOOT onn Na she is betrothed to (see gloss on Lk. 1:27). NCA TOOT' adv.

immediately, forthwith (suff. refers to subject of clause). eth (etoot') prep. to, into the hand of; freq. with verbs of giving, entrusting, etc. Nth (Ntoot') prep. (1) from, from the hand of, from by; (2) with, by, beside; in the hand of; (3) because of, through. ebox Nth from. Nte = Nth. 2Ath (2Atoot') prep. beside, with, near; subject to, under the hand of; virtually interchangeable with 2A2th q.v. 21th (21toot') prep. by the hand of, through the agency of, by, from; (of time:) during, after; (of place) out through, from; ± ebox: expresses agent after passive verb.

тωрπ төрπ- (трп-) торп' vb. tr. to seize, rob (ммо'; from: мтм, 2м, 21); to master, acquire; to carry off (to: 6); as n.m. plunder; маι-тωρπ plunder-loving.

тωρ π τορ π v to tr. to sew, stitch (ммо"; to: ε).

ATT ωρ π unsewn. Νκλ Ν τωρ π needle. ρε чτωρ π tailor.

τωρ π n.m. staircase. τωρ τ ρ τ n.m. ladder, step, degree.

τωρ φ v b. intr. to become red; as adj. red. τροφ, Q τορ φ

to become red. τροφρ φ (τροφρεφ) Q τ ρορ φ (τρεφρωφ) to

TWP \(\bar{z}\), \(Q\) TOP \(\bar{z}\) (TAP \(\bar{z}\), \(TA^2\bar{p}\)) vb. intr. to become sober, alert.

TWC TEC- TOC' \(Q\) THC (TEC) \(\pm \) eBOA vb. tr. (rare) to stiffen, fix; intr. to become stiff, hard, firm, fixed. ATTWC adj. limp. TAC-BAA impudent; MNTTAC-BAA impudence.

become red; as n.m. redness.

TWO TET- TOT' Q THT Vb. tr. to join together, mingle (MMO'); to level; vb. intr. to be agreeable, to agree (with: MN; to: e; on, upon: exN); to be persuaded, satisfied; to be joined; to become even, level. As n.m. agreement, mingling. TWOT M (')2HT, TET-N(')2HT to persuade, satisfy. TWOT N 2HT to consent, agree; as n.m. consent, agreement; ATTWOT N 2HT unconvinced.

TWUE, TWUE, TOTE n.f. fringe, border (of garment).

TWUES, TWEE n.f.m. brick; NAME-TWOES to make bricks.

TWUES, TEEG- TOOS' vb. tr. to repay, requite (MMO'; to:
NA'; for, in place of: 6); as n.m. requittal,

repayment. PEGTOWBE one who repays.

- тоове, тоове Q тоове (тове) vb. tr. to seal, set or stamp with a seal (ммог, е, срм; with: ммог, гм); as n.m. seal, stamp. тоовес n.f.; тоовеч, тоовч n.m. impress of a seal. тыве n.f. seal.
- TWOME, Q TOOME vb. tr. to join; mostly in Q: to be joined (to: ε); to be fitting, suitable (for, to: ε, Νλ΄).
 TWOME, TOOME n.f. purse, wallet.
- TWON (TWN) TEN- TON' Q THN (THHN) vb. tr. to accustom (MMO'; to: e); intr. to become accustomed (e: to), familiar with; as n.m. custom, usage. TONG, TANG, TANG n.f. custom, habit.
- Tww66 (Tw66) Te6- (Tek-) To06 (To6*, Tok*, Took*) Q TH6 (THK) (1) vb. tr. to join, attach (ΜΜΟ*; to: e, exN); to ascribe, impute (ΜΜΟ*; to: e); vb. intr. to join self (to: e, ε2ογη e), to cling; Tww66 ΜΜΟ* 6ΕΟλ to publish. (2) vb. tr. to plant (ΜΜΟ*); as n.m. planting.
- Two Teop Toop Q They vb. tr. to limit, bound, determine
 (ΜΝΟ΄); to appoint, assign, destine (ΜΝΟ΄; to, for,
 over: ε, ε2ογΝ ε, εΧΝ, ΝΛ΄); vb. intr. to become fixed,
 limited, determined; to be moderate. As n.m. ordinance,
 destiny; manner, fashion; affair, matter. λΤΤΦΟ unlimited; immoderate. F-(n) TΦΟ to prepare, put in order.

 †-(n) TΦΟ to give orders (to: ΝΛ΄, ε), provide (for: ε,
 ΝΛ΄). ρε τΦΟ commander. ΤΟΟ, ΤΦΟ (pl. ΤΦΟ, ΤΟΟΟ)?)
 border, boundary, limit; nome; province, district; bishopric; ρΜΝΤΟΟ man of nome; F-ΤΟΟ ΝΛ΄ to be adjacent to;
 †-ΤΟΟ ε to set limits to; Χι-ΤΟΟ to adjoin (ε, ΜΝ).
 ΤΕΦΕ (pl. ΤΕΦΕΕΥ) n.f. neighbor: that which adjoins.
- Tw2 T62- (TA2-) TA2' Q TH2 vb. tr. to mix, stir (ΜΜΟ'; into, with: 6, ΜΝ, ΜΜΟ', 21, 2Ν); vb. intr. to become mixed, disturbed, clouded; as n.m. mixture, disturbance.

 ATTw2 unmixed, distinct. MA1-Tw2 meddlesome. P69Tw2 meddler, mixer; MΝΤΡ64Tw2 confusion.

τως n.m. chaff.

- Twee Tale Q Tale vb. tr. to moisten, soak. eas n.m. leaven; F-eas, x1-eas to become leavened. areas unleavened.
- TW2M TA2M' vb. tr. to chase, pursue (Mmo', Nca).
- Tw2 \overline{N} T62 \overline{N} TA2M Q TA2 \overline{M} (± 620YN) to summon (\overline{M} MO'; to: 6, 6 $\times \overline{N}$); to knock (on, at: 6); as n.m. calling, convocation.
- Tweet Teet Tale (Toet) Q Tale vb. tr. to anoint (MMo; with: MMo, 21, 2N); to pour (MMo, on: e); as n.m. anointing. $x_1-\tau \omega_2 \overline{c}$ to be anointed.
- τως \overline{n} τες \overline{n} (τλς \overline{n} -) τος \overline{n} ' (τλς \overline{n} ') vb. tr. to push (\overline{n} mo').
- τως \overline{p} (τωκ \overline{p} , τω $p\overline{g}$) Q τος \overline{p} (το $p\overline{k}$, τωκ \overline{p}) vb. intr. to be joined, fixed (to: e; in: $2\overline{N}$).
- Tw6c To6c' (Tλ6c') Q To6c vb. tr. to bleach, dye (พักง').

 eвью евью евью Q евьину(т) vb. tr. to make humble,
 humiliate (พักง'); intr. to become humble, be humiliated;
 as n.m. humility.
- өни n.m. sulfur.
- τειο (Θιο) Θια- Θιο' Q Θιμγ vb. tr. to cause to fall, bring down (\overline{N} μο').
- θλο θλο' vb. tr. to cause to fly, to chase away.
- өмко өмкө- өмкө Q өмкнү vb. tr. to afflict, treat badly (ммо*); as n.m. affliction, ill-treatment.
- өмо vb. tr. to warm (ммо°).
- өмсо өмсө- өмсо Q өмсовіт vb. tr. to seat (ммо).
- eno (eno) $\tau \overline{z}$ ne- eno Q enhy (± 620yn) vb. tr. to cause to approach (\overline{M} no'); to hire.
- eno eno" (eno", eeno", exno") Q enhy (eenhy) vb. tr. to pound, crush (\overline{m} mo").
- T20 (00) vb. intr. to become bad; as n.m. badness.
- οσογτ, θωογτ, θωθ n. name of 1st Coptic month.
- τεπο (eno) τεπο vb. tr. to lead, accompany (Μπο; εκολ: forth).
- T2 Pgo (e Pgo) e Pgo- e Pgo vb. tr. to terrify, oppress.
- T6 λ 610 (6 λ 610) (T)6 λ 616- (T)6 λ 610 Q (T)6 λ 61HY Vb. tr. to disgrace, condemn (\overline{M} MO'); vb. intr. to be disgraced,

condemned (to: 6); as n.m. disgrace, condemnation.
61616, 61614 adj. ugly; as n. ugly person; MNT61616
disgrace, ugliness. o N 61616 to be disgraceful.
61 n.m. ugliness.

THOOY : THOOY T-: N-TENOY: OYNOY TA: TO6 TEN-: TWN6, TWWN TNOY": TNOOY TENN, TENNE: TANNTNPOMM6: POMM6 TA-: NA-TAANC: TOON TEPBAGIN: TPBHIN TO: TO6 таате: тооте TEPBEEIN: TPBHIN TORE: TOORE TY6: 106 TEPKO("): TAPKO TOLE: TOE TAGIOY: TOY терчееін: трвнін TOK": T0066 TAI: HAI TEY": TOOY TOX: TAX **TAK: TA6** TEY: THY $TOM\overline{N}T: TWM\overline{N}T$ TAK": TOK TEYNOY: OYNOY TONNG, TONNOY: TWNOY **TAKM: TWKM** TEOE: TWO TONOY: TWNOY TAKP: TOOP тефеек: тефе TONO: TONOY TAX: TOKE TG2TW2(*): TA2T2 TOOR': TOORE TANA": TNO T66-: TW066 TOOBE (C/4): TOOBE **ΤΟΟΚ': Τωω**66 TANZET-: TANZOYT TH: T6, NH TAN2HT: 2HT -TH: 10Y TOOME: TOOME THEE: TAIRE, THEE TANZHYT: TANZOYT TOOT': TWP6 THBT: TET TOOTE: TAATE TAREN: TARRτληζ: τωωπ THHEE: TAIBE TOOYE: 2TOOYE TAPO: TANO тннп: тооп TON': TWNE, TWWN, TWN TAPE: TWPE THK: TWW66 TONG: TWWN TACBAA: TWC της: τωως торе: торе TATC6: TA6C6 тирвии: трвии TOPK: TW6P TA8: TA2T TH4T: TET TOTE: TOTE TAGE-WPK: WPK TOTE: TWKE TH2: TO2 TA2-/*: TW2 TH6: TWW66 TOYHT(*): TOOYTE TA2B(*): TW2B t: toy TOYN-: TOYW' TA26: T126 te: toy TOYN-: TWOYN TA2HY: TA2O 十K: TWK TOYNEC -: TOYNOC TA2M(*): TO2M the: Twne τογοοτε: τογωτ TA2P: TWP2 TA0066: 6A0066 TOYPHC: THY TA2C': TW2C TAO6: 6A06 TOO: TOO TA6N("): TW6N TATIAG: TATA TO6 (*): TOK, TWK TA6C": TW6C **Τλω66: 6λ0066** T06": T0066 TBBE: TWWBE TM 6-/": TMMO TOGC", TOGE: TWKE TENEY: TENH тие: тие тре-: тро TENOOYE: TENH TMO(*): TMMO TPELOOY: TPPE TBC: 十BC тйпсоп: соп TPH: TP6 re: ne TNA-: TANO TPKO": TAPKO TEBE: THHEE ONNT : KNT Trop: Twp TEBET: TOOBE THEY-: THOOY трофра: тора TEBC: +BC THH: GITN TPPOMME: POMME T6K-: T0066 $T\overline{N}NEY-: T\overline{N}NOOY$ TPOPWO: TWPO TENAT: TANO TNO(*): TNO TCBO": TCABO TENA2: TN2 TNOM: TAOM TCEBET: TCABO

TON: TONG, TOOR TW6C: TWKC TENO: TCANO TOPE: TOFF OAB: TW2 B TTO": TT6-TOTP: TOPT ONO : ONG TO : TO TOTE: TOKE 96NO': 9NO TWBA2: TWBZ TOUTE: TOTE BENHY: BNO TWE6: TWWB6 TWKP: TW6P TO6: TOK SIHY: SIO TOMT: TOMNT T066: T0066 ewoyt, ewe: ecoyt TONA, TONE: TONOY TOGE: TOK TX1610: X1610 TOR: TOR

OY

oy interrog. pron. what? less commonly: who? oy epo* what does it profit (me, you, etc.)? oy N what of (partitive)? oyoy what? (with indef. art.). F-oy to do what? to be like what? GTBG OY Why? GTBG OY N 2008 idem. OY MN OY this and that, such and such.

oy indef. art. sing. See 2.1.

- ογα (f. ογει) (1) indef. pron. one, someone; (2) one (the number); see 15.3. For -ογε in μπτογε eleven, etc.; see 24.3. ογα ογα one by one. ογα...ογα... one... the other. εθνογα ογα some (pl.), a few. ε πογα distributive: one each. πογα πογα each one. κεογα another one. F-κεογα to become another, be altered. μπτογα unity unison. F-ογα to become one: to unite with (μπ).
- ογλ n.m. blasphemy; as adj. blasphemous. χι-/χε-ογλ to blaspheme (against: ε, ε2ογν ε). ρεαχι-ογλ blasphemer. Μπτρεαχι-ογλ blasphemy. χλτ-ογλ blasphemer.
- ογλλ intens. pron. -self, alone, only; used appostionally, as in Ντοκ ογλλκ you yourself, you alone; τεczime ογλλc the woman herself. Ν ογλλ idem. Ϝ-ογλλ to become alone.

oyaz n.m. pole, stave.

OYA2864, OYA2M64, OYO2864 vb. intr. to bark, growl (of dog). OYA26 n. oasis.

OYA212HT, OYA261HT adj. cruel. MNTOYA212HT cruelty. F-OYA212HT to become cruel.

OYAZME n.f. storey (of a house or structure).

ογελφ, Ο ογοεφ vb. intr. to become white; as n.m. whiteness. oyour adj. white (aft. n., with or without \overline{N}); F-ογωεο to become white. OYBE (OYBH') prep. against; toward, opposite. oye, 0 oyny vb. intr. to become distant, far, far-reaching (± 680x); oye e to be distant from; idem with prep. Mmo, ebol mmo", clbol mmo", ebol 2N. As n.m. distance: oye enecht distance downward. e nove to a distance (from: Mmo'). M noye at a distance. 21/2M noye idem. ove number one in higher numbers (11, 21, etc.). оуестенти п.m. Greek. митоусствити п.f. Greek (language). oyerne vb. intr. to pass by (subj. usu. period of time). ATOYGING not passing, permanent. OYEICE OYACT Vb. tr. to saw (MMO). PEROYEICE Sawyer. oyerre (oyeere) vb. intr. to waste away, dry up; vb. tr. to dry up, make waste away (Mmo'), ± 680%. oyexoyexe vb. intr. to howl. оуєрнтє, оурнитє n.f. foot, leg (of person, animal, object). OYEPT, OYPT, OYAPT n.f.m. rose. oyez-pacoy interpreter of dreams. oyeepo n.f. doorpost. oyup (f. oyupe) interrog. adj. how much? how many? how great? OTHP ne N? How great is N? (also exclam.). F-OYHP to become how much/many? to amount to how much? \overline{N} oyap by how much? to what extent? λ oyap how long? how much more so (in syllogistic statement). COY-OYHP such and such a day. MEZOYHP the how-many-eth? oyure n.f. calamity. OYTAG, OYGAAG n.f. melody, music. oyxne n. depression in skin (left by disease or sim.). ογμοτ, Q ογομπ (ογομπτ) vb. intr. to become thick. swollen; as n.m. thickness, swelling. oyN- existential predicate: there is/are. See Gr. In. OYNAM n.f. right hand, the right side; 61x N OYNAM the

right hand; & oynam to the right. NCA/21 OYNAM MMO

on the right of. THY \overline{N} OYNAM favorable wind. oynoy (pl. oynooye) n.f. hour. N Teynoy adv. immediately. thereupon, forthwith, 2N Teynoy idem. npoc Teynoy for a while, for a moment, at present. PEGKA-OYNOY astrologer. renoy adv. now; renoy 66 so now, now therefore; ϵ/\overline{N} tenoy now; $\omega \lambda$ tenoy until now; x_{1N} tenoy from now. OYNOG vb. intr. to rejoice (at: exn, 21xn); also used reflex. with Mno. As n.m. joy, gladness. $ov\bar{n}\tau$, over n.m. hollow place, esp. hold of a ship. OYNTE- OYNTA pred. of possession: to have. See 22.1. oyoci, oyoi n.m. rush, swift movement. +-oyoci to go about seeking, search (for: e, ezoyn e, NCA). +-noyoei, † й поуові, †-п(') оуові, †-п(') Noyoві to advance, proceed (to: ϵ ; into: ϵ_2 oyn ϵ ; on to: ϵ_2 PAI ϵ , $\epsilon_x \overline{N}$). 61N+-noyoe1 act of going, proceeding. Noyoe1 = oyoe1. ογοει, ογοι interj. woe! (unto: Να"). Rare as n.m. woe. oyogie (pl. oyegih, oyeih) n.m. farmer, cultivator (of fields and vines). There are many variant spellings. MNTOYOGIG husbandry. MA N OYOGIG farm. PMOYOGIG farmer, peasant (var. pMoye, pMoya). F-oyocie to farm. OYOGIN n.m. light; dawn; eyesight. ATOYOGIN without light. P-cycein to shine, make light (for: e, NA"; on: exN, 21XN). PEAD-OYOEIN one who gives light, illuminator. xi-oyogin to get, receive light. OYOGIT n.m. pillar. OYOGIO n.m. time, occasion. HEOYOGIO ET MMAY at that time. \overline{M} nerwoe \overline{N} oyoerg all this while. \overline{N} oymhuge \overline{N} oyoerg for a long time; n oyoeig nim always. M nioyoeig, n neoyoeig at this/that time. \overline{N} (oy)oyoeig at one time (in the past). ga oyogig later on, at a later time. 2M neoyoeig at the time in question. npoc (oy)oyoeig for a time, transitory. Aroyoeig ill-timed, at a bad time. 2N OYMNTATOYOGIO adv. idem. F-oyogio to spend, pass 6M-noyoeig to find time, have leisure.

oyo: particle of assent: yes. Cf. also oyoe.

eyente, eyeente, eyemnte n.f. tower. oyoma n.m. manger. oyon indef. pron. someone, something, some; in neg. context: no one, nothing. Also used as pl.: Noyon, 26Noyon some, some such. oyon nin everyone (s. or pl.). oyooze, oyoze n.f. scorpion. oyoose, oyose n.f. cheek, jaw. oyon, Q oyaas vb. intr. to become pure, innocent, holy (of, from: c, GBOA 2N); as n.m. purity; 2N oyoyon purely; xi-oyon to acquire purity, be hallowed. ner oyas who/ what is holy, esp. a saint; used with art, or poss, prefixes. OYHHB n.m. priest (Christian or pagan): F-OYHHB to become a priest. HNTOYHHB priesthood. OYOCF n.m. oar. peacek-oyocf, peacok N NoyocF oarsman. ογοςτη, Q ογεςτων to become broad, wide, extensive; rarely tr.: to broaden. As n.m. breadth. oyotoyet, O oyetoywt vb. intr. to become green, pallid; as n.m. greenness, herbs; pallor. ογοφογεφ ογεφογωφ Vb. tr. to beat, strike. OYO4 n.m. lung. ογοσογεσ ογεσογωσ vb. tr. to chew, crush (ΜΜΟ). oypac n. crutch. oypot, Q pooyt vb. intr. to become eager, ready, glad; Q to be fresh, flourishing; as n.m. zeal, enthusiasm, eagerness, gladness; 2N oyoypor gladly, eagerly. +ογροτ to gladden; to incite, arouse (someone: Νλ'). oypw n.m. bean. ογρωε, ογερωε n.f. watch, watch-tower. ΔΝογρωε n.m. guard. oyc n.m. bald person; митоус baldness. ογτλ2 n.m. fruit, produce (rare in Sah.). ογτε (ογτω^{*}) prep. between, among; often in proleptic con-Struction; OYTE X MN Y, OYTE X MN OYTE Y, OYTO'Y between x and y. GBON OYTE from among, from between. oyte is often followed by TMHTE in same constructions.

ογω n.m. news, report. N-noγω κλ' to bring news to.

- p-ογω to respond, reply, say (to: NA*, 2APN). †-ογω to give news. 41-ογω, 4A1-ογω bearer of news, messenger. 26-πογω, 26-π(*)ογω to inquire (about). X1-ογω, X1- πογω to announce (to: ε, NA*, μΔ); pε4X1-ογω informer. 61NX1-ογω Annunciation. 6M-πογω, 6M-π(*)ογω to inquire (about). 2X-ογω n. dispute, argument.
- ογω vb. intr. to cease, stop, come to an end; to cease

 (from: ΝτΝ, 2Ν, 680λ Ν); + Circum.: to stop doing, finish doing; to have already done. + 6/Ν + Inf. idem,
 but rarer. Vb. tr. to stop, bring to an end (Νπο΄).

 λτογω unceasing, unending. †-ογω to cease; to make
 cease (Μπο΄); to release (from: 2Ν); λτ†-ογω unending.
 ογω n. in †-ογω (± 680λ) to bloom, sprout; to bring forth,
- oyω n. in †-oyω (± eBoλ) to bloom, sprout; to bring forth, produce (ΜΜο΄); as n.m. sprout, blossom. x:-oyω to conceive (a child: ΜΜο΄); as n.m. conception.
- ογων ογεν ογον ο Q ογον ο vb. tr. to put to shame, humiliate, defeat (ΜΜΟ'); vb. intr. to bend down (in shame, weakness, defeat); to lean (on: εχΝ, 21χΝ, 620γΝ ε); as n.m. humiliation. ογων ο Ν 2ΗΤ to be discouraged.
- oyom oyem- (oym-) oyom' vb. tr. to eat, consume (mmo'); also fig.: to submit to (e.g. punishment); (subj. the heart) to make repentant. oyom nca to eat away at, gnaw at. oyom (ebox) 2 n to eat away at, consume; to eat some of. oyom as n.m. food, eating. oyam- in cpds.: eater of (e.g. oyam-pome man-eating). atoyom not eating, without food; mntatoyom being without food. mal-oyom fond of eating; mntal-oyom fondness for eating; ma n oyom eating place, refectory. peqoyom glutton; mntpeq-oyom gluttony. einoyom (pl. einoyoom) n.m.f. food.
- ογων, Q ογκν (imptv. λογων, ογν-) vb. tr. to open (πνο', ε); vb. intr. to open (out on, towards: ε, εχν, ε²γλι εχν, νcλ, ογεε); as n.m. opening.
- OYON n.m. part, portion, piece. oyn- in fractions 30.6. OYON \overline{g} n.m. wolf.
- OYONZ OYONZ- OYONZ (t 680%) vb. intr. and reflex.:

- to be revealed, become manifest, appear; vb. tr. to reveal, make manifest, make clear, declare (mmo'; to: e, e₂ογη e, ηλ'). As n.m. (+ eBOλ) manifesting, showing, declaration; 2N ογογων εBOλ openly, publicly. λτογων εBOλ invisible, not manifest. P-λτογων εBΟλ to become invisible.
- ογωρπ ογορπ' vb. tr. to send (Boh., rare in Sah.).

 ογωρΣ ογορΣ ογορΣ 0 ογορΣ vb. tr. to set free, renounce
- ογωρε ογωρε ογορε Vo. tr. to set free, renounce a claim on; vb. intr. to be free, not responsible. ογωρε n.m. free space.
- ογως σογος σογος ογος συν vb. tr. to leave barren, idle; to keep idle; to neglect; vb. intr. to be idle, come to a halt (for: ε); to be brought to naught. As n.m. idleness, ceasing, cessation, ογως σε n.f. idleness.
- оуют оует- оует vb. tr. to make soft, weak; vb. intr. to become green, fresh, raw, soft, weak. оуоте, оуооте n.m. greens, herbs; ма N оуоте garden.
- ογωτ (f. ογωτε, rare) adj. single, sole, one and the same; usu. aft. n. with \overline{N} . ογω \overline{N} ογωτ each one, a single one. Μπτογα \overline{N} ογωτ singleness, unity. \overline{p} -ογα \overline{N} ογωτ to make one, amount to one and the same thing (with: $M\overline{N}$).
- ΟΥΦΤΕ ΟΥΦΤΕ ΟΥΟΤΕ 'Q ΟΥΟΤΕ 'Vb. tr. to pass through (ΜΜΟ'); to transform, translate, exchange; to remove, make/let pass; vb. intr. to change, be altered (usu. of place or situation); ΟΥΦΤΕ 6 to change into, to surpass; Q to surpass. ΟΥΦΤΕ 6ΒΟΧ to cross over, spread over. ΟΥΦΤΕ 62ΟΥΝ to pass in (into: 6). ΟΥΦΤΕ 62ΡΑΙ to pass beyond.
- ογωτ $\overline{\epsilon}$ (forms as above) vb. to pour (into: ϵ ; upon: $\epsilon \times \overline{N}$; down on: $\epsilon n \epsilon c + \tau + 2 \epsilon$).
- ογωτε (forms as above) vb. to pierce. ογωτε n.f. hole.
- ΟΥΦΤΝ ΟΥΟΤΝ' vb. tr. to pour; as n.m. libation. ΟΥΦΤΝ 6ΒΟλ to pour forth (tr. and intr.); as n.m. libation.
- OΥωτε ογοτε ογοτε vb. tr. to cast (metal), to pour (water), to draw (water); as n.m. anything cast or molten.

 Μλ Ν ογωτε crucible, melting-pot. ρεσογωτε cup-bearer,

- drawer of water. ογοτε n.m. cup. ογοτες, ογωτες idem. ογωως, ογωλε, Q ογοολε (ογολε) vb. intr. to become well off, to prosper, flourish (in: Μπο*, ελ); as n.m. prosperity, plenty. εΝ ογογοολες in abundance.
- оушине (оушине) оуоом (оуом) vb. intr. to break down (emotionally); reflex.: to accommodate (someone: мм).
- ογωστε (ογωτε) ογλλτε- ογοοτ' (ογλλτ') Q ογοοτε vb. tr.

 (1) to send (ΜΜο'; forth: εΒολ); (2) to separate, distinguish, choose (esp. in Q), ± εΒολ. ογωτ οr ογετimpers. vb. followed by subj.: is different, distinct.

 ογετ-...ογετ-... the one is..., the other is....
- ογωφ ογωφ ογωφ (ογοφ') vb. tr. to want, wish, desire (ΜΝο'); to love; to be ready, on the point of (doing: e + Inf., ετρε, χε). ογωφ may be prefixed directly to an Inf., as in †ογωφ-ειμε. ογωφ as n.m. desire, love, wish; ογωφ Ν 2μτ heart's desire; Ν π(')ογωφ on one's own; voluntarily, willingly.
- оушу n.m. cleft, gap; interval, pause, holiday; поушу \overline{N} тинте Thursday (?). ка-оушу to set an interval; †-оушу idem. \overline{P} -оушу \overline{G} to wait for. (\overline{N}) оу \overline{G} prep. without.
- oyoge oyege oyoge vb. tr. to answer (someone: e, Na", rarely suff.). ογωσε ΜΝ to converse with. ογωσε Ναλ to repeat after (in response). ογωσε ογες to testify against. ογωσε as n.m. answer.
- ογωφε (ογωωφε) to consume, be consumed (fire, heat).
- оүюым оүеым оүоым Q оүоым vb. tr. to knead, mix, compound (ммо"; with: мм, гі, гм). as n.m. dough. ма м оүюым kneading place.
- ογωμε ογωμε ογομε Q ογομε to become broad, level, flat; to be at ease; also tr. to make broad etc. ογωμε εκολ tr. and intr. to spread out/forth, extend; as n.m. extent, breadth; ease. ογωμε, ογομες, ογομες n.f. breadth.
- OYwyŦ vb. tr. to greet, kiss; to worship (Μ̄Μο', ε, Νλ');
 OYwyŦ εxN/2λ to worship, do obeisance at. φογ-ογωφΤ

adj. to be revered. oyaqre n.f. worship. ογωφι ογεφι ογοφι (ογλωι) Q ογοφι vb. tr. to break down. crush, destroy (Mmo'); vb. intr. to be worn down, broken, crushed, destroyed. As n.m. breakage, destruction. ογω₂ ογ₆₂- ογ_{λ2} 0 ογ_{H2} (1) vb. tr. to put, place set (Mmo'); Q to be placed, situated, set; oyoz Mmo' & to add to, set on, apply to, bring on; oyw2 eroor to add, repeat, do again (vb. complement in Circum. or e + Inf.); ογως εχπ to add (sthg.: ΜΜΟ΄) to; ογως πτοοτ' to leave (a deposit or pledge) with: oyw2 2x to invest/deposit for/with: oyoz EBOA to set (sthg.: MMO') down, leave; ογω₂ επεснτ to set down. (2) vb. reflex. to put or place oneself; oya2 nca to follow. (3) intr. to live. dwell, reside, be situated, be; oyoz 6 live etc. with, by; oyw2 6xN, 62 PAI 6xN to live etc. on, upon; oyw2 MN to live etc. with; sim. with 2x, 2N, 21, and 21xN. As n.m. place where one stops, stands, dwells. Μλ Ν ογω2 idem. 61NOYw2 manner of life. ογω26 n.m. fisherman; ΜΝτογω26 fisherman's profession. ογωεπ ογεεμ- ογλεμο Q ογοεπ vb. tr. to repeat, interpret (Mmo'); intr. to repeat, answer, respond (to: e, exN, NA", NCA); to contradict, object to (OYBE, 2A, 21); reflex. to respond, repeat. OY62 M- may be prefixed to an Inf.: to re-(do), (do) again. As n.m. answer, objection, interpretation. N OYW2M adv. again. PEGOYW2M one who contradicts: ΜΝΤΡ64ΟΥΦ2Μ opposition, disobedience. 6ικογω2 M opposition. ογλ2 M64 n.m. interpreter. oyuxe (oyuuxe) oyeexe- (oyex-) oyoox* (oyox*) vb. tr. to cut out, cut off (mmo"). ογως n. (pl.) architectural term: entrance, portico. ογως π ογες π- ογος π Q ογος π vb. tr. to break, break down, destroy; vb. intr. to break, be broken. As n.m. destruction, breakage. Δτογωέπ unbroken. ογως vb. to collect (a contribution); as n.m. collection. ογωλη n.m. a loan; ε πογωλη on loan. + ε πογωλη to give

on loan (to: Na"); x1 ϵ noygan to receive on loan. oggh (pl. oggooge) n.f. night. Thage \overline{N} теуgh midnight. \overline{N} ogoggh during a night. \overline{N} теуgh by night. \overline{P} -oggh to pass the night.

ογεορ (f. ογεορε, ογεορε, ογεωρε; pl. ογεοορ) n.m. dog.
ογαλι, Q ογοα vb. intr. to become whole, sound, safe, saved
(in religious sense); ογαλι ε to be saved from. As n.m.
health, safety. λτογαλι unsound, incurable. Μῦτλτογαλι
unsoundness, unhealthiness.

OYA N OYOT: OYOT OYAAB: OYUN OYAHOME: OME OYAPT: OYEPT OYACT": OYEICE OYACHE: OYWCH OYATBG: OYWTE OYAT46: OYWTB OYATE6: WTZ ογλώςε: ολφώς ΟΥλφτε: ΟΥψφΤ OYAZME: OYWZM MSWYO : POMSKYO OYAXE: AXE OYEETH: OYOETE OYEETE: OYEITE OYEL: OYA OYGIH: OYOGIG. оүент: чит OYEPE: OYWPE

OYECTON: OYOCTN OYET": OYWT OYET : OYOUTE ΟΥ6ΤΟΥΨΤ: ΟΥΟΤΟΥ6Τ OYETT-: OYWTE OYEON: OYWO ογεφογωφ: ογοφογεφ OYEOCE: OYWOO OY66C-: W6C OYHHB: OYOR OYHN: OYWN OYHY: OYE OYAWM: HPWM OYN-: OYWN OYOM": OYOOM6 оуоните: оуомте OYOMNT: OYMOT ογοολές: ογωωλέ OYOOM': OYOUME оуоомте: оуомте

OYOOTE: OYOT OYOC6": WCE OYOTE: OYWT OYOT26: OYWTZ оүотч("): оүштТ OYOUCE: OYUUT OYO2BE4: OYA2BE4 0Y026: 0Y0026 OYOX: OYXX! OYPWH: MPWM OYTW': OYTE OYW: 6YW OYWME: OYWWME ΟΥΦΤ: ΟΥΦΦΤ6, ΦΤ ΟΥΦΤΕ ΟΥΦΤΕ ογωτε: ωτε OLMOXE: OLOXE OYWOM: WOM PQ : PYO OY20P6: OY2OP

ω

w interj. O, Oh. wsτ, wqτ n.m. goose.

wbg cbg- obg Q obg (1) trans.: to forget, overlook, neglect (6); Q to be forgotten, neglected; also used reflexively. (2) intr. to sleep, fall asleep; Q to be asleep. As n.m. forgetfulness; sleep. 〒-nwbg to become forgetful, to forget; n. obj. with N; pron. obj. with suff. in 下-n(*)wbg; as n.m. forgetfulness. 下戶可以下戶 nwbg one who forgets. Gbg6, Bg6 n.f. forgetfulness,

- carelessness, sleep; \overline{p} -eage to be forgetful; +-eage, + \overline{N} oyeage e n_2 + \overline{N} to make forgetful; $\times 1$ \overline{N} oyeage to be forgetful. oagc n.f. forgetfulness.
- ωκ N 2HT to be content.
- wκm εκm- Q οκm vb. intr. to become dark, gloomy, changed for the worse (toward: ε, ε₂ογν ε); rarely tr. to darken, alter; as n.m. sadness, gloom.
- wa ox Q HA (mainly Boh.) to lay hold of, take, gather in (MMO'); reflex. to withdraw (± ερο' ethical dat.); wa ε to bring/take to; wa εκοά to take away; wa εεογη to bring in; wa εεραι to lift up; to withdraw. As n.m. harvest, in-gathering. ox π.f. what is collected.
- wak, Q oak vb. intr. to become bent, turned aside, distorted (toward: 6); also with adv. 680a, 6π6CHT, 62PAI, NCA πΑ2ΟΥ. wak ga, 6ak-/k-ga to turn up nose, sneer (at: NCA). ολκζ, ολεκζ n.f. bend, corner.
- wam oam Q oam vb. tr. to clasp, embrace (ε, εεογη ε).
 whk emk omk vb. tr. to swallow (mησ); intr. to be
 swallowed.
- wmc emc- omc Q omc vb. tr. to sink, submerge, dip (ммо); vb. intr. to sink, be submerged, dive (into: є, єгоум є, ммо , га, гм), ± євох, єпесит. As n.m. sinking, diving, baptism. речюмс diver. ємсє, мсе n. submersion.
- wmx 6mx- onx' vb. tr. to wean (mmo'; from: 6BOA 2N).
- when.m.(f.) stone. MA N whe stony place. P-whe to become (like) stone. 21-whe to throw stones (at: e).

 NEX-whe idem. BAK-whe stone-thrower. φΑΤ-whe quarry.

 ENE-N-He precious stone, gem.
- wn \overline{K} onκ* (omκ*, onr*) vb. tr. and reflex. to leap (ε: at, upon; $21\times\overline{N}$ from on; 62 μαι up; 62 μαι $6\times\overline{N}$ up onto; 62 μαι $2\overline{N}/2$ μγ from; 68 ολ out).
- שאק (שאק) vb. intr. to be pinched, contracted.
- wng, Q ong (oong) vb. intr. to become dazed, astonished, dumb with astonishment; to gape (at: אוס העם, פאסא to muse, be in a trance. סאס ה. astonishment.

- wNz, Q oNz vb. intr. to live, be alive; as n.m. life. p64-+-ωNz life-giver. 6:NωNz way of life, means of living.
- wn en- on' Q нn vb. tr. to count; to reckon, regard, consider (mmo'; as: mmo'; as belonging to: e); to ascribe (mmo'; to: e); to esteem or be esteemed. Q is esp. freq. in senses: to be ascribed, related, belonging to. wn mm to number or be numbered among/with; to become part of. wn mca to count, enumerate. As n.m. count, reckoning; atwn unesteemed; +-wn, + m nwn to render an account, give an accounting (of); чі-wn to take a count (of: N, e, mN). нпе n.f. number; athne numberless; xi- нпе to take count (of: Mmo'); atxi-нп countless.
- wp \(\omega \) (wp \(\omega \)) ep \(\omega \) = op \(\omega \) (op \(\omega \)); vb. intr. to be shut in, enclosed; as n.m. frame, siege, enclosing, seclusion; \(\alpha \) τop \(\omega \) unlimited. ep \(\omega \), \(\omega \) n.f. pen, enclosure.
- wpk opk vb. tr. to swear (obj. oath: Mmo'); to adjure, swear to (person: e, na'; concerning: ete, exN, 2a; by, upon: exN, 21); as n.m. swearing, oath. wpk N noyx to swear falsely; as n. false oath; peqwpk N noyx one who swears falsely; μητρεφωρk N noyx swearing falsely. ταge-wpk, peqταφε-ωρk one who swears a lot.
- wpw, Q opw (20pw) vb. intr. to become cold; tr. to scorch.
 wpw epw- (pw-) opw Q opw vb. tr. to fasten, bind, imprison; to close (μmo'; against: ε); intr. to be firm, secure. As n.m. firmness, assurance; deed of security.
 2N ογωρω with assurance, with certainty; diligently, carefully. ε nωρω for sure. Μλ Ν ωρω prison; stronghold.
- remain; to be long past, out of date; oft. w. Circum.:
 to remain doing, continue doing; or + 6 + Inf. idem.
 As n.m. duration, continuance, delay.
- তেত (৩৫০) ৫০৫- (০৭৫৫-) ০৫৫ (০৭০৫৫) Q ০৫৫ vb. tr. to anoint, smear (লালত).

ωτ, ογωτ n.m. fat.

- ωτπ ετπ- οτπ' Q οπ' vb. tr. to imprison, shut in (אומס"; in: ε); ωτπ ε2ογκ idem (ε, 2Ν). א א ש ωτπ ε2ογκ place of confinement. εινωτπ ε2ογκ seclusion.
- wтπ orn Q orπ vb. tr. to load (ммо"; with: ммо"). стпω n.f. load, burden; чы-стпω porter, bearer of burden.
- $ωτ\overline{2}$ (ογωτ $\overline{2}$) οτ2 ° Q οτ $\overline{2}$ vb. tr. to weave; to sew, tie (\overline{M} Mo'; to, onto: ϵ , $2\overline{N}$). ογωτ 2ϵ n.m. warp (on loom).
- ωω (ω), Q 66τ (6τ) vb. intr. to become pregnant (with: $\overline{M}MO^{2}$); to conceive (by: $M\overline{M}$, $2\overline{M}$, 680λ $2\overline{M}$); as n.m. conception. $+-\omega\omega$ to be pregnant.
- ωg eg- og (± 680λ) (1) vb. tr. to utter, sound (ΜΜο); to cry out (to, for: e, e2pλι e, exπ, e2oγn e2pπ, oγse);
 (2) vb. tr. to read (ΜΜο); to: e, κλ, on, about: e, exπ; in: 21, 2π); as n.m. reading. κλι-ωg fond of reading: pequa reader, lector. 61Νως art of reading.
- (wg) og (20g, ong) reflex. only, + 620YN 6: to slip into intrude into.
- wgm (ογωgm) εgm- оgm Q ogm vb. tr. to dry up (mmo*), to quench; vb. intr. to be quenched. λτωgm unquenchable. ωч, ωε, ογч n.m. lettuce.
- ωνε (ωκε) εν- ον΄ (οκ΄) vb. tr. to press (Μπο΄; on, onto: επεκτ ε, ενογη ε, $εx\overline{n}$). $φε \overline{n}$ ωνε fuller's club.
- wчт (wвт) вчт- (ввт-) очт (овт) Q очт (овт) vb. tr. to nail, fix (ммо"; to: e, вгоум е, ехм); Q очт м studded with. вгчт, вгвт п.т. nail, spike; ус-м-вгчт blow or wound of nail.
- ω_2 interj. woe! $\omega_2 \in$ woe to ...!
- w2e, λ2e, Q λ2e vb. intr. to stand, stay, remain (with: behind: enλ2oy); to wait (for: e, Nλ*); w2e oybe to stand against, resist, oppose; idem with eboλ e, Nboλ e. λ2epλτ* (reflex. suff.) to stand (e: before), to attend, to resist; λ2epλτ* exN, 21xN to stand upon, at, beside; to stand against, resist. λ2epλτ* MN to stand with. Also w. oybe against, 21pN at, 2λ2TN, 2λTN before, in the presence of. Mλ Nλ2epλτ* place for standing.

- w2c e2c- o2c' (oc2') vb. tr. to reap, mow (ΜΜο'); as n.m. reaping, harvesting; γλ-ως act of reaping. o2c, oc2 n.m. sickle, scythe; χ1-o2c to wield a sickle; χλ1-o2c sickle-bearer, reaper. oc2q, λc2q n.m. knife, sickle.
- wxn exn- oxn' vb. intr. to cease, perish (from: eвол n, eвол гn); vb. tr. to destroy, make cease, put an end to (ммо'); as n.m. ceasing, destruction. xxn wxn without ceasing. xтшхп unceasing.
- $\omega 6\overline{8}$ (2 $\omega 6\overline{8}$) $66\overline{8}$ Q 206 $\overline{8}$ (06 $\overline{4}$) vb. intr. to become cold, freeze; as n.m. cold, frost. $06\overline{8}$ n. cold.
- $\omega_{G\overline{P}}$ osp* Q os \overline{P} vb. intr. to become hard, stiff, frozen; also tr. to freeze, stiffen.

 $ω6\overline{\tau}$ $66\overline{\tau}$ - 06τ ($0x\tau$, $0\tau6$) vb. tr. to choke, throttle ($\overline{\mathsf{M}}\mathsf{M}\mathsf{O}$).

 ω: ωω, Ο
 ωνδ: ωντ
 ω2: ΟΩ2

 ωΒ: ωч
 ωρΨ: ωρΕ
 ω26: Ω26

 ωΒ6: ωч6
 ω4: 2ωΒ
 ω6Γ: ωCΤ

 υΒΤ: ω4Τ
 ω4Τ: ωΕΤ

9

- grefixed vb. (+ Inf.) to be able to, know how to, be allowed to. See 26.2.
- φλ vb. intr. to rise (of the sun), \pm ε₂ ρ_λ , ε₈ ολ. As n.m. rising (of sun). Μλ \overline{N} φλ the east. ρ_λ-φ_λ eastern side.
- φω- vb. to begin; only in cpds. gω-mice, gω-cwnT. See second element.
- φλ n.m. festival; divine service. 2180λ Μ ngλ excommunicated. gλ ω great festival. 200γ N gλ festival day.

 gλ is also used as vb. to keep festival (for: ε, ΝΜΟ΄).

 P-(n)gλ to keep festival (for: ε); Μλ Ν P-gλ church;

 MNΤΡεΨΡ-gλ occasion of festival; †-gλ to give communion.
- $^{\text{QA}}$, $^{\text{QG}}$ ($^{\text{QANT}'}$, $^{\text{QAAT}'}$, $^{\text{QANT}'}$, $^{\text{QANT}'}$, $^{\text{QENT}'}$) n.m. nose. $^{\text{GWBG}}$ $^{\text{R}}$ $^{\text{QA}}$, $^{\text{GE}}$ $^{\text{QA}}$, $^{\text{CE}}$ $^{\text{CE}}$ $^{\text{QA}}$, $^{\text{CE}}$ $^{\text{CE}}$
- (g) (g) prep. (1) to, toward (a person); (2) to, at (a place); (3) till, at, by, for (a time); (4) up to, to

length of (in reckonings); (5) except (i.e. up to but not including). See cpds. under second element.

שבא, שבאץ, goos n.m. skin.

- קאגף, קאף (pl. קאגף) n.m. skin, hide, leather. אורים, אוף tanner. פאר-קאגף ulcer, sore. אורים, man of skin (i.e. the purely physical person).
- gaap, gape, gape n.m. price; γ-gaap to fix a price, bargain (for: ε); γ-gaap idem.
- galpe (galpe, galp) gap Q gap vb. tr. to smite (epo'); as n.m. blow, stroke; речдаре demon, smiter; митречдаре devilry. митдоор idem.

ωλι n.m. fortune. ηωλι also as name of a god.

gal adj. new (after noun, with \overline{N}).

gaire n.f. couch, cohabitation; F-gaire to lie down.

gaipe n.f. sheepfold.

gas n.m. myrrh.

- gaλιογ, gaληγ, gaριογ n.m. administrative official (title). gaλοογ n.f. water-wheel or the like.
- gapsa, gaspa, gaspa n.m. scorching heat; тну м дарва scorching wind; о м дарва to be scorched, parched; †- дарва to scorch (620үм 6).
- умрке n.m. lack of water, drought; ¬¬¬умрке to be dried up.

 уму, умоу, умоу¬ n.m. use, value; as adj. useful, suitable,

 fitting, virtuous; мтуму useless, worthless, obscene;

 ¬¬¬уму (Q о ¬¬) to be useful, suitable (for: 6, мм², 6 쬬);

 to become prosperous; 쬬туму usefulness; propriety, modesty; ¬¬мтуму to become useless, worthless, vain; 쬬туму worthlessness.

 уму¬ worthy of, fit for (in cpds.).
- gay n.m. measure, extent; $\epsilon/\overline{M}/ga$ ngay \overline{N} to the extent of; Na ngay \overline{N} for about (the extent of).
- gay (pl. ghy) n.m. trunk, stump; piece, lump; (או gay gay into many pieces; פּוּףְכּ אַאָּס זְּעָ אַמְע to divide into many pieces.
- удае, удаче, удве, уече, уеве, унче, унве, унче, унве; Q уове vb. intr. to swell up.

ωλ2 n.m. flame, fire; γ-ωλ2 to burn, be alight (± 680λ). waxe vb. tr. to speak, talk, say (MMO'). waxe e to speak to; to speak about, tell of; to speak against. waxe 620YN 6, 62PA1 6 to speak to. WAXE EXN to speak for. on behalf of; enecht exi to speak down (from above). waxe MN to speak with. waxe Nca to speak against, malign. Also with Na2ph before, oybe against, 2a concerning, 2N/N with, in. As n.m. word, saying; thing, matter, affair; story, account, tale. NOG N GAXE boastful words; xe-noe n waxe to boast; peaxe-noe n waxe braggart. 2A2 N WAXE verbosity, garrulousness; MNT2A2 N WAXE idem. P-2A2 N waxe to be garrulous. P-oywaxe N oywr MN to make an agreement with. TAGE-WAXE to talk a lot, multiply words. $x_1-q_{\lambda}x_6$ to accept the word (of: $\overline{N}T\overline{N}$). $6\overline{N}$ φλχε to complain (against: ε20γN ε). φx̄- in various cpds. (see 2nd element). ATQAX6 speechless; unspeakable, ineffable (± mmo', epo'). permaxe eloquent person. ca N waxe babbler. 61 Nouxe speech, saying, tale.

gвс, gче, gese n. off-scouring, filth.

- увир (f. увеере; pl. увеер, увеере) n.m.f. friend, comrade, companion. увр- freq. in cpds.: companion in (often = Gk. prefix συν-). митувир n.f. friendship, community. Горвир (Q о N) to be friend, partner (to, with: ми, е). уви, учи n.f. tale, fable; as adj. fabled, fabulous. усх- уви, ух-уви telling of tales; речхо-уви teller of tales. 98шт, увот (pl. увате) n.m. rod, staff; чал-увит staffbearer.
- g (gμ, g ει) vb. intr. to go (± ethical dat.). This verb is mainly Boh.; its use with adv. and prep. is completely parallel to that of εωκ.
- 9°, 9H, 91 n.m.(f.) wood, beam of wood; many special meanings: cross, gallows, stocks, pillory, shaft, stave. For various woods (96 \overline{N}) see 2nd element.
- ge number: hundred. gut two hundred. Mezge hundredth. ge ge, e nege ge by hundreds. See 30.7.

we, we prep. by (in swearing an oath).

февіну, феввіну, фвеідеїт, фвеідеїт (all Q) to be changed, different (from: e; in regard to: 2N). φвеίω, φείω, φείω, φείω, φείω, φείω, φείω, φείω, φείω, φείω, από n.f. change, exchange, requital; γ-φείω to replace, be instead (of: ммо); χί-(τ)φείω to take requital, be repaid.

φεει (= φε + ει) to go and come, be carried to and fro, wander. As n. derangement, madness.

фексет n.f. bride, daughter-in-law; marriage. ¬р-дексет (Q о N) to become a bride; to make a marriage (for: є; with: мN; to: мх). мх N дексет bridal-chamber, marriage. пх-тдексет bridegroom (may take def. art.). денче, диче, диче, диче n.f. fish-scale. диве, диве n.m. rust, verdigris. ¬р-диве to become rusted. дил, дист, дал n.m. pit, cistern.

gнм n. sign, omen; only in cpds.: x1-gнм to divine, read omens; peqx1-gнм diviner, augur; митречx1-gнм divination. F-gнм to divine.

ghm adj. small, few, young, humble (bef. n. with \overline{n} ; aft. n. without \overline{n}). \overline{n} оуфнм adv. a little. \overline{n} тегге фнм an not only, not merely. коуг фнм little child. \overline{p} нф фнм little by little, (by, into) small amounts. \overline{p} -фнм фнм (Q o \overline{n}) to make small. As n. small person, thing, quantity. \overline{p} нм n.m. tree. \overline{e} 12- \overline{q} 4нм grove. ма \overline{n} фнм idem.

ghpe (gp-; f. geepe, ghhpe; pl. gphy, gpey) n.m.f. son, daughter, child; young of animals. ghpe ghm small child. baby; a youth. geepe ghm f. idem. mntghpe ghm childhood, infancy. atghpe childless; mntatghpe childlessness. mntghpe status of son. p-ghpe (Q o N) to become a child. gp-, gN-, gen- son of, daughter of, in various cpds.; see 2nd element: -eiwt, -haly, -con, -cwne, -oywt, -200yt. gnoya(1), genoya, goyoya n.m.f. nephew, niece. ghtc, gentc, gNc n.m. name of a plant.

gнγε (pl. ? gooγε) n.f. altar.

gı (gei) gı- gıт Q gнү vb. tr. to measure, weigh (ммо');

- with 2nd obj.: to measure out to the amount of; g1 ε2γλι to weigh (6, MN: to a given amount); as n.m. measure, weight, extent, length; moderation. λτg1 immeasurable; MNTλΤg1 immeasurability. κλ-g1 to set a measure or limit (to: Nλ'). P-g1 (Q o N) to make or equal a given weight. †-g1 to set a measure to, restrict (6); λτ†-g1 unmeasured, unrestricted; MNTλΤ†-g1 limitlessness. χ1-g1 to take measure, estimate.
- קוגן, קאו, קאופ, קוגנו; Q дну vb. intr. to be long; as n.m. length. дін, діне, діе n.f.(m.) length; rarely vb. to become, grow long. אב-קוו adj. tall.
- gibe ge- (gebt-, get-, get-) get- (gebt-, get-) Q gobe (goobe, gote) vb. tr. to change, alter (ммо-); vb. intr. and reflex. to change, be altered (to: e; into: м, 2м; in form: м смот); as n.m. change, difference. ммтмы- gibe loving change. это ве unchanging, unaltered; ммт- это ве changelessness. gabwo fearful, strange (lit., changing of aspect). get- n. change. Cf. Q gebihy.
- gike gekT- gakT' (gikT') Q goke vb. intr. to dig (in, into: e, zN; for, after: Nca; down into: enecht e); as n.m. depth. gik, gek, gek n.m. depth, what is dug.
- gine gen(T)- (gn-) gnr' vb. tr. to seek, ask, ask for, inquire after, about (Mmo', Nch); gine e to visit, inquire after, greet, bid farewell; gine Mmo' e to ask someone for something; gine Mmo' Nch/etbe to ask someone about. Also with Ntn from; 2λ for; 2N in, among, into; 21tN through. As n.m. inquiry, request; news, report. gm-noyae good news. Mh N gine place of inquiry, oracle. Peagine inquirer, wizard; Mntpeagine wizardry. Gine Mn(')gine, 6M-n(')gine to search out, visit; as n.m. visitation. 4λ1-gine news-bearer, messenger.
- gine vb. intr. to be ashamed (about: ethe, ex \overline{N} , \overline{N} , 2 λ , 21); rarely tr. to put to shame. gine 2HT' \overline{N} to stand in shame before, to revere. As n.m. shame. λ Tgine

unashamed; F-1701ng to be unashamed; MNTpeqgine shyness. modesty. +-mine to put to shame (NA'); peut-mine one who puts to shame. $x_1-\omega_1$ no to be ashamed (of: erse, $2\overline{N}$; exn; before: мидери). опинт (f. описсте) modest person. wipe, whre (f. weepe) adj. small. 2(e) porpe young servant. youth (opp. of 2xxo); MNT2Fg:P6 state of youth. WITE (WWT) WET- (WAAT-) WAT' (WAAT', WIT') Vb. tr. to demand, extort (MMO', 6; from: MMO'; for, on account of: 2A). ONTE MMO' N OYOCE to exact a fine from. OET-MNTHA to beg for charity. фкак n.m. cry, shout; xi-gкак евох to cry out (to: e, exñ, ϵ_{2} par ϵ_{0}). Agrak to cry out (= Ag-graf, cf. ω_{0}), \pm GBOA (6, 62PA1 6). X!-WKAK, AWKAK n.m. Cry. gкıх, gкнх n. curl of hair. ΦΚλκελ, Φκηλκελ n.m. gnashing, grinding of teeth. φκλιλ, φκελιλ, φκλελ, φκιλειλ, φελειλ, φειλειλ n.m. bell. φκολ n.m. hole. o \overline{N} φκολ φκολ to be full of holes. ψλλ2, φολ2 vb. intr. to be afraid. φχ24, φχ4 n. fear, in cpd. $N62-9\overline{\lambda}2\overline{4}$, $N\lambda2-9\overline{\lambda}2\overline{4}$ to emit fear, be terrified; also as n.m. terror; MNTN62-QNZ24 terror. фан to creep (into: $2\overline{N}$). gan- in gan-oych-noyee to enjoy the odor of incense. φληλ (φλ \overline{x}) vb. intr. to pray (to: e, Nλ°, φλ; for: e, etbg, exn, 21, 21xn); as n.m. prayer. gal6, gae6, gzal6 n.m. spike, sharp instrument; ray, flame. gaon n.m. ply, strand (of cord). gaog n.m. shame, disgrace; as adj. shameful, disgraceful. gx2, gex2, gxe2 n.m. twig, shoot; stave, wand. gX60M, CEA6AM, gXTAM, gEATAM, gEATEM, gAATEM n.f.m. mustard. gma, Q goome (gome) vb. intr. to be light, fine, subtle; as n.m. fineness, subtlety. goods adj. light, fine. gmmo, gmo, gmw (f. gmmw; pl. gmmo) n.m. stranger; as adj. strange. 6 nommo abroad (motion); 21 nommo abroad (static); 2 m помто idem. мптомпо strangeness, foreignness.

mmmo to be hospitable. mnτmacr-ommo hatred of strangers. $\overline{\mathbf{F}}$ - $\overline{\mathbf{o}}\overline{\mathbf{M}}\mathbf{M}\mathbf{O}$ (0 o $\overline{\mathbf{N}}$) to become a stranger, be estranged (from: e. NA"). amoy, amoye, amoyi n.f. peg, stake. фиоун (f. фиоуне) number: eight. митфини(e) eighteen. мег-

omovn eighth. See 15.3; 30.7.

wmoe omoe- omout' vb. tr. to serve (mmo' or אני); as n.m. service, worship, liturgy; MA N whoe place of worship. P-ωπωε to do service (to, for: μλ'). ρεσωπωε server. worshipper: พที่รุยงุญพี่พูธ service. ผูพี่ผู้เราก.m. servant. фмфнее vb. intr. to whisper; as n. whispering. еф-фмфнее, чи-фмфнее to whisper.

φнь n.m. waste-land. F-φнь to become waste, dry.

www n.m. profligate, prodigal (person); мптома profligacy.

φηε, φημ (pl. φημγ, -ε) n.m. net. φηε \vec{n} 210γε casting-net.

фин n.f. garden: пл-тефин gardener.

φιοφ, Q φονφ vb. intr. to stink; as n.m. stench.

ωνς n.m. linen.

φπτω n.f. sheet, robe (of linen).

φπαωτε, φινωωτε n.f. cushion or sim.

go number: thousand. See 30.7.

wo particle: yea!

goein n.m. row, course; N goein goein in rows. gine, goeine n.f. courses (of stones).

φοσιτ in \overline{p} -goσιτ (Q o \overline{N} φοσιτ) to become inspired, possessed, frenzied.

φοειφ n.m. dust; P-goeig to become dusty; x1-goeig idem. (φοκφκ) φεκφωκ* vb. tr. to dig, hollow out, gouge out.

φολ, φλλ n.m. bundle.

ΦΟλ, gwx n.m. molar tooth, tusk.

ΨΟλΜες n.f. gnat.

ψολφλ φ(ε)λφωλ' Q φελφωλ (φρφωρ) vb. tr. to sift, shake in sieve (ммо°); as n.m. shaking.

ФОМ, goom (f. gome; pl. gмoyı) n.m. father-in-law (motherin-law); son (daughter)-in-law. F-gom to become

father-in-law (to: e).

gomnt, gmnt, gemnt, gomt, gmn(т)- number: three (f. gomte, gomnte). gm(n)т-(6) nooy three days ago, heretofore.
мегуомпт third. мптуомте thirteen; мегмптуомте thirteenth. м пуомпт all three (of them). See 15.3; 30.7.
gonte n.f. thorn-tree (acacia nilotica); thorns; thicket of acacias; ge n gonte acacia wood.

φοογ, φοογε, φογ- n.m. incense, perfume.

gooye, Q goγωογ(6) vb. intr. to become dry, dry up, become dessicated, stale. nnet goγωογ the dry land.

gon, goon, gwn n.m. palm, four-fingerbreadth; a set of four. (gongπ) g(G)ngπ-gπgwn' vb. tr. to take in arms, nurse. gopπ, goopπ n.m. awning, veil.

φορφρ φρφρ- φρφωρ Q φρφωρ (φρφορπ) vb. tr. to upset, overturn (ммо°, ε; on, onto: εхм); to destroy; + внеснт idem.

As n.m. overthrow, destruction; pg4gopgp destroyer.

мот, мот (pl. муюте; cf. муюте) n.m. pillow, cushion.
мотут (мохт) метуют Q метуют vb. tr. to cut, carve, hollow out (ммо); to make a hole in (ε, 2N). As n.m. (also муоут) anything carved or hollowed out.

goy, coy in Ngoy prep. without (not standard Sah.).

фоуннв, фоунв, фвів, фоусве, фоусче n.m. persea tree.

goro gore (gor) gorw (goro) (t ebox) vb. tr. to discharge, pour out, empty (ΜΜΟ; from: ΜΜΟ, εβολ ΜΜΟ, εβολ ΣΝ; into: 2Ν); vb. intr. to flow, pour out. goro exN to pour out upon; to crowd against, throng. Also with enecht (e), εγρλι (ε, εχΝ, 2Ν). goreit Q to be empty; net goreit emptiness, vanity (may take art.); μπτηετ goreit emptiness; 2Ν ογμητηετ goreit without cause, vainly; γ-ηετ goreit to act in vain.

фоурн n.f. censer, brazier, altar.

goycooyge, goycwoyge, goycooyze n.m. sacrifice, offering. goywse, φογοse n.f. throat.

φογφογ vb. intr. to boast, brag; reflex. (w. \overline{M} MO') to take pride (in, on, about: e, eTBG, ex \overline{N} , e2PAI ex \overline{N} , 21, 2 \overline{N}).

As n.m. boasting, pride; as adj. proud. MNTWOYWOY, MATGOYGO pride, impudence, +-goygoy to glorify. φογωτ, φωφτ n.m. window; niche, alcove. wow, woow n.m. kind of antelope (bubalis buselaphus). φοφογ, φλφογ, φοφο n.m. pot, jar. wow π n.m. hindrance, impediment; key. +-gog το lock (a door). P-wowT to shut, lock. क्राक्ष, Q क्ष्मुक्ष meaning uncertain: to burrow (?). woxne, waxne vb. tr. to consider (mmo'); to take counsel concerning (e, $\epsilon \times \overline{n}$; with: $m\overline{n}$). As n.m. counsel, design, plan, advice. Ατφοχης ill-considered; μπτατφοχης being without counsel, at a loss, reckless. eige N oygoxne F-goxne to take counsel, make a decision. x1-goxne idem (with: $M\overline{N}$; concerning: 6, 620YN 6); XI-WOXN6 NA to counsel, advise; peaxi-woxne counsellor, advisor. финре n.f. wonder, amazement; miracle; as adj. wonderful, marvelous; 2x onnes wonderful; 2N oronnes wondrously. \overline{P} -gnHpe (Q o \overline{N}) to become amazed, to marvel (at: \overline{M} Ho*, 6, 6TBG, 6XN, G2PA1 6XN, NCA, 2N). φρω n.f. menstruation; $\overline{\mathbf{p}}$ -ωρω (O o $\overline{\mathbf{N}}$) to be menstruous. gcne, cone in 2N ογωcne suddenly, all of a sudden. оте, отн (pl. отну) n.m. mast of a ship. gre n. nest. фтеко, ефтеко (pl. фтекфоу) n.m. prison. тни, от n.f. garment, tunic. 9+ n.m. weaver; warp (on loom). 9τορτρ φτρτρ- φτρτωρ Q φτρτωρ vb. tr. to disturb, agitate, bother, upset, trouble, urge, hasten (mmo'); vb. intr. to become disturbed etc.; as n.m. disturbance, trouble, anxiety, haste; ¿ν ογωτορτρ hastily, quickly, anxiously. λιωτορτρ undisturbed, untroubled; ΜΝτλιφτορτρ tranquillity; F-/+-gropтF to create disturbance. 9тоүнт in †-штоүнт to accuse, bring accusation against (e,

9ω n.m. sand, gravel; κΗ ε πφω being in sandy condition.

N, ελ); ρεч†-ωτογητ accuser.

- gwв (gwq) gg- (geq-) gos (goq') Q gнв (gнq) vb. tr. to shave, clip, tonsure (ммо'); gg-xw to shave the head; as n.m. shaving, clipping, tonsure.
- gob \overline{z} gob \overline{z} Q gob \overline{z} vb. tr. to scorch, wither ($\overline{M}MO^*$); vb. intr. to become scorched, withered.
- gwin.m. what is above, high; always w. art. and usu. in prep. phrases: ε πφωι upward; Μ πφωι ε above; εκολ Μ πφωι from above; cλ-πφωι upper part or direction; cλ πφωι N (prep.) above; εκολ cλ-πφωι from above.
- gwк geк- Q gнк vb. tr. to dig, dig deep; Q = to be deep; as n.m. depth(s). Cf. gike.
- gωκ̄z ωσκ̄z- ωσκ̄z Q ωσκ̄z vb. tr. to dig, dig deep (± επεcht). ωικ̄z n.m. depth. Cf. preceding.
- gωλ gωλ gωλ (gx̄-) goλ vb. tr. to despoil (Mmo'); + esoλ: to spoil, destroy; intr. to be destroyed; as n.m. spoil, booty. λταωλ εκολ indestructible. φολ n.f. spoils.
- gwx gox' Q ghx vb. tr. to loosen, dissolve, paralyze; vb. intr. to flow (into: 6, $2\overline{N}$, 680x $6x\overline{N}$); ghx 680x to be paralyzed, crippled, worthless.
- gωλκ gλκ- goλκ' vb. tr. to stitch, weave (μmo').
- gwam vb. tr. to smell (mmo*); sniff at (ε); as n.m. sense of smell; ma n gwam organ of smell; εικανωπ sense of s.
- gωλ̄ gολ̄ Q gολ̄ vb. tr. to mark, trace line of, make as a mark (Μπο΄); as n.m. mark, marker, stake; †-gωλ̄ to set a mark or boundary.
- gound gard- gark' Q gond vb. tr. to cut (Μπο'); Q to be sharp, sharpened, cutting. Cf. gale.
- gom n.m. tax, tribute; \uparrow -gom to pay tribute; x_1 -gom to receive tribute.
- gow n.m. summer. 26 N gow, 2Ngom n.m. spring.
- goom goom (goom) vb. tr. to wash (clothes: $\overline{M}MO^{\circ}$).
- gwmx gmx- Q gomx vb. tr. to pierce.
- gone vb. intr. to become sick, weak, ill (in, with: ϵ , \overline{N} , $2\overline{N}$); as n.m. sickness, disease. MA \overline{N} netgone infirmary. \overline{P} -gone to become sick; $P \in \mathbb{N}$ gone sick person. For cpds.

- in www-, wan- see 2nd element.
- gon $\overline{\tau}$ (gon $\overline{\tau}$) Q gon $\overline{\tau}$ vb. intr. to quarrel (with: $m\overline{n}$, oyse); as n.m. quarreling.
- gwnT gNT- (genT-) gonT' Q gonT vb. tr. to plait (Mmo').

 φοητε n.f. plaited work.
- gwnq (gwng) genq- (geng-) gonq (gong) Vb. intr. to come together, join; vb. tr. to join, connect (ммо"; to, with: 6, мм, мх"); to convey (to: gx); as n.m. union, unity. 2м оүрwnq jointly, in unison.
- gwn only in genngwn, gnngwn, gnengwn moment, instant; 2π oygenngwn suddenly.
- gwn gen- (gn-, gan-) gon" (gan") Q ghn vb. tr. to receive, accept, take, bear, suffer (ммо"; for, on behalf of: exn, na"; from: ntn, 21tn); to buy (for a price: 2a; with: 2n). Freq. w. ethical dative. Q also = to be acceptable. As n.m. acceptance, purchase. argwn, argon" which cannot be limited or contained. gone n.f. reception, entertainment.
- gone (goone), Q goon vb. intr. to become, come into existence; to happen, take place, occur; to last, endure; Q to be, exist. Acgone impers. it happened that (foll. by coord. vb.). For gone as aux. vb., see §30.9. gone 6, 620γN 6 to be for, intended for, destined for; gone \(\overline{M}\)Mo^{*} (1) to be in; (2) to happen to (a person); (3) to be + pred. noun. gone NA* 6 to act as (6) for (NA*). gone gA to last until; gone 2A to receive, get, have; gone 21 to be/live in the time of. gone 2ATN to be in the care, the charge of. gone 21TN, 680A 21TN to come into existence through, by means of. As n.m. existence, being.

 MA N gone dwelling place, residence. 6gone if, when, since, because.

gone, gove, goode, goode, egone n.m. cucumber.
gone n.m. arm, foreleg; shoulder; name of constellation.
gop gop (goop) Q ghp vb. tr. to stop up, to pile up.
gop π gpn- (gep π-) gop π Q gop π vb. intr. to be early,

first (in, at, to: 6); reflex. idem. gpn- + Inf. to do something first, to have done something previously, already. gwpn n. morning; gwpn m neapacte tomorrow morning, the morning of the next day; nnay n gwpn the morning. gopn (f. gopne) adj. first, earliest; used before of after n., with n; gpn-n- idem. n gwpn, n gopn adv. early. n gopn adv. formerly, at first; n gopn n prep. before. xin (n) gopn, xin e gopn from the beginning. p-gopn (Q o n) to be first, before; + 6 + Inf. to do first, beforehand; to be the first to do.

gwp $\overline{\tau}$ vb. intr. to be demented; tr. to derange (\overline{M} Mo°).

gwc, gwwc, gooc (pl. gooc, gwwc) n.m. shepherd, herdsman; μπτωως shepherding.

gwr, egwr (pl. egore, egare) n.m. trader, merchant; ма π egwr emporium; мπτεgwr trade, commerce; ¬-egwr to trade, deal, traffic (in: 2π); вімер-едwr trade, profit.

gwтв geтв- (gтв-) vb. tr. to muzzle. gтов, gточ n.m. a muzzle, halter.

фоте, фооте n.f. well, cistern.

φωτε n.m. flour, dough.

фютм фетм (фтм) Q фотм vb. tr. to close, seal (ммо ; а-gainst: өрм); vb. intr. to be shut, sealed. фтам (фтом) Q фтам vb. tr. idem. фтом n.m. gate, what is shut. фтм оүшм n.pl. joints.

goome, gome n.f. cliff, precipice.

goone vb. tr. to exclude, deprive (of: ϵ , ϵ); to remove (\overline{M} mo"; from: ϵ).

gowt (got) get- (geet-) galt (galt) Q galt (galt, ght) (1)

vb. tr. to cut (mmo'); to slaughter, slay (with: 2π).

gowt gbol to cut off, cut short; to excommunicate; to

decide; as n.m. excommunication, cutting off. gowt gbol

exπ to condemn. 2π ογφωστ gbol sharply, briefly. galt
in cpds.: who, which cuts (see 2nd elem.). gowt as n.m.

what is cut; sacrifice; decision, verdict. atgoot

uncut. pg4gowt (gbol) cutter, sacrificer.

- (2) vb. intr. to lack (for: G, MMO', 2N); to want, be lacking; as n.m. lack, need, shortage; λτgωωτ without needs. yλλτ N, yλτ G prep. short of, lacking; excepting, apart from. yλλτ G, yλλτ G n.f. part cut off, portion. yλλτ G, yλλτ G n.f. cut, ditch. yτλ vb. intr. to become faulty, deficient; to have defects; as n.m. defect, fault, deficiency.
- φωωσε φεσες (φεσες) φοσες (φοσες) Q φοσσε (φοσες) vb. tr.

 to strike, smite, wound (ΜΜΟς); vb. intr. to be wounded

 (in: ε); as n.m. blow, wound. φελ n.m. blow, wound; γφελ to wound; φελ-ειχ to clap the hands.
- gwm geg- gog Q gнg (± евох) vb. tr. to scatter, spread (мио"; esp. of odor, by wind); vb. intr. idem.
- фиф феф- Q фнф vb. tr. to twist (rope etc.); as n.m. twisting; torture (?).
- φωφ φεφ- φοφ* (φλφ*) Q φηφ (± εδολ, 62γλι) vb. tr. to make equal (Μ̄μο*; to: ε, μῆ); to make level, straight; to lay out straight; Q to be equal (to: ε, μῆ, ογδε). As n.m. equality, sameness, equal status.
- gwu geu- gou' Q ghu vb. tr. to devastate, lay waste, destroy (ммо'); vb. intr. to become desert, laid waste, destroyed; as n.m. devastation, destruction. gwwue, gwwee n.m. barrenness, poverty.
- φωчτ, φοчτ n.m. hollow of hand; handful.
- φωνά φεντ- Q φονά vb. intr. to err, make a mistake (in: ΜΜΟ΄, ε, εΝ); as n.m. error, fault. λτοφωνά unerring. φλντε adj. wicked, iniquitous; μντοματε iniquity; γ-μντ- φλντε to sin (against: ε). φοντε, φοντε, φλντε n. error. φωνε φενε- φονε vb. tr. intr. to wither, scorch.
- φωχε vb. intr. to contend, wrestle, struggle (with: μπ, ογεε); as n.m. contest. μλ π φωχε arena; ρεσφωχε contender. φοειχ n.m.f. athlete, gladiator, contender; μπτφοειχ athleticism; γ-φοειχ to become an athlete, contender. φοειχ n.m. athlete, contender.
- φωχπ φεχπ- φοχπ' Q φοχπ vb. tr. to leave as a remainder,

goop tenant.

to leave behind (MMO'); vb. intr. to be left over, remain; ± 6N220Y idem. As n.m. remainder.

gg6, 6gg6, g6, cg6 (neg. M6gg6, Mgg6) impers. vb. it is fitting, suitable, proper (to, that: e, 6Tp6; see 20.2);

net (6) gg6, Net (6) gg6 that which is proper.

g4w, g8w n.f. a measure of length, schoenus, parasang.

g216, g21x n.m. dust; P-g216 (Q o N) to become dust.

gx6 (pl. gxhy) n.m. locust.

gxhn n.m. garlic.

gx1t, cx16 n.m. name of an occupation: dyer (?).

gxwt n.f. cord (?).

g6An in Ag-g6An to cry out.

g6Op, g6Ox, c6Ox n.m. rent, hire; Atg6Op rent-free; pMN-

ων: ωe WACONT: CONT OBOE: CBOE **ΔΕ : ΤΝΑΚ**Φ QAT(-/*): QITE, gy6: gygy6, gyλ ወልልያ6፡ ወልልያ TOOO ற∈−: மேற $\phi_{\lambda\lambda}T(-/^{\epsilon}): \phi_{\lambda}, \phi_{\lambda}T(\epsilon)$ PA : PATAO **ФЕВВІЮ: ФЕВІН**А QAT6: QUUT TWW g686: g86 σλλτε: σωωτ σλτή: σωωτ **9686: 9246** ወልልተ፫: ወወወተ $\phi \lambda T \overline{C}$: $\phi \omega \phi T$ ወ681: ወአዛ6 ወልልዓ፡ ወልልΒ QATT': QA DEBIW: DEBIHY ውጆላ: 600 февине: выне **ፀ**₽ፈው : ϶₽ፈፈው **Φ68Τ': ΦΨ4**Τ ወልይ6: ወልዛ6 დადე*: დადე gest-/*: g186 φλφογ: φοφογ WABOA: BOA የወው : የፋው DEENE: DOONE ወልነ፡ ወነልነ, ወዘነ ወአባየል: ወአየይአ феере: фіре, фире 0716: 0171 **Φλ46(-): Φ186, ΦΦ4 Φ66Τ-: ΦΦΦΤ** MYXNE: MOXNE ወአነዖ6፡ ወአአዖ6 weeee-: wwwee ODAKT": ODIKE gyB−: gyıB€, gywB 0061: 001, 00E **ወ**ልአ(-): ወወአ QBATE: QBQT gjelk: ajtke **ወ**ልአዘ**ሃ**፡ ወልአ10Υ OBRIAGIT, OBBIO: OJEKT-: OJIKE ወልአለ " : ወወአ 6 Q 6B IHY фектам, фектем: $g\lambda\lambda TEM$, $g\lambda\lambda THM$: $g\overline{\lambda}60M$ **986: Cλ94 Φ**Σ60Μ MAM : - AMA OBEEP(E): OBHP **ден−: дире, диме** DAMICE: MICE BEILEIT: DEBIHY OEN: XIN DANT": DA QECIO: QCBIHY ωenB-: ωωnq φλογ: φλγ **ФВІВ:** ФОУННВ Фене-: Фомие **Φλη(-/*): ωωη, 6ωπε** OBIO: OBBIHY WENOYA: WHPE **ΦΑΡ(*): ΦΑΑΡΕ, ΦΑΑΡ** WERNNE: BINE DENT-: DINE, DONT 92 P 7 5 5 7 5 6 ωв₽−: ωвнр OGNT': OA OBOT: OBOT WENTE: WHIE ወን ነው። ወን አን WAPIOY: WAXIOY ger', gerc: gibe φεημώσι: Φωμ OAPO": OA மு முழ் பிரும் ger-: goot, gire

დ∈ч−: ლთч, ლთ₿	<u>Фи−:</u> Фиье, Фіне	goyw": goyo
ወርче: ወአче	<u> </u>	ლიგლიგ(e): ლიიგe
феатю: Февінд	Фивине: вине	фоляние: 5ние
ውርч ⊤−: ውወч₸, ው∣ Βር	фин, фину(е): фис	уоут: доту⊤
φεчτ ″: ሧነ В€	диолу(1): Фйье	መርፈኒ: መመይ՝ መመፈ
966 6-: 90066	уйс: фнтС	уоче: ф 1ве
ώн: ώε	дйт-/″: фінε, фωн∓	уохт: уотут
фнв: фов, фоүннв	ωΝτ": ΦΥ	യുറ6°, യുറ6€: യൂയയ6€
фике: фенче, фаче	พุทีฯธ: พูธทฯธ	фиемфон: фон
Ф НСІ: ФНІ	whxwx: xwx	фитеете, фитит:
финре: фире	መርያ መንፈብር, መነጸር	g) 1 n6
шнк: шıкс, шwк	றுດвτ(″): றும்ч∓	<u> ши</u> фан: фан
Финас: Фенае	BOEING: BOEIN	gp→: gyHp∈
фнрс: ф 1ре	θχωφ: Εχιθού	ወየአ፡ 2 የአ
унг. 9000т, 9 6	goke: gike	шреу, шрну: шнре
	90x2: 90x2, 9xx2	wpopt: worwp
WHY: gl, glal, gay		alaoli. Aolal
онує: фолинв	фонт, фонтс: фонћт фон": фоюне	
արել: արագ, արա B		φολφλ
фиче: Фуче: Феиче	Фифф : (*) виоф	ogc(ε)−: coω
ற≀: மு∈	фойте: Фωи⊥	g cn λc: λc
фіве: фиве, фуче	фонф: Фиой	<u> </u>
OIC: OIH	φοοβ: ψλλβ	<u> </u>
фін, фінє: філі	goore: gire, gwne	<u>шсенр:</u> сенр
ψlK, wylKT″: gylK6	фоом: фом	ΦΤλ: ΦΦΦΤ, ΧΤΟ
ΦΙΚΣ: ΦωκΣ	доом": дом	ወ ፐል M ፡ ወወፐ፵
gime: goein	фооме: Фиу	дтн: дте
утиуюте: опоюте	доои, дооие: дамие	штну: стє, жто
91T": 91, 91TG	φοοπ: φοπ, φωπε	штй: ш тни
છાય: દાજીવ	တ္တဝုု″ိး ဟ္တမာု	ото(*): жто
фіче: фенче, фхче	goopi: gopi	отов: фот <u>в</u>
glare: gible	စ္ခဝဝင်း စွစ္ဆင့်	ωτο α ωωτ Ε
ркер: сенр	စ္ဘဝ၀γє: စ္ဘဝဂγ	отом: оотн
gkha: gkia	φοογε: φηγε	рторе: тюре
Θ ΚΗλΚΕλ: <u>Θ</u> ΚΣΚΣ	goog: gog	שַּדְקִדוּף: שַדִּסְדִדְּ
WKINGIN: WKNIN	9006°, 90066: 90066	grupe: Tupe
ayein: ayyein	gon <u>ς</u> : απι	ውይ ፐ ፡ ወወ4ፐ
9×62: 9×2	30,000 300 π	DONT: XONT
ψλCG: ψωλG	907116. gw/11 900716. gw/11	90016: 6016
ማአነ6: ወወአቼ		~
ψλ Σ : φληλ	φογ: ego	gwcM: cwgM
Ψ <u>Λ</u> ΤλΝ, <u>Φ</u> ΛΤ6Μ: <u>Φ</u> Λ60Μ	90Y-: 900Y, 9XY,	gwwme: gma
φλη: gλλ2	φογο	душть: дот
Managar mana	фолу: Фире	дче: две, сх <u>о</u> ч
9M-NOY46: 9IN6	фоле: Фолнкв	ወ4ო: ወ8ო
MHHH (E): MMOYN	φογε−: φογο	921X: 9216_
фіїної, фіймо: фіймо	долеве: фолнив	ወጀአ16: ወወአ <u>6</u>
9МПТ (-): фомПТ	φογειτ: φογο	ŵ <u>x</u> −: ῶγχε
фио: фио	доубче: доуннв	джну: Фхе
MOM: SMOM	фолнол: фолнив	gaxoc: 620c
AHUA. BHOAI: BHOA	BOLOBE: BOLOBE	® ₹50 C: 650C
aμολι: aow	дочоча: онре	96): 90066
Фио: Фимо	ወ0Υ0Υωφτ: ΟΥωφτ	ф6нр: С6нр
өйшнт", шйшріт: шйшу€	goycooy₂6: goycooyg€	g6λ61λ, g66λ61λ,

დე6იგ: დე6იթ

q

41 (461) 41- (461-) 417 Q 4HY Vb. tr. to take, carry, bear, sustain (ΜΜΟς, 2λ); oft. w. eth. dat. (6, Nλς). Used w. many prep. and adv. in normal senses. 41 MM to agree with. 41 2λ to tolerate, bear, endure. 41 ΜΜΟς 680λ to take away, remove (from: ΜΜΟς, 2Ν). 41 ΜΜΑΥ to carry etc. from there (± 2Ν, 21ΧΝ: from, from on). For 41- and 4λ1- in Vb. and nom. cpds. see 2nd element. peq41 one who bears (may have object); ΜΝΤΡΕ441 state or condition of bearing.

4ΝΤ, ΒΝΤ, 46ΝΤ, ΟΥ6ΝΤ n.m.f. worm. F-4ΝΤ to become wormy.
40, 80, 800, 401 n.f. canal, water conduit.

- чтооу, втооу (чтеу-, чтоу-; f. чтое, что, вто) number: four.
 мытачте fourteen. мегчтооу (f. -чтое, -что) fourth.
 See §§15.3; 24.3; 30.7.
- 4ω, εω, ογω, 4ωε n.m. hair. ρλτ-4ω, ρητ π 4ω hairy. ογε 2-4ω to let hair grow.
- чюте, вюте n.f.m. sweat. +-чюте to sweat.
- чωтє (вωтє, вотє) чєт- чот vb. tr. to wipe away, off; to obliterate, destroy (ммо). чωтє євоλ (1) idem; (2) intr. to be wiped out, destroyed. λτчωτє євоλ uneffaceable, ineradicable.
- чωбє (вωбє) чєб- чоб° Q чнб (внб) vb. intr. and reflex. to leap, spring (бвох, блеснт, бон, бгоүм, бгрхі); as n.m. impetuosity; реччωбє impetuous person. чобс, вобс п. leaping, dancing; esp. in x1-чобс to dance; ммтреч-
- पळ ६६ (६ळ ६६) प६६- प०६ Q पम६ vb. tr. to seize, snatch, rob (६, लाभ०); рहपपळ ६६ violent person. प०६च n. robber. पळ ६८ (६ळ ६८) rare variant of पळ ६६ to leap q.v.

 4λ1-: 41
 46NT: 4NT
 4Hγ: 41

 4λ1: 8λ1
 46T-: 4ωT6
 4H6: 4ω66

 461, 461-: 41
 466-: 4ω66
 41T*: 41

 407*: 4076

 406*: 4066

 4067: 4066

4T6Y-: 4T00Y 4T0, 4T06: 4T00Y 4T0Y-: 4T00Y 401: 40 4076: 8076 4066: 8076

2

2A, 20 n.m. winnowing fan.

EA, 20 n.m. pole, mast; weaver's beam.

- 2λ (2λρο") prep. (1) under, beneath; often with meaning of bearing, carrying; (2) from under, from the presence of, from the time of; (3) from, by reason of, because of; (4) for, in respect to, on behalf of; (5) in exchange for, for; to, toward (usu. of persons).
- 2λ6, 2λ61H, 2λ1H (f. 2λH, 2λ6; pl. 2λ6γ, 2λ66γ, 2λ6ογ, 2λ6ογ, 2λ6γε) adj. last, final; as n.: end, termination, last part. ε n2λ6, Μ n2λε, Ν ελε, Ν 2λε, εχΝ ελε at last, finally. ωλ 2λε, ωλ ελε until the last, at the last. F-2λε (1) to become last; to be (too) late (for: ε); (2) to be in want (of: ε). χ1-2λε to lag.
- EXCIRCO, EXIRCO, EDIRGO n.f. shade, shelter, shadow; FEXCIRCO to make shade (for, over: e, exm, eixm). xiEXCIRCO to take shade, be shaded, sheltered.
- CACIT, CAIGIT, CAGIHT n.f. gateway, forecourt, porch.
- 2 x 1 n.m. husband. x1-2x1 to take a husband.
- zλκ, zλλκ n.m. tailor.
- מבא adj. sober, prudent, mild (bef. or aft. n., w. א);
 אוּדינאג sobriety, mildness. דְּ-נאג (Q o אֹ נאג) to become sober, prudent.
- 2λκλά, 2λκηλά, 2λκελά, 2λκκλά n.m. a species of lizard.
- n.m.f. servant, slave; rare except in מּאַנּאַג, צַאּנּפּג n.m.f. idem; אַאַדְיַאַא status of slave or servant; דָּ-צאֹנא to serve, become servant (to: אַגּא).
- can only in p-2ax to deceive (μmo*); as n. deceit; μπτρ-2ax deceit, deception; peqp-2ax deceiver; μπτρεφρ-2ax deceit.
 cank, canhk n.f. ring.
- באגאד (pl. פאגאדה, פאגאדה) n.m. bird, any flying creature.

2 AAMH26, 2AAMH2, 2AAMG2, 26AMG26 n.f. boat. 2 AAOYC, 2AAAOYC n.m. spiderweb.

2 Alwm n.m. cheese.

2AM (pl. 2MHY, 2MGY) n.m. craftsman; cf. 2AMge.

ганнр n.m. embrace; P-ганнр ммо"; +-гамнр в to embrace.

2AMO: interj. would that ...!

- гамфе, гамфі (pl. гамфноуе, гамфнує, гамфооує) n.m. carpenter; $m\overline{n}\tau_2$ амфе carpentry.
- 2 λπ, 20π n.m. judgement, inquest; λτ2λπ without going to court. ειρε Μπ(")2λπ, P-2λπ to give a judgement (for: Nλ"; between: ογτε); to go to court; to avenge, i.e. to settle one's case (against: MN). †-2λπ to give a judgement, pass judgement (on: ε, εχΝ); Mλ Ν †-2λπ court, place of judgement; Pε4†-2λπ judge; P-Pε4†-2λπ to act as judge. χι-2λπ, χι Ν ογ2λπ to go to court (against, with: MN, ογκε, 2λ, 21); as n.m. judgement.

2 Ane n.m. the god Apis.

- 2ληορκ, 2ληορτ n.f. saddle, saddle-cloth.
- 2λπc, 20πc impers. vb. (± ne) it is necessary (for someone: e; to do: e, ετρε). See §20.2.
- 2λρ62 (λρ62, 2λρ426, 6ρ62, 6ρ42) vb. tr. to keep, observe, preserve, be careful about (e); to guard, watch, keep (e; from: e, 680λ ΜΜΟ^{*}, 680λ 2Ν̄); as n.m. watch, guard, caution; ΜΝΤΑΤ2λρ62 heedlessness; μλ Ν̄ 2λρ62 place of watch, guardhouse; ρ642λρ62 guard, watcher, watchman.
- apart; (he him)self, by (him)self; other pers. sim.
- 2AC16, 2AC16, 2AC14 n.m. a drowned person; in cpds.: BWK \overline{N} 2AC16, \overline{y} \overline{N} 2AC16, \overline{y} \overline{y} \overline{N} 2AC16 to drown, be drowned.
- 2AT, 2AT6, 2AAT n.m. silver; silver coin(s), money; as adj. silver, white. MAI-2AT money-loving. MGNF-2AT silversmith; peqmenF-2AT idem. CA N 2AT dealer in silver. P-2AT to work silver; (Q o N 2AT) to become silver; peqp-2AT silversmith. +-2AT to pay.

2ATAIA6, 2ATAAH n. name of an eye-disease.

2AT6, 2AAT6 vb. intr. to flow; tr. to pour $(\overline{M}MO^{\sigma})$ ± 6BOA. As n.m. flow. MA \overline{N} 2AT6 channel, water-course.

2 ATHP, 2 ATHPG n.m.f. hammer.

2AOWP, 2OWP name of 3rd Coptic month.

2 AY6AX n.m. anchor.

гафит, гафіт, гаронт n.m. falcon.

22426626, 2242626, 22826626 n.f. lizard.

pron. many; as adj. (bef. or aft. noun, with \overline{N}) many. \overline{P} -2 λ 2 to become or do much/many (+ \overline{N} + noun); \overline{MNT} 2 λ 2 multitude.

2A66, 2AA66 n.m. snare.

2A6IN n.m. mint.

28% n.m. straits, difficulty, distress; \vec{p} -28% (Q o \vec{N} 28%) to become distressed; \uparrow -28% to distress, disturb (N%).

гвес, гевеє, гнуве, гнв (в) с n.m. plow; yoke of animals.

- 28 ΟΡΒ (ΒΟΡΒ) 28 ΡΒ (Β ΡΒ -) 28 ΡΒ Φ (20 Υ ΕΡΟΥ Φ Ρ") Q

 8 ΡΕ Φ (Β ΡΕ ΟΡ Τ) vb. tr. to throw down, push, cast (ΜΜΟ");
 28 ΟΡΒ ΜΜΟ" GBOA to cast forth (on, onto: ε, επες Η τ ε,
 62 ΡΑΙ ε); intr. to fall to pieces. ΚΑ2-Β ΡΕ Φ P unoccupied land; GI Φ2-Β ΕΡΕ Φ P idem or sim.
- 280YP n.f. left hand; as adj. left. (\overline{N}) ca 280YP, 21 280YP on, to the left.
- 28ω n.f. covering; tent.
- 26 (266, 246) Q 2HY to fall (± 6π6CHT, 62PAI down); used with 6, 6xN, 2N, 2ATN, 21xN in ordinary senses. 26 NCA, 26 NTN to become lost to (someone). 26 6BOA to perish, cease (from: 2N; from on, from with: 21, 21xN); to fall away. 26 6 to find, chance upon, light upon, discover; 26 6PO* NCA to find something in the possession of.
- thus. oyn-ee there is a way, it is possible (to: ϵ , \overline{N} , $\epsilon r_{F}\epsilon$); ϵN ee there is no way (to: ϵ , \overline{N} , $\epsilon r_{F}\epsilon$); ϵN ee there is no way (to: ϵ , \overline{N} , $\epsilon r_{F}\epsilon$). ϵN ee ϵN prep. like, in the manner of. ϵN ee + Rel. as, even as, in the same way that. ϵN $\epsilon r_{F}\epsilon$ (1) in this way, thus;

(2) of this sort. N TEIZE THPC SO much, to such an extent. N T(*)2E like (e.g. me), as (I) do, in (my) way or manner. N T(*)2E T(*)2E as (I) was before. Ag N 2E of what sort? κατα θε like (N); as (+ Rel.). κατα τει2E in this way, likewise. P-θε (Q o N θε) (1) to become like; (2) to make like; P-T(*)2E to resume one's former appearance. +-θε to provide means (to: Nα*; so that: 6, εΤΡΕ). εN-θε to find means (to: Nα*; so that: 6, εΤΡΕ). εN-θε to find means (to: N).
2E, 2- n.m. season, in cpds.: 2E-BWWN, 2-BWWN bad season, famine; P-2E-BWWN to have a bad season. 2E-NOY4E good season, plenty; P-2E-NOY4E to be in plenty.

26λne, 2\overline n.f. navel.

26x21x6 n.f. death-rattle.

26N66T6 n.f. monastery, convent. Many variant spellings: 6. н for 66: -н for -6: 2н for 26-.

- 2H, 62H, 2IH (2HT') n.f. front, forepart, beginning; 2HT', 6 2HT' prep. forward (to), before, into the presence of; used idiomatically with certain verbs. 60H adv. forward, ahead, in advance; † 60H to advance, progress. NCX-0H adv. formerly, henceforth. 2A 0H, 2A T(')2H prep. in front of, before (time or place); also used as conj. (+ 6TP6 or MNAT6-). 21 0H, 21 2H at the front, forward, in front; 21 0H MMO' in front of, before, on the front of; P-210H MMO' to precede.
- 2H (2HT") n.f. belly, womb. $M\overline{N}TM\lambda I 2HT\overline{C}$ gluttony. gaze 680A \overline{N} 2HT to ventriloquize. $\overline{N}2HT'$ see $2\overline{N}$.

2H, 26 n.f. storey (of a house).

2нве, 2ннве, 2ннве n.m.f. grief, mourning; P-2нве to grieve, mourn (for: e, eтве, exn, אגי, צוצה); речР2нве mourner.

2нвС, 2ВС, 26вС n.m. lamp.

гике n.f. corn-measure.

2нме, 2 еме, 2 мме, 2 име n.f. freight, fare (on ship or camel); +-2 нме to pay fare; ат2нме free of charge. 2нме n.m., usu. pl., spices, incense. c+-2нме idem;

+-c+-2+NG to offer (burn) incense. goy-2+NG incense;

τλλ6-goy-2+NG G2Fλ1, †-goy-2+NG G2Fλ1 to offer incense.

2HT (2TH*; pl. 2TGGY) tip, edge, end; 2TH* N is the preferred construction before nouns.

2HT n.m. north. 6 2HT, 6N2HT, AN2HT adv. northward. TAN2HT, 2AN2HT adv. (on) the north side. CA N 2HT idem. 2HT (2TH') n.m. heart, mind. A-negght of spoq he came to his senses. MNT2HT N OYOT unanimity, being of a single mind: F-2HT N OYOT to become unanimous. 2HT CNAY doubt; MMT2HT CNAY doubt, hesitation; F-2HT CNAY to become doubtful, hesitant. 2HT WHM impatience; MRT2HT WHM idem; F-2нт онм to become impatient. атгнт senseless; MNTAT2HT senselessness; F-AT2HT to become senseless. BAX-2HT guileless, simple; MNTBAX-2HT guilelessness. pmn2нт wise, a wise person; мптрmn2нт wisdom, understanding; P-PMN2HT to become wise. wc-N-2HT anguish. κω Ν 2 TH*, κλ-2 TH* to set one's heart or mind (on, to: e, exN, 21), to be confident (in); KA-2TH 680A to relax, become careless. P-2TH to regret, repent (concerning: 6, 6xN, NCa); aTF-2TH' unrepentant; MNTP64Fгтн', ммтр-гтн' repentance. сск-пгнт м to persuade. †-2TH' to observe, notice, pay attention to, heed (e, CXN, 21, 2N); NNTAT+-2TH' heedlessness; pent-2TH' attentive; MNTP64+-2TH' attentiveness. gN-2TH' to pity, have pity (on, for: exm, eapal exm, 2a); mntgn-ath pity, mercy; P-gn-2TH to be merciful. For nouns and vbs. cpd. with N 2HT see 1st element. 2A2TN, 2ATN (2A2TH", יבאדאי) prep. with, near, beside.

²ΝΥ, ²HΟΥ n.m. profit, benefit, usefulness, advantage.

^{Mλ 1-2}HΥ profit-loving. F-2HY to be profitable, useful (to: Nλ*). †-2HY to give profit or benefit (to: Nλ*); to gain profit or benefit (in, by, from: MMO*, e, MN̄, ²N̄). 6N̄-2HY to find profit or benefit (in: e, ²N̄).

²H6G to be disturbed, concerned.

21 217" (217") vb. tr. to beat, thresh, rub (Mmo"; on,

- against: exN, 21). As n.m. threshing. pG421 thresher.
 21 (210°, 2100°) prep. (1) on, in, at; (2) (to enquire)
 concerning; (3) and, or, with (connecting two nouns);
 (4) from on, from in, from at; (5) in the time of, in
 the presence of. 21 NA1 adv. thus. GBOA 21 from on,
 from. enecht 21 down from on, down onto; e20YN 21 in
 toward; e2PA1 21 down from, up from, down on.
 (21B6), Q 20B6 (2AB6) vb. intr. to be low, short. 2BB6
- (2186), Q 2086 (2186) vb. intr. to be low, short. 2886
 n.m. lower part or place. 281 n. shortness.
 21801, 26801, 2180Y1, 280Y1, 2180Y1 n.m. ibis.
 216, 214 (pl. 21HY, 21HOY, 216Y) n.m.f. rudder.
 21618, 621618, 26168, 218 (f. 26116186, 21186, 2161186, 216686, 216186, 2186, 2186, 2186, 218186) n.m.f. lamb.
- 216686, 216186, 2186, 21886, 21886) n.m.f. lamb.
 21617, 261617, 217 n.m. pit.
- 21H (pl. 2100γε, 21Hγ) n.f. road, way. 21H N ε1 εΒΟλ exodus, way out. 21H N ε1 ε20γη way in. 21H N ΒωΚ way of going (in: ε20γη). 21H M MOOGE way, road. 21H N X100γ a way for crossing, ford. 21H M πργο the king's road, highway. γΜ(N) 21H traveling companion. †-Τε21H Νλ΄ to provide way or means to (someone).
- zık n.m. magic; as adj. magical. p-21k to bewitch, enchant (6, 21). peqp-21k wizard, magician; ммтречр-21k magic, wizardry. 22ko n.m. magician; ммт22ko magic.
- 21N, 261N n.m. cup, vessel; a liquid measure; peqH-n21N 620γN diviner (by aid of cup).
- 2:Ne vb. intr. to row; tr. idem (MMO'). 2:N16, 26N616 n.m. steering-oar, rudder.
- 2 ING 2NT reflex. to move forward (not properly Sah.). 2 INHB, 2 INH4 to sleep, doze; as n.m. sleep.
- 2 ιογε 21-21" vb. tr. (1) to beat, strike (ΜΜΟ", ε, εΧΝ, 2Ν, ε2ογη ε; with: ΜΜΟ", 2Ν). (2) to cast, throw (ΜΜΟ"; ± εΒΟλ, ε2ρλι); mostly Boh. in this sense. 21-τοοτ"
 - to begin, undertake (to do: ϵ + inf.); also lit., to place one's hand (on: ϵ).
- 21ρ, 261ρ n.m. street, town quarter, road. 6 π21ρ adv.

outside, to the outside. 21p21pc, 26p21p6, 2Hp21p6 n.f. idem.

troubled (with, by, of: etbe, 2λ, 2N); to experience difficulty or distress (in doing: Circum.); to be difficult or troublesome (to, for: e, Nλ"). (2) vb. tr. to weary, distress, trouble. As n.m. weariness, distress, trouble; labor, product of labor. λτ2ιc6 unwearied; without difficulty; MNTMλ1-2ιc6 love of toil. P-2ιc6 to take trouble; to make trouble. +-2ιc6 to give trouble, make trouble (to, for: Nλ"). ογε2-2ιc6 idem. gn-2ιc6 to labor, take trouble, be deeply concerned (for: e, exn; in, concerning: etbe, 2N); as n.m. labor, product of labor; λτgn-2ιc6 unsympathetic; peq-gn-2ιc6 one who labors etc.; MNTgn-2ιc6 labor, suffering. 41 2λ 2ιc6 to bear up under difficulty.

2106 2ACT' Q 2006 to spin (flax etc.).

- 2 ITG 2 GT- 2 AT' (1) vb. tr. to rub, move back and forth (ΜΜο'); to wear out (ΜΜο'); to convulse, torment (ΜΜο'); to flay. (2) vb. intr. to become old, worn out; to loiter, loaf around; to be convulsed, tormented. As n.m. spasm, pain; AT2 ITG unworn; untormented; ΜΝΤΡΕΘ2 ITG convulsion.
- 210ME, 210ME, 20ME n.f. palm, hollow of hand (N 61x).
- 2κο (ε2κο) Q 2κλειτ (2κοειτ, 2οκ) to become hungry (for: ΜΝΟ); as n.m. hunger, famine. 2μκε adj. poor (bef. or aft. noun, with Ν); μντ2μκε poverty; μλι-2μκε loving the poor; μντμλοτ-2μκε hatred of the poor. γ-2μκε to become poor.
- $2\overline{\lambda}$ KOY, 26 λ KOY, 2 λ XKOY n.f.m. sickle.
- 2χλο (f. 2χλω, 2χλογ; pl. 2χλοι) n.m.f. an old person, elder; esp. an older monk; as adj. old (bef. or aft. noun with N). ΜΝΤ2χλο (of women: ΜΝΤ2χλω) old age. ρ-2χλο (Q ο N 2χλο) to become old.
- 2AOGIAG, 2AOGIA, 2AGGIAG, 2GAGIAG Vb. tr. to bear, carry

 $(\overline{N}MO^{s})$, usu. on surface of water; intr. to be borne, carried; to float.

2 NOMAM n.m. entanglement, snare.

2 λοολε vb. tr. to nurse (a child: ΜΜΟ°); to carry a child during pregnancy or infancy. ρεα2λοολε π. nurse.

2 Aon n.m. a vessel (for pouring).

2λοπλ \overline{n} (2λοπλen) 2 $\overline{\lambda}$ πλen $^{\prime}$ Q 2λeπλen vb. tr. to weary, plague (e, ex \overline{n}); intr. to become weary, despondent; as n.m. weariness, distress.

2λοςτπ, 2λοςτεμ n.m. mist; F-2λοςτπ to become misty, dark; ή-2λοςτπ to darken.

2λογλωογ Q to be high, exalted.

2λο6, Q 20λ6 to become sweet, delightful; 2λλ6- in cpds.: sweet in, sweet of (e.g.-yλx6 speech, -2μτ heart). As n.m. sweetness, delight. μπτ2λο6 idem. †-2λο6 to make sweet, pleasant. 26λ66, 2λ66 n.f. sweetness. 2λμ66 n.f. idem.

2 NOM, \$20M, 2 NOM n.m. louse, flea.

2λω64 vb. intr. to be easy, pleasant.

zme number: forty (see §30.7), $\pi e_2 me \overline{n}$ zooy Lent. $me_2 zme$ fortieth.

2Mene number: eighty (see §30.7).

2Μme in P-2Mme to steer, guide (Mmo*). P-2Mme n.m. guidance. ΔτP-2Mme unguided. PeqP-2Mme pilot, guide.

гМитыре, геметоре, геветыре etc. n.m. sign, token; password.

2мом (gмом) Q 2нм to become hot; as n.m. fever, heat. 2Ммє n.f.m. heat, fever: +-2ммє to give off heat.

2Mooc vb. intr. to sit, sit down, be seated (± 62 pl); to dwell, remain. Used with most prep. in normal senses.

MA N 2Mooc (1) seat; (2) privy, latrine; (3) anus.

61N2MOOC manner of sitting, dwelling. 2MA1C n.m. buttocks.

2мот n.m. grace, gift, favor; gratitude, thanks, credit. ΔT 2моT graceless, thankless. F-2моT, E1P6 N O72моT to grant a favor, give grace, give as a gift. +-2моT to

- give grace, to benefit, be kind to (NA*); † MMO* N
 2MOT to give as a gift or favor. $g\bar{n}-2MOT$ $\bar{N}T\bar{N}$ to thank,
 give thanks to (for: $ex\bar{N}$, 21, 2A); as n.m. thanksgiving; $\lambda Tg\bar{n}-2MOT$ ungrateful; $peqg\bar{n}-2MOT$ a grateful person; $M\bar{N}Tpeqg\bar{n}-2MOT$ gratitude. x1-2MOT to obtain grace
 or favor (from: eROX $2\bar{N}$, $\bar{N}T\bar{N}$; for someone: $ex\bar{N}$, $21X\bar{N}$). $e\bar{N}-2MOT$ to find favor or grace.
- 2MOY n.m. salt. \overline{p} -2MOY to become salt. \uparrow -2MOY to add salt. x_1 -2MOY to be salted. x_1 2MOY unsalted. x_1 2MOY salt-dealer, salt-seller.
- 2MOX, Q 20MX to become sour. 2MX, 26MX, 2HMX n.m. vinegar.

 P-2MX to become sour. + 6 π2MX to start to turn sour.

 2MC, 26MC, 2HMC n.m. ear of grain.
- $2\overline{M}2\overline{M}$ vb. intr. to roar, neigh; as n.m. neighing, roaring. $\overline{p}-2\overline{M}2\overline{M}$ idem.
- 2N (N2HT") prep. (1) of place: in, within, on, at, among; from in, from; (2) of time: at, in, during; (3) of agent, means, instrument: with, by, through; (4) for adv. phrases 2N ογ... see 21.3; (5) for 2N HTP64- see 20.1. eBOA 2N from in, from within, out of; e2OYN 2N into, toward, at, within; N2OYN 2N in, within; 2PA1 2N in.
- מאאץ, אואסץ n.m. vessel, pot, container; thing (any material object), property. אוּדער state of being without property.
- where (62NG-) 2NA' (62NA') impers. vb. it pleases (suff. is objective); nGT 62NG- that which pleases (someone), that which (someone) desires; often followed by 6 + inf.

 P-2NA' to be willing, desire (to do: 6, 6TPG). See 20.2.

 ZÜKG n.m. beer.
- 20, 2λ (2ρλ*) n.m. face (of man or animal); surface, side.
 20 MN 20 face to face. 20 0γ86 20, 20 21 20 idem. N
 20, M π20, 2M π20 by sight. 2λ π(*)20 from before.

 † M π(*)20 to direct one's attention (to: 6, 6xN).

 9π-2γλ* (N) to beseech, ask; to receive, accept. 412γλ* (680λ, 6π9ωι) to look up. x1-20, x1-π20, x1 M π20

 (\overline{N}) to heed, pay attention to, respect, favor; x_1-20 as n.m. favoritism; atx1-20 impartial; MNTATX1-20 impartiality: pegx1-20 one who is partial. x1-2px', x1 N 20 (0 x1-2PA61T), suff. is reflex.: to amuse oneself, occupy oneself; to be distracted; to attend (to: 6); to converse (with: $M\overline{N}$); to reflect (on: 21, $2\overline{N}$); to sport, play (with: MN, 2N); MNTX1-2PA distraction; p64x1-2PA trifler. +-20 6 to beseech (Boh., rare in Sah.). $62P\overline{N}$ ($62P\lambda^{2}$) prep. toward (the face of), among; 680λ 62 PN out to: 62 OYN 62 PN in to, before, at, against. $NA2P\overline{N}$, $\overline{N}NA2P\overline{N}$, $(\overline{N})NA2PA^*$ in the presence of, before. 21 2PA on the surface of, on the face of. 20, 20 n.m. a grain measure. 2061M (p1. 2HM6, 21MH) n.m. wave. \overline{p} -2061M (Q o \overline{N} 2061M) to become agitated. +-2061M, 41-2061M to cast up waves. 2061NG, 201NG indef. pron. pl. some, certain (ones, people, things); as pred.: such, of this sort. 2061P6, 2061A6, 201P6 n.f. dung (human or animal). 2061T6, 201T6 n.f. hyena. 2061TE, 201TE n.m.f. garment: +-2061TE 6xN to clothe. 201 n. in F-201 meaning uncertain, prob.: to make an effort, strive (to do: ϵ , \overline{N} + Inf.); \uparrow -201 NA* to vex. 201 (pl. 2166Y, 2166Y6) n.m. (1) field: (2) water-wheel. zome n.f. cup. 20MNT, 20MGT, 20MT n.m. copper, bronze; coin, money. +-2 OHNT to pay (someone: Na'; for: 2A). gon 2A 2 OHNT to buy with money. $x_1-20M\overline{N}T$ to accept a bribe. Mai-2 OMNT money-loving; MNTMA1-2 OMNT love of money; MNT-MACT-20MNT hatred of money. F-20MNT to become copper; речр-гомит coppersmith; са и гомит copper-dealer. 20NB6 n.f. spring, well. 20NT, 20NT n.m. pagan priest. 200 λ 6 n.f.(m.) moth. \overline{P} -200 λ 6 to become moth-eaten, to

perish. ATF-200x6 incorruptible, indestructible. 2007 n.m. day. M n6200Y in, during the day. N 0Y200Y

for a day. $2\overline{N}$ oy200Y 680% $2\overline{N}$ oy200Y from day to day. x_{1N} 200Y 6 200Y idem. 200Y 200Y, n_{62} 00Y n_{62} 00Y idem. \overline{p} -200Y to spend a day. n_{00} Y adv. today; \overline{N} n_{00} Y idem; n_{00} Y from today onward; n_{00} Y n_{00} Y n

- 200γ Q to be bad, wicked, putrid. πεθοογ, πετ 200γ used as nominal: what is bad; evil, wickedness (may take def. or indef. art.). γ-πεθοογ, ειγε π πεθοογ to do evil; γεαγ-πεθοογ evil-doer; μπτρεαγ-πεθοογ wickedness; ca π πεθοογ evil-doer; μπτα π πεθοογ evil.
- 200YT, 6200YT, 260YT, 26YT (20YT-) n.m. male (of men or animals); freq. as adj., aft. n., with or without N: male, wild, savage. 20YT-C21M6 male-female, bisexual. мNT200YT maleness.

200γτπ n.m. road, highway; a furlong.

200 yo to abuse, curse (ϵ , $\epsilon x \overline{n}$).

20n, 200n (2An-) n.m. marriage feast; bridle-chamber.

2008, 20086 n.f. market; pMN2008 market-man.

20cm, 20chm, 20cme, 2Acm, 20cem n.m. natron.

- 20TE, 20T, 20TE, 20T in \overline{M} $\Pi(f)$ 20TE, \overline{M} Π 20TE \overline{N} in the vicinity of, in the presence of.
- 20Te n.f. fear; as adj. fearful. ΔΤ20TE fearless; ΜΝΤΔΤ20TE fearlessness; P-ΔΤ20TE to become fearless. 2Δ
 20TE in fear; fearful, fearsome (as pred.). P-20TE (Q
 0 N 20TE) to become afraid (of: e, exN, etbe, 2ΔΘΗ Ν,
 680λ 2N, 2ΗΤ΄ Ν); PEΨP-20TE fearing, respectful; ΜΝΤPEΨP-20TE fear, respect. +-20TE to terrify, frighten
 (G, NΔ΄, exN); PEΨP-20TE dreadful. x1-20TE to frighten
 (ΜΜΟ΄).

POTE n.f. hour, moment; F-20T6 to spend time.

20τc, 2λτc n.f. a vessel or measure.

20Τ2〒 26Τ2〒- 26Τ2ωτ Q 26Τ2ωτ vb. tr. to examine, investigate, inquire into (ΜΜΟ', ε, Ναλ, 2Ν); as n.m. inquiry, question; λτ26τ2ωτ unfathomable: peq20Τ2〒 inquirer:

MNTP6420T27 inquiry.

20γειτ (f. 20γειτε, 20γιτε; pl. 20γλτε) adj. bef. or aft. n. with π: first, foremost, leading. 20γειτε n.f. beginning; 2N τε20γειτε in the beginning; xiN τε20γειτε from the beginning.

20γHT (pl. 20γλτ6) n. passenger, crewman (?).

20γN n.m. inner part, interior. ΜπεογΝ ΜΜο prep. inside, within (spatial or temporal). P-π(*)20γN ε to enter. ε20γN adv. to the inside, into, toward: ε20γΝ ε prep. to, toward, into; ε20γΝ is also used to reinforce ε2 ρΝ, εΧΝ, ΝΑ*, ΝΑΣ ρΝ, ΦΑ, 2λ. Ν20γΝ adv. within, inside (static location); Ν20γΝ 2λ under; Ν20γΝ 2Ν in: Ν20γΝ ΜΜο in. cλ-20γΝ adv. inside, within; + ε/ΜΜο idem as prep. cλ Ν 20γΝ n.m. inner part, interior. Φλ 20γΝ ε prep. until. 21 20γΝ adv. within; ετ 21 20γΝ adj. phrase: inner, interior. ρΜΝ20γΝ title of official.

greatness; as adj. bef. n. without \overline{N} or aft. n. with \overline{N} : great, much; before adj.: more, greater. 20Y6- as proclitic form of adj., used like preceding entry. 20Y0 e, 20Y6 more than, beyond. ϵ 20Y0 e, ϵ 20Y6 (e) more than, rather than. ϵ ne20Y0 adv. greatly, very. \overline{N} 20Y0 adv. much, greatly, very, much more so; \overline{N} 20Y0 ϵ more than. \overline{N} 20Y0 \overline{N} 20Y0 idem (emphatic). \overline{P} -20Y0 to exceed, be more than (ϵ); to have or do more (than: ϵ); with immediately following noun or verb: to be or do all the more. \overline{P} -20Y6- proclitic form of preceding.

20γρε- (20γρ-, 20γρω-) 20γρο* (20γρω*) vb. tr. to deprive (someone: suff. obj.) of $(\overline{M}MO^*, \epsilon)$.

20 ypir, 20 pir (pl. 20 ypare) n.m. watchman, guardian. Aug \overline{N} 20 ypir head-watchman.

20Y4 n.m. vetch, pulse.

20726 n.m. untimely birth.

204, 208, 20π, 2ω8 (f. 24ω, 28ω; pl. 28ογι) n.m.f. snake. 20x2x (20x26x, 20xx) 26x2x- (26xx-) 26x2ωx Q 26x2ωx vb. tr. to distress, restrict, straiten ($\overline{M}MO^*$); to compel, force; vb. intr. to become distressed, restricted, narrow; as n.m. distress, need.

2 not, 2 not n. a fathom.

- 2PA (ΦPA) vb. tr. to drive, compel (MMO', NCA), ± 680A.

 2PAI, 2PE n.m. upper part (very rare as n.); 2PAI reinforces other prep., no diff. in meaning. 62PAI adv. upward (see §8.1). 62PAI forms cpds. with many prep. (including 6, 6XN, 62PN, 0Y86, CA, 2A, 2I, 2N), usually, but not necessarily, with the added nuance of "up," e.g. up to, up onto, etc. N2PAI adv. above (static; §28.7). also freq. cpds., as in N2PAI 6XN up on, etc. CA-2PAI adv. above, on the upper side. 9A 2PAI adv. upward; 9A 2PAI 6 up to, even to. 2I 2PAI, 2I 2P6 adv. upward. CA-2P6 n.m. in 6 n(')CA-2P6 prep. above.
- 2 γιι n.m. lower part, rare except in cpds.: 62 γιι adv. downward, down; 62 γιι 6 down to, into, onto; 62 γιι 6x π down onto. π2 γιι adv. below. c1-2 γιι adv. downward, down. 91 2 γιι 6 prep. down to.
- $2\overline{p}B$ n.m. form, likeness; $x_1-2\overline{p}B$ to assume a form, likeness. $2\overline{p}B\omega T$, $2\overline{e}pB\omega T$
- 2PG, 2epe (pl. 2pHye, 2peoye) n.m.f. food (of man or animals); F-2pe (Q o N 2pe) to become food; †-2pe, † N ογ2pe to give food (to: Nλ*). x1-2pe to get food.
 2pes n.m. chisel.
- 2PHPE n.m.(f.) flower. P-2PHPE to bloom, blossom. τεκ2PHPE EBOX idem. ΟΥΧΜ-2PHPE beetle (lit., flower-eater).
 2PHG, 2PHX vb. intr. to become still, calm, quiet.
 2PIM n.m. pelican.
- ² PMAN, 26PMAN n.m. pomegranate (tree or fruit); Βω Ν̄ ² PMAN pomegranate tree.
- calm, quiet; to cease; rarely tr. to still. As n.m. stillness, quiet; †-2pok to calm, quiet (NΔ*). 20pkq adj. silent, quiet.

2 POOY (2 POY-, 2 P-; 2 PA*) n.m. voice; sound, noise, cry.

AT2 POOY voiceless; C2A1 AT2 POOY a consonant. Nex
2 POOY, NOYXE N OY2 POOY (± 6BOλ) to let out a cry. C6K2 POOY to snort. +-2 POOY (± 6BOλ) to speak, give voice,
promise; C2A1 64 +-2 POOY a vowel. 69-2 POOY 6BOλ to make
a sound, utter a cry. 41-2 POOY, 41-2 PA* (± 6BOλ, 62 PA1)
to raise one's voice, to utter, speak. X1 N 2 PA* to
cry out; X1 M n62 POOY to hear the sound (of). MNT
NA9T-2 POOY being hard-voiced. 2 POY-M-n6 n.m. thunder.
2 POY-BA1 n.f. thunder; +-2 POY-BA1 to thunder. 2 POYO,
2 POYW boastful talk; MNT2 POYO boastfulness; F-MNT2 POYO
to boast.

reported vb. tr. to flap or spread (wings); to blink (eyes).
royxε n. pebbles.

2 POO 2 PO- (2 e PO-) Q 2 O PO vb. intr. to become heavy, difficult (for someone: e, exπ, 62 Pλ1 exπ; in, with something: MHO*, 2π); to be slow (to do: e + Inf.); rarely tr.: to make difficult. As n.m. weight, burden. λτ-2 POO weightless; †-2 POO Nλ* to add weight to. 2 POO N 2 HT to become long-suffering, patient; 2 λ PO-2 HT adj. patient, long-suffering; MNT2 λ PO-2 HT patience; P-2 λ PO-2 HT to be patient. 2 PHO e, 2 e PO e n.f. weight.

2 P \propto P \propto V b. tr. to grind or gnash (the teeth; at, against: $ε_2$ O Y N ε, $ε_2$ O Y N ε, $ε_2$ P \propto N \approx N. M. gnashing of teeth.

2 pω n.f. oven, furnace.

ιρωτ n.f. wine-press, vat.

2₱2₱ vb. intr. to snore.

2TAI (2TAGI, 620AI) to become fat. As n.m. fat.

гтн n.f. shaft of spear; mast.

erir n.m. onion.

270, 270, 6270 (f. 270pe, 2700pe; pl. 2700p, 62700p, 270p) n.m.f. horse. MAC \overline{N} 6270 foal. MANG-270 horse-groom. \overline{PMN} 270 horseman.

гтомты гтыты- Q гтытомт to become dark, be darkened; as

n.m. darkness, mist. 2 TOOYS, TOOYS n.m. dawn, morning. HNAY N 2 TOOYS dawn, early morning. 6 2 TOOY6, N 2 TOOY6, 21 2 TOOY6 at dawn. @A 2 TOOY6 until morning. XIN 2 TOOY6 from morning (on). 2 Ton n.m. (1) fall, destruction; (2) name of a measure. TOP n.m. necessity, constraint; 2N oy2 TOP out of necessity. 2A/2M n(*)2TOP of one's own accord, on one's own authority. F-2 TOP to constrain (6); F-n(')2 TOP to exercise authority. †-2 top to constrain (6); to give authority (to: e). 2ω impers, vb. it suffices, is enough (for someone: ε; to, that: e + Inf., erpe, Circum.). Also used with pers. subject: to have enough, be satisfied; to cease, stop (c + Inf., erpe, Circum.); often + e as ethical dative. 2ωΒ 2λΒ vb. tr. to send (Mmo'; for, after: Nch). 2ωΒ, 2ω4, ω4, 204 (pl. 2ΒΗγε) n.m. (1) work, product of work; (2) thing, object; (3) matter, affair, business. oy ne n(*) 208 what is the matter (with ...)? oy ne n208 What is the use of? OY N 208 What? OYN-208 MN (neg. $M\overline{N}-2WB$ $M\overline{N}$ there is (not) a matter; this and the same constructions with the corresponding possessives (0yN-TAI etc.) express the general idea of having a (legal) problem with or involving another person. 2008 N 61x handiwork, handicraft. F-208 to work (at, on: 6; for: 21; in, with: 2N); as n.m. work, working; peqF-20B worker; μπτρεάρ-2ωε work, labor; ωερ-ρ-2ωε fellow-worker. 2ω8 k vb. tr. to prick, incite. 280k, 2806 n. prick, stab. 2ωBC 26BC- (2Ec-) 20BC Q 20BC vb. tr. to cover, shelter. protect, clothe (Mmo*, 6, 6xN, 21xN; with: Mmo*, 2N); 2ωβ G GBON GXN idem; vb. intr. to become covered etc. PG42WBC coverer, protector. 2WBC, 2OBC, 2BC, 2BBC, 2HBC, 2ληC n.m. covering, 1id. 2BOOC, 2BOC (pl. 2BOOC, 28wc) n.m.f. covering, garment; linen. 28co, 268cw (pl.

εωκ εεκ- εοκ' Q εμκ vb. tr. to smite, crush (Μπο', εxπ).

2 Εcooγε) n.f. garment, clothes, cloth.

- 2ωλ, Q 2Hλ vb. intr. to fly. 2ωλ 680λ to fly forth; Q to be distraught. Other adv. and prep. in normal senses.

 MA N 2ωλ exit. p642ωλ flier.
- 200A (200AE, 2000AE) vb. intr. to become hoarse.
- $2\omega\lambda$ 26λ $(2\overline{\lambda}$ -) 20λ ' vb. tr. to throw, cast.
- 2ωλΚ (2ωλδ) 20λΚ Q 20λΚ vb. tr. to twist, braid, roll (ΜΜΟ'); as n.m. plait, twist.
- 2ωλδ, Q 20λδ vb. tr. to embrace (6, 620γN 6); as n.m. embrace.
- 200 $2\overline{N}$ (26N -) 20N Q 2N vb. tr. to tread, trample, beat (\overline{M} MO'; on: 6, 62 \overline{N}) 6, 62 \overline{N} , 62 \overline{N} , 62 \overline{N} , 21); as n.m. treading, trampling.
- 2ων 2ν- 2ον Q 2Ην (± 62ογν) vb. intr. to approach, draw near (to: 6); to be about (to do: 6 + Inf.); Q to be nigh, near; to be related (to), in compliance (with); rarely vb. tr. or reflex. to bring near. Δτ2ων 6γο unapproachable.
- 2ων 2ον' vb. tr. to command, order (someone: ετπ, πτπ; to do: ε, ετρε); to give (an order, command: ммо'; to: ετπ, πτπ). As n.m. command.
- 2ωN vb. intr. to go aground. Μλ \overline{N} 2ωN shallows. o \overline{N} 2ωN (Q) to be shallow.
- 200N n. in x_1-200 N to betroth (\overline{M} Mo"; to: $N\lambda$ "). 200NG n.f. canal.
- 2ωΝΚ (2ωΝF) 26ΝΓ vb. tr. to consecrate, appoint.
- 2ωΝΤ 26ΝΤ- (2ΝΤ-) 20ΝΤ' Q 2ΗΝΤ vb. intr. to approach; rare in Sah.; uses parallel those of 2ωΝ (approach) q.v.
- $2\omega N\overline{x}$ vb. tr. to entreat, exhort (e). Very rare in Sah.
- 2ωογ 2ογ- vb. intr. to rain (down on: exπ, e2γλι exπ; from: eBOλ 2π); also tr. As n.m. rain, moisture; μογ-π-2ωογ idem. 2ογ-π-πε n.m. rain.
- 2ωπ 26π 20π° Q 2μπ vb. tr. to hide, conceal (Μπο°; from: 6); intr. to hide, become hidden (from: 6). As n.m. hiding; 2N ογ2ωπ in hiding, secretly; 2N π2ωπ idem; N 2ωπ idem. λτ2ωπ unhidden. Μλ Ν 2ωπ hiding-place.

- 2ωnφ n.m. palm-branch with hanging dates.
- 2ωρ 2F- 20p* Q 2HP vb. reflex. to guard against, take heed for (6).
- 2ωρ 2P- (26P-) 2OP vb. tr. to milk; 2P-6Pωτ6 idem.
- 2ωP the god Horus.
- $2\omega P\overline{B}$ ($2\omega P\overline{A}$) $2\Omega PB''$ Q $2\Omega P(G)A$ vb. tr. to break ($\overline{M}MO''$); intr. to be broken.
- 2ωρκ, Q 20ρκ vb. intr. to sit quietly (as in ambush).
- zωρπ, Q zopπ vb. intr. to sleep, doze.
- 2ωρπ 2pn- (2ερπ-) 2ορπ' Q 2ορπ vb. tr. to soak, drench, wet (ΜΜΟ*; with: 2N, ΜΜΟ*); also intr.: to become wet.
- 2ωρδ (2ωλδ, 2ωρΣ) 2ερδ- 2ορχ Q 2ορδ (2ορΣ) vb. tr. to heap up, pile up (ΜΜο΄); to put into order, arrange; vb. intr. to be heaped up, put into order; as n.m. order, harmony. 2ροχ vb. idem (rare).
- 2ωc, 2ωωc, 2ογc n.m. thread, cord.
- 2ωc 2ec- 2oc* vb. tr. to block up, cover up, stop up (ΜΜο*; $ex\overline{N}$, $21x\overline{N}$); vb. intr. to be blocked up etc.
- 2ωc vb. intr. to sing, make music; as n.m. song; peq2ωc singer.
- 2ωτ n.m. sack, bag.
- 2ωτ in \overline{p} -2ωτ to sail, float (to: ϵ , $\varphi \lambda$; in, on: $2\overline{N}$); $M \lambda \overline{N}$ \overline{p} -2ωτ sailing course.
- 2 ωτΕ (2ωτεβ) 2 ετΕ- 20ΤΕ Q 20ΤΕ vb. tr. to kill (ΜΜΟ');
 2ωτΕ Να to massacre. 2λΤΕ in cpd.: slaying, as in
 2λΤΕ-9μρε child-slaying. As n.m. slaughter, murder;
 Corpse; ρεμ2ωτΕ slayer, murderer; μΝτρεμ2ωτΕ murder,
 slaughter; ρ-ρεμ2ωτΕ to slay (ε). 2λΤΕ α n.f. slaying;
 thing slain.
- εωτε vb. to bruise, pierce.
- ²ωτε, ₂ωτ n.f. rod, pole; ge N ₂ωτ wooden pole.
- 2ωτħ (2ωπτ) 26τħ~ 20τπ (20πτ) Q 20τπ vb. intr. to set,
 sink (of celestial bodies); to become reconciled (to,
 with: e, μΝ); vb. tr. to reconcile (ΜΜο); to, with: e,
 μΝ); as n.m. reconciliation; sunset. μλ Ν 2ωτπ the

- west. F-2ωτπ to reconcile.
- 2ωτρ (2ωτερ) 2ετρ 2οτρ Q 2οτρ vb. tr. to join (ΜΜος; to:
 ε; with: ΜΝ); to hire; vb. intr. to be joined (to: ε);
 to be hired (for: ε); to be in harmony (with: ΜΝ). As
 n.m. joining, yoke, harmony. ρεν2ωτρ hireling. 2λτρ
 (pl. 2λτρεεγ, 2λτρεγε) n.m. twin, double; as adj. doubled. 2λτρες n.f. yoke (pair) of animals.
- 2000, 200 emphatic or intensive pronoun, used appositionally with other pronominal elements: (I) myself, (I) too, for my part, on the contrary, on the other hand.
 2004 adv. (no pron. agreement) on the other hand, however (expressing contrast or opposition).
- 2ωωκ (2ωκ) 2 GK 2 OK (20 OK) Q 2 HK vb. tr. to gird, arm (ΜΜΟ ; with: 2Ν, ΜΜΟ ; for, against: e, ογβ e), ± GBOλ, G2 OγΝ. 2ωκ ΜΜΟ Η ΜΑΤΟΙ to gird someone as a soldier. As n.m. girding, breastplate, protective armor.
- 2ωωκε (2ωκε, 2ωωκ, 2ωκ) 2εεκε- (2εκε-, 2εκ-) 2ΟΚ΄ (2ΟΟΚ΄)
 Q 2ΟΟΚε vb. tr. (1) to scrape, scratch, esp. as means
 of torture (ΜΜΟ΄); (2) to shave (ΜΜΟ΄); as n.m. baldness, shaven condition. 2ωωκε n.m. fleece.
- 2ωωλε (2ωλε) 2λλ- 20λ (200λ) vb. tr. to pluck.
- zωωμε (zωμε, zωμ) Q zam (zam) vb. intr. to become lean, thin; + εβολ: to pine away, be blighted.
- 2ωg 26g- 20g" Q 2Hg vb. tr. to distress, afflict (ΜΗΟ', 6); intr. to be distressed (by, with: 6TB6, 2λ, ΜΝ, ΝΤΝ); as n.m. distress, straits. 2λgc n.f. constraint.
- 2004 Т (208 Т, 204 Т) 264 Т 204 Т (208 Т) vb. tr. to steal (МНО ; from: NTN, 21, 2N, 6BOX 2N); as n.m. theft. p64-204 Т thief.
- (2047) 2647- 2047' \pm 680 λ vb. tr. to eject, send forth.
- 202 202 vb. tr. to scrape, scratch (MMo'); vb. intr. to be scraped; to itch; as n.m. itching, scratching.
- 2ω2Ψ, 2ω2Ε n.f. hand (as a measure).
- 2ωx (20x) Q 2Hx vb. intr. to be in straits, be dying; vb.
 tr. to distress, put in straits (ΜΜΟ*, 6); as n.m.

straits. 2xx n.m. illness; name of a disease.
2mx\overline{\tau} 26x\overline{\tau} - 20x\overline{\tau} vb. tr. to shut (\overline{\tau}mo'), shut in, enclose;
as n.m. shutting, sealing.

2ω6Ε (2ω6Ψ, 2ωκΨ) 266Ε- (266Ψ-) 206Ε Q 206Ε (206Ψ) vb. tr. to wither, destroy (ΨΜο΄); vb. intr. to wither away, fade, expire. 2λ6Ε- in cpds.: weak in, feeble of. As n.m. feebleness. λτ2ω6Ε unfading.

2xonxπ (xonxπ, 2nosng) vb. intr. to feel, grope (for: ε, ε2ογη ε).

26: 2H, 626 2 A: 2 O SYLOASE: bolse 26886: 2886 2 A A K : 2 A K 2 AAM: 2 WWM E 2 X P W . PO 2 EB ETWPE: 2MNTWPE 2 A A T : 2 A T Sybal: sboo 26BC: 2HBC 26BCW: 2WBC STAS : STAG 2 A PODHT: 2 A ODHT 22766: 5766 2 A C 6 1 6 : 2 A C 1 6 26801: 21801 2 A B *: 2 W B 2ACM: 2OCH 2 6 B W W N : 26-2 A B 6: 2 O B 6 2ACT-/*: 21C6 266: 26 2AT": 21T6, 21 2661T: 2A61T 2 A B 10 Y 1: 2 1 B W 1 $2\lambda B\lambda GG\lambda G$: $2\lambda 4\lambda GG\lambda G$ $2\lambda T\overline{B}-$: $2\omega T\overline{B}$ 261A6186: 2161B **2 λ Β**Ολ: **Β**ωλ 2ATBGC: 2WTB 2616B: 2161B 2 A 6 B-: 2 W 6 B 2 X T 6: 2 X T 26161T: 2161T 948 :H1948 26K-: 2WK, 2WWK, 2 ATHY: THY 2 A 6 6 Y: 2 A 6 SATN: TWP6 2 WWK€ 3 X 6 OY: 2 X 6 2ATOOT': TOP6 26K6-: 200K6 2 X 6 Y (6) : 2 X 6 ZATF: 2WTF 2626126: 2206126 2 AH: 2 A 6 2 ATP 6 6 Y (6): 2 WTP 2 6 3 KOY: 2 X KOY 2 X 1 B 6 : 2 1 6 1 B 2 ATPEC: 2 WTP 26AM62G: 2AAMH26 2 A I H : 2 A 6 2ATC: 2OTC 26266: 2706 2 λ K Η λ Ψ : 2 λ K λ q 2AOIT: 2AOHT 26M6: 2HM6 2 A KO: 21K 5 7 00 C: 5 00 0 **геметоре: 2МИтюре** 2 A A -: 2 W W A 6 2 AX: 2 0X 2 6MX: 2MOX THKES : STEELS 2 XXN, 2 XXW": XW" 26NH(H)TE: 26NEETE 2AAHK: 2AAAK 2 ENT': 2 ONK 2BA1: 21B6 2 A A KOY: 2 TKOY 2BB6: 21B6 26N616: 21N6 SAYYOAC: SYYOAC 2BBC: 2WBC 26NOY96: 26-2276-: 5706 2BHHTE: CZBHHTE 260YT: 200YT SAM: SOOME 2BHY6: 2WB 26PBOODE: 2PBOT 2 AMNTOP: 2MNTOP6 2BOK: 2WBK 2676: 2P6 PXXKS : 2AKX9 2800C, 280C: 2WBC 26P07-: 2POg 2 ANSHT: 2HT 2BOY1: 204, 21B01 sebae: sboa OF : OTHES 2806: 208K 26P21P6: 21P nos :nks 2BC: 2HBC, 2WBC 26T-: 21T6 2λπC: 2ωBC 2 BCOOY6: 2 WBC 2 6 YT: 200 YT sybeione: eloue 2 6XX-: 20X2X 2 BCW: 2 WBC Sybad: Ybades 280: 204 266M-: 206B SPAS: STABES 2BWC, 2BWWC: 2WBC 2HB(B)6: 2BB6 2 A P N : PO 2800N: 26-2HBC: 2WBC

246: 26	2100": 21	εολμμε: εφολ
2H61T: 2A61T	21272 20	20YP=: 20YP &-
2 H H B G: 2 H B G	2 1 X N , 2 1 X W " : X W "	20YPATE: 20YPIT
2ннпе, 2ннте: elc	2KAGIT, 2KOGIT: 2KO	20YP@(@) : 20YPG-
2H1B6: 2HB6	2266176: 5706176	20YC: 2WC
2HK: 2WWK	2 X H 6 6: 2 X O 6	20YT-: 200YT
2HK6: 2K0	2 AOM: 2 AWM	20 9*: wy
2HM6: 2061M	2 X O Y W : O Y W	204: 2WB
2 HMC: 2 MC	2766: 2706	20x: 2wx
2HMX: 2MOX	2M-: 2AM, 2WM	20xx: 20x2x
2HM: 2MOM	2MAIC: 2MOOC	206B: W6B
2HN6T6: 26N66T6	2MGY, 2MHY: 2AH	१ ०६४: १७६ <u>६</u>
2HNT: 2WNT	2MM6: 2HM6, 2MOM	2 no 6 ne, 2 nox nx:
гноу: гну	2Mтшр: 2MNтшре	$\overline{\pi}$ xn oxs
2HP21PG: 21P	2M2AA, 2M26A: 2AA	2 NOT: 2 NOT
2HT = 2H	ZMX: ZMOX	2 PA": 20, 2 POOY
2HT6: 61C	YANS :YAANS	2PAK: 2POK
2HY: 26	2 NT : 2 I N 6	2P6: 2PA1
2HYB6: 2BB6	20: 2 \	2 P 6 0 Y 6 : 2 P 6
2HX: 2AX	20B: 204	2 PHY6: 2 P 6
21-: 21076	20BG: 21BG	zpHgg€: zpOg
21286, 2126186:	208 T: 2 W 4 T	2 p K p i k 6: p K p i k 6
21618	206126: 2061P6	2 POYB (B) A1: 2 POOY
218, 2186: 21618	20186C: 2A6186C	2 POΥΜΠ6: 2 POOY
2 BOX: BWX	20K": 200K, 200K6	ερογο, ερογω:
2 BOY1: 2 BW1	20KF: 2KO	27007
216686: 21618	20X 2 2 W W X 6	2 POYWP": 2BOPBP
21667(6): 201	2026: 2206	2 Pg -: 2 POg
216Y: 216	2 OM 6 T : 2 OM NT	2TH": 2HT
2161286, 216186:	20MT: 20MNT	2T00P6: 2T0
21618	2OMOTOP: 2MNTOPE	2TW: 2TO
21H: 21G, 26, 2H	soux: soux	2 T W P G : 2 T O
2 I H B 6: 2 I 6 I B	200K*: 200K6, 200K	2 TOUP: 2 TO
21HOY: 216	200K6: 200K6	2007: 2200P
21HY: 216, 21H	200% 200%	200: 20 20: 20
2127: 77	2007: 200	2 0. 200 =
2 IME: 2 HME, C2 IME	ουν. 20η 1 κς , μος 1 πος	2008: 209
2 IMH: 206 IM	201C: 2017	2087: 2047
21N16: 21N6	20πτ [*] : 2ωτπ	200K: 200K
2 IOM6: C2 IM6,	20P64: 2WPB	ξωκ. ξωωκ ξωκ(ε): ξωωκε
2 1 WM 6	20PK: 2POK, 2WPK	20KM: 206B
\$ 100Y6: \$1H		
2172176: 217	20 P 9 : 2 P 0 9 , W P 5 20 P 7 : 2 W P 8	გთგ ნ: გთთგ ნ, გთგ გთგ ნ: გთხ <u>ნ</u>
21pN: po	506x(2): 506g	2ωλδ. 2ωγδ 2ωλ δ: 2ωλ Κ
• •	20C6: 21C6	
2 l poyze: poyze		2 WW : 2 WWW 6
2 1 pω": po	20CHM: 20CM	¿wme: ¿wwme,
2 1T: 2161T 2 1T: 21, 210Y6	20YAT6: 20Y61T,	3 MW 1 3
	20YHT	ZWNT: ZWNK
2 I TN: TwpG	20Y6-: 20Y0	2WNT: 2ONT
2 I TOOT": TWP6	20Yelte: 20Yelt	2017: 20TII
2 ΙΤΟΥΝ-: ΤΟΥΦ"	20Υ6ΡΟΥωρ*: 2BΟΡΒ <u>Ρ</u>	20PIT: 20YPIT
ειτογω": τογω "	SOLILE: SOLEIL	zwp4: zwpB

 2WT: 2WT6, 2OT, 2OT6
 2WW4: 2WW
 2WGE: W6E

 2WT6: 2OT, 2OT6
 2W2E: 2W2T
 2W6T: 2W6E

 2WWA6: 2WA
 2W4: 2WB
 24W: 2O4

 2WWC: 2WC

x

XAAX6 vb. tr. to clap (hands: MMO', 2N).

жавів, жаів, жав n.m. desert. 21/2й пжавів in the desert.
маі-жавів loving solitude. мйтжавів desolation. кш/†/
віре ймо° й жавів to make desert. ——жавів to become
desert, waste.

xx610 (TXX610) xx610° vb. tr. to display (MHO°).

xak vb. tr. to clap (hands: ммо°); to flap (wings); as n.m. clapping, flapping; речхак one who claps.

жын n.f. calm.

xane, xaane, xoone, xann n. ark, box.

xaces n. in x1-xaces to repair, put in order.

xate (xaate), Q xote vb. intr. to become ripe, mature; to advance in age. xtal, Q xHt idem.

жатме n. heap (of grain).

хатче, хатве n.m. snake, reptile.

xa4, xas n.m. frost.

xλ2x2 (xλ2x7, xλx27, 6λ2x27, 6λ262, 6λ267, 6026) x62xω2* vb.

tr. to beat, strike, gnash (ΜΜο*; against: 6xN); as n.m.

beating, gnashing; as adj. beaten, (of metal) refined.

xλx n.m. sparrow. xλx N λ1λ name of a bird.

xaxe (xaaxe), Q xaxω(ογ) vb. intr. to become rough, hard, harsh. atxaxe not harsh (of voice).

xaxc (pl. xixeey, xixeeye, xixeoy, xinxeeye, xin-xeye) n.m.f. enemy. Mai-xaxe loving enmity, quarrelsome; MNTXAXE enmity (toward: e2OYN ε). F-xaxe (Q o N) to be at enmity (with: ε, MN).

 $x\overline{k}B\overline{c}$, $x\overline{k}Rec$, xebec, xebec, $x\overline{k}c$, $xhhb\overline{c}$, $xhb\overline{c}$, xhibec, xhel- Bec n.f. coal, charcoal.

xmın n.m. blemish. אדג without blemish.

xe, Nxe conj. see 30.11 for full discussion of uses.

жевна, жвна, жівна, жевеа n.m. spear; a shoot.

xek n.m. shell, sherd.

EGRAC, EGRANC conj. so that, in order that; usu. followed by Future III or II. See 27.4.

x6Kx1K n. an insect (ant?).

 $x_{6\lambda 2HC}$, x_{72HC} , x_{72C} , $x_{6\lambda\lambda HC}$ vb. intr. to become exhausted, to pant; as n.m. exhaustion, panting.

xemnez, xmnez, xmnhz, xenhz, xhnez, xinez n.m. apple. xemenwp n.f. roof.

xepo (xepω) xepe- (xeepe-) xepo' (xepω') vb. tr. to kindle, set afire; intr. to be ablaze, burn.

жн n.m. speck, mote (of straw, chaff, sawdust).

жн n.f. dish, bowl.

xunc n.f. bowl, censer.

хнр vb. intr. to be merry, enjoy oneself; to be wanton; as n.m. merriment, fun; wanton behavior. речхнр wanton. херхр n.m. wanton behavior.

жыре, жеере n.f. threshing-floor; threshing season. xi n.m. a metal vessel.

x: (xe:) x:- (xe-) x:r* Q xHy vb. tr. (1) to seize, take (MMO*); to receive, accept; (2) to buy, acquire; (3) to strike, reach (of arrows, teeth, etc.); (4) to learn by heart. In basic meaning (1) all prep. and adv. occur with normal meanings. x: e to affect, relate to, impinge on; (± e20YN) to lead to, be conducive to, introduce to. x: MMO* exN to borrow (suff. on exN is reflex.). x: MN to touch, be in contact with. For x:- and xi-in vb. and nom. cpds. see 2nd element.

xieipe n.m. pod.

XIAAGC, XIAAHC, XGAAHC, KGAAHC n.m. box.

xin, $x\overline{n}$, $x \in N$, $k\overline{n}$, $6\overline{n}$, $g \in N$ prep. from, since, starting from; conj. since (see § 30.3); while yet (+ Circum.). $x \in N$ in \overline{N} , $x \in N$ in $x \in N$ in $x \in N$ in $x \in N$ from $x \in N$ to $x \in N$ in $x \in N$ from $x \in N$ in $x \in N$ so $x \in N$ from $x \in N$ in $x \in N$ so $x \in N$ is

occasionally preceded by 6, \overline{N} , 2λ , 21.

no purpose, for no reason. N xINXH idem.

x10Y6 vb. tr. to steal (MMo"; from: $2\overline{N}$, 680 λ $2\overline{N}$); to rob (6, \overline{N} c λ); as n.m. theft, fraud. \overline{N} x10Y6 adv. stealthily secretly; unbeknownst (to: 6). 41 \overline{M} Mo" \overline{N} x10Y6 to steal. MA \overline{N} x10Y6 secret place. PGYX10Y6, CA \overline{N} x10Y6 thief.

xip n.m. brine; salted fish. ANXIP brine-lotion (as soap).

wice xecT- xxcT" (xicT") Q xoce (± e2pxi) vb. tr. to raise up, exalt (ммо"; over, above: e, exn, 2ixn); vb. intr. to become exalted, raised up; as n.m. height(s), top. net xoce the Most High (of God). xice n 2HT to become arrogant, proud, vain; xxci-2HT proud, arrogant; p-xxci-2HT to become vain, proud; ммтххсi-2HT pride, arrogance. xoce n.m. exalted person or place.

xice n.f. back, spine.

xice, xece, xuce n.f. a land measure.

xiq adj. sparing, niggard.

x12 n.m. spittle.

xixwi, 6ixwi n. single lock or braid of hair.

XXWM, XXWB, XXW4 n.m. brazier.

אגאיא n.pl. testicles.

xπ, xen, xin, xe conj. or. xm mmon/mme or not. xm mmωp or rather.

אוא (צפוא, אופ) אופ- אפוא vb. tr. to quench, put out (אומס"); intr. to be quenched. אדאה unquenchable.

xnx vb. tr. to send, send away.

xna* (xno*, xnaa*) vb. tr. to strike (with: \overline{N} or zero).

אואץ (אגאאץ) vb. intr. to delay (in doing: e); as n.m. sloth. אדאאץ without delay; אאדאדאאץ promptness; פּיִּלאַאאן sluggard; אאדן sloth, delay.

XNA2 (pl. xnay2) n.m. forearm, wing; force, violence. N

XNA2 with effort. †-xna2 na* to treat violently. x1

MMO* N xna2 to force, compel. x1 N OYXNA2 to use force;

MNTX1 N xna2 force, violence; peqx1-xna2 violent; MNTpeq-

- $x_1-x_{N\lambda_2}$ violence. Moyp \overline{N} xN λ_2 n.f. scapular (of monk). xNG, xNH, xHNH n.m. beets, greens.
- xnooy, xnaay (pl. xnooye) n.m. threshing-floor, grain on threshing-floor. γxnooy, γεxnooy, λεxnooy, γιχnooy n.f. idem.
- xNoy xNe- (xN-) xNoy" (xNoy", xeNoyoy") vb. tr. to ask, question (dir. obj. of person asked; the thing asked is indicated by ε or ετεε); (rarely) to tell. As n.m. inquiry, questioning.
- xnou, xenou, xenou n.m. basket, container.
- xñxwn' vb. tr. to ask about.
- xo xe- xo' Q xhy vb. tr. to sow, plant (seed: mmo'; in: 2N, 21xN); to plant (a field; mmo', e; with: mmo'); as n.m. sowing, planting. peqxo sower.
- xo xe- (x1-) xo* (usually + eBOλ) vb. tr. (1) to spend, expend, dispose of, use up (MMO*); (2) to put forth, send forth (MMO*; to, onto: e, exN, e2OγN e). xe-NOγNe eBOλ to take root.
- xo (pl. xwoy) n.m. arm-pit; o N xo to be hunch-backed.
- xoe, xoie, xoei, xoi, xo (pl. exh) n.f. wall. $xe-\overline{N}$ -тмнте, xeneтмнтe n.f. middle wall.
- xoeic, xoic (abbrev. xc; pl. xicooye, xicooy) n.m.f. lord, lady; with def art. the Lord; master, owner. φ-xoeic to become lord, rule (over: ε, εχπ, εγρι εχπ); ρεσφχοεic ruler. Μπτχοείc lordship.
- xoeit, xaeit (xit-) n.m. olive-tree, olives; n.m.f. testicle. BW N xoeit olive-tree. MA N xoeit olive grove.

 ge N xoeit olive wood. 6E-N-xoeit olive-leaf. πτοογ
 N xoeit the Mt. of Olives.
- xoı, xoeı (pl. exey) n.m. ship, boat.
- xok, xxk n.m. hair.
- xοκxκ, xεκxωκ (xεκxοκτ) vb. tr. to stamp, brand, mark (ΜΜο΄); as n.m. stamp, brand.
- xox Q to be least, smallest.
- xολxx (xολxελ) xxxx- xxxωλ vb. tr./intr. to drip, let drip.

- x_{OAXX} x_{GAXX} $(x_{XXX}$ -) x_{XX} x_{XX} x
- xoolec n.f. moth. F-xoolec to become moth-eaten, decayed.
- xooy (xoy, xxy) x6y- (xooy-, xxy-) xooy" (xoy") vb. tr. to send (\overline{M} Mo"; to: 6, 6pxt", 6x \overline{N} , Nx", gx) \pm 680% out, off, away; 620YN in; 62px1 up; 2x0H ahead. xooy \overline{N} Cx to send after.
- xooүт adj. base, lowly, rejected. митхооүт, митречхооүт baseness. F-хооүт to become base, lowly.

xooya n.m. papyrus.

xon n.m. bowl, dish.

xopxF sepsor' Q xepxor vb. tr. to overcome; Q to be hard.

xογωτ (xογτ-, xxγτ-, xωτ-, xοτ-; f. xογωτε, xογογωτε) number: twenty. See 30.7.

- xoy4 (xoy8, xnoy4, xo4) xe4- Q xн4 (xн8) vb. tr. to burn, scorch (ммо*); intr. to be sharp, bitter; as n.m. burning, ardor. xoy4 N 2HT n.m. warmth of heart, esp. in 2N оухоу4 N 2HT warmly, sincerely, ardently.
- xογ4 (xω4) xos Q xh4 vb. intr. to be costly, rare; tr. to value.

xoyee vb. intr. to limp.

xoyxoy, 60y60y vb. intr. to fly (or sim., of birds).

жочты in ы жочты headlong, over the edge.

- xoqxq (xosxg, xoqxeq) xeqxwq Q xeqxwq vb. tr. to burn, cook; intr. idem.
- xπ-, xen- n.m.f. hour; usually prefixed to number, as in xπ-нทτογε the llth hour. $\overline{\mathbf{H}}$ nnxy $\overline{\mathbf{N}}$ x $\overline{\mathbf{n}}$ -X at about the Xth hour.
- צחו-, צחם- vb. must; usually prefixed to Inf., as in אורד. משנה שורד. מורד. מ
- xпіо xпіє- xпіо' Q xпінт vb. tr. to blame, scold, reproach (ммо'; for: етве, єхм, гх, гм); as n.m. blame, reproach. ммтхпінт modesty.
- xno xno-xno' vb. tr. (1) to beget, give birth to $(\overline{M}MO')$; (2) to acquire, get, obtain $(\overline{M}MO')$, oft. + eth. dat. w.

- NA*. As n.m. birth, begetting; acquisition, gain, possession. ATERIO4 unbegotten. PE4ENO maker, begetter; MNTPE4ENO begetting.
- xpo (6po) Q xpxeit (xpoeit, 6poeit) vb. intr. to become strong, firm, victorious (over: e, exn); vb. tr. to make strong; as n.m. strength, victory. +-xpo nx' to encourage, confirm. chn-xpo to establish victory. xt-xpo unconquerable. hal-xpo victory-loving. peqxpo victor, victorious. xoop Q to be strong, bold, hard. xxp-bax bold of sight, staring; hntxxp-bax staring. xxp-2ht firm of heart, bold; hntxxp-2ht courage, boldness; +-hntxxp-2ht to give courage (to: nx'); x1-hnt-xxp-2ht to take courage. xwwpe, xwp adj. strong, bold (bef. or aft. n. with n). F-xwwpe to become strong. hntxwwpe strength, prowess.
- хто (gro) хте- хто (gro, grx) Q хтнү (grнү) vb. tr. to lay down (ммо; on: e, exm, гм, гим); intr. to lie down. хто e ngwne to succumb to sickness.

xw n.m. cup.

- xw* n.m. head (§28.6). Rare except in prep. phrases or as the obj. in certain verbal expressions. exN exw* prep.

 (1) on, upon, over, above; (2) for, on account of; (3) at, against; (4) to, unto; (5) in addition to. eBOX exN out upon; e2OYN exN unto; e2PAI exN up/down onto, upon.

 2AXN 2AXW* prep. before, in front of. 21XN 21XW* prep.

 (1) on, upon, over; (2) in, at, beside; (3) ± eBOX from on, from at; ner 21XN the one in command of; 2PAI 21XN on, upon.
- xω xe- xo' vb. tr. to sing; as n.m. song. ρεчxω (pl. ρεчxooye) singer, minstrel.
- xw xe- (x1-) xoo' (imptv. xx1-, xx1') vb. tr. to say, speak (Μπο'; to: e, κλ'; about, concerning: e, eτεe, exπ, e2γλι exπ; against: πcλ, ογεe). λτxw, λτxοο' ineffable. γc4- xe- one who says; κπτρεθχε- saying, telling. xeγo- (for xw eγo') to mean, signify; to say to. nexe-, nexx'

- said (before direct quotation; see 20.3).
- xok xek- xok' Q xнк (± eвох) vb. tr. to finish, complete, fulfill, accomplish (ммо'); vb. intr. to become finished, completed, fulfilled, ended; as n.m. completion, end; total; fulfillment. xrxok without end.
- хюкм хекм- хокм (хакм) Q хокм vb. tr. to wet, wash (мно; in, with: 2N, евох 2N); as n.m. washing, cleansing. +-хюкм их to bathe, baptize. хи-хюкм to be bathed, baptized. атхюкм unwashed; ммтатхюкм being unwashed.
- XWKF X6KF- XOKP' Q XOKF vb. tr. to salt, season.
- xwxk vb. tr./intr. to sink, submerge.
- xωλκ xελκ- xολκ Q xολκ (± εεολ) vb. tr. to extend, stretch κων to: ε, ε2ογν ε); to sew together. xωλκ εεολ as n.m. stretching, strain; extent; endurance, continuation. xλλκ n.m. strain; punishment. xολκ n.f. strain, tension.
- xωλΜ (xωρΝ) Q xολΝ (xορΝ) vb. intr. (1) to make merry; (2) to become implicated, involved (in, with: MN, 2N); as n.m. (1) festivity, dissipation; (2) care, distraction. xολΝ(ε)c, xορΝ(ε)c n. care, distraction.
- $x\omega\lambda\overline{z}$ $x\overline{\lambda}z$ $xo\lambdaz$ vb. tr. to cut, prune.
- xwxz (xwxx2, xwpx2) x6x62- x0x2" (± 680x) vb. tr. to draw, scoop (MMo"). x0x26c, x0x2c, x0x(6)c n.f. vessel for pouring.
- xwm n.m. generation. XIN xwm ga xwm, gyxwm N Nxwm, N 2en-xwm N xwm from generation to generation. gaxe Nxwm genealogy.
- XWNT XNT- (XENT-) XONT' Q XONT Vb. tr. (1) to try, test (MMO', 6; With: 2N); (2) to begin, start; as n.m. trial; MA N XWNT place of testing. XONTO n.f. trial, test.

 XNIT in XI-XNIT to test, try (MMO', NCA); as n.m. test, trial; peqxi-XNIT tester.
- בּשׁאַ (שְּשִׁאַ Q בְּאַסְאָפּיִּ vb. impersonal: to happen, befall by chance; personal: to happen to be; vb. tr. to meet with (6) by chance; as n.m. chance.
- xwp Q xmp vb. tr. to blacken.

- xwp xoop* vb. tr. to study, examine. As n.m. spy, scout.
- xwp xep- xop' Q xmp vb. tr. to sharpen; as n.m. sharpness.
- жшрм Q жорм vb. intr. to make a sign (to: є, оувє; with: ммо', гм), to beckon; vb. tr. to indicate (ммо'); as n.m. sign, indication.
- хшрм Q хорм vb. tr. to urge on, hasten (мно'); intr. to ride fast, hasten (after: мсл). на м хшрм training stable. речхшрм rider.
- xωρπ vb. intr. to stumble, trip. xpon n.m. obstacle, impediment; xrxpon unimpeded; F-xpon to become an obstacle, difficulty; +-xpon to trip up (κx*), cause difficulty for: x1-xpon to stumble, trip, be impeded.
- xwc Q xHc vb. tr. to load, pack (MMo"; with: MMo"); intr. to become hard, solid.
- xωτε (xωτ) xετ- xοτ' (± ε2ογη) vb. tr. to pierce, penetrate (Μπο'; to, as far as: ε, ωλ, 2η); as n.m. penetration, separation.
- xwT2 Q xoT2 vb. intr. to fail, cease.
- хоове (хооче, хоче) хееве- хоове vb. tr. to reach, pass, surpass (ммое); атхоове impassable.
- xωωκε (xωκε, xογογκε) xεεκε- (xεκ-) xοοκ* vb. tr. to sting, prick, goad (Μπο*). xοοκει n.m. goad.
- xwwxe Q xooxe vb. intr. to be hindered.
- xwwλe (xwλe) xeeλe- (xeλe-) xooλ' (xoλ') vb. tr. to gather, harvest (ΜΜο'); as n.m. harvest. γεαχωωλε harvester. xλλε n. gleanings, left-over crops.
- жимие, жиме n.m. book, document, book-roll, sheet of parchment; as adj. book- (with parts or types of books); жимие N им reading book.
- xωωρε (xωρε) xeepe- (xepe-, 6ep-) xoop* (xop*) Q xoope vb. tr. to scatter, disperse (ΜΜο*); + 680% idem; to hinder, bring to naught (ΜΜο*); as n.m. scattering, dissolution.
- xww66 (xw66) x66- x06' (xox', xxκ') Q xH6 vb. tr. to dye, stain (ΜΝΟ'; with: 2N, 680λ 2N); intr. to become dyed, stained; as n.m. dyeing; p64x66- dyer of. xH66, xH66,

xeke n.m. purple dye; as adj. purple; elen-xhee purple embroidery; ca \overline{N} xhee seller of purple.

 $x\omega_2$ (xo_2) xe_2 - Q xH_2 vb. tr. to touch (ϵ , e_2oyn ϵ); as n.m. touching, contagion. $\lambda T x\omega_2$ $\epsilon Po'$ untouchable.

 $x\omega_2 x\varepsilon_2 - x\lambda_2' = Q xH_2 vb.$ tr. to smear, anoint ($\overline{M}Mo^2$, ε ; with: $\overline{M}Mo^2$, $2\overline{N}$).

xω2 M x62 M xλ2 M Q xλ2 M vb. tr. to defile, pollute (MMO'); to become defiled, polluted (with, by: 2N, 680λ 2N); as n.m. pollution, uncleanness. λτχω2 M undefiled. p64-xω2 M defiled person.

xwx, xnxwx n.m. head, chief. $g\overline{N}$ -xwx headache. \overline{p} -xwx to become head, chief.

X66X6-: XWWX6 XHHBC: XBBC XYYNE: XYNE жеере: жире XHIBEC: XEEC XXXTE: XXTE XYYXE: XYXE xeepe-: xepo, xwwpe xhke: xww6e XXB: XX4 XEK-: XWOKE XHNH: XNE XX6: XX616 X6K6: XH66 XHUES: XEWUES XACIBEC: XBBC xere-: xwore XHC6: XIC6 XXEIT: XOEIT **x**€λ**€x**−: 6ωλx XHT: XXT6 xx1-: x1 XEXXHC: XEX2HC XHY: XI, XO XXIE: XXEIE XEXXHC: XIXXEC PYOX : PHX XXK": X0066 XEN: XIN XH6: XWW66 XXK: XOK XGNA(*): XNA XH66: XWW66 XXKM": XOKM XENGTMHTE: XOE x1-: x1, x0, xw **አንዘዘ: 6ን**ጆዛዘ XCHOB, XCHO4: XHO4 жівна: жевна XYNH: XYNE XENOYOY": XNOY XIN: XN XXP-: XPO XENTMHTE: XOE XINOY": XNOY XXCI-: XIC6 XERH2: XEMRE2 XINOY": XNOY XXCT": XIC6 xepe-: xwwpg XINTHY: THY XATEE: XATEE xepe-: xepo XINXEEA(E): XYXE XAY(-): X00Y xepo-: xw XINXEYE: XXXE XAYT": XOYOT xepw(*): xepo XINXIN: 6N6N xx2 : xw2 XEPXF: XHP X100P: 6100P MS () MS 4X xece: xice XING2: XEMNE2 XX2X: XX2XZ XECT-: XICE XIPO4: PO XXXW(OY): XXX6 XIT": XI, XOEIT xer-: xwre XXXZ: XX2XZ X6Y-: X00Y XICOOY(6): XOEIC жена: жевна XICT": XICE X64-: X0Y4 XEC: XEBC **አ**ፅዓወን: ወን XIXEEY(E). X6: XN, 66 XIXEOY: XXXE xezxwz : xxzxz xe-: xo, xw, x1 xex-: 6woxe **Χ**Σλε: **Χ**ωωλε x_{GBBGC} , x_{GBBC} : x_{BBC} x_{GG} : x_{GGG} XX2HC: XEX2HC жевех: жевнх XH: XINXH XX2C: X6X2HC 20086-: X.0086 XHB: XOY9 ж**мж**м: 6 й 6 й XCCKET: XWWKE XHBC: XBBC YOUX , NIX YNOY

XNAAY: XNAY, XNOOY XOONE: XOONE Tron: xopT XOONE: XANE XPAGIT: XPO XNAY2: XNA2 XNG(-): XNA, XNOY PHOX : PANOOX XPOSIT: XPO XC: XOEIC XOOP': XOUPE XNH: XN6 XNIT: XONT xoop': xwp XTAI: XATE XNO': XNA' XOOP: XPO XT6-: XTO PYOX : PYONX XOOYE: XW XTHY: XTO XOUXI: 5XOUXI XOKE: XOOKE **∡**₩**x**₩: 6₩6₩ xo': xo, xw XOP': XOUPE XULE: XDULE X01X: 601X xo: xoe MKOX: HYOX XOE': XOY4 XOPHEC: XWAH XOM6: XOOME X00Y-: X0 XOBXE: XO4X4 XOCE: XICE X061: X01, X06 XOT": XOTE xwp(6): xwwpe, xpo XOT-: XOYOT XUPA: XUAZ XOI, XOIE: XOE X0X*: X00X6 XOT6: XATE HAWK: HAWK XOLEC: XWAZ XOY, XOY': XOOY XOT-: XOYOT XOAMEC: XWAN PYOX: XOY4 XOT-: XOTE XONC: XWNZ XOYOYKE: XOOKE XOTZ: 60TZ XOX26C, XOX2C: XOX2 XOYOYOT6: XOYOT XOOP: XPO XOXX', XOXX: 6WXX XOYT-: XOYOT xoupe: xpo XONTC: XONT XOX': XOUGE SAWOX: SPOOK X008': X0086 X06': X0066 PYOX : POX xne-: xno, xni-XOOK": XOOKE XOTE: XOOBE XOOKEY: XOOKE X116-: X110 xw6€: xww6€ XOOX': XWWXE XILIAX: XILIO X20C: 620C

6

618618, 614614, K14K14, 6186H8 n. chick-pea. 61112T, K1112T n.f. pot.

6116, 6114 (pl. 61166γ, 61166γε, 6116γε) adj. lame, crippled; Mπτ6116 lameness; F-6116 (Q o N) to become lame. 6111τε n.f. name of vessel or measure.

6220YB12 n.m. bald-headed person.

6AM n. bull. (Doubtful.)

61MOYA, KAMOYA (f. 61MAYA6, KAMOOYA6, KAMHA6; pl. 61MAYA6, 61MOYA6, KAMOOYA6) n.m.f. camel, camel-load. MAN-61-MOYA camelherd. MAC N 61MAYA6 baby camel.

61N12 n. or adj. maimed; F-61N12 (Q o \overline{N}) to become maimed. 610YON, 61YON, 61YOYON, K1YON n.m.f. slave, servant. \overline{N} T-610YON service, servitude. \overline{F} -610YON (Q o \overline{N}) to become a slave.

6AOYON, 6AYON n.m. a beverage.

sameixe, samixe, samixe, kamixe, samixoy n.m.f. a dry

measure.

eaneen (sensil) vb. intr. to be hurried, anxious.

6APATE n. carob pod.

6xx6 n.m. earring.

6ax14, 6ax18, kax14 n.m. ant. \overline{p} -6ax14 to suffer from itch or warts.

GAXMH, GAXME, XAMH n.f. fist, handful. GAXMEC n.f. idem.

6A6:TON(6) n.m.f. coarse linen, tow; coarse linen garment.

6 бве, Q 600 в (600 ч) vb. intr. to become feeble, timid; as n.m. weakness. 6 в в - 2 нт weak, feeble; м м т 6 в в - 2 нт weak-ness, timidity; ¬ - 6 в в - 2 нт (Q о м) to become feeble. 6 о в adj. weak, feeble; м м т 6 ов weakness, folly; ¬ - 6 ов (Q о м) to become weak; 6 г р 6 м м о г м 6 ов to make weak.

6801, 6806 n.m. arm (of person); leg (of animal).

66, xe postpositive particle (1) then, therefore, for;

(2) with neg.: no more, not again. TENOY 66 now then, and now, now moreover.

66AMAI, 6 $\overline{\Lambda}$ MAI, 66AMA, K6AMA, 6AAMA, 66AMHN N.M. jar, Vase. 66A \overline{Z} , 6 $\overline{\Lambda}$ Z, 6AAZ n.m. shoulder.

GENNHYT Q to be hard, stiff.

бепн, біпн vb. intr. to hurry, hasten, come quickly; may be used reflex. w. ммог. As adv. quickly, in haste; usu. in phrase 2N оубепн. речбепн one who is hasty, quick; мхтречбепн hastiness.

беров, бероч (pl. бероов, бероов) n.m. staff, rod. +- беров to beat (NA*, 6). $\oplus \overline{C}-\overline{N}$ -беров a blow.

6ние n.f. cloud.

GIE, GIGIE, GIH n.m. he-goat.

61N-, KIN-, $6\overline{N}$ - prefix added to any inf. to form an abstract noun (f.) of action or manner of action.

GING 6N- (6GN-, GIN-) 6NT' (6GNT', KNT', GHNT', ΓΝ') vb.

tr. to find (ΜΜΟ'). 6NTC to find that (+ Circum. or

xe); also: perhaps, suppose that. GING ΜΜΟ' Ναλ to find

someone (Ναλ) guilty of (ΜΜΟ'). 6N-2HT to learn wisdom.

GING as n.m. finding, thing found. peqGING finder.

61NMOYT, 61MMOYT, 61NMOT, $\kappa \overline{\text{NMOYT}}$, $6\overline{\text{MMOYT}}$ n.f. the Pleiades. 61NOYHA, 66NOYHN, 61NOYBAA, κ 1NBHA n.m. kind of ship. 61N2OYT, 61M2OYT, $6\overline{\text{M2OYT}}$, $6\overline{\text{M6AW}}$, 61N6AW, 61N6AW, 61N6AW, 61N6AW, $6\overline{\text{M66AW}}$, $6\overline{\text{M66AW}}$, $6\overline{\text{M6AW}}$,

61N6WP n.m. talent (weight).

61TPE n. kind of fruit, lemon.

61x n.f. hand; script-hand; hand as measure. 61x N ογναμ right hand. 2λ τ(*)61x under one's control. F-NO6 N 61x to become generous. +-61x to promise (someone: νλ*). 6X, 6λλ n.m. a weapon (exact meaning not clear).

6AA, 6AO, KAA in †-6AA to sway, stagger.

6A1A n.m. burnt-offering.

6Am, 6eam, 6Aam, Keam n. dry sticks, twigs.

6xo n.m. vanity, futility.

6AOMAM 6AMAOM- (6AEMAOM-) 6AMAOM' (6AEMAOM') Q 6AMAOM (6AM-AOMT, AMAOM, AMAHM) vb. intr. to become twisted (up with, up in: 6, 2N); to become implicated, involved, complicated; also tr. to embrace. As n.m. complication.

6λ006ε, 6λ06ε, κλοΓε, τλ006ε, τλω6ε n.f. ladder.

6x06, Tx06 n.m. bed, bier.

6x06, 66x06 n.m. gourd.

6λω, 6λογ n.f. twigs, firewood.

- sawt (pl. or dual: 6λοοτε, 6λοτε, 6λοο6ε) n.m.f. kidney;
 pl. also = internal organs in general, viscera.
- GNON, Q GHN (GON) vb. intr. to become soft, smooth, weak; as n.m. softness. †-GNON to weaken. GON, GOONG, GON adj. soft; also of a condition of wine.
- бмбм (бембем, бмбм, хмхм, химхи, хмхм) vb. intr. to make music (vocal or instrumental); as n.m. music.
- 60ειλε (60ιλε) 6λλε- Q 6λλωογ (κλλωογ, 6λληγ, 6λληγτ, 6λλοογτ) vb. intr. to dwell, sojourn, reside (at, in: ε);
 μλ Ν 60ειλε dwelling-place, inn. ρΜΝ60ειλε sojourner,
 lodger; ρ-ρΜΝ60ειλε (Q ο Ν) to become a sojourner. (2)
 (additional forms: 6λλω, 6λλωω, Q κελοιτ) to deposit
 (ΜΜο, with: ε), entrust to. 60ειλε n.m. sojourn,

residence: furnishings; deposit.

60λ n.m. (1) a lie; (2) a liar. F-60λ to lie, be false;
61P6 ΜΜΟ΄ Ν 60λ to make false, present or take as false.
x1-60λ to tell a lie; xτx1-60λ sincere; μητατχ1-60λ sincerity; p64χ1-60λ liar; μητρ64χ1-60λ lying.

60x86 n.f. woolen garment.

60 λ x" 6 vb. reflex. to abstain from. 60 λ \overline{\overline{X}} n.m. abstinence. 60 λ 6\overline{X} (60 λ 6\overline{A}) 6\overline{A}6\overline{A}') Q 6\overline{A}6\overline{A} vb. tr.

to spread to dry (MMo'); as n.m. spreading to dry.

- 60M n.f. power, strength, might, authority. Δτ60M powerless; MNTΔΤ60M powerlessness, inability; P-ΔΤ60M (Q o
 N) to become powerless; PMN60M mighty man. κΔ-60M 680λ
 to lose strength, be exhausted. P-60M, 61P6 N 0γ60M to
 do wondrous deeds. 61P6 N τ(*)60M to do one's utmost.

 --60M to give power (to: NΔ*). 0γN-60M MM0* (one) has
 the strength, power, ability (to do: 6, 6TP6); 0γN-960M
 NM0* idem; (one) is able (to do: 6, 6TP6); (M)MN-(9)60M
 MM0* neg. of preceding. 6M-60M, 6N-60M to find strength,
 to be able (to do: 6); to prevail (over: 6, 6XN, 62PΔ1
 6XN, 2N, 21XN). Φ6M-60M idem.
- 60м6м (60м66м) 6м6юм vb. tr. to touch, grope for (e); as n.m. sense of touch. אדה without untouchable.
- GON n.m. low place, hollow. GOONE n.f. idem.
- GONG n. violence, might, force, usu. only in cpd. xi N GONG to use violence, act violently; to harm, hurt, illtreat, constrain (MMO'); as n.m. violence, iniquity; MNTXI N GONG idem; P-XI N GONG to act violently; PG4XI N GONG Violent, harmful; MNTPG4XI N GONG Violence.
 GOOAGC n.f. thigh.
- 600YNE, 6AYNE n.f. hair-cloth, sack cloth; as measure: a sack. ca \overline{N} 600YNE sack-seller. ca2($\overline{\tau}$)-600YNE weaver of sacks.

600γρε, ελγρε a term of contempt; slave (?).
60π, κλπ n.f. sole of foot, foot.

60n n.f. a cutting instrument.

- 60m6, 60m, 61me, 61me, 61me, n. small vessel, small amount; 60m6 60m6 little by little.
- борте, корте, бърте n.f. knife, sword. этборте without a knife; uncut.
- 60P4" (KOP4") vb. tr. to nip off.
- 60 $p\overline{x}$, 60 $p\overline{x}$ 6 n.m. filth. \overline{p} -60 $p\overline{x}$ (Q o \overline{N}) to become filthy.
- soc, кос n.m. half. oysoc (added to a quantity) and a half. sic-, sec- cpd. form, as in sicтные half a fingerbreadth.
- 60c™ n.m. darkness, stormy darkness.
- 60c6€ (60c6€c) vb. intr. to dance; as n.m. dancing.
- such. size; age; form, sort. N телоот of this sort, such. му N бот of what sort? F-тоот (Q o N) to become like (N or poss. prefix).
- боуна n.m. kind of locust.
- 60γx, κογx, 6ωx, 60γ6, κογκ π.m. safflower, cardamum.
- $6096\overline{g}$ (6096eg) 6eg6eg— 6eg6wg Q 6eg6wg vb. tr. to sprinkle.
- $60x6\overline{x}$ (60x66x) 66x60x (66760x, 667606) Q $66x60x\overline{t}$ vb. tr.
- to cut, smite, slaughter (MMO°); as n.m. cutting etc.
- брн vb. tr. to dig (Ммо°).
- брипе n.f. diadem, sceptre.
- брибе n.f. dowry.
- dove. spmngan n.f. turtledove.
- броомпє, громпє n.m. name of a vessel and measure.
- sperm; progeny. Δτορος without seed, without progeny. χι-6ρος to be impregnated.
- 6 pw2, kpw2, 6 pww2, kpo2 n.m. need, want, lack. \overline{p} -6 pw2 to be in want (of: \overline{N}); as n.m. need.
- 600 Q 666T (6HHT) vb. intr. (1) to remain, wait (for: 6, NA*; with: $M\overline{N}$; in, within: $2\overline{N}$); (2) to continue, persist (in doing: Circum.); (3) to cease, stop, cease functioning.
- 600 $6\overline{\lambda}$ (661) 601 (6001, KO1) Q K61 Vb. tr. to collect, gather. p646 $\overline{\lambda}$ -96 wood-gatherer.
- $6\omega\lambda$ (K $\omega\lambda$) $6\overline{\lambda}$ $60\lambda'$ ($600\lambda'$) Q $6H\lambda$ vb. tr. to roll up (like

- a scroll: MMO'); intr. to roll up, back; to curl up.
 600 680 to turn back, return (tr. or intr.).
- 6ωλπ (κωλπ) 66λπ- (6⊼π-) 60λπ' (κολπ') Q 60λπ (κολπ) ± 680λ vb. tr. to uncover, reveal (ΜΜΟ'; to: 6, Ναλ); vb. intr. to become revealed, uncovered, manifest; as n.m. revelation, uncovering; λτ6ωλπ covered. 6λλπ- in cpd. one who uncovers.
- 60λΣ (xωλΣ) 6λ6- (66λΣ-, x6λ6x-) 60λχ' (60λ6', x0λχ') Q
 60λΣ (x0λΣ) vb. tr. to entangle, ensnare (ΜΜο'; in, with:
 6, ΜΜο'); reflex. and intr. to become entangled, entwined;
 to adhere, be swathed (in: ΜΜο', 2Ν); as n.m. entanglement.
- 60м, 60м, ком (pl. 600м, каам) n.m. garden, vineyard, property. 6м6, 6мн (pl. 6мну, 6мноу, 6м66у) n.m. gardener, vinedresser. атбые untilled; ыйтбые vinedressing.
- 60NA6, 60YNA6, 60NB, 60YNA66C, KAYNAK6C n.m. cloak.
- GONT, Q GONT Vb. intr. to become angry, furious, raging
 (at, against: 6, exn); as n.m. wrath, anger, fury. ΔΤGONT incapable of anger; μπτωτοωντ ability to control
 one's anger. ρεσιωντ wrathful, quick-tempered person;
 μπτρεσιωντ quick-temperedness. †-6ωντ to provoke to
 anger (νω*); ρεστ-6ωντ one who provokes to anger; μπτρεστ-6ωντ provoking to anger. 6νωτ vb. intr. to become
 angry; as n.m. anger; ρεσίονωτ given to anger; †-6νωτ to
 provoke to anger; ρεστ-6νωτ provoking to anger.
- 660NT (6606) 6NG- (66NT-) vb. tr. to wring, nip off. 66NGN (660NGN) 6NGN- idem.
- 6ωογ 6οογ' Q 6ηγ (6ηογ) vb. tr. to make narrow; intr. to become narrow, crowded; as n.m. narrowness.
- 6ωογ 6εγ- vb. tr. to push; + εκοχ: to put (a ship: Μπο') to sea, to set sail, push off.
- 6ωογ6, Q 6οογ6 vb. tr. to twist, make crooked (ΜΜΟ*); intr. to become crooked, twisted. 2Ν ογεωογ6 crookedly.
- быпе (кыпе, дыпе, бып, кып) беп- (бп-, дып-, бып-, бып-, бып-, бып-, кып', кып', кып', кып', бып') Q бып (кып, беп, кеп) vb. tr. (1) to seize, take (ммо'); to

- take up, begin (from: xin, $2\overline{N}$); (2) to have a claim against (6); Q to be guilty (of: 6), liable for, responsible for; (3) to entrap (in, by: $2\overline{N}$), inculpate. $60\overline{NC}$ n. capture.
- εωρΣ, κωρΣ, εωρλ2 n.m. night.
- сωρδ, Q сорδ vb. tr. to hunt (ε), lie in ambush for; as n. m. snare. ма й сωρδ hunting place; речсωρδ hunter. сорб n.f. snare, ambush; prey. серно (pl. серьсе) n.m. hunter.
- swpe sope Q sope (soλe) vb. tr. to prepare, provide (ΜΜο΄). as n.m. preparation. peqswpe preparer.
- σωρδ σορδ Q σορδ vb. tr. to populate, people, inhabit (ΜΜο΄); intr. to be inhabited, peopled.
- 6ωτ n.f. drinking trough.
- 6ωτπ σετπ- 6οτπ Q σοτπ vb. tr. to overcome, defeat (ΜΜΟ); intr. to become defeated, overcome, wearied, discouraged. as n.m. intimidation, discouragement. 6ωτπ 6ΒΟλ to frighten away. 6ωτπ Ν 2ΗΤ to be afraid; as n.m. fear. λτ6ωτπ unconquered, undefeated; ΜΝΤλΤ6ωτπ invincibility. ρε46ωτπ, ρε46ετπ-ρωμε εΒΟλ kidnapper. 60της n.f. defeat.
- $6ωτ\overline{z}$ ($xωτ\overline{z}$, $κωτ\overline{z}$, $6ωz\overline{\tau}$) 6οτz Q $6οτ\overline{z}$ vb. tr. to pierce, wound ($\overline{M}MOz$); as n.m. hole. \overline{p} - $6ωτ\overline{z}$ $6ωτ\overline{z}$ Q O \overline{N}) to become all holes. 6λτz6 n. hole.
- 600086, 6086 (6 \overline{B} -) n.f. leaf. $6\overline{B}$ in cpds. e.g. $6\overline{B}$ -xoeit olive-leaf. At60086 leafless. x_1 -60086 to glean grapes. 6000A6, KOA6 n.m. flat cake, loaf.
- 6ωωλε (6ωλε) 6ε6λε- (6ελε-, 6\over 1.00 000λε (6λλε) Q 600λε vb.
 tr. to swathe, clothe, cover (6; with: \(\over \ndeta \ndeta
- 6ωωμε 6εεμε- (6εμε-) Q 600με (6λλμε) vb. tr. to twist, pervert (μμο'); intr. to be twisted, crooked; as n.m. perversion; μπτεοομε crookedness; μπτρεμεοομε perversion.
- 6000x6 (600x6) xex- 60x2° Q 600x6 (± 680λ) vb. tr. to cut, cut off, hew ($\overline{M}MO$ °). MA \overline{N} xex- $\omega N6$ quarry.
- 6ωωπ, Q 6οωπ vb. intr. to look, glance, gaze (at: ε, ε2ογη

 ϵ , $\epsilon \times \overline{N}$, $\overline{N}c\lambda$, $2HT^{\epsilon}$); to pay heed (to: ϵ); to look forward (to: ϵ); as n.m. look, glance. $\epsilon \omega g \overline{\tau}$ $\epsilon B \circ \lambda$ idem; as n.m. idem. MA \overline{N} $\epsilon \omega g \overline{\tau}$ a look-out.

66Y-: 6WOY 6A: T6A610 6 \ 2 C 6 : 6 2 O C 6AB-: 6BB6 $6\lambda_2x\overline{2}$: $x\lambda_2x\overline{2}$ 6 ex-: 6 wwx.e 6226, 62262: X22X2 66x=: 606 6AAME: 6WWM6 62266: 6W6 GAXE: 6WXE 6HHT: 60 6A616, 6A61H: T6A610 6A6*: 6W6 6HH: 6NON 6HNT": 61N6 6A610: T6A610 6266: 6WG **6λλ*: 6ωωλ6** 6B-: 6wwBe 6HOY: 6WOY 6 A A : 6 A 6B06: 6B01 бип: бфпе 6AA6-: 6061AG 6 Eggl, 6 Egg G: ggl **6ние: кние** 6AX66Y(6): 6AX6 66: K6 6HY: 6WOY GHX: GWG, GWWXE 6AXEYE: GAXE 666X6-: 600X6 6 A A H : 6 A A 6 GEEMET: GWWME 61616: 616 6AAHY(T): 6061A6 666T: 6W 61H: 61E 6AXIX: KAXKIX 661A661A: KAAKIA 6 IMMOYT: 6 INMOYT GAAMA: GEAMAI 66x6-: 600x6 61M2OYT: 61N2OYT 6ALOOYT: 6061A6 GEAGIA: KAAKIA 61N-: 61NG **6λλΟΠΟΥ: ΚλλωΠΟΥ** 66M6-: 600M6 61NTXW: 61N6XW 627<u>1</u>-: 607<u>1</u> 66M2OYT: 61N2OYT 61 NH: 66 NH 6AAW", 6AAWW": 61C-: 60C 66N-: 61N6 606116 6 ENOYHA: 6 I NOYHA 61XWI: XIXWI 6AAWOY: GOGIAG 66NT': 6INE 616XW: 61N6XW банауаб: баноуа 66N6XW: 61N6XW 6x-: 6∞ωλ€ 6An-, 6An': 6wne 6en-, 6en: 6wne 6 A A 2 : 6 \overline{\lambda}2 GARE, GARH, GAREI: бепн: кнпе 617 M: 61 M 66P-: XWWP6 6 x 0: 6 x x eno9 6ARIXE, 6ARIXOY: 66n6<u>11</u>: 62n6en 6X00T6: 6XWT 9X191X6 GEPAGE: 6WPG 6X0066: 6XWT 62рт6: 60рт6 66PH6: 6WP6 **6λΟΤ6: 6λωΤ** GATZE: 6WTZ беромпе: броомпе **6λΟΥ: 6λΦ** 67ANE: GOOANE бероов: бершв 6x066: 6x0066 6AYON: 6AOYON берффв: берфв 6λχ-: Κωλχ 6AYOYON: 6AOYON 6ep6wp": xopxp 6ME: 6WM 6AYP6: 600YP6 66C-: 60C 6MGEY: GWM 6A46A4: 6A86A8 66T6WX*, 66T6W6*: 60X6X

6MH: 6WM 600ME: 600ME 6POGIT: XPO 600NE: 6NON, 60N брйпфан: броомп6 6MHY: 6WM **бймоүт: біммоүт** 600Y": 6WOY 6p006: 6p06 6M2OYT: 61N2OYT 600Y6: 600Y6 6 P W W 2 : 6 P W 2 6M6M: 6N6N 6004: 6BB6 6pww6: 6p06 6N-: 61N-, 61NE 600x6: 600x6 6PW6: 6PO6 6N: XIN 60066: 6w6 6COYP: KCOYP **6ΝΔΤ: 6ωΝΤ** 60n: 60n6 6WB: 6BB€ 6NT': 61N6 60n-/*: 6wne 6WB6: 6WWB6 eonc: emue 6N66AO, 6N66Aw: 6016: 60016 618630 60 pxe: 60 px 6026620: 61N6620 60p6C: 6wp6 6N6N-: 6wn6 6wn: 6non 60λ6C: 6ωωλ6 **60ΤΝΖ: 6ωΤΠ** 60NE: 60NA6 60λΧ: ΚωλΧ 607N76 (ec): emn76 6wn6N: 6wn6 6wn-/*, 6wn: 6wn6 6016 : 601 X 60Y6: 60YX, 6w6 60λ**6:** 6ωρ**6** GOYGOY: XOYXOY 6 m p w : K m p w 60M: 6WM 602C: 620C **60P4**: **K0P4** 600B: 6BB6 6026: XX2X2 602 T: 60T ₹ 600X*: 6wwxe 60xq: 6wxB 60X: 60YX 600x*: 6wx 60x2 *: 6wwxe 6wxe: 6wwxe 600λ6: 6φωλ6 6∏-: 6wn€ 6 w x 4: 6 w x B 600X6C: 600X6 6 P A : T P A 606: 60NE

Addenda

61 6802 $2\overline{N}$ to survive (an ordeal), come through successfully. (NO2N $\overline{2}$) N62NOY2 Q N62NOY2 vb. tr. to shake, shake down. now \overline{N} 6 to assist.

cx20 in moy-w-cx20 lukewarm water.

22610, Acto part. yea, verily; also of entreaty.

6po: xpo

86P6-: 8 wwp6

AAAM: Awwm6

PAKT6: PIK6

CANAWT: CAANW

TOY6IH: TOOY

WOOP: WAAP6

600M: 6WM

ZABA6HEIN: ABA6HEIN

2 PEBOT: EBOT

Glossary of Greek Words

(Greek verbs are cited in their normal dictionary form: 1st pers. sing. indicative active or middle.)

άγαθόν n. what is good. άγαθός good. άγαπή f. love. άγγεῖον n. name of a vessel. αγγελος m. angel. άγορά f. agora, forum. άήρ m. air, atmosphere. άθετέω to disregard. αίθριον n. atrium, courtyard. αίσθητήριον n. sense-organ. αίτέω to ask, ask for. αίχμάλωτος m. prisoner. αίών m. period of time, age: eternity: world. άκαθαρσία f. uncleanness. άμάθαρτος unclean. άκατάληπτος incomprehensible. άκτίς, -ῖνος f. ray, beam. άλλά but, but rather. άληθῶς truly. άμήν amen; truly, verily. άνάγμη f. necessity. άναστροφή f. turning; life(-time). αύξάνω to grow up.

and live in desert as a hermit.

άναχωρέω to retire, withdraw, go άφελής simple.

άναχωρητής m. anchorite. άνομία f. lawlessness. άνοχή f. a holding back. άπαντάω to meet, confront. άπαρχή f. first-fruits. απιστος unbelieving.

άπογραφή f. registration. άπογράφω to register. άποθήκη f. storehouse, barn. άπόστολος m. apostle. άποταμτικός m. anchorite, hermitmonk. άποτάσσω to renounce, give up. apa (introduces question). άρετή f. goodness, virtue. ἄρχω to begin. άρχή f. beginning. άρχιεπίσμοπος m. archbishop. άρχιερεύς m. high priest. αργων m. ruler: Archon. άσεβής impious. άσθενής weak, without strength. άσκός m. leather bag; wine-skin. άσπάζομαι to greet. άσπασμός m. greeting. άσώματος incorporeal. άτοπος odd, strange.

άπλους simple, sincere.

βαλλάντιον n. purse. βαπτίζω to baptize. βάπτισμα n. baptism. Báoavos f. torture, anguish. βάσις f. course. βημα n. platform, judgement seat. βίος m. life.

βλάπτω to harm, injure.

βοήθεια f. help, aid, support.

γάρ for, since, because.

YEVEA f. generation.

γένος n. race.

γραμματεύς m. secretary, scribe.

γραφή f. writing, scripture.

δαιμόνιον n. evil spirit; demoniac.

δαίμων m. evil spirit.

δέ but, however.

δεμανοί m.pl. the decans.

δήμιος m. executioner.

διάβολος m. the Devil.

διαθήμη f. will, testament,

covenant.

διακονέω to wait on, serve.

διστάζω to hesitate.

δίκαιος just.

δικαιοσύνη f. justice.

δικαίωμα n. justice, ordinance.

δόγμα n. decree.

δοκιμάζω to prove, test.

δυνάστης m. ruler.

δῶρον n. gift.

ἔαρ n. springtime.

έβδομάς f. week.

€Ovos n. nation, people.

εί μή τι if not, unless 30.10

είδος n. kind, sort.

είκών f. likeness.

είρήνη f. peace.

είτε ... είτε either (whether)

... or.

έκκλησία f. church.

ἔλος n. marsh.

έλπίζω to hope for.

έλπίς f. hope.

ένεργία f. function, action.

ένοχλέω to trouble, disturb.

έντολή f. command. commandment.

έξομολογέω to confess,

acknowledge.

έξουσία f. power, authority.

έπεί since, because.

έπειδή since, because.

έπειδήπεο inasmuch as.

έπιβουλή f. plot.

έπιθυμέω to desire, be eager

(for).

έπίσμοπος m. bishop.

έπιστολή f. letter, epistle.

έπιτιμάω to rebuke.

έρῆμος f. desert, wilderness.

έτάζω to examine, test.

ἔτι still, yet.

εύαγγέλιον n. gospel.

εύχαριστέω to give thanks.

ñ or.

ήγεμονία f. rule.

ἡγεμών m. governor.

ήδονή f. pleasure, delight.

ήλικία f. age, time of life.

ήμερος mild, tame.

ήσυχάζω to be still, quiet.

θάλασσα f. sea. θεωρέω to observe, look at. θλίβω to afflict, distress. θρόνος m. throne.

θυσία f. offering, sacrifice.

θυσιαστήσιον n. altar.

ίδιώτης m. layman, uninformed person.

καθαρός pure.

καθηγέομαι to instruct. καθολικός universal, catholic.

καί γάρ for surely.

καίτοι and yet, although, albeit.

κακία f. evil. badness.

καλῶς well.

นฉึง (even) if.

καπνός m. smoke.

καρπός m. fruit.

κατά in accordance with; see 30.10, μέχρι even up to, even including.

καταλαλέω to slander.

ματαλαλία f. slander.

κελεύω to order, bid, command.

κέραμος m. tile.

κεραστής f. horned-(viper).

κηρύσσω to announce, proclaim.

κινδυνεύω to be in danger.

κλάσμα n. piece.

κλήρος m. portion, inheritance.

κοινωνός m. partner.

κόλασις f. punishment, correction.

κοσμικός worldly, secular.

κόσμος m. world.

κούφον n. (empty) vessel.

κοάτιστος most excellent.

κρίνω to judge.

κρύσταλλος m. ice.

utious f. world, creation.

μυριακή f. Sunday.

λαός m. people.

λύπη f. grief.

μαθητής m. pupil, disciple.

μακαρίζω to bless, deem blessed.

μακάριος blessed.

μάλιστα especially.

μέν...δέ see 30.10.

μερίς f. portion, share.

μέρος n. part, member.

ueditno m. mediator, intercessor.

μετάνοια f. repentance; obeisance.

μετανοέω to repent.

μετέχω to partake (of: ε).

μή (introduces question; 30.10).

μήποτε so that not (+ Coni.).

 $\mu\eta\pi\omega\varsigma$ so that not (+ Conj.).

 $u\acute{n}\tau\iota = u\acute{n}$.

μόγις with difficulty, hardly,

scarcely.

μοναχός in. monk.

μόνον only, alone; but (w. neg.).

μορφή f. form, shape.

μυστήριον n. mystery.

νηστεία f. fasting.

νηστεύω to fast.

νοέω to think.

νομοδιδάσκαλος m. teacher of the law. νόμος m. law. νοῦς m. mind.

οίκονόμος m. steward, manager. οίκουμένη f. world. δλοκόττινος m. gold coin. δλοσηρικός silken. δυοίως adv. likewise. ομολογία f. confession. ὄργανον n. instrument. όργή f. wrath. όρεινή f. hill-country. όρφανός m. orphan. όσον as long as (+ Circum.), while. πλήν except; but, however. όταν when, whenever, if (+ Cond.), πνεύμα n. spirit. OUN therefore. ούδέ and not, nor. OÜTE ... OÜTE neither ... nor. όψώνιον n. wages.

πάθος n. suffering.
πανούργος m. villain.
παντοκράτωρ m. the Almighty.
πάντως wholly, completely.
παραβολή f. parable.
παραγγέλλω to order, command.
παράγω to pass by, away.
παράδεισος m. Paradise, Eden.
παρακαλέω to exhort.
παράνομος lawless, unjust.
παρθένος f. virgin.
παρρησία f. freedom, openness.

πάσχα n. Passover.

πατριά f. family, clan, nation. $\pi \epsilon i \vartheta \omega$ to persuade. πειράζω to tempt, experience. πειρασμός m. temptation. περιεργάζομαι to be overly concerned. περίχωρος f. surrounding countryside. πίναξ m. writing-tablet. πιστεύω to believe. πίστις f. faith, trust. πιστός faithful, true. πλανάω to err. πλάνη f. error, erring. πλάσσω to form, mould. πνευματικά n. spiritual matters. πόλις f. city. πονηρός bad, wicked. πόρνη f. prostitute. ποτήριον n. wine-cup, cup of wine. πρεσβύτερος m. elder. προάστειον n. suburbs, environs. προκόπτω to progress, advance. πρός in accordance with. προσευχή f. prayer. προφητεύω to prophesy. προφήτης m. prophet. πύλη f. gate. πως how? why?

σάββατον n. sabbath. σαΐτιον n. keg. σάρξ f. flesh. σεμνός holy, august. σίκερα n. strong drink. σμάνδαλον n. impediment: bad behavior.

σμεπάζω to cover, shelter. σκηνή f. tent, "tabernacle".

σοφία f. wisdom.

σπέρμα n. seed, offspring.

σπήλαιον n. cave.

σταυρός m. the Cross.

στήθος n. chest. breast.

στιγμή f. moment.

στρατιά f. army.

συγγενής m. kinsman.

συγκλητικός of noble rank.

σύμβολον n. mark, token.

συμβουλεύω to advise, give counsel, χαλινός m. bridle.

σύμβουλος m. counsellor.

συναγωγή f. synagogue.

σχημα n. garb; monk's habit.

σῶμα n. body.

Jωτήρ m. savior, redeemer.

ταλαίπωρος wretched, miserable.

τάξις f. order, rank, post.

τάφος m. tomb.

άχα quickly.

έλειος perfect, complete.

ελώνης m. tax-collector.

ελώνιον n. tax-house.

ετράρχης m. tetrarch, petty prince.

Εχνίτης m. craftsman.

ιμή f. price, value.

5τε then, thereupon.

ράπεζα f. table.

τροφή f. food, nourishment.

ὔλη f. woods, forest.

ὑμνέω to sing hymns.

ὑπηρέτης m. custodian.

ὑπομένω to be patient under, submit to.

ὑπομονή f. patience, endurance.

ωαρισαῖοι m. the Pharisees.

φθόνος m. ill-will, jealousy.

φορέω to wear.

φυλή f. tribe, people, nation.

σύσει by nature, naturally.

φύσις f. nature.

χαῖρε Greetings!

χαλάω to lower, let down.

xápic f. grace.

χήρα f. widow.

χιών f. snow.

xopóc m. chorus, choir.

xpdoual to use.

χρεία f. need, necessity.

χρήμα n. goods, money.

χρηστός useful, beneficial.

χριστός m. the Christ.

χώρα f. land, country.

ψάλλω to recite the psalter.

ψαλμός m. psalm.

ψυχή f. soul.

ω (vocative particle).

ώς (see 30.10).

ώστε (see 30.10).

ώφελία f. advantage, profit.

Bibliography

The field of Coptic studies has never been a particularly neglected area, and with the resurgence of interest brought about by recent Manichaean and Gnostic finds, the bibliography of the field has expanded to enormous proportions. We shall restrict ourselves here to mentioning a few essential bibliographical, grammatical, and lexical works with which the student who wishes to continue his studies should become familiar.

A. Bibliographical Works

- Kammerer, W. A Coptic Bibliography. Ann Arbor, 1950.
- Mallon, A. *Grammaire copte*. 4th ed. revised by M. Malinine; Beirut: Imprimerie catholique, 1956. Contains a valuable bibliography pp. 254-398.
- Scholer, D. M. Nag Hammadi Bibliography 1948-1969. Leiden: E. J. Brill, 1971. This is updated annually in Novum Testamentum.
- Simon, J. "Contribution à la bibliographie copte des années 1940-45," Bulletin de la Société d'archéologie copte (Cairo) 11 (1945), 187-200.
- . "Bibliographie copte," appears regularly in Orientalia from 1949 onward.
 - B. Grammatical Works (including Dialect Studies)
- Jernstedt, P. "Die koptische Praesens und die Anknüpfungsarten des näheren Objekts," Doklady Akademii Nauk S. S. R. 1927, pp. 69-74.
- Kahle, P. E. Bala'izah. 2 vols.; London: Oxford University Press, 1954.
- Plumley, J. M. An Introductory Coptic Grammar (Sahidic Dialect). London, 1948.
- Polotsky, H. J. Études de syntaxe copte. Cairo: Publications de la Société d'archéologie copte, 1944.
- . "Modes grecs en copte?" Coptic Studies in Honor of

- W. E. Crum. Boston, 1950.
- . Review of W. Till, Koptische Grammatik, in Orientalistische Literaturzeitung 52 (1957), 219-34.
- _____. "The Coptic Conjugation System," Orientalia 29 (1960), 392-422. (These and other articles are reprinted in H. J. Polotsky, Collected Papers. Jerusalem: Magnes Press, 1971.)
- Steindorff, G. Lehrbuch der koptischen Grammatik. Chicago: University of Chicago Press, 1951.
- Stern, L. Koptische Grammatik. Leipzig, 1880.
- Till, W. C. Koptische Grammatik (Saïdischer Dialekt). 2nd ed.; Leipzig: Harrassowitz, 1961.
- . Koptische Dialektgrammatik. 2nd ed.; Munich, 1961.
- Vergote, J. Phonétique historique de l'égyptien: Les consonnes. Louvain: Bureaux du Muséon, 1945.
- . Grammaire copte, Vol. Ia, Ib. Louvain: Edit. Peeters, 1973.
- Wilson, M. R. Coptic Future Tenses: Syntactical Studies in Sahidic. The Hague: Mouton, 1970.
- Worrell, W. H. Coptic Sounds. Ann Arbor: University of Michigan Press, 1934.

C. Dictionaries and Concordances

- Crum, W. E. A Coptic Dictionary. Oxford: Clarendon Press, 1939.
- Spiegelberg, W. Koptisches Handwörterbuch. Heidelberg: C. Winters, 1912.
- Wilmet, M. Concordance du nouveau testament sahidique, II. Les mots autochtones. Corpus scriptorum christianorum orientalium; Subsidia, vol. 11. Louvain, 1957.

Grammatical Index (Coptic)

(All references are to the numbered paragraphs of the Lessons unless "p." is specified.)

- imptv. prefix 17.1 (e continued) w. verbs of motion 7.2 AN neg. of adv. pred. 2.2: 3.1 of nom. pred. 6.1 direct object marker 10.1 indirect object marker 10.2 of Imperfect 21.1 AN- indep. pron. we, procl. 6.1 w. Inflected Inf. 20.1 6- rel. pron. Imperfect 21.1 ANT- indep. pron. I, procl. 6.1 6/6P6- circum. converter 23.1; ANAY imptv. of NAY 17.1 24.2: 25.1 ANI- ANI imptv. of eine 17.1 AMH imptv. of 61 17.1 e/epe- second tense converter AMHELTN imptv. of ϵ 1 17.1 24.2; 25.1 AMOY imptv. of el 17.1 6BOA 21TN as agent 13.4 API- API' imptv. of eipe 17.1 eBOA 2N nominalized 27.2 eie, eeie in apodosis 29.1 APIPE imptv. of eipe 17.1 AT- neg. adj. prefix 27.1 as mark of interrog. 30.11 λY^- for λ -oy 7.1 e-Ne- in contrary-to-fact λyω 8.2; 30.11 protases 29.1 AYWN imptv. of OYWN 17.1 ene as mark of interrog. 29.1 enta- rel. pron. Perf. I 12.1 Am interrog. pron. 6.2; 16.1 (ογ) λφ N 16.1 - ephy reciprocal pron. 28.2 A26PAT 19.2 6po*: € 9.1; 30.6 **λx!** - **λx!** ' imptv. of **xw** 17.1 epw": epN 9.1 ет rel. pron. 3.1; 19.1 BOX in directional adv. 8.1 **€T ™MAY 3.1** in cpd. adv. and prep. 28.7 ет/етере- rel. converter 24.2; 25.1 Aε Gk. δέ 28.2 етвинт": етве 9.1 ете rel. pron. 5.1; 12.2; 19.1; e prep. in comparative 29.3 21.1 w. Inf. 13.3 ете път пе 5.1

 6T6P6- rel. pron. 19.1
 MMO': N 9.1

 6T00T': 6TN 10.4
 MN prep. as conj.

 6TP6- Infl. Inf. w. 6- 20.1
 MN- neg. existent

 6g- = g 26.2
 w. possessive

 6gwne 29.1
 in Bipartite (MN- (g) 60M MMo'

 6xN prep. w. vbs. of motion 7.2
 MNNCA TP6- 20.1

 6xw': 6xN 9.1
 MNT- for 'teens 2

 61AT' (eye) in cpds. 28.6
 MNT- abstract present the street pre

KAN 29.1

-KG- 4.3; 28.5

KGOYA, KGOYGI 28.5

KGT, KGTG 28.5

KOOYG 28.5

λλλγ indef. pron. 16.3 λλλγ Ν 16.3 ΟΥλλλγ as pred. 16.3 Ν λλλγ adv. 16.3

M assimilated form of N

MA- imptv. prefix 26.3

MA, MA- imptv. of † 17.1

MA N in cpds. 23.2

MAT' imptv. of † 17.1

MAYAA(T)' intens. pron. 28.3

Megage neg. of gage 20.2

Me2- ordinal prefix 30.7

MHe1' imptv. of † 17.1

MMIN MMO' intens. pron. 28.3

MMO': N 9.1

MN prep. as conj. 1.4; 30.11

MN- neg. existential pred. 2.2

w. possessive 22.1

in Bipartite C. 24.2; 18.1, 2

MN-(g) 60M MMO' impersonal 20.2

MNNCA TP6- 20.1

MNT- for 'teens 24.3

MNT- abstract prefix 27.2

MNT6-, MNTA' possessive pred. 22.1

MNP- neg. imptv. prefix 17.1

in neg. of Injunctive 30.1

N, Who' direct object marker 10.1 omitted in possessive 22.1 omitted after verb 26.3 partitive 16.5

N... AN negation in Bipartite C. 24.2 in copulative pred. 5.1 in Fut. I 18.2 in impersonal constructions

20.2 in Pres. I 18.1

N after anticipatory pron. suffix 10.4

N (linking) w. adj. 15.1 w. nouns as adj. 23.2 w. numbers 15.3

N, NA prep. indirect object
marker 10.2
ethical dative 30.6
N-, N6- def. art. pl. 1.3
N genitive 2.3

NA": N 10.2

NTOOT': NTN 10.4 NA- abs. rel. pron. 22.2 N2HT": 2N 9.1 NA: dem. pron. 5.2 N61 7.1 NA2PA": NA2PN 9.1 Ne see NA"; Nomitted aft. ESTON 24.1 Ne pron./copula 5.1 o \overline{N} as Q in cpd. vbs. 26.1 NE/Nepe- imperfect converter 24.2; 25.1 -ooye as pl. of Gk. n. 3.2 Nei- dem. adj. 4.2 NENT-, NET- nominalized rel. n- def. art. m.s. 1.3 3.1; 12.3; 20.2 na- abs. rel. pron. 22.2 NH dem. pron. 30.8 na: dem. pron. 5.2 mapa mapo' in comparative 29.3 NI- dem. adj. 30.8 w. pron. suffixes 29.3 NEM interrog. pron. 6.2 nazov in directional adv. 8.1 NIM N 16.1 w. second tenses 14.2 in cpd. adv. and prep. 28.7 + ne + rel. 13.2 ne pron./cop. w. Imperfect 21.1 w. imperfect converter 24.2 NIM each, every 16.2 2 WB NIM 16.2 ne, Te, Ne pron./cop. 5.1 NMMA : MM 9.1 w. converters 25.1 NOY poss. pron. 22.2 + rel, clause 13.2 ne- def. art. 1.3 NCA (except) 30.11 NCABHA XG 29.1 nei- rei- Nei- dem. adj. 4.2 NCφ": NCλ 9.1 ment- nominalized rel. 12.3 Ντλ*: see Ντε necht in directional adv. 8.1 N⊤€ N⊤x genitive in cpd. adv. and prep. 28.7 aft. indef. noun 2.3 ner nominalized rel. 3.1; 20.2; 27.2 aft. n. w. dem. adj. 4.2; 22.1 nexe- nexx* (said) 20.3 aft. n. w. modifier 15.1 пн, ты, мн dem. pron. 30.8 w. oy $\overline{N}/M\overline{N}$ in possessives 22.1 n₁-, r₁-, n₁- dem. adj. 30.8 NTe- indep. pron. you (f.s.), no, ro, noy poss. pron. 22.2 procl. 6.1 now: in directional adv. 8.1 NTETN indep. pron. you (pl.), in cpd. adv. and prep. 28.7 procl. 6.1 NTK- indep. pron. you (m.s.), F- in cpd. verbs 26.1; 27.1 procl. 6.1 F-ANA impers. vb. 20.2

F-2NA reflex. verb 20.2 THE in directional adv. 8.1 in cod. adv. and prep. 28.7 par' (foot) in agepar' 19.2 in cpds. 29.4 τω' poss. pron. 22.2 TWN interrog. adv. 14.2; 24.1 Peq- noun prefix 27.2 em in directional adv. 8.1 $p\overline{M}(\overline{N})$ - noun prefix 27.2 pw' (mouth) in cpds. 28.6 рюме as adj. 23.2 oy- indef. art. 2.1 w. oy (what?) 6.2 ca (side) in cpd. adv. and prep. 28.7 omission of: see Subject Index oy interrog. pron. 6.2 $C\lambda$ \overline{N} (seller of) in cpds. 23.2 -coy for -oy 22.1; 20.4 w. second tenses 14.2 case impers. vb. 20.2 OY N 16.1 $-c\overline{q}$ for -q 22.1; 20.4 OY MN OY 16.1 C21Me as adi. 23.2 oy ne + rel. 13.2 oya as indef. pron. 16.5 T- def. art. f.s. 1.3 OYBH": OYBE 9.1 τλ- abs. rel. pron. 22.2 OYAA(T) f intens. pron. 28.3 TAI dem. pron. 5.2 OYON indef. pron. 16.3 Te pron./copula f.s. 5.1 OYON NIM 16.3 Te- def. art. f.s. 1.3 $oy\overline{N}$ - existential pred. 2.2 τει- dem. adj. 4.2 in Bipartite C. 24.2 TENT-, TET- nominalized rel. w. converters 25.1 3.1; 12.3; 20.2 w. indef. subject Pres. I тн dem. pron. 30.8 18.1; Fut. I 18.2 тнр" (all) 16.4 in possessive pred. 22.1 †- dem. adj. 30.8 ογπ-(g) 60m mmo impersonal 20.2 TM negative OYNTE- OYNTA" pred. of possesof Conditional 29.1 sion 22.1 in Clause Conjugations 30.3 of Infl. Inf. 20.1 w vocative particle 17.2 of Temporal 13.1 TNAY, THNAY interrog. adv. 14.2 o- aux. vb. 26.2 TOOT* in cpds. 10.4; 28.6 ga prep. w. vbs. of motion 7.2 τογω- τογω- (bosom) in cpds. **ΦλΡΟ': Φλ** 9.1

moon Mmo' possession 22.1

29.4

goon NA* possession 22.1
ggs impersonal vb. 20.2

2 Anc impersonal vb. 20.2 2 APO : 2 A 9.1 2 A 2 TH ": 2 A 2 TN 9.1 26N- for 2N- 2.1 2 6N- indef. art. pl. 2.1 2 нт (belly) 29.4 2HT* (front) 29.4 21 as conj. 30.11 2 1 Pw : 2 1 PN 9.1 2 I TOOT": 2 I TN 9.1; 10.4 2 1 w (w) *: 2 1 9.1 2 1xw : 2 1xN 9.1 2M nTP€- 20.1 2 N- for 26N- 2.1 2N oy- adv. phrases 21.3 20Y6 in comparative 29.3 20YN in directional adv. 8.1

in cpd. adv. and prep. 28.7

20yo e in comparative 29.3 2PA (face) in cpds. 29.4 2PA (voice) in cpds. 29.4 2PA1 in directional adv. 8.1 in cpd. adv. and prep. 28.7 2TH* (tip) 29.4 2тн (heart) in cpd. 29.4 $_2\omega(\omega)$ intens. pron. 28.3 xe conj. w. Fut. III 27.4 uses of 30.11 xekaac, xekac conj. w. Fut. III 27.4 x:N- w. Perf. II 30.3 xooc xe 12.5 xw' (head) in cpds. 28.6 XW MMOC XE 12.5

66 postpositive particle 30.11 66 pron. 28.5 61N- noun prefix 27.2

Table of Principal Verbal Conjugations

First Present		Rel. of Pres. I		$Circumstantial^1$	
†	τÑ	6†	€T N	61	6 N
ĸ	тетй	€TK	етет П	ек	етет П
те(р), тр		ете		ep (e)	
4	ce, coy	ет ч	етоү	€q	eγ
C		ετ C		€C	
zero-N		стерс-N		epe−N	
Imperfect		Fut. I		Fut. II	
NEI	NEN	†n2	Д (N) ['] ЙТ	EINA	ения
NEK	нететП	KNA	теҐ <mark>ก</mark> '(ท) ่ม	EKNA	етет <u>и</u> ,(и) т
иере		тена, тера		ерена	
N 6 4	иеү	L NP	CENA	64NA	еуна
NEC		CNY		6CN7	
nepe-N		zero-N NA-		ере-N мж -	
Fut. III		Neg. Fut.	III	Imperf.	of Fut.
6 I E	ene	ผ หม	พิทธท	NEINA	иения
eke	EYETNE	Nnek	йнетй	иеких	метет па
ере		ที่หล		иереня	
646	6Y6	ที่พธฯ	инел	иечия	иеуна
ece		ที่ทธด		ИВСИУ	
epe−N		πne−N		N E P E - 1	V на-
Perfect I		Neg. Perf. I		Perfect II ²	
3.1	λN	ភាព រ	<u> </u>	Ντ λΙ	ИДТЙ
λK	хтет₩	й пєк	йпетй	ЙΤλΚ	йтатетй
λp(e), λ		พีทธ(ๆ), พีท	ιογ	йтъре, йт	ra (p)
PΑ	λY	កា ne ។	Йπογ	PATΠ	NTAY
λC		м пес		ΝΤλC	
λ−N		™ne-N		NTA-N	

¹Second Present = Circumstantial.

 $^{^2}$ Relative of First Perfect = Second Perfect with or without prefixed e-.

Habitual		Negative	Habitual	Injuncti	ve
וגע	ØγΝ	мет	MEN	нарі	МРИ
gyk	φλτετÑ	мек	мететП		
φλρ(ε)		мере			
ወልч	ወኦሃ	ME4	мвү	нареч	мароу
ωyc		Mec		марес	
у ъре−N		мере- N		маре- N	
Conditional		Conjunctive		Fut. Conj. of Res. 1	
€ I @ A N	енфун	Δ τ (<u>Ν</u>)	ЙТЙ		TAPÑ
EKWYN	6.161 <u>и</u> фуи	ÑΓ, ΝΓ	NTETN	тарек	TAPETN
6 L 6 A Y N		П тє		таре	
ечфуи	6үфүи	<u>Р</u> и , РИ	NCE	ТАРЕЧ	ΤΆΡΟΥ
€C ∰ N		ÑС, NC		тарес	
ерфуи-И		NTE-N		търе-N	
Temporal		"Until"		"Not yet"	t
Птері	ที่ระคม	ወ ል N ተ ²	ФУИТИ	ל גח א	ЙТАПМ
птерек	Пте ретП	GYNTK	фунтети	<u></u> ጠልተ፳	МпатетÑ
йтере		ФУИТЕ		йиуте	
Птереч	йтероу	$\overline{\Gamma}$ TN \mathcal{L} Q	YOTNAQ	₽τፈnΉ	Йπλτογ
йтерес		⊕ ANTC		ЙПДТ С	
п тере−N		данте-N		Mnate-N	
Inflected Infinitive					

тра трен трек трететя тре треч треу трес

тре-М

 $^{^{1}}$ May have prefixed $\overline{\mathbf{N}}$ -.

Or wanta.

Subject Index

(All references are to the numbered paragraphs of the lessons unless "p." is specified.)

Achmimic dialect p. ix adiectives 15.1 "all" 16.4 "any" 16.3 attributive 15.1 comparative 29.3 demonstrative 4.2; 30.8 "each, every" 16.1 Greek 15.1 negative compound 27.1 "other" 4.3 predicate 15.2 predicate inflected 29.2 substantivized 15.1 adverbs directional 8.1 with $2\overline{N}$ oy- 21.3 interrogative 14.2 of static location 28.6 agent with passive 13.4 "all, entire" 16.4 alphabet p. x anticipatory suffixed pronoun 10.4 asyndeton 8.2 article definite 1.3 indefinite 2.1 omission (deletion) of 2.2; 4.3; 15.3; 16.2; 18.1; 23.2; 26.1 assimilation p. xvi Bipartite Conjugation 24.2

Bohairic dialect p. viii-ix Causative Infinitive: see Inflected Infinitive causative 30.4 verbs of type TAKO 26.3 Circumstantial 23.1; 24.2 circumstantial: see clause types circumstantial converter w. copulative clauses 25.1 w. existential and possessive predications 25.1 w. Fut. I 25.1 w. Imperfect 24.2 w. Habitual 28.1 w. Perf. I 25.1 w. Pres. I 24.2 clause types formal w. adjectival predicate 15.2; 29.2 w. adverbial predicate 1.4; 2.2 w. existential predicate 2.2 w. nominal predicate (copulative) 5.1; 6.1; 6.2; 15.2 w. possessive predicate 22.1 w. verbal predicate 7.1 functional circumstantial 23.1 relative 3.1; 5.1; 12.1; 12.2; 13.2; 19.1; 21.1 purpose/result 27.4; 30.2

"each, every" 16.2 temporal 13.1; 23.1; 30.3; emphasis 28.2 30.10; 30.11 ethical dative 30.6 Clause Conjugations 30.3; see existential predication: see clause also names of individual types, formal conjugations w. converters 25.1 cleft sentences in Coptic 13.2 in possessive predication 22.1 in English 13.2; 14.1; 28.2 Favvumic dialect p. ix cohortative 30.1 finalis 30.2 First Future 18.2 comparative of adjectives 29.3 compound nouns 27.2 w. converters 25.1 compound adjectives with AT- 27.1 negative 18.2 compound verbs 26.1 relative 19.1 conditional sentences 29.1 First Perfect 7.1 Conditional and its neg. 29.1; w. converters 25.1 30.3 negative 10.3 conjunctions negative relative 12.2 Coptic 1.4; 8.2; 30.3; 30.11 relative 12.1 Greek 30.10 First Present 18.1 Conjunctive and its neg. 25.2; w. indefinite subject 18.1 30.2; 30.3; w. Gk. conjuncnegative 18.1; 19.1 tions 30.10 relative 19.1 converters: see individual names Future Conjunctive of Result 30.2, 3. copula 5.1 Greek copulative: see clause types, adjectives 15.1 formal conjunctions 30.10 copulative clauses w. converters nouns 3.2 25.1 prepositions 29.3; 30.10 dative 10.2 verbs 18.5 ethical 30.6 gender 1.1 demonstrative: see pronouns, genitive 2.3; 15.1; 22.1 adjectives Habitual 28.1 dialects p. viii-ix w. converters 28.1 directional adverbs 8.1 negative 28.1 dummy object with xw 12.5 Imperative 17.1; 30.1

w. ma- 26.3	(Infinitive, types)
Imperfect 21.1	солс х, дторт р 20.4
as bipartite 24.2	miscellaneous 20.4; 12.4
w. converters 24.2	тако 26.3
negative 21.1	Inflected Infinitive 20.1; 30.3
relative 21.1	negative 20.1
imperfect converter	as causative 30.4
w. copulative clauses 25.1	Injunctive 30.1
w. existential and possessive	negative 30.1
predication 25.1	intransitive (verbs) 18.3
w. Future I 25.1	as term 17.4; 18.3
w. Habitual 28.1	qualitative of 22.3
w. Perf. I 25.1	use of transitive verbs 18.3
w. Pres. I 24.2	Jernstedt's Rule 24.2; 26.1
imperfectum futuri 25.1	jussive 30.1
impersonal verbs and expressions	linking (adjectival) \overline{N} 15.1; 23.2;
20.2	27.2
independent pronouns: see pro-	negation: see names of individual
nouns. in apposition for	verbal conjugations and clause
emphasis 28.2	types; see also an, $\overline{N} \dots an$,
preposed 28.2	and $T\overline{M}$ in Gr. Index.
Infinitive	nominalized: see substantivized
in bipartite and tripartite	nominal subjects in verbal clauses
conjugations 24.2	7.1
Causative: see Inflected Inf.	"not yet" Conjugation 30.5
with e of purpose 13.3	nouns
forms of 11.1	as adj. with $\overline{\mathbf{N}}$ 23.2
w. object suffixes 11.2	gender 1.1
types: + 12.2	Greek 3.2
кют 13.5	number 1.2
MOΥP, ΟΥW2 (ΟΥλ2°) 14.3	plural 1.2
місє 17.3	w. pronominal suffixes 10.4; 19.2
кште, моуже, фшшбе 18.4	28.6; 29.4
сштй, оүшгй (оүлгм"),	number 1.2
моуоут, моуг™ 19.3	

numbers	(pronouns, interrogative)
cardinal (1-5) 15.3; (6-10)	as adjectives 16.1
16.5; (11-19) 24.3	"other" 4.3
higher 30.7	personal
w. partitive 16.5	independent 6.1; 28.2
syntax of 15.3	suffixal
fractional 30.7	anticipatory 10.4
ordinal 30.7	forms 9.1; 20.4
object	on Infinitive 11.2
direct 10.1	on nouns 28.6; 29.4
dummy 12.5	on possessive pred. 22.1
indirect 10.2	on prepositions 9.1
Optative 30.1	possessive 22.2
ordinals: see numbers	possessive adjectival 4.1
participium conjunctivum 27.3	relative: see clause types,
participle, proclitic 27.3	relative
passive 13.4	resumptive 12.1; 30.3
periphrastic future 30.9	omission of 30.3
possession	pronunciation p. x-xv
w. genitive 2.3	qualitative 21.2
predication of 22.1	in Bipartite Conjugation 24.2
pronominal 4.1; 10.4; 28.6; 29.4	forms of 21.2
preposing 28.2	of intransitive verbs 22.3
prepositions 1.4	of transitive verbs 21.2
compound 8.1; 28.6	reflexive verbs 19.2
w. pronominal suffixes 9.1	relative clauses
w. verbs of motion 7.2	circumstantial clauses as 23.1
proleptic: see anticipatory	in cleft sentences with ne, re,
pronouns	NG 13.2
absolute relative 22.2	substantivized 3.1; 12.3
demonstrative 5.2; 30.8	see also clauses, relative;
emphatic 28.3	relative converter
indefinite 16.3; 16.5	relative converter
intensive 28.3	w. copulative clauses 25.1
interrogative 14.2; 6.2	w. existential and possessive

(relative converter)

pred. 25.1

w. Fut. I 25.1

w. Imperfect 24.2

w. Habitual 28.1

w. Perf. I 25.1

w. Pres. I 24.2

resumptive pronouns 12.3; 12.1

pl. w. NIM 16.2

Sahidic dialect p. viii

Second Perfect 14.1

negative 24.1

Second Present 24.1

as Bipartite 24.2

negative 24.1

second tense converter

w. copulative sentences 25.1

w. Fut. I 25.1

w. Habitual 28.1

w. Perf. I 25.1

w. Pres. I 24.2

second tenses: see also names of

individual conjugations

w. interrogative pron. and adv.

14.2

as emphasis 28.2

negation of 24.1

sentence: see clause

Sentence Conjugations 30.3

Subachmimic dialect p. ix

substantivized

adjectives 15.1

relative clauses 3.1; 12.3; 20.2

supralinear stroke p. xiv; 18.1

syllabification pp. xiv-xv

Temporal 13.1; 30.3

negative 13.1

Third Future 27.4

negative 27.4

topicalization 28.2

transitive (verbs)

as term 17.4

qualitative of 21.2

Tripartite Conjugation 24.3; see

also names of individual conju-

gations

"until" Conjugation 30.3

verbs: see names of individual

conjugations and principal parts;

transitive; intransitive; reflex-

ive; impersonal; compound

vocative 17.2